## **DD Form 1494 Preparation Guide**

## INTRODUCTION

A principal goal of the DoD's spectrum management program is to develop and efficiently manage the DoD's use of the spectrum during the frequency allocation, allotment, and assignment processes. Achieving this goal minimizes the potential for interference during the fielding and employment of spectrum dependent equipment. The Frequency Allocation-to-Equipment Process supports the DoD's spectrum management goal. Achievement of this goal is critical due to significant increases in the use of the available electromagnetic spectrum due to technological advancements and spectrum being reallocated away from military use, or being opened to sharing with commercial use.

DoD acquisition policy states that the funds for the acquisition, research, development, production, purchase, lease, or use of weapons systems, information management systems, electronic warfare (EW) systems, or other systems that require use of the electromagnetic spectrum will not be released by the obligating authority until an application for frequency allocation has been approved.

In the research, development, production and procurement cycles, adherence to acquisition policy must begin as early as possible. This process includes policies, responsibilities, and procedures for Army commands, agencies, and the Program Executive Officers (PEOs) and Product Managers (PMs) involved in conceptualizing research, development, production, procurement, modification/product improvement, and the leasing of spectrum dependent equipment.

The Frequency Allocation-to-Equipment Process is a spectrum requirement and exists to determine

- (1) if equipment can operate in spectrum bands per the national and international tables of spectrum allocation; and that
- (2) equipment being brought into the Army inventory conforms to applicable spectrum management regulations, directives, standards, and specifications.

For almost 50 years, the Joint Frequency Panel of the United States Military Communications-Electronics Board (MCEB) has reviewed the characteristics of radio frequency (RF) equipment purchased, or developed, by the Department of Defense (DoD). This review process is called the Joint Frequency Allocation-to-Equipment Process, or as it is commonly known, the J-12 Process.

The form currently used to submit systems information to the MCEB for frequency allocation approval is the DD Form 1494, last revised in August 1996. All major military RF systems, as defined in Chapter 10 of the National Telecommunications and Information Administration (NTIA) Manual of Regulations & Procedures for Federal Radio Frequency Management, are reviewed in parallel with the J-12 process by the Interdepartment Radio Advisory Committee (IRAC) Spectrum Planning Subcommittee (SPS).

The Chief Information Officer (CIO) G6 Staff has the responsibility to review, coordinate, and approve applications for frequency allocations submitted by Army activities. Within the Army, that function has been assigned to the Army Spectrum Management Office (ASMO).

The submission of incomplete and inaccurate DD Form 1494 applications consumes unnecessary manpower and causes processing delays. A review of past DD Form 1494 submissions by the PM AIT and the JSC suggests that the information being sought and the correct method of presenting that information is sometimes unclear.

To minimize problems encountered when DD Form 1494 applications are submitted, the PM Automated Information Technology (PM AIT) office requested that the Joint Spectrum Center (JSC) update and prepare an electronic version of the "Preparation Guide" for completing the DD Form 1494.

# **ABOUT THIS GUIDE**

This guidebook is divided into seven sections: *Introduction, General Instructions, Guide to the DD Form 1494 Pages, Sample Completed DD Form 1494s, More About, Points of Contact, and References.* 

The *Introduction* provides background on the Allocation-to-Equipment Process and the DD Form 1494 requirement.

The *General Instructions* section provides detailed instructions and a template for filling out the DD Form 1494.

The Sample Completed DD Form 1494s section contains two examples of completed and reviewed DD Form 1494s. One is for a Radio Frequency Identification (RFID) system and the other is for a Wireless Local Area Network (LAN) application.

The *More About...* section includes comprehensive information about specific aspects of the DD Form 1494 fields.

The remaining sections provide *Points of Contact*, and a list of additional *References*.

## **GENERAL INSTRUCTIONS**

The DD Form 1494 Preparation Guide is an Adobe Acrobat 6.0 PDF document. To take advantage of the features of this document, download, at no cost, Adobe Acrobat Reader (6.0, or greater), available from <u>http://www.adobe.com</u>

The technical specifications of the spectrum dependent equipment must be obtained prior to starting this process. Transmitter, receiver, and antenna characteristics of the proposed system are also required. In some cases, access to a measurement lab will be required to obtain this information. Please note that for Federal Communications Commission (FCC) Part 15 Type Accepted equipment, an FCC test report can provide some of this information. For commercial off-the-shelf (COTS) equipment, manufacturer technical specifications provide a significant source of data to complete the DD Form 1494.

### **GUIDE TO THE DD FORM 1494 PAGES**

The DD Form 1494 is composed of six basic pages and a system line diagram. The main pages are titled:

- DoD General Information
- Transmitter Equipment Characteristics
- Receiver Equipment Characteristics

- Antenna Equipment Characteristics
- NTIA General Information
- Foreign Coordination General Information.

If the proposed system will operate only within the United States & Possessions (US&P), all pages, except the Foreign Coordination page, must be completed. The Foreign Coordination page is **only** required if the planned system will operate outside of the US&P.

On the inside left of the DD Form 1494 Preparation Guide is a set of bookmarks that can be used to navigate through the Guide. The actual document contents are found on the right (see Figure 1).

Begin processing a DoD General Information page by positioning the mouse and clicking on the '+' sign located to the left of the field "DoD General Information." This will expand the field descriptions for the DoD General Information Page as shown in Figure 2.

Image: Second
Image: Control of the con

Figure 1

Figure 2

Model Reader - [DD1494PreparationGuidePrototype5.pdf]   Ele Edt View Document Tools Window Help   Image: State Stat	L□× L□ E × Search Adobe PDF files

Click on each topic for instructions for each field. (See Figure 3.)



Figure 3

Scroll to the bottom of the page to view pages titled *Common Errors* and *More About*.

*Common Errors* describes the most common mistakes made on DD Form 1494s. These mistakes can lead to delays in obtaining equipmente certification.

*More About* provides hot links for a more in depth review of the word or phrase selected. Simply click on the word or phrase to be directed to the additional reference sources.

Blank forms are provided and should be completed using the required data.

**Note:** While the free, downloadable Adobe Acrobat Reader will permit the user to enter data on the DD 1494 Forms and to print the completed forms, it will not permit the user to save that information. To save the data, the user must obtain a copy of the fully-licensed product: Adobe Acrobat 6.0.

To navigate back to the DD Form 1494, simply click the Previous View button on the bottom of the Adobe reader. (See Figure 4.)



Figure 4

APPLICATION FOR EQUIPMEN	т	CLASSIFICATION	DATE	Form OMB	Form Approved OMB No. 0704-0188			
FREQUENCY ALLOCATION				PAGE 1	OF PAGES			
The public reporting burden for this collection of information is estimated to average 24 hours per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing the burden, to Department of Defense, Washington Headquarters Services, Directorate for Information Operations and Reports (0704-0188), 1215 Jefferson Davis Highway, Suite 1204, Arlington, VA 22202-4302. Respondents should be aware that notwithstanding any other provision of law, no person shall be subject to any penalty for failing to comply with a collection of information if it does not display a currently valid OMB control number. PLEASE DO NOT RETURN YOUR FORM TO THE ABOVE ADDRESS. RETURN COMPLETED FORM TO THE USING AGENCY OR CONTRACTING AGENCY, AS APPROPRIATE.								
DOD GENERAL INFORMATION								
то		FROM						
1. APPLICATION TITLE								
2. SYSTEM NOMENCLATURE								
3. STAGE OF ALLOCATION (X one)   a. STAGE 1 - CONCEPTUAL b. STAGE 2 - EXPERIMENTAL   c. STAGE 3 - DEVELOPMENTAL d. STAGE 4 - OPERATIONAL								
4. FREQUENCY REQUIREMENTS a. FREQUENCY(IES) b. EMISSION DESIGNATOR(S)								
5. TARGET STARTING DATE FOR SUBSEQUE	ENT STAC	GES						
a. STAGE 2	b. STAG	BE 3	c. STAGE 4					
6. EXTENT OF USE								
7. GEOGRAPHICAL AREA FOR								
a. STAGE 2								
b. STAGE 3								
c. STAGE 4								
8. NUMBER OF UNITS								
a. STAGE 2	b. STAGE 3		c. STAGE 4	c. STAGE 4				
9. NUMBER OF UNITS OPERATING SIMULTANEOUSLY IN THE SAME ENVIRONMENT								
10. OTHER J/F 12 APPLICATION NUMBER(S)	TO BE	11. IS THERE AN	11. IS THERE ANY OPERATIONAL REQUIREMENT AS DESCRIBED IN					
a. SUPERSEDED J/F 12/		THE INSTRU	CTIONS FOR PARAG	SRAPH 11?				
b RELATED J/F 12/		a. YES	b. NO		c. NAvail			
12. NAMES AND TELEPHONE NUMBERS		(1) COMMERCIAL						
				(1)//01010101				
b. PROJECT ENGINEER		(1) COMMERCIAL		(2) AUTOVON				
13. REMARKS								
DOWNGRADING INSTRUCTIONS		CLASSIFICATIO	N					

## **DOD GENERAL INFORMATION**

# CLASSIFICATION

If the multi-page DD Form 1494 contains classified information, it must be handled in accordance with appropriate agency security directions. For the US Army, the DD Form 1494 classification level shall be in accordance with DoD Directive 5200.1-R and US Army Information Security Program Regulation AR 380-5.

For classified applications, the highest classification for each item, or sub-item, must be indicated by a (U) for UNCLASSIFED, (C) for CONFIDENTIAL, or (S) for SECRET, alongside the item, or sub-item, title. Downgrading instructions must be indicated.

Frequently, individual pages are classified in different ways. For example, a system might have SECRET data on the Transmitter and Receiver Equipment Characteristics pages, CONFIDENTIAL data on the Antenna Equipment Characteristics page, and UNCLASSIFED data on the other pages. In this case, the overall classification is SECRET, and each page is marked with the highest security classification indicated on that page. Each page has a classification block at the top and bottom and the appropriate terms (UNCLASSIFIED, CONFIDENTIAL, or SECRET) should be typed into those blocks.

If a page is classified as CONFIDENTIAL or SECRET, every block on that page shall be marked with its appropriate classification, based on the data entered in that block. This block-by-block marking should conform to the portion marking, as defined in DoD Directive 5200.1-R.

The overall (highest) classification of the document must be conspicuously stamped on the top and bottom of the front page and the top and bottom of the back of the last page. If the front page has a classification that is lower than the overall classification, the classification block at the top and bottom of the front page must reflect the classification of the page (e.g., "This Page is UNCLASSIFIED," or "This Page is CONFIDENTIAL," etc., in addition to the overall classification stamp.

The application title (Item 1 of the DoD, NTIA, and Foreign Coordination General Information pages) should <u>never</u> be classified.

The terms "Competition Sensitive" or "Proprietary Data" are not classifications and, therefore, should not be entered into the classification block. If needed, the words "Competition Sensitive" or "Proprietary Data" should be typed within the top and bottom margins of the form, and/or stamped in the REMARKS block. The J-12 process is designed to handle and protect proprietary data. No proprietary data will be abstracted, or entered into any reports, or data bases, not directly involved with the J-12 process. A designation of "Competition Sensitive" may only be applied when two or more contractors are bidding for a contract, and should be removed within a reasonable time period. After contract award, the data on the successful contractors' DD Form 1494 will not be considered proprietary.

**Common Errors:** Classification is not entered on the form.

#### More About: DoD Directive 5200.1-R, AR 380-5

#### DATE

Enter the application date. Include the **Day, Month,** and **Year**. Examples: 15 July 2004, or 15/07/2003.

**Common Errors:** The date field is left blank.

### PAGE NUMBER

Enter the page number. The DoD General Information page is <u>always</u> page 1, and is preprinted "Page 1 of \_ Pages." The total number of pages should be inserted in the space provided. The correct number to insert is the total number of equipment characteristics pages (transmitter, receiver, and antenna), plus two (for the DoD and NTIA General Information pages), plus the number of continuation pages, line diagram pages, and space systems data pages. **Common Errors:** Page numbering is not a common problem. The US Army DD Form 1494 processing office often re-numbers the pages before submitting them to the MCEB.

Three DD 1494 pages (DoD General Information, NTIA General Information, and Foreign Coordination General Information) have "Downgrading Instructions" blocks at the bottom. The downgrading instructions for the information presented must be supplied for classified systems. All downgrading instructions must conform to the requirements of DoD Directive 5200.1-R and AR 380-5. For unclassified systems the entry "NA" is applicable.

# то

Enter the addressee for US Army DD Form 1494 submissions. The current addressee is:

US ARMY NETWORK ENTERPRISE TECHNOLOGY COMMAND Attn: James Thomas 2461 Eisenhower Ave., Suite 1204 Alexandria, VA 22331-2200 Phone: 703-325-8227 DSN 221-8227

The e-mail address for submitting US Army DD Form 1494 applications is:

James.E.Thomas@us.army.mil

The e-mail address for classified US Army DD Form 1494 applications is:

Thomas3J@ncr.disa.smil.mil

**Common Errors:** Electronic delivery of these applications must be accompanied by an electronic signature of the government representative making the request for spectrum certification

### FROM

This block should include the mailing address of the government office originating the DD Form 1494. This may be the command that is procuring, developing, modifying, or using the equipment. This block should <u>not</u> contain the address of a government contractor.

**Common Errors:** Required information is not provided, or the mailing address is not correct.

## **1. APPLICATION TITLE**

Enter the Government's nomenclature of the equipment, or enter the manufacturer's name and model number, and a short descriptive title. If a Joint Electronics Type Designation System (JETDS) nomenclature (AN nomenclature) has been assigned to the RF component being allocated, it should be listed in this block along with the descriptive name of the equipment (e.g., AN/TPQ-37 Artillery Locating Radar). If a JETDS nomenclature is not assigned, a unique descriptive title must be entered (e.g., Symbol Technologies Spectrum 24 Wireless). In the case of commercially produced off-the-shelf equipment, the title should include the manufacturer's name and model number, in addition to a short descriptive title (e.g., SAVI Model STR7200 Hand-Held Interrogator, Reader Module).

If multiple equipment are allocated, this should be reflected in the Application Title block. Example: "SAVI EchoPoint System" would be a correct entry. The manufacturer's name is not important in the case of competition sensitive equipment; it is perfectly acceptable to omit this information. Instead, the identifier "Model A" or "Contractor B" should be substituted.

**Note:** "Title" and "Nomenclature not yet assigned" are not acceptable entries.

**Note:** The Application Title blocks on the NTIA General Information and Foreign Coordination pages must be identical to the information listed on the DoD General Information page.

**Common Errors:** The JETDS nomenclature (if available) is not included in the title. For COTS equipment, the equipment manufacturer's name and model number is not provided in the title.

#### More About: JETDS

#### 2. SYSTEM NOMENCLATURE

Enter the nomenclature of the overall system for which this equipment is a subsystem (e.g., Ammunition Automatic Identification Technology (AMMO AIT), Patriot, or Global Positioning System).

The System Nomenclature is a title describing the system requiring the allocated RF elements. In cases where an allocation is needed for a complex system made up of several subsystems, a separate DD Form 1494 must be submitted for each separate subsystem. For example, the Airborne Adverse Weather Weapon System (AAWWS) has two distinct RF components: the RF Hellfire Modular Missile System and the Fire Control Radar. An application for allocation for the AAWWS would require one form for the RF Hellfire Modular Missile System and one for the Fire Control Radar. Both of the DD Form 1494s would list the system nomenclature as the AAWWS, but the entry in Item 1, Application Title, would be different, depending on the subsystem being allocated.

**Note:** Because the two applications are related, each application should show the J/F 12 number of the other application in Item 10b of Page 1.

For less complex systems, the Application Title and System Nomenclature may be identical. In this case, the same entry as in Item 1, APPLICATION TITLE, should be inserted. Leaving this block blank may lead to confusion as to the exact relationship, or lack thereof, between this allocation application and a similar one.

Equipment should be addressed on an "RF link" rather than "platform" basis. Normally, each link is considered separately. For example, an air-launched guided missile might have two RF links: a Telecommand (aircraft-to-missile) link and a Video (missile-to-aircraft) link. Each of these RF links should be addressed separately, and should include one transmitter, one receiver, and one, or more, antenna characteristics pages. The data on the technical characteristics pages should be related, as they all pertain to one RF link (one set of equipment designed to function together). If the equipment is allocated on a platform basis, each allocation would still have one, or more, transmitter, receiver, and antenna page, and the only relationship would be the physical location of the equipment. A missile allocation would have the video transmitter with its associated antenna; and the telecommand receiver with its associated antenna. Obviously, allocations on an RF link basis present a clearer picture than allocations on a platform basis and are, therefore, less confusing.

**Common Errors:** System nomenclature information is not provided.

## 3. STAGE OF ALLOCATION

Stages of allocation correspond to the stages of life cycle management. The amount and type of data required on the DD Form 1494 vary with the stage of allocation. Read the general NTIA description of each stage, given below, and mark the appropriate block.

**STAGE-1 - CONCEPTUAL.** Initial planning has been completed, including proposed frequency bands and other available characteristics. A conceptual allocation is required prior to releasing funds for studies, or assembling "proof-of-concept" test-beds. Planned or estimated details concerning the equipment should be entered on the DD Form 1494. Where information has not been determined, the entry "NAvail" or "Not Available" should be used. Stage-1 is for system planning and does not involve actual transmission or reception.

**STAGE-2 - EXPERIMENTAL.** The preliminary design has been completed; onair radiations, using test equipment or preliminary models, may be required. Prior to the release of funds, an experimental allocation is required for building a radiating test model, or assignment of a frequency for experimental usage, including, but not limited to, units tested within a laboratory. Estimated and calculated data can be used, where appropriate, for nearly all DD Form 1494 technical items. **STAGE-3 - DEVELOPMENTAL.** The major design has been completed; radiation may be required. Prior to release of funds for engineering development models, a Stage-3 allocation must be obtained. All applicable items of the DD Form 1494 should be filled in with measured data. Where measured data are not available, calculated data can be substituted. An entry of "NAvail" or "Not Available" must include a reason.

**STAGE-4 - OPERATIONAL.** Development has been essentially completed and final operating constraints required to assure compatibility need to be identified. Prior to release of funds for production units, an operational frequency allocation is mandatory. Measured data for all technical characteristics such as emission bandwidth, harmonic level, spurious level, etc., should be provided when submitting a Stage-4 frequency allocation application.

All COTS equipment with unmodified RF characteristics fall into this type of allocation. Although the equipment may be used within experimental or developmental systems, by definition, it is operational equipment, because of its off-the-shelf status (for example, an off-the-shelf telemetry system used during development of a missile).

**Common Errors:** COTS equipment submitted as other than Stage-4. However, if COTs equipment is modified in any way, a Stage-2 or -3 application should be submitted.

More About: Data requirements for each stage

# 4. FREQUENCY REQUIREMENTS

4.a. Enter the required frequency band(s), indicating units, (e.g., kHz, MHz, or GHz). For equipment designed to operate only at a single frequency, enter this frequency and units.

4.b. Enter the emission designator(s), including the necessary bandwidth for each designator, as described in Chapter 9 of the NTIA manual (e.g., 40M0PON). For systems with a frequency hopping mode, identify each mode as "hopping" or "non-hopping" (e.g., 64M0F3E Hopping).

Enter other information pertinent to frequency requirements, such as minimum frequency separation or special relationships involving multiple discrete frequencies in Block 13, titled REMARKS.

**Common Errors:** Frequencies are not submitted. The frequency bands requested do not support the planned radio service. Emission designator is often left out. Lockouts have not been identified in the REMARKS section.

More About: Emission Designators, Frequencies, Lockouts

# 5. TARGET STARTING DATE FOR SUBSEQUENT STAGES

For each subsequent stage, enter the proposed date of application submission.

These dates mark when various life-cycle management phases of the equipment are scheduled to begin. A projected date for the start of all future phases should be provided in the blocks. An "NA" or "Not Applicable" should be entered on the line for current and accomplished phases. For example: A system moving from the current Advanced or Engineering Development stage to the Operational or Production stage in November of 2005 will list on the DD Form 1494 "NA" for STAGES 2 and 3 and "November 2005" for STAGE-4.

Occasionally, equipment is developed that was never intended to be introduced into the operational inventory (e.g., a proof-of-concept test bed, or a one-of-a-kind piece of laboratory equipment). Indicate this on the DD Form 1494 by entering an "NA" for the STAGE-4 date and including an explanation in Block 13, REMARKS.

**Common Errors:** Dates are not in correct sequence. Dates for the stage have already past.

More About: Life-cycle management, COTS Equipment

# 6. EXTENT OF USE

Describe the extent of use, continuous or intermittent, that will apply to Stage-4. If intermittent, an estimate of the total time of equipment operation should be given. For equipment intended primarily for peacetime use, this should include the number of hours per day, and days per week. For weapons systems and wartime-oriented systems, an estimated number of hours of operation per mission should be given. Most equipment (wartime systems included) have some use in a training role that should also be listed in this block. A sample entry in this block would read "Peacetime training, testing, and use will include approximately 2 hours per day, 5 days per week of intermittent transmissions. During wartime, system will operate at all times with intermittent 1-minute transmissions, approximately 10 per hour."

Common Errors: Information not supplied.

# 7. GEOGRAPHICAL AREA FOR

Enter geographic location(s), or area(s) of use, for this stage and subsequent stage(s). Provide geographical coordinates (degrees, minutes, seconds), if available. Enter "NA" for all accomplished stages.

#### Stage-2 and -3

Whenever possible, specific sites, identified by geographic location and coordinates should be listed. When a large number (e.g., >10) of US test sites are anticipated, but the final sites have not been chosen, it is acceptable to add "and various other test sites in the US&P."

All test sites outside the Continental United States (OCONUS) should be identified by base, station or military unit name, and the country.

#### Stage-4

The locations of operations should be as specific as possible. For fixed equipment, in most cases, it should be possible to list the site locations by latitude and longitude. For mobile systems where no specific location can be identified, all of the countries where standard coordination is to be accomplished must be listed. General rather than specific locations lead to time-consuming questions throughout the review process.

If the space on the DD Form 1494 is too small to list all locations and specific sites, use the REMARKS section or continuation pages.

**Common Errors:** Organizations often do not know where the system will be operated. Entering "worldwide" is not acceptable; specific countries are needed. The fewer locations to coordinate, the less time required for approval. Japan and Korea require specific geographic locations.

### 8. NUMBER OF UNITS

Enter the total number of units planned for the current allocation stage and for any remaining stages.

This item should be the sum of the number of units being built, procured, or used at each stage of allocation. If the number of transmitters differs from the number of receivers, they should be identified and listed separately. When there is a one-for-one correspondence of transmitters to receivers, a single figure may be listed.

Common Errors: Required data is not supplied.

# 9. NUMBER OF UNITS OPERATING SIMULTANEOUSLY IN THE SAME ENVIRONMENT

Enter the total number of units planned to be operating simultaneously in the same environment during Stage-4 use.

The environment is defined as the physical area over which the equipment signal can communicate or interfere. For a fixed microwave radio, this may be two units, the transmitting station and the receiving station. For radars, there may be only a single radar in the environment. In the case of a base police radio net, there may be dozens of "handie-talkies" within reception range of each other. When the number of systems operating may vary, a "normal" and a "maximum" value could also be given. The size of the area (e.g., geographical coordinate and a radius) should also be stated. The entry in Item 9 can never be "NA."

Common Errors: Required data is not submitted.

# 10. OTHER J/F 12 APPLICATION NUMBER(S) TO BE

Mark the appropriate boxes(s) and enter J/F 12 numbers for superseded and/or related applications.

#### BOX 10.A. SUPERSEDED

This is the box most often used when requesting a new allocation for an already allocated system. As a system transitions from conceptual to operational, it must be allocated at each stage. The title describes the action that will be taken by the USMCEB Frequency Panel and the Department of the Army. This block should also be used to indicate that the system being allocated is replacing an existing system. The system being replaced may continue to be operational, but until it is phased out of the inventory, it will still need a valid frequency allocation. Include the J/F 12 number to be superseded on the line to the right of the "Superseded J/F 12/" box. Write "None" when no system is being replaced.

#### BOX 10.B. RELATED

When two separately allocated systems are being used in conjunction with each other, such as a newly developed earth station and an existing space station, the space station's J/F 12 number should be entered opposite the checked box marked RELATED J/F 12. On the DD Form 1494, RELATED does not extend to different systems performing the same or similar purposes, unless the two systems work together. For example, the Motorola H34SX Hand Held and the General Electric MPD Series radios perform a voice communications function and have similar RF characteristics, but they are not RELATED.

Write "None" for new systems that have never been allocated and are not related to other systems.

**Common Errors:** Submitter indicates other systems that are not directly related to this application.

# **11. OPERATIONAL REQUIREMENT**

If this equipment will operate with the same, or similar, equipment used by other US Military Services, DoD Components, US Government Agencies, or Allied Nations, mark "YES" and specify, the Services, Agencies or countries (to include the country's services) in Box 13 REMARKS.

**Common Errors :** Answering "YES," but providing no explanation in Box 13 REMARKS.

#### More About: Remarks

#### **12. NAMES AND TELEPHONE NUMBERS**

The name and telephone number (both Commercial and DSN) of the Program Manager and a Project Engineer are required to permit contact between the US Army reviewing office and the Project Office during allocation processing. An important reason for this contact is to obtain missing or incomplete information. The "Project Engineer" should be somebody familiar with the RF parameters of the system and, where possible, the DD Form 1494. In addition to the two required contacts, it is not uncommon to list a contractor point of contact, if the equipment is being produced or developed by such a contractor. Unless otherwise specified, it is assumed that direct liaison is authorized with all contacts listed. When the organization and address of the Program Manager, and Project Engineer are different from the application originator, this fact should be supplied in the Block 13 REMARKS section.

**Common Errors :** Contractors are listed as primary points of contact. A Government POC is always required.

### 13. REMARKS

This block may be used as a continuation of any of the preceding blocks. The continued block, or blocks, should be identified by block number. If any of the DD Form 1494 is classified, the security classification of each item of information contained in the REMARKS block must be <u>explicitly marked</u>.

Any additional information that clarifies the function or operation of the equipment may be entered in this block.

**Common Errors:** Not including a reference number in the REMARKS block for one or more of the numbered blocks on the form.

CLASSIFICATION	PAGE			
TRANSMITTER E	QUIPMENT CHARACTERISTICS			
1. NOMENCLATURE, MANUFACTURER'S MODEL NO.	2. MANUFACTURER'S NAME			
3. TRANSMITTER INSTALLATION	4. TRANSMITTER TYPE			
5. TUNING RANGE	6. METHOD OF TUNING			
7. RF CHANNELING CAPABILITY	8. EMISSION DESIGNATOR(S)			
9. FREQUENCY TOLERANCE				
10. FILTER EMPLOYED (X one)				
a. YES b. NO				
11. SPREAD SPECTRUM (X one)	<b>12. EMISSION BANDWIDTH</b> (X and complete as applicable)			
a. YES b. NO	CALCULATED MEASURED			
13. MAXIMUM BIT RATE	a3 dB			
	b20 dB			
14. MODULATION TECHNIQUES AND CODING	c40 dB			
	d60 dB			
	e. OC-BW			
	15. MAXIMUM MODULATION FREQUENCY			
16. PRE-EMPHASIS (X one) a. YES b. NO	17. DEVIATION RATIO			
	18. PULSE CHARACTERISTICS			
19. POWER	a. RATE			
a. MEAN	b. WIDTH			
b. PEP	c. RISE TIME			
20. OUTPUT DEVICE	d. FALL TIME			
	e. COMP RATIO			
	21. HARMONIC LEVEL			
22. SPURIOUS LEVEL	a. 2ND			
	b. 3RD			
23. FCC TYPE ACCEPTANCE NO.	c. OTHER			
24. REMARKS				
ULASSIFICATION				

# TRANSMITTER EQUIPMENT CHARACTERISTICS

## 1. NOMENCLATURE, MANUFACTURER'S MODEL NO.

Enter the Government assigned JETDS equipment nomenclature. If no AN nomenclature has been assigned, a manufacturer's model name, or part number, should be entered. The number should be easily identified by the manufacturer if inquiries are made by the government. If the transmitter has neither type of identifier, as may occur in a conceptual or experimental allocation action, enter the system name followed by the word "transmitter." In a multi-transmitter system, a nomenclature describing the purpose of each transmitter would be appropriate on each respective transmitter page (e.g., "Command Guidance System (CGS) Projectile Subsystem Transmitter" and "Command Guidance System (CGS) Ground Subsystem"). Under no circumstances is the entry "None" or "Not Yet Assigned" acceptable.

**Common Errors:** Form is submitted without either the Government assigned nomenclature or manufacturer's model number or part number.

#### More About: JETDS

### 2. MANUFACTURER'S NAME

Enter the manufacturer's name, if available. Even if a manufacturer's model number is listed in Item 1 (Nomenclature, Manufacturer's Model No.) this block must still be completed. If the development is solely within an Army R&D element, it is appropriate to list the name of the organization in this block. In the case where a manufacturer has not been selected, as may occur in a conceptual (Stage-1) system, an entry of "NAvail" is acceptable here.

**Common Errors:** Manufacturer's subsidiary name is given, rather than that of the parent company.

#### More About: Manufacturer's

## 3. TRANSMITTER INSTALLATION

List the specific type(s) of vehicle(s), ship(s), plane(s), building(s), etc., where the transmitter will be installed. Include the platform(s) on which the transmitter is mounted (e.g., attached to a building, attached to a shipping container, a fixed site atop a mountain, an S-120 shelter, on a High Mobility Multipurpose Wheeled Vehicle (HMMWV), aboard a helicopter, etc.). All military vehicles should be identified by their proper military nomenclature. If the system is portable and not mounted in a vehicle, the user should be identified (e.g., man-portable carried by military police, hand held reader used by logistics personnel, or a handie-talkie worn by flight deck crew or by base police).

Common Errors: Required information is not supplied.

## 4. TRANSMITTER TYPE

Enter the generic class of the transmitter. The general modulation type and transmitter purpose should be entered as a two- or three-word description. Examples are: FM Communications, Pulse Doppler Radar, 802.11b Wireless LAN, FSK Data Communications, Spread-spectrum Communications, etc. As shown in the examples, the format is "general modulation type" followed by "purpose." An entry of "NAvail" or "NA" is unacceptable.

**Common Errors:** Required information is not supplied.

### 5. TUNING RANGE

Enter the frequency range (e.g., 225-400), and units (e.g., kHz, MHz or GHz) over which the transmitter is capable of being tuned. For equipment designed to operate only at a single frequency, enter this frequency.

The entry should be the lowest tunable center frequency through the highest tunable center frequency. As a matter of convention, this tuning range does not include the transmitter's emission bandwidth.

If the transmitter is normally not tunable (e.g., fixed crystal), the range of the lowest frequency obtainable by crystal substitution or by cavity adjustment through the highest frequency, should be listed. This is true even if only one of

several specific crystal frequencies is intended for use. At a later date, it may be necessary to shift the tuned frequency of the equipment, and, by correctly identifying the capabilities of the system, the Army spectrum manager can assess the ability of the system to operate relative to this shift.

If the equipment is designed for use at a single frequency only, cannot be tuned, or in any way adjusted in frequency, the center frequency of the emission or reception should be listed in this block. An example of a fixed frequency system is an ATCRBS IFF/SIF interrogator that would have the entry - "1030 MHz."

If there is more than one transmitter being used to cover the frequency ranges or if the transmitter is capable of tuning over a greater frequency range than is being requested for this allocation, the range listed in this block may differ from the range(s) given on page 1, Item 4, FREQUENCY REQUIREMENTS,

**Common Errors:** Specifying the frequencies planned for use instead of the transmitter range. For example, a software defined radio (SDR) can typically operate over a much wider range of frequencies than will be allowed for its operation.

#### More About: Software Defined Radio

#### 6. METHOD OF TUNING

Enter the method of tuning and frequency generation device (e.g., crystal, synthesizer or cavity). If the equipment is not readily tunable in the field, indicate the complexity of tuning in the Block 24 REMARKS. Include complexity factors such as skill levels involved, major assemblies involved, time required, and location (factory or depot) where equipment is to be tuned.

This is another two- or three-word descriptor, usually containing information about the method of effecting change and the device that generates the output frequency. Examples are: manually adjusted klystron cavity, fixed crystal, interchangeable crystal, manually adjusted crystal synthesizer, etc. Even if no tuning is possible, the second half of the descriptor is valid. Thus, a non-tunable radar would still have an entry appropriate for this block (e.g., sealed cavity magnetron). Common Errors: Required information is not supplied.

## 7. RF CHANNELING CAPABILITY

Describe the radio frequency channeling capability. For uniformly spaced channels, enter the center frequency of the lowest channel, the increments between consecutive center frequencies, and the number of channels (e.g., 406 MHz, 100-kHz increments, 20 channels). For continuous tuning, enter the lowest frequency and the word "continuous." For nonuniform channelization, or in cases where channel selection is under software control, enter a detailed description in Block 24, REMARKS. Any constraints for using any of these channels must also be described in Block 24, REMARKS (e.g., degraded channels, internal hardwiring limitations, or lockout capability for frequency hopping systems).

The block's title contains the acronym RF, for radio frequency. This is to qualify that only RF output channeling information is appropriate. Combining multiple input channels through time division multiplexing (TDM) or frequency division multiplexing (FDM) into an RF channel need not be described in this block. However, multiplexing information should be listed in Block 14, MODULATION TECHNIQUES AND CODING, or Block 24, REMARKS.

**Common Errors:** The entire frequency band is submitted.

More About: Radio frequency channel, Lockouts

# 8. EMISSION DESIGNATOR(S)

The emission designator is a string of seven alphanumeric characters that describe both the necessary bandwidth and the emission classification of the signal. Enter the emission designator(s) including the necessary bandwidth for each designator (e.g., 16K0F3E). The method for formulating an emission designator is described in the NTIA Manual of Regulations & Procedures for Federal Radio Frequency Management, included in the Guide. Necessary

bandwidth can be determined by following the guidance in Annex J of the NTIA Manual.

For systems with a frequency hopping mode as well as a non-hopping mode, enter the emission designators for each mode as hopping or non-hopping.

**Common Errors:** Bandwidth indicated in the Emission Designator does not reflect the values supplied in Item 12 EMISSION BANDWIDTH

More About: Emission Designators, Necessary Bandwidth

### 9. FREQUENCY TOLERANCE

Enter the frequency tolerance (i.e., the maximum departure of a transmitter from its assigned frequency after normal warm-up time has been allowed). Indicate the units in parts per million (ppm) for all emission types except single sideband (SSB), which shall be indicated in Hertz (Hz).

Frequency tolerance is defined as follows:

Freq tolerance (ppm) =  $\frac{\text{maximum transmitter drift (Hz)}}{\text{center frequency (MHz)}}$ 

It should be expressed in parts per million (ppm), rather than actual frequencies or a percentage of the (referenced or assigned) center frequency. A 1-MHz center frequency transmitter that can drift 1 Hz has a frequency tolerance of 1 ppm. A 100-MHz transmitter with a drift of 10 Hz has a tolerance of 0.1 ppm.

Note: The exception is that SSB frequency tolerance may be expressed in Hz.

The tolerance, or drift, of a transmitter from its assigned center frequency is a function of many factors. It may depend on transmitter temperature, age, or center frequency. The value entered should represent the frequency tolerance at the operating temperature and at a time sufficiently into the equipment's life cycle to exclude "Break-in" or "Pre-Burn-In" variations. It is not necessary to list these conditions in Block 9.

**Common Errors:** The information is not supplied, or does not meet the NTIA frequency drift standards.

#### More About: Warm-up time, SSB, NTIA Frequency Drift Standards

#### **10. FILTER EMPLOYED**

If a filter is employed between the final RF stage and the transmitter antenna, it should be indicated here. Use Block 24, REMARKS, to provide information on the type of filter (e.g., lowpass, highpass, or bandpass) as well as any additional specifications). An example of this additional information would be an entry in Block 24 REMARKS: "Low-Pass filter with 1.5-dB insertion loss and a minimum of 85-dB attenuation 25 MHz removed from the tuned frequency."

**Common Errors:** Item is marked YES, but the type of filter is not specified.

#### More About: Filter types

#### **11. SPREAD SPECTRUM**

If spread spectrum techniques are being employed, check the YES box and provide a complete description of how the spreading is achieved in Block 14, MODULATION TECHNIQUES AND CODING and Block 24, REMARKS.

Spread spectrum is defined by the Institute of Electrical and Electronic Engineers (IEEE) as "a modulation technique for multiple access or for increasing immunity to noise and interference. Spread spectrum systems make use of sequential noise-like signal structures, for example, pseudonoise (PN) codes, to spread the normally narrowband information signal over a relatively wide band of frequencies. The receiver correlates these signals to retrieve the original information signal." The principal spread spectrum methods are direct-sequence coding, frequency-hopping, or a hybrid of the two. Examples of a REMARKS entry would be "direct-sequence pseudorandom noise code is used to spread the bandwidth of the telecommand signal. Signal information rate = 14.6 kbits/s and chip rate = 18.5 Mbits/s" or "Frequency-hopping is employed using 21 discrete software selected channels within the tuning range. The hop rate is 20 k hops/s."

**Common Errors:** User selects YES, but does not specify the type of Spread Spectrum.

#### More About: Spread Spectrum

#### **12. EMISSION BANDWIDTH**

This item should contain information regarding the spectral energy distribution of the transmitted signal. The emission bandwidth is defined as the signal appearing at the antenna terminals and includes any significant attenuation contributed by filtering in the output circuit, or transmission lines. The bandwidths of the signals at points -3, -20, and -60 dB, relative to the fundamental signal level, are required, and should not include tuning range or carrier movement. The bandwidth at -40 dB shall also be entered if the transmitter is a pulsed radar transmitter. Values of emission bandwidth specified should be indicated as calculated or measured by marking the appropriate block. Indicate units used, e.g., Hz, kHz, or MHz. Note that the occupied bandwidth (Block 12.e) is defined as the bandwidth such that, below its lower and above its upper frequency limits, the mean powers radiated are each equal to 0.5% of the total mean power radiated.

During conceptual and experimental stages, the waveform envelope may be calculated. For developmental and operational allocations, measured data should be provided.

The bandwidth of a frequency-hopping signal should be measured at any instantaneous frequency.

This should fall somewhere between the -20 and -40 dB points of the emission bandwidth on most FM systems. Under no circumstances can the occupied bandwidth be less than the necessary bandwidth listed in Block 8, EMISSION DESIGNATORS(S).

A complete set of bandwidth data must be listed for each different pulse rate, pulse width, pulse rise time, and pulse fall time listed in Block 18, PULSE CHARACTERISTICS. In addition, a complete set of emission bandwidth data must be included for each different emission designator listed in Block 8. For instance, a radar system may be capable of operating as a pulse-Doppler radar or a frequency modulated (FM) chirp radar. In this case, two emission designators would be indicated in Block 8, and correspondingly, two sets of emission bandwidth data must be listed in Block 12. Additional sets of bandwidth data should be put in Block 24, REMARKS, or on continuation pages.

**Common Errors:** Information is not submitted. An emission designator and bandwidth is not supplied for multiple data rates. The data rates that apply to each emission designator are not provided.

#### More About: Decibels, Occupied bandwidth

#### **13. MAXIMUM BIT RATE**

Enter the maximum information bit rate for digital equipment, in bits per second (bps). If spread spectrum is used, enter the bit rate after encoding. This item ordinarily applies only to digital communication systems. Enter the maximum bit rate, etc., associated with the transmission of information and error correction, in bps. Chip rate information associated with spread-spectrum signals should be provided separately in Block 14, MODULATION TECHNIQUES AND CODING.

**Common Errors:** Required information is not submitted. Each bit rate should have an emission designator.

More About: Chip rate

### 14. MODULATION TECHNIQUES AND CODING

Describe, in detail, the modulation and/or coding techniques employed. For complex modulation schemes, such as direct-sequence spread-spectrum, frequency-hopping, frequency-agile, etc., provide full details in Block 24, REMARKS. An example of an entry would be "A 13-bit Barker Code is used to compress the radar pulses."

**Common Errors:** Information not supplied. More detail, other than that provided by the emission designator is often needed.

#### More About: Modulation Techniques and Coding

#### **15. MAXIMUM MODULATION FREQUENCY**

Enter the maximum modulation, or baseband, frequency. This applies to all analog and digital FM or phase-modulated (PM) signals, and is the maximum 3-dB down point on the high side of the modulating spectrum (i.e., the highest input frequency that is transmitted in that application). For a standard analog voice communications radio, the maximum modulation frequency might be designated as 3 kHz. For multichannel frequency-division multiplexed radio, the maximum modulation frequency of the highest channel after being multiplexed. (In some cases, the pilot tone is placed at the upper end of the modulation envelope. The pilot tone becomes the highest modulating frequency.) For example, if 10 3-kHz channels are multiplexed, the highest modulation frequency is 10 X 3 kHz, or 30 kHz. Indicate the units (e.g., Hz, kHz, or MHz).

For other kinds of systems, the maximum modulation frequency is not needed; the correct entry is "NA."

Common Errors: Required information is not supplied.

More About: FM Modulation.

#### 16. PRE-EMPHASIS

This item is applicable only to analog FM systems. Mark the appropriate block to indicate whether pre-emphasis is available. Pre-emphasis refers to preferential amplification of the higher modulating frequencies. Typically, the amplification increase is 6 dB per octave. If the exact amount of pre-emphasis is known, it should be entered in the space provided in the left-hand portion of the block. Pre-emphasis is common in COTS equipment, but is not generally found in tactical military radios.

Check the NO block for other systems, such as amplitude modulated (AM) and pulse-modulated (PM) systems.

**Common Errors:** Required information is not supplied.

#### More About: FM Pre-emphasis

## **17. DEVIATION RATIO**

For FM or PM transmitters, enter the deviation ratio computed with the formula:

 $Deviation Ratio = \frac{Maximum Frequency Deviation}{Maximum Modulation Frequency}$ 

This item applies only to FM or PM equipment. For FM systems, the deviation ratio is the ratio of the maximum amount a one-sided RF emission deviates from the center frequency to the maximum frequency of the input modulating signal. For FM systems, the deviation ratio is directly proportional to the amplitude of the modulating signal, since the amount of carrier frequency "swing" is tied to the amplitude variance of the modulator. In PM systems, the deviation ratio is tied to both the amplitude of the modulating signal and the phase deviation constant of the modulator.

For example, a deviation ratio of 1 for an FM system indicates that a 3-kHz input frequency will cause a peak instantaneous frequency deviation of 3 kHz. A deviation ratio of 3 is the result of a 9-kHz deviation of the emission when modulated with a 3-kHz signal.

For AM frequency-shift-keying (FSK) systems, the deviation ratio is not applicable.

Enter "NA" in this block for all AM and pulse-modulated systems.

Common Errors: Required information is not supplied.

# **18. PULSE CHARACTERISTICS**

Pulse Characteristics only apply to pulse-modulated systems. Do not enter data concerning short duration FM and AM emission transmitters (e.g., frequency-hopping) in these blocks. Information concerning transmission rates of repetitive

AM or FM emissions should be entered in Block 24, REMARKS, or on a continuation sheet.

#### Block 18.a. RATE

Pulse rates should be listed for individual pulses. While "pulse pairs" are commonly used in IFF system specifications, it is not correct to list the pulse-pair's repetition rate in this block. Rather, the number of individual pulses per second is the correct figure, and the remarks concerning pulse pairs should be included in Block 14, MODULATION TECHNIQUES AND CODING, or Block 24, REMARKS. If several different pulse rates are employed by the system, each rate or range should be listed rather than just the maximum rate. If different pulse widths are associated with different pulse rates, then the association should be indicated.

Block 18.b. WIDTH

The pulse width is that time period during which the baseband voltage (or RF envelope at the antenna terminals) exceeds 50% of the highest voltage that will be conducted to the antenna. This time period is measured in microseconds ( $\mu$ s). Obviously, the pulse width of a transmitter cannot exceed one divided by the pulse rate (1/pulse rate); to do so would yield a transmitter duty-cycle greater than 100%.

Block 18.c. RISE TIME

The pulse rise time is the time period between the 10% and 90% points of the RF voltage envelope measured on the leading edge of the pulse. The time is measured in microseconds. There is an overlap between the rise time period and the pulse width period. The pulse rise and fall times are major contributing factors to the overall spectral power distribution, with the faster of the two having the greatest effect. Thus, different rise and fall times will cause different RF emission bandwidths. For this reason, each different pulse rise and fall time listed in Block 18 must be accompanied by its attendant RF emission bandwidth. These bandwidths should be listed in column form in Block 12, EMISSION BANDWIDTH, or Block 24, REMARKS.

Block 18.d. FALL TIME

The pulse fall time is the time period between the 90% and 10% points of the RF voltage being supplied to the antenna on the trailing edge of the pulse. This time period is measured in microseconds. This time period overlaps the pulse width time period.

Block 18.e. COMPRESSION RATIO

The previous four pulse parameters (rate, width, rise time, and fall time) apply to all pulse-modulated systems and must be supplied. The last parameter in Block 18, the pulse compression ratio, applies only to linear or non-linear FM pulse modulation systems, or to phase-coded pulse-modulation systems. A linear FM pulse-modulation system is one in which the carrier frequency is changed linearly during the pulse period. The compression ratio is the ratio of the uncompressed pulse width to the compressed pulse width, where the compressed pulse width (in µs) is 1/frequency displacement (in MHz). (This can also be expressed as the uncompressed pulse width multiplied by the carrier frequency displacement.) Thus, a 50-µs pulse width with a linear FM chirp of 1 MHz has a pulse compression ratio of 50. A phase-coded pulse modulation system divides a pulse into N subpulses and then phase-modulates the carrier accordingly. The code used to describe the phase of each subpulse is generally a "Barker Code." The compression ratio is N, the number of subpulses. Since the longest Barker Code is only 13 bits long, to achieve compression ratios larger than 13, several Barker Codes are often combined to form longer strings.

Enter "NA" in this block for all AM systems.

Common Errors: Required information is not supplied.

#### More About: Pulsed Radar Systems

#### 19. POWER

Enter the mean power delivered to the antenna terminals for all AM and FM type emissions. For all other classes of emissions enter the peak envelope power (PEP). Any unique situations such as interrupted continuous wave (CW), should be provided in detail in Block 24, REMARKS. Indicate the units (e.g., W or kW).

Block 19.a. MEAN POWER

Applicable to all forms of modulation except single sideband (SSB) amplitude modulation (AM), the mean, or average power is defined as the power supplied to the antenna transmission line averaged over a time that is long when compared with the period of the lowest frequency encountered in the modulation. A time of 1/10 second, during which the mean power is greatest, is normally selected. For a pulsed system, the mean power is calculated as:

Mean power = peak power X duty cycle

where duty cycle = pulse rate X pulse width

Block 19.b. PEAK ENVELOPE POWER (PEP)

This item must be completed for SSB AM systems, PM systems, and AM television systems. The PEP is defined as the average power supplied to the antenna transmission line by a transmitter during one radio frequency cycle at the highest crest of the modulation envelope, taken under conditions of normal operation.

The power of a pulsed system must be given twice, once as mean and once as peak power.

**Common Errors:** Required information is not supplied or is incorrect.

More About: SSB, CW, AM, FM

#### 20. OUTPUT DEVICE

Enter a description of the output device used in the transmitter output stage. Output device refers to the final RF power output device. Examples include: Klystron, magnetron, traveling wave tube (TWT), transistor pair in Class C amplifier configuration, Impatt diode, Gunn diode, etc. The specific device designation should also be supplied (e.g., Varian VTS5751A1 TWT).

**Common Errors:** Required information is not submitted.

#### More About: Output devices

#### 21. HARMONIC LEVEL

All RF transmitters emit energy at various places in the spectrum outside the necessary bandwidth. These out-of-band emissions are referred to as "spurious emissions." The highest levels of spurious energy generally occur at frequencies that are multiples or "harmonics" of the center frequency. Thus, a transmitter tuned to 30 MHz will also emit energy on 60 MHz, 90 MHz, 120 MHz, etc. Block 21 denotes harmonic spurious emissions. Block 22 denotes non-harmonic spurious emissions.

Block 21 requires the 2<sup>nd</sup>, 3<sup>rd</sup>, and "Other" harmonic power level specified in decibels relative to the peak output power of the carrier signal (dBc). The 2<sup>nd</sup> harmonic emission falls at the frequency equal to two times the fundamental frequency. Similarly, the 3<sup>rd</sup> harmonic falls at the frequency equal to three times the fundamental frequency. The "Other" referred to in Item 21c is the greatest harmonic emission power level above the 3<sup>rd</sup> harmonic.

Harmonic emission power levels tend to decrease as the harmonic frequencies increase. Thus, the 2<sup>nd</sup> harmonic power level is generally greater than the 3<sup>rd</sup>, with the 3<sup>rd</sup> expected to be greater than that listed under "Other." The carrier signal, or fundamental emission, is always designated as 0 dB. Thus, a harmonic or spurious emission with a peak power of 80 dB below that of the fundamental has a relative level of -80 dBc. The level should always be expressed as a relative dB level, never as an absolute level (dBm or dBW). Whenever possible, the power levels should be measured from the radiated spectrum of the transmitter. If radiated spectrum measurements are not possible, the power levels should be measured at the antenna terminals. Indicate where the measurement was made (radiated, antenna terminals, or transmitter final stage).

**Common Errors:** Information is not provided, or is submitted in wrong units.

## 22. SPURIOUS LEVEL

Enter the maximum value of spurious emission in dB relative to the fundamental, which occurs outside the –60-dBc point on the transmitter fundamental emission spectrum but does not occur on a harmonic of the fundamental frequency.

**Common Errors:** Information is not provided or is submitted in wrong units.

# 23. FCC TYPE ACCEPTANCE NO.

This applies only to COTS equipment. Civil equipment parameters are reviewed by the FCC in a manner much like the DoD frequency allocation process. The FCC assigns a type acceptance number, which generally includes the manufacturer's model number. For example, the FCC type acceptance number for the Savi Hand Held Reader Module, Model STR7200, is "P93-STR7200-1."

Equipment with an FCC-type acceptance is not exempt from the DoD frequency allocation process. A frequency allocation is still required for all equipment developed, leased, modified, or procured by any DoD agency, or contractor, in support of a DoD contract.

If commercial equipment is being modified, or does not have a type acceptance number, "NA" is the proper response in this block.

**Common Errors:** Incorrect number is supplied.

**More About:** Commercial off-the-shelf equipment, FCC Type Acceptance Number

### 24. REMARKS

This block may be used as a continuation of any of the preceding blocks. The continued block, or blocks, should be identified by block number. If any of the DD Form 1494 is classified, the security classification of each item of information contained in the REMARKS block must be <u>explicitly marked</u>.
Any additional information that clarifies the function or operation of the equipment may be entered in this block.

**Common Errors:** Not including a reference number in the REMARKS block for one or more of the numbered blocks on the form.

		RECEIVI	ER EQUIPME	INT CHARACTERISTICS			
1. NOMENCLATURE, MANUFACTURER'S MODEL NO.				2. MANUFACTURER'S NAME			
3. RECEIVER INSTALLATION				4. RECEIVER TYPE			
5. TUNING RANGE				6. METHOD OF TUNING			
7. RF CHANNELING CAPABILITY				8. EMISSION DESIGNATOR(S)			
9. FREQUENCY TOLERANCE							
10. IF SELECTIVITY	1ST	2ND	3RD	11. RF SELECTIVITY (X and complete as applicable)			
a3 dB				CALCULATED MEASURED			
b20 dB				a3 dB b20 dB			
c60 dB				c60 dB			
				d. PRESELECTION TYPE			
a. 1ST				13. MAXIMUM POST DETECTION FREQUENCY			
b. 2ND				14. MINIMUM POST DETECTION FREQUENCY			
c. 3RD							
15. OSCILLATOR TUNED	1:	ST 2ND	3RD	16. MAXIMUM BIT RATE			
a. ABOVE TUNED FREQUENCY				17. SENSITIVITY			
b. BELOW TUNED FREQUENCY				a. SENSITIVITY	dBm		
c. EITHER ABOVE OR BELOW TUNED FREQUENCY			b. CRITERIA				
18. DE-EMPHASIS (X one)				c. NOISE FIG dB			
a. YES b. NO				d. NOISE TEMP Kelvin			
19. IMAGE REJECTION				20. SPURIOUS REJECTION			
21. REMARKS							
CLASSIFICATION							

# **RECEIVER EQUIPMENT CHARACTERISTICS**

## 1. NOMENCLATURE, MANUFACTURER'S MODEL NO.

Enter the Government assigned JETDS equipment nomenclature. If no "AN" nomenclature has been assigned, a manufacturer's model name or part number should be entered. The number should be easily identified by the manufacturer if inquiries are made by the Government. If the receiver has neither type of identifier, as may occur in a conceptual or experimental allocation action, the system name followed by "receiver" should be entered. In a multi-receiver system, a nomenclature describing the purpose of each receiver would be appropriate on each respective receiver page. Examples: Command Guidance System (CGS) Projectile, Subsystem Receiver or Command Guidance System (CGS) Ground Subsystem. Under no circumstances are the entries "None" or "Not Yet Assigned" acceptable.

**Common Errors:** Form is submitted without either Government assigned nomenclature or manufacturer's model number.

#### More About: JETDS

## 2. MANUFACTURER'S NAME

Enter the manufacturer's name, if available. Even if a manufacturer's model number is listed in Block 1 NOMENCLATURE, MANUFACTURER'S MODEL NUMBER, this item must be completed. If the development is solely within an Army R&D element, list the name of the organization in this block. If a manufacturer has not been selected, as may occur in a conceptual (Stage-1) system, then an entry of "NAvail" is acceptable.

**Common Errors:** Manufacturer's name listed is a subsidiary instead of the parent company.

#### More About: Manufacturer's

# 3. RECEIVER INSTALLATION

List the specific type(s) of vehicle(s), ship(s), plane(s), or building(s), etc., where the receiver(s) will be installed (e.g., attached to a building, attached to a shipping container, a fixed site atop a mountain, an S-120 shelter, a HMMWV, aboard a helicopter, etc.). All military vehicles should be identified by their proper military nomenclature. If the system is portable and not mounted in a vehicle, the user should be identified (e.g., man-portable carried by military police, hand held reader used by logistics personnel, or a handie-talkie worn by flight deck crew or by base police).

Common Errors: Information is not supplied.

# 4. RECEIVER TYPE

Enter the generic receiver type. The general format for entries in this item is "number of superheterodyne stages", "general modulation type", and "purpose."

If the receiver is not a superheterodyne type, either "heterodyne," "homodyne," "tuned radio frequency (TRF)," or "regenerative" should be substituted, depending on the form of detection used.

In heterodyne detection, the incoming RF signal is directly mixed with the local oscillator (LO).

A homodyne receiver mixes the incoming RF signal with a signal of the same carrier frequency, thereby converting the RF signal to baseband without using an intermediate frequency (IF) stage. This form of demodulation is quite often used with Doppler radars. Doppler radars leak a small portion of the CW oscillator power that generates the transmitted pulses into the receiver to take the place of the LO. The incoming RF is then mixed with this LO. The frequency of the resultant baseband signal is the amount of the Doppler-induced shift. (This is sometimes referred to as coherent detection, since the phase of the transmitted signal is preserved in the reference signal.)

In a TRF receiver the detector immediately follows the RF amplifier stages. This technique is rarely used in modern communications and radar systems.

Examples: Single Conversion FM Communications, Homodyne Doppler Radar, and Double Conversion Spread Spectrum Communication. An entry of "NAvail" or "NA" is not permitted.

**Common Errors:** Required information is not supplied.

More About: Receiver Types.

## 5. TUNING RANGE

Enter the frequency range, through which the receiver is capable of being tuned (e.g., 225-400 MHz). For equipment designed to operate only at a single frequency, enter this frequency. Indicate units (kHz, MHz, or GHz).

This should be the lowest tunable center frequency through the highest tunable center frequency. Do not include the receiver acceptance bandwidth as part of this range.

**Common Errors:** Required information is not supplied.

# 6. METHOD OF TUNING

Enter the method of tuning, (crystal, synthesizer, or cavity). If the equipment is not readily tunable in the field, indicate in Block 21 REMARKS, the complexity of tuning. Include complexity factors such as skill levels involved; major assembles involved, time required, and location (factory or depot) where equipment is to be tuned.

This is another two- or three-word descriptor, with the format of "method of effecting change" and "device ensuring frequency control." Examples: manually adjusted klystron cavity, fixed crystal, interchangeable crystal, manually adjusted crystal synthesizer, etc. If no tuning is possible, the second half of the descriptor is still valid. Thus, a non-tunable hand held RFID reader would still have an entry appropriate for this block: Surface Acoustic Wave (SAW) Resonator.

**Common Errors:** Required information is not supplied.

# 7. RF CHANNELING CAPABILITY

Describe the radio frequency channeling capability. For uniformly spaced channels, enter the center frequency of the lowest channel and the increments between consecutive center frequencies and the number of channels (e.g., 406 MHz, 100-kHz increments, 20 channels). For continuous tuning, enter the lowest frequency and the word "continuous." For nonuniform channelization, or in cases where channel selection is under software control, enter a detailed description in Block 24, REMARKS including any constraints for using any of these channels (e.g., degraded channels, internal hardwiring limitations, or lockout capability for frequency hopping systems).

Common Errors: Required information is not supplied.

# 8. EMISSION DESIGNATOR(S)

The emission designator is a string of seven alphanumeric characters that describe both the necessary bandwidth and the emission classification of the signal. Enter the emission designator(s) including the necessary bandwidth for each designator (e.g., 16K0F3E). The method for formulating an emission designator is described in the NTIA Manual of Regulations & Procedures for Federal Radio Frequency Management.

For systems with a frequency hopping mode as well as a non-hopping mode enter the emission designators for each mode. Identify each mode as hopping or non-hopping.

**Common Errors:** Information is not supplied. The emission designator should be paired with the transmitter so it should match the emission designator on the transmitter page of the application.

### More About: Emission Designator

# 9. FREQUENCY TOLERANCE

Enter the frequency tolerance, (the maximum departure of a receiver from its assigned frequency after normal warm-up time has been allowed). Indicate the units in ppm for all emission types except SSB which shall be indicated in Hz.

Frequency tolerance is defined as follows:

Freq tolerance (ppm) =  $\frac{\text{maximum receiver drift (Hz)}}{\text{center frequency (MHz)}}$ 

Frequency tolerance should be expressed in ppm, rather than actual frequencies or a percentage of the (referenced or assigned) center frequency. A 1-MHz center frequency receiver that can drift 1 Hz has a frequency tolerance of 1 ppm. A 100-MHz receiver with a drift of 10 Hz has a tolerance of 0.1 ppm. Note the exception that SSB frequency tolerance may be expressed in Hz.

**Common Errors:** The information is not supplied, or does not meet the NTIA frequency drift standards.

### More About: Warm Up Time, SSB, NTIA Frequency Drift Standards

## **10. IF SELECTIVITY**

Enter the bandwidth for each IF stage at the -3, -20, and -60 dB levels, and indicate units (kHz or MHz). This applies to all heterodyne and superheterodyne receivers. When there is more than one IF-stage in the receiver, a separate set of bandwidth data must be given for each IF-stage. The number of IF-stages corresponds to the number of IF frequencies listed in Block 12 IF FREQUENCY and to the number of conversions listed in Block 4, RECEIVER TYPE. Thus, a receiver described as a "triple conversion superheterodyne" must have three sets of bandwidth data listed in Block 10, IF Selectivity, and three frequencies listed in Block 12, IF FREQUENCY.

Generally, but not always, the IF bandwidth data are smaller than the RF bandwidth data, with each successive IF-stage having bandwidth data that is no larger than the previous one. Since the final IF filter must pass the modulated signal at the translated carrier frequency, its 3-dB bandwidth should equal, or exceed, the necessary bandwidth listed in Block 8, EMISSION DESIGNATOR(S). However, the IF-stage is responsible for rejecting unwanted carriers in the immediate vicinity of the desired signal or "adjacent-channel selectivity." Thus, the IF 3-dB bandwidths should not be much larger than the necessary bandwidths.

If the receiver is a homodyne or a TRF type, the appropriate entry in this block is "NA."

**Common Errors:** An IF selectivity that is wider than the RF selectivity is a common red flag. A note to explain this situation in the Block 21 REMARKS can reduce processing delays.

# **11. RF SELECTIVITY**

Enter the RF bandwidth at the -3, -20 and -60 dB levels. The RF bandwidth includes any significant attenuation contributed by filtering in the input circuit or transmission line. Values of RF bandwidths specified should be indicated as calculated or measured by marking the appropriate block. Indicate units (kHz or MHz). Enter the preselection type (e.g., tunable cavity).

The acceptance bandwidth data of the RF circuits preceding the first mixer or demodulator are required in this item. The bandwidth data should reflect the attenuation of the received signal by the antenna, the antenna-transmission line, any RF preselection-filters, and any RF preamplifiers.

The bandwidth curve of the RF selectivity should be greater than, or equal to, the emission curve of the received signal. If not, information will be lost due to the inability of the receiver to accept the entire signal. The significant types of preselection employed must be entered in Block 11.d PRESELECTION TYPE. Possible entries are waveguide cutoff, YIG filter, 6-pole Butterworth filter, etc.

The primary purpose of the RF stage is to reject the "image" frequency before it gets to the mixer. Receivers that omit the RF stage such as a homodyne or that heterodyne to an audio frequency (omitting the IF stages), are much more likely to experience interference from spurious signals located on the image frequencies.

**Common Errors:** The information is not supplied.

More About: Receivers, Preselection Types

# **12. IF FREQUENCY**

Enter the tuned frequency of the 1<sup>st</sup>, 2<sup>nd</sup>, and 3<sup>rd</sup> IF stages. Indicate units (e.g., kHz or MHz). In a heterodyne or superheterodyne receiver, each conversion stage yields an IF stage frequency. The IF frequency is determined by computing the sum or difference between the input RF signal and the local oscillator (LO) frequency. For multiple-conversion superheterodyne receivers, the input signal for the subsequent converters is the IF signal from the preceding IF stage. Block 12 has blanks for three IF stages. There must be a different IF frequency listed for each set of IF selectivity bandwidth data listed in Block 10, IF SELECTIVITY. The number of IFs listed should also correspond to the number of stages of conversion indicated in Block 4, RECEIVER TYPE. Additional IFs can be listed in Block 21, REMARKS.

If the receiver is a homodyne or TRF receiver, "NA" should be entered on all three lines of this block. If only one IF is employed, an "NA" should be entered in the 2<sup>nd</sup> and 3<sup>rd</sup> columns.

**Common Errors:** Required IF frequency is not given for each stage of a superheterodyne receiver.

### **More About: Receivers**

# **13. MAXIMUM POST DETECTION FREQUENCY**

Enter the maximum post detection frequency; the nominal frequency at the –3-dB point on the high frequency side of the receiver baseband. Indicate units (kHz or MHz).

This is the highest baseband frequency that can be recovered and demodulated by the receiver. Specifically, this frequency should have less than 3-dB attenuation relative to the least attenuated multiplexed signal. For a single channel analog voice system, this frequency is generally 3 kHz. For a multichannel, frequency-division multiplexed receiver, the maximum post-detection frequency is the highest frequency of the highest channel, measured after demodulation but prior to demultiplexing. For a pulse modulated system, the correct entry would be "NA." If the low frequency response is a significant parameter, then the 3-dB low frequency cutoff should be provided in the remarks block.

**Common Errors:** Required information is not supplied.

# **14. MINIMUM POST DETECTION FREQUENCY**

For multichannel FM systems enter the minimum post detection frequency; the nominal frequency at the -3-dB point on the low frequency side of the receiver base band. Indicate units (kHz or MHz).

This item applies only to multi-channel FM frequency-division multiplexed receivers. The minimum post-detection frequency is the lowest frequency of the lowest channel, measured after detection but prior to demultiplexing.

For AM, pulse-modulated, and single-channel FM systems, the appropriate entry is "NA."

Common Errors: Required information is not supplied.

# 15. OSCILLATOR TUNED

Mark the appropriate block to indicate the location of the 1<sup>st</sup>, 2<sup>nd</sup>, and 3<sup>rd</sup> oscillator frequencies with respect to the associated mixer input signal.

This block answers whether the local oscillator is tuned above or below the center frequency of the input to the mixer. If the receiver is a homodyne or TRF type, the correct entry is "NA."

If more than one stage of conversion is indicated in Block 4, "RECEIVER TYPE", Block 10, IF SELECTIVITY, and Block 12, IF FREQUENCY, the relationship of each local oscillator relative to its IF should be listed in Block 15, OSCILLATOR TUNED.

**Common Errors:** Required information is not supplied for superheterodyne receivers.

### More About: Receivers

## **16. MAXIMUM BIT RATE**

Enter the maximum information bit rate for digital equipment, in bps. If spread spectrum is used, enter the bit rate after encoding. This item ordinarily applies only to digital communication systems. Maximum bit rate does not mean pulse repetition rate for a radar. Enter the maximum bit rate, etc., associated with the transmission of information, including error correction, in bps. Chip rate information (associated with spread spectrum signals) should be provided separately in Block 21, REMARKS.

**Common Errors:** Information is not submitted. Each bit rate should have a emission designator.

## **17. SENSITIVITY**

Sensitivity is divided into four parts: minimum acceptable signal level (sensitivity), the criteria defining an acceptable signal, noise figure, and equipment noise temperature. Each of the parts is explained here.

Block 17.a. SENSITIVITY

Enter the minimum acceptable signal level required to be present at the antenna input terminals for a specified performance. Signal levels must be specified in decibels referenced to one mW (dBm). Field strengths in microvolts per meter are not acceptable.

Block 17.b. CRITERIA

Enter the sensitivity criteria. For many receivers, this is the signal-to-noise ratio (S/N), or the signal+noise+distortion to noise+distortion ratio (SINAD) required to produce an acceptable system output, measured in decibels. For some receivers, other types of criteria are applicable. Often, the criterion used in data systems is an acceptable bit-error rate (BER), or a certain probability of successful reply. Other acceptable criteria include Minimum Discernible Signal (MDS) and Automatic Gain Control (AGC) response.

### Block 17.c. NOISE FIGURE

Enter the receiver noise figure. Noise figure is the measure of the internal receiver noise present in the output. It is the ratio of the input S/N ratio to the output S/N ratio at the standard temperature of 290 K. It is applicable only to terrestrial systems. For space systems, an acceptable entry is "NA."

### Block 17.d. NOISE TEMPERATURE

Enter the receiver noise temperature (space systems only). These data should be provided for all types of space system receivers. The noise temperature corresponds to the temperature of a thermal noise source that produces the same amount of noise as the receiver. Typically, a cooled satellite earth-station receiver has an equivalent noise temperature of 30-70 K. The acceptable entry for terrestrial systems is "NA."

**Common Errors:** The sensitivity is provided without criteria. Noise figure or noise temperature is not provided. Incorrect units are used for sensitivity.

### More About: Sensitivity, Noise Figure, Noise Temperature

## **18. DE-EMPHASIS**

For FM or PM receivers, mark the appropriate block to indicate whether deemphasis is available.

De-emphasis in the receiver works in conjunction with pre-emphasis in an FM transmitter. It is a noise suppression technique where the high frequencies that were artificially amplified prior to transmission are compressed to their original levels. De-emphasis is common in COTS FM radios. It is not generally found in military-developed radios.

If the subject system is an AM, pulse-modulated, PM, or FM system not using deemphasis, the NO block should be checked.

**Common Errors:** If the associated transmitter uses pre-emphasis the receiver should use de-emphasis.

### More About: Pre-emphasis

## **19. IMAGE REJECTION**

The image rejection is the receiver attenuation, in dB, of a signal at the image frequency. It should always be a positive number. For example, if a receiver had a sensitivity of -100 dBm and could receive an "image" signal with a power of -20 dBm causing the standard response (criteria in 17.b), then the receiver image rejection would be 80 dB. Enter the image rejection in dB.

This block is applicable to superheterodyne receivers only.

The "image" signal generated in a receiver is a special case of a spurious response. This response in a receiver occurs when there is an additional RF signal located at a frequency twice the IF frequency away from the tuned RF signal such that when it is combined with the LO, it produces the IF frequency. The image frequency is defined as that RF frequency which is displaced from the LO frequency by the IF frequency and which appears on the opposite side of the LO frequency as the signal frequency.

The "image" signal must be attenuated before reaching the 1<sup>st</sup> mixer in a superheterodyne receiver; otherwise, the undesired "image" frequency signal will be processed in the IF amplifier in the same manner as the desired RF signal. Filtering the RF band in preselection and the RF stages sufficiently will attenuate the "image" frequency signal to a point where it will no longer cause interference.

For heterodyne, homodyne, or TRF receivers, "NA" is the appropriate entry in this block.

**Common Errors:** Incorrect units are used for image rejection. Image rejection should be provided in dB.

#### More About: Receivers

# **20. SPURIOUS REJECTION**

Enter the single level of spurious rejection that the receiver meets or exceeds at all frequencies outside the -60-dB IF bandwidth. Enter the spurious rejection in dB. Spurious rejection is the ratio of the power level of an out-of-band signal required to produce a specified output, to the desired signal power level required to produce the same output.

A spurious response is any undesired response of an electronic device, through its intended input terminal, to energy outside its tuned frequency. Spurious receiver responses arise when strong undesired signals and the receiver LO combine in the mixer to produce a frequency on, or near, the IF frequency. The power required to excite a spurious receiver response is normally much higher than that required for the desired response. Spurious rejection is always a positive number. For example, a receiver with a sensitivity of -120 dBm would provide 80 dB of spurious rejection if the smallest spurious signal that could cause the standard response had -40 dBm of power.

For homodyne and TRF receivers, "NA" is an appropriate entry for this block.

**Common Errors:** Incorrect units are used for image rejection. Image rejection should be provided in dB.

#### More About: Receivers

## 21. REMARKS

This block may be used as a continuation of any of the preceding blocks. The continued block, or blocks, should be identified by block number. If any of the DD Form 1494 is classified, the security classification of each item of information contained in the REMARKS block must be <u>explicitly marked</u>.

Any additional information that clarifies the function or operation of the equipment may be entered in this block.

**Common Errors:** Not including a reference number in the REMARKS block for one of the numbered blocks on the form.

CLASSIFICATION	PAGE				
	IT CHARACTERISTICS				
1a. TRANSMITTINGb. RECEIVING	c. TRANSMITTING AND RECEIVING				
2. NOMENCLATURE, MANUFACTURER'S MODEL NO.	3. MANUFACTURER'S NAME				
4. FREQUENCY RANGE	5. TYPE				
6. POLARIZATION	7. SCAN CHARACTERISTICS				
	a. TYPE				
a. MAIN BEAM	D. VERTICAL SCAN (1) MAX ELEV				
b. 1ST MAJOR SIDE LOBE	(2) MIN ELEV				
	(3) SCAN RATE				
9. BEAMWIDTH	c. HORIZONTAL SCAN				
a. HORIZONTAL	(1) SECTOR SCANNED				
b. VERTICAL	(2) SCAN RATE				
	d. SECTOR BLANKING (X one)				
10. REMARKS	(1) YES (2) NO				
CLASSIFICATION					

# ANTENNA EQUIPMENT CHARACTERISTICS

# 1. FUNCTION

To indicate the type of function the antenna performs, mark the appropriate block: TRANSMITTING, RECEIVING, or TRANSMITTING AND RECEIVING. For multi-antenna systems, use one page for each antenna. If separate, but identical, antennas are used for transmitting and receiving (as might be the case in an RFID tag), only one Antenna Equipment Characteristics page should be submitted, and the TRANSMITTING AND RECEIVING block should be checked If two identical antennas are being used it should be noted in Block 10, REMARKS. Similarly, if two or more identical antennas are employed in a space-diversified or multi-area coverage system, only one page is required for each antenna type, and how the antenna is being used must be explained further in Block 10, REMARKS.

**Common Errors:** If TRANSMITTING is selected for antenna function, then a separate receiver antenna page is required.

# 2. Nomenclature, Manufacturer's Model No.

Enter the Government-assigned alphanumeric equipment designation; if unavailable, enter the manufacturer's model number (e.g., DS6558), and complete Block 3. If neither is available, enter a short descriptive title (e.g., EchoPoint SMR-640P-110 Mobile Reader Antenna). If the antenna is an integral component of the transmitter (e.g., molded into the case), submit the model number of the transmitter.

**Common Errors:** Required information is not submitted.

More About: Manufacturer's

# 3. MANUFACTURER'S NAME

Enter the manufacturer's name, if available. If a manufacturer's model number is listed in Block 2, this item must be completed.

**Common Errors:** Required information is not submitted.

# 4. FREQUENCY RANGE

Enter the range of frequencies (and units – kHz, MHz, GHz) for which the antenna is designed.

The antenna tuning range may be listed in one of two ways. First, the efficient operating frequency range of the antenna, the range over which the antenna's radiated output does not vary by more than 3 dB when measured at a fixed location in the main beam, may be entered. If this efficient operating range does not cover the tuning range of its associated transmitter, receiver or transceiver, the range over which it will be used should be noted in Block 10, REMARKS (along with the resulting variation in gain).

**Common Errors:** Antenna(s) should cover a frequency range that is at least equal to the frequency range covered by the transmitter or receiver. This is a common red flag during application review.

### More About: Antennas

# 5. TYPE

Enter the generic name of the antenna or describe general technical features. The description should include information regarding either the physical or electrical size of the antenna. Examples: quarter-wave dipole, three-meter hornfed parabolic reflector, 10-meter Cassegrain reflector with half-wave dipole feed, etc. The description should be as explanatory as possible. If more space is needed, the description of any and all antenna peculiarities should be continued in Block 10, REMARKS. Reference to a technical article should be provided for unique antenna designs. All physical measurements should be expressed in metric units of measurement. If available, illumination functions (such as cosecant, cosecant-squared, etc.) should be provided in this block. Illumination functions are discussed further in the instructions for Block 9, BEAMWIDTH.

**Common Errors:** Required information is not submitted.

## More About: Antenna Types

# 6. POLARIZATION

Enter the antenna polarization (e.g., horizontal or vertical linear); if circular, indicate whether it is right or left hand.

Polarization refers to the orientation of the electric field vector of the radiated electromagnetic wave relative to the ground plane. The particular polarization(s) of the antenna being described on this page should be entered in this block. If the polarization of the antenna can be changed, this should be noted in the Block 10, REMARKS, along with a description of how the change is accomplished.

**Common Errors:** Required information is not submitted.

More About: Antenna Polarization

# 7. SCAN CHARACTERISTICS

A complete description of the antenna scan pattern, or its range of motion, should be provided in Block 7. Since antenna scanning has so many variables and can be rather complex, Block 10, REMARKS, should also be used to explain any antenna scanning that cannot be clearly described in Block 7. Side lobe adaptive antennas can be adequately described using Block 10, REMARKS. A simple illustration is especially useful in describing antennas, their radiation patterns, and their scan formats.

### Block 7.a. TYPE

Antennas may be fixed or scanned. There are two basic types of scanning, electronic and mechanical. The term "electronic" denotes an antenna that does not physically move, but rather shifts the direction of the main beam by means of phasing, or switching antenna elements on and off. The movement of an antenna by a servomotor, or rotor, is mechanical scanning. Scanning may also be accomplished by a combination of these two types.

In addition to the type of scan, the scan format should also be entered in Block 7.a. Examples of some common scan formats are search while track, conical scan, box scan, raster scan, 3D, vertical only, and horizontal only. Thus, the complete entry in Block 7.a. might be electronic conical scan or mechanical vertical only scan. In some cases, both electronic and mechanical scanning may be used simultaneously. Rather than just state mechanical and electronic raster scan, full details should be given in Block 10, REMARKS. The same is true if two or more different types of scan formats are used or selected by the operator.

If the antenna, its main beam, and its side lobes do not move, enter "Fixed" in Block 7.a. If the antenna can be set up in only one orientation, such as vertical for a monopole whip antenna, the rest of the lines in Block 7 should be filled in with "NA." If the antenna mounting provides for various adjustments resulting in different antenna orientations, only Blocks 7.b.(3) and 7.c.(2) should be filled in with "NA." The questions in Blocks 7.b.(1), 7.b.(2), and 7.c.(1) are still applicable and should be answered.

#### Block 7.b. VERTICAL SCAN

If vertical scanning is possible, enter how the vertical portion of the scan is accomplished. If no vertical scanning is possible enter "Fixed." If the antenna does not scan but can move or be set up in different vertical orientations, the entry should be "None" and Blocks 7.b.(1) and 7.b.(2) should be completed.

Block 7.b.(1) MAX ELEV. This is the highest vertical angle to which the antenna can either scan or be set. If the antenna can scan or be set up to radiate/receive directly overhead, enter "+90 degrees" in Block 7.b.(1). If the antenna can scan only up to the horizon enter "0 degrees". As stated before, if the antenna does

not scan but can be changed in orientation, the maximum skyward adjustment should be entered in Block 7.b.(1).

Block 7.b.(2) MIN ELEV. This is the lowest vertical angle that the antenna can scan or be pointed and it ranges from directly overhead (+90 degrees) to straight down into the ground (-90 degrees). Even if the antenna does not scan, but can be set up or adjusted to different orientations, the minimum elevation should still be entered in Block 7.b.(2).

Block 7.b.(3) SCAN RATE. Two different rates are required. The first rate is the number of degrees of vertical scan per second. If the antenna performs a horizontal scan in one second as a part of a raster scan and drops down 1 degree after each sweep, it is scanning vertically at 1 degree /second. The second rate is the number of complete vertical scans per minute. If the same antenna completed a scan in 30 seconds, the vertical scan rate would be 2 scans/minute. Block 7.b.(3) should be divided in half, with the degrees/second rate listed first and the scans per minute rate last. A valid entry would be 1 degree/s, 2 scans/min.

If the antenna does not have a standard scan format, the scans/minute rate may not be applicable; only the degree/second rate need be entered. An "NA" in the second column of Block 7.b.(3) is not necessary. The degree/second rate is applicable to all antennas with any vertical scan and must be listed. Any other pertinent scan rates should also be given in Block 10, REMARKS.

Block 7.c. HORIZONTAL SCAN

Enter how the horizontal portion of the scan, if any, is accomplished. If no horizontal scanning is possible, the correct entry is "Fixed". If the antenna does not scan but can move or be set up in different horizontal orientations, enter "None" and complete Block 7.c.(1) SECTOR SCANNED.

Block 7.c.(1) SECTOR SCANNED. This is the portion of a circle describing an antenna's scan, or range of motion. An antenna that scans a complete circle has a 360 degree sector scan. If the antenna does not scan but its mounting enables it to move or be set up in different orientations, the maximum horizontal motion or adjustment should be entered.

Block 7.c.(2) SCAN RATE. Two different rates are required. The first rate is the number of degrees of horizontal scan per second. An antenna that sweeps a complete circle once every second has a sweep rate of 360 degrees/second. The second rate is the number of horizontal scans completed per minute. It is not necessarily the "Sector Scanned" figure times the "degrees/second". If a significant portion of time is spent vertically scanning in between horizontal sweeps, the horizontal scan rate would be lower. Both entries should be made on Block 7.c.(2). List the degrees/second rate first, and the scans per minute rate last.

If an antenna does not have a standard scan format, the scans/minute rate may not be applicable. In this case, only the degrees/second rate need be entered. An "NA" in the second column of Block 7.c.(2) is not necessary. The degrees per second rate is applicable to all antennas with any horizontal scan and must be listed.

### Block 7.d. SECTOR BLANKING

Indicate whether the antenna is capable of sector blanking. If yes, enter the details in Block 10, REMARKS.

**Common Errors:** Required information is not supplied.

## 8. GAIN

Block 8.a. MAIN BEAM

Enter the maximum gain in dBi. The gain to be provided is the power gain and must be given relative to an isotropic antenna. If the antenna efficiency is known, it should be provided in Block 10, REMARKS.

The main beam is that solid angle radiating away from the antenna in which the highest output power is transmitted or the maximum reception is possible. It encompasses the entire solid angle over which the output power is not less than 3 dB below the highest output power. For an ideal point source in free space, this is a 360 degree X 180 degree solid angle (a sphere). A dipole antenna main beam is 360 degrees X 78 degrees. A high-gain millimeter-wave parabolic

antenna may have a main beam solid angle of 0.25 degrees X 0.25 degrees. The main beam gain is the maximum gain within the main beam angle. Some antennas, such as full earth coverage antennas, may have complex patterns and no unique maximum. Such antennas must be explained in Block 10, REMARKS.

# Block 8.b. 1<sup>ST</sup> MAJOR SIDE LOBE

In addition to the main beam, most antennas also exhibit secondary areas of increased transmission or reception capability. These areas are called side lobes. Side lobes tend to fan out from the centerline of the main beam. The side lobe closest to the main beam is called the first side lobe. The farther away from the centerline of the main beam that a side lobe forms, the less will be its gain. Usually, the first side lobe has the most gain of any antenna side lobe. Also, antennas are normally designed to minimize side lobe gains. Two items of information about the first side lobe are required in Block 8.b. The first is the gain of the side lobe relative to the gain of the ideal isotropic antenna. The second is the angular position of the side lobe relative to the main beam. The angular position of a first side lobe that is at right angles to the main beam is 90 degrees. A first side lobe formed at the back of an antenna would have an angular position of 180 degrees. For example, the entry in Block 8.b might be "10 dBi at 38 degrees." If the antenna pattern is not symmetrical in the horizontal and vertical planes, the side lobe gain and side lobe position in both the planes should be given.

Quite often during antenna development, antenna radiation patterns are measured. These are essentially graphic outlines of the main beam and all the side lobes. If radiation patterns are available, they should be attached to the DD Form 1494. Radiation patterns on commercial antennas are often available from the equipment supplier or manufacturer.

**Common Errors:** Required information is not supplied.

#### More About: Antenna Radiation Pattern, Side lobes

## 9. BEAMWIDTH

The antenna main beam is that angle radiating away from the antenna in which the highest output power is transmitted or the maximum reception is possible. Main beam beamwidth is the angle over which the antenna gain is not less than 3 dB below the main beam gain. A high-gain millimeter-wave parabolic antenna may have a main beam width of 0.25 degrees X 0.25 degrees. When the angle is not centered on the main beam axis, the limits relative to the main-beam axis should be stated in the remarks section.

The radiation pattern of an antenna is a function of the amplitude and phase distribution of the current across the aperture. The aperture distribution or illumination controls the main beam gain, beamwidth, and side lobe levels. The aperture distributions are uniform, cosine, cosecant, and cosecant-squared.

Block 9.a. HORIZONTAL BEAMWIDTH

This is the angle in degrees between the 3-dB down points measured in a plane containing the main-beam axis and a horizontal line.

Block 9.b. VERTICAL BEAMWIDTH

This is the angle in degrees between the 3-dB down points measured in a plane containing the main-beam axis and a vertical line.

**Common Errors:** Confusing Antenna Beamwidth with Antenna Bandwidth.

### More About: Antenna Patterns

## **10. REMARKS**

This block may be used as a continuation of any of the preceding blocks. The continued block, or blocks, should be identified by block number. If any of the DD Form 1494 is classified, the security classification of each item of information contained in the REMARKS block must be <u>explicitly marked</u>.

Any additional information that clarifies the function or operation of the equipment may be entered in this block.

**Common Errors:** Not including a reference number in the REMARKS block for one or more of the numbered blocks on the form.

	CLASSIFICATION			PAGE					
APPLICATION FOR									
SPECTRUM REVIEW									
NTIA GENERAL INFORMATION									
1. APPLICATION TITLE									
2. SYSTEM NOMENCLATURE									
3. STAGE OF ALLOCATION (A One)									
a. STAGE 1 - CONCEPTUAL b. STAGE 2 - EXPERIMENTAL c. STAGE 3 - DEVELOPMENTAL d. STAGE 4 - OPERATIONAL   4. FREQUENCY REQUIREMENTS									
a. FREQUENCY (IES)									
b. EMISSION DESIGNATOR(S)									
5. PURPOSE OF SYSTEM, OPERATI	ONAL AND SYSTEM CONCEPT	S (WARTIME USE	) (X one)	a. YES b. NO					
6. INFORMATION TRANSFER REQU									
0. INFORMATION TRANSFER REQUIREMENTS									
7. ESTIMATED INITIAL COST OF TH	E SYSTEM								
8. TARGET DATE FOR			T						
a. APPLICATION APPROVAL	b. SYSTEM ACTIVATION	N C. SYSTEM TE		ERMINATION					
9. STOTEM RELATIONSHIP AND ES	SENTIALITY								
10. REPLACEMENT INFORMATION									
11. RELATED ANALYSIS AND TEST	DATA								
12. NUMBER OF MOBILE UNITS									
13 GEOGRAPHICAL AREA FOR									
a. STAGE 2									
b. STAGE 3									
c. STAGE 4									
14. LINE DIAGRAM		15. SPACE SYSTEMS							
(See Page(s)									
16. TYPE OF SERVICE(S) FOR STAG	iE 4	17. STATION CLASS(ES	) FOR STAGE	= 4					
18 REMARKS									
DOWNGRADING INSTRUCTIONS		CLASSIFICATION							

# NTIA GENERAL INFORMATION

# **1. APPLICATION TITLE**

Enter the Government's nomenclature of the equipment, or enter the manufacturer's name and model number, and a short descriptive title. If a JETDS nomenclature (AN nomenclature) has been assigned to the RF component being allocated, it should be listed in this block along with the descriptive name of the equipment (e.g., AN/TPQ-37 Artillery Locating Radar). If a JETDS nomenclature is not assigned, a unique descriptive title must be entered (e.g., Symbol Technologies Spectrum 24 Wireless). In the case of commercially produced, off-the-shelf equipment, the title should include the manufacturer's name and model number, in addition to a short descriptive title (e.g., SAVI Model STR7200 Hand-Held Interrogator, Reader Module).

If multiple equipment are included in the allocation request, this should be reflected in the Application Title block. Example: "M/A-2.5 TX and MODEL SUPER 2 MR RX" would be a correct entry. The manufacturer's name is not important in the case of competition sensitive equipment; it is perfectly acceptable to omit this information. Instead, the identifier "Model A" or "Contractor B" should be substituted.

**Note:** "Title" and "Nomenclature not yet assigned" are not acceptable entries.

**Note:** The Application Title blocks on the NTIA General Information and Foreign Coordination pages must be identical to the information listed on the DoD General Information page.

**Note:** This item must be identical to Block 1 on the DoD General Information page.

**Common Errors:** The JETDS nomenclature (if available) is not included in the title. The equipment manufacturer's name and model number are not always included in the title for COTS equipment.

### More About: JETDS

## 2. SYSTEM NOMENCLATURE

Enter the nomenclature of the system for which this equipment is a subsystem (e.g., Ammunition Automatic Identification Technology (AMMO AIT), Patriot or Global Positioning System).

The System Nomenclature is a title describing the system containing the allocated RF elements. In cases where an allocation is needed for a complex system, made up of several subsystems, a separate DD Form 1494 must be submitted for each separate subsystem. For example, the Airborne Adverse Weather Weapon System (AAWWS) has two distinct RF components: the RF Hellfire Modular Missile System and the Fire Control Radar. An application for allocation for the AAWWS would require one form for the RF Hellfire Modular Missile System and one for the Fire Control Radar. Both of the DD Form 1494s would list the system nomenclature as the AAWWS, but the entry in Block 1, Application Title, would be different, depending on the subsystem being allocated.

Note: Because the two applications are related, each application should show the J/F 12 number of the other application in Block 10b of Page 1.

For less complex systems, the Application Title and System Nomenclature may be identical. In this case, the same entry as in Block 1, APPLICATION TITLE, should be inserted. Leaving this block blank may lead to confusion as to the exact relationship, or lack thereof, between this allocation application and a similar one.

Equipment should be addressed on an "RF link" rather than "platform" basis. Normally, each link is considered separately. For example, an air-launched guided missile might have two RF links: a Telecommand (aircraft-to-missile) link and a Video (missile-to-aircraft) link. Each of these RF links should be addressed separately, and should include one transmitter, one receiver, and one, or more, antenna characteristics pages. The data on the technical characteristics pages should be related, as they all pertain to one RF link (one set of equipment designed to function together). If the equipment is allocated on a platform basis, each allocation would still have one, or more, transmitter, receiver, and antenna page, and the only relationship would be the physical location of the equipment. A missile allocation would have the video transmitter with its associated antenna; and the telecommand receiver with its associated antenna. Obviously, allocations on an RF link basis present a clearer picture than allocations on a platform basis and are, therefore, less confusing.

**Note:** This item must be identical to Block 2 on the DoD General Information page.

Common Errors: System nomenclature information is not provided.

# 3. STAGE OF ALLOCATION

Stages of allocation correspond to the stages of life cycle management. The amount and type of data required on the DD Form 1494 vary with the stage of allocation. Read the general NTIA description of each stage, given below, and mark the appropriate block.

**STAGE-1 - CONCEPTUAL.** Initial planning has been completed, including proposed frequency bands and other available characteristics. A conceptual allocation is required prior to releasing funds for studies, or assembling "proof-of-concept" test-beds. Planned or estimated details concerning the equipment should be entered on the DD Form 1494. Where information has not been determined, the entry "NAvail" or "Not Available" should be used. Stage-1 is for system planning and does not involve actual transmission or reception.

**STAGE-2 - EXPERIMENTAL.** The preliminary design has been completed; radiation, using test equipment or preliminary models, may be required. Prior to the release of funds, an experimental allocation is required for building a radiating test model, or assignment of a frequency for experimental usage, including, but not limited to, units tested within a laboratory. Estimated and calculated data can be used, where appropriate, for nearly all DD Form 1494 technical items.

**STAGE-3 - DEVELOPMENTAL.** The major design has been completed; radiation may be required. Prior to release of funds for engineering development models, a Stage-3 allocation must be obtained. All applicable items of the DD Form 1494 should be filled in with measured data. Where measured data are not available, calculated data can be substituted. An entry of "NAvail" or "Not Available" must include a reason.

**STAGE-4 - OPERATIONAL.** Development has been essentially completed and final operating constraints required to assure compatibility need to be identified. Prior to release of funds for production units, an operational frequency allocation is <u>mandatory</u>. Measured data for all technical characteristics such as emission bandwidth, harmonic level, spurious level, etc., should be provided when submitting a Stage-4 frequency allocation application.

All COTS equipment with unmodified RF characteristics fall into this type of allocation. Although the equipment may be used within experimental or developmental systems, by definition, it is operational equipment, because of its off-the-shelf status (for example, an off-the-shelf telemetry system used during development of a missile).

**Common Errors:** COTS equipment submitted as other thanStage-4. However, if COTs equipment is modified in any way, a Stage-2 or -3 application should be submitted.

More About: Data requirements for each stage

# 4. FREQUENCY REQUIREMENTS

4.a. Enter the required frequency band(s), indicating units, (e.g., kHz, MHz, or GHz). For equipment designed to operate only at a single frequency, enter this frequency and units.

4.b. Enter the emission designator(s), including the necessary bandwidth for each designator, as described in Chapter 9 of the NTIA manual (e.g., 40M0PON). For systems with a frequency hopping mode, identify each mode as "hopping" or "non-hopping" (e.g., 64M0F3E Hopping).

Enter other information pertinent to frequency requirements, such as minimum frequency separation or special relationships involving multiple discrete frequencies in Block 13, titled REMARKS.

Note: This item must be identical to Block 4 on the DoD General Information page.

**Common Errors:** Frequencies are not submitted. The frequency bands requested do not support the planned service. Emission designator is often left out. Lockouts have not been identified in the REMARKS section.

## More About: Emission Designators, Frequencies, Lockouts

# 5. PURPOSE OF SYSTEM, OPERATIONAL AND SYSTEM CONCEPTS

Enter a summary description of the function of the system or subsystem. Example: Collect and disseminate meteorological data using satellite techniques; Transmission of radar data for air traffic control (ATC); Remote control of ATC radars; Provide for the transmission and reception of digital voice and data by means of line-of-sight (LOS) or tropospheric modes of propagation; Provide navigational signals from which a broad spectrum of users are able to derive navigational data. Also include information on operational and system concepts. Mark whether the system has a wartime function.

**Common Errors:** Required information not supplied.

# 6. INFORMATION TRANSFER REQUIREMENTS

This block should contain a description of the types and forms of information to be transmitted or received. Examples include analog voice, digital voice, multichannel pulse code modulation (PCM) data, a tone, several coded-tone sequences, unmodulated radar pulses, pulse-compression radar pulses, standard video, etc. If several modulation techniques are to be employed, each one should be listed. If processing prior to final modulation is to be performed, such as spread spectrum modulation or multiplexing, this should be described. In the case of direct-sequence spread spectrum, this must include the chip rate, data rate, and noise-equivalent occupied bandwidth. The description of frequency-hopping spread spectrum must include the number of channels, the hop rate, the total hopped-frequency range, and whether channels or groups of channels can be "locked out" or avoided. If the equipment will transmit or receive several channels of information that have been multiplexed, specify the number of separated channels and the means of multiplexing (TDM, FDM, etc.). If the equipment transmits on several RF channels simultaneously, this should be described and the total number of RF channels listed.

**Note:** Data rates listed should be the maximum information data rate rather than nominal values. In addition, if several discrete data rates are available, this should be indicated and the discrete data rates listed.

**Common Errors:** The coded data rate is provided instead of the information rate.

More About: Frequency-hopping, Spread Spectrum

# 7. ESTIMATED INITIAL COST OF THE SYSTEM

This block is for information to show the general size and complexity of the system. It is not intended to be a determining factor in system review. For Stage-2, enter the research cost; forStage-3, enter the development cost; forStage-4, enter the unit cost of equipment and expected number of equipments/systems to be procured.

Example: 10 million dollars for 100 Operational Systems.

Common Errors: Required information not supplied.

# 8. TARGET DATA FOR

For the stage review requested, enter the appropriate dates. Funds must not be obligated prior to the approval of this application.

### Block 8.a. APPLICATION APPROVAL

The application approval date is the date by which approval of the DD Form 1494 is required. Generally, this is determined by the scheduled date for contract signing. Funds must not be obligated prior to the approval of this application. If foreign coordination is not required, then approximately one year must be allowed for application approval. If foreign coordination is required, allow approximately two years for application approval.

**Note**: In all cases, the application approval target date must predate the proposed target starting date for subsequent stages listed in Block 5 of the DoD General Information page.

### Block 8.b. SYSTEM ACTIVATION

The system activation date is the date when the system will be activated in the stage being allocated. This date must post date that of the application approval.

### Block 8.c. SYSTEM TERMINATION

The system termination date is that date when the current phase of the life-cycle model is scheduled to end. For a Stage-2 application, the date entered is the date when the system is scheduled to enter Stage-3. This date may be an estimate. For a Stage-4 application, the date entered should be the planned date when the system will no longer be used, or when a new system is expected to replace it.

**Common Errors:** Required information is not supplied. Application approval date is after the proposed target starting date for subsequent stages listed in Block 5 of the DoD General Information page.

## 9. SYSTEM RELATIONSHIP AND ESSENTIALITY

Enter the essentiality and a statement of the relationship between the proposed system and the operational function it is intended to support.

This item should identify required interfaces with other systems or platforms and provide a brief statement of how these interfaces are achieved. For example, a tactical radio might interface with the local telephone system through a Radio-Wire-Integration (RWI) control device. Such an interface, if planned, should be explained in this block. A basic justification for the existence of the system should be included.

**Common Errors:** Required information is not provided.

# **10. REPLACEMENT INFORMATION**

Identify existing system(s), which may be replaced by the proposed system. State any known additional frequency requirements.

No action to cancel frequency assignments or allocations will be taken based on the information provided.

Common Errors: Required information not provided.

# 11. RELATED ANALYSIS AND TEST DATA

Identify available reports that document previous electromagnetic compatibility (EMC) studies, predictions, analyses, or prototype EMC testing and that are relevant to the assessment of the system under review. In addition, identify any reports regarding frequency-assignment problems or algorithms.

Each reference should include the report number and title, the agency or contractor responsible for the report, the date of its publication, and downgrading (if classified). Reports currently in process, but not yet published, should be referenced.

**Note:** The availability of this information can reduce the time required to obtain spectrum certification.

**Common Errors:** Required information not supplied.

# **12. NUMBER OF MOBILE UNITS**

Enter the number of mobile units capable of operating from different locations.

**Common Errors:** Required information not supplied.

# 13. GEOGRAPHICAL AREA FOR

Enter the geographical location(s) or area(s) of use for this stage and subsequent stage(s). Provide geographical coordinates (degrees, minutes, seconds), if available. Enter "NA" for all accomplished stages.

### Stages-2 and -3

Whenever possible, list the specific sites identified by geographic location and coordinates. If a large number of US test sites are anticipated, but the final sites have not been chosen, it is acceptable to add "and various other test sites in the US&P." Specific locations should be given whenever possible.

All test sites OCONUS should be identified by base, station or military unit name, and the country.

### Stage-4

Operation locations should be as specific as possible. For fixed equipment, in most cases, it should be possible to list the site locations by latitude and longitude. For mobile systems, where no specific location can be identified, list all of the countries where standard coordination is to be accomplished. General, rather than specific, locations lead to time-consuming questions throughout the review process.

Use the REMARKS section or continuation pages when the space on the DD Form 1494 is too small to list all locations and specific sites.

The fewer locations where you need to coordinate, the less time required for approval. Specific geographic locations are needed for Japan and Korea.

**Note:** This item must be identical to Block 7 on the DoD General Information page.

**Common Errors:** Often organizations do not know where the system will be operated. Entering "worldwide" is not acceptable. Specific countries are needed.

## 14. LINE DIAGRAM

Enter the page number of the line diagram(s). The only acceptable entry in this block is "See page\_\_\_\_\_". Do not draw the line diagram on the form. A separate sheet with the line diagram containing appropriate security classification markings is <u>mandatory</u> for all allocation applications. If the line diagram is contained in an attached file (e.g., Line\_Diagram.pdf or Line\_Diagram.ppt), reference the file. A sample line diagram of the EchoPoint RFID system is shown in Figure 5.

The line diagram should show all of the RF links of the overall system or platform and how they interrelate. All RF links should be labeled with directions of transmission and frequency range. A detailed breakdown of internal components, i.e., a block diagram, is not desired. Multiple diagrams showing different interconnections or models are encouraged for complex systems.

**Common Errors:** Line diagram is not supplied.





# 15. SPACE SYSTEMS

Enter the page number of the space system data. Attach to this page a copy of the space system data as described in the NTIA Manual, Paragraph 10.7.3, Data Requirement.

**Common Errors:** Required information is not provided.

More About: NTIA Manual Paragraph 10.7.3
# 16. TYPE OF SERVICE(S) FOR STAGE 4

Enter the appropriate type of service(s) that applies, or will apply, to the equipment in the operational stage (Stage-4). If the service is not in accordance with the allocation tables, full justification must be entered.

The type of service should not be confused with the station class. The station class describes the function of the equipment. The services being referred to in this block are the International Telecommunication Union (ITU)-recognized services as they appear in national and international allocation tables. Examples are radio-location, aeronautical radionavigation, broadcasting satellite, fixed, mobile, aeronautical radionavigation-satellite, etc.

If more than one service is applicable, list all that apply. Thus, a radio might be used while in motion on the ground (mobile); it might also be permanently installed and used to communicate with other fixed radios (fixed); or, it might be installed in an aircraft (aeronautical mobile). All three service types should be listed. Some systems do not have a specific service, such as electronic warfare (EW). For those systems, the correct entry is "no specific service (EW)".

If the frequency band requested on this application is not allocated to this type of system, or if the system is multifunctional and the frequency band is not allocated to one of its types of service, then an explanation must be included in Block 18. This explanation must contain a justification for use of this band that is based on functional and operational requirements. Explanations involving cost, availability of commercial equipment, etc. are not acceptable. A frequency allocation will not be granted to an "out-of-band" system without a sufficient and valid explanation.

**Common Errors:** Required information is not supplied.

More About: Service Types

# 17. STATION CLASS(ES) FOR STAGE 4

Enter the appropriate station class(es).

This item refers to the station classes defined in the NTIA Manual Chapter 6; only the symbols listed in Chapter 6 are acceptable entries in this block. If several different station classes describe the equipment, all the descriptive station class symbols should be listed. Multiple code listings may be used when the equipment accomplishes more than a single function. The station class is the equipment purpose and function, and must fall within those classes related to the services listed in Block 16, TYPE OF SERVICE.

Consider the following rules when determining station class applicability:

1. All equipment, fixed or otherwise, transmitting to a mobile receiver is considered in the mobile service.

2. No general station classes exist for "transportable" equipment. Both "fixed" and "mobile" station classes applicable to the equipment should be listed in this block.

3. Several types of systems, such as electronic warfare and simulators, have no station class. For the station class of these systems, write "NA (EW)" or "NA (Simulator)".

**Common Errors:** Station class information is not provided.

#### More About: NTIA Station Classes

## 18. REMARKS

This block may be used as a continuation of any of the preceding blocks. The continued block, or blocks, should be identified by block number. If any of the DD Form 1494 is classified the security classification of each item of information contained in the REMARKS block must be <u>explicitly marked</u>.

Any additional information that clarifies the function or operation of the equipment may be entered in this block.

**Common Errors:** Not including a reference number in the REMARKS block for one or more of the numbered blocks on the form.

Any additional information that clarifies the function or operation of the equipment may be entered in this block.

**Common Errors:** Not including a reference number in the REMARKS block for one or more of the numbered blocks on the form.

	CLASSIFICATION		PAGE						
APPLICATION FOR FOREIGN									
FC	REIGN COORDINATION								
2 SYSTEM NOMENCI ATURE									
2. STSTEM NOMENCLATORE									
3. STAGE OF ALLOCATION (X one)									
a. STAGE 1 - CONCEPTUAL b. STA	GE 2 - EXPERIMENTAL	c. STAGE 3 - DEVELOPMENTAL	d. STAGE 4 - OPERATIONAL						
4. FREQUENCY REQUIREMENTS									
b. EMISSION DESIGNATOR(S)									
5. PROPOSED OPERATING LOCATIONS OUT	SIDE US&P								
6. PURPOSE OF SYSTEM, OPERATIONAL AN	ID SYSTEM CONCEPTS								
7. INFORMATION TRANSFER REQUIREMENT	S								
8. NUMBER OF UNITS OPERATING SIMULTA	NEOUSLY IN THE SAME	EENVIRONMENT							
		-							
9. REPLACEMENT INFORMATION									
10. LINE DIAGRAM		11. SPACE SYSTEMS							
(See Page(s))		(See Page(s))							
12. TROJECTED OF ERAHORAE DEFECTMEN									
13. REWARKS									
DOWNGRADING INSTRUCTIONS		CLASSIFICATION							

## FOREIGN COORDINATION GENERAL INFORMATION

# **1. APPLICATION TITLE**

Enter the Government's nomenclature of the equipment, or enter the manufacturer's name and model number, and a short descriptive title. If a JETDS nomenclature (AN nomenclature) has been assigned to the RF component being allocated, it should be listed in this block along with the descriptive name of the equipment (e.g., AN/TPQ-37 Artillery Locating Radar). If a JETDS nomenclature is not assigned, a unique descriptive title must be entered (e.g., Symbol Technologies Spectrum 24 Wireless). In the case of commercially produced off-the-shelf equipment, the title should include the manufacturer's name and model number, in addition to a short descriptive title (e.g., SAVI Model STR7200 Hand-Held Interrogator, Reader Module).

If multiple equipment are included in the allocation request, this should be reflected in the Application Title block. Example: "M/A-2.5 TX and MODEL SUPER 2 MR RX" would be a correct entry. The manufacturer's name is not important in the case of competition sensitive equipment; it is perfectly acceptable to omit this information. Instead, the identifier "Model A" or "Contractor B" should be substituted.

Note: "Title" and "Nomenclature not yet assigned" are not acceptable entries.

**Note:** The Application Title blocks on the NTIA General Information and Foreign Coordination pages must be identical to the information listed on the DoD General Information page.

**Note:** This item must be identical to Block 1 on the DoD General Information page.

**Common Errors:** The JETDS nomenclature (if available) is not included in the title. The equipment manufacturer's name and model number are not always included in the title for COTS equipment.

## 2. SYSTEM NOMENCLATURE

Enter the nomenclature of the system for which this equipment is a subsystem (e.g., Ammunition Automatic Identification Technology (AMMO AIT), Patriot, or Global Positioning System).

The System Nomenclature is a title describing the system containing the allocated RF elements. In cases where an allocation is needed for a complex system, made up of several subsystems, a separate DD Form 1494 must be submitted for each separate subsystem. For example, the Airborne Adverse Weather Weapon System (AAWWS) has two distinct RF components: the RF Hellfire Modular Missile System and the Fire Control Radar. An application for allocation for the AAWWS would require one form for the RF Hellfire Modular Missile System and one for the Fire Control Radar. Both of the DD Form 1494s would list the system nomenclature as the AAWWS, but the entry in Block 1, Application Title, would be different, depending on the subsystem being allocated.

Note: Because the two applications are related, each application should show the J/F 12 number of the other application in Block 10b of Page 1.

For less complex systems, the Application Title and System Nomenclature may be identical. In this case, the same entry as in Block 1, SYSTEM NOMENCLATURE, should be inserted. Leaving this block blank may lead to confusion as to the exact relationship, or lack thereof, between this allocation application and a similar one.

Equipment should be allocated on an "RF link" rather than "platform" basis. Normally, each link is allocated separately. For example, an air-launched guided missile might have two RF links: a Telecommand (aircraft-to-missile) link and a Video (missile-to-aircraft) link. Each of these RF links should be allocated separately, and should include one transmitter, one receiver, and one, or more, antenna characteristics pages. The data on the technical characteristics pages should be related, as they all pertain to one RF link (one set of equipment designed to function together). If the equipment is allocated on a platform basis, each allocation would still have one, or more, transmitter, receiver, and antenna page, and the only relationship would be the physical location of the equipment. A missile allocation would have the video transmitter with its associated antenna; and the telecommand receiver with its associated antenna. Obviously, allocations on an RF link basis present a clearer picture than allocations on a platform basis and are, therefore, less confusing.

**Note:** This item must be identical to Block 2 on the DoD General Information page.

**Common Errors:** System nomenclature information is not provided.

# 3. STAGE OF ALLOCATION

Stages of allocation correspond to the stages of life cycle management. The amount and type of data required on the DD Form 1494 vary with the stage of allocation. Read the general NTIA description of each stage, given below, and mark the appropriate block.

**STAGE-1 - CONCEPTUAL.** Initial planning has been completed, including proposed frequency bands and other available characteristics. A conceptual allocation is required prior to releasing funds for studies, or assembling "proof-of-concept" test-beds. Planned or estimated details concerning the equipment should be entered on the DD Form 1494. Where information has not been determined, the entry "NAvail" or "Not Available" should be used. Stage-1 is for system planning and does not involve actual transmission or reception.

**STAGE-2 - EXPERIMENTAL.** The preliminary design has been completed; radiation, using test equipment or preliminary models, may be required. Prior to the release of funds, an experimental allocation is required for building a radiating test model, or assignment of a frequency for experimental usage, including, but not limited to, units tested within a laboratory. Estimated and calculated data can be used, where appropriate, for nearly all DD Form 1494 technical items.

**STAGE-3 - DEVELOPMENTAL.** The major design has been completed; radiation may be required. Prior to release of funds for engineering development models, a Stage-3 allocation must be obtained. All applicable items of the DD Form 1494 should be filled in with measured data. Where measured data are not available, calculated data can be substituted. An entry of "NAvail" or "Not Available" must include a reason.

**STAGE-4 - OPERATIONAL.** Development has been essentially completed and final operating constraints required to assure compatibility need to be identified. Prior to release of funds for production units, an operational frequency allocation is <u>mandatory</u>. Measured data for all technical characteristics such as emission bandwidth, harmonic level, spurious level, etc., should be provided when submitting a Stage-4 frequency allocation application.

All COTS equipment with unmodified RF characteristics fall into this type of allocation. Although the equipment may be used within experimental or developmental systems, by definition, it is operational equipment, because of its off-the-shelf status (for example, an off-the-shelf telemetry system used during development of a missile).

**Common Errors:** COTS equipment submitted as other than Stage-4. However, if COTs equipment is modified in any way, a Stage-2 or -3 application should be submitted.

More About: Data requirements for each stage

# 4. FREQUENCY REQUIREMENTS

4.a. Enter the required frequency band(s), indicating units, (e.g., kHz, MHz, or GHz). For equipment designed to operate only at a single frequency, enter this frequency and units.

4.b. Enter the emission designator(s), including the necessary bandwidth for each designator, as described in Chapter 9 of the NTIA manual (e.g., 40M0PON). Identify each mode as "hopping" or "non-hopping" (e.g., 64M0F3E Hopping).

Enter other information pertinent to frequency requirements, such as minimum frequency separation or special relationships involving multiple discrete frequencies in Block 13, titled REMARKS.

**Note:** This item must be identical to Block 4 on the DoD General Information page.

**Common Errors:** Frequencies are not submitted. The frequency bands requested do not support the planned service. Emission designator is often left out. Lockouts have not been identified in the REMARKS section.

More About: Emission Designators, Frequencies, Lockouts

# 5. PROPOSED OPERATING LOCATIONS OUTSIDE THE US&P

Enter proposed Host Nation locations or areas of use. List each country and, if known, the geographical coordinates in degrees, minutes, and seconds. Japan and Korea require specific geographic locations. It generally requires less time to obtain host nation approval if the system will only operate in a limited geographic area.

**Note:** The fewer locations to coordinate, the less time required for approval.

**Common Errors:** Organizations often do not know where the system will be operated. Entering "worldwide" is not acceptable. Specific countries are needed.

#### More About: Country List

# 6. PURPOSE OF SYSTEM, OPERATIONAL AND SYSTEM CONCEPTS

Enter a brief summary description of the function of the system or subsystem. Example: Collect and disseminate meteorological data using satellite techniques; Transmission of radar data for ATC; Remote control of ATC radars; Provide for the transmission and reception of digital voice and data by means of LOS or tropospheric modes of propagation; Provide navigational signals from which a broad spectrum of users are able to derive navigational data. Also include information on operational and system concepts.

**Note:** This item must be identical to Block 5 on the NTIA General Information Page.

**Common Errors:** Required information not supplied.

## 7. INFORMATION TRANSFER REQUIREMENTS

This block should contain a description of the types and forms of information to be transmitted or received. Examples include analog voice, digital voice, multichannel PCM data, a tone, several coded-tone sequences, unmodulated radar pulses, pulse-compression radar pulses, standard video, etc. If several modulation techniques are to be employed, each one should be listed. If processing prior to final modulation is to be performed, such as spread spectrum modulation or multiplexing, this should be described. In the case of directsequence spread spectrum, this must include the chip rate, data rate, and noiseequivalent occupied bandwidth. The description of frequency-hopping spread spectrum must include the number of channels, the hop rate, the total hoppedfrequency range, and whether channels or groups of channels can be "locked out" or avoided. If the equipment will transmit or receive several channels of information that have been multiplexed, specify the number of separated channels and the means of multiplexing (TDM, FDM, etc.). If the equipment transmits on several RF channels simultaneously, this should be described and the total number of RF channels listed.

**Note:** Data rates listed should be the maximum information data rate rather than nominal values. In addition, if several discrete data rates are available, this should be indicated and the discrete data rates listed.

**Note:** This item must be identical to Block 6 on the NTIA General Information page.

**Common Errors:** The coded data rate is provided instead of the information rate.

#### More About: Frequency-hopping, Spread Spectrum

# 8. NUMBER OF UNITS OPERATING SIMULTANEOUSLY IN THE SAME ENVIRONMENT

Enter the total number of units planned to be operating simultaneously in the same environment during Stage-4 use.

The environment is defined as the physical area over which the equipment signal can communicate or interfere. For a fixed microwave radio, this will usually be two units, the transmitting station and the receiving station. For radars, there may be only a single radar in the environment. In the case of a base police radio net, there may be dozens of "hand-helds" within reception range of each other. When the number of systems operating may vary, a normal and a maximum value could also be given. The size of the area should also be stated (usually a radius value). The entry in Block 8 can never be "NA".

**Common Errors:** Data is not submitted.

## 9. REPLACEMENT INFORMATION

Identify existing system(s) which may be replaced by the proposed system. State any known additional frequency requirements.

No action to cancel frequency assignments or allocations will be taken based on the information provided.

Common Errors: Required information not provided.

#### **10. LINE DIAGRAM**

Enter the page number of the line diagram(s). The only acceptable entry in this block is "See page\_\_\_\_\_." Do not draw the line diagram on the form. A separate sheet with the line diagram containing appropriate security classification markings is <u>mandatory</u> for all allocation applications. If the line diagram is

contained in an attached file (e.g., Line\_Diagram.pdf or Line\_Diagram.ppt), reference the file. A sample line diagram of the EchoPoint RFID system is shown in Figure 6.

The line diagram should show all of the RF links of the overall system or platform and how they interrelate. All RF links should be labeled with directions of transmission and frequency range. A detailed breakdown of internal components, i.e., a block diagram, is not desired. Multiple diagrams showing different interconnections or models are encouraged for complex systems.

**Common Errors:** Line diagram is not supplied.

## 11. SPACE SYSTEMS

Enter the page number of the space system data. Attach an additonal page with the space system data as described in the NTIA Manual, Paragraph 10.7.3, Data Requirement.

Common Errors: Required information not provided.

#### More About: NTIA Manual Paragraph 10.7.3

## 12. PROJECTED OPERATIONAL DEPLOYMENT DATE

Enter the date the equipment will become operational at its foreign location. This date should be the same as or later than, the Stage-4 target starting date listed in Block 5.c of the DoD General Information page.

**Common Errors:** Required information not provided.

Figure 6



# 13. REMARKS

This block may be used as a continuation of any of the preceding blocks. The continued block, or blocks, should be identified by block number. If any of the DD Form 1494 is classified the security classification of each item of information contained in the REMARKS block must be <u>explicitly marked</u>. Any additional information that clarifies the function or operation of the equipment may be entered in this block.

**Common Errors:** Not including a reference number in the REMARKS block for one or more of the numbered blocks on the form.

	CLASSIFIC		ATION	DATE		Form Approved					
APPLICATION FOR EQUIPMENT			aggified	15/04/2004	OMB No. 0704-0			-0188			
FREQUENCY ALLOCATION		Unc	assilled	15/04/2004	P	AGE 1 O	<b>F</b> 6	PAGES			
L L L L L L L L L L L L L L L L L L L											
	D	OD GENERA		N							
то			FROM								
US Army Network Enterprise Technology Command Attn: James Thomas 2461 Eigenbeuger Aug. Suite 1204											
Alexandria VA 22331-2200			Lorton VA 22	2079-1442							
1. APPLICATION TITLE			201001, 1112								
EchoPoint Reader, Model SR-640-101											
2. SYSTEM NOMENCLATURE											
Automatic Identification Technology											
3. STAGE OF ALLOCATION (X one) X											
a. STAGE 1 - CONCEPTUAL b. STAGE	E 2 - EXPE	RIMENTAL	c. STAGE	3 - DEVELOPMENTAL	X d.	STAGE 4	- OPE	RATIONAL			
4. FREQUENCY REQUIREMENTS											
a. FREQUENCY(IES) 000.00 MHz											
b. EMISSION DESIGNATOR(S) 150K0F1D											
5. TARGET STARTING DATE FOR SUBSEQUEN	IT STAG	ES									
a. STAGE 2 b	D. STAGE	3		c. STAGE 4							
					15/0	4/2005					
6. EXTENT OF USE											
Continuous											
7. GEOGRAPHICAL AREA FOR											
a. STAGE 2 NA											
b. STAGE 3 NA											
US & P and those countries list	ted in R	emarks.									
8. NUMBER OF UNITS											
a. STAGE 2 b	D. STAGE	3		c. STAGE 4		1000					
						1000					
9. NUMBER OF UNITS OPERATING SIMULTANE	LUUSLI	IN THE SAM	200	• •							
10. OTHER J/F 12 APPLICATION NUMBER(S) TO	) BE		11. IS THERE A	NY OPERATIONAL R	EQUIREN	ENT AS	DESC	RIBED IN			
a. SUPERSEDED J/F 12/			THE INSTRUCTIONS FOR PARAGRAPH 11?								
X b RELATED J/F 12/ 745]			X a. YES	b. NC	)		c. NA	vail			
12. NAMES AND TELEPHONE NUMBERS											
Anne Scotti			703-33	9-4400 x 102	(2) AUTO	NA <sub>1</sub>	vail				
b. PROJECT ENGINEER			(1) COMMERCIAL		(2) AUTO						
Ginny Cook			703-33	9-4400 x 107	(_,	NA	vail				
13. REMARKS											
Item 7.c: Afghanistan, Djibouti, Austria, Belgium, Bahrain, Bosnia, Croatia, Egypt, FYROM, Germany, Hungary, Honduras, Iraq, Italy, Japan, Kuwait, Kosovo, Kyrgyzstan, Netherlands, Oman, Pakistan, Panama, Qatar, Republic of Korea, Saudi Arabia, Sweden, Switzerland, Turkey, UAE, United Kingdom, Uzbekistan.											
Item 11: All DoD components and all US Gov	vernme	nt Agencies									
DOWNGRADING INSTRUCTIONS			CLASSIFICATI	ON							
			Une	classified							
L											

CLASSIFICATION		PAGE
Unclassified		2 of 6
		2 01 0
TRANSMITTER EQUIPM	ENT CHARACTERISTICS	
1. NOMENCLATURE, MANUFACTURER'S MODEL NO.	2. MANUFACTURER'S NAME	
EchoPoint Reader, Model SR-640-101	Savi Technology, Inc.	
3. TRANSMITTER INSTALLATION Multiple sites and vehicles	4. TRANSMITTER TYPE FM communicat	ions
5. TUNING RANGE	6. METHOD OF TUNING	
00.00 MHz	SAW resona	tor
7. RF CHANNELING CAPABILITY	8. EMISSION DESIGNATOR(S)	
NA	150K0F1D	
9. FREQUENCY TOLERANCE		
1.57 ppm		
10. FILTER EMPLOYED (X one)		
X a. YES b. NO		
11. SPREAD SPECTRUM (X one)	12. EMISSION BANDWIDTH (X and comple	ete as applicable)
a. YES X b. NO		MEASURED
13. MAXIMUM BIT RATE	a3 dB 810 kHz	
30 kbps	b20 dB 2160 kHz	
14. MODULATION TECHNIQUES AND CODING	c40 dB NA	
FSK	d60 dB 32500 kHz	
	e. OC-BW 176.3 kHz	
	15. MAXIMUM MODULATION FREQUENC	Y
	30 kHz	
16. PRE-EMPHASIS (X one)	17. DEVIATION RATIO	
a. YES X b. NO	1.66	
	18. PULSE CHARACTERISTICS	
19. POWER	a. RATE NA	
a. MEAN 3.0 mW	b. width NA	
b. PEP NA	c. RISE TIME NA	
20. OUTPUT DEVICE	d. FALL TIME NA	
Transistor	e. COMP RATIO NA	
	21. HARMONIC LEVEL	
22. SPURIOUS LEVEL	a. 2ND 74	-1D -
-46 dBc	-/4	dBc
	b. 3RD 57	dD.
23. FCC TYPE ACCEPTANCE NO.	-30	Dabe
KL7-600R-V2	c. OTHER	UdD a
	-4/	abc
24. REMARKS		
Item 6: See attached SAW resonator spec sheet (66702101B.pdf,	433.92 MHz)	
Item 10: Type of filter: Band Pass		
In-band insertion loss: 5 dB		
Minimum 10 dB attenuation at 20.4 MHz removed from	the tuned frequency.	
CLASSIFICATION		
TT 1 '0' 1		
Unclassified		

CLASSIFICATION						PAGE				
			Unclas	sified			3 of 6			
			RECEIVE	ER EQUIPME	NT CHARACTERISTICS					
1. NOMENCLATURE, MAN	NUFACTUR	ER'S MOI	DEL NO.		2. MANUFACTURER'S N	NAME				
EchoPoint Reader, Mode	el SR-640-	101			Savi Technology, Inc.					
3. RECEIVER INSTALLAT	ION				4. RECEIVER TYPE					
Multi	iple sites a	nd vehicl	es		Single c	conversion superheterodyr	ne			
5. TUNING RANGE	000 00 1				6. METHOD OF TUNING					
	000.00 N	AHZ				SAW resonator				
7. RF CHANNELING CAP					8. EMISSION DESIGNAT	I OR(5)				
					150K0F1D					
3. TREGOLIGITOLERA	1.57 pr	om								
10. IF SELECTIVITY	107 Pr		2ND	3RD	11. RF SELECTIVITY (X)	and complete as applicable)				
a3 dB	460 kH	Iz			CALCULATED		)			
	752 1.1	T-			a3 dB 810 kHz					
b20 dB	/ 35 KI	12			<b>b20 dB</b> 2160 kHz					
a 60 dB	10211	Ц <sub>7</sub>			c60 dB 32500 kH	Z				
C00 0B	1051 K	пг			d. PRESELECTION TYP	PE				
12. IF FREQUENCY										
a. 1ST 10.7 MHz					13. MAXIMUM POST DETECTION FREQUENCY					
b. 2ND					14. MINIMUM POST DETECTION FREQUENCY					
c. 3RD					NA					
15. OSCILLATOR TUNED		1ST	2ND	3RD	16. MAXIMUM BIT RATE					
a. ABOVE TUNED FREQUENCY					17. SENSITIVITY	30 kbps				
b. BELOW TUNED FREQUENCY		×			a. SENSITIVITY	-97	dBm			
c. EITHER ABOVE OR BI TUNED FREQUENCY	ELOW				b. CRITERIA	20 dB S/N				
18. DE-EMPHASIS (X one)					c. NOISE FIG	4.5	dB			
a. YES		X b.	NO		d. NOISE TEMP	NA	Kelvin			
19. IMAGE REJECTION	26.11				20. SPURIOUS REJECTI	ON CON				
	26 di	3				56 dB				
Item 6: See SAW resona	tor spec sł	neet (667(	)2102A.p	df, 433.92 N	ИНz)					
CLASSIFICATION										
			Unclas	sified						

CLASSIFICATION	PAGE
Unclassified	4 of 6
1a. TRANSMITTINGb. RECEIVING	C. TRANSMITTING AND RECEIVING
2. NOMENCLATURE, MANUFACTURER'S MODEL NO.	3. MANUFACTURER'S NAME
EchoPoint Reader, Model SR-640-101	Savi Technology, Inc.
4. FREQUENCY RANGE	5. TYPE
000.00 MHz	Tunable loop antenna
6. POLARIZATION	7. SCAN CHARACTERISTICS
Linear	a. TYPE Fixed
8. GAIN	b. VERTICAL SCAN
a. MAIN BEAM	(1) MAX ELEV
2 dBi	
b. 1ST MAJOR SIDE LOBE	(2) MIN ELEV NA
NA	(3) SCAN RATE NA
9. BEAMWIDTH	c. HORIZONTAL SCAN
a. HORIZONTAL	(1) SECTOR SCANNED
360 deg	(2) SCAN RATE
b. VERTICAL	NA
30 deg	d. SECTOR BLANKING (X one)
	(1) YES X (2) NO
CLASSIFICATION	
Unclassified	

	CLASSIFICATION PAGE										
APPLICATION FOR SPECTRUM REVIEW		Une	5 of 6								
NTIA GENERAL INFORMATION											
1. APPLICATION TITLE EchoPoint Reader, Model SR-640	-101										
2. SYSTEM NOMENCLATURE											
Automatic Identification Technolo	ogy										
3. STAGE OF ALLOCATION (X one)	_			_			1				
a. STAGE 1 - CONCEPTUAL	b. STAGE 2	- EXPERIMENTAL		c. STAGE 3 - DEVEL	OPMENTAL	X	d. STAGE 4 - OPERATIONAL				
4. FREQUENCY REQUIREMENTS											
b EMISSION DESIGNATOR(S)	.00 MHZ										
150	K0F1D										
5. PURPOSE OF SYSTEM, OPERAT	IONAL AND S	YSTEM CONCEPT	s	(WARTIME USE	<b>)</b> (X one)	X	a. YES b. NO				
Provides visibility of assets includ	ling ordinanc	e by reporting loc	ation	and status of assets	s by use of w	vireles	s communications				
Trovides visionity of assets merud	ing oraniane	e by reporting loca	ution	and status of assets	<i>b b y use or v</i>	vii eiee	ss communications.				
PCM data up to 30 kbps, Manches	ster data up to	o 27.7 kbps									
7. ESTIMATED INITIAL COST OF TH \$500k	IE SYSTEM										
8. TARGET DATE FOR											
a. APPLICATION APPROVAL	b.	SYSTEM ACTIVATION	N		c. SYSTEM T	ERMIN	ATION				
01/04/2005		01/0	4/200	5		0	01/04/2010				
9. SYSTEM RELATIONSHIP AND ES	SENTIALITY										
These RF transmitters (readers), c	ommunicate	information to trai	nspon	ders located on ass	sets in order	to tra	ck principle assets in				
facility or in transit.											
10. REPLACEMENT INFORMATION											
This system replaces manual proce	esses within t	racking, inventory	y, and	manifesting system	ns.						
11. RELATED ANALYSIS AND TEST	DATA										
NA											
12. NUMBER OF MOBILE UNITS											
2 STAGE 2 NA											
b. STAGE 3 NA											
c. STAGE 4 US & P and those of	countries liste	ed in Remarks.									
			15 0								
(See Page(s) see attached file	e echo line a	liagram.ppt	15. 5	See Page(s) N	A						
16. TYPE OF SERVICE(S) FOR STAG	GE 4	and branning pr	17. S	TATION CLASS(ES	) FOR STAGE	E 4					
Fixed & Mobile			FX I	МОР							
18. REMARKS											
Item 13.c: Afghanistan, Djibouti,	Austria, Belg	ium, Bahrain, Bos	snia, (	Croatia, Egypt, FY	ROM, Germ	nany, l	Hungary, Honduras, Iraq,				
Italy, Japan, Kuwait, Kosovo, Kyr	rgyzstan, Net	herlands, Oman, F	Pakista	an, Panama, Qatar,	Republic of	f Kore	a, Saudi Arabia, Sweden,				
Switzerland, Turkey, UAE, United	d Kingdom, U	Jzbekistan.									
DOWNGRADING INSTRUCTIONS			CLAS	SIFICATION							
				Unclessified							
				Unclassified	L						

	CLASSIFICATION			PAGE
APPLICATION FOR FOREIGN SPECTRUM SUPPORT		Unclassified		6 of 6
FC	 	N GENERAL INFORM		
1. APPLICATION TITLE EchoPoint Reader, Model SR-640-101				
2. SYSTEM NOMENCLATURE Automatic Identification Technology				
3. STAGE OF ALLOCATION (X one)				
a. STAGE 1 - CONCEPTUAL b. STA 4. FREQUENCY REQUIREMENTS	GE 2 - EXPERIMENTAL	c. STAGE 3 - D	EVELOPMENTAL	X d. STAGE 4 - OPERATIONAL
a. FREQUENCY(IES)   000.00 MHz     b. EMISSION DESIGNATOR(S)   150K0F1D				
5. PROPOSED OPERATING LOCATIONS OUT	SIDE US&P			
See Remarks.				
6. PURPOSE OF SYSTEM, OPERATIONAL AN	ID SYSTEM CONCEPT	S		
Provides visibility of assets including ordin	ance by reporting loc	ation and status of a	ssets by use of w	vireless communications.
7. INFORMATION TRANSFER REQUIREMENT	S			
PCM data up to 30 kbps, Manchester data u	p to 27.7 kbps			
8. NUMBER OF UNITS OPERATING SIMULTA	NEOUSLY IN THE SAM	IE ENVIRONMENT		
200				
<b>9. REPLACEMENT INFORMATION</b> This system replaces manual processes with	nin tracking, inventor	y and manifesting sy	/stems.	
<b>10. LINE DIAGRAM</b> (See Page(s)) see attached file echo lin	ne diagram.ppt	11. SPACE SYSTEM (See Page(s))	IS N/A	
12. PROJECTED OPERATIONAL DEPLOYMEN	IT DATE		10/11	
13 REMARKS				
Item 5: Afghanistan, Djibouti, Austria, Belg Italy, Japan, Kuwait, Kosovo, Kyrgyzstan, I Switzerland, Turkey, UAE, United Kingdor	gium, Bahrain, Bosnia Netherlands, Oman, P n, Uzbekistan.	a, Croatia, Egypt, FY akistan, Panama, Q	ROM, German atar, Republic of	y, Hungary, Honduras, Iraq, SKorea, Saudi Arabia, Sweden,
DOWNGRADING INSTRUCTIONS		CLASSIFICATION		
		<u>म</u> न्द्र 1	: C . J	
		Unclass	intea	
1				

		CLASSIFICATION		DATE			Form Approved				
APPLICATION FOR EQUIPMENT		Unclassified		15/04/2004			OMB No. 0704-0188			0188	
FREQUENCY ALLOCATION		Unclassified		1	13/04/2004		PAGE 1	I OF	7	PAGES	
The public reporting burden for this collection of information is estimated to average 24 hours per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including											
suggestions for reducing the burden, to Department of Defense, Washington Headquarters Services, Directorate for Information Operations and Reports (0704-0188), 1215 Jefferson Davis Highway, Suite 1204, Arlington, VA 22202-4302. Respondents should be aware that notwithstanding any other provision of law, no person shall be subject to any penalty for failing to comply with a collection of											
PLEASE DO NOT RETURN YOUR FORM TO THE ABOVE ADDRESS. RETURN COMPLETED FORM TO THE USING AGENCY OR CONTRACTING AGENCY, AS											
DOD GENERAL INFORMATION											
то			FRO	M							
US Army Network Enterprise Technology Command PM Automatic Identification Technology (AIT)											
Attn: James Thomas			8580	Cinder E	Bed Road						
2461 Eisenhower Ave., Suite 1204			Suite	e 1400	070 144	•					
Alexandria, VA 22331-2200			Lorto	on, VA 22	2079-144	2					
1. APPLICATION IIILE Symbol Technologies Spectrum 24 Wireless											
2 SYSTEM NOMENCI ATURE											
Automatic Identification Technology											
3. STAGE OF ALLOCATION (X one) X											
a. STAGE 1 - CONCEPTUAL b. STAG	E 2 - EXPE	RIMENTAL		c. STAGE	3 - DEVEL	OPMENTAL	×	d. STAG	9E 4 - C	OPER.	ATIONAL
4. FREQUENCY REQUIREMENTS											
a. FREQUENCY(IES) 000.00 MHz											
b. EMISSION DESIGNATOR(S) 22M0G1D											
5. TARGET STARTING DATE FOR SUBSEQUE	NT STAG	ES									
a. STAGE 2	b. STAGE	3				c. STAGE 4					
							1:	5/04/20	05		
6. EXTENT OF USE											
Continuous											
7. GEOGRAPHICAL AREA FOR											
a. STAGE 2 NA											
	. 1.' D	1									
US& P and those countries list	ted in Re	marks.									
8. NUMBER OF UNITS											
a. STAGE 2	b. STAGE	3	NTA			c. STAGE 4		1000			
					NT.			1000			
3. NOMBER OF ONTS OF ERATING SIMULTAN	LOUGET		50		• •						
10. OTHER J/F 12 APPLICATION NUMBER(S) TO	O BE		11. IS THERE ANY OPERATIONAL REQUIREMENT AS DESCRIBED IN							IBED IN	
a. SUPERSEDED J/F 12/			Т		UCTIONS	FOR PARAG	RAPH	11?			
X b RELATED J/F 12/ 7771/1				a. YES		X b. NO			C.	NAva	ail
12. NAMES AND TELEPHONE NUMBERS			(1) CC				(2) A11	TOVON			
Anne Scotti			(1) 00	703-33	- 9-4400 x	102	(2) AU	N	JAvai	1	
b. PROJECT ENGINEER			(1) CC	MMERCIAL	-		(2) AU	TOVON		-	
Ginny Cook				703-33	9-4400 x	107		1	VAvai	il	
13. REMARKS											
Item 7.c: Germany, Japan, ROK, Brazil, Colu	umbia, H	Ionduras, N	icarag	gua, Panar	na, Peru,	Venezuela,	Austr	alia, Ne	w Zea	alanc	1,
Kuwait, Saudi Arabia, and Egypt.											
DOWNGRADING INSTRUCTIONS			CLAS	SIFICATI	ON						
				I Tra	alassifiad						
				Ulle	ciassified						
	וחח										

CLASSIFICATION		PAGE				
Unclassified		2 of 7				
TRANSMITTER FOLLIPM	ENT CHARACTERISTICS					
1. NOMENCLATURE, MANUFACTURER'S MODEL NO.	2. MANUFACTURER'S NAME					
PDT8046, AP4131	Symbol					
3. TRANSMITTER INSTALLATION Multiple Sites	4. TRANSMITTER TYPE PSK and CCK Commu	unications				
5. TUNING RANGE	6. METHOD OF TUNING					
000.00 MHz to 000.00 MHz	Syn	thesizer				
7. RF CHANNELING CAPABILITY	8. EMISSION DESIGNATOR(S)					
See Remarks	22M0G1D					
25 ppm	-					
10. FILTER EMPLOYED (X one)						
X a. YES b. NO						
13. MAXIMUM BIT RATE	a -3 dB 10.5 MHz	WEASURED				
11 Mbps	b20 dB 18 MHz					
14. MODULATION TECHNIQUES AND CODING	c40 dB 36 MHz					
DBPSK: 1 Mbps; DQPSK: 2 Mbps	d60 dB 50 MHz					
CCK 5.5 Mbps & 11 Mbps	е. ос-вw 22 MHz					
	15. MAXIMUM MODULATION FREQUENC	Y				
	NA					
	18. PULISE CHARACTERISTICS					
19. POWER	a. RATE NA					
a. MEAN 100 mW	b. WIDTH NA					
b. PEP NA	c. RISE TIME NA					
20. OUTPUT DEVICE	d. FALL TIME NA					
Transistor	e. COMP RATIO NA					
	21. HARMONIC LEVEL					
	a. 2ND	61				
-40 dB	b. 3RD					
23. FCC TYPE ACCEPTANCE NO.		61				
H9PLA4121 for PDT8046, H9PLA4131M for AP4131	c. OTHER	61				
24. REMARKS						
Item 1: PDT8046 Portable Data Collection Terminal AP-4131 RF Access Point Base Station						
Item 7. IEEE 802.11b frequency plan channels 1-14 will be utilized between channels 1-13 is 5 MHz. Channel spacing between	ed following host nation frequency restrict n channel 13 & 14 is 12 MHz.	ctions. Frequency spacing				
Item 10: The SAW has a -20 dBc bandwidth of 25 MHz. A front end RF filter is employed that has a -20 dBc bandwidth of 125 MHz.						
Item 11: Direct Sequence spreading with a chip rate of 11 Mcps.						
CLASSIFICATION						
Unclassified						

CLASSIFICATION						PAGE				
			Unclas	sified			3 of 7			
			RECEIVE	R EQUIPME	NT CHARACTERISTICS					
1. NOMENCLATURE, MAN PDT8046, AP4131	UFACTURI	ER'S MOE	DEL NO.		2. MANUFACTURER'S NA Symbol	AME				
3. RECEIVER INSTALLATI	<b>ON</b> Multiple S	Sites			4. RECEIVER TYPE Single Co	onversion Superhetero	dvne			
5. TUNING RANGE					6. METHOD OF TUNING					
000.	00 MHz to	00.000	MHz		Syn	thesizer				
7. RF CHANNELING CAPA	BILITY				8. EMISSION DESIGNAT	OR(S)				
	See Rema	arks			22M0G1D					
9. FREQUENCY TOLERAN	25 ppn	n								
10. IF SELECTIVITY	1ST		2ND	3RD	11. RF SELECTIVITY (X a	nd complete as applicabl	e)			
a3 dB	21.6 MI	Hz					ED			
b20 dB	27.3 MI	Hz			a3 dB 10.5 MHz					
					b20 dB 18 MHz					
c60 dB	48 MH	Iz			c60 dB 50 MHZ	F				
12. IF FREQUENCY					SAW Filte	- r				
274347					13. MAXIMUM POST DET	ECTION FREQUENCY				
a. 1ST 3/4 MHz					_	11 MHz				
b. 2ND					14. MINIMUM POST DETECTION FREQUENCY					
c. 3RD										
15. OSCILLATOR TUNED		1ST	2ND	3RD	16. MAXIMUM BIT RATE					
a. ABOVE TUNED FREQUENCY					17. SENSITIVITY	11 Mbps				
b. BELOW TUNED FREQUENCY		×			a. SENSITIVITY	-76	dBm			
c. EITHER ABOVE OR BE TUNED FREQUENCY	LOW				b. CRITERIA	20 dB S/N				
18. DE-EMPHASIS (X one)			•	·	c. NOISE FIG	6	dB			
a. YES	[	X b.	NO		d. NOISE TEMP	NA	Kelvin			
19. IMAGE REJECTION					20. SPURIOUS REJECTIO	DN .				
	50 dB	3				50 dB				
21. REMARKS   Item 1: PDT8046 Portable Data Collection Terminal AP-4131 RF Access Point Base Station   Item 7: IEEE 802.11b frequency plan channels 1-14 will be utilized following host nation frequency restrictions. Frequency spacing between channels 1-13 is 5 MHz. Channel spacing between channel 13 & 14 is 12 MHz.   CLASSIFICATION										
			Unclas	sified						

CLASSIFICATION	PAGE	
TT 1 '0' 1	A _ C 7	
Unclassified	4 OI /	
	IT CHARACTERISTICS	
1a. TRANSMITTINGb. RECEIVING	C. TRANSMITTING AND RECEIVING	3
2. NOMENCLATURE, MANUFACTURER'S MODEL NO.	3. MANUFACTURER'S NAME	
PDT8046	Symbol Technologies, Inc.	
	Januartad E	
0000 MHZ - 0000 MHZ	Inverted F	
6. POLARIZATION	7. SCAN CHARACTERISTICS	
vertical	a. TYPE Fixed	
8. GAIN	b. VERTICAL SCAN	
	NA NA	
0 dB1	(2) MIN ELEV	
b. 1ST MAJOR SIDE LOBE	NA	
NA	(3) SCAN RATE	
	NA	
9. BEAMWIDTH	C. HORIZONTAL SCAN	
	NA	
360 deg	(2) SCAN RATE	
b. VERTICAL	NA	
15 deg	d. SECTOR BLANKING (X one)	
	(1) YES X (2) NO	
Item 2: For AD 4121 second point		
item 2. For AF-4151 access point.		
CLASSIFICATION		
Unclassified		
Chelubbilled		

CLASSIFICATION					PAGE
	Unclassified				5 of 7
	ANTENNA EQUIPMEN	IT CHARACTERISTICS			
1a. TRANSMITTING	b. RECEIVING		X C.	TRAN	SMITTING AND RECEIVING
2. NOMENCLATURE, MANUFACTURER'S MO	DEL NO.	3. MANUFACTURER'	SNAME		
ML-2499-APA2-01		Symbol Technologie	es, Inc.		
			,		
4. FREQUENCY RANGE		5. TYPE			
0000 MHz - 0000 MHz		Dipole Omnidired	ctional		
6. POLARIZATION		7. SCAN CHARACTE	RISTICS		
Linear		a. TYPE Fixed			
8. GAIN		b. VERTICAL SCAN			
a. MAIN BEAM		(1) MAX ELEV			
+2 dBi				NA	4
		(2) MIN ELEV		N	•
b. 1ST MAJOR SIDE LOBE				IN A	A
NA		(3) SCAN RATE		N	Δ
9. BEAMWIDTH		c HORIZONTAL SCA	N	1 12	
a. HORIZONTAL		(1) SECTOR SCAL	NNED		
360 dag		(,,		NA	A
500 deg		(2) SCAN RATE			
b. VERTICAL				NA	ł
70 deg		d. SECTOR BLANKIN	IG (X one)		
		X (1) YES			(2) NO
item 2: For AP-4131 access point.					
ULAJJIFICA HUN					

Unclassified

	CLASSIFI	CATION			PAGE
APPLICATION FOR SPECTRUM REVIEW	Unclassified		6 of 7		
			AL INFORMATION		
1. APPLICATION TITLE Symbol Technologies Spectrum24	Wireless				
2. SYSTEM NOMENCLATURE Automatic Identification Technolog	gy				
3. STAGE OF ALLOCATION (X one)					
a. STAGE 1 - CONCEPTUAL	b. STAGE	2 - EXPERIMENTAL	c. STAGE 3 - DEVEL	OPMENTAL	X d. STAGE 4 - OPERATIONAL
4. FREQUENCY REQUIREMENTS					
a. FREQUENCY(IES) 000	00 MHz				
b. EMISSION DESIGNATOR(S) 22M	10G1D				
5. PURPOSE OF SYSTEM, OPERATI	IONAL AND	SYSTEM CONCEPT	S (WARTIME USE	<b>:)</b> (X one)	X a. YES b. NO
Provide a wireless data communica shipping/receiving and distribution	ation system 1.	m for distributed da	ata collection in manager	ment of inve	ntory, logistic, and material
6. INFORMATION TRANSFER REQU Direct Sequence, High Rate, 11 M	IREMENTS bps				
7. ESTIMATED INITIAL COST OF TH \$500k	E SYSTEM				
8. TARGET DATE FOR					
a. APPLICATION APPROVAL	t	D. SYSTEM ACTIVATIO	N 04/2005	c. SYSTEM T	
		01/0 V	J4/2005		01/04/2010
9. STSTEM RELATIONSHIP AND ES		T 1 1 1			
accomplished.	ion to the n	lanual barcode proc	cedure. It will increase th	le efficiency	of the tasks to be
10. REPLACEMENT INFORMATION NA					
11. RELATED ANALYSIS AND TEST	DATA				
<b>12. NUMBER OF MOBILE UNITS</b> 20					
13. GEOGRAPHICAL AREA FOR					
a. STAGE 2 NA					
b. STAGE 3 NA					
c. STAGE 4 US& P and those co	ountries lis	ted in Remarks.			
14. LINE DIAGRAM			15. SPACE SYSTEMS		
(See Page(s) see attached file	e line_diagi	ram.ppt	(See Page(s) N	A	
16. TYPE OF SERVICE(S) FOR STAG	6E 4		17. STATION CLASS(ES	) FOR STAG	Ξ4
			FAMO		
Item 13.c: Germany, Japan, ROK, Brazil, Columbia, Honduras, Nicaragua, Panama, Peru, Venezuela, Australia, New Zealand, Kuwait, Saudi Arabia, and Egypt.					
DOWNGRADING INSTRUCTIONS CLASSIFICATION					
Unclassified					

		PACE		
APPLICATION FOR FOREIGN SPECTRUM SUPPORT	Unclassified	7 of 7		
FC	REIGN COORDINATION GENERAL INFO	RMATION		
1. APPLICATION TITLE Symbol Technologies Spectrum24 Wireless				
2. SYSTEM NOMENCLATURE Automatic Identification Technology				
3. STAGE OF ALLOCATION (X one)				
a. STAGE 1 - CONCEPTUAL b. STA	GE 2 - EXPERIMENTAL C. STAGE 3 -	DEVELOPMENTAL X d. STAGE 4 - OPERATIONAL		
4. FREQUENCY REQUIREMENTS				
a. FREQUENCY(IES) 0000 MHz				
b. EMISSION DESIGNATOR(S) 22M0G1D				
5. PROPOSED OPERATING LOCATIONS OUT	SIDE US&P			
Germany, Japan, ROK, Brazil, Columbia, F Arabia, and Egypt.	Ionduras, Nicaragua, Panama, Peru, Ve	enezuela, Australia, New Zealand, Kuwait, Saudi		
6. PURPOSE OF SYSTEM, OPERATIONAL AN	ID SYSTEM CONCEPTS			
Provide a wireless data communication syst shipping/receiving and distribution.	em for distributed data collection in ma	anagement of inventory, logistic, and material		
7. INFORMATION TRANSFER REQUIREMENT	S			
Direct Sequence, High Rate, 11 Mbps				
8. NUMBER OF UNITS OPERATING SIMULTA	NEOUSLY IN THE SAME ENVIRONMENT			
This system replaces manual processes with	in tracking, inventory, and manifesting	g systems.		
10. LINE DIAGRAM	11. SPACE SYSTE	EMS		
(See Page(s)) see attached file line_dia	gram.ppt (See Page(s))	N/A		
12. PROJECTED OPERATIONAL DEPLOYMEN	IT DATE			
13. REMARKS				
DOWNGRADING INSTRUCTIONS				
	Uncla	ssified		
1				

#### MINIMUM DATA REQUIREMENTS FOR EACH STAGE OF ALLOCATION FOR THE DD FORM 1494

The following tables provide the minimum data requirements for each stage of allocation for the DD Form 1494.

DoD General Information Page				
Item No.	Item Name	Requirement		
	CLASSIFICATION	All Stages		
	DATE	All Stages		
	PAGE	All Stages		
	ТО	All Stages		
	FROM	All Stages		
1.	APPLICATION TITLE	All Stages		
2.	SYSTEM NOMENCLATURE	All Stages		
3.	STAGE OF ALLOCATION	All Stages		
4.a.	Frequency(ies)	All Stages		
4.b.	Emission Designator	Stage-2 (Estimated)		
		Stage-3 (Calculated)		
		Stage-4 (Measured)		
5.	TARGET DATE FOR SUBSEQUENT STAGES	All Stages		
6.	EXTENT OF USE	All Stages		
7.	GEOGRAPHICAL AREA FOR	All Stages		
8.	NUMBER OF EQUIPMENTS	All Stages		
9.	NUMBER OF EQUIPMENTS OPERATING	All Stages		
	SIMULTANEOUSLY IN THE SAME			
	ENVIRONMENT			
10.	OTHER J/F 12 APPLICATION NUMBER(S) TO BE:	All Stages		
11.	IS THERE ANY OPERATIONAL REQUIREMENT	All Stages		
	AS DESCRIBED IN THE INSTRUCTIONS FOR			
	PARAGRAPH 11?			
12.	NAMES AND TELEPHONE NUMBERS	All Stages		
13.	REMARKS	All Stages (As Required)		
	DOWNGRADING INSTRUCTIONS	All Stages		
	CLASSIFICATION	All Stages		

TRANSMITTER EQUIPMENT CHARACTERISTICS				
Item No.	Item Name	Requirement		
	CLASSIFICATION	All Stages		
	PAGE	All Stages		
1.	NOMENCLATURE, MANUFACTURER'S MODEL	Stage-2 (Interim Designation)		
	NO.	Stages-3, -4		
2.	MANUFACTURER'S NAME	Stages-2, -3, -4		
3.	TRANSMITTER INSTALLATION	All Stages		
4.	TRANSMITTER TYPE	Stages-1, -2 (Specification)		
		Stages-3, -4		
5.	TUNING RANGE	Stage-1 (Specification)		
		Stages-2, -3, -4		

TRANSMITTER EQUIPMENT CHARACTERISTICS				
Item No.	Item Name	Requirement		
6.	METHOD OF TUNING	Stage-2 (Specification)		
		Stages-3, -4		
7.	RF CHANNELING CAPABILITY	Stage-2 (Specification)		
		Stages-3, -4		
8.	EMISSION DESIGNATOR(S)	Stage-2 (Estimated)		
		Stage-3 (Calculated)		
		Stage-4 (Measured)		
9.	FREQUENCY TOLERANCE	Stage-2 (Specification)		
		Stages-3, -4		
10.	FILTER EMPLOYED	Stage-2 (Specification)		
		Stages-3, -4		
11.	SPREAD SPECTRUM	All Stages		
12.	EMISSION BANDWIDTH	Stage-2 (Specification)		
		Stage-3 (Calculated)		
10		Stage-4 (Measured)		
13.		Stage-2 (Specification)		
		Stages-3, -4		
14.	MODULATION TECHNIQUES AND CODING	Stage-2 (Specification)		
45		Stages-3, -4		
15.		Stage-2 (Specification)		
		Stage-3 (Calculated)		
16		Stages 1 - 2 (If Known)		
10.				
17		Stage 2 (Specification)		
17.	DEVIATION RATIO	Stage-2 (Specification)		
		Stage-4		
18	PULSE CHARACTERISTICS	Stage-2 (Specification)		
10.		Stage-3 (Calculated)		
		Stage-4 (Measured)		
19.	POWER	Stage-1, -2 (Specification)		
		Stages-3, -4		
20.	OUTPUT DEVICE	Stage-2 (Specification)		
_		Stages-3, -4		
21.	HARMONIC LEVEL	Stages-2, -3 (Estimated)		
		Stage-4 (Measured)		
22.	SPURIOUS LEVEL	Stages-2, -3 (Estimated)		
		Stage-4 (Measured)		
23.	FCC TYPE ACCEPTANCE NO.	Stage-4		
24.	REMARKS	All Stages (As Required)		
	CLASSIFICATION	All Stages		

Item No.	Item Name	Requirement		
	CLASSIFICATION	All Stages		
	PAGE	All Stages		
1.	NOMENCLATURE, MANUFACTURER'S MODEL	Stage-2 (Interim Design)		
	NO.	Stages-3, -4		
2.	MANUFACTURER'S NAME	Stages-2, -3, -4		
3.	RECEIVER INSTALLATION	All Stages		
4.	RECEIVER TYPE	Stages-1, -2 (Specification) Stages-3, -4		
5.	TUNING RANGE	Stage-1 (Specification) Stages-2, -3, -4		
6.	METHOD OF TUNING	Stage-2 (Specification) Stages-3, -4		
7.	RF CHANNELING CAPABILITY	Stage-2 (Specification) Stages-3, -4		
8.	EMISSION DESIGNATOR	Stage-2 Estimated Stage-3 Calculated Stage-4 Measured		
9.	FREQUENCY TOLERANCE	Stage-2 (Specification) Stages-3, -4		
10.	IF SELECTIVITY	Stage-2 Estimated Stage-3 Calculated Stage-4		
12.	IF FREQUENCY	Stage-2 Estimated Stage-3 Calculated Stage-4 Measured		
13.	MAXIMUM POST DETECTION FREQUENCY	Stage-2 (Specification) Stages-3, -4		
14.	MINIMUM POST DETECTION FREQUENCY	Stages-2, -3, -4		
15.	OSCILLATOR TUNED	Stages-2, -3, -4		
16.	MAXIMUM BIT RATE	Stages-1, -2 (Specification) Stages-3, -4		
17.	SENSITIVITY	Stage-2 (Specification) Stages-3, -4		
18.	DE-EMPHASIS	Stage-2 (Specification) Stages-3, -4		
19.	IMAGE REJECTION	Stages-2, -3 (Estimated) Stage-4 (Measured)		
20.	SPURIOUS REJECTION	Stage-2, -3 (Estimated) Stage-4 (Measured)		
21.	REMARKS	All Stages (As Required)		
	CLASSIFICATION	All Stages		

ANTENNA EQUIPMENT CHARACTERISTICS				
Item No.	Item Name	Requirement		
	CLASSIFICATION	All Stages		
	PAGE	All Stages		
1.	TRANSMITTING, RECEIVING, TRANSMITTING AND RECEIVING	All Stages		
2.	NOMENCLATURE, MANUFACTURER'S MODEL NO.	Stage-2 (Interim Designation)		
3.	MANUFACTURER'S NAME	Stages-2, -3, -4		
4.	FREQUENCY RANGE	Stages-1, -2 (Specification) Stages-3, -4		
5.	TYPE	Stage-2 (Generic Specification) Stage-3 (More Details) Stage-4 (Complete)		
6.	POLARIZATION	Stages-1, -2 (Specification) Stages-3, -4		
7.	SCAN CHARACTERISTICS	Stages-3, -4		
8.	GAIN	Stages-2, -3 (Estimated or Calculated) Stage-4 (Measured)		
9.	BEAMWIDTH	Stages-2, -3 (Estimated or Calculated) Stage-4 (Measured)		
10.	REMARKS	All Stages (As Required)		
	CLASSIFICATION	All Stages		

NTIA GENERAL INFORMATION				
Item No.	Item Name	Requirement		
	CLASSIFICATION	All Stages		
	PAGE	All Stages		
1.	APPLICATION TITLE	All Stages		
2.	SYSTEM NOMENCLATURE	All Stages		
3.	STAGE OF ALLOCATION	All Stages		
4.a	Frequency(ies)	All Stages		
4.b	Emission Designator(s)	Stage-2 (Estimated)		
		Stage-3 (Calculated)		
		Stage-4 (Measured)		
5.	PURPOSE OF SYSTEM, OPERATIONAL AND	All Stages		
	SYSTEM CONCEPTS	-		
6.	INFORMATION TRANSFER REQUIREMENTS	Stages-2, -3, -4		
7.	ESTIMATED INITIAL COST OF THE SYSTEM	All Stages		
8.	TARGET DATE FOR	All Stages		
9.	SYSTEM RELATIONSHIP AND ESSENTIALITY	All Stages		
10.	REPLACEMENT INFORMATION	All Stages		
11.	RELATED ANALYSIS AND/OR TEST DATA	All Stages		

#### NTIA GENERAL INFORMATION

Item No.	Item Name	Requirement
12.	NUMBER OF MOBILE UNITS	All Stages
13.	GEOGRAPHICAL AREA FOR	All Stages
14.	LINE DIAGRAM	All Stages
15.	SPACE SYSTEMS	All Stages
16.	TYPE OF SERVICE FOR STAGE-4	All Stages
17.	STATION CLASS FOR STAGE-4	All Stages
18.	REMARKS	All Stages (As Required)
	DOWNGRADING INSTUCTIONS	All Stages
	CLASSIFICATION	All Stages

FOREIGN COORDINATION GENERAL INFORMATION				
Item No.	Item Name	Requirement		
	CLASSIFICATION	All Stages		
	PAGE	All Stages		
1.	APPLICATION TITLE	All Stages		
2.	SYSTEM NOMENCLATURE	All Stages		
3.	STAGE OF ALLOCATION	All Stages		
4.a	Frequency(ies)	All Stages		
4.b	Emission Designator(s)	Stage-2 (Estimated) Stage-3 (Calculated) Stage-4 (Measured)		
5.	PROPOSED OPERATING LOCATIONS OUTSIDE OF US&P	All Stages		
6.	PURPOSE OF SYSTEM, OPERATIONAL AND SYSTEM CONCEPTS	All Stages		
7.	INFORMATION TRANSFER REQUIREMENTS	Stage-2 (Specification) Stages-3, -4		
8.	NUMBER OF EQUIPMENTS OPERATING SIMULTANEOUSLY IN THE SAME ENVIRONMENT	All Stages		
9.	REPLACEMENT INFORMATION	All Stages		
10.	LINE DIAGRAM	All Stages		
11.	SPACE SYSTEM	All Stages		
12.	PROJECTED OPERATIONAL DEPLOYMENT DATE	All Stages		
13.	REMARKS	All Stages, as required.		
	DOWNGRADING INSTUCTIONS	All Stages		
	CLASSIFICATION	All Stages		

## **More About**

Detailed information about specific DD Form 1494 topics are listed alphabetically in this section of the guidelines.

**Amplitude Modulation (AM):** A method of impressing data onto a carrier wave which is varied in accordance with the modulating signal. The instantaneous amplitude (overall signal power) varies depending on the instantaneous amplitude of the modulating data.

In AM, the carrier wave does not fluctuate in amplitude. Instead, the modulating data appears in the form of signal energy at frequencies slightly higher and lower than that of the carrier. These components are called sidebands. The lower sideband (LSB) appears at frequencies below the carrier frequency; the upper sideband (USB) appears at frequencies above the carrier frequency. The LSB and USB are essentially "mirror images" of each other. The sideband power accounts for the variations in the overall amplitude of the signal. In Single Side Band (SSB) AM one sideband is removed to reduce bandwidth requirements.

AM: See Amplitude Modulation.

**Antennas:** Any structure or device used to collect or radiate electromagnetic waves, often referred to as "radio waves." Most antennas are resonant devices operating over a relatively narrow frequency band. To avoid impairing reception or transmission, an antenna must be tuned to the frequency band of the radio system to which it is connected.

Common antenna types:

Corner Reflector Dipole Dipole Array Discone Log Periodic Ground Plane Horn Helical Lens Long Wire Loop Magnetic loop Metal-plate lens Monopole Parabolic Reflector Phased Array Rhombic Rod Slotted Waveguide Sloper Spiral Stub Whip Vee Yagi-Uda

**Antenna polarization:** The orientation of the electric field of an electromagnetic wave that an antenna is designed to radiate or receive. The initial polarization of a radio wave is determined by the type of antenna. Linear polarization and circular polarization are two often-used cases of elliptical polarization.

In linear polarization, the electric field vector stays in the same plane. In circular polarization, the electric field vector appears to be rotating, with a circular motion, about the direction of propagation, making one full rotation for each RF cycle. The rotation may be oriented right-hand or left-hand.

Common polarization types:

Linear – Vertical Linear – Horizontal Circular – Right Hand Circular (RHC) Circular – Left Hand Circular (LHC)

**Antenna patterns:** The relative strength of the radiated field at a fixed distance from an antenna as the direction from the antenna is varied. The radiation (antenna) pattern is also a "reception pattern," as it describes the antenna's receiving properties. The radiation pattern is three-dimensional, but it is difficult to display the three-dimensional radiation pattern in a meaningful manner, it is also time consuming to measure a three-dimensional radiation pattern. Often radiation patterns are measured as slices of the three-dimensional pattern.

These two-dimensional radiation patterns can be displayed easily on a screen or piece of paper. These pattern measurements are presented in either a rectangular or a polar format.

Antenna types: See Antennas.

AR 380-5: To view this document in its entirety, click here.

**Chip rate:** The rate at which the information signal bits are transmitted as a pseudorandom sequence of chips in direct-sequence-modulation spread spectrum systems. A chip is the most elemental component of a spread spectrum signal when it is decompressed in time

**Coding:** The class of signal transformations designed to improve communications performance by enabling a receiver to recover the transmitted signal in the presence of noise, fading, and jamming. Coding reduces the probability of error or the required  $E_b/N_o$ , at the cost of bandwidth.

**Commercial off-the-shelf equipment (COTS):** Items customarily used for Non-Government purposes (DoD 5000.2R and FAR 2.101 contain detailed definitions of "commercial" and Non-Developmental items). A commercial off-the-shelf product is used "as-is." COTS equipment is mass-produced, designed to install easily, is relatively inexpensive, and interoperates with existing system components. Examples include: wireless local area network cards, Part 15 certified radio devices, etc.

**Continuous Wave (CW):** A radio transmission of constant amplitude and constant frequency (containing no modulation). Also used to describe transmission on a single frequency using on/off keying (e.g., Morse Code).

# Country List: Countries recognized by the United Nations

Af – Dj	Do – Lu	Ma - Sen	Ser - Zim
Afghanistan	Dominica	Macedonia (The Former Yugoslav	Serbia and Montenegro
Albania	Dominican Republic	Republic of)	Sevchelles
Algeria	Ecuador	Madagascar	Sierra Leone
Andorra	Egypt	Malawi	Singapore
Angola	El Salvador	Malaysia	Slovakia
Antigua and Barbuda	Equatorial Guinea	Maldives	Slovenia
Argentina	Eritrea	Mali	Solomon Islands
Armenia	Estonia	Malta	Somalia
Australia	Ethiopia	Marshall Islands	South Africa
Austria	Fiii	Mauritania	Spain
Azerbaijan	Finland	Mauritius	Sri Lanka
Bahamas	France	Mexico	Sudan
Bahrain	Gabon	Micronesia (Federated States of)	Suriname
Bandladesh	Gambia	Monaco	Swaziland
Barbados	Georgia	Mongolia	Sweden
Belarus	Germany	Morocco	Switzerland
Belgium	Ghana	Mozambique	Svrian Arab Republic
Belize	Greece	Myanmar	Taiikistan
Benin	Grenada	Namihia	Tanzania (United Republic of)
Bhutan	Guatemala	Nauru	Thailand
Bolivia	Guinea	Nenal	Timor-Leste
Bosnia and Herzegovina	Guinea-Bissau	Netherlands	
Botswana	Guinea-Dissau	New Zealand	Topgo
Brozil		Nicoragua	Tripidad and Tobago
Brupai Darugaalam	Handuraa	Nicar	Tunicia
Biuliei Dalussalam Bulgaria		Niger	Turkov
Bulyana Bulyana Easa	loolood	Nervov	
Burundi	India	Oman	
Combodio	Indonesia	Dilidi	Llagado
Camproon	Inconesia Iran (lalamia Banublia of)	Pakisian	Uganua
Canada		Palau	United Areb Emirotea
Canada Cana Varda	IIaq	Panama Depus New Cuines	United Kingdom of Croot
Cape Velue	Ireialia	Papua New Guinea	Dritein and Northern Iroland
Central Amcan Republic	Isidei	Paraguay	
Chilo	lanon	Pelu Dhilippingg	United States of America
China	Japan	Pland	Ulthekisten
Colombia	Joiddii	Polatu	Vanuatu
Colombia	Kazakristan	Pollugai	Vanualu
Control Componentia	Kellya	Qala Depublic of Koroo	
Congo (Democratic	Kindali Karaa (Democratic Decele's	Republic of Noldayo	Viet Nam
Republic of the)	Rorea (Democratic People's	Republic of Moldova	Zambia
	Republic or)	Romania Duccion Fodoration	Zampia
Cote divoire	Kuwait	Russian Federation	Zimbabwe
Citalia	Kyrgyzstan		
	Lao (People's Democratic	Saint Kitts and Nevis	
Cyprus One als Danuts lie	Republic)	Saint Lucia	
	Latvia	Saint Vincent and the Grenadines	
Denmark	Lebanon	Samoa San Marina	
Djibouti	Lesotho		
		Sao Tome and Principe	
	Libyan Arab Jamaniriya	Saudi Arabia	
		Senegal	
	Litnuania		
	Luxembourg		
**CW:** See Continuous Wave.

**Data requirements:** To view data requirements for each stage of allocation, click <u>here</u>.

**Decibel (dB):** Represents the relationship between two values of power. Decibels describe numbers of significantly different magnitude, such as 11.23 and 56,000,000. For convenience, we find the ratio between the two numbers and convert that into a common logarithm and multiply it by 10.

The formula for dB = 10Log (Power X/Power Y)

Example: If Power X = 11.23 Watts and Power Y = 56,000,000 Watts, the Power X is -66.97 dB less than Power Y.

For Voltage (e.g., Electric field strength), the formula for dB = 20Log (Voltage X/Voltage Y).

Other dB units represent the power levels in decibels with reference to one milliwatt (dBm), and one watt (dBW) and represent antenna gain in decibels with reference to an isotropic radiator (dBi).

DoD Directive 5200.1-R: To view this directive, click here.

**Emission Designators:** To view information about the construction of emission designators, click <u>here</u>.

**FCC Type Acceptance #:** The FCC ID is assigned for equipment authorization under various rules including Parts 15, 90, 101. An FCC ID consists of two elements, the grantee code and an equipment product code.

The Grantee Code is the first three characters of the FCC ID, consisting of Arabic numerals, capital letters, or a combination of the two. The Grantee Code is permanently assigned by the Commission to a company for authorization of all radio frequency equipment subject to certification, type acceptance and notification

The Equipment Product Code (EPC) is assigned by the applicant (minimum of 1 character and maximum of 14 characters) and may consist only of capital letters, Arabic numerals, and the hyphen or dash (-). Applicants often assign a hyphen (-) as the first digit of the EPC.

**Filter types:** An RF filter is an electrical circuit specifically designed with respect to the transmission or attenuation of various frequencies that may be applied to it. Four general types of RF filters exist: high-pass, low-pass, band-pass, and band-stop.

High-pass filter - permits all frequencies above the specified cut-off frequency to be transmitted with little or no loss, and will attenuate all frequencies below the cut-off.

Low-pass filter - permits all frequencies below the specified cut-off frequency to be transmitted with little or no loss, and will attenuate all frequencies above the cut-off frequency.

Band-pass filter - transmits a selected band of frequencies with substantially no loss, and attenuates all frequencies higher or lower than the desired band.

Band-stop filter - attenuates a selected band of frequencies and passes all frequencies higher or lower than the stop band.

**Frequency/Frequencies:** Frequency refers to the number of cycles per unit time of an electromagnetic wave (e.g., an RFID tag operating on the frequency 433.92 MHz is using electromagnetic waves that cycle at 433.92 million times per second).

**Frequency Modulation (FM):** A method of impressing data onto a carrier wave by varying the wave's instantaneous frequency. This scheme can be used with analog or digital data.

**FM Pre-emphasis:** In FM transmission systems pre-emphasis of the higher modulating frequencies results in an improved signal-to-noise power ratio when

the receiver recovers the modulation. At the output of the FM receiver, a deemphasis network restores the original signal power distribution.

**JETDS:** To view the JETDS, click <u>here</u>.

**Lockout:** A frequency channel deliberately disabled from use (locked out) to avoid interference with another service.

**Life cycle management:** The cradle-to-grave management of products, including such common phases as: mission need justification, concept studies, concept exploration and definition, demonstration and validation, development decision, development phase, production decision, production and deployment, and operations and support. Frequency supportability must be obtained at each phase of the product life cycle. For products that are commercial off the shelf (COTS), non-developmental items, only Stage-4 – Operational approval needs to be obtained.

**Manufacturers:** If the system is built or designed by a contractor, the name of the prime contractor (not a subcontractor or subsidiary) should be provided.

**Modulation Techniques:** Used to transmit information (digital or analog) by encoding the information onto a radio frequency signal. The information signal is modulated onto an analog signal at a specific frequency called the carrier. The three basic modulation techniques are:

AM (amplitude modulation) FM (frequency modulation) PM (phase modulation)

All three modulation techniques utilize a radio frequency carrier signal. A carrier signal is a single frequency that is used to carry the intelligence (data). For digital, the intelligence is either a 1 bit or 0 bit. Modulating the carrier changes its characteristics to represent either a 1 or 0.

Amplitude Modulation modifies the amplitude of the carrier to represent 1s or 0s. For example a 1 is represented by the presence of amplitude modulation on he carrier for a predefined period of 4 cycles of carrier. A lack of modulation on the carrier for 4 cycles indicates a 0.

Frequency Modulation changes the frequency of the carrier to represent the 1s or 0s. For example a 0 is represented by the initial carrier frequency and a 1 by a second frequency.

Phase Modulation modifies the phase of the carrier to represent a 1 or 0. The carrier phase is switched at every occurrence of a 1 bit, but remains unaffected for a 0 bit. The phase of the signal is measured relative to the phase of the preceding bit. The bits are timed to coincide with a specific number of carrier cycles.

**Necessary Bandwidth:** The width of the frequency band which is just sufficient to ensure the transmission of information, for a given class of emission, at the rate and with the quality required under specified conditions.

**Noise Figure:** The contribution by the receiver itself to thermal noise at its output. The noise figure is usually expressed in decibels (dB), with respect to thermal noise power at the system impedance, at a standard noise temperature (usually 20 degrees C, 290 K) over the bandwidth of interest. It is determined by (a) measuring (determining) the ratio, in dB, of the thermal noise power at the output, to that at the input, and (b) subtracting from that result, the gain, in dB, of the system. Typical noise figures range from 0.5 dB for very low noise devices, to 4 or 8 dB. In some systems (e.g., superheterodyne systems) total output noise power includes noise from sources other than thermal, such as spurious contributions from image-frequency transformation. However, noise from these sources is not considered in determining the noise figure.

**Noise temperature:** Method used to specify the contribution of a receiver to the thermal noise at its output. It is used often used with space systems. The formula for noise temperature is:

$$(T) = 290X (10^{(Noise Figure/10)-1}) K$$

**NTIA Frequency Tolerance Standards:** To view the NTIA Frequency Telerance Standards, click <u>here</u>.

**NTIA station classes:** To view the NTIA Table of Services, Station Classes, and Stations, click <u>here</u>.

**Occupied bandwidth:** The width of a frequency band such that below the lower and above the upper frequency limits the mean powers radiated are each equal to B/2 percent of the total mean power of the emission. Unless otherwise specified by the CCIR for the appropriate class of emission, the value of B/2 should be taken as 0.5%. In some cases (e.g., multichannel frequency-division multiplexing systems), use of the 0.5% limits may lead to certain difficulties in the practical application of occupied and necessary bandwidth; in such cases, a different percentage may prove useful.

**Output devices:** A description of the device used to generate the RF energy delivered to the antenna.

RF Output Devices:

Transistor Gunn Diode Field Effect Transistor Reflex Klystron Magnetron Traveling Wave Tube (TWT) Transistor pair in Class D operation Transistor pair in Class B operation Impatt Diode RF Integrated Circuit "Manufacturer Name" and Model "#"

If available, the specific device or RF module designation should be supplied, for example:

Varian VTS5751A1 TWT or Texas Instruments TRF 6900A Single Chip RF Transceiver

**Phase Modulation:** A form of modulation in which the carrier waves phase is varied in accordance with the modulating signal. Phase modulation is a form of angle modulation.

Pre-emphasis: See FM Pre-emphasis.

**Preselection types:** Preselection filtering reduces the potential for out of band interference due to front end overload and intermodulation effects. Some receivers use no preselection, others that must operate in more adverse RF environments require complex filtering circuitry.

Preselection filter types:

Tunable cavity filter Microstrip filter Band-pass filter Band-stop filter Low-pass filter High-pass filter SAW filter Two-pole electronically tuned filter

**Pulsed Radar Systems:** There are two basic types of radar systems: pulsed radars, used to search, detect and measure the range of targets, and continuous wave radars used to perform target velocity measurements. In a pulsed radar system, the frequency generation and timing system periodically cause the transmitter to generate a pulse, or burst, of electromagnetic energy. The width of the pulse can vary between nanoseconds and milliseconds. The radar system which transmitted the RF pulse waits for the echo to return from the target, where it is detected and processed.

**Radio Frequency Channel:** Radio frequency assignment is a synonym for frequency assignment, which is an authorization, given by an administration, for a radio station to use a radio frequency, or band of frequencies, under specified conditions.

**Receiver types:** In the context of a DD Form 1494 application, receiver types refers to basic receiver architecture. Most high quality receivers are superheterodyne, meaning that the incoming radio frequency signal is converted to an intermediate frequency for additional processing (e.g., IF filtering and

amplification) prior to demodulation. Other receiver types commonly encountered are the homodyne, tuned radio frequency (TRF), and regenerative.

**Receivers:** Any device that detects and demodulates transmitted electromagnetic energy radiated from either a man-made or natural source.

**Remarks:** May be used to continue the description in preceding blocks. Each must be identified by block number and if the DD Form 1494 is classified each must contain a security classification

**Sensitivity:** The minimum input signal level required at the antenna terminals for the receiver to produce a specified level of performance after demodulation and processing.

Sensitivity formula:

Sensitivity 
$$dBm = (S/N)_{min} + 10 \log (kT_oB) + (NF) + 30$$

where

(S/N) <sub>min</sub>	=	Minimum input signal-to-noise power ratio needed to process a
		signal at the required level of performance, in dB
k	=	Boltzmann's Constant = 1.38 x 10 <sup>-23</sup> Joule/degrees K
To	=	Absolute temperature of the receiver input (Kelvin) = 290
В	=	Receiver 3-dB IF bandwidth, Hz
NF	=	Receiver Noise figure, in dB

**Service types:** The following are the ITU-recognized services that appear in national and international allocation tables. Service types should not be confused with station class.

Valid service types:

Amateur Broadcasting Broadcasting-Satellite Earth Exploration Satellite *Meteorological Satellite* 

Fixed Aeronautical Fixed Fixed-Satellite Inter-Satellite Meteorological Aids Mobile Aeronautical Mobile Land-Mobile Maritime Mobile Mobile-Satellite Aeronautical Mobile-Satellite Land Mobile Satellite Maritime Mobile-Satellite Radio Astronomy Radiodetermination Radiolocation Radionavigation Aeronautical-Radionavigation Maritime Radionavigation Radiodetermination-Satellite Radionavigation-Satellite Aeronautical Radionavigation-Satellite Maritime Radionavigation-Satellite Space Operation Space Research Standard Frequency and Time Signal Standard Frequency and Time Signal-Satellite No Specific Service

**Single Side Band (SSB):** An amplitude modulated waveform that has one sideband suppressed to reduce the necessary bandwidth.

**Software Defined Radio (SDR):** A rapidly evolving software technology that implements functional modules of a radio system, such as modulation/demodulation, signal generation, coding and link-layer protocols. A complete hardware based radio system has limited utility since parameters for each of the functional modules are fixed. An SDR system extends the utility of the system for a wide range of applications that use different link-layer protocols and modulation/demodulation techniques.

**Spread Spectrum:** A group of modulation techniques in which a signal is transmitted in a bandwidth considerably greater than the frequency content of the original signal. These techniques decrease the potential interference to other receivers while achieving privacy and increasing the immunity of spread spectrum receivers to noise and interference. Spread spectrum generally makes use of a sequential noise-like signal structure to spread the normally narrowband information signal over a relatively wide band of frequencies. The receiver correlates the signals to retrieve the original information signal. Frequency hopping, direct sequence spreading, time scrambling, and combinations of these techniques are forms of spread spectrum.

### POINTS OF CONTACT

### Product Manager AIT

Attn: SFAE-PS-EI-AIT 8580 Cinder Bed Road, Suite 1400 Lorton, VA 22079-1442 Telephone: 703-339-4400 x101 Facsimile: 703-339-4401 Frequency Management: 703-339-4400 x121 ann.scotti@us.army.mil or mike.morrissey@us.army.mil

### Joint Spectrum Center

Acquisition Support (JSC/J8) 2004 Turbot Landing Annapolis, MD 21402-5064 DSN-281-2681 410-293-2681 j8divisionchief@jsc.mil

### US ARMY NETWORK ENTERPRISE TECHNOLOGY COMMAND

Army Spectrum Management Office (ASMO) Attn: James Thomas 2461 Eisenhower Ave., Suite 1204 Alexandria, VA 22331-2200 703-325-8227 james.e.thomas@us.army.mil

### REFERENCES

Manual of Regulations and Procedures for Federal Radio Frequency Management, Washington, DC: US Department of Commerce, National Telecommunications and Information Administration, May 2003 (revision September 2003).

M. Alexander, D. Kenney, B. Pavon, R. Orsulak, and T. Maguire, *DD Form 1494 Preparation Guide for Army Frequency Allocations*, ECAC-CR-90-079, DoD Electromagnetic Compatibility Analysis Center, January 1991.

*Army Management of the Electromagnetic Spectrum*, Army Regulation 5-12, Washington DC: Department of the Army, 1 October 1997.

*Department of the Army Information Security Program*, Army Regulation 380-5, Washington DC: Headquarters Department of the Army, 29 September 2000.

*Joint Electronics Type Designation System (JTEDS)*, Army Regulation 105-19, Washington DC: Headquarters Department of the Army, the Air Force, the Navy, and the US Marine Corps, 11 February 1974

Department of Defense Electromagnetic Compatibility Program, DoD Directive 3222.3, Washington DC: ASD C3I, 20 August 1990.

*DoD Information Security Program*, DoD Directive 5200.1-R, Washington DC: ASD C3I, 13 December, 1996.

### ATTACHMENTS

Management

# Army Management of the Electromagnetic Spectrum

Headquarters Department of the Army Washington, DC 1 October 1997



# SUMMARY of CHANGE

AR 5-12 Army Management of the Electromagnetic Spectrum

This regulation --

- o Revises AR 5-12 and consolidates AR 105-3, AR 105-4, AR 105-24, and AR 105-28.
- o Revises Army Spectrum Management Policy (Chapter 1).
- o Revises Major Army Commands Spectrum Management Responsibilities (Chapter 2).
- o Outlines the Army's Electromagnetic Compatibility, Deconfliction, and Interference Programs (Chapter 3).
- o Outlines the Frequency Allocation-to-Equipment Process (Chapter 4).
- o Defines the spectrum coordination channels, spectrum assignment policy, and types of spectrum actions (Chapter 5).
- o Outlines the Radio Stations Identification procedures for the allocation of international call signs and call sign assignment authority (Chapter 6).
- o Outlines the Army's Interference Resolution Program (Appendix C).

Effective 1 November 1997

### Management

### Army Management of the Electromagnetic Spectrum

materiel requiring frequency spectrum support. It also describes the Army spectrum management functional processes necessary to implement the National Telecommunications and Information Administration (NTIA) Manual of Regulations and Procedures for Federal Radio Frequency Management and the provisions of Department of Defense Directive (DoDD) 4650.1.

**Applicability.** This regulation applies to all active Army, the Army National Guard of the United States, and the US Army Reserves. It also applies to Army components of Unified Commands if it does not contradict guidance issued by the Unified Commands.

**Proponent and exception authority.** The proponent of this regulation is the Director, Information Systems for Command, Control, Communications, and Computers. The proponent has the authority to approve exceptions to this regulation that are consistent with law or regulation. Proponents may delegate the approval authority, in writing, to a division chief within the proponent agency in the grade of colonel or the civilian equivalent.

Army management control process.

This regulation contains management control provisions in accordance with AR 11-2 and contains checklists for conducting management control reviews.

**Supplementation.** US Army Major Command Supplements to this regulation that implement and/or delegate responsibilities to subordinate commands will be submitted for approval to HQDA (SAIS-PAS-M), Washington, DC 20310-0107.

**Suggested Improvements.** Users are invited to send comments and suggested improvements on DA Form 2028 (Recommended Changes to Publications and Blank Forms) directly to HQDA (SAIS-PAS-M), Washington, DC 20310-0107.

**Distribution.** Distribution of this publication is made in accordance with initial distribution number (IDN) 093495, intended for command levels D and E for Active Army, Army National Guard, and U.S. Army Reserve.

Contents (Listed by paragraph and page number) Chapter 1 Introduction, page 1 Purpose • 1–1, page 1 References • 1–2, page 1

61 1.6

History. This printing publishes a revision of

this publication. Because the publication has

been extensively revised, the changed portions

Summary. This revision updates and con-

solidates several regulations regarding policies and responsibilities for Army

management of the electromagnetic spec-

trum. It covers the coordination and integra-

tion of the research, development, test,

acquisition, fielding and operation of Army

have not been highlighted.

Togo D. West, Jr

Secretary of the Army

Abbreviations and terms • 1–3, page 1 Responsibilities • 1–4, page 1 Goals • 1–5, page 1 Management of the electromagnetic spectrum • 1–6, page 1 Management of the electromagnetic spectrum in the IM environment • 1–7, page 1 Policies • 1–8, page 1

### Chapter 2

Responsibilities, page 3 Introduction • 2–1, page 3 Command and staff channels • 2–2, page 3 Technical channels • 2–3, page 4 Applicability • 2–4, page 4 Responsibilities • 2–5, page 4 Director of Information Systems for Command, Control, Communications, and Computers (DISC4) • 2–6, page 4

The Army Spectrum Manager • 2-7, page 4 US Army Communications-Electronics Services Office (USACESO) • 2-8, page 5 MACOM commanders • 2–9, page 5 Major subordinate command (MSC) commanders • 2-10, page 5 US Army installations • 2-11, page 6 Major command DCSIM • 2-12, page 6 MACOM Spectrum Manager • 2-13, page 6 Major Subordinate Component Command (MSCC) (Corps and CONUSAs) ACSIM/SO • 2-14, page 6 MSCC Spectrum Manager • 2-15, page 6 Army Garrison Commander • 2-16, page 6 Director of Information Management (DOIM) • 2-17, page 6 Commanding General, US Army TRADOC (CG, TRADOC) 2–18, page 7 Commanding General, US AMC (CG, AMC). • 2-19, page 7 Commanding General, US Army Forces Command (CG, FORSCOM) • 2–20, page 7 Commanding General, US Army Reserve Command (CG, USARC) • 2-21, page 8 Commanding General, US Army Intelligence and Security Command (CG, INSCOM) • 2-22, page 8 Commanding General, US Army Signal Command (CG, ASC) • 2-23, page 8

\*This regulations supersedes AR 5-12, 15 May 1983, and rescinds AR 105-3, 31 July 1986; AR 105-4, 22 September 1977; AR 105-24, 28 March 1977; and AR 105-28, 25 July 1973.



### Contents—Continued

- Commanding General, US Army Special Operations Command (SOC) (CG, USASOC) 2-24, page 8
- Commanding General, US Army Medical Command (CG, USAMEDCOM) 2–25, page 9
- Commanding General, US Army Operational Test and Evaluation Command (CG, OPTEC) • 2–26, page 9
- Commanding General, US Army Space and Strategic Defense Command (CG, SSDC) • 2–27, page 9
- Commanders of Reserve Component (RC) Units 2-28, page 9
- PEO/PM and Separate PM 2-29, page 9
- Army Director, JSC 2-30, page 9
- Special Frequency Responsibilities for Units, Organizations, and Activities 2-31, page 10

### Chapter 3

### Army Electromagnetic Compatibility, Deconfliction, and Interference Control Programs, *page 10*

Introduction • 3-1, page 10

- Scope 3-2, page 10
- Management Requirements for Achieving Electromagnetic Compatibility (EMC) 3-3, page 10
- USASC Services for the Electromagnetic Compatibility Program (EMCP) 3-4, page 11
- US Army Test and Evaluation Command (USATECOM) Services for the EMCP 3-5, page 11
- JSC Services for the EMCP 3-6, page 11
- Deconfliction 3–7, page 12
- Army Interference Resolution Program (AIRP) 3-8, page 12
- Army E3 Objectives Relating to Spectrum-Dependent Devices • 3–9, page 12

### Chapter 4

### Frequency Allocation-to-Equipment Process, page 12

- Introduction 4-1, page 12
- Policies Concerning DD Form 1494 4-2, page 13
- Application for Spectrum Certification 4–3, page 13
- Processing DD Form 1494 4-4, page 14
- EMC and DD Form 1494 Reviews Required by the NTIA 4–5, page 14
- Waivers and Exceptions 4-6, page 14
- EMC Standards for Telecommunications Equipment 4-7,
- *page 15* Duties • 4–8, *page 16*

#### Chapter 5

#### Spectrum Requests and Assignments, page 16

Introduction • 5–1, page 16
Spectrum Coordination Channels • 5–2, page 17
Types of Spectrum Actions • 5–3, page 17
Frequencies that do not require specific authorization • 5–4, page 17
Army Policy for Spectrum Assignments • 5–5, page 17
Spectrum Coordination Procedures • 5–6, page 18
Spectrum Requests in Non-Government Bands • 5–7, page 18
International Registration • 5–8, page 19
Electronic Attack (EA) Training • 5–9, page 19

### Chapter 6

Radio Station Identification, page 19
General • 6–1, page 19
Responsibility for Radio Station Identification • 6–2, page 19
International Call Signs • 6–3, page 19
International Call Sign Assignment Authority • 6–4, page 19
Publications • 6–5, page 21
Security • 6–6, page 21
Positive Voice Identification (PVI) • 6–7, page 21
Tactical Call Signs • 6–8, page 22
Special Call Signs/Call Words for Army Aircraft • 6–9, page 22

Amateur Radio Call Signs • 6-10, page 22

### Appendixes

- A. References, page 23
- B. Army Frequency Coordinators, page 24
- C. Program Plan for Interference Resolution, page 25
- D. Management Control Evaluation Checklist, page 34
- E. Management Control Evaluation Checklist, page 34
- F. Management Control Evaluation Checklist, page 35
- G. Management Control Evaluation Checklist, page 36
- H. Management Control Evaluation Checklist, page 37
- I. Management Control Evaluation Checklist, page 37

### Glossary

Index

### Chapter 1 Introduction

### 1–1. Purpose

This regulation:

*a.* Issues Army policy and assigns responsibilities for Army management of the electromagnetic spectrum (hereafter referred to as "spectrum management."

*b.* Issues Army policy and assigns responsibilities for Army participation in service, joint, national, and international spectrum management activities.

c. Issues policies and responsibilities for spectrum allocations, allotments, and assignments by Army commands, agencies and activities within the United States and Possessions (US&P).

*d.* Issues guidance for spectrum allocations, allotments, and assignments to Army components of Unified commands.

*e.* Issues guidance for implementing spectrum management functions contained in AR 25-1, The Army Information Resources Management Program.

*f.* Issues policies and responsibilities for submitting US Army requirements for a DD Form 1494 (Application for Equipment Frequency Allocation) as outlined in Chapter 4.

*g.* Prescribes duties and responsibilities for Army commanders at all levels and for users at installation and unit levels who perform spectrum management duties.

*h*. Issues policy and responsibilities for achieving the Army spectrum-dependent electromagnetic environmental effects  $(E^3)$  and electromagnetic compatibility (EMC) objectives during the design, development, acquisition, and use of spectrum-dependent equipment.

*i.* Issues policies, guidance, and responsibilities for the implementation and reporting of interference under the Army Interference Resolution Program (AIRP).

j. Issues policy for Army radio station identification.

### 1-2. References

Required and related publications and prescribed forms are listed in Appendix A.

### 1–3. Abbreviations and terms

Abbreviations and special terms used in this regulation are explained in the glossary.

### 1-4. Responsibilities

Responsibilities are listed in Chapter 2.

### 1-5. Goals

The goals of the Army spectrum management program for efficient use of the spectrum are:

*a.* To develop and efficiently manage Army use of the spectrum during the allocation, allotment, and assignment processes and thereby minimize the potential for interference during the fielding and employment of spectrum dependent equipment.

b. To obtain and manage the frequency resources to support Information Mission (IM) responsibilities, activities, and programs relating to the disciplines of telecommunications, automation, visual information, records management, and publications and printing.

*c.* To provide spectrum signal characterization, measurement, and enforcement procedures for continued operation in the Army, joint and combined environment.

### 1-6. Management of the electromagnetic spectrum

The Army's management of the spectrum is accomplished in the following terms:

*a. Allocation.* An allocation is the designation of frequency bands for use in performing specific functions or services. Allocations are made to communications services such as fixed, mobile, broadcast, and amateur. This process is not to be confused with the Army J/F-12 (Frequency Allocation-to-Equipment) process (permission to build or buy equipment for use in a particular frequency band) outlined in Chapter 4.

b. Allotment. An allotment is the designation of specific frequency bands or groups of frequencies within a prescribed allocation.

*c. Assignment.* An assignment is the designation of a specific frequency or frequencies for use by a radio station under specified conditions. An assignment grants permission to operate or turn on authorized equipment.

*d. Enforcement.* Enforcement is the process of identifying and eliminating unauthorized use of the frequency spectrum with potentially punitive measures.

# 1–7. Management of the electromagnetic spectrum in the IM environment

AR 25-1, The Army Information Resources Management Program, identified three IM environments in which the Army spectrum management processes will be performed. The three environments are:

a. Theater/Tactical. This environment is defined as the operational Army Theater "area of operations." Types of information resources managed in this environment are those needed to direct, coordinate, and support deployable combat, combat support, and combat service support forces. The boundary of this environment extends from the tactical area of interest of the forwardmost deployed forces back to the Theater Army rear boundary, and includes the headquarters of joint, unified, or combined commands.

b. Sustaining Base. This environment encompasses area and information resources usually located outside the "area of operations." The environment encompasses the information resources and activities which have the responsibility to raise, organize, train, equip, and eventually, deploy and sustain Army and other assigned forces into the operational theaters - the Theater/Tactical environment. The types of information resources managed in this environment cover all functional areas other than those needed to actually direct tactical forces in the execution of their operational missions in the Theater/ Tactical environment. The boundary of this environment is usually geographically located in the continental United States (CONUS) but, during peace and transition to conflict, extensions of the Sustaining Base may be found in the geographical operational area of the Theater/Tactical environment.

*c. Strategic Environment.* This environment pertains specifically to the type of information resources that support decision making during a national crisis. The type of information used is usually that which is concerned with large unit readiness and deployability status (time phased force deployment data (TPFDD), ports and strategic movement capabilities, strategic intelligence, and strategic contingency plans and deployment schedules. Headquarters Department of the Army (HQDA) agencies, major Army commands (MACOMs), and the Army components of deployed Army forces are major users and providers of this type of information.

### 1-8. Policies

The following policies address some of the major technological and operational requirements that must be met to achieve the Army's spectrum management goals:

*a.* The Army satisfies its spectrum resource requirements based on priority of operational needs. The Army satisfies these requirements by following the Army's Spectrum Management Master Plan (ASM2P), using operations plans contingency plans, TPFDD, and using computerized techniques, databases, and assignment systems.

*b.* Policies and guidelines for use of the spectrum by all Federal Government agencies have been established in the National Telecommunications and Information Administration (NTIA) Manual and provisions of DoDD 4650.1. The Army is obligated to comply with these policies unless waived by the Army Spectrum Manager.

c. Funds for the acquisition, research, development, production, purchase, lease, or use of weapons systems, information management systems or electronic warfare (EW) systems or other systems that require use of the electromagnetic spectrum will not be released by the obligating authority until a DD Form 1494 (Application for Frequency Allocation) has been approved. Sources of funding that are subject to the DD Form 1494 approval process are DoDR 5000.2-R, AR 25-3, and AR 70-1 acquisitions as well as Commander in Chief (CINC) initiative funds. The approval authority for DD Form 1494 within the Army is the Army Spectrum Manager (SAIS-PAS-M). Waivers may be obtained from the Army Spectrum Management Office. See Chapter 4 of this regulation.

*d.* Army materiel, which depends on or affects the use of the electromagnetic spectrum, will be introduced in the Army only after the results of EMC analyses have shown the proposed materiel is compatible with the intended electromagnetic environment and can be supported in its intended spectrum environment. Electronic Attack (EA) equipment is not usually required to be compatible with the electromagnetic environment. However, EA equipment developers must comply with the certification process (outlined in Chapter 4).

*e.* Spectrum resources allotted to Army installations or combat training center will be controlled by the garrison commander.

*f.* Requirements for the use of national and international spectrum resources will be consolidated at the US Army Communications-Electronics Services Office (USACESO).

g. For host nations, unified and joint commanders will coordinate and obtain authorizations to use spectrum resources with the host government. Diplomatic channels or established spectrum management channels will be used to obtain the authorizations required.

*h*. The Army spectrum management process must be responsive to the Army Command and Control System (ACCS). The ASM2P will be based upon the ACCS.

*i.* The Army Spectrum Manager is the principal negotiator for Army international spectrum management discussions, either through the Department of State on a Government-to-Government level; through "Status of Forces Agreements" already in force with host governments; or through the International Telecommunication Union (ITU) structure or meetings. *j.* The Army Spectrum Manager is responsible for designating individuals or agencies to represent the Army spectrum management interests at the national level. Persons or agencies outside of the HQDA presenting a unilateral Army position, plan, purpose, or objective involving the electromagnetic spectrum must ensure that the position is coordinated with the Army Spectrum Manager prior to release.

*k.* No agency of the Army is authorized participation in Federal Communication Commission (FCC) legislative matters except as authorized by the Army Spectrum Manager.

*l.* The Army Spectrum Manager has designated the USACESO to be responsible for Army representation to the Radio Technical Commission for Aeronautics; the Radio Technical Commission for Maritime Services; the US Military Communications-Electronics Board (USMCEB); the Interdepartmental Radio Advisory Committee (IRAC); and the US National Committee representative to the Radiocommunications Bureau (BR). The Army Spectrum Manager may request specific commands or activities to provide representation to Army specific working parties, ad hoc groups/committees, or panels; however, the oversight and direction are the responsibility of the Army Spectrum Manager. Army participation in national and international spectrum management organizations is shown in Table 1-1.

*m.* The Army Spectrum Manager coordinates Army positions in cases of Congressional actions that could result in the transfer, sale, auction, or removal of spectrum resources from the Government/Military to industry or the civil/private sector.

*n*. Due to the degree and complexity of coordination required for effective spectrum management, informal discussions and communications are authorized and encouraged.

#### Table 1–1 Army Participation in Spectrum Management Orga

Army Participation in Spectrum Management Organizations			
Spectrum Management Organization	USACESO	AMC	
1. FMG, RCC	М		
2. J/EP. LISMCEB	M	_	
a. Allocations Working Group (J-12)	M	_	
b. Electromagnetic Compatibility Working Group (J-208a)	M	_	
c. Frequency Resource Record System Working Group (J-208b)	M		
d. FRRS Automation Working Group (J-208f)	M		
e. Space Frequency Matters Working Group (J-208i)	М	М	
f. JTIDS Frequency Coordination Working Group (J-208j)	М	_	
g. WRC Preparation Working Group (J-208w)	Μ	М	
h. Spectrum Management Communications-Computer Systems Architecture Working Group (J-208z)	М	Μ	
3. IRAC	М		
a. FAS	M		
b. SPS	M	М	
c. TSC	М	М	
d. ING	М	М	
4. US BR National Committee	М		
5. US BR Study Groups (US BR SGs)			
a. SG-1 (Spectrum Utilization)	Μ	М	
b. SG-2 (Space Research & Radio Astronomy)	М	М	
c. SG-4 (Fixed Satellites)	Μ	М	
d. SG-5 (Propagation in Nonionized Media)	Μ	М	
e. SG-6 (Ionosphere Propagation)	М	М	
f. SG-7 (Standard Time & Frequency)	M	_	
g. SG-8 (Mobile)	M	Μ	
h. SG-9 (Fixed, LOS & Tropo)	M	—	
i. SG-10 (Broadcasting, Sound)	M	_	
j. SG-11 (Broadcasting, Television)	М	_	
k. General and Specialized WRC and ITU sponsored international meetings	М	_	

Table 1–1 Army Participation in Spectrum Management Organizations—Continued				
Spectrum Management Organization	USACESO	AMC		
6. ABCA	М	М		
M = member Terms defined in Glossary.				

### Chapter 2 Responsibilities

### 2-1. Introduction

*a.* Various echelons within the Army have command and functional responsibilities for obtaining and managing systems and spectrum resources necessary to support the Army's requirements discussed in Chapter 1. The commander or director of an organization is responsible for identifying and validating the spectrum requirements needed to accomplish the organization's assigned mission. Within a functional area, it is the functional proponent's responsibility to determine the information requirements needed to accomplish the functional mission. Figure 2-1 shows the functional Army spectrum management coordination channels along with spectrum management office designations.



*b*. Support is provided through two channels of responsibility — command and staff channels and technical channels.

### 2-2. Command and staff channels

The commander or director of an organization, agency, or activity at all levels is responsible for—

*a.* Identifying, validating, and managing their spectrum requirements and resources in meeting the organization mission in accordance with Army policy, doctrine, and assigned responsibilities per this regulation.

*b.* Assuring that USMCEB spectrum guidance (Certification of Frequency Supportability) is received prior to contractual obligations to procure or develop equipment that radiates or receives electromagnetic energy in its intended environment.

c. Assuring that equipment directly procured or leased from commercial suppliers is acceptable for licensing by the FCC ("FCC type accepted" ).

*d.* Assuring that equipment procured under the exemption granted by the Army Spectrum Manager is FCC type accepted.

e. Implementing the policies of the Internal/Review and Audit

Compliance Program outlined in AR 11-7, and conducting internal management control using the procedures and checklists modeled after those contained in AR 11-2. An example of an internal/management control checklist is provided in Appendix D.

### 2–3. Technical channels

*a.* Organizations, activities, and individuals are assigned responsibility for performing technical research, development engineering, allocation, allotment, and assignment missions that support Army spectrum management. Spectrum management conducted within these technical channels will be conducted within the limits of established Army policy.

b. There is no strict hierarchy within the technical channels. Coordination of issues with one or many offices is expected to occur. Issues are to be resolved at the lowest possible level of command. Issues that cannot be resolved within these channels are referred to command and staff channels for action.

### 2-4. Applicability

The responsibilities assigned by this regulation apply to all Active Army, Army National Guard, and US Army Reserve organizations organized under Tables of Organization and Equipment (TOE) and Tables of Distribution and Allowances (TDA).

#### 2-5. Responsibilities

Responsibilities for Army spectrum management are outlined in this chapter and depicted in Figure 2-2.



Figure 2-2. Army spectrum management organizations

#### 2–6. Director of Information Systems for Command, Control, Communications, and Computers (DISC4) The DISC4 will—

a. Direct and provide oversight to the Army spectrum management program.

*b.* Perform spectrum planning to satisfy Army warfighter requirements for spectrum resources during peacetime and wartime.

*c.* Advise the Secretary of the Army as well as the Chief of Staff of the Army on spectrum matters.

*d.* Support and defend resource requirements for spectrum management in the Army Planning, Programming, and Budgeting System (APPBS).

*e*. Direct Army representation in the national and international spectrum regulatory process.

*f.* Represent the Secretary of the Army for implementation of the Electromagnetic Compatibility Program per DoDD 3222.3.

g. Advise the Assistant Secretary of the Army for Research, Development and Acquisition (ASA (RD&A)) on spectrum management considerations in acquisition strategies for weapon, command and control communication, intelligence, and information management systems per AR 70-1.

h. Provide the Army Spectrum Manager.

### 2-7. The Army Spectrum Manager

The Army Spectrum Manager also serves as the Director of the USACESO. The Army Spectrum Manager will—

a. Develop and promulgate Army spectrum policy and planning guidance in support of all Army spectrum management activities.

*b.* Implement responsive spectrum management processes to meet Army needs and requirements at both the national and international levels. This regulation authorizes the Army Spectrum Manager to review and update the Army spectrum management structure and assign specific functions within the Army as required.

c. Implement the Army Interference Resolution Program (AIRP) in accordance with the guidance contained in Chapter 3 and Appendix C of this regulation.

d. Coordinate spectrum management matters within the office of the Secretary of the Army (OSA) and the Army Staff (ARSTAF).

*e.* Identify, budget, and provide a direct fund line in the APPBS for Army-wide spectrum management/EMC initiatives, studies, and analyses program on a recurring FY basis.

*f*. Budget and provide a direct fund line for the operational EMC services implemented by the US Army Signal Command (USASC).

g. Budget and provide a direct fund line for the Army J/F-12 Program on a FY basis.

*h*. Represent the Secretary of the Army for DoD management and use of the spectrum per DoD Directive 4650.1, as required.

*i.* Manage and be the principal Army negotiator for Army international discussions, either through the US Department of State on a Government-to-Government level; through "Status of Forces" in cases of host government relationships; or for ITU structure or meetings.

*j*. Validate and approve all spectrum management databases and analysis capabilities developed for Army-wide use.

*k.* Oversee the Army spectrum management automation architecture to ensure compliance with DoD automation standardization initiatives.

*l.* Coordinate with the Deputy Chief of Staff for Personnel, for priority of assignment of personnel to critical spectrum management positions identified by the MACOMs.

*m.* Assist the commanders of MACOMs in defining and justifying requirements for TOE/TDA spectrum management positions at Army posts, camps, stations, and combat training centers.

*n*. Review Army materiel objectives and requirements to identify potential effects on the spectrum per AR 70-1 and AR 25-3.

*o*. Ensure Army spectrum management policies, plans, programs, and procedures are compatible with the life cycle management of Army materiel per DoD 5000 and 8000 series guidelines and governing regulations.

*p*. Serve as the central Army manager for processing and approval of Army DD Form 1494s prior to submission of the DD Form 1494 to the Joint Spectrum Center (JSC).

q. Represent spectrum management requirements in Army Systems Acquisition Review Councils per AR 15-14.

*r*. Provide the Army member to the IRAC and direct Army participation in IRAC activities.

*s.* Provide the Army member to the Radio Communications Bureau (BR) and direct Army participation in BR study groups.

t. Provide the Army member to the USMCEB Joint Frequency Panel.

*u.* Provide the Army member to the Combined Communications-Electronics Board (CCEB) Frequency Panel.

v. Prepare and publish the ASM2P.

*w.* Monitor the spectrum management and EMC support provided by the JSC to the ARSTAF, MACOMs, and Army commands, agencies and activities. Coordinate the assignment of the JSC/Army Director with the US Army Personnel Command, and the Commander, JSC.

x. Report and coordinate the resolution of serious electromagnetic interference (EMI) incidents with appropriate Army and other Service activities, to include the JSC Joint Spectrum Interference Resolution (JSIR) team, US civilian telecommunications organizations, and host nations, if required.

y. Provide ARSTAF supervision of the US Army Signal Operation Instructions (SOI)/Communications-Electronics Operation Instructions (CEOI) Program per AR 25-1.

z. Identify spectrum management and EMC requirements in the Army E3 Program.

### 2–8. US Army Communications-Electronics Services Office (USACESO)

The USACESO is a field operating agency that assists the Army Spectrum Manager in managing the Army's spectrum management program. The USACESO performs unique and specialized operational IM functions at the direction of the Army Spectrum Manager. The USACESO will:

*a.* Serve as the focal point for the Army spectrum management program and perform assigned IM and spectrum management responsibilities as directed by the Army Spectrum Manager.

*b.* Communicate with the JSC, ARSTAF, other military services, MACOMs, other Government agencies, and activities in the private sector in the performance of assigned duties.

*c*. Command and exercise technical control over Army Frequency Management Office-CONUS (AFMO CONUS), Fort Sam Houston, TX; DoD AFC State of Arizona, Fort Huachuca, AZ (AFC-AZ); and DoD AFC White Sands Missile Range, White Sands, NM (AFC-WSMR).

*d.* Exercise technical control over the Army Frequency Coordinator, Military District of Washington (AFC-MDW), Fort Lesley J. McNair, Washington, DC; and Army Frequency Coordinator, USACE (AFC-USACE), Washington, DC.

*e*. Coordinate specialized spectrum management requirements with the USACE and the MDW. Coordinate specialized spectrum management requirements for the National Training Center (NTC)

at Fort Irwin, CA, and the Joint Readiness Training Center (JRTC) at Fort Polk, LA. Coordinate spectrum management requirements for the states of Hawaii and Alaska through the Commander, United States Army, Pacific (USARPAC).

*f.* Coordinate and obtain spectrum resources and make frequency allotments and/or assignments to Army operational requirements in the US&P.

g. Communicate directly with the Army Spectrum Manager for Army spectrum management policy and guidance on the allocation of Government spectrum resources to frequency-dependent equipment being procured or developed for Army use.

(1) Recommend to Program Executive Officers/Program Managers (PEOs/PMs), separate PMs, and others in the materiel development community the initiation of frequency supportability assessments before Requests for Frequency Allocation to Equipment (DD Form 1494) actions are submitted.

(2) Ensure that required EMC analyses are conducted before DD Forms 1494 are submitted.

(3) Ensure that the results of EMC analyses accompany each DD Form 1494.

(4) Ensure that the results of the EMC analysis describe the EMC of proposed system(s) with its coexisting electromagnetic environment.

*h*. Serve as the Army coordinator of Army test bed programs for emerging spectrum-dependent systems and models.

*i*. Implement international, national, Department of Defense (DoD), Joint, and HQDA spectrum management policy and guidance as directed by the Army Spectrum Manager.

*j.* Represent the Army on committees, groups, and organizations that address spectrum management issues when directed by the Army Spectrum Manager.

*k.* Provide the Army member on the Frequency Assignment Subcommittee (FAS), International Notification Group (ING), the Spectrum Planning Subcommittee (SPS) of the IRAC, and various ad hoc groups of the NTIA.

*l.* Provide members to the USMCEB Frequency Panel and coordinate Army participation in USMCEB working groups.

*m.* Provide the Army member to the USMCEB Frequency Panel. *n.* Process requests for spectrum resources for Army commands

*n*. Process requests for spectrum resources for Army commands and activities. *o*. Review and provide comment on spectrum proposals of other

*o*. Review and provide comment on spectrum proposals of other Government and non-Government activities and foreign governments which impact on current and future Army interest.

*p.* Identify and provide the required Army support for processing, storing, and retrieving frequency assignment records in the DoD Frequency Resource Record System (FRRS).

*q*. Serve as Army focal point for entering data in the Government Master File (GMF) and the FRRS.

*r*. Provide spectrum resources to support Army components of unified commands as required.

*s.* In coordination with the US Army Training and Doctrine Command (TRADOC), identify requirements for E3, EMC/ spectrum management training, monitoring, and education.

*t*. Serve as the focal point for the AIRP for collecting and reporting all instances of interference in accordance with procedures outlined in Chapter 3 and Appendix C.

u. Coordinate spectrum management information and requirements for Major Automated Information Systems per AR 25-3.

### 2-9. MACOM commanders

MACOM commanders will coordinate, plan, program, and fund for adequate management and supervision of the spectrum. Normally, the Deputy Chief of Staff for Information Management (DCSIM), a principal staff officer, will serve as the staff component for management and supervision of this finite resource. MACOM commanders will appoint a frequency spectrum coordinator for spectrum requirements for each Major Subordinate Command (MSC).

### 2-10. Major subordinate command (MSC) commanders

MSC commanders will coordinate, plan, program, and fund for adequate management and supervision of the spectrum. Normally,

the Assistant Chief of Staff for Information Management/Signal Officer (ACSIM/SO) will be responsible for management of this finite resource.

### 2-11. US Army installations

Installation commanders will coordinate, plan, program, and fund for adequate management and supervision of the spectrum. Installation commanders are responsible for all devices that emit electromagnetic radiations from their installation. Normally, Installation commanders will have a Director of Information Management (DOIM), as a principal staff officer, who is responsible for management of this finite resource. Where no fully resourced installation configuration exists, the owning MACOM will establish areas or regions and will designate an installation to provide information management support.

### 2–12. Major command DCSIM

The MACOM DCSIM/G-6 will-

a. Serve as the MACOM staff component for spectrum management.

*b.* Coordinate the development of the MACOM information resources management program and supervise its implementation.

c. Identify and validate requirements for authorized Spectrum Manager positions at the MACOM headquarters, adequate to discharge the functions.

*d.* Identify and validate requirements for authorized Frequency Manager positions at the MSCs headquarters, adequate to discharge assigned functions.

*e*. Identify and validate requirements for authorized Frequency Manager positions at CONUS-based MACOMs.

*f.* Identify and validate requirements for authorized Frequency Manager positions at installations.

g. Coordinate critical spectrum management personnel requirements internally with MACOM personnel and budgeting directorates and externally with the Army Spectrum Manager.

*h.* Serve as the staff proponent for the MACOM EMC Program (EMCP) which includes developing and maintaining the EMCP, funding, and interfacing, as required, with HQDA and other Army agencies.

*i*. Serve as MACOM staff proponent for the information management staff activities in the IM disciplines.

*j.* Conduct MACOM headquarters oversight for spectrum resourcing and other spectrum management related activities at MACOM installations.

*k.* Determine peace and wartime communications equipment, spectrum resources, and computer system requirements and obtain sufficient capabilities as appropriate.

### 2-13. MACOM Spectrum Manager

The MACOM Spectrum Manager will-

*a.* Manage the spectrum resources used by the MSCs, CONUSAs (as applicable), and installations.

*b*. Consolidate requirements for spectrum resources and support during mobilization and deployment planning, training, operations, and contingencies.

c. Coordinate requirements with the Army Spectrum Manager and the USACESO to determine spectrum supportability.

d. Identify and validate identified requirements for spectrum resources.

*e*. Plan, program, and budget for resources to satisfy the MACOM spectrum management program requirements per AR 1-1 and this regulation.

*f.* Identify MACOM spectrum resource requirements in the development of information systems requirements per ARs 25-1 and 25-3.

g. Assist the USACESO and other MACOMs in the development of spectrum management doctrine.

*h.* Through coordination with the USACESO, identify, budget for, and enhance the procurement of approved automated EMC tools

for the assignment of spectrum resources, determine technical solutions for spectrum related problems and enforce adherence to specified technical assignment parameters.

*i.* Prepare and provide MACOM spectrum management requirements for the ASM2P.

j. Implement the AIRP throughout the MACOM.

### 2–14. Major Subordinate Component Command (MSCC) (Corps and CONUSAs) ACSIM/SO

The MSCC ACSIM/SO will—

a. Serve as the staff component for spectrum management.

b. Coordinate the development of and implement the MSC Information Resources Management (IRM) program.

c. Identify and validate requirements for spectrum management positions at the MSCC headquarters to the appropriate MACOM.

*d.* Coordinate with the Deputy Chief of Staff for Personnel and the Comptroller to assure continuity of spectrum management. Coordinate with the MACOM DCSIM for critical shortages in spectrum management personnel.

*e*. Serve as the staff proponent for the MSCC EMCP, include its development, maintenance, funding, and interfacing with the MACOM.

### 2–15. MSCC Spectrum Manager

The MSCC Spectrum Manager will-

a. Manage the spectrum resources used by the Commander.

b. Serve as the single point of contact for the MSCC spectrum and callsign usage.

c. Consolidate SOI information for generating SOIs for the MSCC using the Revised Battlefield Electronic CEOI System (RBECS).

*d.* Serve as the consolidating, validating, and prioritizing office for requests for satellite service.

*e*. Assemble necessary information and data for Integrated Satellite Data Base submissions to assure access on needed space platforms.

*f.* Identify and validate MSCC spectrum requirements in operational, crisis, wartime, and contingency planning.

g. Identify and validate MSCC spectrum requirements for exercises, deployments, employments, and peacetime operations.

*h.* Coordinate requirements through the chain of command to obtain spectrum supportability.

*i*. Provide experts knowledgeable of spectrum engineering and modeling systems for subordinate and attached units.

*j.* Assist in the development of spectrum management doctrine. *k.* Obtain automated spectrum management, EMC assurance, SOI

generation, and EW tools and systems sufficient to assure mission accomplishment.

*l*. Ensure MSCC adherence to the spectrum resource processes. *m*. Obtain assets to enforce the MSCC's adherence to assigned spectrum resources.

*n*. Review and update MSCC operations and contingency plans to reflect changes necessary to accomplish spectrum management.

o. Present the MACOM Spectrum Manager with concerns and requirements that affect the ASM2P.

*p*. Identify MSCC spectrum resource requirements in the development of information systems per ARs 25-1 and 25-3.

q. Validate identified requirements for spectrum management.

*r*. Prepare and provide MSCC input to the MACOM for inclusion in the ASM2P.

### 2-16. Army Garrison Commander

The Garrison Commander will-

*a.* Oversee the spectrum resources as related to the performance of the garrison mission.

*b.* Support the Director of Information Management (DOIM) in the performance of its Information Management Authority.

2-17. Director of Information Management (DOIM) The DOIM will—

a. Serve as the Information Management Authority for the installation. This includes, but is not limited to:

(1) Provide operational and training spectrum resources which are authorized for such use on the installation.

(2) Process new requests for frequencies.

(3) Educate the installation and tenant activities on Army spec-

trum management procedures, doctrine, and policy.

(4) Perform limited technical analysis.

(5) Report or resolve interference problems according to the AIRP as outlined in Chapter 3 and Appendix C.

(6) Validate the installation information and functional area requirements necessary to accomplish the installation's assigned mission.

b. Through coordination with the installation commander, the MACOM DCSIM, and the USACESO, identify and forward to the installation resource manager budgeting requirements for procurement of Army-approved automated hardware and user-friendly software to perform base-level spectrum management and technical analysis functions (e.g., sustaining base information services software).

*c.* Determine peace and wartime communications equipment, spectrum resource, and computer system requirements and obtain sufficient capabilities as appropriate to the installation (e.g., increased mobilization and/or training base requirements).

*d*. Coordinate with other installation directorates to ensure that frequency-dependent equipment being developed or procured by or for use on the installation are fully spectrum supportable. See Chapter 4 of this regulation.

*e*. Assure that spectrum authorizations used within their areas of responsibility are valid. Such authorizations must be obtained per this regulation Chapters 4 and 5.

*f.* Ensure that garrison spectrum emitters operate within geographical and technical parameters to promote electromagnetic compatibility among equipment.

g. Serve as the point of contact for spectrum and non-tactical call sign requirements and usage within the installation, including tenant activities and units conducting training on the installation.

*h*. Keep records on the types of equipment, locations of equipment, and use of the spectrum and non-tactical call signs assigned to the installation.

*i*. Process and forward requests for spectrum and call sign assignment, which cannot be met from authorized resources, to the supporting AFC.

*j.* Review all spectrum assignments at least every five years or sooner, as required in Paragraph 5-3f.

*k.* Program, budget, and coordinate with the Installation Commander and appropriate directorates for financial resources for executing assigned spectrum management responsibilities per AR 1-1 and this regulation.

*l*. Perform other duties as assigned by AR 25-1.

### 2–18. Commanding General, US Army TRADOC (CG, TRADOC)

The CG, TRADOC will-

a. Implement spectrum requirements in all combat development responsibilities to include:

(1) Conducting initial spectrum supportability assessments in the analysis of identified threats and operational capability goals.

(2) Identifying spectrum analysis needs for evaluating technical concepts.

(3) Coordinating spectrum requirements in all Operational Readiness Documents with the Army Spectrum Manager per DoD 5000.2.

*b.* Ensure spectrum resource requirements are incorporated in the development of information management direction and guidance as doctrine for the Strategic, Theater/Tactical, and Sustaining Base environments.

c. Prepare the TRADOC input to the ASM2P.

*d.* Program, budget, and provide financial resources to support assigned TRADOC spectrum management responsibilities per AR 1-1 and this regulation.

*e.* Participate in the preparation of spectrum management portions of IM regulations at the request of the Army Spectrum Manager.

*f.* Review TRADOC personnel authorization documents to ensure requirements for authorized spectrum management positions are identified for all TRADOC garrisons, installations, and organic organizations. Where necessary, justify and obtain authorized spectrum management spaces. Coordinate critical spectrum management personnel vacancies with the Army Spectrum Manager.

g. Perform spectrum management responsibilities as requested by the Army Spectrum Manager.

h. Implement the AIRP throughout TRADOC.

*i*. Provide certification information for all proposed or conceptual developments that will impact on the electromagnetic spectrum.

*j.* Provide for E3, EMC/spectrum management training, monitoring, and education at all TRADOC schools.

### 2–19. Commanding General, US AMC (CG, AMC).

The CG, AMC will-

*a.* Provide E3 life cycle management guidance to Army materiel developers for research, development, acquisition, and product improvement to ensure system compatibility as outlined in Chapter 3.

*b.* Develop and provide automated tools to assist Army materiel developers in performing E3 assessments.

*c.* Develop spectrum engineering techniques and perform spectrum studies for developing and evolving Army systems and equipment developed by the MACOM and all subordinate commands.

*d.* Appoint a frequency spectrum coordinator for frequency spectrum matters, for IM and frequency spectrum requirements for each MSC.

*e*. Identify spectrum requirements in Outline Development Plans (ODPs).

*f*. Provide spectrum management support plans for all Army systems that impact on the spectrum resource.

g. Provide cost and operational equipment analysis (COEA) data to the Commander, TRADOC, as required.

*h*. Fulfill spectrum management responsibilities during research, development, and acquisition (RD&A) of Army materiel as directed by AR 70-1.

i. Prepare and forward the AMC input to the ASM2P.

*j*. Direct the AMC MSCs coordinators to interface with the USACESO and the Army Spectrum Manager, as required.

*k.* Program, budget, and provide resources to support assigned spectrum management responsibilities per AR 1-1 and this regulation.

*l*. Review AMC personnel authorization documents to identify requirements for authorized spectrum management positions at AMC installations and organizations. Where necessary, justify and obtain authorized spectrum management spaces and the qualified personnel to fill spectrum management requirements. Coordinate critical AMC spectrum management vacancies with the Army Spectrum Manager.

m. Implement the AIRP throughout AMC.

*n*. Perform spectrum management requirements as requested by the Army Spectrum Manager.

### 2–20. Commanding General, US Army Forces Command (CG, FORSCOM)

The CG, FORSCOM will-

*a.* Identify the exercise Executive Agent to coordinate the spectrum resource requirements with the USACESO for Army contingency planning, field training exercises, and command post exercises.

*b.* Identify the exercise Executive Agent to coordinate the spectrum resource requirements with the USACESO for Army component use in joint contingency planning and operations, field training, and command post exercises.

c. Implement and integrate Army spectrum management doctrine, policy, and procedures to support Army and joint contingencies, field training exercises, and command post exercises in peacetime and wartime.

*d.* When requested by the Army Spectrum Manager, represent the Army on national and international spectrum management panels that affect the active Army, ARNGUS, and USAR to ensure total force representation is present as agreements (e.g., Standardization Agreements (STANAGs), Quadripartite Standardizations Agreements (QSTAGs)) are developed.

*e*. Apply spectrum management training criteria and doctrine to the total force.

*f.* Implement total force spectrum management training. Provide recommended instructional changes to Army doctrine, procedures, and training to the USACESO and the Army Spectrum Manager.

*g.* Maintain spectrum management readiness posture for the command.

*h.* Program, budget, and provide resources to support assigned FORSCOM, and its subelements, spectrum management responsibilities per AR 1-1 and this regulation.

i. Prepare the FORSCOM input to the ASM2P.

*j.* Review FORSCOM personnel authorization documents to identify requirements for authorized spectrum management positions at all FORSCOM installations and organizations. Where necessary, justify and obtain authorized spectrum management spaces and the qualified personnel to fill spectrum management requirements. Coordinate critical FORSCOM spectrum management vacancies with the Army Spectrum Manager.

*k.* Perform direct technical and staff support as requested by the Army Spectrum Manager.

*l*. Implement the AIRP throughout FORSCOM.

*m.* Maintain the updated RBECS SOI/CEOI/JCEOI database requirement for all CONUS based forces.

### 2–21. Commanding General, US Army Reserve Command (CG, USARC)

The CG, USARC will-

*a.* Fulfill spectrum requirements in all Army Reserve responsibilities to include:

(1) Maintaining the technical proficiency and expertise of assigned spectrum managers.

(2) Identifying spectrum resource requirements in Mobilization Plans.

*b*. Coordinate spectrum requirements for Army Reserve installations and organizations.

c. Coordinate EMC analysis requirements with the Army Spectrum Manager.

*d.* Provide spectrum support plans for all Army Reserve operations that impact on the spectrum.

*e*. Prepare and forward, through the chain of command, the USARC portion of the ASM2P.

f. Program, budget, and provide resources to support assigned spectrum management responsibilities per AR 1-1 and this regulation.

g. Review personnel authorization documents to identify requirements for spectrum managers at USARC installations and organizations.

*h.* Justify and obtain spectrum manager positions to fulfill mission and function requirements. Coordinate critical spectrum manager personnel vacancies with the Army Spectrum Manager.

*i.* Perform spectrum management requirements as requested by the Army Spectrum Manager.

j. Implement the AIRP throughout USARC.

# 2–22. Commanding General, US Army Intelligence and Security Command (CG, INSCOM)

The CG, INSCOM will-

a. Assist the CG, TRADOC in Army spectrum management training to support INSCOM requirements.

*b.* Program, budget, and provide for resources for executing assigned spectrum management responsibilities per AR 1-1 and this regulation.

c. Prepare the INSCOM spectrum management input to the ASM2P.

*d.* Participate in joint military, national, and international spectrum management activities as requested by the Army Spectrum Manager.

*e*. Coordinate and assist Army commands, agencies, and activities in fulfilling assigned spectrum management responsibilities related to the INSCOM mission and requirements.

f. Implement the AIRP throughout INSCOM.

# 2–23. Commanding General, US Army Signal Command (CG, ASC)

The CG, ASC will-

*a.* Identify spectrum resource requirements to the Army Spectrum Manager in the development of information system requirements per AR 25-1 and AR 25-3.

*b.* Fulfill spectrum management responsibilities within the ASC mission for Army combat developments.

c. Perform spectrum planning and engineering for-

(1) Army air traffic control facilities through coordination with the Commander, US Army Aviation Center, Fort Rucker, AL and the USACESO.

(2) Portions of the Defense Communications System (DCS) assigned to the Army.

*d.* Prepare the USASC spectrum management input to the ASM2P.

*e.* Program, budget, and provide resources for executing ASC, and its subelements, spectrum management responsibilities per AR 1-1 and this regulation.

*f.* Review ASC personnel authorization documents to identify requirements for authorized spectrum management positions. Where necessary, justify and obtain authorization for spectrum management spaces and the qualified personnel to fill the spectrum management requirements. Coordinate critical personnel requirements with the Army Spectrum Manager.

g. Provide radio propagation technical services to the military services and other Government agencies per AR 10-87.

*h.* Perform field electromagnetic spectrum management and conduct electromagnetic radiation hazard (RADHAZ) surveys per ARs 10-87 and 5-50.

*i*. Program and obtain resources to engineer and operate mobile spectrum monitoring facilities to support Army spectrum management activities.

*j.* Perform direct technical and staff support as requested by the Army Spectrum Manager.

k. Implement the AIRP throughout ASC.

# 2–24. Commanding General, US Army Special Operations Command (SOC) (CG, USASOC)

The CG, USASOC will—

*a.* Fulfill spectrum requirements in all Army Special Operations responsibilities to include:

(1) Identifying spectrum requirements in operations and contingency plans.

(2) Maintaining continuity of operations through judicious acquisition and use of the spectrum resource.

b. Coordinate spectrum requirements for Special Forces Groups/ Teams/Organizations.

c. Coordinate EMC analysis requirements with the Army Spectrum Manager.

*d.* Provide spectrum support plans for all Army Special Operations that impact on the spectrum resource.

*e*. Prepare and forward, through the chain of command, the SOC portion of the ASM2P.

*f.* Program, budget, and provide resources to support assigned spectrum management responsibilities per AR 1-1 and this regulation.

g. Review personnel authorization documents to identify requirements for spectrum managers at SOC installations and organizations. Coordinate the filling of critical spectrum manager personnel positions with the Army Spectrum Manager.

*h.* Perform spectrum management requirements as requested by the Army Spectrum Manager.

### 2–25. Commanding General, US Army Medical Command (CG, USAMEDCOM)

The CG, USAMRMC will-

a. Fulfill spectrum management responsibilities during RD&A of Army medical materiel as directed by The Surgeon General.

b. Provide the Army Spectrum Manager with data concerning the location and electromagnetic characteristics of all Army medical materiel that either depends on, or affects the use of, the electromagnetic spectrum.

c. Analyze and measure RADHAZ to personnel from all Army electromagnetic materiel per AR 40-46 and TB Med 523.

d. Prepare the USAMEDCOM spectrum management input to the ASM2P.

e. Program, budget, and provide resources for executing assigned USAMEDCOM spectrum management responsibilities per AR 1-1 and this regulation.

f. Participate in joint military, national, and international spectrum management activities when requested by the Army Spectrum Manager.

g. Coordinate operational management of spectrum resources peculiar to USAMEDCOM (e.g., X-rays, infrared).

h. Implement the AIRP throughout the USAMEDCOM.

### 2-26. Commanding General, US Army Operational Test and Evaluation Command (CG, OPTEC)

The CG, OPTEC will-

a. Coordinate and conduct user testing and evaluation of spectrum-dependent Army materiel, tactical spectrum management systems, and spectrum management doctrine not otherwise assigned.

b. Implement the AIRP throughout OPTEC.

### 2-27. Commanding General, US Army Space and Strategic Defense Command (CG, SSDC)

The CG, SSDC will-

a. Provide E3 life cycle management guidance to Army strategic and tactical missile defense and space developers for research, development, acquisition, and product improvement to ensure system compatibility as outlined in Chapter 3.

b. Develop spectrum engineering techniques and perform spectrum studies for developing and evolving Army strategic and tactical missile defense and space systems and equipment developed by the MACOM and all subordinate commands.

c. Coordinate spectrum requirements for SSDC installations and organizations.

d. Coordinate EMC analysis requirements with the Army Spectrum Manager.

e. Fulfill spectrum management responsibilities during research, development, and acquisition of Army strategic and tactical missile defense and space systems as directed by AR 70-1.

f. Program, budget, and provide resources to support assigned spectrum management responsibilities per AR 1-1 and this regulation.

g. Prepare the SSDC spectrum management input to the ASM2P.

h. Participate in joint military, national, and international spectrum management activities as requested by the Army Spectrum Manager.

i. Coordinate and assist Army commands, agencies, and activities in fulfilling assigned spectrum management responsibilities related to the SSDC mission and requirements.

j. Implement the AIRP throughout SSDC.

### 2-28. Commanders of Reserve Component (RC) Units

Commanders of RC units, i.e., ARNGUS and USAR, will-

a. Submit applications for spectrum and call sign assignments to the AFC within whose area or jurisdiction the unit is located. A copy of the application will also be forwarded to the AFC in whose area the frequencies and call signs will be employed. The coordination channels described below are for spectrum coordination purposes only. Actions which impact on command policies and directives will be forwarded through normal command channels. The spectrum coordination channels are:

(1) ARNGUS units will forward their requests through the State Adjutant General to the supporting AFC (Appendix B).

(2) USAR units will forward their requests through the Major USAR Command (MUSARC) to the Supporting AFC (Appendix B).

b. Normally, RC units are issued spectrum assignments for training in garrison or at specified training locations. If an RC unit conducts training at a new location, it must coordinate its spectrum requirements with the appropriate installation spectrum manager. Coordination must be completed at least 90 days prior to the conduct of the training, and information copies must be sent to the supporting AFC.

c. Assure that units conducting outside CONUS (OCONUS) training obtain spectrum resources from the sponsor unit through procedures outlined in command directives, exercise directives, or other appropriate documents.

d. Assure that spectrum resources for mobilization are obtained under the provisions of Commanding General, Forces Command policy and directives.

e. Implement the AIRP throughout the RC.

### 2-29. PEO/PM and Separate PM

The PEO/PM will-

a. Conduct technical research and engineering analyses to identify potential effects on the spectrum.

b. Coordinate the resolution of critical or specialized spectrum management requirements/issues identified during the acquisition process with the Army Spectrum Manager.

c. Ensure compliance with the policies and procedures for the Army Frequency Allocation to Equipment (Army J/F-12) Program as described in Chapter 4 by initiating the DD Form 1494.

d. Provide resource funding for the conduct of spectrum supportability assessments and EMC analyses prior to and during the Life Cycle System Management Model (LCSMM) processes (refer to par 4-2).

e. Integrate existing Army-approved EMC/spectrum hardware and software tools into system procurements whenever possible. Coordinate identification, potential use, and capabilities of existing hardware and software tools with the Army Spectrum Manager.

f. Perform spectrum management requirements as identified by the Integrated Product Team.

#### 2–30. Army Director, JSC

The Army Director, JSC will-

a. Represent the Army on the JSC staff and manage Army projects being performed by the Center.

b. Provide liaison between the Army Spectrum Manager, the USACESO, and the Commander, JSC, to expedite coordination and actions requested by the Army Spectrum Manager.

c. When requested by the Army Spectrum Manager, represent the Army at meetings concerning EMC, electromagnetic vulnerability (EMV), E3, EW, and spectrum management in the technical and functional areas of diagnostics and operational EMC analyses. Provide telephonic, written, and/or facsimile reports of the meetings to the Army Spectrum Manager.

d. Provide spectrum management, EMC, EMV expertise to operational commanders, PEO/PMs, and spectrum managers.

e. Provide technical assistance during the acquisition process, Test and Integration Working Groups (TIWGs), and other committees requested and approved by the Army Spectrum Manager.

f. Provide reports on Army projects being performed by the Center and perform studies for the Army Spectrum Manager.

g. At the request of CESO, advise PEO/PMs, combat developers, and system developers on the J/F-12 (Allocation of Frequency to Equipment) process and procedures for submission of DD Form 1494 (Application for Equipment Frequency Allocation), in cooperation with policies set forth by the Army Spectrum Manager.

*h.* Develop suitable computer hardware/software suites to adequately support Army operational requirements for spectrum management and EMC.

### 2–31. Special Frequency Responsibilities for Units, Organizations, and Activities

Each unit, organization, or activity authorized to use frequencies will—

*a*. Assure that an authorization document for each spectrum resource used is retained by the operating activity.

b. Forward requests for renewal of temporary spectrum assignments at least 60 days prior to the expiration date.

c. Provide the information required by AR 380-5 for classified spectrum assignments.

d. Assure that the operation of communications-electronics (C- $E^3$ ) equipment complies with the spectrum authorization limitations and tolerances.

*e*. Assure that current Army directives and procedural publications concerning spectrum management are available and being followed.

*f*. Review spectrum authorizations annually and identify those no longer required to the supporting AFC for deletion.

g. Identify to the supporting AFC or the USACESO the unit point of contact for all spectrum matters.

### Chapter 3

# Army Electromagnetic Compatibility, Deconfliction, and Interference Control Programs

#### 3–1. Introduction

*a.* Since the primary spectrum management objective is efficient use of authorized spectrum resources, the effects of the electromagnetic environment must be considered when developing spectrum-dependent materiel.

b. As the spectrum becomes congested with more users, and as the spectrum is reallocated away from military use, the capability of equipment to operate without causing or experiencing unacceptable EMI becomes critical. The term used for the process of reducing EMI and thus optimizing the use of the spectrum for the spectrum management, intelligence and EW (IEW) communities is deconfliction.

c. The Army continues to assess measures for implementing the enforcement of proper spectrum use. Army users of the spectrum rely upon spectrum sharing agreements with other agencies, host nations, the FCC, and other Services in order to meet their increasing spectrum requirements. Increases in the density of spectrumdependent equipment operating in the same bands result in increased operational conflict and a higher potential for interference. Without proper application of deconfliction procedures for the control of interference, operational failures will occur and liability claims may result from interference to commercial systems.

### 3–2. Scope

This chapter applies to all active Army, ARNG, and USAR operations and the operation of spectrum dependent equipment required by the materiel development community involved. Such operations involve equipment herein defined as that designed to function using the radio frequency electromagnetic spectrum. Secondary effects of electromagnetic interference (EMI) to or from electric or electromechanical equipment, i.e., computers or electric motors, which are not intended to use the electromagnetic spectrum, are the responsibility of the Army E3 and equipment development organizations. The environmental effects of these devices must be considered and minimized through system analysis and design concepts. Overall system compatibility between spectrum-dependent systems must be assured.

### 3–3. Management Requirements for Achieving Electromagnetic Compatibility (EMC)

*a.* The Army considers EMC an essential characteristic of spectrum-dependent equipment in the same sense as reliability and the ease of maintenance. Thus, EMC analyses must be conducted to support each step of the design, development, acquisition, and use of this equipment.

b. As a time cost consideration, prior to entering any of the four phases of the Life Cycle System Management Model (LCSMM) or procuring commercial off-the-shelf (COTS) spectrum-dependent equipment, it is highly recommended that concept evaluators, PEOs, PMs, combat developers, and, in some cases, units themselves request that a spectrum supportability assessment be conducted to determine if the proposed concept or requirement will meet spectrum supportability and equipment compatibility requirements in all CONUS and/or OCONUS environments (see Chapter 4).

*c.* Each active Army, ARNGUS, and USAR organization and each individual that participates in managing the acquisition and use of spectrum-dependent equipment must stress the need to achieve EMC by accomplishing the following actions:

(1) Implement EMC decisions and actions during the LCSMM. EMC decisions and actions that occur during the four phases of the LCSMM are:

(a) Concept Exploration and Definition Phase.

*1.* For telecommunications equipment, there are two major milestones in the Concept Exploration Phase. The first milestone is the preliminary selection of the frequency band, including spectrum supportability, type of channelization, and other main characteristics of the system. The second milestone is the submission of a Stage 1, 2, and 3 DD Form 1494. Ensure the DD Form 1494 contains sufficient data for a subsequent review and evaluation of spectrum supportability. If required, obtain a spectrum assignment for experimental testing. Obtain an approved Stage 3 application or waiver for Milestone I Army Systems Acquisition Review Council or appropriate In-Progress Review (IPR).

2. For non-telecommunications systems (e.g., generators, vehicle motors, etc.) which do not use the spectrum, determine system technical characteristics. This determination must contain sufficient information to permit evaluation of the potential for unintentional susceptibility or unintentional radiation interference to the environment where their equipment will be operating.

(b) Demonstration and Validation Phase.

*1.* For this phase, first prepare the EMC-related portion of the equipment performance specifications for the prototype equipment. Second, develop plans for the EMC portion of developmental and operational test. Third, review results of EMC testing as part of Development Test (DT) and Operational Test (OT). Fourth, verify that potential EMC problems have been averted or can be expected to be resolved during engineering development.

2. Before Milestone II or appropriate IPR, the Stage 4 DD Form 1494 or waiver must be approved. The DD Form 1494 and all further forms must include the new data if there are changes to previously approved frequency allocations.

(c) Engineering and Manufacturing Development Phase. For EMC considerations and actions, complete the following actions:

*1.* Prepare EMC portions of equipment development specifications.

2. Prepare EMC portions of test plans for DT and OT.

3. Review EMC test results.

4. Verify EMC performance.

5. Prepare EMC portions of equipment specifications for initial production.

6. Obtain an operational frequency assignment if required.

7. Approve TOE material and training material.

(d) Production and Deployment Phase. For EMC considerations and actions, first prepare EMC portions of the production equipment specifications. Then verify performance by reviewing operational performance reports and reports of interference or EW. (2) Conduct EMC environmental surveys. Conduct surveys either prior to or when installing or modifying spectrum-dependent equipment. If desired, these surveys may be accomplished by the USAISEC (AMSEL-IE-TS), Fort Huachuca, AZ 85635-5300, or the JSC, Army Director (AR), 120 Worthington Basin, Annapolis, MD 21402-5064.

(3) Resolve interference. Interference can be classified as local, regional, and/or national. Commands and agencies will attempt to resolve instances of interference within their area of responsibility. Reporting procedures for interference problems will be described in Paragraphs 3-8b and 3-8c and Appendix C.

### 3–4. USASC Services for the Electromagnetic Compatibility Program (EMCP)

*a.* The USASC Spectrum Engineering Branch is the implementing organization for the operational EMC and propagation engineering (PE) program areas. These responsibilities are contained in AR 10-87.

*b*. The USASC will provide an operational EMC assurance capability to—

(1) Detect, report, and coordinate the solution and correction of operational EMC problems with the USACESO, and when necessary, the JSC.

(2) Assist the Army Spectrum Manager and the Army commanders in resolving operational EMC problems by providing technical assistance in spectrum engineering, propagation services, and the electromagnetic environment using both on- and off-site personnel and facilities.

(3) Conduct Electromagnetic Radiation Hazard (EMRH) surveys at conventional and special munitions storage and handling sites/ areas for all levels of command.

(4) At the direction of the Army Spectrum Manager, and in coordination with the JSC Army Director and the CG, TRADOC, develop, manage, and operate EMC assurance training programs.

c. Provide a Quick Reaction Capability (QRC) for EMC assurance capable of responding to field missions immediately and providing an EMC team on-site within 24 to 72 hours to resolve interference problems. Required resource planning data will be coordinated with the USACESO.

*d.* Provide EMC engineering and analysis support to eliminate potential EMC/EMI problems for planned system upgrades, modifications, or new C-E system installations.

*e*. Provide radio wave propagation technical services to the Army and to other military services and Federal agencies as resources permit. Perform radio wave propagation path surveys, analyze radio frequency (RF) system performance, and design and analyze antennas. The results of the technical services (e.g., surveys, analyses, etc.) will be forwarded to the USACESO and the JSC Army Director.

*f.* Plan, program, budget, and provide resources for executing assigned spectrum management, EMC assurance, and EMC testing and evaluation responsibilities per this regulation.

g. Address requests for EMCP assistance to Commander, USASC, ATTN: DSED, Fort Huachuca, AZ 85613-5300, DSN 879-3068 (Commercial 520-538-3068).

# 3–5. US Army Test and Evaluation Command (USATECOM) Services for the EMCP

*a.* USATECOM has the responsibility to provide a variety of EMC, EMV, database services and spectrum management process support to developers of Army electromagnetic spectrum-dependent equipment. These services are provided through the operation of the following subordinate elements located at the Electronic Proving Ground (EPG): the Battlefield Electromagnetic Environments Office (BEEO), Electromagnetic Environmental Test Facility (EMETF), and the EMI/TEMPEST branch.

b. The BEEO will-

(1) Support the Army EMCP through development, operation, and maintenance of databases for C-E compatibility and vulnerability analyses and concept studies.

(2) Develop, maintain, and operate the database element of the EMCP to provide timely scientific and technical support to the Army spectrum management program. These databases include equipment characteristics and measurements, organizational data such as TOE and Basis of Issue, tactical concepts and doctrine, and threat documentation.

(3) Develop, maintain, and operate simulated tactical deployments, based on approved scenarios, to include geographical locations, communications netting, frequency assignments and spectrum use. These deployments are down to the individual equipment operator level for US, Allied, and Threat forces.

(4) Conduct research and studies to design and develop other systems/capabilities as required to fulfill special database requirements.

c. The EMETF will-

(1) Assess the ability of Army systems and equipment to operate compatibly in their intended operational electromagnetic environment. Assessments will include the activities relevant to the C-E equipment systems under consideration and doctrine of both threat and friendly forces. Provide assessment service relative to both unintentional (compatibility) and intentional (vulnerability) interference.

(2) Assess the influence of the intended operational electromagnetic environment on Army systems and equipment concepts and the doctrine for their implementation.

(3) Maintain its support capability through three main capabilities which are:

(a) Electronic and electromagnetic measurements of equipment parameters and performance including electro-optical equipment employed by the US, Allied, and Threat forces.

(b) Databases describing the structure, activity, equipment parameters and geographic environment of deployments up to the level of a deployed Army corps, and its opposing threat force, in their expected tactical situation.

(c) Analytical capability which incorporates the electromagnetic results of the measurements and deployment database into computer models of C-E equipment employed by the Army in opposition to a corresponding threat force. The analysis will give measures of performance for systems being evaluated in either a one-on-one or many-on-many electromagnetic emitter environment.

*d.* The EMI/TEMPEST branch will make a TEMPEST assessment of the equipment or system ability to process classified information without risking compromise. The branch has a measurement capability that detects emanations from electronic and electro-mechanical equipment conformity to requirements in this area specified by NSA. Measurement data can be analyzed to ascertain the probability of the system under test being able to complete an assigned mission without risking compromise of classified information.

*e*. The EMETF facility and the EMI/TEMPEST branch will provide support for all phases of the LCSMM.

f. EMEFT services are provided to Army agencies on a cost reimbursable basis.

g. Address requests for EMCP services to Commander, EPG, ATTN: STEWS-EPG-TT, Fort Huachuca, AZ 85613-7110, DSN 879-4860 (Commercial 520-538-4860).

### 3-6. JSC Services for the EMCP

*a*. As defined by DoD Directive 3222.3 and DoD Regulation 5000.2-R, the JSC is a Joint services activity that provides frequency spectrum engineering and EMC analysis support to all DoD components developing or operating telecommunications equipment.

*b.* Army components, as well as the joint services, have access to databases that contain electromagnetic environmental, electro-optical, equipment electromagnetic characteristics, frequency assignment and use, space systems orbital, tactical deployment, and topographical data. The JSC has the responsibility to maintain the frequency resource file for the DoD.

c. The JSC has both automated and manual EMC models ranging from antenna and propagation models to terminal-device-performance models. The JSC has adapted in-house models to meet the

specific needs of other DoD analysis centers and services operational units.

d. JSC services are available on-site at Army field commander locations worldwide.

e. The JSC provides services to Army agencies on a cost reimbursable basis.

*f.* Army organizations requesting spectrum management support or analytical services pertaining to existing or anticipated EMC problems should address their requests to the Commander, JSC, ATTN: AR, 120 Worthington Basin, Annapolis, MD 21402-5064, DSN 281-2103, Commercial (410) 293-2103, facsimile (410) 293-2631.

### 3–7. Deconfliction

*a.* Deconfliction is a systematic management procedure to coordinate the use of the electromagnetic spectrum for operations, communications, and intelligence functions. It is an element of electromagnetic spectrum management.

b. Deconfliction is comprised of three categories. They are:

(1) Friendly communications equipment interfering with friendly electronic support (ES) equipment (interfering with intercept operations).

(2) Friendly EA equipment interfering with friendly C-E equipment (jamming missions interfering with normal C-E operations).

(3) Hostile EA equipment interfering with friendly C-E equipment.

c. The Army Spectrum Manager shall advise the ARSTAF, MACOMs, and other Army organizations of trends in very high power emitters as a result of its coordination at Joint, national, and international levels; and, with awareness of the susceptibility levels identified by the E3 program, ensure that the spectrum management process disseminates appropriate alerts and coordination.

*d.* To accomplish deconfliction, Army battlefield spectrum managers continually coordinate the assignment of frequencies to battlefield systems to minimize electromagnetic spectrum conflicts or interference with the IEW units of friendly forces.

*e*. Information on the development and availability of automated deconfliction tools is available from the USACESO/(SAIS-PAS-M), DSN 227-0199.

### 3-8. Army Interference Resolution Program (AIRP)

a. The AIRP revolves around four functions:

(1) Direction Finding (DF). A DF capability is often the key to locating the source of interference and is an integral part of resolving and analyzing incidents and problems. The degree of accuracy is dependent on the environment and frequency band.

(2) Signal Monitoring. Signal monitoring or spectrum surveillance incorporates a frequency spectrum analyzer or surveillance receiver covering all spectrum bands of use. These systems perform real-time evaluation of spectrum usage and interference in a specific area.

(3) Signal Analysis. Analysis of DF and monitoring data is required to determine the source of interference and misuse of the spectrum.

(4) Transportability/Mobility. Degree, circumstances, and geographic location of the types of interference incidents and problems will determine transportability and mobility requirements. Mobile/transportable DF and monitoring equipment is a requirement for tactical units, training areas, and for incidents not necessarily confined to a specific geographical area. Man portable equipment should also be considered for certain instances and conditions. Fixed equipment would be required for those areas which require real-time solutions in a defined geographical area.

b. Responsibilities.

(1) The Army Spectrum Manager is responsible for reporting and coordinating the resolution of serious incidents of spectrum interference with appropriate HQDA and other Service activities, to include US civilian telecommunications organizations and host nations, if required. (2) USASC is responsible for the dispatch of monitoring and DF teams to the affected site within 24 to 72 hours.

(3) AMC will maintain a database of spectrum interference incidents reported to HQDA/USACESO.

*c*. The procedures for reporting incidents of spectrum interference to Army forces located within CONUS are:

(1) CONUS-based Army forces involved with sustaining base operations will report spectrum interference to the supporting DOIM, who will make every effort to resolve problems locally within the installation/MACOM-owned assets.

(2) If the spectrum interference cannot be resolved locally, the supporting DOIM will report the problem to the supporting DoD AFC as depicted in Figures B-1 and C-2.

(3) If the AFC cannot resolve the problems regionally with other federal agency field offices (e.g., FCC, Federal Aviation Administration (FAA) Regions), the AFC will request the spectrum manager submitting the spectrum interference report to resubmit the spectrum interference report through command channels.

*d.* The procedures for reporting incidents of spectrum interference to Army forces located OCONUS are:

(1) Army forces stationed OCONUS will report spectrum interference to the supporting Joint Frequency Management Office (JFMO) in accordance with the guidelines contained in FM 24-35-1 and the JSIR Program. When required, reports may also be submitted to the spectrum managers supporting Army community commanders.

(2) Most instances of spectrum interference will be resolved locally with the Division and Corps. For those incidents of spectrum interference which cannot be resolved at echelons below Corps, a report will be sent to the Theater Army/CINC Joint Frequency Management Office (JFMO) headquarters for resolution. Where required, the JFMO will interface with the host nation through the Status of Forces Agreements.

(3) In cases where the JFMO cannot resolve the spectrum interference problem, Army users will submit reports to HQDA, ATTN: SAIS-PAS-M (The Army Spectrum Manager) and the USACESO. Army users may also be required to report to other activities through existing CINC directives.

(4) Interference to high frequency (HF), satellite, or troposcatter communications may involve reporting the interference to activities outside the Theater/CINC area of responsibility.

*e.* Incidents of harmful interference, whether occurring within CONUS or OCONUS, which cannot be resolved locally or regionally, may cause a serious threat to life or limb (i.e., interference to air traffic control, detonation of fuses to explosives, etc.); or may cause high-cost expenditure to the Army (e.g., delay or cancellation of a test), and therefore should be reported through Army interference resolution organizations (see Figure C-2). Action will be initiated by these offices to coordinate resolution of the interference problem with appropriate Army activities.

### 3–9. Army E3 Objectives Relating to Spectrum-Dependent Devices

a. The Army E3 objectives that relate to spectrum-dependent devices are:

(1) To achieve EMC for all spectrum-dependent equipment operated by the Army components.

(2) To attain built-in design compatibility, rather than achieve EMC through the use of planned product improvements (PPI) after development.

(3) To promote EMC philosophies and techniques in the acquisition and use of spectrum-dependent equipment.

(4) To promote total EMC test and analysis techniques of all systems with spectrum-dependent materiel.

b. Detailed procedures for implementing the Army  $E^3$  Program are contained in Army Acquisition Executive (AAE) Policy Memorandum 91-3.

### Chapter 4 Frequency Allocation-to-Equipment Process

#### 4–1. Introduction

*a.* The Frequency Allocation-to-Equipment Process supports the Army spectrum management goal as stated in Paragraph 1-6. Due to an exponential increase in electromagnetic spectrum use resulting from technological advancements, and as the spectrum is reallocated away from military use, achievement of this goal is critical. In the research, development, production, and procurement cycle, policy implementation must begin as early as possible. A means for early policy implementation is the Frequency Allocation-to-Equipment Process. This process prescribes policies, responsibilities, and procedures for Army commands, agencies, PEOs/PMs involved in conceptualizing research, development, and lease or use of spectrum-dependent materiel.

b. The Frequency Allocation-to-Equipment Process is a spectrum requirement and exists to determine that:

(1) If equipment can operate in spectrum bands per the national and international tables of spectrum allocation.

(2) Equipment being brought into the Army inventory conforms with applicable spectrum management regulations, directives, standards, and specifications.

c. A basic goal of the process is to assure spectrum supportability and equipment EMC. Procedures and requirements outlined in this chapter apply variously to different types of equipment methods of acquisition. See Table 4-1 for the summary of equipment requiring spectrum applications.

### 4-2. Policies Concerning DD Form 1494

*a.* Funds for the research, development, production, procurement, modification/production improvement, and lease or use of spectrum dependent equipment will not be released by the obligating authority until an approved DD Form 1494 has been obtained per Paragraph 4-2f. DoD Regulation 5000.2-R, paragraph 4.4.7, mandates the determination of spectrum supportability prior to initiating cost estimates for development or procurement. The objective of the DD Form 1494 is the determination of spectrum supportability of spectrum dependent equipment and systems. Waivers for submitting the form are in Paragraph 4-6.

*b*. The DD Form 1494 establishes that a particular equipment/ system has valid spectrum requirements and is not an approval document for use of frequency(ies). An approved DD Form 1494 is necessary to obtain a frequency assignment.

c. Compliance with guidance in the following documents is mandatory:

(1) The approved DD Form 1494.

(2) Memoranda and letters of transmittal (LOTs) from the Army Spectrum Manager, the USMCEB, and possibly, the NTIA, for equipment that operates within the US&P.

*d.* Contracting officers will comply with Defense Acquisition Regulations (DFARS) 235.071 and use the contract clause found at DFARS 252.235-7003 in all solicitations and contracts requiring frequency authorization.

#### Table 4–1

### Summary of Equipment Requiring DD Form 1494

Equipment Category	DD Form 1494 Required	Applicable Paragraph
All telecommunications emitters, e.g., radars, navigation, meteorological, ATC, mari-	Yes	4-3b(1)
time		
EA Equipment/Threat Simulators	Yes	4-3b(2)
Receivers and antennas developed independently	Yes	4-3b(3)
Certain types of nontactical, commercial, intrabase radios	Yes	4-3b(4) or
Operations/test at—	Determined by local DoD AFC	USAKÁ 4-
White Sands Missile Range		6b(1)
Army Electronic Proving Ground		
Yuma Proving Ground		
Kwajalein Atoll (USAKA)		
Incidental radiation devices	No	4-6b(2)
Fuses and detonators	No	4-6c(5)
Commercial satellites	No	4-6c(6)
Low power devices	Yes	4-6c(1)
Industrial, scientific, and medical equipment	Yes	4-6c(2)
	(see limitations in applicable paragraph)	( )
Ultrasonic equipment	Yes	4-6c(3)
	(see limitations in applicable paragraph)	(.)
Lasers	Determined by USACESO/SAIS-PAS-M	4-6c(4)
Terms defined in Glossary.		

*e.* Prior to entering the four phases of the LCSMM or procuring COTS spectrum-dependent equipment, it is required that concept evaluators, PEOs, PMs, combat developers and, in some cases, units themselves request that a frequency supportability assessment be conducted. These assessments will be conducted to determine if the proposed equipment will meet spectrum supportability and EMC in its intended operating environment. These assessments can take from 3-9 months to perform, users should plan accordingly in both time and money. Spectrum assessment can be conducted by several different organizations (e.g., JSC, ISEC).

*f.* An approved DD Form 1494 or waiver (Paragraph 4-6a) is required, as indicated below, in the LCSMM (AR 70-1 and DA Pam 11-25), and DoD Regulation 5000.2-R paragraph 4.4.7.

(1) During Phase I (Concept Exploration and Definition Phase)— Conceptual and Experimental Application (Stages 1 and 2).

(2) At Milestone I for systems that emit or receive Hertzian

waves; at appropriate IPR for non-major systems—Developmental Application (Stage 3).

(3) At Milestone II for systems that emit or receive Hertzian waves; at appropriate IPR for non-major systems—Operational Application (Stage 4).

#### 4-3. Application for Spectrum Certification

*a.* Instructions for completing DD Form 1494 are included with the form. The certification request will include modifications of military standards, if needed, and supporting rationale, EMC studies already performed, a concept of operations, and a statement releasing the information to foreign nations (if equipment is intended to be used OCONUS). The completed form and supporting documents will be sent through command channels to USACESO (SFIS-FAC-P, 200 Stovall St., Room 9S65, Alexandria, VA 22332-2200).

b. A DD Form 1494 must be completed for all systems and

equipment that emit or receive Hertzian waves. Those systems and equipment include:

(1) All telecommunications emitters. All telecommunications equipment except those exempted by Paragraph 4-6.

(2) EA equipment and Threat simulators. Developers of EA equipment and Threat simulators will forward to USACESO, through appropriate command channels, a completed DD Form 1494 for each end-item of equipment. USACESO will provide MACOMs a copy of the DD Form 1494 for their use in developing spectrum engineering tools and conducting EMC analyses. When necessary, USACESO will, after coordination with appropriate command spectrum coordinators, furnish guidance to the developing activity regarding impact of equipment use on established Army services.

(3) Receivers and antennas developed independently. Developers of receivers and antennas will complete a DD Form 1494 for each receiver and antenna developed independently of transmitters or receivers.

(4) Non-tactical, commercial, land mobile radios (LMRs) procured without a DD Form 1494/ military nomenclature. To maintain accurate information and databases of Army telecommunications equipment using Army spectrum resources, MACOMS will submit a DD Form 1494 for those equipment already procured without a DD Form 1494 and in use at the MACOM. For anticipated procurement of new non-tactical, commercial, intrabase radios, commanders of MACOMs (posts, camps, and stations) must submit a DD Form 1494 for approval prior to procurement of the equipment to the USACESO (SFIS-FAC-P). Also, prior to contractual commitments, frequency assignments must be available as determined by the USACESO or AFC.

### 4-4. Processing DD Form 1494

*a.* Army personnel involved in the conceptualizing of research, development, production, procurement, modification/product improvement, and lease or use of spectrum-dependent materiel will initiate the Frequency Allocation-to-Equipment Process by completing and submitting DD Form 1494 with supporting documents (e.g., spectrum supportability/EMC assessments) to addressees indicated in Paragraph 4-3a.

*b.* USACESO will review the DD Form 1494 for completeness, accuracy, and initial determination of spectrum supportability. The DD Form 1494 will then be distributed to other organizations in parallel. This distribution may include the USMCEB, AFCs, holders of DD Form 1494s, the JSC (AR), and the NTIA/SPS if appropriate.

c. The military departments, unified commands CINCs as appropriate, and the JSC will review the form and provide EMC comments on the DD Form 1494 to the J-12 permanent Working Group.

*d.* The J-12 permanent Working Group will review the DD Form 1494 and the EMC comments to determine if the equipment or system can receive spectrum support in the geographic areas specified on the form. If the equipment or system is to be used outside of the US&P, unified command (CINCs) and host nation concurrence is required prior to approval.

*e*. The USMCEB guidance memorandum will outline general considerations and restrictions that apply to particular equipment. The guidance will also include recommendations of specific actions for approval or implementation. The memorandum will be returned to the originator through the USACESO. In addition to the memorandum, USACESO will prepare a LOT for approval by the Army Spectrum Manager that contains additional guidance, restrictions, and specific actions. The originator must comply with the instructions in both the memorandum and the LOT.

*f.* If there is a conflict between the USMCEB and the Army Spectrum Manager, then the papers (USMCEB memorandum, the DD Form 1494, and the JSC EMC comments) will be forwarded to the USMCEB Joint Frequency Panel (JFP) for resolution.

g. If the memorandum and the Army Spectrum Manager's LOT indicate approval with implementing guidance, the memorandum is then signed by the J-12 Working Group chairman.

*h.* The USACESO and USMCEB reviews normally take 90 to 120 days, depending on the type of equipment and its operational

environment. It may take 6 to 12 months to coordinate with host countries for frequency spectrum support. Therefore, originators will submit the DD Form 1494 during the LCSMM, as follows:

(1) Stage One—Planning (Conceptual); Phase I, LCSMM: submit per type of system below:

(*a*) Space Systems (including earth stations and terminals): submit 4 to 6 years, but not less than 2 years, before planned satellite launch.

(b) Terrestrial Systems: submit 3 years, but not less than 1 year, before planned procuring for experimentation.

(2) Stage Two—Experimental; Phase I, LCSMM: submit not less than 6 months before procuring for experimentation.

(3) Stage Three—Developmental: submit in order to be approved before Milestone I Review, or appropriate IPR, per type of system below:

(a) Space Systems: submit 2 years, but not less than 1 year, before a development contract award.

(b) Terrestrial Systems: submit not less than 6 months before a development contract award.

(4) Stage Four—Operational: submit in order to be approved before Milestone II Review or appropriate IPR. Submit at least 6 months before acquisition actions for all equipment that use satellites or spacecraft or have significant impact on the electromagnetic frequency spectrum. For all other equipment, submit at lest 4 months before acquisition actions. Data on DD Form 1494 should be validated through measured data when possible.

### 4–5. EMC and DD Form 1494 Reviews Required by the NTIA $% \left( {{\left( {{{\rm{DD}}} \right)}_{\rm{T}}}} \right)$

*a.* Reviews of certain new Federal Government telecommunications systems by the SPS are required prior to committing funds, entering into contracts, and assigning frequencies. These reviews apply to the following types of telecommunications systems:

(1) New space telecommunications systems or subsystems, as well as major modifications to existing systems, that involve the use of satellites or spacecraft.

(2) New terrestrial telecommunications systems or subsystems, as well as major modifications to existing systems, that have significant impact on the use of the electromagnetic spectrum.

(3) New, major, digitized voice systems, including modifications to existing systems or subsystems, that involve the 30-88, 138-174, and 406-420 MHZ bands.

(4) All systems in the 14.4-15.35 GHZ band.

(5) Systems or facilities that the NTIA, IRAC, or other Federal Government agencies refer to the SPS. Referral may result from factors such as system costs, importance, or from estimates of unusual or potential impact on other spectrum uses.

*b.* USACESO will submit the DD Form 1494 to the SPS for review. After reviewing the proposed system, the SPS will submit recommendations, with supporting documents on spectrum availability, through the IRAC to the NTIA.

c. The NTIA review will be returned to the Army originator through USACESO. In addition to the NTIA review, USACESO will prepare a LOT for approval by the Army Spectrum Manager which may include additional guidance, restrictions, and specific actions. The originator must comply with the instructions in the review and the LOT. The originator will submit a letter to USACESO stating how the recommendations and guidance of the review and LOT will be implemented. USACESO will keep the Army Spectrum Manager, MACOM spectrum coordinators, and other interested Army commands and agencies informed of the originator's progress in complying with the review and LOT implementation guidance.

*d.* If the NTIA review and the Army Spectrum Manager LOT both indicate disapproval, the originator will resubmit the DD Form 1494 when corrective action has been completed.

### 4–6. Waivers and Exceptions

*a.* Allocation approval waiver. If exceptional circumstances will not allow enough time (normally 90-120 days) to process a DD Form 1494, a "Release of Funds Waiver" may be requested. Such a

request will be forwarded to USACESO (SFI-FAC-P, 200 Stoval St., Alexandria, VA 22332-2200) along with the DD Form 1494 and special justification. The justification must show that:

(1) Eventual approval of the form can reasonably be anticipated.

(2) Waiting the normal processing time may not be in the best interest of the Army.

(3) As a result of coordination with the USACESO, it has been determined that an operating frequency could be assigned. USACESO will forward the "Release of Funds Waiver" to the Army Spectrum Manager (SAIS-PAS-M) for his signature of approval.

*b.* Exceptions. Exceptions to the requirements for submitting DD Form 1494 are limited to the following:

(1) White Sands Missile Range (WSMR), NM; the Electronic Proving Ground (EPG), Fort Huachuca, AZ; Yuma Proving Ground (YPG), AZ; and Kwajalein Atoll (USAKA). These installations are part of the National Military Test Ranges operated under the responsibility of the Army. The DoD AFC WSMR, and the DoD AFC AZ, have allocation approval authority for all emitting and radiating equipment involved in range activities in their geographic areas. Equipment approved by the DoD AFC for use in support of range activities may or may not require a DD Form 1494. The DoD AFC will independently decide on each equipment proposal. Approval by the DoD AFC does not assure that the form can or will be subsequently approved for use in other geographic areas.

(2) Incidental radiation devices. Exceptions are usually allowed for incidental radiation devices that radiate RF energy in the course of normal operation but are not designed to intentionally generate RF energy. Examples are clothes dryers, washing machines, electric typewriters, computers, and microwave ovens. These devices do not require a DD Form 1494 if the operating agency will either eliminate any harmful interference caused to an authorized radio service or obtain a waiver. For this equipment, the waiver process is covered by MIL-STD-481A, Paragraph 6.5. Waivers must be approved or disapproved by the local Materiel Review Board (MRB) or, in the absence of such MRB, the Contract Administrative Officer.

*c.* Partial processing. The Army Spectrum Manager may approve some equipment for use without completing the Frequency Allocation-to-Equipment Process, but only after the DD Form 1494 is submitted. This equipment is listed below:

(1) Low power devices. Low power or restricted radiation devices as specified in the NTIA Manual.

(2) Industrial, scientific, and medical (ISM) equipment. Radiation devices which use radio waves for ISM purposes. Purposes include the transfer of energy by radio. ISM equipment will not be used for radio communication unless constructed and operated per the limitations defined in the NTIA Manual.

(3) Ultrasonic equipment. Any equipment which generates RF energy and uses that energy to excite or drive an electromechanical

transducer for the production of sonic or ultrasonic mechanical energy. This energy is for ISM use and will not be used for radio or other communication purposes unless constructed and operated in accordance with the limitations defined in the NTIA Manual.

(4) Lasers. Use of lasers for telecommunications is an area of considerable study and research. In those instances where laser coding techniques are used in range finding, target acquisition/identification research and development, or any other type of laser coding experimentation that will lead to spectrum occupancy, materiel developers must coordinate with the USACESO to determine if a DD Form 1494 is required.

(5) Fuses and detonators. Items of materiel used solely to activate a fuse or detonator do not require a DD Form 1494. In some congested areas, controls may become necessary to avoid EMI caused by or to these devices. In a noncombat situation, the operation of such devices must not cause interference to authorized spectrum users. Assistance and guidance in the selection of appropriate operating spectrum bands is available from the USACESO.

(6) Commercial satellites. Satellites are registered by commercial companies and operate in commercial spectrum bands. A DD Form 1494 is not required.

*d.* Radar design objectives and engineering criteria. The provisions of MIL-STD-469A and the NTIA Manual, Section 5.3, are mandatory for all new radars. However, existing radar systems and equipment will not be converted solely to comply with the requirements of Section 5.3. In any instance of harmful interference involving the use of conforming and nonconforming equipment, the activity using the nonconforming equipment will make adjustments to eliminate the interference.

### 4-7. EMC Standards for Telecommunications Equipment

a. Categories. Certain military and national standards for frequency spectrum-dependent equipment should be reviewed before the DD Form 1494 is completed. Some of these major standards are shown in related references in Appendix A and in Table 4-2. The equipment limitations, as shown on the DD Form 1494, must be compared to these standards during the Frequency Allocation-to-Equipment Process. Standards are in the following categories:

(1) International standards and agreements. Many factors that influence standards, equipment operation, and design are related to international agreements. Some of these, such as international aviation agreements, affect worldwide operations. Others, such as the North Atlantic Treaty Organization (NATO) standardization agreements, affect C-E operations only in certain areas of the world. The ITU Radio Regulations (RRs) and the BR recommendations constitute the basic international agreements affecting spectrum management.

Table 4–2 Applicability of Federal Government/Military Documents to Common Equipment Characteristics						
Characteristics	MIL-STD	MIL-STD	MIL-STD	NTIA	NTIA	NTIA
Transmitter:	188C	461	1572	5.1	5.3	5.8
Harmonic <sup>1,6</sup> - attenuation	X <sup>1</sup>	Х	Х	Х	Х	Х
Spurious attenuation	X <sup>1</sup>	Х	Х	Х	Х	X <sup>5</sup>
Tunability <sup>3</sup>	Х		Х	Х	Х	
Frequency stability	X <sup>2</sup>		Х	Х	Х	Х
Maximum emission bandwidth	X <sup>1</sup>	Х			Х	
Receiver:						
Spurious response	X <sup>1</sup>	Х	Х		X <sup>4</sup>	
Image response			Х		Х	
Frequency stability			Х		Х	

Table 4–2 Applicability of Federal Government/Military Documer	nts to Commo	on Equipmen	t Characteristi	cs—Continue	ed	
Characteristics Transmitter:	MIL-STD 188C	MIL-STD 461	MIL-STD 1572	NTIA 5.1	NTIA 5.3	NTIA 5.8
Minimum required acceptance bandwidth					X <sup>4</sup>	
votes: In some bands, MIL-STD 461 applies for transmitter attenuation and receiver response requirements.						

<sup>2</sup> MIL-STD 188C requires a maximum necessary bandwidth.

<sup>3</sup> Where a standard has a channelization requirement, it is considered under tunability.

<sup>4</sup> Paragraph 5.3.1 (Criteria B), NTIA Manual, has no image response or antenna requirements - see paragraph 5.3.2 (Criteria C), NTIA Manual.

<sup>5</sup> Both the military and national standards define spurious emissions to include harmonic emissions; however, the military has separate requirements forharmonic attenuations.

<sup>6</sup> Identification, Friend or Foe (IFF) Radar is lowered for criteria (FederalAviation Administration Regulation 1010.51A and Air Traffic Control RadarBeacon Systems, IFF Mark XII Systems (AIMS) Technical Standard 65-1000). Where the criteria differs, the AIMS standard governs.

(2) US national standards. These apply to the design and use of equipment that is common to the military, other Federal Government agencies, and civil users. NTIA Manual, Chapter 5, lists many standards that apply to Federal Government agencies, including the DoD. Military equipment that will be used for tactical exercises or training in the US&P is subject to the provisions of the NTIA Manual. Included in the NTIA Manual are radar design criteria, standard emission designators, channeling plans, out-of-band emission limits, and other criteria. Army systems will be judged against the NTIA Manual throughout the system life cycle, and the NTIA Manual, Paragraph 8.3, should be followed when initiating a DD Form 1494.

(3) *Military documents*. Many military documents contain information important to the reduction of EMI. These documents are available through publication distribution channels. See Table 4-2 for applicability of Government and military documents to common equipment characteristics.

(4) USMCEB documents. USMCEB documents may provide design guidance and are referenced when developing new equipment and assigning frequencies for existing equipment. Often these documents are issued as standards, plan, etc. In general, those documents that affect equipment allocations are made available to the concerned Army commands. If not, contact USACESO (SFIS-FAC-P) for guidance.

(5) Range Commanders' Council (RCC) standards and documents. RCC standards and documents are national and service range agreements on certain performance levels and characteristics that apply to equipment used on the national and service ranges. Standards are formulated to make efficient use of the spectrum and provide for spectrum compatibility.

b. Waivers. A request for waiver of military standards requirements that pertain to spectrum supportability for spectrum-dependent equipment must be made through USACESO to the Army Spectrum Manager. Requests must include technical justification as to why standards should be waived and the economic impact if the request is not approved. The USACESO will assess the waiver justification and provide the Army Spectrum Manager with a written recommendation for approval or disapproval.

### 4–8. Duties

*a.* The Army Spectrum Manager will approve or disapprove Army waivers to provisions of the Frequency Allocation-to-Equipment Process.

b. The USACESO will—

(1) Coordinate the processing of DD Form 1494 and associated documents with the Army Spectrum Manager (SAIS-PAS-M), the JSC, the USMCEB-FP, and the SPS.

(2) Return the following completed documents through command channels to the originator:

(a) The DD Form 1494.

(b) The USMCEB approval or disapproval.

(c) As appropriate, the NTIA approval or disapproval.

(d) An Army Spectrum Manager approved LOT with any further

guidance or restrictions. The LOT will include recommended actions. The originator will notify USACESO if there is a change in the proposed schedule.

(3) Provide the MACOM spectrum coordinators and director for spectrum requirements, PEOs, and other DoD/Army organizations and activities with a semiannual summary of the progress, problems, and trends of all DD Forms 1494 being processed. The semiannual summary will be provided not later than the last working day of March and September.

(4) Forward to appropriate Army commands or agencies those non-Army DD Forms 1494 which may have an impact on their operation, and collect impact analyses.

c. The MACOMs through USACESO will:

(1) Provide the MACOM coordinators and directors of spectrum requirements who will assist the PEOs/PMs, separate PMs, and other materiel development organizations and agencies in the processing of DD Forms 1494.

(2) Assist the Army Spectrum Manager and the USACESO in identifying spectrum management requirements in ODPs.

(3) Coordinate, within the materiel development community, the initiation of spectrum supportability assessments before Frequency Allocation-to-Equipment Process actions (DD Form 1494) are submitted.

### Chapter 5 Spectrum Requests and Assignments

### 5–1. Introduction

*a.* The Army Spectrum Manager also serves as the director of the USACESO. The Army Spectrum Manager has staff responsibility for development of Army concepts, plans, and policies for the employment of the electromagnetic spectrum and the allocation of frequencies.

*b*. The USACESO is responsible for managing spectrum and call sign assignments for the Department of the Army (DA) within the US&P, developing concepts, plans, and policies and, when tasked by the director, initiating actions for spectrum management to be implemented by the Army worldwide.

*c.* Spectrum requirements of Army components within unified commands will be satisfied by procedures established by the unified command. Call sign requirements for Army components will be satisfied using the provisions in this regulation and in allied communications publications.

d. The policy contained in this chapter is provided to:

(1) Accomplish effective use of the limited electromagnetic spectrum available for support of the Army commands, agencies, and activities within the US&P and Army components of unified commands.

(2) Promote rapid and direct action in responding to requirements submitted for spectrum and associated call sign assignments.

#### 5–2. Spectrum Coordination Channels

CONUS and OCONUS spectrum management coordination channels are shown in Figure 2-1, Chapter 2.

*a. CONUS.* Commanders of CONUS posts, camps, stations, and activities will submit spectrum actions to the supporting AFC (Appendix B). Exceptions are applicants in Alaska and Hawaii, who will submit their applications as outlined in Chapter 2, paragraph 2-8*e*.

b. OCONUS. Overseas applicants will submit requirements according to directives of unified commands and/or host nation agreements. The CINC of each unified command will be responsible for all military use of frequencies within the CINCs geographical area.

c. Army contractors. Contractors will submit spectrum requirements in direct support of Army contracts through the appropriate PM Office responsible for administration of the contract. A contractor will request frequencies for a Joint-service contract from the military department that is the executive service for that contract. Frequency support for contractor communications needs, not required to meet specifications of the contract, must be obtained by the contractor through FCC channels.

### 5-3. Types of Spectrum Actions

Applicants will submit spectrum requests using the Standard Frequency Action Format (SFAF) unless directed to use another format by a unified command or host nation agreement. ACP 190, US Supplement-1(C) Appendix D, describes types of spectrum actions and coordination channels, and provides guidance and procedures for processing assignment requests.

*a. New assignments.* Request for a new spectrum assignment must contain the information required in the SFAF and any additional information necessary to provide a clear and accurate description of the requirement. Organizations and activities requesting temporary and/or exercise assignments may submit requests by message to the supporting AFC in abbreviated SFAF format.

*b. Lead-time for request.* If a request for assignments does not allow for the prescribed lead-times listed below, the applicant must provide an impact statement justifying the urgency.

(1) Normally a minimum of 120 days lead-time is required to process an assignment spectrum request.

(2) For requests that require coordination with the FCC, the FAA, or host nation, the lead-time is 180 days.

(3) SFAF message requests in support of temporary requirements and exercises require 60 days lead-time.

c. Assignment deletions. Using activities will submit requests for deletions of authorized frequencies by letter, message, e-mail, or by the bulletin board system using the SFAF, through spectrum management channels. Using activities will not submit deletion requests when the requirement for the spectrum ceases within 120 days of the expiration date of the authorization. Navigational aids (NAVAIDS) identifiers will be deleted when no longer required.

*d. Assignment modifications.* An applicant may request modification of any item in an assignment except for the frequency, serial number, and transmitter state/country fields. Changes for the actions listed in paragraphs 5-3a, 5-3b, and 5-3c require an application for a new assignment be submitted according to 5-3a, above. A request for modification must arrive at the supporting AFC not later than 120 days before the current expiration date or new required date.

e. Renewals of frequency assignments with expiration dates. If the user desires to renew a frequency with an expiration date, the user must submit a request, using the SFAF, to reach the supporting AFC no later than 120 days prior to the expiration date. When submitting a request for frequency renewal, the user will update the operational, technical, and geographical data which indicates how the frequency assignment is being used.

*f. Five-year review program.* Commanders of Army posts, camps, stations, and activities will establish a program of continuing review of frequency assignments and will delete or amend such assignments as appropriate. The supporting AFC will provide users a listing of frequency records requiring a five-year review/update.

### 5-4. Frequencies that do not require specific authorization

The following types of frequency usage require no specific authorization within CONUS:

- a. International distress and emergency.
- b. Miscellaneous radiating devices to include:
- (1) Restricted radiation devices.
- (2) ISM equipment.
- (3) Radio receivers.
- (4) Cordless telephones.
- (5) Electronic fuzes.

#### 5-5. Army Policy for Spectrum Assignments

General policy for spectrum assignments to be used within the Army are:

*a. Air traffic control.* Air Traffic Control frequencies will be used to control the movement of aircraft taxiing, departing and approaching Army airfields, and en route in controlled air space.

b. Amateur frequencies. The Army will not use frequencies designated for amateur radio users within the US&P during normal peacetime conditions, except as authorized by the NTIA or FCC. Frequencies and emissions shown in Figure 5-1 are for use in emergency areas, when required to make initial contact with Radio Amateur Civil Emergency Services (RACES) units. Activities may also use these frequencies for communications with RACES stations on matters requiring coordination.

FREQUENCY	(REFERENCE FREQUENCY)	EMISSION
3997 kHz		6K00A3E
3998.5 kHz	(3997 kHz)	3K00H3E
53.3 MHZ		40K00F3E
3997 kHz 3998.5 kHz 53.3 MHZ	(3997 kHz)	6K00A3E 3K00H3E 40K00F3E

Figure 5-1. RACES frequencies

*c. Land Mobile Radio (LMR).* Due to congestion in the 148-150.8 MHZ and 162-174 MHZ bands, units requesting LMR frequencies will usually receive assignments in either the 138-144 MHZ or 406-420 MHZ bands unless there are operational needs which require the use of another band. Before deploying equipment overseas, users must assure that LMR frequency authorizations are available. The following policy also applies:

(1) Coordination must be made with supporting AFCs to determine if procurement of LMRs is frequency supportable.

(2) Currently, all LMR frequency assignments in the 148-150.8 MHZ band are governed by a USMCEB channeling plan, based on 25-kHz channelization. Command spectrum managers will replace LMR frequency assignments not divisible by 25-kHz (e.g., 148.065, 150.195, 150.195) with valid on-channel assignments at the earliest practical date. When command spectrum managers determine LMR frequencies are not in conformance with the USMCEB plan, the following special provisions will apply:

(*a*) When an Army unit is planning to replace off-channel radio equipment, the spectrum manager will determine if an on-channel frequency assignment can be obtained prior to ordering the new equipment.

(b) When an off-channel LMR is receiving interference from an on-channel system, and a frequency change is the most economical way to solve the problem, the off-channel net will change.

(c) If all the radio equipment on an off-channel net is turned in, the frequency assignment will be deleted immediately. Although the radios may be issued later to another unit or transferred to a new location, the off-channel frequency assignment will not be "held in reserve" for the new unit, nor will it be reassigned to a new location.

*d. Citizen band (CB) radios.* The following policy applies to Army users of CB radios:

(1) Regulations governing the use of CB radios are found at 47

Code of Federal Regulations Part 95, Subpart D. Army radio stations may use frequencies in the 26.97-27.41 MHZ band provided:

(a) Justification indicates such an assignment is necessary for inter-communications with non-Government stations.

(b) Approval for use on post, camp, or stations is subject to local requirements and restrictions of the installation commander.

(2) Possession of CB radios OCONUS is subject to national and international regulations. CB radios are not authorized for use OCONUS without host nation approval. Unified command directives apply. Coordinate CB radio use with the appropriate Army Component Spectrum Management Office. For more information on CB radios, see AR 25-1.

*e. Special considerations for CONUS HF requests.* The use of HF for domestic, point-to-point service within CONUS is limited to the following conditions:

(1) When providing instantaneous transmission of emergency, command control and alerting traffic of such importance as to affect the immediate defense and survival of the nation. In such cases, the following policy applies:

(a) Circuits will be in an operational status at all times and there will be on-the-air tests to assure readiness.

(b) Frequency assignment for such circuits will be protected commensurate to the importance of the communications requirement.

(2) When required in an emergency where life, public safety, or important property is in danger and other communications means are nonexistent, temporarily disrupted, or inadequate. Spectrum managers will ensure that command user training and equipment tests on frequencies in this category are conducted.

(3) When there is a need to provide a communications system manned by qualified operators who are military reservists, military affiliate radio station (MARS) affiliates, or personnel knowledgeable in tactical or training systems. These frequencies will not be used for traffic that can be routinely handled by other means.

(4) When other telecommunications facilities, such as the DCS and MARS, do not exist, are not practical for installation, and the use of frequencies above 30 MHZ is not practical.

*f. Maritime mobile (MM) frequencies.* The 156.2475-157.45 MHZ band is allocated for MM communications. In addition to the Government allocated sub-band, several channels are also available for government use as outlined in the NTIA Manual.

g. Specialized mobile radio (SMR) service. Government agencies, including the Army, are authorized to use the SMR service in the 806-824 MHZ, 851-869 MHZ, 896-901 MHZ and 935-940 MHZ land mobile bands under the following conditions:

(1) The Army will not establish an SMR system or provide an SMR service in the bands listed above. Army elements shall operate only as an "end user" with an FCC-licensed private carrier on a contractual basis. Since the SMR service is not considered to be in the common carrier service, spectrum authorization to federal agencies, including the Army, will be contingent upon the continuation of the negotiated contract with the private carrier.

(2) SMR systems are established by private commercial carriers and licensed through the FCC. After negotiating a contract to satisfy an Army requirement with a private carrier, Army elements will obtain spectrum authorization through NTIA to operate in the band corresponding to that in which the private carrier has been licensed, in the geographic area, by the FCC when becoming an "end user" in the SMR service. Federal agencies, including the Army, will not request SMR frequencies from the FCC.

(3) Army users submitting applications to NTIA to obtain frequency spectrum authority for SMR services will include the system name and the private carrier's name in SFAF Item 705, and the exact number of mobile receivers in SFAF Item 341.

h. Trunked land mobile radio (LMR) system.

(1) A trunked LMR system is a spectrum efficient method to meet non-tactical LMR operational requirements. Army installations may be the lead agency for such systems, and Army units may share use of an existing or planned system sponsored by another military service or federal agency. AR 25-1 and AR 25-3 provide IM and acquisition policy and procedures for these systems.

(2) Policy:

(*a*) Army units and agencies managing trunked LMR systems shall allow access by other federal agencies to the trunked system where it is operationally and technically feasible.

(b) Army and NTIA validated LMR system approval is required prior to purchase.

(3) Procedures: If the Army is the lead agency for a trunked LMR system, the Army installation or activity commander will comply with procedures outlined in the NTIA Manual. The procedures are essentially a two-step process: 1) Obtain Federal system approval from the SPS; and 2) obtain frequency assignments for the trunked system. NTIA will not issue frequency assignments for a trunked LMR system until the SPS has written system review approval.

(4) Army units and agencies are required to submit usage reports to USACESO (SFIS-FAC-P, 200 Stovall St., Room 9S65, Alexandria, VA 22332-2200). NTIA requires information for the first five years of operation of a trunked LMR system. Information to be reported is outlined in the NTIA Manual, and the reports must be submitted annually. The reported information is required to justify and defend trunked LMR systems nationally and as the basis for justification for additional channels.

i. Satellite communications. The following procedures apply:

(1) Defense Satellite Communications System (DSCS). Procedures for coordinating use of the DSCS and assigning of frequencies in the 7.25-7.75 GHZ and 7.75-8.4 GHZ bands, see Field Manual (FM) 24-11.

(2) Ground Mobile Forces (GMF). Procedures for coordinating use of GMF satellite access and assigning frequencies in the 225-400 MHZ band, see Field Manual (FM) 24-11.

(3) Commercial Satellite Communications (SATCOM). Procedures for coordinating and assigning frequencies for commercial SATCOM are contained in the NTIA Manual.

### 5-6. Spectrum Coordination Procedures

Specific coordination requirements and procedures are contained in ACP 190, US Supplement - 1(C), Appendix D.

### 5-7. Spectrum Requests in Non-Government Bands

The Army is authorized by Chapter 7 of the NTIA Manual to use frequencies in certain non-Government bands to meet peacetime tactical and training requirements, as well as certain other bands for test range requirements. Frequencies will be assigned by the supporting AFCs only when spectrum requirements cannot be satisfied in Government bands and when operation will not cause interference to non-Government service. The Army will have to accept interference caused by authorized non-Government users. Military use of a particular frequency in these bands will not preclude new non-Government assignments on that frequency. Specific policy concerning assignments in the above bands is:

a. Government users in non-Government bands. Government users may obtain agreement from the FCC to use frequencies allotted to non-Government operations. Such cases must meet the following minimum criteria before submitting the request through spectrum coordination channels:

(1) The assignment must be essential for communications with non-Government activities and cannot be met through use of regularly designated Government bands. For example an Army installation Provost Marshal wishes to operate in a local county or municipal police net, Army Medical Evacuation (MEDEVAC) helicopters wish to operate in a state-wide medical evacuation net, or an installation fire department wishes to operate in a local county or city fire department net.

(2) The FCC licensee and the requesting agency have concluded a mutually approved arrangement, and the licensee has provided written authorization for the Army unit to operate on the particular frequency. The requestor will forward a copy of this authorization to the supporting AFC.

(3) The intended operation will not prohibit expansion of the

non-Government services for which the frequencies are allotted and will be:

(a) Conducted in the geographical area of the licensee.

(b) Restricted to the purpose for which the particular frequency is authorized to the non-Government stations.

(c) Operated per FCC rules and regulations.

(d) Terminated if it causes harmful interference to the non-Government stations.

b. Operation and registration of FCC-licensed stations on Army installations. CB, amateur, taxi companies, and other radio stations that are FCC-licensed may transmit on Army installations, but will be subject to limitations imposed by the installation commander. Limitations, if any, will be an installation regulation. The regulation must not impose limitations so severe that they unnecessarily infringe on the rights of the individual to operate a radio according to FCC rules and regulations. Users must coordinate with the installation DOIM or spectrum manager prior to operation of such equipment on the installation.

### 5-8. International Registration

The following policy applies to the registration of Army frequencies used in foreign countries:

*a.* Frequencies used by Army activities on foreign soil can be registered with the International Frequency Registration Board (IFRB) by the Army per USMCEB-M88-83 or by the foreign country in its own name.

*b.* If the foreign country insists on registering frequencies used by the US military forces in its territory, the agreement will at least provide that:

(1) The registration will not affect existing US registrations.

(2) The country will cancel, upon request by the US authorities, any registrations made to support US operations.

*c*. Arrangements with foreign countries for international registration of frequencies are of joint interest to the military Services. Therefore, the Services will coordinate with the joint FP, USMCEB before concluding any arrangements.

### 5–9. Electronic Attack (EA) Training

*a.* Approved DD Form 1494 and frequency assignments are not approval documents to conduct EA operations (e.g., jamming, chaff drops).

b. EA frequency clearances are granted as follows:

(1) CONUS: Under provisions of AR 105-86, AFC AZ, AFC WSMR, and AFMO-CONUS may provide local EA clearances limited to certain frequency bands under specified conditions. EA clearances requiring national level coordination with the other Military Services and Government agencies are issued by USACESO (SFIS-FAC-S, 200 Stovall St., Room 9S65, Alexandria, VA 22332-2200).

(2) OCONUS: Overseas applicants will submit EA requirements according to directives of unified commands and/or host national agreements. The CINC of each unified command will be responsible for all military use of frequencies including EA within the geographical area.

(3) Users of the equipment must comply with local EA frequency clearances, host nation agreements, unified command directives, USMCEB policy and procedures, and DD Form 1494 with Army Spectrum Manager's LOT.

### Chapter 6 Radio Station Identification

### 6-1. General

The three main categories of radio station identification used by the US Army are international call signs, positive voice identification (PVI), and tactical call signs. In Army usage, a call sign is a combination of letters and digits. Call signs are used to establish and maintain communications. Call signs are to identify the radio stations of command authorities, activities, facilities, units, elements, or

individual positions. Call Signs Are Not Used to Identify People. PVI is a less formal means of identification used instead of international call signs for many non-tactical purposes. Call words, e.g., RED DOG 6 or SPEEDY TULIP 3, are authorized for communications ONLY when operating in a secure mode. This applies to aeronautical, maritime, and ground stations alike.

### 6-2. Responsibility for Radio Station Identification

The Military Communications Procedures and Publications Panel of the USMCEB is responsible for radio station identification within DOD. As a principal member of the USMCEB, HQDA DISC4 (SAIS-PAS-M) ensures decisions of the USMCEB are carried out within the Army. USACESO is responsible for nontactical call sign management within the Army (Mailing address: USACESO, ATTN: SFIS-FAC-M, 200 Stovall St., Room 9S65, Alexandria, VA 22332-2200).

### 6-3. International Call Signs

International call signs are governed by the rules of the International Telecommunication Union (ITU). These call signs use specific characters in the first, or first two positions, to identify the nationality of the station. These characters are listed in the Table of Allocations of International Call Sign Series contained in the ITU radio regulations. The following types of international call signs, together with specific suballocations are used within the US Army.

a. Fixed service radio stations.

b. Land mobile service.

(1) Base radio stations. (In Army use, the basic call sign normally includes the associated land mobile stations.)

(2) Land mobile radio stations (independent land mobile stations only).

c. Aeronautical mobile service.

(1) Aeronautical radio stations.

(2) Aircraft radio stations.

d. Maritime mobile service.

(1) Coast radio stations.

(2) Ship radio stations.

(3) Ship radio teleprinters.

e. Space service earth stations.

f. Experimental radio stations.

### 6-4. International Call Sign Assignment Authority

Army authority for assigning international call signs is further delegated as follows:

*a.* Fixed service radio station call signs for the USACE. These call signs will be assigned and maintained by the US Army Communications-Electronics Services, 200 Stovall Street, Attn: SFIS-FAC-M, Alexandria, VA 22332-2200. The block of call signs WUA through WUZ has been suballocated for this purpose.

(1) A three letter call sign is assigned to all USACE activities.

(2) The district call sign may be suffixed by one digit.

*b.* Military Affiliate Radio System (MARS). The Chief, Army MARS will assign call signs for the Army MARS Program. MARS call signs are assigned per AR 25-6 and the call sign patterns are assigned per FM 11-490-7. Correspondence will be addressed to Commander, US Army Information Systems Command, Attn: Chief Army MARS, Fort Huachuca, AZ 85613.

c. Shipboard Radio Stations.

(1) A ship radio authorization (SRA) is required for the operation of a non-tactical shipboard radio station used on any watercraft that is:

(a) Owned and operated by the Army.

(b) Operated by the Army with Army (civilian or military) radio operators, such as watercraft under bareboat, time voyage or other similar charter.

(2) SRAs are not issued for:

(a) Army watercraft operating solely in a tactical environment, and which do not have a requirement to employ non-military frequencies and call signs. Tactical frequencies and call signs will be assigned by appropriate service or unified command authorities.

(b) Shipboard radio stations whose radio operators are not Army

(civilian or military) personnel. An FCC license is required when a watercraft is chartered or otherwise used under Army direction, and the shipboard radio station is operated by personnel not directly responsible to Army authority.

(3) Correspondence for the assignment, renewal or cancellation

of an SRA will be forwarded through appropriate command channels to USACESO, ATTN: SFIS-FAC-M, 200 Stovall St., Room 9S65, Alexandria, VA 22332-2200.

(4) Requests for SRAs require the information contained in Table 6-1.

#### Table 6–1 Request for SRA

ITEM	DATA REQUIRED	EXPLANATION
1.	Official watercraft name or designation	
2.	Former name(s) or designation(s) of watercraft, if any	
3.	Army authority over watercraft	a. Owned b. Operated c. Leased d. Other
4.	Description of transmitting equipment (radar, communications, position finding)	<ul> <li>a. Equipment make or manufacturer</li> <li>b. Model number</li> <li>c. Frequency tuning range</li> <li>d. Specific frequencies to be used</li> <li>e. Frequency band required for radar</li> <li>f. Type(s) of emission required</li> <li>g. Power output</li> <li>h. TELEX ID</li> <li>i. Is DSC ID required?</li> <li>j. Will vessel sail in international waters?</li> </ul>
5.	Controlling authority	<ul><li>a. Army command exercising operational control</li><li>b. USACE districts</li><li>c. Point of contact and telephone number</li></ul>

(5) SRAs will remain in effect for a period of 3 years from the last day of the month in which issued. Renewal requests must be received by USACESO at least 30 days before the expiration date indicated on the SRA. SRAs not renewed within the above time periods will be cancelled, and may be issued to another station.

(6) Significant changes in data items in Table 6-1 will be reported to USACESO within 90 days.

(7) Users will submit to USACESO a request for cancellation of an SRA when radio equipment is permanently removed from the watercraft, or the watercraft is sunk, scrapped, transferred from Army control or disposed of by other means.

(8) Assignment authority for SRA call signs. All shipboard radio station call signs are composed of four letters and are assigned by the USACESO. Call signs will only be assigned in conjunction with an SRA and are allocated as follows:

<ul> <li>AAAA-ADZZ</li> </ul>	US Army owned/or operated watercraft,
	less Corps of Engineers
<ul> <li>AEAA-AEZZ</li> </ul>	Corps of Engineers floating vessels
• ALAA-ALZZ	Spares for future assignment

(9) Assignment authority for Shipboard Radio Teleprinter (TELEX) identifiers. These identifiers will be assigned in conjunction with an SRA. TELEX have a five digit numeric identifier, in addition to the ship radio call sign, and are required for the operation of radio teleprinter equipment onboard Army watercraft.

(10) Digital Selective Calling (DSC) Identification Number. A nine digit numeric identifier, in addition to ship radio call sign, is required to operate digital maritime telecommunications systems. These identification numbers will be assigned by USACESO in conjunction with a SRA.

d. Nontactical earth stations in the space service. These stations will be issued international call signs. Call signs will consist of two letters followed by three digits and are allocated as shown:

• AA200-AA299 AFMO CONUS (for Eastern US)

• AA300-AA399 AFMO CONUS (for Central US) • AA400-AA499 AFMO CONUS (for Western US) • AA500-AA599 AFC-WSMR • AA600-AA699 AFC-AZ AFC-MDW • AA700-AA799 • AA800-AA899 **USACESO** • AB200-AB999 USARPAC (for Pacific) • AC200-AC999 USACESO • AD200-AD999 USACESO • AE200-AE9995th Signal Command (for Europe and the Middle East) • AL200-AL999 USARPAC (for Alaska)

*e.* Experimental call signs. Each call sign will consist of two letters followed by a number, the letter "X", and another letter. Assignment authorities for experimental call signs are AFMO CON-US, AFC-AZ, AFC-WSMR, and USACESO. These call signs will:

(1) Be assigned solely to identify emissions that are definite parts of an experimental operation.

(2) Not be used to identify communications support operations.(3) Be allocated as listed below:

• AA2XA-AE2XZ	AFMO CONUS (Eastern US)
• AL2XA-AL2XZ	AFMO CONUS (Eastern US)
• AA3XA-AE3XZ	AFMO CONUS (Central US)
<ul> <li>AL3XA-AL3XZ</li> </ul>	AFMO CONUS (Central US)
• AA4XA-AE4XZ	AFMO CONUS (Western US)
• AL4XA-AL4XZ	AFMO CONUS (Western US)
• AA5XA-AE5XZ	AFC-WSMR
• AL5XA-AL5XZ	AFC-WSMR
• AA6XA-AE6XZ	AFC-AZ
<ul> <li>AL6XA-AL6XZ</li> </ul>	AFC-AZ
• AA7XA-AE7XZ	AFC-MDW
• AL7XA-AL7XZ	AFC-MDW
<ul> <li>AA8XA-AL8XZ</li> </ul>	Reserved for USACESO
---------------------------------	----------------------
<ul> <li>AL8XA-AL8XZ</li> </ul>	Reserved for USACESO
<ul> <li>AA9XZ-AE9XZ</li> </ul>	Reserved for USACESO
<ul> <li>AL9XA-AL9XZ</li> </ul>	Reserved for USACESO

*f*. Land mobile stations in the land mobile service. Generally, land mobile stations derive their call signs as subunits of their base station's call sign.

(1) The individual land mobile stations operating within those nets will share the base station call sign by adding digits or letters for unit identification as determined by the net control (for example, STATION CHARLIE; UNIT ONE-SIX; or MOBILE SEVEN).

(2) However, if a land mobile station is set up independent of a base station, the mobile station must be assigned its own international land mobile call sign.

g. Land mobile call signs are assigned to independent land mobile stations. They may be assigned in instances of a mobile operation in a nontactical environment and may transit over two or more Army areas or have a broad range. Generally, positive voice identification is used. These land mobile call signs consist of two letters followed by four digits. Call signs are allocated as indicated below:

• AA2000-AA2999	AFMO CONUS
• AA3000-AA3999	AFMO CONUS
• AA4000-AA4999	AFMO CONUS
• AA5000-AA5999	AFC-WSMR
• AA6000-AA6999	AFC-AZ
• AA7000-AA7999	AFMO CONUS
• AA8000-AA8999	USACESO
• AA9000-AA9999	USACESO

*h.* Army aircraft in nontactical aeronautical mobile service. Normally, these aircraft operate using voice transmissions (telephony) and will use FAA identification procedures. If Army nontactical aircraft must transmit in other modes (for example, teletypewriter or continuous wave), an international call sign must be obtained. Aircraft call signs consist of five letters and are assigned from the block ADAAA through ADZZZ. Call signs are assigned by USACESO.

*i*. Army international call sign prefixes for all other fixed service and land radio stations (base, aeronautical, and coast). USACESO has suballocated these prefixes to the AFC or spectrum managers for specific geographical areas. In the past, fixed and land radio station call signs were assigned from the Army's resources. More recently, within the US&P, the provisions of PVI (para 6-7) have been substituted wherever practical. However, configured call signs will be used where the geographical location of the station, the authorized power and/or the propagation characteristics are such as to create instances of possible harmful frequency interference beyond US&P boundaries. The suballocated prefixes for specific geographical areas are indicated below:

- AAB AFMO CONUS (Eastern US)
  AAC AFMO CONUS (Eastern US)
  AAD AFMO CONUS (Eastern US)
  AAE AFMO CONUS (Central US)
- AAF AFMO CONUS (Central US)
- AAG AFMO CONUS (Western US less Arizona and WSMR)
- AAH AFMO CONUS (Western US less Arizona and WSMR)
- AAK USACESO

*Note.* For transportable communications vans-worldwide. When used, to be suffixed with the last three digits of the authorized serial number of the van. Note: This call sign to be used only when operating independently. When replacing or augmenting an existing Defense Communications System station, the call sign of that station will be used.

• AAW AFC-WSMR

- AAZ AFC-AZ (State of Arizona)
- ABA-ABZ USARPAC (Hawaii and East PAC)
- ACA-ACZ AFMO CONUS (Caribbean)
- ADA USARPAC (Far East)
- ADB USARJ (Okinawa Prefecture)
- ADC USARPAC (Far East)
- ADD USARJ (Japan)
- ADE-ADZ USARPAC (Far East)
- AEA-AEZ 5th Signal Command (Europe and Middle East)
- WAR AFC MDW (MDW)

(1) Call signs that consist of three letters will be assigned to fixed stations serving as theater or area administrative net control stations.

(2) Call signs that consist of three letters suffixed with one digit will be assigned to coast stations.

(3) Call signs that consist of three letters suffixed with two digits will be assigned to all other fixed stations and to aeronautical stations.

(4) Call signs that consist of three letters suffixed with three digits will be assigned to base stations in the land mobile service (for example, guard, fire, and vehicle dispatching nets). Digits 0 and 1 will not be used after the third letter of the prefix.

*j.* Repeater and relay stations. International call signs will not be assigned to a station mainly to identify repeater or relay functions. Emissions from repeater and relay stations are normally identified by the call sign of the distant station keying the equipment. For local testing of such equipment, a form of PVI may be used (for example, FORT BLISS FIVE NET REPEATER - TESTING).

#### 6-5. Publications

Army assigned international call signs are published in the documents listed below. The activity assigning the call signs ensures that pertinent information on all new assignments, deletions, or changes is furnished, in decode format, to the preparing agency as soon as possible for inclusion in the proper publications.

a. ACP 100, US SUPP-1.

- · Fixed service stations
- Land mobile service stations
- · Aeronautical mobile service stations
- Coast stations
- · Experimental stations
- Space service stations
  - b. ACP 113. Ship stations

#### 6-6. Security

International call signs assigned to stations operating in permanent places must be unclassified. While it may be necessary to classify information on a communications facility or specific nontactical operation during planning and development, once the station becomes operational, the call sign and location can no longer be protected. If the specific mission or function of the station is not given during transmissions, that type of information might continue to be classified. However, unclassified listing of the call sign, geographical location, and description must be published in appropriate international listings. These listings will help to identify harmful interference. Guidance on the security of call signs is in ACP 121, Chapter 5, and ACP 122, Chapter 4.

## 6-7. Positive Voice Identification (PVI)

*a.* Within the US&P the use of PVI procedures in place of international call signs is encouraged. PVI may be used for most nontactical radio nets using LOS type transmissions. However, PVI will not be used when any combination of the geographical location of the station, authorized output power, and normal radio propagation characteristics could cause harmful frequency interference across the boundaries of any foreign country. In this case, international call signs must be used. Those very high frequency, ultrahigh frequency, and super high frequency stations that transmit within 200 kilometers of the boundary of a foreign country and all very low frequency, low frequency, medium frequency, and high frequency radio stations, regardless of location, are potential sources of interference.

b. PVI will consist of the word "ARMY," the geographical location of the station or operation, and the type of operation. Examples are: ARMY FORT KNOX POST TAXI DISPATCHER and ARMY FORT BELVOIR MAP SURVEY BASE.

c. The associated mobile stations will be identified by suffixing numbers or letters. Examples are: ARMY FORT KNOX POST TAXI UNIT TWO-ONE and ARMY FORT BELVOIR MAP SUR-VEY TEAM THREE. During reliable communications, when it appears that there is no conflict with other operations, PVI may be abbreviated to aid operations. However, at the beginning and ending of operations, and hourly during operations, the base stations should use the full PVI at least once.

*d.* When using telephony, Army nontactical aeronautical stations will normally use PVI. PVI will consist of the names of the aeronautical station, followed by the function. Examples are: DAVISON TOWER and GOODMAN GCA.

#### 6-8. Tactical Call Signs

*a.* General. Tactical call signs consist of various combinations of letters and digits. The use of character sequences configured in accordance with ITU Radio Regulations, article 19, should be avoided if possible. Tactical call sign systems are developed to meet specific military requirements under an exemption to the ITU Radio Regulations. See AR 25-1 for information on specific tactical call sign systems and their use.

*b*. Composition. Tactical call signs assigned to US Army units for use in intra-Army or joint operations, when the joint commander concurs, consist of basic letter-number-letter combinations to indicate the command. The numbers "01 through 99" are suffixed to indicate the specific subordinate command, title or position, or mission or functions of the using element. The two-digit suffix may be expanded by the letters "A through Z," when required, to further identify associated subelements.

#### 6-9. Special Call Signs/Call Words for Army Aircraft

Non-tactical Army aircraft may be granted the use of call words. Only those call words authorized by the Federal Aviation Administration (FAA), and listed in the Joint Communications Publications (JANAP 119) will be used. Requests for call words must be forwarded through appropriate command channels to USACESO, ATTN: SFIS-FAC-M, 200 Stovall Street, Room 9S65, Alexandria, VA 22332-2200.

#### 6-10. Amateur Radio Call Signs

Amateur radio station call signs are not assigned by the Army. A license that includes a proper call sign for the desired type of operations must be obtained from the FCC or host foreign government (if applicable). Examinations are required for an amateur radio license. The license conveys major operating privileges. AR 105-70 encourages and supports amateur radio activities. Before operating an amateur radio station on an Army post, camp, or station, coordinate with the local C-E officer, DOIM, or spectrum manager. See AR 105-70 for additional information.

#### Section I Required Publications

# AR 10-87

Major Army Commandas in CONUS (Cited in para 2-23g, 2-23h, and 3-4a.)

# AR 25–1

The Army Information Resources Management Program (Cited in para 1-1e, 1-7, 2-7z, 2-171, 2-23a, 5-5d, 5-5h, and 6-8a.)

# AR 25–3

Army Life Cycle Management of Information Systems (Cited in para 1-8c, 2-8u, 2-13f, 2-15p, 2-23a, and 5-5h.)

# AR 70–1

Army Acquisition Policy (Cited in para 1-8c, 2-6g, 2-7n, 2-19h, and 4-2f.)

# DFARS 235.071

Defense Acquisition Regulations (Cited in para 4-2d.)

# DoDD 3222.3

Department of Defense Electromagnetic Compatibility Program (Cited in para 2-6f and 3-6a.)

# DoDD 4650.1

Management and Use of the Radio Frequency Spectrum (Cited in summary, para 1-8b, and 2-7h).

# **DoDD 5000.1**

Defense Acquisition.

# DoDR 5000.2-R

Mandatory Procedures for Major Defense Acquisition Programs (MDAPs) and Major Automated Information Systems (MAIS) Acquisition Programs (Cited in para 1-8c, 3-6a, 4-2a, and 4-2f.)

# DoDD 5200.1R

Information Security Program Regulation.

# NTIA Manual

Manaual of Regulations and Procedures for Federal Radio Frequency Managment (Cited in para 4-6c, 4-6d, 4-7a, 5-5f, 5-5h(3), 5-5i(3), and 5-7.) This document is available from the Superintendent of Documents, P.O. Box 371954, Pittsburgh, PA 15250-7954, Stock Number 903-008-00000-8, Telephone: (202) 512-1800.

# Section II Related Publications

# OMB Circular A-11

Office of Management and Budget Circular A-11, Preparation and Submission of Budget Estimates.

# Administrative Procedures Manual

Administrative Procedures Manual, US Military Communications-Electronics Board (USMCEB) This document is available from the USMCEB, Room 1E833, Pentagon, Washington, DC 20310.

# AAE 91-3

Army Acquisition Executive (AAE) Policy Memorandum 91-3. (Army Electromagnetic Environmental Effects  $(E^3)$  Program Implementation).

# AR 1–1

Planning, Programming, Budgeting and Execution System.

#### AR 5–3 Installation Management and Organization.

# AR 11–2

Management Control.

# AR 11–7

Internal Review and Audit Compliance (IRAC) Program.

# AR 25–6

Military Affiliate Radio System (MARS).

# AR 40–46

Control of Health Hazards from Lasers and Other High Intensity Optical Sources.

# AR 71–9

Materiel Objectives and Requirements.

# AR 95–2

Air Traffic Control, Airspace, Airfields, Flight Activities, and Navigational Aids.

# AR 105–70

Amateur Radio Operations.

# DA Pamphlet 25–1

Army Information Architecture.

# DA Pamphlet 25–2

Information Mission Area Planning Process.

# TB Med 523

Control of Hazards to Health from Microwave and Radio Frequency Radiation and Ultrasound.

# DoD ECAC Frequency Resource Record System Handbook

This document is available from the JSC, 120 Worthington Basin, Annapolis, MD 21402-5064.

# **ITU Radio Regulations**

This document is available from the ITU, Geneva, Switzerland.

# MIL-STD-188-100

Common Long Haul and Tactical Communications System Technical Standards.

# MIL-STD-461.D

Requirements for the Control of Electromagnetic Interference Emissions and Susceptibility.

# MIL-STD-469A

Radar Engineering Design Requirements for Electromagnetic Compatibility.

# FM 24-11

Tactical Satellite Communications.

# JSC-CR-90-079

DD Form 1494 Preparation Guide for Army Frequency Allocations. This preparation guide is available electronically on the Army Spectrum Management Home Page at—

http://www.spectrum.gen.va.us/ceso/engr/spec\_engineer.htm

#### Section III Prescribed Forms

**DD Form 1494** Application for Equipment Frequency Allocation.

#### Section IV Referenced Forms

DA Form 2028

Recommended Changes to Publications and Blank Forms.

# Appendix B Army Frequency Coordinators

# B-1. Areas of Responsibility for AFCs

Areas of responsibility for Army Frequency Coordinators/Management Offices are shown in Figure B-1.



Note: Alaska and Hawaii: USARPAC

Figure B-1. Areas of responsibility for Army Frequency Coordinators

#### **B-2. Frequency Coordinators Addresses**

The Frequency Coordinators addresses and areas of responsibility are listed below.

a. Army Frequency Management Office, Continental United States

Fort Sam Houston, TX 78234-5032, Telephone: (210) 221-2820 DSN: 471-2820/FAX:-2844.

Alabama, Arkansas, California (less units located on or conducting training of the Army National Training Center, Fort Irwin, CA), Colorado, (less the area west of 108EW), Connecticut, Delaware, Florida, Georgia, Idaho, Illinois, Indiana, Iowa, Kansas, Kentucky, Louisiana, Maine, Maryland (less certain areas which are included in the Military District of Washington), Massachusetts, Michigan, Minnesota, Mississippi, Missouri, Montana, Nebraska, Nevada, New Hampshire, New Jersey, New York, North Carolina, North Dakota, Ohio, Oklahoma, Oregon, Pennsylvania, Rhode Island, South Carolina, South Dakota, Tennessee, Texas (less Fort Bliss and US Territory within 240-kilometer radius of White Sands Missile Range), Utah (less the area east of 111EW), Vermont, Virginia (less certain areas which are included in the Military District of Washington), Washington, West Virginia, Wisconsin, Wyoming, Puerto Rico, and the Virgin Islands.

*b.* Area Frequency Coordinator, State of Arizona, Fort Huachuca, AZ 85613-5000, Telephone:(520) 538-6423 DSN:879-6423/FAX:-8525

Arizona

c. Area Frequency Coordinator, White Sands Missile Range, New Mexico 88002-5526, Telephone: (505) 678-5417 DSN: 258-3702/FAX:-5281

New Mexico, US territory within a 240-kilometer radius of (*hite sands missile range (including fort bliss, tx*) plus the area of

Utah and Colorado that lies within 108N and 111N West.

d. Deputy Chief of Staff for Information Management, Military District of Washington, ATTN: ANMY-IMO-O, Fort Lesley J. McNair

Washington, DC 20319-5050 Telephone: (202) 475-2799 DSN: 335-2799/FAX:-2767

Army installations and activities within the District of Columbia;

Arlington and Fairfax Counties in Virginia; Montgomery and Prince George's counties in Maryland; Fort Ritchie, MD; Fort Holabird, MD; Fort Meade, MD; and Fort A.P. Hill, Fort Belvoir, and Vint Hill Farms, VA.

e. Commander, US Army, Pacific (USARPAC), ATTN: APIM-OEO Fort Shafter, HI 96858-5410, Telephone: (808) 477-1054

DSN: 477-1054/FAX:-0691 Alaska and Hawaii

# Appendix C Program Plan for Interference Resolution

#### C-1. Introduction.

*a.* Army operations must share spectrum resources with other users in both the Government and non-Government sectors, thereby increasing spectrum congestion and the potential for radio frequency interference both from and to Army spectrum-dependent materiel.

*b*. As portions of the spectrum are reallocated in the future, the potential for interference may increase. The Army and its officials could be found liable for interference with Civilian telecommunications in the vicinity of Army installations under the Federal Communications Act and implementing regulations. The possibility of interference must be monitored and prompt corrective action taken as needed.

c. Freedom from potential interference is a fundamental purpose of the Army E3 Program. The structure of the Army acquisition program areas was designed for a logical succession of efforts to minimize the potential for Army spectrum-dependent materiel to either cause, or be susceptible to, radio frequency interference, beginning with standards and specifications, through equipment design and development, and finally with operational spectrum support after equipment is fielded.

#### C-2. Purpose.

In recognition of the fact that interference will and does occur in operational environments, a program has been established to define policies and procedures for resolving interference when it occurs. This Appendix describes that program including technical considerations, organizational responsibilities, descriptions of technical assets, and interference resolution procedures which can help the Army overcome the effects of interference during Army operations.

#### C-3. Types of Interference.

*a.* EMI. EMI is any electromagnetic or electrostatic disturbance that disrupts, interrupts, or otherwise degrades or limits the effective performance of spectrum-dependent materiel. Such interference can be caused by either intentional or unintentional sources.

*b.* Harmful Interference. Harmful interference is interference which endangers the functioning of a radionavigation service or other safety services or seriously degrades, obstructs, or repeatedly interrupts a radiocommunication service.

#### C-4. Army Interference Resolution Authority.

The responsibility for interference reporting was assigned by the Deputy Chief of Staff, Operations (DCSOPS) to the Director of Information Systems for Command, Control, Communications, Computers (DISC4) on 1 September 1992.1

## C-5. Scope.

These provisions of the AIRP apply to all active Army, ARNGUS, and USAR organizations.

## C-6. Program Relationships.

*a.* EMCP. Provisions for resolving interference affecting Army operations is an integral part of the EMCP. This relationship is exemplified in Figure C-1 in which various parts of the program function together to facilitate Army operations.



b.  $E^3$  Program. The principal relationship between interference and the  $E^3$  program is information flow from the USACESO who provides interference reports to the  $E^3$  program coordinator at the AMC and to the EPG. The coordinator analyzes these reports to identify the need for changes to  $E^3$  standards and specifications to provide permanent solutions to recurring interference associated with technical characteristics of Army spectrum-dependent materiel or other materiel which affects the use of Army spectrum materiel for future procurements. The EPG adds the information to their equipment measurement database.

#### C-7. Interference Resolution Functional Responsibilities.

Army organizations and activities with responsibilities regarding interference resolution are identified in Figure C-2. Specific responsibilities are:

a. Headquarters, Department of the Army (HQDA). HQDA, DIS-C4, Spectrum Management Office (SAIS-PAS-M) is responsible for providing policy and program guidance for the AIRP.



Appropriate AFC. See Fig B-1 of Appendix B (Army Frequency Coordinators)

Figure C-2. Army interference resolution organization.

b. Army Spectrum Manager. The Army Spectrum Manager (SAIS-SM) directs the AIRP.

c. US Army Communications-Electronics Services Office (USACESO). USACESO, on behalf of the Army Spectrum Manager, will:

(1) Provide day-to-day management of the AIRP worldwide.

(2) Coordinate the AIRP with the JSC JSIR program and initiate Army requests for assistance.

(3) Coordinate interference incidents involving space operations with US Space Command (USSPACECOM).

(4) Serve as the Army focal point for interservice interference incidents and for incidents involving other Federal agencies or the civilian sector in the CONUS.

(5) Assist the JSC in resolving Army interference incidents when JSIR assistance has been requested or is being provided.

(6) Develop interference resolution training requirements and coordinate the requirements with the US Army TRADOC.

(7) Establish interference resolution requirements for CONUS installations and validate Army interference resolution requirements from TRADOC and other MACOMs.

(8) Manage, coordinate, and distribute automated interference resolution tools to meet the validated requirements.

(9) Coordinate with AMC concerning the exchange of data and other information with the Army E3 Program and Army E3 database.

(10) Prepare, publish, and distribute a list of current Army interference resolution points-of-contact (POCs) to all Army installations.

(11) Program funds to support USACESO interference resolution responsibilities and activities.

d. US Army Materiel Command (AMC). AMC will:

(1) Coordinate action within the materiel development and acquisition community regarding changes to E3 standards and specifications or other corrective action.

(2) Program and budget for AMC interference resolution responsibilities and activities.

(3) Ensure that appropriate data from interference reports and other sources is provided to EPG for incorporation into the E3 database.

e. US Army Training and Doctrine Command (TRADOC). TRADOC will:

(1) Coordinate with USACESO regarding interference resolution training requirements for Army personnel.

(2) Conduct interference resolution training of Army personnel through the TRADOC schools.

(3) Program and budget for interference resolution responsibilities and Army-wide interference resolution training.

f. US Army Signal Command (ASC). ASC will:

(1) Provide equipment and manpower for three (3) transportable DF and monitoring field teams.

(2) Deploy the field team(s) within 24 to 72 hours after an initial

request for interference resolution assistance to detect, monitor, locate the source of, and assist in resolving interference situations.

(3) Document the results of interference resolution assistance in an after action report. Provide a copy of the documentation to USACESO.

(4) Maintain the automated historical database of Army interference reports.

(5) Program and budget for ASC interference responsibilities and activities.

g. Commanders of Army Major Commands (MACOMs). Commanders of Army MACOMS will:

(1) Designate a MACOM Spectrum Manager as the POC for interference resolution.

(2) Coordinate interference resolution requests between the MACOM post, camp, and station DOIM, frequency managers, and ASC through the MACOM POC for interference resolution.

(3) Program and budget for interference resolution responsibilities and activities to include ASC assistance, JSIR assistance, and the acquisition of automated interference resolution hardware and software tools for use by the MACOM, DOIMs, and frequency managers.

(4) Provide guidance and direction to subordinate commands, installations, and activities.

h. Army Post, Camp, and Station Director of Information Management (DOIM). The DOIM will:

(1) Serve as the installation POC for coordinating and assisting in resolving interference incidents involving the installation in accordance with this regulation and AR 5-3.

(2) Report all incidents of interference occurring on the installation that impact on Army operations, and submit copies of the interference report to USACESO and ASC.

(3) Assist and coordinate with tenant Army units, other military services, Government agencies, and civilian activities within 80 kilometers of the installation in resolving incidents of harmful interference.

(4) Maintain records of all authorized frequency assignments within the geographical area of the installation and maintain records of all interference incidents for the area of responsibility.

(5) Process requests for new military frequency assignments or modification of existing military assignments required to resolve interference situations, when necessary.

(6) Resolve incidents of interference with local assets and information whenever possible.

(7) Program and budget for procurement of hardware and software tools for use in the installation interference resolution activities.

*i. All Army Units and Activities.* Army units and activities will: (1) Contact the installation DOIM or frequency manager immedi-

ately in all cases of actual or suspected interference.

(2) Provide the DOIM or frequency manager with essential information concerning incidents of interference (see Paragraphs C-11 and C-12).

(3) Operate all spectrum-dependent systems and equipment in strict accordance with radio frequency authorizations or frequency plans, particularly with respect to assigned frequency and emission/ bandwidth, transmitter power limitations, and antenna height and direction.

(4) Attempt to locate and identify interference sources with organic resources including expedient antennas for DF when possible.

(5) Provide access to unit facilities, to installation personnel, or to other authorized technicians who are assisting in resolving the interference incident.

#### C–8. General, Defining the Environments Where Harmful Interference Occurs, Resolution Assets, Reporting Procedures, and Coordination Procedures

a. General. The DOIM or frequency manager at each Army installation is the "first line of offense" and is responsible for directing and coordinating all actions to resolve interference among Army units, tenant military activities, and civilian spectrum users (including host government users involved.

(1) Interference resolution during tactical exercises and operations is very similar to the manner in which interference is resolved while in garrison.

(2) High densities of spectrum-dependent equipment, used to support both offensive and defensive operations, are found in a relatively small radio LOS geographical area. HF reuse ratios are common on battlefields and during field training exercise. These situations create a potential for frequency interference to occur. The most significant difference is that the local interference resolution coordinating authority and POC is the Division Frequency Manager, and the regional coordinating authority is the Corps Frequency Manager. Depending on the tactical organization, the Army Component Frequency Manager plays a role in coordinating interference resolution activities when interference extends beyond a corps boundary or within the theater of operations.

(3) Interference resolution is viewed from the victim's perspective. In almost all cases, resolving interference involves a set of activities that tend to be the same, regardless of the details of the interference incident. This is illustrated in Figure C-3. The impact of each interference incident is unique, and no standard procedure can be established that will guarantee resolution in every case. A reasoned, logical, step-by-step approach, however, will reduce time and cost in resolving interference situations.





(4) Resolving interference involves following a logical progression of induction and deduction. Figure C-4 is a logic diagram for instances when an Army activity is the victim in a sustaining base

environment. Figure C-5 is a logic diagram for instances when an Army unit is the victim in tactical operations or a training exercise environment.







Figure C-5. Interference resolution (tactical operations/training exercise environment (victim)).

(5) Figure C-6 is a logic diagram that shows the activities for interference resolution when an Army activity is the source of interference in Sustaining Base and tactical operations/training environments.

(6) Sometimes the operator of spectrum-dependent equipment will find that differentiating between interference and other cases of disruption or unacceptable degradation is difficult. Operators should try, through the process of elimination, to determine if the disruption or degradation is due to interference before notifying the DOIM or frequency manager of an interference situation.

(7) When harmful interference is known to be the cause of interference, the military activity responsible for operations will contact the installation DOIM or frequency manager immediately by the most expeditious means, usually by telephone. Army units and activities are responsible for using whatever organic resources are available, such as radio receivers or electronic test equipment to determine whether the cause of the disruption is interference in order to gain as much information about the nature of the interference as possible. Army unit and activity personnel will be prepared, during this initial contact with the DOIM or frequency manager, to provide as much information about the known or suspected interference as possible. As a minimum, the Army activity should furnish the essential information described in Paragraphs C-11 and C-12.



Figure C-6. Interference resolution (sustaining base and tactical operations/training environment (source)).

*b.* Defining the Environments Where Harmful Interference Occurs. For administrative and coordination purposes, environments where interference to Army operations occur have been subdivided into four categories: local, regional, national, and space. The subdivision designations are a function of the area of extent of Army operations involved in an interference situation. The definitions of the four environments are:

(1) Local interference. Local interference is defined as an interference situation within the boundaries of an Army post, camp, or station; or within 80 radio LOS kilometers from the edge of the installation boundary when Army operations are involved; or within a contiguous training area. The source and/or victim (military or civil) of the interference may be located either on or off the installation.

(2) Regional Interference. Regional interference is defined as an interference situation that involves multiple installations or locations within a command or geographical area.

(3) National Interference. National interference is defined as an interference situation affecting a wide area or an entire area of command responsibility that may include situations where the source of interference cannot be identified or located using local or regional assets.

(4) Space Interference. Space interference is defined as an interference situation involving military, other Government, civil, or commercial space operations. *c*. Resolution Assets. There are several procedures and automated hardware and software tools available to aid in resolving interference at local, regional, national, or space environments. These resolution assets are:

(1) Records of previous local interference incidents maintained by the DOIM or unit frequency manager can be instrumental in resolving recurring interference. Such records can be searched and accessed rapidly if maintained in automated format.

(2) Frequency assignment records for the installation and the surrounding area from the frequency resource records (FRRS), government master file (GMF), and FCC files are an invaluable tool for ascertaining possible sources of interference, particularly when the interference is known or suspected to be fundamental-to-fundamental frequency interference. Subsets of these databases can be obtained from the regional AFCs or USACESO. Overseas Army activities involved in interference resolution can obtain frequency assignment information from the military FMOs, or from foreign host nation government and military spectrum management authorities.

(3) Equipment technical characteristics of spectrum-dependent materiel (e.g., the Army J-12 E3 database) are useful in resolving interference situations. In many instances, referring to equipment characteristics will lead to identifying the equipment causing the interference or suggest why the interference is disrupting performance of the victim equipment.

(4) Monitoring the interference while it is occurring provides

valuable information about the source of interference, e.g., type modulation (data, voice, pulse), antenna rotation rate, bandwidth, or other data about the source of interference.

(5) Hand-held DF equipment can be used to determine the location of interference sources. Army research, test, and evaluation facilities have sophisticated DF and monitoring equipment for use in accomplishing their assigned mission. Such activities include the White Sands Missile Range, NM; the Electronic Proving Ground, Fort Huachuca, AZ; the National Training Center, Ft Irwin, CA; the Joint Readiness Training Center, Ft Polk, LA; and the Combat Maneuver Training Center, Hohenfeld, GE. These facilities may be available for use in resolving interference situations on a case-bycase basis and should not be overlooked.

(6) In instances in which the source of the interference cannot be identified or located with locally available assets, technical assistance can be obtained from either ASC which maintains quick reaction field teams for field deployment or, through USACESO, from the JSC under the JSIR program.

(7) Other sources of technical assistance and information include commercial equipment manufacturers, developers of Army spectrum dependent materiel, or regional support activities of the FCC or FAA.

*d.* Reporting Procedures. It is the responsibility of all Army personnel to report incidents of interference. Reporting procedures are described below.

(1) Installation Procedures.

(*a*) Army units and other post, camp, and station tenant activities will notify, by the most expeditious means available, the supporting DOIM or frequency manager immediately when interference occurs. Personnel of affected activities will furnish the DOIM with as much information as possible about the interference during initial notification. Essential information required is defined in the interference reporting procedures described in Paragraphs C-11 and C-12.

(*b*) The DOIM supporting the affected Army activities will prepare and forward an initial interference report to ASC immediately in accordance with the format and procedures contained in Paragraphs C-11 and C-12. A copy of the initial report and subsequent interference reports will also be provided to the AFMO-CONUS, DoD AFC WSMR, DoD AFC Arizona, USACESO, and the MACOM DCSIM. The DOIM will prepare and forward a follow-up report when the interference has been resolved. The follow-up report will identify the causes of the interference and the corrective action taken.

(c) If safety of life is involved or potentially affected, the DOIM will report the interference incident immediately by the most expeditious means to the local command authority and the Office of the Army Spectrum Manager, Washington, DC.

(d) Incidents involving interference to Army operations in national and/or space environments must be reported and coordinated with ASC and USACESO. In cases of interference to space/satellite systems, USACESO must be contacted. USACESO will then coordinate with USSPACECOM to resolve the interference. Additional reporting and information collection requirements will be decided on a case-by-case basis by either ASC, USACESO, or the Army Spectrum Manager (SAIS-PAS-M).

(2) Tactical Operations/Training Exercise Procedure.

(a) In tactical operations or unit training exercises, unless it is specifically known that the interference is hostile, the interference incident should be reported to the unit signal officer.

(b) The signal officer will contact the unit or parent unit's frequency manager to assist in resolving the interference.

(c) If the frequency manager determines that the interference is hostile, the S2/S3 or G2/G3 representatives will be notified. The format for hostile interference reporting is found in the unit Standard Operating Procedure (SOP).

(d) If the frequency manager determines that assistance is required to resolve the interference, the frequency manager will contact the local DOIM or other regional POCs for assistance. An interference report is required. The format for reporting and a list of POCs is contained in Paragraphs C-11 and C-12. (e) The frequency manager should refer to the exercise directive, service component, and/or CINC guidance, directive, or SOP on reporting interference. The frequency manager should also refer to the Air Land Sea Application (ALSA) Center Tactics, Techniques and Procedures (TTP) Pamphlet of Joint Spectrum Management (SM-J) to obtain reporting interference procedures when the unit is part of a Joint Task Force (JTF) organization.

(f) If safety of life is involved, or potentially affected, or if an interference incident could seriously affect international spectrum usage agreements, the frequency manager will report the interference incident immediately by the most expeditious means to the local command authority, the regional AFC, and the Office of the Army Spectrum Manager, Washington, DC.

(g) Incidents involving interference to Army operations in national and/or space environments must be reported and coordinated with ASC and USACESO. Additional reporting and information collection requirements will be decided on a case-by-case basis by either ASC, USACESO, the Army Spectrum Manager (SAIS-PAS-M), or the responsible Army Component Commander.

*e. Coordination Procedures.* It is important that the DOIM or the frequency managers coordinate with installation activities when incidence of interference is reported. Likewise, those Army activities whose operations are affected by harmful interference must take action to report and coordinate efforts to resolve the interference quickly. The coordination procedures are:

(1) Affected units must coordinate immediately by the most expeditious means with the supporting DOIM or frequency manager when an interference situation occurs.

(2) Commensurate with local command procedures, the DOIM or frequency manager supporting the local affected activity may coordinate directly with ASC for advice and assistance when in doubt regarding interference resolution procedures.

(3) When locally available interference resolution assets are not sufficient to resolve the interference, the DOIM or frequency manager supporting the affected activity will coordinate directly with ASC for assistance. Informal, initial coordination with ASC is acceptable; however, the DOIM or frequency manager will prepare and forward a formal request through normal command channels to ASC within 24 hours following the informal request.

(4) If the interference incident involves other Federal agencies, the civilian community, or if safety of life is imminent, the DOIM or frequency manager will coordinate immediately by the most expeditious means with local command authority, ASC, AFCS, USACESO, and the Office of the Army Spectrum Manager, Washington, DC.

(5) National and space interference incidents will be reported and coordinated with USACESO and ASC.

(6) International interference incidents will be reported and coordinated in accordance with host government agreements, or through the ITU structure or meetings.

#### C-9. Possible Sources and Causes of Interference.

a. Possible sources and causes of interference include:

(1) Improper or inadequate frequency planning or engineering, e.g., lack of consideration of frequency intermodulation products in cosite environments.

(2) Not operating spectrum-dependent materiel in accordance with the RFA, e.g., antenna off azimuth, exceeding antenna height or transmit power limitations.

(3) Improper adjustment of tunable transmitters and receivers.

(4) Unknown and uncoordinated changes in equipment locations, e.g., installation of radio transmitters or repeaters in the civilian environment near Army activities.

(5) Operation of high power equipment such as radar in the vicinity.

(6) Lack of, or deterioration of, bonding and grounding provisions for spectrum-dependent materiel.

(7) Radiation from commercial industrial, scientific, and medical or other equipment not requiring licensing in the vicinity of telecommunications systems.

(8) Mixing two radio signals in a non-linear device external to

the transmitter and receiver and retransmitting intermodulation products of the two original signals. Although rarely identified, the socalled "rusty bolt problem" does occur.

(9) Operating two or more transmitters/receivers in close proximity (cosite) of each other.

(10) Using a frequency that is not assigned to the transmitter commonly referred to as "bootlegging."

(11) When a victim receiver is in the direct path of a transmitter and its intended receiver (overshot)

b. Different causes and sources of interference require different measures to locate and correct them. In some instances, resolution is as simple as changing frequencies (with proper authorization) or increasing the physical separation between the source of the interference and the victim. In other cases, monitoring and recording the interference with sophisticated DF equipment will be required to locate the source; spectrum analysis instrumentation may be needed to characterize the interference through measurement in order to identify possible interference sources.

c. Usually in simple situations, interference can be resolved locally; other times, technical assistance may be required from outside agencies. Permanent solution may require extensive coordination with various command, technical assistance, and operating activities. Changes in standards and specifications for equipment may be needed for future equipment development and acquisition, or modification to doctrine may be needed for using the spectrum-dependent materiel.

d. Procedures for identifying and resolving interference are discussed later. Fundamentally, reports of all interference incidents are required for analysis and permanent corrective actions. The information required in interference reports and the procedures for preparation and submission are contained in Paragraphs C-11 and C-12.

#### C-10. Interference Resolution and Prevention—Factors and Capabilities.

A number of factors and capabilities can be applied to resolving and preventing interference. The factors and capabilities are as follows:

a. Frequency Assignments. Properly planned and engineered frequency assignments are key to precluding interference. Possible frequency intermodulation products must be considered when choosing frequencies for cosited equipment. Assignments must observe frequency ranges allocated for specific telecommunications services. Transmitter power, polarity, radiation direction, and antenna height limitations must be observed to prevent transmitted energy from being transmitted beyond intended areas of reception.

b. Training. Training in equipment operations, spectrum discipline, and identifying and resolving interference is critical for effective telecommunications operations. Personnel at all levels must be familiar with provisions for reporting and eliminating interference. As more and more automated tools become available to assist the DOIMs and frequency management personnel, they must be trained in their operation and when and how to exploit these tools for rapid detection, location, and resolution of interference.

c. Coordination. In almost all instances of interference, coordinating resolution activities will involve personnel of operating units, supporting DOIMs, and frequency managers at posts, camps, and stations, command headquarters, and organizations providing administrative and technical assistance. Personnel must be familiar with the various schedules of activities occurring in the environment and coordination procedures and channels which have been established to resolve interference rapidly and effectively.

d. Automated Tools. Various automated tools, such as the Joint Spectrum Management System (JSMS), have been developed or are being developed to assist the DOIMs and frequency managers in resolving intency managers in identifying possible sources of interference. Formats, i.e., SFAF, have been created to expedite spectrum management matters. Electronic means such as Army E-Mail, bulletin board systems, and the Defense Data Network (DDN) also serve to expedite reporting activities. Personnel at all levels involved in interference resolution need to be aware of the existence of these tools and databases and become familiar with their use.

Automated tools in the hands of an experienced frequency manager will go a long way in resolving interference. The DOIMs and frequency managers can perform a valuable function by providing feedback to USACESO concerning ways that automated tools can be improved.

e. Monitoring. Monitoring the interfering signal can assist the DOIM and frequency manager in identifying the type of interference being encountered. In some cases, the radio receiver that is being victimized by interference can be used for monitoring; otherwise, more sophisticated spectrum analysis instruments will be required. Organizations equipped with more sophisticated equipment include ASC (operational EMC/EMI DF and monitoring support) and the JSC (JSIR support).

f. DF. Inexpensive off-the-shelf, hand-held DF equipment can be procured for use by DOIMs and frequency managers at Army posts, camps, and stations as an aid in determining the location of the source of interference. This equipment allows the DOIM or frequency manager to quickly locate radio transmissions in the vicinity of the post, camp, and station. More sophisticated DF equipment is available through ASC and, on a case-by-case basis, through the Army training centers, proving grounds, and test range facilities.

g. Interference Reporting. Historically, interference was reported only occasionally, if at all. Army personnel had little or no confidence in the ability of the "system" to resolve interference situations. Since the Army now has resumed responsibility for analyzing interference reports, this information has great potential value in categorizing interference and for providing input to a number of activities which are key to permanent solution of recurring interference situations. Included are changes to doctrine for Army telecommunications services and changes to EMC standards and specifications in conjunction with the Army E3 program.

#### C-11. Interference Reporting Format and Procedures.

Interference reports are a critically important part of the Army program for resolving interference. These reports, maintained by ASC and by each DOIM and frequency manager, service several purposes.

a. The DOIM or frequency manager uses the file of interference reports as a data resource for identifying and resolving recurring interference incidents on the local installation.

b. The database of interference reports at ASC and the E3 database at EPG can be accessed to assist the DOIM or frequency manager in identifying a possible interference source(s). These reports can also be analyzed by USACESO to indicate interference trends and the need for recommending to the AMC that equipment design characteristics requirements be changed. USACESO can also recommend to the Army that Army doctrine for using spectrumdependent materiel be changed. USACESO may utilizes the interference reports to develop requirements for interference resolution training of Army personnel.

#### C-12. Essential Information.

a. Initial Interference Reports. Initial interference reports are required from the DOIM or frequency manager as soon as possible following notification of interference by an Army unit or by any other Federal agency or civil activity within 80 kilometers of the installation, or in accordance with existing host government agreements. As a minimum, interference reports should include the following information.

- (1) Classification and downgrading Instructions (as appropriate)
- (2) Date of report year-month-day (YYMMDD)
- (3) Originator Information:
- (a) POC Title.(b) POC Name (Last, First, MI).
- (c) Organization.
- (d) Office Symbol.
- (e) Address (Street/building/suite).
- (f) Address (City, State, Zip Code).
- (g) Phone numbers (commercial and DSN).
- (4) Victim of Interference Information:
- (a) POC Title.

- (b) POC Name (Last, First, MI).
- (c) Organization.
- (d) Phone number (commercial and DSN).
- (5) Interference Occurred:
- (a) Date (YYMMDD).
- (b) Time Period (Start and End).
- (c) State/Country.
- (d) Location.
- (e) Coordinates (Latitude/Longitude).
- (6) Impact of Interference on the System.
- (7) Description of the System which was disrupted or degraded:
- (a) Equipment nomenclature.
- (b) Station Class (fixed (FX), land mobile (ML)).
- (c) Emission designator.
- (d) Receiving frequency.
- (8) Description of the Interference:
- (a) Type (noise, voice).
- (b) Repetition.
- (c) Duration (days, hours, minutes).
- (9) Local actions taken to resolve the interference.
- (10) Assistance required (type).
- b. Follow-up Interference Reports.

(1) Description of the system which caused the interference (if known).

(a) Equipment Nomenclature.

- (b) Frequency.
- (c) Location/Call Sign.
- (2) Technical Assistance Obtained:
- (a) Date (YYMMDD).
- (b) From Organization.
- (3) Cause of the Interference.
- (4) Action taken to resolve the interference.

(5) Recommendations for improving resolution techniques or for precluding any future interference.

# Appendix D Management Control Evaluation Checklist

## **D–1.** Function

The function covered by this checklist is to establish awareness for Radio Frequency Spectrum Allocation-to-Equipment Assessment. Information Mission Activities: Radio Frequency Spectrum Management.

#### D-2. Purpose

The purpose of this checklist is to assist program/project/product managers, contracting officers, or commanders at all levels responsible for; coordination and integration of the research and development, test, acquisition, fielding, procurement, leasing, modification, installation, or operation of Army materiel requiring radio frequency spectrum support under AR 5-12. To apply appropriate and cost-effective frequency allocation-to-equipment measures. The controls listed provide reasonable assurance that Army resources are adequately safeguarded.

#### **D-3.** Instructions

Answers must be based on the actual testing of key management controls such as document analysis, direct observation, interviewing, sampling, and simulation. Explain rationale for AYES@ responses or provide cross-reference to where rationale can be found. For ANO@ response, cross-reference to where corrective action plans can be found. For response of ANA,@ explain rationale. This checklist must be used 120 days after publication and every two years thereafter. Certification that this evaluation has been conducted must be accomplished on DA Form 11-2-R (Management Control Evaluation Certification Statement).

### D-4. Test Questions

a. Is the commander, program/project/product manager/contracting officer aware of the frequency allocation-to-equipment process? Response YES/NO/NA Remarks\*

Note. Cover off-the shelf, discretionary funding purchases.

b. Are the requirements for submission of DD Forms 1494 known?

Response YES/NO/NA

Remarks\*

c. Are records of DD Form 1494 and supporting documentation being maintained so information can be retrieved if necessary? Response YES/NO/NA Remarks\*

*d.* Has data been obtained from appropriate sources? Response YES/NO/NA

Remarks\*

e. Is the supporting Army Frequency Coordinator (AFC) known? Response YES/NO/NA

Remarks\*

*f.* Is the local installation/activity frequency manager known? Response YES/NO/NA

Remarks\*

g. Have the appropriate operations, research and development, procurement and telecommunication personnel read and understood AR 5-12?

Response YES/NO/NA Remarks\*

Note. . Cover off-the-shelf, discretionary funding purchases.

#### D-5. Comments

Help make this a better tool for evaluating management controls. Submit comments to HQDA ATTN: SAIS-PAS-M (The Army Spectrum Manager) Army, Pentagon, WASH DC 20310.

#### Appendix E Management Control Evaluation Checklist

#### E-1. Function

The function covered by this checklist is to define responsibilities of commanders at all levels. Information Mission Activities: Radio Frequency Spectrum Management.

## E-2. Purpose

The purpose of this checklist is to assist program/project/product managers, contracting officers, or commanders at all levels responsible for; coordination and integration of the research and development, test, acquisition, fielding, procurement, leasing, modification, installation, or operation of Army materiel requiring radio frequency spectrum support under AR 5-12. To assign responsibilities and establish objectives and procedures in writing. To designate responsibilities and to develop implementing regulations, SOPs, or other guidance. The controls listed provide reasonable assurance that Army resources are adequately safeguarded.

#### E-3. Instructions

Answers must be based on the actual testing of key management controls such as document analysis, direct observation, interviewing, sampling, and simulation. Explain rationale for AYES@ responses or provide cross-reference to where rationale can be found. For ANO@ response, cross-reference to where corrective action plans can be found. For response of ANA,@ explain rationale. This checklist must be used 120 days after publication and every two years thereafter. Certification that this evaluation has been conducted must be accomplished on DA Form 11-2-R (Management Control Evaluation Certification Statement).

#### E-4. Test Questions

*a.* Is all equipment, within the command, that radiates radio frequency (RF) energy essential to the mission? Response YES/NO/NA

Response Remarks\*

*b.* Are RF transmitters operated without causing harmful interference to receivers (Government and non-Government) located on or near the installation or activity?

Response YES/NO/NA Remarks\*

c. Are contractual obligations made for equipment that transmits or receives RF energy, before efforts are made to assure that it is frequency supportable?

Response YES/NO/NA Remarks\*

Note. Cover off-the-shelf, discretionary funding purchases.

*d.* Is equipment procured or leased, off-the-shelf, acceptable for licensing by the Federal Communications Commission (FCC) when it operates only in the non-Government frequency bands? Response YES/NO/NA Remarks\*

Note. Cover off-the-shelf, discretionary funding purchases.

*e.* Are local frequency managers and supporting AFCs allowed to participate in all phases of information systems planning, including plans to modify equipment? Response YES/NO/NA

Remarks\*

*f.* Are frequency allocation-to-equipment policies in AR 5-12 followed for transmitters and receivers of RF energy? Response YES/NO/NA Remarks\*

# E-5. Test Questions for commanders of Army posts, camps, stations, installations, and activities

Response Remarks\* YES/NO/NA

a. Are radio frequencies within the commander's area of responsibility authorized for use?

Response YES/NO/NA Remarks\*

*b.* Have policies and procedures been established for efficient RF spectrum management within the commanders' area of responsibility?

YES/NO/NA

Response Remarks\*

*c*. Has an installation or activity frequency manager been appointed for management of radio frequency and nontactical call sign assignments?

YES/NO/NA

Response Remarks\*

*d.* Are records being kept of the use of radio frequencies and nontactical call signs?

Response YES/NO/NA Remarks\*

*e.* Are procedures established to obtain frequency and call sign assignments from the supporting AFC? Response YES/NO/NA

Remarks\*

f. Has a 5-Year Review Program been established and implemented per AR 5-12? Response YES/NO/NA

Remarks\*

#### E-6. Comments.

Help make this a better tool for evaluating management controls.

Submit comments to HQDA ATTN: SAIS-PAS-M (The Army Spectrum Manager) Army, Pentagon, WASH DC 20310.

#### Appendix F

# Management Control Evaluation Checklist

#### F-1. Function

The function covered by this checklist is to determine Frequency Supportability for Equipment or System Research and Development (R&D). Information Mission Activities: Radio Frequency Spectrum Management.

## F-2. Purpose

The purpose of this checklist is to assist program/project/product managers, contracting officers, or commanders at all levels responsible for; coordination and integration of the research and development, test, acquisition, fielding, procurement, leasing, modification, installation, or operation of Army materiel requiring radio frequency spectrum support under AR 5-12. To plan for appropriate frequency supportability considerations early in the acquisition life cycle (e.g. during Concept Exploration Phase), and include in appropriate documentation. To insure frequency supportability requirements are included in all documentation used for the research and development (R&D) of all devices, equipment, and systems that will transmit and receive RF energy. The controls listed provide reasonable assurance that Army resources are adequately safeguarded.

#### F-3. Instructions

Answers must be based on the actual testing of key management controls such as document analysis, direct observation, interviewing, sampling, and simulation. Explain rationale for AYES@ responses or provide cross-reference to where rationale can be found. For ANO@ response, cross-reference to where corrective action plans can be found. For response of ANA,@ explain rationale. This checklist must be used 120 days after publication and every two years thereafter. Certification that this evaluation has been conducted must be accomplished on DA Form 11-2-R (Management Control Evaluation Certification Statement).

#### F-4. Test Questions

a. Has the operational capability of the proposed systems been identified? Response YES/NO/NA

Response	
Remarks*	

*b*. Based on operational capability, have specific missions to be supported by the system, and functions of the subsystem, been identified?

Response YES/NO/NA

Remarks\*

c. Have telecommunications requirements needed to support the proposed mission been identified (e.g., data rates, circuit quality/ reliability)? Response YES/NO/NA

Response Remarks\*

*d.* Have alternate communications systems or subsystems not requiring frequency supportability been identified? Response YES/NO/NA

Response Remarks\*

e. Have signal and/or software requirements been considered? Response YES/NO/NA

Remarks\*

*f.* Have appropriate EMC considerations been given to the frequency tunability of the contemplated equipment/hardware? Response YES/NO/NA

Remarks\*

g. Has consideration been given to intra-system implications on flexibility and EMC, and side effects on personnel and explosives? Response YES/NO/NA Remarks\*

h. Can off-the-shelf equipment be used? Response YES/NO/NA Remarks\* i. If off-the-shelf equipment cannot be used, is new equipment necessary? Response YES/NO/NA Remarks\* j. If new equipment is necessary, does this mean advancing the safeguarded. state-of-the-art to meet the telecommunications requirements? Response YES/NO/NA Remarks\* k. Based only on technical factors, have idealized frequency bands or ranges been identified which could best support the telecommunications requirements? YES/NO/NA Response Remarks\* *l*. Do existing allocations provide for the services required by the system in the idealized frequency bands? Response YES/NO/NA Remarks\* m. If allocation support does not exist in the idealized bands, are there any appropriately allocated bands that could satisfy the telecommunications requirements? Response YES/NO/NA Remarks\* Response *n*. Have penalties (in terms of time, cost, and performance) been Remarks\* associated with the available frequency allocation alternatives? YES/NO/NA Response Remarks\* Response o. Have areas or locations been identified in which the proposed Remarks\* system is to be deployed? Response YES/NO/NA Remarks\* Response p. Has adequate consideration been given to the electromagnetic Remarks\* environment including: idealized versus available frequencies; and tradeoffs of time, cost, and performance? sidered? Response YES/NO/NA Response Remarks\* Remarks\* q. Has a final study been published which supports rationale for band selection? YES/NO/NA Response Remarks\* Response r. Are there any system limitations that may be imposed by Remarks\* electromagnetic environment factors? Response YES/NO/NA Remarks\* Response Remarks\* F–5. Comments Help make this a better tool for evaluating management controls. Submit comments to HQDA ATTN: SAIS-PAS-M (The Army Spectrum Manager) Army, Pentagon, WASH DC 20310.

# Appendix G Management Control Evaluation Checklist

#### G-1. Function

The function covered by this checklist is to determine frequency supportability modifications in the experimental phase (Frequency Supportability for Equipment or System Research and Development (R&D)). Information Mission Activities: Radio Frequency Spectrum Management.

# G-2. Purpose

The purpose of this checklist is to assist Program/project/product managers, contracting officers, or commanders at all levels responsible for; coordination and integration of the research and development, test, acquisition, fielding, procurement, leasing, modification, installation, or operation of Army materiel requiring radio frequency spectrum support under AR 5-12. To plan for experimentation criteria upon which to validate or modify frequency supportability considerations originally identified in the planning phase. To insure frequency supportability requirements are included in all documentation used in experimentation with all devices, equipment, and systems that will transmit or receive RF energy. The controls listed provide reasonable assurance that Army resources are adequately

#### G-3. Instructions

Answers must be based on the actual testing of key management controls such as document analysis, direct observation, interviewing, sampling, and simulation. Explain rationale for AYES@ responses or provide cross-reference to where rationale can be found. For ANO@ response, cross-reference to where corrective action plans can be found. For response of ANA,@ explain rationale. This checklist must be used 120 days after publication and every two years thereafter. Certification that this evaluation has been conducted must be accomplished on DA Form 11-2-R (Management Control Evaluation Certification Statement).

#### G-4. Test Questions

a. Have frequency supportability changes been identified from previously submitted planning data?

YES/NO/NA

b. Have aspects of systems, along with telecommunications characteristics, requiring experimentation been identified? YES/NO/NA

c. Have specified, alternative telecommunications requirements been identified in detail?

YES/NO/NA

d. Have signal and/or software philosophy techniques been con-

YES/NO/NA

e. Has consideration been given to EMC potential of alternative telecommunications designs been investigated?

YES/NO/NA

f. Has appropriate EMC consideration been given to the frequency tunability of the proposed equipment/hardware?

YES/NO/NA

g. Has adequate consideration been given to intra-system implications on flexibility and EMC, and the side effects on personnel and explosives?

YES/NO/NA

Respone Remarks\*

h. Have all factors been identified that affect spectrum requirements and EMC?

YES/NO/NA Response

Remarks\*

i. Have pertinent economic trade-offs with respect to spectrum use been identified?

YES/NO/NA Response Remarks\*

j. Are spectrum considerations included in any request for procurement (RFP) preparation?

Response YES/NO/NA

Remarks\*

k. In updating the frequency band selection alternatives in this phase, were the most satisfactory frequency bands identified? Response YES/NO/NA Remarks\*

*l*. Are there other studies available about the frequency selection process to prepare for filing for frequency support of the system? Response YES/NO/NA Remarks\*

#### G-5. Comments

Help make this a better tool for evaluating management controls. Submit comments to HQDA ATTN: SAIS-PAS-M (The Army Spectrum Manager) Army, Pentagon, WASH DC 20310.

## Appendix H Management Control Evaluation Checklist

#### H-1. Function

The function covered by this checklist is to determine frequency supportability modifications in the developmental stage (Frequency Supportability for Equipment or System Research and Development (R&D)). Information Mission Activities: Radio Frequency Spectrum Management.

#### H-2. Purpose

The purpose of this checklist is to assist Program/project/product managers, contracting officers, or commanders at all levels responsible for; coordination and integration of the research and development, test, acquisition, fielding, procurement, leasing, modification, installation, or operation of Army materiel requiring radio frequency spectrum support under AR 5-12. To plan for development criteria upon which to validate or modify frequency supportability considerations originally identified in previous stages. To insure frequency supportability requirements are included in all documentation used in all testing and developmental documentation. The controls listed provide reasonable assurance that Army resources are adequately safeguarded.

#### H-3. Instructions

Answers must be based on the actual testing of key management controls such as document analysis, direct observation, interviewing, sampling, and simulation. Explain rationale for AYES@ responses or provide cross-reference to where rationale can be found. For ANO@ response, cross-reference to where corrective action plans can be found. For response of ANA,@ explain rationale. This checklist must be used 120 days after publication and every two years thereafter. Certification that this evaluation has been conducted must be accomplished on DA Form 11-2-R (Management Control Evaluation Certification Statement).

#### H-4. Test Questions

a. Have frequency supportability changes been identified from previously submitted data?

YES/NO/NA Response

Remarks\*

b. Have aspects of systems or subsystems been identified which require development?

YES/NO/NA Response

Remarks\*

c. Have proper adjustments been made to overall system configuration based on experimental stage?

Response YES/NO/NA

Remarks\*

d. Have the following been considered: instrumentation, spurious emissions, electromagnetic coupling, emission spectrum, antennas, receiver, and signal processing requirements and/or limits? Response YES/NO/NA Remarks\*

e. Do EMC specifications comply with prevailing standards and criteria?

YES/NO/NA Response

Remarks\*

f. Can difference from prevailing standards and criteria for EMC be justified? Response

YES/NO/NA

Remarks\*

g. Have areas of current design been identified in which system performance is degraded due to EM environment, and will EMI problems be created in other systems? YES/NO/NA

Response

Remarks\*

h. Based on prior planning experience, has basis for modification of EMC/EMI standards been identified?

Response YES/NO/NA

Remarks\*

i. Have plans for special tests, measurement techniques, and simulation efforts been defined which will aid in validating design? Response YES/NO/NA Remarks\*

j. Did Development Phase review of frequency band selection change from previous stage review?

YES/NO/NA Response

Remarks\*

k. If frequency band selection was changed in Development Stage review, was change technologically justified?

YES/NO/NA Response

Remarks\*

l. Were specific frequency requirements identified for development systems-integration testing? Response YES/NO/NA

Remarks\*

#### H-5. Comments

Help make this a better tool for evaluating management controls. Submit comments to HQDA ATTN: SAIS-PAS-M (The Army Spectrum Manager) Army, Pentagon, WASH DC 20310.

# Appendix I Management Control Evaluation Checklist

#### I-1. Function

The function covered by this checklist is to validate frequency supportability prior to procurement of equipment, systems, or facilities which are spectrum-dependent. (Frequency Supportability for Equipment or System Research and Development (R&D)). Information Mission Activities: Radio Frequency Spectrum Management.

#### I-2. Purpose

The purpose of this checklist is to assist Program/project/product managers, contracting officers, or commanders at all levels responsible for; coordination and integration of the research and development, test, acquisition, fielding, procurement, leasing, modification, installation, or operation of Army materiel requiring radio frequency spectrum support under AR 5-12. To validate frequency supportability before procurement of devices, equipment, systems, or facilities, and include all appropriate documentation. To insure frequency supportability requirements are included in all documentation for procurement of all devices, equipment, systems, and facilities that will transmit or receive RF energy. The controls listed provide reasonable assurance that Army resources are adequately safeguarded.

#### I-3. Instructions

Answers must be based on the actual testing of key management controls such as document analysis, direct observation, interviewing, sampling, and simulation. Explain rationale for AYES@ responses or provide cross-reference to where rationale can be found. For ANO@ response, cross-reference to where corrective action plans can be found. For response of ANA,@ explain rationale. This checklist must be used 120 days after publication and every two years thereafter. Certification that this evaluation has been conducted must be accomplished on DA Form 11-2-R (Management Control Evaluation Certification Statement).

I–4 Test Questions	Kennai
<i>a</i> . Have changes in all previous stages been identified and sub- mitted for consideration?	<b>I–5. C</b> Help 1
Response YES/NO/NA	Submi
Remarks*	Spectr
b. Have basic technical characteristics been clearly defined, i.e.	
transmitted power, emission characteristics (bandwidth, modulation,	
data rate), antenna orientation and directivity, receiver characteris-	
Desponse VEC/NO/NA	
Response IES/NO/NA Remarks*	
c Have all special technical characteristics peculiar to the pro-	
posed system and having potential EMC problems been adequately	
identified and described (i.e. complex modulation schemes, filters,	
special receiver circuitry, signal processing, etc.)?	
Response YES/NO/NA	
Remarks*	
d. Has all data developed during previous phases of the system's	
life cycle been considered and evaluated in completing the telecom-	
munications requirements and/or characteristics?	
Response YES/NU/NA Pemerks*	
a Have previous FMC analyses been reviewed and undated?	
Response YES/NO/NA	
Remarks*	
f. Are additional studies required to assess changes to system	
design, deployment, and other factors?	
Response YES/NO/NA	
Remarks*	
g. Have equipment tests been specified to identify and validate	
EML identified in provious and undeted analyses?	
Response YFS/NO/NA	
Remarks*	
h. Have both static and dynamic electromagnetic environments	
for testing been considered?	
Response YES/NO/NA	
Remarks*	
<i>i</i> . Have in-field EMC assessment techniques (frequency assign-	
ment aids, equipment usage aids, grading for performance, measure-	
support procurement and deployment planning for the system?	
Response YES/NO/NA	
Remarks*	
j. Has a discrete frequency plan been proposed?	
Response YES/NO/NA	
Remarks*	
k. Has the number of frequencies required, within a specified	
band of bands, been identified?	
Remarks*	
<i>l</i> . Have channeling limitations, transmit/receive separation, and	
other pertinent limitations and system peculiarities been identified?	
Response YES/NO/NA	
Remarks*	
m. Has all deployment information (mobile, transportable, and/or	
fixed) for the system(s) been developed?	
Response YES/NO/NA	
remarks"	
<i>n</i> . may notations been identified for testing and operation? Response VFS/NO/NA	
Remarks*	
o. Have operational altitudes and orbital locations and character-	
istics been identified for airborne and satellite systems?	
Response YES/NO/NA	
Remarks*	

p. Have EMC procedures and techniques been identified to educate users on deployment of the system(s)? Response YES/NO/NA Remarks\*

## –5. Comments

Help make this a better tool for evaluating management controls. Submit comments to HQDA ATTN: SAIS-PAS-M (The Army Spectrum Manager) Army, Pentagon, WASH DC 20310.

#### Glossary

Section I Abbreviations

AAE Army Acquisition Executive

ABCA Australia, Britain, Canada, and America

ACCS Army Command and Control System

ACP Allied Communications Publication

ACSIM Assistant Chief of Staff for Information Management

AEPG Army Electronic Proving Ground

AFC Army Frequency Coordinator

AFMO Army Frequency Management Office

AIMS Army IFF Mark XII Systems

AIRP Army Interference Resolution Program

ALSA Air Land Sea Application

AM Amplitude Modulation

AMC Army Materiel Command

APPBS Army Planning, Programming, and Budgeting System

AR Army Regulation

ARC Army Reserve Command

ARFA Allied Radio Frequency Agency

ARNG Army National Guard

ARSTAF Army Staff

ASA Assistant Secretary of the Army

ASM2P Army Spectrum Management Master Plan ATC Air Traffic Control

AWIS Army WWMCCS Information System

AZ Arizona

**BEEO** Battlefield Electromagnetic Environments Office

**BR** Radiocommunication Bureau (French translation)

BSMO Battlefield Spectrum Management Office or Officer

C3CM Command, Control, and Communications Countermeasures

C-E Communications-Electronics

CB Citizen Band

CCEB Combined Communications-Electronics Board

CECOM Communications-Electronics Command

**CEOI** Communications-Electronics Operation Instructions

CG Commanding General

CINC Commander-in-Chief (Unified Command)

**CINCEUR** Commander-in-Chief, Europe

CINCPAC Commander-in-Chief, Pacific Command

CINCSOU Commander-in-Chief, Southern Command

COEA Cost and Operational Effectiveness Analysis

CONUS Continental United States

CONUSA Continental United States Army

COTS Commercial off-the-shelf

CTC Combat Training Center CWG Consultative Working Group

**DA** Department of the Army

**DACC** The Department of the Army Command and Control System

DCS Defense Communications System

**DCSIM** Deputy Chief of Staff for Information Management

**DCSOPS** Deputy Chief of Staff, Operations

**DDN** Data Defense Network

**DECON** Deconfliction

**DET** Detachment

DF Direction Finding

**DIRLAUTH** Direct Liaison Authorized

DISC4 Director of Information Systems for Command, Control, Communications and Computers

**DoD** Department of Defense

**DoDD** Department of Defense Directive

**DOIM** Director of Information Management

**DSC** Digital Selective Calling

DSCS Defense Satellite Communications System

DSN Defense Switched Network

**DT** Development Test

E<sup>3</sup> Electromagnetic Environmental Effects

EA Electronic Attack

**ECAC** Electromagnetic Compatibility Analysis Center (Now the Joint Spectrum Center) **ELINT** Electronics Intelligence

**EMC** Electromagnetic Compatibility

**EMCP** Electromagnetic Compatibility Program

**EME** Electromagnetic Environment

EMETF Electromagnetic Environmental Test Facility

**EMI** Electromagnetic Interference

**EMP** Electromagnetic Pulse

**EMRH** Electromagnetic Radiation Hazard

**EMV** Electromagnetic Vulnerability

**EP** Electronic Protect

**EPG** Electronic Proving Ground

ES Electronic Support

**EW** Electronic Warfare

FAA Federal Aviation Administration

FAS Frequency Assignment Subcommittee

FCC Federal Communications Commission

FM Field Manual

FMG Frequency Management Group

FMO Frequency Management Office

FORSCOM Forces Command

FP Frequency Panel

FRRS Frequency Resource Record System

FY Fiscal Year

**GHZ** Gigahertz GMF Government Master File

**HF** High-Frequency

HQDA Headquarters, Department of the Army

**Hz** Hertz

**IEEE** Institute of Electrical and Electronics Engineers

IEW Intelligence and Electronic Warfare

IFF Identification, Friend or Foe

IFRB International Frequency Registration Board

IM Information Mission

ING International Notification Group

INSCOM Intelligence and Security Command

IPR In-Process Review

**IRAC** Interdepartment Radio Advisory Committee or Internal Review Audit Compliance

IRM Information Resource Management

ISC Information Systems Command

ISEC Information Systems Engineering Command

**ISM** Industrial, Scientific, and Medical

ITU International Telecommunication Union

JCS Joint Chiefs of Staff

JETDS Joint Electronics Type Designation System

JFMO Joint Frequency Management Office

JFP Joint Frequency Panel

JRTC Joint Readiness Training Center JSC Joint Spectrum Center

JSIR Joint Spectrum Interference Resolution

JSMS Joint Spectrum Management System

JTF Joint Task Force

**kHz** Kilohertz

LCSMM Life Cycle System Management Model

LMR Land Mobile Radio

LOS Line-of-Sight

LOT Letter of Transmittal

MACOM Major Army Command

MARS Military Affiliate Radio Station

MCS Maneuver Control System

MDW Military District of Washington

MED Medical

MEDEVAC Medical Evacuation

MHZ Megahertz

MM Maritime Mobile

MRB Materiel Review Board

MSC Major Subordinate Command

MSCC Major Subordinate Component Command

MSE Mobile Subscriber Equipment

MUSARC Major United States Army Reserve Command

NASA National Aeronautics and Space Administration NATO North Atlantic Treaty Organization

NAVAIDS Navigational Aids

NCA National Command Authority

NGB National Guard Bureau

NSA National Security Agency

NTC National Training Center

NTIA National Telecommunications and Information Administration

**OCONUS** Outside Continental United States

**ODP** Operational Development Plan

**OPTEC** Operational Test and Evaluation Command

**OSA** Office of the Secretary of the Army

OT Operational Test

PAC Pacific

PAM Pamphlet

PE Propagation Engineering

**PEO** Program Executive Office

PM Program or Project Manager

**PPBES** Planning, Programming, Budgeting, and Execution System

PPI Planned Product Improvement

**PUB** Publication

**PVI** Positive Voice Identification

**QRC** Quick Reaction Capability

**QSTAG** Quadripartite Standardization Agreement RACES Radio Amateur Civil Emergency Services

RADHAZ Radiation Hazard

**RBECS** Revised Battlefield Electronics CEOI System

RC Reserve Component

RCC Range Commanders' Council

RFA Radio Frequency Authorization

RD&A Research, Development and Acquisition

RF Radio Frequency

RFL Restricted Frequency List

**RR** Radio Regulation

**RTCA** Radio Technical Commission for Aeronautics

RTCM Radio Technical Commission for Marines

SATCOM Satellite Communications

SFAF Standard Frequency Action Format

SG Study Group

SIDPERS Standard Installation/Division Personnel System

SIGINT Signal Intelligence

SM-J Joint Spectrum Management

SMO Spectrum Management Office

SMR Specialized Mobile Radio

**SO** Signal Officer

SOC Special Operations Command

SOI Signal Operation Instruction SOP Standard Operating Procedure

SPS Spectrum Planning Subcommittee

SRA Ship Radio Authorization

SSDC Space and Strategic Defense Command

STANAG Standardization Agreement

STANFINS Standard Financial System

STD Simulated Tactical Deployment

STN Station

TAG The Adjutant General

**TAMMIS** Theater Army Medical Management Information System

TB Technical Bulletin

**TDA** Table of Distribution and Allowance

**TECOM** Test and Evaluation Command

TELEX ID Teleprinter Identifier

TIWG Test Integration Working Group

TOE Table of Organization and Equipment

**TPFDD** Time Phased Force Deployment Data

TRADOC Training and Doctrine Command

TTP Tactics, Techniques, and Procedures

US&P United States and Possessions

USA United States Army

**USACE** United States Army Corps of Engineers

USACESO United States Army Communications-Electronics Services Office USAKA US Army Kwajalein Atoll

USAMEDCOM US Army Medical Command

USAR United States Army Reserve

USARC US Army Reserve Command

USAREUR United States Army, Europe

USARJ United States Army Japan

USARPAC United States Army Pacific

USASSDC US Army Space and Strategic Defense Command

#### USMCEB

United States Military Communications-Electronics Board

USNG United States National Guard

USSPACECOM United States Space Command

WRC World Radio Conference

#### WSMR

White Sands Missile Range

#### WWMCCS

Worldwide Military Command and Control System

YPG

Yuma Proving Ground

Section II Terms

#### Allocation

An allocation is the designation of frequency bands for use in performing specific functions or services. Allocations are made to communications services such as fixed, mobile, broadcast, and amateur.

#### Allotment.

An allotment is the designation of specific frequency bands or frequencies within a prescribed allocation. Within the Federal Government, allotments are made to specific Government agencies.

#### Army Frequency Coordinator.

The AFC serves as the focal point for radio frequency and call sign requests within the US&P which required registration and clearance at the national level. The AFC coordinates with counterparts of other Military Services and Government agencies prior to making frequency and call sign assignments. M2P).

# Army Spectrum Management Master Plan (ASM2P)

A document that issues spectrum management policies, plans, and programs. It includes directives and programs to accomplish such plans and policies. The ASM2P also provides the management baseline data to periodically review the effectiveness of spectrum management programs.

## Assignment.

An assignment is the designation of a specific frequency or frequencies for use by a radio station under specified conditions.

#### Deconfliction.

An integral part of spectrum management, deconfliction is a process of optimizing the usage of the spectrum incorporating both the requirements of the battlefield spectrum managers and the EW operations.

# Department of Defense Area Frequency Coordinator.

An office empowered by the USMCEB to provide overall management and use of the spectrum in areas on or near a National Test Range or other designated complex.nation and resolution of harmful radio frequency interference caused by or to DoD users. Terms of reference for the DoD AFCs is contained in ACP 190 US SUPP-1, Annex B.

#### Electromagnetic Compatibility.

The capability of electrical and electronic systems, equipment, and devices to operate in their intended electromagnetic environment within a defined margin of safety, and at design levels of performance without suffering or causing unacceptable degradation as a result of electromagnetic interference.

#### Electromagnetic Environment.

All electromagnetic radiation, manmade and natural, emanating from emitters at the lowest alternating current to the highest RF in the environment are included.

#### **Electromagnetic Environmental Effects.**

The impact of the electromagnetic environment upon the operational capability of military forces, equipment, systems, and platforms. It encompasses all electromagnetic disciplines, including EMC/EMI; EMV; electromagnetic pulse; electronic warfare; hazards of electromagnetic radiation to personnel, ordnance, and volatile materials; and natural phenomena effects of lightning and pstatic.

# Electromagnetic Interference.

Any electromagnetic disturbance that interrupts, obstructs, or otherwise degrades or limits the effective performance of electronics/electrical equipment. It can be induced intentionally, as in some forms of EW warfare, or unintentionally, as a result of spurious emissions and responses, intermodulation products, or the like.

#### Electromagnetic Pulse.

The electromagnetic radiation from a nuclear explosion caused by Compton-recoil electrons and photoelectrons from photons scattered in the materials of the nuclear device or in a surrounding medium. The resulting electric and magnetic fields may be coupled with electrical/electronic systems to produce damaging current and voltage surges. The electromagnetic pulse may also be caused by non-nuclear means.

#### Electromagnetic Spectrum.

The range of frequencies of electromagnetic radiation from zero to infinity. It is divided into 26 alphabetically designated bands.

#### **Electromagnetic Radiation.**

Radiation made up of oscillating electric and magnetic fields and propagated with the speed of light. This radiation includes gamma radiation; x-rays; ultraviolet, visible, and infrared radiation; and radar and radio waves.

#### **Electromagnetic Radiation Hazards.**

Those electromagnetic radiations which are a source of direct danger to the human body or those that could possibly detonate or ignite explosives, flammable gases or vapors, dust, or easily ignitable particles or fibers.

#### Electronic Protect (EP).

The division of EW involving actions taken to ensure friendly, effective use of the electromagnetic spectrum despite the enemy's use of EW.

#### Electromagnetic Vulnerability.

The characteristics of a system which cause it to suffer a definite degradation (incapability to perform the designated mission) after being subjected to a certain level of effects in an unnatural (manmade), hostile environment. Electromagnetic vulnerability measures the system's incapacity to perform in the presence of hostile EA. Electromagnetic vulnerability is measured only in its own operational environment (actual or simulated) and under conditions which take into account:

a. How susceptible the system is.

b. How easily it can be intercepted by hostile intercept and direction-finding activities.

*c*. The nature and extent of the hostile EW threat.

#### Electronic Attack (EA).

Actions taken to prevent or reduce an enemy's effective use of the electromagnetic spectrum.

Note. This definition was extracted from FM 34-40-7.

#### **Electronic Attack Clearance.**

An approval or authorization to conduct EA (e.g., jamming, chaff drops) in a given geographical area under specified conditions and controls to prevent harmful interference to other authorized spectrum users. The provisions of AR 105-86 apply in CONUS. OCONUS, appropriate Unified directives apply. A frequency assignment does not provide authority to conduct EA.

#### Electronic warfare.

Military action involving the use of electromagnetic energy to determine, exploit, reduce, or prevent hostile use of the electromagnetic spectrum, and action which retains friendly use of the electromagnetic spectrum. There are three divisions of electronic warfare: EA, EP, and ES measures.

#### EMC analysis.

An objective investigation into the potential frequency spectrum resource requirements of Army frequency spectrum-dependent systems or equipment. Recommended to be conducted prior to entering the phases of the LCSMM as a cost-saving measure and to determine if proposed system or equipment is frequency supportable in its proposed environment. EMC analysis will present the trade-offs regarding use of the electromagnetic spectrum in various technical concepts for fulfilling Army materiel requirements.

# Frequency Allocation-to-Equipment Process.

The process is a spectrum requirement and is used to determine that Army spectrum-dependent equipment operates in frequency bands according to national and international frequency allocation tables and conforms to all other applicable spectrum management regulations, directives, standards, and specifications. The process is started as early as possible in the research, development, production, and procurement cycle. This early start is to efficiently assure future spectrum assignments and EMC. (See Chapter 4 for more details).

# Information Resources Management (IRM).

The planning, budgeting, organizing, directing, training, promoting, controlling, and management activities associated with the burden, collection, creation, maintenance, utilization, dissemination, and disposition of information regardless of media. IRM includes the management of information and information related resources and systems, whether manual or automated, such as records management activities, privacy and security of records, agency sharing and dissemination of information, and acquisition and use of automatic data processing, telecommunications, and other information technology.

#### Interference.

See electromagnetic interference.

#### Occupied Bandwidth.

The bandwidth within which 99 percent of the total emitted energy is contained. The occupied bandwidth must encompass the necessary bandwidth. If not, the transmitter will not emit a signal wide enough to successfully convey all the information.

#### Radio Wave Propagation.

The transfer of energy by electromagnetic radiation at radio frequencies.

#### Spectrum-Dependent Equipment.

Army telecommunications and command and control systems (including weapon systems), subsystems, or equipment which either depend on or affect the use of the electromagnetic spectrum. Further defined by DoD Regulation 5000.2-R as, "systems and equipment that emit or receive Hertzian waves."

#### Spectrum Management.

The management of the use of electromagnetic spectrum resources. The goal of Army spectrum management is to support telecommunications, weapons systems, and electronic warfare requirements. This goal will be accomplished through the acquisition of spectrum resources, the efficient use of those resources, and the attainment of electromagnetic compatibility.

#### Spectrum Management Doctrine.

Fundamental principles which guide Army use of the electromagnetic spectrum for operation of Army telecommunications and command and control systems (including weapons systems), subsystems, and equipment. These principles are official and require sound military and technical judgment in application. Principles may be based on: *a*. Basic physical phenomena associated

with radio wave propagation and radiation.

*b*. National or international regulatory constraints on the use of the frequency spectrum.

*c*. The need for coordination and cooperation among users of the frequency spectrum. *d*. EW implications.

#### Spectrum Plan.

An organized and documented scheme which identifies the specific spectrum resources required for a military operation (such as a contingency operation or field training exercise) or to operate a telecommunications system (such as a satellite communications system). Spectrum plans will be engineered to ensure communicability and to reduce interference, both among the frequencies in the plan and between the frequencies in the plan and those in use or planned in the coexistent electromagnetic environment.

#### Spectrum Resources.

Allocations, allotments, or assignments of portions of the electromagnetic spectrum for the accomplishment of a specific function or telecommunications service. In an operational environment, spectrum resources are the specific number and types of frequency assignments needed to operate items of spectrumdependent materiel.

#### Spectrum Support.

The potential availability of operating frequencies (frequency assignments) to meet specific type-of-service and operational requirements.

#### **Telecommunications.**

Any transmission, emission, or reception of signs, signals, writing, images, and sound or information of any nature by wire, radio, or other electromagnetic or optical systems.

#### **Telecommunications Service.**

A specific function performed by a system, subsystem, or equipment such as fixed, mobile, broadcasting, or mobile satellite.

#### Waiver.

A written authorization to accept a system, subsystem, or equipment which, having been submitted for inspection, does not meet specified requirements but is considered suitable for use 'as is' or after rework by an approved method.

#### Index

This index is organized alphabetically by topic and subtopic. Topics and subtopics are identified by paragraph number.

**Electromagnetic Spectrum** 

Management, 1-6

Management in the IM Environment, 1-7

Policies, 1-8

- Army Spectrum Management Responsibilities
- **Functional Coordination Channels, 2-2**
- Army Spectrum Management Organizations, 2-2
- Director of Information Systems for Command, Control, Communications, and Computers (DISC4), 2-6
- The Army Spectrum Manager, 2-7
- US Army Communications Electronics Services Office (USACESO), 2-8
- Areas of Responsibility for Army Frequency Coordinators, B-1
- Army Electromagnetic Compatibility, Deconfliction, and Interference Control Programs
- Introduction, 3-1
- Management Requirements for achieving Electromagnetic Compatibility (EMC), 3-3
- **Deconfliction**, 3-7
- Army Interference Resolution Program (AIRP), 3-8
- Frequency Allocation-To-Equipment process for DD Form 1494
- Policies, 4-2
- Application for Spectrum Certification, 4-3
- Processing DD Form 1494, 4-4
- EMC Standards of Telecommunications Equipment, 4-8
- Waivers/exceptions, 4-6
- Spectrum Requests and Assignments Spectrum Coordination Channels, 5-2
- Types of Spectrum Actions, 5-3
- Army Policy for Spectrum Assignments, 5-5
- Spectrum Coordination Procedures, 5-6
- Radio Station Identification Responsibility for Radio Station Identification, 6-2
- International Call Signs, 6-3 International Call Sign Assignment Authority, 6-4
- Publications, 6-5

Security

Department of the Army Information Security Program

Headquarters Department of the Army Washington, DC 29 September 2000



# SUMMARY of CHANGE

AR 380-5 Department of the Army Information Security Program

This revision--

- Emphasizes the responsibilities of Headquarters Department of the Army, the Commander, the Command Security Manager, the Supervisor, and the Individual (chap 1).
- Updates information relating to Restricted Data, Formerly Restricted Data, Sensitive Compartmented Information, Communications Security Information, and Special Access Program Information (chap 1).
- o Expands and clarifies exceptional situations and waivers (chap 1).
- o Specifies the process for applying original classification (chap 2).
- Removes the declassification statement Originating Officials Determination Required or "OADR" (para 2-11).
- Changes the distribution process for security classification guides (para 2-18).
- o Outlines the procedures for challenges to classification (para 2-22).
- o Updates information regarding the Army Declassification Program (chap 3).
- o Updates information relating to the deadline for automatic declassification, adding an additional 18 months (para 3-5).
- o Gives clearer information in regards to destruction, declassification, purging, and clearing (chap 3).
- o Replaces the statement "US ONLY" with "NOFORN" (para 4-6).
- o Updates Warning Notices for printed documents and those on Automated Information Systems (para 4-12).
- o Lists the obsolete restrictions and control markings (para 4-13).
- o Outlines the requirements relating to For Official Use Only (FOUO) information (chap 5).
- Replaces the statement "Limited Official Use" with "Sensitive But Unclassified" (para 5-7).
- o Outlines the requirements relating to Drug Enforcement Administration Sensitive Information (chap 5).

- o Outlines the requirements relating to Department of Defense Unclassified Controlled Nuclear Information (chap 5).
- o Outlines the requirements relating to sensitive information and the Computer Security Act of 1987 (chap 5).
- Outlines the requirements relating to distribution statements on scientific and technical documents (para 5-24).
- o Outlines the requirements regarding the SF 312 (Classified Information Nondisclosure Agreement (NDA) (chap 6).
- Explains the requirements to access Department of the Army classified information by individuals or agencies outside the Executive Branch (para 6-8).
- o Adds guidance to the use of speakerphones (para 6-14).
- Updates guidance on classified meetings, conferences, and acquisition meetings (para 6-18).
- Removes the requirement for the Entry/Exit Inspection Program and the Two-Person Integrity Program (para 6-36).
- o Updates the federal specifications for locks (para 7-4).
- o Updates procedures for handcarrying of classified material (chap 8).
- o Outlines requirements for transportation plans (para 8-7).
- Emphasizes annual refresher training requirements for the security education program (para 9-7).
- o Rescinds DA Form 2134 (Security Violation(s) Report) (para 10-3).
- Updates security controls on dissemination and marking of warning notices on intelligence information to include the latest version of Director of Central Intelligence Directive 1/7 (app D).
- o Addresses security procedures for documents created for and on automated information systems and Internet web-based display (app E).
- o Updates the Management Control Evaluation Checklist (app F, section I).
- o Lists and describes the modern army recordkeeping system file numbers associated with this Regulation (app F, section II).
- o Updates information of Special Access Programs (app I).
- o Updates the abbreviations and terminology lists (glossary).

Headquarters Department of the Army Washington, DC 29 September 2000

# \*Army Regulation 380–5

Effective 31 October 2000

# Security

# Department of the Army Information Security Program

By Order of the Secretary of the Army:

ERIC K. SHINSEKI General, United States Army Chief of Staff

Official:

had B Hul

JOEL B. HUDSON Administrative Assistant to the Secretary of the Army

**History.** This publication publishes a revision of this publication. Because the publication has been extensively revised, the changed portions have not been highlighted.

**Summary.** This regulation implements the policy set forth in Executive Order (E. O.) 12958, "Classified National Security Information", April 17, 1995, with amendments, and DOD 5200.1–R, "Information Security Program." It establishes the policy for classification, downgrading, declassification, and safeguarding of information requiring protection in the interest of national security.

**Applicability.** This regulation applies to all military and civilian members of the Active Army, Army National Guard of the United States (ARNGUS), and U.S. Army Reserve (USAR) and Department of the Army (DA) personnel. During mobilization, chapters and policies contained in this regulation may be modified by the proponent.

**Proponent and exception authority.** The proponent of this regulation is the Deputy Chief of Staff for Intelligence. The Deputy Chief of Staff for Intelligence has the authority to approve exceptions to this regulation that are consistent with controlling law and regulation, and as stated in this regulation. Paragraph 1–20 of this regulation further delineates waiver and exception to policy authority.

Army management control process.

This regulation contains management control provisions and identifies key management controls that must be evaluated.

**Supplementation.** Supplementation of this regulation is permitted at command option, but is not required.

**Suggested Improvements.** Users are invited to send comments and suggestions on DA Form 2028, Recommended Changes to Publications and Blank Forms, through command channels to HQDA DCSINT (DAMI-CH), 2511 Jefferson Davis Highway, Suite #9300, Arlington, VA 22209–3910.

**Distribution.** This publication is available in electronic media only and is intended for command levels A, B, C, D, and E for the Active Army, the Army National Guard, and the U.S. Army Reserve.

**Contents** (Listed by paragraph and page number)

# Chapter 1 General Provisions and Program Management, *page 2*

Section I Introduction, page 1 Purpose • 1–1, page 1 References • 1–2, page 1 Explanation of abbreviations and terms • 1–3, page 1

Section II Responsibilities, page 1 Secretary of the Army • 1–4, page 1 Headquarters Department of the Army (HQDA) • 1–5, page 1 The Commander • 1–6, page 2 The Command Security Manager • 1–7, page 2 The Supervisor • 1–8, page 3

\*This regulation supersedes AR 380–5, 25 February 1988; AR 380–150, 15 September 1982; AR 381–1, 12 February 1990; DA Pamphlet 380–1, 23 March 1990, and rescinds DA Form 2134, Security Violation(s) Report, January 1983.

AR 380-5 • 29 September 2000

The Individual • 1-9, page 3 Section III Program Management, page 4 Applicability definition • 1-10, page 4 General principles • 1-11, page 4 Legal authority  $\cdot$  1–12, page 4 Recordkeeping Requirements • 1-13, page 4 Section IV Program Direction, page 4 Background • 1-14, page 4 Information Security Oversight Office Responsibility • 1-15, page 5 Section V Special Types of Information, page 5 Atomic Energy Information (Restricted Data (RD)/Formerly Restricted Data (FRD)) • 1-16, page 5 Sensitive Compartmented Information, Communications Security Information, and Special Access Programs Information • 1–17, page 5 Section VI Exceptional Situations, page 5 Military Operations, Exercises, and Unit Deactivations • 1-18, page 5 Waivers and Exceptions to Policy • 1-19, page 5 Section VII Corrective Actions and Sanctions, page 6 General • 1-20, page 6 Sanctions • 1–21, page 6 Reporting of Incidents • 1-22, page 6 Section VIII Reports, page 7 Reporting Requirements • 1-23, page 7 Command security inspections • 1-24, page 7 Chapter 2 Classification, page 7 Section I Classification Principles, page 7 Original vs. derivative classification • 2-1, page 7 Policy • 2–2, page 8 Delegation of authority • 2-3, page 8 Required Training • 2-4, page 8 Section II Derivative Classification, page 8 Policy • 2-5, page 8 Accuracy responsibilities • 2-6, page 8 Section III The Original Classification Process, page 9 General • 2-7, page 9 Classification criteria • 2-8, page 9 Possibility of Protection • 2-9, page 9

Levels of classification • 2–10, page 10 Duration of classification • 2–11, page 10 Communicating the Classification Decision • 2–12, page 11 Compilation • 2–13, page 11 Acquisition Systems • 2–14, page 11 Limitations and prohibitions • 2–15, page 11

Section IV Security Classification Guides, page 12 Policy • 2–16, page 12 Content • 2–17, page 12 Approval, distribution, and indexing • 2–18, page 12 Review, revision, and cancellation • 2–19, page 12

Section V Non-Government Information, page 13 Policy • 2–20, page 13 Classification determination • 2–21, page 13 Classification challenges • 2–22, page 14

# Chapter 3 Declassification, Regrading, and Destruction, *page 15*

Section I Army Declassification Program, page 15 General • 3-1, page 15 Special Program Manager • 3-2, page 15 Declassification of Restricted Data and Formerly Restricted Data • 3-3, page 15 Declassification of other than Army information • 3-4, page 15

Section II The Automatic Declassification System, page 16 General • 3–5, page 16 Exemption from automatic declassification • 3–6, page 16 Marking of documents exempted from automatic declassification at 25 years • 3–7, page 17

Section III Mandatory Review for Declassification, page 17 General • 3–8, page 17 General • 3–9, page 17

Section IV Regrading, page 17 Downgrading information • 3–10, page 17 Downgrading policy • 3–11, page 17 Upgrading • 3–12, page 18

Section V Classified Material Destruction Standards, page 18 General • 3–13, page 18 Concepts of destruction • 3–14, page 18 Approved routine methods of destruction • 3–15, page 18 Appropriate material destruction techniques and methods for non paper–based material • 3–16, page 19 Technical advice on approved destruction devices and methods • 3–17, page 20 Clearing, purging, declassifying, and destroying media • 3–18, page 20

Chapter 4 Marking, page 21

#### Section I

Marking Documents, page 21 Purpose and policy • 4-1, page 21 Exceptions • 4-2, page 21 Requirements • 4-3, page 22 Overall classification marking • 4-4, page 22 Date, command, office of origin, and agency • 4-5, page 22 Page and portion marking • 4-6, page 22 Sources of classification - overview • 4-7, page 23 Sources of classification - procedures • 4-8, page 24 Reason for original classification • 4-9, page 24 Declassification instructions—"Declassify on" line • 4-10, page 25 Sources that were created prior to 1976 • 4-11, page 27 Warning notices • 4-12, page 27 Obsolete Restrictions and Control Markings • 4-13, page 29 Downgrading instructions • 4-14, page 29 The Modern Army Recordkeeping System • 4-15, page 29

# Section II

Marking Special Types of Documents, page 30 Documents with component parts • 4–16, page 30 Transmittal documents • 4–17, page 30 Classification by compilation • 4–18, page 30 Translations • 4–19, page 30 Electronically transmitted messages • 4–20, page 30 Documents marked for training purposes • 4–21, page 31 Files, folders, and groups of documents • 4–22, page 31 Printed documents produced by AIS equipment • 4–23, page 31

# Section III

Marking Special Types of Material, page 31 General policy • 4–24, page 31 Telephone or communications directories • 4–25, page 32 Blueprints, schematics, maps, and charts • 4–26, page 32 Photographs, negatives, and unprocessed film • 4–27, page 32 Slides and transparencies • 4–28, page 32 Motion picture films and videotapes • 4–29, page 32 Sound recordings • 4–30, page 32 Microfilms and microfiche • 4–31, page 32 Removable AIS storage media • 4–32, page 33 Fixed and internal AIS storage media • 4–33, page 33 Standard form (SF) labels • 4–34, page 33

# Section IV

Changes in Markings, page 34 Downgrading and declassification in accordance with markings • 4-35, page 34 Downgrading and declassification earlier than scheduled • 4-36, page 34 Upgrading • 4-37, page 34 Posted notice on bulk quantities of material • 4-38, page 34 Extensions of duration of classification • 4-39, page 34

Section V Remarking and Using Old Classified Material, page 35 Old markings • 4–40, page 35 Earlier declassification and extension of classification • 4–41, page 35

Section VI

Safeguarding Joint Chiefs of Staff Papers, page 35 General • 4-42, page 35 References • 4-43, page 35 Responsibilities • 4-44, page 35 Requirements • 4-45, page 35 Access • 4-46, page 35 Familiarization requirements • 4-47, page 35 Distribution of JCS documents • 4-48, page 36 Release and Distribution of Joint Strategic Planning System Documents • 4-49, page 36 Release and Distribution of Joint Operation Planning System Documents • 4-50, page 36 Release of JCS information to Army Service Schools • 4-51, page 36 Release of information to organizations outside DA • 4-52, page 37 Reproduction of JCS documents • 4-53, page 37

Section VII Foreign Government Information, page 37 Policy • 4–54, page 37 Equivalent U.S. classification designations • 4–55, page 37 Marking NATO documents • 4–56, page 37 Marking Other Foreign Government Documents • 4–57, page 37 Marking Foreign Government Information Provided in Confidence • 4–58, page 38 Marking of Foreign Government Information in Department of the Army Documents • 4–59, page 38

# Chapter 5

Controlled Unclassified Information, page 57

Section I For Official Use Only Information, page 57 General • 5–1, page 57 Description • 5–2, page 57 Marking • 5–3, page 58 Access to FOUO information • 5–4, page 58 Protection of FOUO information • 5–5, page 58 Further guidance • 5–6, page 58

Section II

Sensitive But Unclassified and Limited Official Use Information, page 59 Description • 5–7, page 59 Marking • 5–8, page 59 Access to SBU information • 5–9, page 59 Protection of SBU information • 5–10, page 59

Section III Drug Enforcement Administration Sensitive Information, page 59 Description • 5–11, page 59 Marking • 5–12, page 59 Access to DEA sensitive information • 5–13, page 59 Protection of DEA sensitive information • 5–14, page 60

Section IV DOD Unclassified Controlled Nuclear Information, page 60 Description • 5–15, page 60 Marking • 5–16, page 60 Access to DOD UCNI • 5–17, page 60 Protection of DOD UCNI • 5–18, page 61

Section V Sensitive Information (Computer Security Act of 1987), page 61 Description • 5–19, page 61 Marking • 5–20, page 61 Access to sensitive information • 5–21, page 61 Protection of sensitive information • 5–22, page 61 Further guidance • 5–23, page 61 Technical documents • 5–24, page 61

# Chapter 6

Access, Control, Safeguarding, and Visits, page 63

Section I Access, page 63 Responsibilities • 6–1, page 63 Nondisclosure Agreement • 6–2, page 63 Signing and filing the NDA • 6–3, page 63 Refusal to execute the NDA • 6–4, page 64 Debriefing and termination of classified access • 6–5, page 64 Communication and cooperation between command officials • 6–6, page 65 Access to restricted data, formerly restricted data, and critical nuclear weapons design information • 6–7, page 65 Access by persons outside the Executive Branch • 6–8, page 66

# Section II

Control Measures and Visits, page 67 Responsibilities for maintaining classified information. • 6–9, page 67 Care during working hours • 6–10, page 67 End-of-Day security checks • 6–11, page 67 Emergency planning • 6–12, page 68 Telephone conversations • 6–13, page 68 Speakerphone guidance • 6–14, page 68 Removal of Classified Storage and Information Processing Equipment • 6–15, page 69 Visits • 6–16, page 69 Classified visits by Department of Energy personnel and to DOE facilities • 6–17, page 70 Classified meetings and conferences • 6–18, page 71 Information processing equipment • 6–19, page 73 Receipt of classified material • 6–20, page 73

Section III Accountability and Administrative Procedures, page 73 TOP SECRET information • 6–21, page 73 SECRET and CONFIDENTIAL information • 6–22, page 74 NATO and Foreign Government material • 6–23, page 74 Working papers • 6–24, page 74

Section IV Reproduction of Classified Material, page 74 Policy • 6–25, page 74

Approval for reproduction • 6-26, page 75

Section V Disposition and Destruction of Classified Material, page 75 Policy • 6–27, page 75 Methods and standards for destruction • 6–28, page 76 Records of destruction • 6–29, page 77

Section VI Waivers, page 77 General • 6–30, page 77 Unique situation and compensatory measures • 6–31, page 77 Duration • 6–32, page 77 Documentation • 6–33, page 77 Prior waivers • 6–34, page 77

Section VII Inspections, page 77 Self Inspection • 6–35, page 77 Entry Exit Inspection Program and Two Person Integrity for TOP SECRET Information • 6–36, page 77

# Chapter 7

Storage and Physical Security Standards, page 78

Section I General, page 78 Policy • 7–1, page 78 Physical security policy • 7–2, page 78

Section II Storage Standards, page 78 Standards for storage equipment • 7–3, page 78 Storage of classified information • 7–4, page 78 Procurement of New Storage Equipment • 7–5, page 80 Residential storage • 7–6, page 80 Safeguarding of U.S. Classified Information Located in Foreign Countries • 7–7, page 81 Equipment Designations and Combinations • 7–8, page 81 Repair of Damaged Security Containers • 7–9, page 82 Maintenance and Operating Inspections • 7–10, page 83 Turn–in or Transfer of Security Equipment • 7–11, page 83

Section III Physical Security Standards, page 83 General • 7–12, page 83 Vault and Secure Room (Open Storage Area) Construction Standards • 7–13, page 83 Intrusion Detection System Standards • 7–14, page 84 Selection of equipment • 7–15, page 85 IDS Transmission • 7–16, page 85 System Requirements • 7–17, page 85 Installation, Maintenance and Monitoring • 7–18, page 86 Access Controls While Material is Not Secured in Security Containers • 7–19, page 86 Minimum standards for deviations to construction standards for open storage areas • 7–20, page 87

Section IV Lock Replacement Priorities, page 88

Priorities for Replacement of Locks • 7-21, page 88

Chapter 8 Transmission and Transportation, page 90

#### Section I

Methods of Transmission and Transportation, page 90 Policy • 8–1, page 90 TOP SECRET Information • 8–2, page 90 SECRET information • 8–3, page 90 CONFIDENTIAL information • 8–4, page 91 NATO restricted material • 8–5, page 92

Section II Transmission of Classified Material to Foreign Governments, page 92 General • 8–6, page 92 Procedures • 8–7, page 92 Shipment of freight • 8–8, page 94

# Section III

Preparation of Material for Transmission, page 94 Envelopes or containers • 8–9, page 94 Addressing • 8–10, page 95 Mail channels with the Department of Energy • 8–11, page 95

# Section IV Escort or Handcarrying of Classified Material, page 95 General provisions • 8–12, page 95 Documentation • 8–13, page 96 Security requirements for temporary duty travel outside the United States • 8–14, page 96 Handcarrying or escorting classified material aboard commercial passenger aircraft • 8–15, page 97 Consignor/consignee responsibility for shipment of bulky material • 8–16, page 98

# Chapter 9

Security Education, page 100

Section I Policy, page 100 General policy • 9–1, page 100 Methodology • 9–2, page 100

Section II Briefings, page 100 Initial orientation • 9–3, page 100 Cleared personnel • 9–4, page 100 Briefing upon refusal to sign the NDA, SF 312 • 9–5, page 101 Briefing uncleared personnel • 9–6, page 101 Refresher briefing • 9–7, page 101 Foreign travel briefing • 9–8, page 101

Section III Special Requirements, page 102 General policy • 9–9, page 102 Original classifiers • 9–10, page 102 Derivative classifiers • 9–11, page 103 Security Program Management personnel • 9–12, page 103

Critical Nuclear Weapons Design Information Briefing • 9–13, page 103 Others • 9–14, page 103

Section IV Termination Briefings, page 104 General Policy • 9–15, page 104

Section V Program Oversight, page 104 General policy • 9–16, page 104

# Chapter 10 Unauthorized Disclosure and Other Security Incidents, *page 104*

Section I

Policy, page 104 General policy • 10–1, page 104 Reaction to discovery of incident • 10–2, page 105 The preliminary inquiry • 10–3, page 105 Reporting results of the preliminary inquiry • 10–4, page 106 Reevaluation and damage assessment • 10–5, page 106 Debriefings in cases of unauthorized access • 10–6, page 107 Management and oversight • 10–7, page 108 Additional investigation • 10–8, page 108 Unauthorized absences, suicides, or incapacitation • 10–9, page 108 Negligence • 10–10, page 108

Section II

Extracts of Espionage Laws and Federal Statutes, page 108

United States Code, Title 18, Section 641 - Public Money, Property Or Records • 10-11, page 108

- United States Code, Title 18, Section 793 Gathering, Transmitting, Or Losing Defense Information 10-12, page 108
- United States Code Title 18, Section 794 Gathering Or Delivering Defense Information To Aid Foreign Government 10–13, page 109

United States Code, Title 18, Section 795 - Photographing And Sketching Defense Installations • 10-14, page 110

United States Code, Title 18, Section 796 – Use Of Aircraft For Photographs Of Defense Installations • 10–15, page 110

United States Code, Title 18, Section 797 – Publication And Sale Of Photographs Of Defense Installations • 10–16, page 110

United States Code, Title 18, Section 798 - Disclosure Of Classified Information • 10-17, page 110

United States Code, Title 50, Section 797 - Violate Regulations And Aiding And Abetting • 10-18, page 111

United States Code, Title 18, Section 952 - Diplomatic Codes And Correspondence • 10-19, page 111

United States Code, Title 18, Section 1001 - False And Fraudulent Statements • 10-20, page 111

United States Code, Title 18, Section 1924 – Unauthorized Removal And Retention Of Classified Documents Or Material • 10-21, page 111

United States Code, Title 50, Sections 783 (B) And (D) • 10-22, page 111

UNIFORM CODE OF MILITARY JUSTICE Article 106a ESPIONAGE • 10-23, page 111

# Appendixes

- A. References, page 116
- B. Presidential Executive Orders (EO) 12958, EO 12972 and EO 13142, page 121
- C. Special Procedures for Use in Systematic and Mandatory Review of Cryptologic Information, page 136
- D. Security Controls on Dissemination and Marking of Warning Notices on Intelligence Information, page 137
## **Contents**—Continued

- E. Security Procedures for Documents Created for and on Automated Information Systems and Internet Web-based Display, *page 151*
- F. Management Control, page 165
- G. Security Classification Guide Preparation, page 184
- A. CLASSIFICATION FACTORS, page 198
- B. CLASSIFYING DETAILS, page 202
- C. ITEMS OF INFORMATION, page 204
- D. Recommended Format For A Security Classification Guide, page 205
- E. FORMAT VARIATIONS, page 210
- H. Instructions Governing Use of Code Words, Nicknames, and Exercise Terms, page 211
- I. Special Access Programs (SAPs), page 214

#### Table List

- Table F-1: File Titles and Dispositions for Records, page 180
- Table F-2: File Numbers and Descriptions for Records, page 182
- Table C-1: Classification guidance, page 194
- Table C-2: HUMINT classification guide, page 196
- Table C-3: Classification of notes and transcripts, page 197
- Table C-4: Classification topics, page 198
- Table C-1: Strategic and Tactical Capabilities and Vulnerabilities, page 204
- Table D-1: Performance and capabilities topics, page 207
- Table D-2: Specification topics, page 208
- Table D-3: Administrative data topics, page 209
- Table D-4: Hardware classification, page 209
- Table E-1: Format variation topics, page 210

#### Figure List

- Figure 4-1: Sample of Marking an Originally Classified Document, page 39
- Figure 4–2: Sample of Marking a Classified Document that is Exempt from the 25–Year Automatic Declassification, page 40
- Figure 4-3: Sample of Marking an Originally and Derivatively Classified Document, page 41
- Figure 4-4: Sample of Marking a Document Derivatively Classified from Information in Old Document, page 42
- Figure 4–5: Sample of Marking a Document When each Portion is Unclassified but Together are classified by Compilation, page 43
- Figure 4-6: Sample of Marking a Document Derivatively Classified from One Source Classified under the Current System, page 44
- Figure 4-7: Sample of Marking a Document Derivatively Classified from Source Classified under the Old System and a Source Classified under the Current System, page 45
- Figure 4-8: Sample of a Document Derivatively Classified from Multiple Sources, page 46
- Figure 4–9: Sample of Marking a Document Where the Cover Memo is Unclassified but the Attachments are Classified, page 47
- Figure 4–10: Sample of Marking a Document Where the Cover Memo is Unclassified but the Attachments are Classified, page 48
- Figure 4-11: Sample of Marking Foreign Government Information, page 49
- Figure 4-12: Sample of Marking Working Papers, page 50
- Figure 4-13: Equivalent Foreign Security Classification, page 51
- Figure 4-13: Equivalent Foreign Security Classification-Continued, page 52
- Figure 4-13: Equivalent Foreign Security Classification-Continued, page 53
- Figure 4-13: Equivalent Foreign Security Classification-Continued, page 54
- Figure 4-13: Equivalent Foreign Security Classification-Continued, page 55

## **Contents**—Continued

Figure 4-13: Equivalent Foreign Security Classification-Continued, page 56

- Figure 5-1: Distribution Statements for Technical Documents, page 62
- Figure 7-1: Lock Replacement Priorities, page 89
- Figure 8-1: Federal Aviation Administration (FAA) Air Transportation Security Field Offices, page 99
- Figure 10-1: Sample Preliminary Inquiry Report, page 113
- Figure 10-1: Sample Preliminary Inquiry Report-Continued, page 114
- Figure 10-1: Sample Preliminary Inquiry Report-Continued, page 115
- Figure D-1: Director of Central Intelligence Directive (DCID) 1/7, Security Controls on the Dissemination of Intelligence Information, page 139
- Figure D-2: Sample of Marking an Originally Classified Intelligence Community Document, page 149
- Figure D-3: Sample of Marking Foreign Government Intelligence Information, page 150
- Figure E-1: Unclassified Warning Banner, page 153
- Figure E-2: Intelink Security Banner, page 154
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites, page 156
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 157
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 158
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 159
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 160
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 161
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 162
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 163
- Figure E-3: Guidance for Management of Publicly Accessible U.S. Army Websites-Continued, page 164
- Figure G-1: DOD Directive 5200.1-H, page 187
- Figure G-1: DOD Directive 5200.1-H—Continued, page 188
- Figure A-1: CLASSIFICATION FACTORS, page 199
- Figure A-1: CLASSIFICATION FACTORS-Continued, page 200
- Figure A-1: CLASSIFICATION FACTORS-Continued, page 201

#### Glossary

Index

RESERVED

## Chapter 1 General Provisions and Program Management

## Section I Introduction

## 1-1. Purpose

This regulation establishes the policy for the classification, downgrading, declassification, transmission, transportation, and safeguarding of information requiring protection in the interests of national security. It primarily pertains to classified national security information, now known as classified information, but also addresses controlled unclassified information, to include for official use only and sensitive but unclassified. For the purposes of this regulation, classified national security information, or classified information, is defined as information and/or material that has been determined, pursuant to EO 12958 or any predecessor order, to require protection against unauthorized disclosure and is marked to indicate its classified status when in documentary or readable form. This regulation implements Executive Order (EO) 12958 and Department of Defense Regulation 5200.1-R. This regulation contains the minimum Department of the Army (DA) standards for the protection of classified information and material. Such standards may be enhanced but never lessened at command option. This regulation also establishes the DA policy on the safeguarding of Restricted Data (RD) and Formerly Restricted Data (FRD) as specified by the Atomic Energy Act of 1954. This regulation also provides guidance on the proper handling of sensitive unclassified material. A restatement and interpretation of the policy concerning the protection of controlled unclassified information is included in this regulation as chapter 5. This regulation does not establish the special, additional policy for the safeguarding of special category information to include Sensitive Compartmented Information (SCI) or Communications Security (COMSEC). which can be found in AR 380-28 and AR 380-40 respectively. It does address the protection of information in an automated environment (app E) and Special Access Programs (SAPs) (app I).

## 1-2. References

Required and related publications and prescribed and referenced forms are listed in appendix A.

## 1–3. Explanation of abbreviations and terms

Abbreviations and special terms used in this regulation are explained in the glossary.

## Section II Responsibilities

#### 1-4. Secretary of the Army

The Secretary of the Army (SECARMY) will-

*a.* Appoint a senior agency official to be responsible for direction and administration of the program within the Army. The SECARMY may designate a separate senior official to be responsible for overseeing SAPs within the Army, if necessary.

b. Commit necessary resources to the effective implementation of the information security program.

c. Establish procedures to ensure that the head of each Major Command (MACOM) that creates, handles or stores classified and sensitive information, appoints an official to serve as security manager for the command, who will provide management and oversight of the command's information security program.

#### 1-5. Headquarters Department of the Army (HQDA)

*a.* The Deputy Chief of Staff for Intelligence (DCSINT), Headquarters, DA (HQDA), is designated as the DA Senior Official of the Intelligence Community (SOIC), to direct, administer, and oversee the Army's information security program. The Chief of the Counterintelligence/Human Intelligence Division (DAMI–CH), Intelligence Policy Directorate, provides staff support for these functions. The DCSINT will—

(1) Promulgate (or cause to promulgate) policy, procedures, and programs necessary for the implementation of EO 12958 and resulting national and Department of Defense (DOD) Directives (DODD).

(2) Monitor, evaluate, and report on the administration of the Army's information security program. Ensure MACOM, Major Subordinate Commands (MSC) and other agencies, establish and maintain an ongoing self-inspection program, to include periodic reviews and assessments of their classified and sensitive products.

(3) Respond to information security matters pertaining to classified and sensitive information that originated in an Army command that no longer exists and for which there is no successor in function.

(4) Delegate SECRET and CONFIDENTIAL Original Classification Authority (OCA) to other Army officials. The SECARMY is the only Army official that may delegate TOP SECRET original classification authority.

(5) Commit the necessary resources for the effective policy development and oversight of the programs established by this regulation.

b. The Chief, Technology Management Office (TMO), HQDA, is the Army primary contact for the management and oversight of Army and Army-supported SAPs. The Chief, TMO will implement SAPs information security measures and all associated policies necessary to execute the SAPs security policy that is established and directed by the HQDA, DCSINT. See appendix I of this regulation and AR 380–381 for further guidance on SAPs.

c. The Deputy Chief of Staff for Personnel (DCSPER), HQDA, will execute the provisions of section 5.6(c)(7) of EO 12958, and establish policy to ensure that the systems used to evaluate or rate civilian and military personnel performance, include the management of classified, and when possible sensitive, information as a critical element/item/ objective to be evaluated in the rating of—

(1) Original classification authorities.

(2) Security managers and security specialists.

(3) All other personnel whose duties significantly involve the creation or handling of classified and sensitive information.

*d*. The Comptroller of the Army, HQDA, will execute the provisions of section 5.6(c)(8) of EO 12958, establish and implement a system for accounting for the costs associated with the Army implementation of EO 12958, and for reporting those costs to the Director of the Information Security Oversight Office (ISOO) for publication.

#### 1-6. The Commander

Security is a command function. Commanders, Officers in Charge (OIC), and heads of agencies and activities (referred to as commanders), will effectively manage the information security program within their commands, agencies, activities, or areas of responsibility (referred to as commands). Commanders may delegate the authority to execute the requirements of this regulation, where applicable, but not the responsibility to do so. Security, including the safeguard-ing of classified and sensitive information and the appropriate classification and declassification of information created by command personnel, is the responsibility of the commander. The commander will—

*a*. Establish written local information security policies and procedures and an effective information security education program.

b. Initiate and supervise measures or instructions necessary to ensure continual control of classified and sensitive information and materials.

c. Ensure that persons requiring access to classified information are properly cleared.

d. Continually assess the individual trustworthiness of personnel who possess a security clearance.

*e.* Designate a Command Security Manager (CSM) by written appointment. The CSM will be of sufficient rank or grade to effectively discharge assigned duties and responsibilities. As a general requirement, the CSM will be a commissioned officer (0–3 or above), warrant officer, or civilian in the grade of GS–12 or above. The MSC commander may, subject to MACOM or Administrative Assistant to the Secretary of the Army (HQDA SAAA) for the HQDA staff for policy approval, designate a CSM at a lower rank or grade in situations in which the rank or grade of the individual selected is sufficient to effectively discharge assigned responsibilities. The CSM will have direct access to the commander on matters affecting the information security program.

f. Ensure the CSM is afforded security training consistent to the duties assigned.

g. Ensure adequate funding and personnel are available to allow security management personnel to manage and administer applicable information security program requirements.

h. Review and inspect the effectiveness of the information security program in MSCs.

*i.* Ensure prompt and appropriate responses are given, or forward for higher echelon decision, any problems, suggestions, requests, appeals, challenges, or complaints arising out of the implementation of this regulation.

*j*. Ensure the prompt and complete reporting of security incidents, violations, and compromises, related to classified and sensitive information, as directed herein.

k. Ensure prompt reporting of credible derogatory information on assigned/attached personnel, to include recommendations for or against continued access.

#### 1–7. The Command Security Manager

The command security manager is the principal advisor on information security in the command and is responsible to the commander for management of the program. The CSM will—

a. Advise and represent the commander on matters related to the classification, downgrading, declassification, and safeguarding of national security information.

b. Establish and implement an effective security education program as required by chapter 9 of this regulation.

c. Establish procedures for assuring that all persons handling classified material are properly cleared. The clearance status of each individual must be recorded and accessible for verification.

d. Advise and assist officials on classification problems and the development of classification guidance.

e. Ensure that classification guides for classified plans, programs, and projects are properly prepared, distributed, and maintained.

f. Conduct a periodic review of classifications, assigned within the activity, to ensure that classification decisions are proper.

g. Consistent with operational and statutory requirements, review all classified and sensitive documents in coordination with the Command Records Management Officer. Continually reduce, by declassification, destruction, or retirement, unneeded classified and sensitive material.

h. Submit Standard Form (SF) 311 (Agency Information Security Program Data) to DAMI-CH annually, as required.

*i*. Supervise or conduct security inspections and spot checks and notify the commander regarding the compliance with this regulation and other security regulations and directives.

*j*. Assist and advise the commander in matters pertaining to the enforcement of regulations governing the access, dissemination, reproduction, transmission, transportation, safeguarding, and destruction of classified and sensitive material.

k. Make recommendations, based on applicable regulations and directives, on requests for visits by foreign nationals, and provide security and disclosure guidance if the visit request is approved.

*l*. Ensure the inquiry and reporting of security violations is completed, including compromises or other threats to the safeguarding of classified and sensitive information, in accordance with chapter 10 of this regulation. Recommend to the decision official whether or not administrative sanction is warranted, and/or indicate corrective action that should be taken concerning security violations.

*m*. Ensure proposed public releases on classified and sensitive programs be reviewed to preclude the release of classified information or other sensitive unclassified information covered under the Freedom of Information Act (FOIA).

*n*. Establish and maintain visit control procedures in cases in which visitors are authorized access to classified information.

o. Issue contingency plans for the emergency destruction of classified and sensitive information and material and, where necessary, for the safeguarding of classified and sensitive information and material used in or near hostile or potentially hostile areas.

p. Be the single point of contact to coordinate and resolve classification or declassification problems.

q. Report data as required by this regulation.

#### 1-8. The Supervisor

Supervisory personnel (to include those in command positions) have a key role in the effective implementation of the command's information security program. Supervisors, by example, words, and deeds, set the tone for compliance by subordinate personnel with the requirements to properly safeguard, classify, and declassify, information related to national security. The supervisor will—

a. Ensure subordinate personnel who require access to classified information are properly cleared and are given access only to that information, to include sensitive information, for which they have a need-to-know.

b. Ensure subordinate personnel are trained in, understand, and follow, the requirements of this regulation, and local command policy and procedures, concerning the information security program.

c. Continually assess the eligibility for access to classified and sensitive information of subordinate personnel and report to the CSM any information that may have a bearing on that eligibility.

d. Supervise personnel in the execution of procedures necessary to allow the continuous safeguarding and control of classified and sensitive information.

*e*. Include the management of classified and sensitive information as a critical element/item/objective in personnel performance evaluations, where deemed appropriate, in accordance with Army personnel policy and paragraph 1–5c of this regulation. Supervisors should include the protection of classified and sensitive information as a performance evaluation factor or objective for other personnel as the supervisor deems appropriate.

f. Lead by example. Follow command and Army policy and procedures to properly protect classified and sensitive information and to appropriately classify and declassify information as stated in this regulation.

#### 1–9. The Individual

All DA personnel, regardless of rank, grade, title, or position, have a personal, individual, and official, responsibility to safeguard information, related to national security, that they have access to. All DA personnel will report, to the proper authority, the violations by others that could lead to the unauthorized disclosure of classified and sensitive information. This responsibility cannot be waived, delegated, or in any other respect, excused. All DA personnel will safeguard all information and material, related to national security, especially classified information, which they access, and will follow the requirements of this and other applicable regulations.

## Section III Program Management

#### 1–10. Applicability definition

This regulation governs the Department of the Army information security program and applies to all DA personnel to include military and civilian members of the Active Army, Army National Guard (ARNG), and Army Reserve (USAR). Information relating to national security will be protected by DA personnel and employees against unauthorized disclosure. For the purposes of this regulation, DA personnel includes any active or reserve military personnel or National Guard, assigned or attached to a Department of the Army installation or activity, and persons employed by, assigned to, or acting for, an activity within the Department of the Army, including contractors, licensees, certificate holders, and grantees, and persons otherwise acting at the direction of such an activity.

## 1–11. General principles

a. All DA personnel, regardless of rank, title, or position, have a personal, individual, and official responsibility for the proper safeguarding and protection of the information they have access to, in particular, classified information.

*b.* Information will be classified, or protected as sensitive, only when it is in the interest of national security, and downgraded or declassified when it is determined that the information requires, in the interest of national security, a lower degree of protection against unauthorized disclosure than is currently required. Information or material, that requires protection against unauthorized disclosure, in the interest of national security, shall be classified as one of the three following categories or levels, as defined in this regulation:

(1) TOP SECRET.

- (2) SECRET.
- (3) CONFIDENTIAL.

c. Information and material, classified under this regulation, will be afforded the level of protection, against unauthorized disclosure, commensurate with the level of classification or sensitivity assigned, under the varying conditions that may arise in connection with its use, dissemination, storage, movement, transmission, or destruction. Responsible officials will ensure that classified and sensitive information and materials are adequately protected from compromise. Everyone must be continually aware of possible threats from all–source intelligence efforts of potential adversaries.

d. Access to classified information is authorized only to the following personnel:

(1) Persons with the appropriate need-to-know for the information in order to perform a lawful and authorized governmental function;

(2) Persons who have been granted a security clearance and access authorization at the appropriate level of clearance.

(3) Persons who have executed an appropriate non-disclosure agreement. AR 380-67 contains policy on the personnel security clearance program. Note: The holder, not the potential receiver, of the information, determines the need-to-know and is responsible for verifying the clearance and access authorization of the potential receiver. No person will be granted access to classified information solely by virtue of rank, title, or position.

e. Classified and sensitive information will be maintained only when necessary for the operation of the organization or when its retention is required by law, regulation, or records management policy.

## 1–12. Legal authority

The statutory authority for this regulation is derived from Title 18, of the United States Code (USC), the Atomic Energy Act of 1954, as amended, Executive Orders, issuances from the Office of Management and Budget (OMB), and the Security Policy Board (SPB).

## 1–13. Recordkeeping Requirements

This regulation requires the creation, maintenance, and disposition, of records, to document and support the business processes of the Army. Recordkeeping requirements are found in section II of appendix F, of this regulation, and AR 25–400–2.

## Section IV Program Direction

## 1–14. Background

Within the Federal Government, and the Office of Management and Budget, are a number of offices designed to oversee and implement EO 12958. These offices issue directives, as necessary and these directives are binding to all the components. Directives issued by these offices establish standards for—

- a. Classification and marking principles.
- b. Agency security education and training programs.

- c. Agency self-inspection programs.
- d. Classification and declassification guides.

## 1–15. Information Security Oversight Office Responsibility

The Director of the Information Security Oversight Office (ISOO) is delegated the responsibility for the implementation and monitorship functions of these programs. The details on the make–up and responsibilities of these offices are contained in EO 12958, a reprint of which can be found in appendix B, of this regulation.

## Section V

## **Special Types of Information**

## 1-16. Atomic Energy Information (Restricted Data (RD)/Formerly Restricted Data (FRD))

The primary purpose of this regulation is to implement EO 12958 and its implementing Department of Defense directives. EO 12958 does not apply to information classified as Restricted Data (RD) or Formerly Restricted Data (FRD). Nothing in EO 12958 supersedes any requirement made by, or under, the Atomic Energy Act of 1954, as amended. Of particular importance is that neither RD nor FRD information is subject to the automatic declassification provision of EO 12958, as specified in chapter 3 of this regulation. RD and FRD information will not be declassified without the specific permission of the Department of Energy (DOE). RD and FRD shall be safeguarded, classified, downgraded, and declassified, per the provisions of the Atomic Energy Act of 1954, as amended, and by DOD and Army policy. The Army policy on the marking and safeguarding, of RD/FRD information, is contained in chapters 4 and 6, respectively, of this regulation. RD and FRD information will be safeguarded, sa required by this regulation, for other information of a comparable level of security classification. The policy on the classification, downgrading, and declassification, of RD and FRD information, is stated in classification and declassification guidance promulgated by the DOE, or in guidance issued jointly by the DOD and DOE.

# 1–17. Sensitive Compartmented Information, Communications Security Information, and Special Access Programs Information

Security classification and declassification policies apply to SCI, COMSEC, and SAPs information in the same manner as other classified information (see app C for guidance on declassification of cryptologic information and appendix I, of this regulation, for more information on SAPs). SCI, COMSEC, and SAPs information will be controlled and safeguarded in accordance with AR 380–28, AR 380–40, and AR 380–381, respectively.

## Section VI Exceptional Situations

## 1–18. Military Operations, Exercises, and Unit Deactivations

a. Military Operations. The provisions of this regulation may be modified, but not lessened, by the commander or senior official, as necessary to meet local conditions relating to combat, combat operations, emergency conditions under operations other than war, to include peacekeeping operations, and any other emergency situation where that operation or situation requires exceptional measures to protect life or Department of the Army assets. The criteria for classification and sensitivity of national security information remains for all situations; however, nothing in this regulation prohibits commanders from protecting any other information they deem necessary, to carry out the military operations and emergency situations identified above. Classified and sensitive information may be introduced into combat areas or zones or areas of potential hostile activity, but only as necessary to accomplish the military mission.

b. Military Exercises. Military exercises pose a unique situation where the handling and protection of classified and sensitive information are concerned. During exercises troops are told to "train as you would fight." When material, used in exercises, are "Classified For Training Only," it will be handled as if it were real world classified and/or sensitive. This is good security practice and can help prevent possible security violations in the future. When real–world classified and/or sensitive material is introduced and/or used in military exercises, every effort will be used to prevent compromise and/or loss. While there may be circumstances and situations where inadvertent disclosure of classified and/or sensitive information may occur, it is up to the command security manager, and ultimately the commander, to ensure that the provisions of chapter 10, of this regulation, are followed.

c. Unit Deactivation. Original classification authority is assigned to a duty position not to an individual person. When an organization has been deactivated, the OCA's responsibilities will revert to their higher headquarters or that organization assuming responsibility over the out–going command's security decisions. Challenges to classification decisions, of the deactivated organization, will be directed to that headquarters with the security responsibilities of the outgoing unit.

## 1–19. Waivers and Exceptions to Policy

*a*. This regulation is based on national policy that is applicable to all U.S. Government departments, agencies, and DA personnel. In order to ensure the protection of information related to national security, and allow all agencies to

have confidence in the sharing of information with other agencies, the national, DOD, and DA policy, contained in this regulation, will be followed.

*b.* Unless otherwise noted, requests for waivers to the requirements contained in this regulation, will be submitted, through command channels, to DAMI–CH. Waivers to DOD requirements will be forwarded by DAMI–CH, for decision to the Assistant Secretary of Defense for Command, Control, Communications, and Intelligence (ASD(C3I)). For requirements related to Two–Person Integrity (TPI), RD, Foreign Government Information (FGI) (including North Atlantic Treaty Organization (NATO)), and security arrangements for international programs, waivers will be forwarded to the Under Secretary of Defense (Policy)(USD(P)). Waivers for SAPs will be submitted, through SAPs channels, to DAMI–CH for coordination with TMO and, as required, forwarded to the Under Secretary of Defense (Special Programs) (USD(SP)). The ASD(C3I) and USD(P) are responsible for notifying the Director of the ISOO of the waivers approved that involve EO 12958 and its implementing directives.

c. Before submitting a request for waiver, the requesting authority will consider risk management factors such as criticality, sensitivity, and value of the information, analysis of the threats both known and anticipated, vulnerability to exploitation, and countermeasure benefits versus cost (national security cost and resource cost). Requests for waiver must contain sufficient information to permit a complete and thorough analysis to be made of the impact on national security if the waiver is approved. The waiver request will also describe all the factors creating the special situation and the alternative or compensatory measures which make sure the protection afforded the information is sufficient to reasonably deter and detect loss or unauthorized disclosure. The requesting command will maintain documentation regarding approved waivers, including the alternative or compensatory measures approved and in use, and furnish this documentation, upon request, to other agencies and to other Army commands, with whom classified information or secure facilities are shared.

Note: Waivers granted before the effective date of this regulation are canceled no later than one year after the effective date of this regulation. New/updated waiver requests may be submitted prior to cancellation date.

*d.* Throughout this regulation there are references to policy subject to MACOM approval or subject to policy as the MACOM directs. Where that language, in substance, is used, the MACOM commander, or the HQDA SAAA, for cases involving HQDA and its Field Operating Agencies (FOA), can delegate such approval authority. The delegations will be in writing. A copy of such delegations will be maintained by the appointing official and reviewed periodically for review of need for continuation. Where this regulation specifically specifies waiver authority to a MACOM commander or the HQDA SAAA, that authority resides solely with the MACOM commander or HQDA SAAA and will not be further delegated.

## Section VII

#### **Corrective Actions and Sanctions**

#### 1-20. General

Commanders will establish procedures to make sure that prompt and appropriate action is taken concerning a violation of the provisions of this regulation, especially in those cases involving incidents which can put classified information at risk of compromise, unauthorized disclosure, or improper classification of information. Such actions will focus on a correction or elimination of the conditions that caused or contributed to the incident.

#### 1-21. Sanctions

- a. DA personnel will be subject to sanctions if they knowingly, willfully, or negligently-
- (1) Disclose classified or sensitive information to unauthorized persons.
- (2) Classify or continue the classification of information in violation of this regulation.
- (3) Violate any other provision of this regulation.

b. Sanctions can include, but are not limited to warning, reprimand, suspension without pay, forfeiture of pay, removal, discharge, loss or denial of access to classified information, and removal of original classification authority. Action can also be taken under the Uniform Code of Military Justice (UCMJ) for violations of that Code and under applicable criminal law, if warranted.

c. Original classification authority will be withdrawn for individuals who demonstrate a disregard or pattern of error in applying the classification and sensitivity standards of this regulation.

#### 1-22. Reporting of Incidents

EO 12958, paragraph 5.7(e)(2), requires that the director of the ISOO be advised of instances in which classified information is knowingly, willfully, or negligently disclosed to unauthorized persons, or instances of classifying, or continuing the classification of, information in violation of this regulation. Reports of those instances will be submitted through command channels to DAMI–CH for forwarding to the director of the ISOO and other defense officials as appropriate. See chapter 10 for reporting of other security incidents.

## Section VIII Reports

## 1-23. Reporting Requirements

HQDA is required to report data necessary to support various requirements of EO 12958. Commanders will respond to those data calls when so notified. MACOMs and the HQDA SAAA will also submit a consolidated annual report, for all units under their security responsibility, on SF 311, to reach DAMI–CH no later than 1 October, or other date specified by DAMI–CH, each fiscal year. The report will cover the preceding fiscal year. DAMI–CH will consolidate and submit the annual SF 311 report for the Army. Interagency Report Control Number 0230–GSA–AN applies to this report.

## 1-24. Command security inspections

MACOM, agency, and MSC commanders will establish and maintain a self-inspection program for their command, and a program to inspect their subordinate units. The program must be based upon program needs and the degree of involvement with classified and sensitive information. The purpose of the program will be to evaluate and assess the effectiveness of the command's protection of classified and sensitive information and adherence to Army policy contained in this regulation. Inspections will be conducted annually unless the command's higher headquarters determines that the quantity of classified and sensitive holdings and material generated does not warrant that frequency. In those cases, inspections will occur not less frequently than once every other year. This will not dismiss other annual requirements outlined in this regulation.

## Chapter 2 Classification

## Section I Classification Principles

#### 2-1. Original vs. derivative classification

*a.* Original classification is the decision to designate a certain item of information as classified, at a particular level, and for a certain duration of time. Often these decisions are communicated in a published Security Classification Guide (SCG). These decisions can only be made by persons designated in writing by either the SECARMY or the DCSINT as Original Classification Authorities (OCA). There are relatively few officials in the Army that have the authority to apply original classification, and relatively few instances of original classification, in most Army commands. Derivative classification is the incorporating, restating, paraphrasing, or generating in new form, information that has already been determined to be classified, and ensuring that it is classified and handled at the level that the OCA has already determined will be done. Derivative classification can be accomplished by any properly cleared personnel. Derivative classifiers are not required to be appointed or designated unless so directed by Command option. Most DA personnel that classify information do so in a derivative manner from some other document or source. Derivative classification is most commonly accomplished by marking classified material based on the guidance from an SCG or from the source document. The derivative classifier must have enough subject matter knowledge to properly interpret and apply the instruction of the classification guidance. The original classification authority decides what portion(s) of a plan, program, or project needs to be classified. The derivative classifier applies that decision to the same type of information restated or generated in a new form.

*b.* For example, an OCA could make the decision that the maximum effective range of Missile XYZ is classified. The classification authority issues a security classification guide that states that the maximum effective range of the missile will be classified at the SECRET level. When the missile is tested and the results are documented, the person who writes the report, states that the maximum effective range of Missile XYZ is 250 miles, derivatively classifying that item of information as SECRET. In this case, the classification is derived from the security classification guide. Most classification in the Department of the Army is done in a derivative manner. Those DA officials authorized to apply original classification decisions are relatively few in number.

## 2-2. Policy

Original classification is the initial determination by an OCA that an item of information could be expected to cause damage to national security if subjected to unauthorized disclosure. Damage to the national security means harm to the national defense or foreign relations of the United States from the unauthorized disclosure of the information, to include the sensitivity, value, and utility of that information. It includes military operations in support of national objectives when those operations involve information that meets the criteria of classification. This decision will be made only by persons specifically authorized in writing to do so, have received training in the exercise of this authority, and have program or program support responsibility or cognizance over the information. The decision to

originally classify must be made based on the requirements of this regulation. Delegations of original classification authority will be limited to the minimum required and only to officials who have a demonstrable and continuing need to exercise it.

## 2-3. Delegation of authority

*a.* The Secretary of the Army has been granted original classification authority by the President of the United States. TOP SECRET OCA can be delegated only by the SECARMY. SECRET and CONFIDENTIAL original classification authority can only be delegated by the DCSINT or by the SECARMY. Delegation of authority includes information at that level and any lower level(s) of classification. This authority cannot be redelegated.

*b*. Requests for OCA will be submitted, through command channels, to DAMI–CH. These requests will specify the position title for which the authority is requested and detailed justification for the request. Original classification authority is assigned to a position title and not to an individual person. In order to ensure that the number of OCAs is strictly limited, the request must address why another OCA, within that official's command or area, cannot assume this responsibility.

c. Requests for original classification authority will be granted only when:

(1) Original classification is required during the normal course of operations;

(2) Sufficient expertise and information is available to the prospective original classification authority to allow effective classification decision making;

(3) The need for original classification cannot be handled by other existing OCAs; or

(4) Referral of decisions to existing original classification authorities, at the command or at higher levels in the chain of command, is not practical.

## 2-4. Required Training

Officials who have been delegated original classification authority will receive training, as required by chapter 9 of this regulation, before exercising this authority.

## Section II Derivative Classification

## 2–5. Policy

DA personnel who generate material which is to be derivatively classified are responsible for making sure that the classification is properly applied based on the original source material marking and local security classification guides. DA personnel who apply derivative classification should take care to determine whether their paraphrasing, restating, or summarizing of classified information has removed all or part of the basis for classification. Certain information that would otherwise be unclassified, may require classification when combined or associated with other unclassified information. This is referred to as classified by compilation. However, a compilation of unclassified items of information of unclassified items of information provides an added factor that warrants classification. Similarly, a higher classification may be assigned to compilations of information that warrants higher classification than that of its component parts. Classification on this basis shall be fully supported, in writing, accompanying the compilation document. See paragraph 2–8 for specific classifying criteria.

## 2-6. Accuracy responsibilities

Officials who sign or approve derivatively classified material are responsible for the accuracy of the derivative classification. This applies to all forms of material and information regardless of the media involved. Personnel accomplishing derivative classification will—

a. Observe and respect the classification determinations made by original classification authorities.

*b*. Apply markings or other means of identification to the derivatively classified material, as required by this regulation, at the level and for the duration specified by the classification guide or source document. Where classification instructions do not reflect the new marking requirements of EO 12958, mark the level of classification as directed by the classification guide or source document and follow this regulation for all other marking requirements. Derivative classifiers are encouraged to keep informal records of which portions of a draft document are classified and by which source to make the classification of the finished product easier.

c. Use only authorized sources such as classification guides, other forms of official classification guidance, and markings on source material, from which the information is extracted. Refrain from guesswork.

*d*. Use caution when paraphrasing or restating information extracted from a classified source to determine whether the classification could have been changed in the process.

*e*. Take appropriate and reasonable steps to resolve doubt or conflicts in classification. In cases of apparent conflict between an SCG and a classified source document, concerning a discrete item of information, the instructions in the SCG will take precedence unless the source document is signed by the original classification authority. In such cases,

the OCA, or the point of contact for answering questions on classification, will be consulted. In the event that it is not possible to consult the OCA, the more restrictive classification instruction will be followed.

*f*. Make a list of sources used when material is derivatively classified based on "Multiple Sources" (more than one SCG, classified source document, or any combination). A copy of this list will be included in, or attached to, the file or record copy of the material. Derivative classifiers are encouraged to include this listing with all copies of the document, to make later declassification review easier if the file or record copy is unavailable.

g. Contact the classifier of the source document for resolution in cases in which the derivative classifier believes the classification applied to the information is not accurate.

#### Section III

#### **The Original Classification Process**

## 2–7. General

The decision to apply original classification requires the application of judgment, on the part of the classifier, that the unauthorized disclosure of the information could reasonably be expected to cause damage to the national security, and that the probable damage can be identified or described. It is not necessary for the original classifier to produce a written description of the damage at the time of classification, but the classifier must be prepared to do so if the information becomes the subject of a classification challenge, a request for mandatory review for declassification, or a request for release under the Freedom of Information Act. The decision to classify also has operational and resource impacts as well as impacts affecting the United States technological base and foreign relations. The decision to classify should consider all relevant factors. If there is doubt about classification, the OCA will research the matter to make an informed decision. If, after such research, there is a significant doubt about the need to classify information, it will not be classified. In making a decision to originally classify an item of information, an original classification authority will—

a. Determine that the information has not already been classified.

b. Determine that the information is eligible for classification pursuant to paragraph 2–8 of this regulation.

c. Determine that classification of the information is a realistic course of action and that the information can be protected from unauthorized disclosure when classified.

d. Decide that unauthorized disclosure could reasonably be expected to cause damage to the national security and that this disclosure is identifiable and can be described.

e. Select the appropriate level or category of classification and/or sensitivity to be applied to the information, based on a judgement as to the degree of damage unauthorized disclosure could cause.

f. Determine and include the appropriate declassification, downgrading, and/or exemption category instruction(s) to be applied to the information, when applicable.

g. Make sure that the classification decision is properly communicated so that the information will receive appropriate protection. Security classification guides will be used in this regard where appropriate (see paragraph 2-16).

#### 2–8. Classification criteria

U.S. classification can only be applied to information that is owned by, produced by or for, or is under the control of, the United States Government. This is determined by the original classification authority that the unauthorized disclosure of the information reasonably could be expected to result in damage to the national security, and the information falls within one or more of the following categories specified in section 1.5 of EO 12958:

- a. Military plans, weapons systems, or operations.
- b. Foreign government information.
- c. Intelligence activities (including special activities), intelligence sources or methods, or cryptology.
- d. Foreign relations or foreign activities of the United States, including confidential sources.
- e. Scientific, technological, or economic matters relating to the national security.
- f. United States Government programs for safeguarding nuclear materials or facilities.

g. Vulnerabilities or capabilities of systems, installations, projects or plans relating to the national security. Note: When used, these seven classification categories are referred to by their reference letter, preceded by "1.5," the reference location within the EO. For example, "Military plans, weapons systems, or operations" would be "1.5(a)." See paragraph 4–9 for further details on "Classified by" marking.

## 2-9. Possibility of Protection

a. The OCA must determine that, if classification is applied or reapplied, there is a reasonable possibility that the information will be provided protection from unauthorized disclosure.

b. The reclassification of information which was once classified but was declassified and officially released to the public, by an authorized Army official, and had wide-spread access by the public, is prohibited. Army information that has not previously been disclosed to the public, under proper Army authority, can be classified or reclassified. This includes after a command has received a request for it. However, only if the reclassification is accomplished on a

document-by-document basis, with the participation, or under the direction of, the SECARMY, the Under Secretary of the Army, or the DCSINT. Guidance from DAMI-CH will be requested in those instances. The information that is reclassified must meet the criteria for classified information established in EO 12958 or successor orders and directives. In considering issues of reclassification or classification of previously unclassified information, the OCA will—

(1) Determine that control of the information has not been lost and can still be prevented from being lost; and

(2) In the case of information released to secondary distribution centers, determine that no secondary distribution has been made and can still be prevented.

c. Classified information will not be declassified automatically as a result of any unauthorized disclosure of identical or similar information. In these cases, the OCA will review the situation to determine if continued classification is warranted. However, such disclosures require immediate determination of the degree of damage to the national security and reevaluation of the information to determine whether the publication has so compromised the information that downgrading or declassification is warranted.

#### 2–10. Levels of classification

*a.* Once a decision is made to classify, information will be classified at one of the three levels listed below. For each level, the OCA must be able to identify or describe the damage that unauthorized disclosure reasonably could be expected to cause to the national security. These levels are:

(1) TOP SECRET – Will be applied to information in which the unauthorized disclosure could reasonably be expected to cause exceptionally grave damage to the national security.

(2) SECRET – Will be applied to information in which the unauthorized disclosure could reasonably be expected to cause serious damage to the national security.

(3) CONFIDENTIAL – Will be applied to information in which the unauthorized disclosure could reasonably be expected to cause damage to the national security.

*b*. If there is doubt about the classification level, the OCA will research the matter to make an informed decision. If significant doubt still remains about the classification level to be assigned, the lower level will be assigned.

#### 2–11. Duration of classification

Information will be declassified as soon as it no longer meets the standards for classification. Information will remain classified as long as it is in the interest of national security and meets the criteria stated in this regulation. At the time an item of information is originally classified, the original classifier must decide the length of time the information will require classification and select an appropriate declassification date or event. The term "time or event phased declassification date," used for acquisition programs, is also synonymous with the term "declassification date" as used in this regulation. The declassification date indicates when the information no longer requires protection in the interests of national security. When deciding on the declassification date or event, the following options are the only ones available to the OCA:

*a.* At the time of original classification, the original classification authority will attempt to establish a specific date or event for declassification based upon the duration of the national security sensitivity of the information. The OCA will attempt to determine a date, within ten years from the date of classification, upon which the information can be automatically declassified. If that is not possible, they will attempt to determine a specific event, reasonably expected to occur within 10 years, that can be set as the signal for automatic declassification of the information. This is referred to as the "ten–year rule." The date or event will not exceed the time frame in subparagraph c, below.

*b*. If information has originally been assigned a date or event for declassification of ten years or less, in accordance with subparagraph a above, and the OCA later has reason to believe longer protection is required, the classification can be extended for successive periods of up to ten years at a time, not to exceed the time period in subparagraph e, below, where applicable.

c. If unable to determine a date or event that is ten years or less, the OCA will assign an exemption designation to the information, if the information qualifies for exemption from automatic declassification in ten years. This could be done if the unauthorized disclosure of the information could reasonably be expected to cause damage to the national security, if specific information requires a period beyond 10 years from the date of original classification, and the release of the information could reasonably be expected to result in one or more of the following:

(1) Reveal an intelligence source, method, or activity, or a cryptologic system or activity.

(2) Reveal information that could assist in the development or use of weapons of mass destruction.

(3) Reveal information that could impair the development or use of technology within a United States weapon system.

(4) Reveal United States military plans or national security emergency preparedness plans.

(5) Reveal foreign government information.

(6) Damage relations between the United States and a foreign government, reveal a confidential source, or seriously undermine diplomatic activities that are reasonably expected to be ongoing for a period greater than ten years.

(7) Impair the ability of responsible United States government officials to protect the President, the Vice President, and other individuals for whom protection services, in the interest of national security, are authorized.

(8) Violate a statute, treaty, or international agreement. Note: When used, these eight exemption categories are either completely written out or referred to by their reference number preceded by the letter "X." For example, "Violate a statute, treaty, or international agreement" or "X8." See paragraph 4–10 for further details on exemption marking.

*d.* Information marked for an indefinite duration of classification under prior orders, for example, "Originating Agency's Determination Required" (OADR), or information classified under prior orders that contain no declassification instructions, will be declassified in accordance with chapter 3 of this regulation. The term OADR will no longer be used. When an exemption category is selected, there is no requirement to select a specific date or event for declassification at the time of original classification. In those cases in which the original classifier does not select a declassification date, the following will apply:

(1) The information, if placed in records that have been determined to have permanent historical value under Title 44, USC (see "permanent" files under AR 25–400–2), will be automatically declassified in 25 years from the date of original classification, unless specifically exempted or unless this policy is changed before that time.

(2) The information, if not placed in such records (mentioned in subparagraph (1) above), will remain classified until destroyed, or until the OCA determines a change in classification.

e. For information in records determined to have permanent historical value, successive extensions may not exceed a total of 25 years from the date of the information's origin. Continued classification of this information is governed by the automatic declassification provisions of this regulation contained in chapter 3.

*f.* Decisions to extend classification must take into account the potential difficulty of notifying holders of the extension, including the possible inability to ensure continued, uniform protection of the information. Officials who decide to extend a declassification date are responsible for notifying all known holders of the information of the decision and for obtaining assurance from those holders that notification has been made to organizations that were provided the information under further dissemination by those holders.

#### 2–12. Communicating the Classification Decision

An original classification authority who has made a decision to originally classify information is responsible for communicating that decision to persons who will likely be in possession of that information. This will be accomplished by issuing classification guidance, discussed in section V of this chapter, or by making sure that a document containing the information is properly marked to reflect the decision. Marking requirements for classified material, including page and paragraph markings, are covered in chapter 4 of this regulation.

#### 2–13. Compilation

Generally, a compilation of unclassified items of information is not classified. In unusual circumstances, compilation of items of information that are individually unclassified can be classified if the compiled information reveals an additional association or relationship that matches criteria for classification as described in paragraph 2–8 of this regulation. Classification by compilation will be fully supported by a written explanation that will be provided on, in, or with, the material containing the information. An OCA must be consulted if guidance is required concerning whether or not the compilation results in classification.

#### 2–14. Acquisition Systems

Classification and safeguarding of information involved in the DOD acquisition process will conform to the minimum standards of this regulation, as well as the requirements of DODD 5000.1 and DOD Instruction (DODI) 5000.2 (or successor directives and instructions). The term "time or event phased declassification date", used for acquisition systems, is synonymous with the term "declassification date" used in this regulation.

#### 2–15. Limitations and prohibitions

EO 12958 and the Atomic Energy Act of 1954 provide the only basis to classify information. Information will only be classified when it requires protection in the interest of national security as specified in this regulation. Classification cannot be used to conceal violations of law, inefficiency, or administrative error, or to prevent embarrassment to a person, organization, agency, or to restrain competition. Basic scientific research and its results can be classified only if it clearly relates to the national security. Section VI of this chapter covers information that is a product of non–government research and development, that does not incorporate, or reveal, classified information to which the producer, or developer, was given prior access.

## Section IV Security Classification Guides

## 2-16. Policy

A Security Classification Guide (SCG) will be issued for each system, plan, program, or project in which classified information is involved. Agencies with original classification authority will prepare classification guides to facilitate the proper and uniform derivative classification of information. These guides will conform to standards contained in directives and regulations issued under EO 12958 and this regulation.

## 2-17. Content

Security classification guides will, at a minimum, include the following information:

*a.* Identify specific items or elements of information to be protected and the classification level to be assigned each item or element. When deemed useful, specify the items or elements of information which are unclassified or which were previously classified and now are declassified.

b. Provide declassification instructions for each item or element of information, to include the applicable exemption category for information exempted from declassification within ten years. See paragraph 2-11 for exemption categories.

c. Provide a concise reason for classification for each item, element, or category, of information which, at a minimum, cites the applicable classification category or categories from section 1.5 of EO 12958 and that are listed in paragraph 2-8 of this regulation.

d. Identify any special handling caveats or warning notices or instructions, which apply to the items, elements, or categories of information.

*e*. Identify by name, or personal identifier, and position title, the OCA approving the guide, and the date of the approval. A personal identifier is any grouping of letters or numbers used in an organization code that the command uses to identify a particular position. Classification guides will normally be signed by the OCA and, where that is the case, the name and position title, rather than the personal identifier and position title, will be used.

*f*. Provide a point of contact, with telephone number, for questions concerning the guide, challenges to classification, and suggestions for improvement. Provide a statement in the guide encouraging personnel to informally question the classification of information before resorting to a formal challenge. Provide an address for formal classification challenges.

## 2–18. Approval, distribution, and indexing

*a.* Security classification guides will be personally approved in writing by the original classification authority who is authorized to classify information at the highest level designated by the guide, and who has program support or supervisory responsibility for the information or for the command's information security program.

*b*. Security classification guides will be distributed to those commands, contractors, or other activities expected to be derivatively classifying information covered by the guide.

*c*. One paper document copy of each approved SCG (less those for SAPs or programs involving SCI) and its changes will be sent to the Director of Freedom of Information and Security Review, Office of the Assistant Secretary of Defense. Also, one copy, in paper document (hard copy) and/or automated format (soft copy), will be sent to the Army Declassification Special Program Office. See AR 380–381 for guidance on distribution of classification guides for SAPs, and AR 380–28 for guidance on SCI programs.

*d.* Two copies of each guide, other than those covering SAPs or SCI information, will be provided to the Administrator, Defense Technical Information Center (DTIC). Each guide furnished to DTIC must bear the appropriate distribution statement required by DODD 5230.24. Security classification guides issued under this regulation, will be indexed in the DOD Index of Security Classification Guides (DOD 5200.1–I). The originator of the guide will submit DD Form 2024 (DOD Security Classification Guide Data Elements) to the Administrator, DTIC, upon approval of the guide. If the originator determines that listing the guide in DOD 5200.1–I would be inadvisable for security reasons, issuance of the guide will be separately reported, with an explanation of why the guide cannot be listed, to the Director, Special Programs, ODTUSD(P)PS, along with a separate memorandum to DAMI–CH. Report Control Symbol DD–C31 (B&AR) 1418 applies to the reporting requirements of this paragraph.

#### 2–19. Review, revision, and cancellation

*a.* Security classification guides will be revised whenever necessary to promote effective derivative classification. When a guide is revised or reissued, and a specific date was selected for declassification instruction, computation of declassification instructions will continue to be based on the date of the original classification of the information, and not on the date of the revision or re–issuance. Guides will be reviewed by the originator for currency and accuracy at least once every five years, or if concerning a defense acquisition program, prior to each acquisition program milestone, whichever occurs first. Changes identified in the review process will be promptly made. If no changes are

required, the originator will advise the Administrator, DTIC, and DAMI-CH in writing, and the record copy of the guide will be so annotated with the date of review.

b. Guides will be cancelled only when:

(1) All information specified as classified by the guide has been declassified;

(2) When the system, plan, program, or project classified by the guide has been cancelled, discontinued, or removed from the inventory and there is no reasonable likelihood that information covered by the guide will be involved in other classified programs or will be the subject of derivative classification; or

(3) When a major restructure has occurred as the information is incorporated into a new classification guide and there is no reasonable likelihood that information covered by the guide will be the subject of derivative classification.

c. Impact of the cancellation on systems, plans, programs, and projects provided to other nations under approved foreign disclosure decisions, and impact of such decisions on existing U.S. SCGs of similar systems, plans, programs, or projects, will be considered in the decision. When a classification guide is cancelled because the system, plan, program, or project has been cancelled, discontinued, executed, or removed from the inventory, the information covered by the guide is not automatically declassified. That decision rests with the OCA and authorized declassification authorities within the Army. Upon cancellation of a guide, the OCA, or other designated declassification official, with the concurrence of the OCA, will consider the need for publication of a declassification guide. In place of a separate declassification guide, declassification guidance can be included in a classification guide for a similar, current system, plan, program, or project.

*d*. Revision, re-issuance, review, and cancellation of a guide will be reported to DTIC on DD Form 2024 as required for new guides. Copies of changes, reissued guides, and cancellation notices will be distributed as required for new guides as stated in paragraph 2-18 of this regulation.

## Section V Non–Government Information

#### 2–20. Policy

Information that is a product of contractor or individual Independent Research and Development (IR&D) or Bid and Proposal (B&P) efforts, conducted without prior or current access to classified information associated with the specific information in question, cannot be classified unless:

a. The U.S. Government first acquires a proprietary interest in the information.

*b*. The contractor, conducting the IR&D/B&P, requests the U.S. Government activity to place the information under the control of the security classification system, without relinquishing ownership of the information.

#### 2-21. Classification determination

*a.* The individual or contractor conducting an IR&D/B&P effort could believe that information, generated without prior access to classified information, or current access to classified information, associated with the specific information in question, might require protection in the interest of national security. The contractor would then safeguard the information and submit it to an appropriate Army, or other U.S. Government activity, for a classification determination.

b. The Army command receiving such a request will issue security classification guidance as appropriate if the information is to be classified. If the information is not under that command's OCA, the command will refer the matter to the appropriate OCA or inform the individual or contractor to take that action. The information will be safeguarded until the matter has been resolved.

c. The Army command that holds classification authority over the information will verify whether or not the individual or contractor is cleared and has authorized storage capability. If not, the appropriate contracting authority for the command will advise whether or not to process clearance action.

d. If the individual or contractor refuses to be processed for a clearance, and the government does not acquire a proprietary interest in the information, the information cannot be classified.

## 2-22. Classification challenges

*a.* If authorized holders of information have substantial reason to, in good faith, believe that the information is improperly or unnecessarily classified, they will communicate that belief through their command security manager, to the OCA of the information, to bring about any necessary correction. This can be done informally, or by submission of a formal challenge, to the classification as provided for in EO 12958 and this regulation. Informal questioning of classification is encouraged before resorting to formal challenge. Commanders will establish procedures through which authorized holders of classified information within their Commands, can challenge a classification decision, and will ensure that Command personnel are made aware of the established procedures. An authorized holder is any person who has been granted access to specific classified information being challenged. OCAs will establish written procedures through which authorized holders of classified information can challenge classification decisions. At a minimum, security classification guides will contain a point of contact to informally communicate classification challenges and an address to communicate formal classification challenges. EO 12958 establishes the Interagency Security Classification

Appeals Panel (ISCAP). One of the roles of the panel is to decide upon appeals by authorized holders of the information who have made a formal classification challenge as described in this section. See section 5.4, EO 12958, a reprint of which is found at appendix B of this regulation, for more details on the composition and function of this panel.

(1) Formal challenges to classification, made under this subsection, will include a sufficient description of the information being challenged, to permit identification of the information and its classifier, to include the OCA, where known, with reasonable effort. Challenges to classification made by Army personnel will include the reason why the challenger believes that the information is improperly or unnecessarily classified. Use of DA Form 1575 (Request for/ or Notification of Regrading Action) may be used to make a formal challenge. The challenge request should be unclassified, if possible. The classification determination of the OCA will be upheld and carried forward until otherwise determined by the appropriate authorized official.

(2) Commanders will make sure that no retribution is taken against any personnel for making a challenge to a classification.

b. The following will be established by each OCA:

(1) A system for processing, tracking, and recording formal challenges to classification. The system used will differentiate the classification challenges with other reviews for possible declassification (for example, FOIA requests). Requests for information made under the FOIA will be handled as directed by AR 25–55.

(2) The OCA will provide a written response to the challenge within 60 calendar days following the receipt of the challenge. If the OCA cannot respond fully to the challenge within 60 calendar days from receipt, the challenge will be acknowledged and an expected date of response provided. This acknowledgment will include a statement that, if no response is received within 120 calendar days following receipt of the challenge, the challenger has the right to forward the challenge to ISCAP. The challenger can also forward the challenge to the ISCAP if the OCA has not responded to an internal appeal within 90 calendar days of receipt. An internal appeal is when the challenge comes from DA personnel to a Department of the Army OCA. An information copy of the request for appeal, submitted by DA personnel, whether or not to a Department of the Army OCA, will be sent to the original classification authority.

(3) If the challenge is denied and the original classification authority determines that the information is properly classified, the OCA will advise the challenger of the right to appeal the decision. The first level of appeal will be to the first superior general officer in the chain of command of the original classification authority. That general officer will either rule on the appeal, in an impartial manner, or will designate an impartial official, or panel of officials, knowledgeable in the subject matter of the information being challenged, to decide upon the appeal. Both the challenger and the OCA will be advised of the appeal decision. The same time frames and notification to the challenger, stated in subparagraph b, above, apply to the first level of the appeal procedure. If, as a result of the first level of appeal authority will advise the challenger of the right to appeal the decision to the ISCAP. The Director of the ISOO serves as the Executive Secretary of the ISCAP. The correct address to furnish the challenge for appeals to that panel is to the Executive Secretary of the Interagency Security Classification Appeals Panel, c/o ISOO. As of the publication date of this regulation, the mailing address for ISOO is: Information Security Oversight Office (ISOO), National Archives and Records Administration, 700 Pennsylvania Avenue, NW, Room 5W, Washington D.C. 20408.

(4) If a challenge is received concerning information that has been the subject of a challenge, within the preceding two years, or which is the subject of pending litigation, the original classification authority need not process the challenge. The OCA has the option of whether or not to process the challenge. If the challenge is not processed, the challenger will be informed of the situation and that the matter may be appealed to the ISCAP.

(5) If a challenge is received concerning information that has been classified by another OCA within the Army, or by another agency in the U.S. Government, the challenger will be informed of this fact and directed to resubmit the challenge to the appropriate official.

(6) If a challenge is received concerning information classified by a foreign government or international organization, the receiver of the challenge will forward the request for classification review to the appropriate foreign government agency that classified the information. The request to the foreign government for classification review will state that within the United States it is the procedure to respond to these challenges or notify the challenger within 60 calendar days of receipt of the request, and that it would be appreciated if this same response time could be observed. The correspondence to the foreign government will also inquire if there is any appeal authority, and if so, that this authority be listed in the response if the challenge is denied. The challenger will be advised of this referral, if applicable. Army OCAs will be responsive to such informal inquiries and will recognize that the Army has no control over a timely, or lack of, response from the foreign government. The reply from the foreign government, upon receipt, will be forwarded by the requester to the challenger.

c. Information that is the subject of a classification challenge will continue to be classified and appropriately safeguarded until a decision is made to declassify it.

## Chapter 3 Declassification, Regrading, and Destruction

## Section I Army Declassification Program

## 3-1. General

Information will be declassified when it no longer meets the standards and criteria for classification. The authority to declassify information resides with the OCA for that information and those appointed as declassification authorities, subject to the criteria specified in EO 12958 and/or successor orders and directives. Department of the Army files and records will not be declassified without prior review to determine if continued classification is warranted and authorized. EO 12958 contains provisions for four declassification programs as follows:

- a. Original classification authority action.
- b. Automatic.
- c. Mandatory.
- d. Systematic.

## 3–2. Special Program Manager

*a.* The Deputy Chief of Staff for Personnel is the Army Special Program Manager (SPM) for the execution of the centralized portion of the Army's automatic declassification program. The SPM will coordinate declassification actions with Army commands, as required. The authority to decide whether information meets the criteria for continued classification and/or exemption from automatic declassification, remains with the OCA for that information. The declassification decisions of the OCA for the information will be executed by the SPM and by other officials so designated by the OCA. OCAs will publish declassification guides or guidance and will forward them to the SPM and any other officials as deemed appropriate. Declassification guidance can be included in security classification guides or can be published separately, at the option of the OCA.

*b*. In executing the centralized portion of the Army declassification program, the SPM will review, or cause to be reviewed, those files subject to the automatic or systematic declassification provisions of EO 12958, located in the National Archives, Washington National Records Center (WNRC), and Presidential Libraries. The SPM will coordinate the declassification actions of Army commands concerning the review of records subject to the automatic declassification program when those records are located in repositories other than the National Archives, WRNC, and Presidential Libraries. Records stored at other locations will be reviewed by the command responsible for retiring the records, or by that command's successor in function, as appropriate.

c. MACOMs will establish programs to make sure that such records are reviewed and either declassified or exempted prior to the date for automatic declassification. Army commands will provide the SPM with the statistics concerning the declassification review, as the SPM directs.

#### 3-3. Declassification of Restricted Data and Formerly Restricted Data

Restricted Data (RD) and Formerly Restricted Data (FRD) are not subject to EO 12958. This information is classified under the Atomic Energy Act of 1954, as amended. Declassification of RD and FRD information will only be affected with the express specific approval of the classification authority for the information. Generally, this is the Department of Energy (DOE) or DOE in conjunction with DOD.

#### 3-4. Declassification of other than Army information

*a.* Information classified by other U.S. Executive Branch agencies or by foreign governments or international organizations, including foreign contractors, will be referred to the originating agency, or its successors in function. In the case of a foreign government, refer to its legitimate successor for a declassification decision.

b. Every effort will be made to make sure that Foreign Government Information (FGI) is not subject to declassification without the consent of the originating government. FGI can exist in two forms; foreign documents provided to the U.S. and included in Army files and foreign government classified information that is included in a U.S. classified document. If these documents are included in permanently valuable records of the U.S. Government and are subject to the 25-year automatic declassification provisions of EO 12958, declassification officials will consult with the originating foreign government for exemption from automatic declassification in accordance with section II of this chapter. FGI will be tabbed and the Department of State will be notified. The Office of the Deputy to the Under Secretary of Defense (Policy) (ODUSD(P)) for Policy Support should be contacted for assistance and guidance.

c. See appendix C for declassification guidance concerning cryptologic information primarily under the classification authority of the National Security Agency (NSA).

## Section II The Automatic Declassification System

#### 3-5. General

*a.* EO 12958 sets forth policy on the declassification of information. In particular, EO 13142, the amendment to section 3.4 of the EO 12958, requires the automatic declassification of all U.S. classified documents (other than RD or FRD) contained in records that are more than 25 years old on 17 October 2001, or are determined to have permanent historical value under Title 44, USC, unless that information has been exempted from automatic declassification. The declassification requirement will exist for all records of permanent historical value under Title 44, USC, unless that information has been exempted from automatic declassification. The declassification requirement will exist for all records of permanent historical value under Title 44, USC, unless that information has been exempt. If not exempt, automatic declassification will occur whether or not the records have been reviewed. Records that are reviewed and are then exempted will not be automatically declassified.

*b.* Army files subject to automatic declassification will be reviewed prior to the date for automatic declassification. That date is 17 October 2001, for records created prior to 17 April 1976, and 31 December each year thereafter, as records become 25 years old. For records otherwise subject to this section for which a review or assessment conducted by the agency and confirmed by the Information Security Oversight Office has determined that they contain information that was created by or is under the control of more than one agency, or are within file series containing information that almost invariably pertains to intelligence sources or methods, all classified information in such records will be automatically declassified, whether or not the records have been reviewed, within 8 years from the date of EO 12958, except as provided in paragraph 3–6. This date will be 17 April 2003. This is a change to EO 12958, as prescribed by EO 13412, the amendment to Executive Order 12958, dated 19 November 1999 (see appendix B for a copy of the amendment).

## 3-6. Exemption from automatic declassification

*a.* In accordance with EO 12958, the Army has identified and proposed specific designated file series, with description and identification of information in those file series, to be exempt from the 25-year automatic declassification.

b. Files containing information described in the list of exempt file series, must be located and marked to reflect the exemption from automatic declassification, the applicable exemption category, and the date or event for future declassification. Officials conducting a declassification review will apply the exemption markings at the time of the review.

c. Information contained in files not determined to be of permanent historical value under Title 44, USC, is subject to automatic declassification. DA retention and destruction requirements apply.

d. Information exempted from automatic declassification at 25 years remains subject to the mandatory and systematic declassification review provisions of this regulation.

*e*. Classified information not contained in the file series, mentioned in subparagraph a, above, exempted from the automatic declassification system, may be exempted from declassification if it falls within one of the nine exemption categories that are listed below. Under the provisions of EO 12958, section 3.4, the exempting official can exempt from automatic declassification specific information the release of which would be expected to:

(1) Reveal the identify of a confidential human source, reveal information about the application of an intelligence source or method, or reveal the identity of a human intelligence source when the disclosure of that source would damage the national security interest of the United States.

(2) Reveal information that would assist in the development or use of weapons of mass destruction.

(3) Reveal information that would impair U.S. cryptologic systems or activities.

(4) Reveal information that would impair the application of state of the art technology within a U.S. weapon system.

(5) Reveal U.S. military war plans that remain in effect.

(6) Reveal information that would seriously impair relations between the United States and a foreign government, or demonstrably undermine ongoing diplomatic activities of the U.S.

(7) Reveal information that would clearly impair the current ability of United States Government officials to protect the President, Vice President, and other officials for whom protection services, in the interests of national security, are authorized.

(8) Reveal information that would seriously impair current national emergency preparedness plans.

(9) Violate a statute, treaty of international agreement.

*f.* Requests for exemption for other than listed exemptions, must be reported to DAMI-CH, as a "Notice of the Intent to Exempt Information from Automatic Declassification." The notice will—

(1) Describe the specific information to be exempted.

(2) Explain why the information must remain classified.

(3) Except for the identity of a confidential human source or a human intelligence source, provide a specific date or event upon which the information will be declassified.

g. DAMI-CH will review the request for conformity with current policy. If the document meets that policy,

DAMI-CH, as the Senior Agency Official, will notify the ISSO of the intent to exempt this information from automatic declassification. See paragraph 1–15 for further information on the ISSO. Notification must be received by ISSO, acting as the Executive Secretary of the ISCAP, 180 days before the information is scheduled for automatic declassification. The notice must contain the information identified in paragraph 3–6e.

h. In the case of foreign documents, the declassification review official will review the provisions of paragraph 3-4b and then determine if exemption categories 6 or 9 (25X6 or 25X9) or both should be applied.

#### 3–7. Marking of documents exempted from automatic declassification at 25 years

*a.* Documents that are exempted from automatic declassification after 25 years, will be marked with the designation "25X", followed by the number of the exemption category (see categories listed in paragraph 3–6e), or by a brief reference to the pertinent exemption. For example, "25X–Human Source," notes the information is exempted from automatic declassification because it reveals the identity of a human source, as described in exemption 1. It could also be marked 25X1. If the document is exempt because of a human source, do not mark a declassification date. All other exemptions will be marked with a future declassification date, or event, established by the exempting authority. For example, a document created in 1974 is reviewed and found to contain information that requires classification beyond 17 October, 2001, because of exemption category 4, "Reveal information that would impair the application of state of the art technology within a U.S. weapon system." The exemption authority determines that the information must remain classification date in ten–year blocks, per paragraph 2–11. This means the information could be declassified on 31 December 2019 (1999 + 20 = 2019). In this case, the declassification instructions for the document could be written as: "Declassify on 25X4, 31 December 2019".

*b*. Files, documents or other material that are subject to final storage, at the National Archives or Federal Records Center, must be marked in such a way that it is clear whether the material has been exempted or declassified, in addition to the markings stated in paragraph 3–7a. The SPM, under the direction of ISSO, will provide guidance on the method of marking or tabbing those files.

## Section III Mandatory Review for Declassification

#### 3-8. General

*a.* Any individual or organization may request a review for declassification of information. Upon receipt of such a request, the command will follow the general policy for classification challenges (see chap 2, section VII). Response time frames for mandatory review can be extended at command option. The request will be referred to the originating or successor agency for declassification review. In either a classification challenge or request for mandatory review for declassification, the command will refuse to confirm or deny the existence or non–existence of requested information, when the fact of its existence or non–existence is properly classified. A mandatory declassification will not be conducted if declassification review occurred within the preceding two years.

b. Information originated by the incumbent President, the incumbent President's White House staff, committees, commissions, or boards appointed by the incumbent President, or other entities within the Executive Office that solely advise and assist the incumbent President, is exempt from the provisions of this section.

#### 3-9. General

*a*. Heads of HQDA activities and MACOMs will, as permitted by available resources, establish systemic programs to review for declassification, information classified under EO 12958, and predecessor orders, in the custody of the Army that is contained in permanently valuable historical records, and is exempt from automatic declassification.

*b*. These efforts will concentrate on records that have been identified to have significant value for historical or scientific research, or promoting the public welfare, and have reasonable probability of being declassified upon review.

#### Section IV Regrading

#### 3-10. Downgrading information

Downgrading information is appropriate when the information no longer requires protection at the originally assigned level. Classified information can be upgraded to a higher level of classification only if holders of the information can be notified of the change so that the information will be uniformly protected at the higher level. The OCA is authorized to downgrade and upgrade information and is responsible for notifying holders of the change in classification.

#### 3–11. Downgrading policy

a. Purpose and authority. Downgrading of information to a lower level of classification is appropriate when the information no longer requires protection at the originally assigned level and can be properly protected at a lower level.

The principal purpose of downgrading is to conserve security resources by avoiding protection of information at too high a level. Downgrading is accomplished by the OCA for the information only.

b. Downgrading decisions during original classification. Downgrading should be considered when OCAs are deciding on the duration of classification to be assigned. If downgrading dates or events can be identified, they must be specified along with the declassification instructions. Note that downgrading instructions do not replace declassification instructions.

c. Downgrading at a later date. Information may be downgraded by any official who is authorized to classify or declassify the information, namely the original classification authority. The OCA making the downgrading decision will notify holders of the change in classification.

## 3–12. Upgrading

Classified information may be upgraded to a higher level of classification only by officials who have been delegated the appropriate level of original classification authority, in accordance with chapter 2, section II, of this regulation. Information may be upgraded only if holders of the information can be notified of the change so that the information will be uniformly protected at the higher level. The OCA making the upgrading decision is responsible for notifying holders of the change in classification.

## Section V

## **Classified Material Destruction Standards**

## 3-13. General

Classified material will be destroyed completely to preclude recognition or reconstruction of the classified information contained in or on the material. Destruction methods include burning, crosscut shredding, wet pulping, melting, mutilation, chemical decomposition, and pulverizing. This section contains basic concepts and guidelines that assist in determining the sufficiency of the various destruction techniques. This section also provides residue dimension standards that will assist in achieving secure destruction. Destruction will be accomplished in accordance with these guidelines.

## 3-14. Concepts of destruction

The guidelines in this chapter are acceptable when employed in a timely manner to prevent excessive accumulation and in conjunction with the "secure volume" and "data density" concepts of destruction, explained below.

a. "Secure Volume" Concept. The "secure volume" concept of destruction processing stresses that security is enhanced, not only by small residue particle size, but also by restricting the chances of successful reconstruction of that residue, by increasing the number of pieces involved. This increase can be achieved in two ways. Prohibit destruction until a quantity of no less than 20 similar pages of classified paper are destroyed at one time or add sufficient similar type of unclassified pages of paper, not blank paper, to arrive at the minimum 20 similar page count. Either method will result in a "secure volume" of residue. The "secure volume" concept will be a standing operating procedure for the use of all office-type approved security shredders. Bulk feeding procedures for the larger, high-volume destruction equipment systems (pulpers and pulverizers) normally allow for "secure volume" destruction.

b. "Data Density" Concept. The following standards apply to the destruction of graphic materials where "data density," print or image ratio to blank space per square centimeter, is no greater than that employed to print this paragraph. Where smaller print is employed, the data density per square centimeter is greater than that appearing in the paragraph. Examples of high data density material are microfilm, microfiche, and aerial photography. Consequently, a more stringent destruction standard is necessary when processing high data density materials, than is established here, for office copy paper–based items. To achieve a more stringent standard, a smaller sized security screen is employed, or the material is completely destroyed by burning. The data density determination and subsequent security screen size required to be used, is the responsibility of the Security Manager at each installation and activity. No single security screen standard for all graphic material destruction is established due to differences in data density. Therefore, several security screen sizes are needed for each mechanical system using such screens. The screens should be within reach of the operator, or otherwise easily accessible, to preclude insecure destruction.

#### 3-15. Approved routine methods of destruction

The methods for routine destruction of classified material, shown below, are approved for use by Army commands when the "secure volume" and "data density," where applicable, concepts are employed.

*a. Burning.* A means of pyrolysis (high temperature multistage), by forced-air incinerators, or by any other incinerator or incendiary equipment which reduces the material to an ash such that reconstruction of the information is not possible. No other single destruction method has been found to be as effective, versatile, and secure, as burning. However, since there are limitations, in some areas, on bulk incineration, for environmental reasons, pyrolytic furnaces, as well as other mechanical destruction systems, have replaced incineration at many commands. Pyrolytic furnaces operate in compliance with Federal Clean Air Act Regulatory Standards, and are to be given preference over other incinerators, when possible. Commands are advised to obtain a written guarantee, from the pyrolytic furnace seller,

attesting to the unit's ability to be licensed, and to operate within the standards applicable at the point of installation, since standards vary from state to state. Commands located outside the United States, will conform to host nation standards, if they are more restrictive than U.S. standards, unless emergency destruction is necessary. Pyrolytic furnace ash residue and ash residue from other forms of burning will not contain unburned product. If unburned product is found, it will be treated as classified waste and maintenance personnel will be instructed to correct this fault in the furnace's burn cycle. The ash must be stirred to make sure destruction is complete and reconstruction is impossible. Ash residue is to be examined and reduced by physical disturbance and will be considered destroyed when capable of passing through a 1/2–inch (13–mm) square wire sieve. It is recommended that furnace operators be permanently assigned and trained to perform necessary adjustments and maintenance, and be cleared for access to the highest level of material being routinely destroyed.

b. Shredding. Crosscut shredders are the only authorized shredders approved for use in the destruction of classified information. The crosscut shredding machine must reduce the material to shreds no greater than 1/32nd of an inch (plus 1/64th inch tolerance) by 1/2 inch crosscut. The Class I shredder, identified by GSA Interim Federal Specifications FF–S–001169, meets this standard and is approved for use when the "secure volume" concept is employed. Any other cross–cut shredder whose residue particle size, total area, is equal to or smaller than that of the above Class I shredder, is similarly approved for classified destruction, when used in accordance with the secure volume concept of operation. Classified microfilm, microfiche, or similar high data–density material will not be destroyed by shredding to the standards described in this section. They can only be destroyed as indicated in paragraph 3–16.

*c. Pulping.* Standard wet process pulpers, with a 1/4–inch or smaller diameter perforated security screen, are approved for the destruction of classified paper–based documents. The Interim Federal Specifications FF–P–00800A, with Amendment 2, specifies the perforated screen or ring used in the masticating unit, through which all pulp must pass, will have 1/4–inch (6.35–mm) or smaller diameter perforations, and therefore, meets this standard. Since the pulping process entails wetting and dissolving action, plastic–based or other water–repellent–type papers will not be put through this system. However, if wetting additives are used and the ratio of soluble to non–soluble paper is kept high (16 to 1 or greater), the masticating unit will tolerate that material. This toleration is totally dependent upon the sharpness of the pulper's cutters. Foreign matter, such as metal and glass, must be excluded from charge loads by visual inspections. Since pulpers generally destroy only paper products, staples, paper clips and other fasteners must be removed so they do not clog the security screen. Commands will make sure that random samples of residue from pulpers are collected for periodic examination.

*d. Pulverizing.* Interim Federal Specifications FF–P–00810A, with Amendment 3, covers pulverizing as a dry destruction process. It does not, however, specify a specific dry destruction method; thus, within this category are hammer mills, choppers, hoggers, and hybridized disintegrating equipment.

(1) *Hammer mills*. Hammer mills destroy by a flailing action. Paper, lightweight plastics and wood, glass slides, and aluminum–offset plates, as well as other easily broken materials, can be destroyed in a hammer mill. This process is extremely destructive, very noisy, and can be dusty if the air–handling system is not kept in good repair. Equipped with a 1/4–inch (6.35–mm) or less diameter security screen, hammer mills are approved for destruction of classified paper–based materials and aluminum offset plates, provided the "secure volume" concept is employed. When required to destroy non–paper–based classified material or high–density substances such as classified microfilm and microfiche, the security screen size will be reduced to a diameter of at least 1/16th of an inch (1.588–mm) or smaller. If used to destroy plastic film–based material, care must be exercised in the feeding of the hammer mill because of the high heat buildup that can result, causing film to melt, fuse or burn. To prevent this, paper and plastic–based films are to be alternately fed into the hammer mill.

(2) *Choppers*. Choppers cut by a scissors action between one or more fixed and one or more rotating square–edged surfaces. This system's waste volume expansion is the most compact of the various dry mechanical destruction systems. Choppers are approved for destruction of classified paper–based documents using a 3/16–inch (5–mm) diameter perforated security screen, provided the "secure volume" processing concept is used.

(3) *Hoggers and hybrids.* Hoggers and other hybridized disintegrating equipment are principally for high–volume destruction operations, such as destroying one or more tons per day. Because there are many hogger and hybrid designs on the market, a better description of this destruction methodology, and the appropriate security screen size for each, cannot be given here. It is recommended that "secure volumes" and a security screen size of 1/4–inch (6.35–mm) be employed for classified paper–based materials processed in these systems. If the command security manager determines that the residue from such a screen size consistently reflects excessive destruction, the security screen perforation size may be increased to 5/16–inch (7.94–mm), provided all processing through the device employs the "secure volume" concept and 50–pound minimum loads. In addition, the security manager will make sure that frequent residue examinations are made to determine that the destruction of the information is complete.

## 3-16. Appropriate material destruction techniques and methods for non paper-based material

Shredding, pulping, and pulverizing machines built to produce the above residue standards are used primarily in the destruction of classified paper-based products. Classified waste containing typing ribbon, aluminum and plastic offset printing mats, and other non-paper-based products require special handling. They must be segregated, marked to reflect their content and classification, and dealt with on an individual basis. These items can cause serious damage if

allowed to accidentally enter some of these machines. Thus, every effort must be made to keep foreign matter out of burn bags. Non-paper-based classified material is to be disposed of as follows when a pyrolytic furnace is not available or is inappropriate:

a. Non water-soluble plastic coated, waxed paper, plastic acetate, or similar material. This material will be burned in a pyrolytic furnace or other incinerator or incendiary device, destroyed in one of the high-capacity dry pulverizing systems, or shredded. Such material will not be allowed to enter wet pulping systems. Carbon paper is an exception, since it has relatively low tensile strength.

b. Magnetic Storage Media (MSM), such as materials for audio and video recorders, computers, and Automated Data Processing (ADP) office equipment. See AR 380–19 and/or section VII below for standards for destruction and degaussing of classified information on MSM.

c. Typewriter ribbons and cassettes (mylar, nylon, and cotton-based ribbon). This category of material should be destroyed by burning, since any other method involves both a serious risk of damage to the mechanical destruction equipment and the attendant mess of manual handling. Shredding, chopping, and hammer mill pulverizing requires the necessity of removing the ribbon from its reel by radially slitting with a razor blade. This ensures that no one strip is longer than 10-inches (25.4-cm). Longer strips have a tendency to become entangled in destruction equipment. Once cut from the reel, this material is to be fed into the destruction system intermixed with paper-based material, sufficient to assist with its being purged from the system. A heavy-duty (1.5 horsepower or larger) crosscut security shredder can be used if fed slowly; however, the standard office-type shredder cannot be used. When using a heavy-duty shredder, the strips of ribbons must be fed in so that they are also sliced across their longest dimension. This will minimize the possible jamming of the machine by having any strips wrap around the cutting reel. When any other dry destruction process is used for ribbon strips, a security screen appropriate for TOP SECRET paper-based material must be used.

*d. Original microfilm and microfiche and other silver–based photographic material.* This category of material (having a silver content), to include black and white and colored photographs and negatives, x–rays, aerial films and photographs, and unexposed, expired film, must always be segregated and destroyed by pyrolysis in a silver reclamation furnace for both security and economic reasons. The silver content of these items remains with the ash and can be salvaged.

*e. Duplicate microfilm and microfiche.* Microfilm and microfiche duplicates normally are made by processes that do not employ silver. Therefore, this type of material can be burned along with other paper and plastic materials in a pyrolytic furnace or other incinerator. Where burning is not permitted, consideration is to be given to a centralized collection point for destruction by burning at another location. Several destruction devices, for use with classified non–COMSEC plastic base micrographic products, have been approved. They produce extremely fine particulate and, when employed in conjunction with the "secure volume" concept, achieve the proper level of security. Further information on these commercial devices is available from: Commander, Intelligence Materiel Activity (IMA), Fort Meade, Maryland 20755–5315. As a last resort, a properly screened (1/16–inch (1.588–mm) or smaller) hammer mill can be used. Hammer mills must be fed plastic film and other plastic–based materials very slowly to avoid heat build–up. It is best to wait and mix in batches of paper between charges of film to allow cooling and to remove softened plastic from the hammer mill. Further information on commercial devices is available from IMA.

*f. Equipment and devices.* Equipment, devices and other solid objects are best destroyed by burning, preferably in a pyrolytic furnace. Where destruction by exposure to flame is insufficient to achieve the necessary secure level of destruction, other means must be used. Dependent upon the nature of the item to be destroyed, the means selected must achieve the desired results, which is the information is destroyed completely so as not to allow recognition or reconstruction of the classified information, and involve a minimum of hazard for personnel involved. Several common methods are listed below.

- (1) Burning and melting with an oxyacetylene torch.
- (2) Sledge hammer and hacksaw demolition.
- (3) Use of local smelter or foundry retort or open hearth or other furnace to melt beyond recognition.
- (4) Crushing by hydraulic press beyond recognition.
- (5) Hogging in a heavy-duty, industrial-type hogger equipped with a suitable security screen.

#### 3-17. Technical advice on approved destruction devices and methods

Destruction devices generally can be obtained through the National Supply System (FSC Group 36, Part II). Technical guidance concerning appropriate methods, equipment, and standards for the destruction of classified electronic media and processing equipment components, can be obtained by submitting all pertinent information to the National Security Agency, Attention: NSA/CSS Directorate for Information Systems Security, Fort Meade, MD 20755. Technical guidance concerning the standards for the destruction of other storage media can be obtained from: Intelligence Materiel Activity (IMA), Fort Meade, MD 20755–5315.

## 3-18. Clearing, purging, declassifying, and destroying media

a. Clearing of media means erasing or overwriting all information on the media without the totality and finality of purging. The clearing procedure is adequate when the media will remain within the facility; however, removable media

must continue to be controlled at their prior classification or sensitivity level. Purging or sanitizing of media means to erase or overwrite, totally and unequivocally, all information stored on the media. Declassifying of media refers to the administrative action taken after it has been purged. Declassifying is required when the media must leave the facility under the control of uncleared personnel; for example, for maintenance operations.

*b*. The decision to declassify media will be made only after comparing the inherent risks (in the Magnetic Media Remanence Guide – Rainbow Series) with the financial or operational benefit of media declassification. For example, destruction of media is normally more appropriate than declassification and reuse, given the low cost of the media.

c. Media can be declassified only after purging. The appropriate ISSO must verify that the technique chosen for purging (or sanitizing) meets applicable requirements. Additionally, the ISSO must establish a method to periodically verify the results of the purging. As a minimum, a random sampling will be taken to verify each purge.

*d.* Degaussing must be accomplished using NSA–approved equipment from the Degausser Products List of the Information Systems Security Products and Services Catalogue. Information on degaussers is available through the information systems security management structure. Some listed products may be used only to degauss magnetic media that has coercivity no greater than 350 oersteds (also known as type I media), while others are approved for media with coercivity no greater than 750 oersteds (also known as type II media). Certain tape media have a coercivity greater than 750 oersteds (also known as type II media). Certain tape media have a coercivity greater than 750 oersteds (also known as type II media).

e. A CD-ROM will be destroyed by scratching both surfaces with an abrasive substance, to render the CD unreadable, prior to breaking the CD into numerous pieces with an impact device, such as a hammer.

f. Storage media containing Sensitive Compartmented Information (SCI) will be handled as stated in AR 380–19, and media containing Special Access Program (SAPs) material will be handled as stated in AR 380–381.

## Chapter 4 Marking

## Section I Marking Documents

#### 4-1. Purpose and policy

Marking is the principal means of informing holders of classified and sensitive information of its classification/ sensitivity level and protection requirements. Within the Department of the Army, classified and sensitive material will be identified clearly by marking, designation, electronic labeling, or if physical marking of the medium is not possible, by some other means of notification. The term "marking" as used in this regulation is intended to include all these methods of notification. The term "document" as used in this section is meant to apply to all classified and unclassified material, no matter what form (paper, electronic, etc.) it is in. Classification/sensitivity markings must be conspicuous. Original and derivative classifiers are responsible for application of the appropriate classification/sensitivity markings. The requirements for marking information and material within the intelligence community are a little different. These requirements can be found in appendix D, of this regulation, and successor Director of Central Intelligence Directives (DCID). The requirements of this chapter do not apply to the marking of security containers. The only markings allowed on security containers are those outlined in paragraph 7–8 of this regulation. Marking serves these purposes:

- a. Alerts holders to the presence of classified and sensitive information.
- b. Identifies, as specifically as possible and feasible, the exact information needing protection.
- c. Indicates the level of classification/sensitivity assigned to the information.
- d. Provides guidance on downgrading (if any) and declassification.
- e. Gives information on the source(s) and reason(s) for classification of the information.
- f. Warns holders of special access, control, dissemination, or safeguarding requirements.

#### 4-2. Exceptions

*a.* Public Media — Classification and/or other security markings will not be applied to an article or portion of an article that has appeared in a newspaper, magazine, or other public medium. If such an article is evaluated to see if it contains classified and/or sensitive information, the results of the review will be properly marked, if classified and/or sensitive, and will be kept separate unless both the article and the results of the review are protected (stored and otherwise safeguarded as classified/sensitive information). DA personnel will neither confirm nor deny the presence of classified and/or sensitive information or the accuracy of such information when that information has appeared in the public media.

*b*. Confidential Source or Relationship — Classified documents and material will be marked in accordance with this regulation unless the markings themselves would reveal a confidential source or relationship not otherwise evident in the document, material, or information.

c. Restricted Data/Formerly Restricted Data — The marking requirements for the date or event for declassification do not apply to documents or other material that contain, in whole or part, RD or FRD information. Such documents or other material or portions thereof will not be declassified without approval of the Department of Energy with respect to Restricted Data or Formerly Restricted Data information, and with respect to any national security information contained therein, the approval of the originating agency.

## 4-3. Requirements

General requirements are shown in this section. Each of these requirements is explained in more detail in a separate section of this chapter. Figures 4–1 through 4–13, at the end of this chapter, provide examples of the most typical situations. These figures are not intended to cover all situations. Material other than paper documents require the same markings and must have the same information either marked on it or made available to holders by other means of notification. While not a requirement, the holder of an improperly marked classified document should contact the document originator to obtain correct markings. Classified and sensitive material will bear the following markings:

a. The overall (highest) classification/sensitivity of the information.

b. The command, office of origin, date, and if not evident by the name of the command, the fact that the document was generated by the Department of the Army.

c. Identification and date of the specific classified information in the document and its level of classification (page and portion markings).

*d.* Identification of the source(s) of classification ("Classified by" or "Derived from" line), and, for originally classified information, the concise reason(s) for classification. In cases of derivative classification, the reason(s) the source of the classified portion(s) is/are derived from.

e. Declassification instructions ("Declassify on" line), and downgrading instructions, if any downgrading applies.

f. Warning and sensitivity notices and other markings, if any, that apply to the document.

## 4-4. Overall classification marking

Classified and sensitive documents will be marked to show the highest classification/sensitivity of information contained in the document. For documents containing information classified at more than one level, the overall marking will be at the highest level. For example, if a document contains some information marked "SECRET" and some information marked "CONFIDENTIAL", the overall marking would be "SECRET". This marking must be conspicuous enough to alert personnel handling the material that it is classified and must appear in a way that will distinguish it clearly from the text of the document. The overall classification/sensitivity will be conspicuously marked, stamped, or affixed (with a sticker, tape, etc.), top and bottom, on the front and back covers (if the document has covers), on the title page (if there is one), and on the first page, in letters larger than those on the rest of the page. If it is not possible to mark classification/sensitivity in letters which are larger than the rest of the text (for example, on covers of documents or graphics), apply classification/sensitivity markings in any manner that is immediately noticeable. To promote reproducibility, classification/sensitivity and associated markings will be applied in black or other dark ink. The use of red ink is discouraged. If the document or other material has no front cover, the first page will be the front page. If it has a cover, the first page is defined as the first page that can be seen when the cover is turned back or opened. In some documents, the title page and first page can be the same.

## 4-5. Date, command, office of origin, and agency

Classified and sensitive documents will be marked on the face of the document with the date of the document, the command that originated it, the office or agency which originated it, and "U.S. Army" or "Army" if it is not clear from the name of the command that it is a DA activity originating the document. This information will be clear enough to allow the recipient of the document to contact the preparing office if questions or problems about classification arise.

## 4–6. Page and portion marking

Each classified and/or sensitive document must show, as clearly as possible and feasible, which information in it is classified and/or sensitive and at what level. That will be done in the following manner:

*a*. Each interior page of a classified and/or sensitive document (except blank pages) will be conspicuously marked, top and bottom, with the highest classification/sensitivity of the information on the page. The marking must be conspicuous enough that it is clearly distinguishable from the regular text of the document. Blank interior pages are not required to be marked. This is the preferred method of page marking. As an alternative to marking pages according to individual page content, the interior pages can be marked with the highest overall classification/sensitivity of information within the document. If this alternative method is used, portion marking must be used and cannot be excepted as described in paragraph 4–6c, below.

b. Each section, part, paragraph, and similar portion of a classified and/or sensitive document will be marked to show the highest level of classification/sensitivity of information it contains, or that it is UNCLASSIFIED. "Portion marking" is the term used to meet this requirement. The term "paragraph marking" is generally used interchangeably with "portion marking". Whether referred to as portion or paragraph marking, the term includes the marking of all portions of a document, not just paragraphs. When deciding whether a subportion (such as a subparagraph) will be

marked separately as a "similar subportion", the deciding factor is whether or not the marking is necessary to eliminate doubt about the classification/sensitivity of its contents. Unless the original classification authority or originator of the document indicates otherwise on the document, each classified and/or sensitive portion of a document will be presumed to carry the declassification instructions (date, event, or exemption category) of the overall document.

(1) Each portion of text will be marked with the appropriate abbreviation ("TS" for TOP SECRET, "S" for SECRET, "C" for CONFIDENTIAL, or "U" for UNCLASSIFIED), placed in parentheses immediately before the beginning of the portion. If the portion is numbered or lettered, the abbreviation will be placed in parentheses between the letter or number and the start of the text. Some agencies permit portion marking at the end of the portion, rather than at the beginning. The Department of the Army does not. When extracts from non–DA documents are made and incorporated into DA documents, the portion marking will be placed at the beginning of the portion.

(2) Portions containing Restricted Data (RD) and Formerly Restricted Data (FRD) will have abbreviated markings ("RD" or "FRD") included with the classification marking, for example, "(S-RD) or (S-FRD)". Critical Nuclear Weapons Design Information (CNWDI) will be marked with an "N" in separate parentheses following the portion marking, for example, "(S-RD)(N)".

(3) The abbreviation "FOUO" will be used in place of "U" when a portion is UNCLASSIFIED but contains "For Official Use Only" information. AR 25–55 contains the definition and policy application of FOUO markings. See chapter 5, of this regulation, for further guidance, as well.

(4) Portions of DA documents containing foreign government or NATO information will include identification of the foreign classification in the marking in parentheses. For example, "(UK–S)" for information classified "SECRET" by the United Kingdom; and "(NATO–C)" for North Atlantic Treaty Organization (NATO) information classified as "CONFIDENTIAL".

(5) Paragraph 5–410, DODD 5200.1–R stated that the caveat "NOFORN" will no longer be used. This has since been rescinded. Effective with the release of Director of Central Intelligence Directives (DCID) 1/7 (see app D) and DCID 5/6, both dated 30 June 1998, the use of "US ONLY," to mark information that must be restricted to U.S. nationals, will cease. Until revoked, this type of information will be marked "NOFORN." This applies to all media, including hard copy, digital, and graphic.

(6) The subject and title of classified documents will be marked to show the classification of the information in the subject or title. The same abbreviations ("TS", "S", "C", "U", or "FOUO") will be used but the abbreviations will be placed in parentheses at the end of the subject or title.

(7) Charts, graphs, photographs, illustrations, figures, tables, drawings, and similar portions will be marked with the unabbreviated classification/sensitivity, such as "UNCLASSIFIED", based on the level of classified and/or sensitive information revealed. The marking will be placed within the chart, graph, etc., or next to it, such as on the frame holding the document. Captions and titles of charts, graphs, etc., will be marked as required for text portions (such as paragraphs) and will be placed at the beginning of the caption or title.

(8) See appendix D for an explanation of marking certain intelligence control markings (for instance, ORCON and PROPIN). Portion marking of those intelligence control markings will follow the policy as stated in DCID 1/7 and successor directives.

(9) See appendix I for an explanation of marking Special Access Programs (SAPs) material.

c. If an exceptional situation makes individual marking of each paragraph or other portion clearly impracticable, a statement can be substituted describing the fact that portion markings were not used, and which portions are classified and/or sensitive and their level of classification/sensitivity. Such a statement will identify the information as specifically as would have portion markings. For classification by compilation, the statement required by paragraph 2-13 meets this requirement.

d. Documents containing information classified by compilation will be marked as follows:

(1) If portions, standing alone, are UNCLASSIFIED, but the document is classified by compilation (see para 2–13), mark the portions as "(U)" and the document and pages with the classification of the compilation. You must also add an explanation of the classification as required in paragraph 2–13 of this regulation.

(2) If individual portions are classified and/or sensitive at one level, but the compilation results in a higher classification/sensitivity, mark each portion with its own classification/sensitivity and mark the pages, and the overall classification/sensitivity of the document, with the higher classification/sensitivity of the compilation. An explanation of the classification/sensitivity by compilation is required to be placed in the document, preferably on the cover or title page.

(3) DAMI-CH will be contacted for guidance on submission of waivers or exceptions to policy concerning the marking of documents classified and/or sensitive by compilation.

#### 4–7. Sources of classification – overview

Each classified document will be marked with the source of the classification. For originally classified documents, that identification will be preceded by the term "Classified by". In cases of derivative classification, the source of classification is derived from either:

(1) A classification guide or guidance.

(2) A classified source document that was used to extract, summarize, restate, or paraphrase information from the source document into the new document.

(3) When compilations of items of information that are individually unclassified can be classified if the compilation reveals an additional association or relationship that matches criteria for classification pursuant to paragraph 2–8 of this regulation. For derivatively classified documents, the term "Derived from" will precede the identification of the source of classification. This is a change from previous policy. Previous policy required the use of the "Classified by" line for both originally and derivatively classified documents. Current policy requires the use of the "Classified by" line only for original classification documents and combination original and derived documents, and requires the use of the "Derived from" line for only wholly derived classified documents. See chapter 2 for a further explanation of the differences between, and requirements for, originally and derivatively classified documents.

#### 4–8. Sources of classification – procedures

*a.* Originally classified documents. Each originally classified document will have a "Classified by" line placed on the face of the document. The "Classified by" line will identify the original classification authority responsible for classification of the information contained in the document. The OCA will be identified by name or personal identifier (see paragraph 2–17e for an explanation of the term "personal identifier"), and position title. If the information required to be included in the "Classified by" line would reveal classified information not evident from either the rest of the document or not evident from the face of the document, the "Classified by" line will be completed with an UNCLASSIFIED identification (such as an UNCLASSIFIED personal identifier) that can be traced through secure channels.

b. Derivatively classified documents. Each derivatively classified document will have a "Derived from" line placed on the face of the document. The term "Classified by" will not be used on classified documents that are wholly derivative. The "Derived from" line will be completed as follows:

(1) If all the information was derivatively classified using a single security classification guide (or guidance) or only one source document, identify the guide or the source document on the "Derived from" line. Include the date of the guide or document. If using a source document that cites a guide as classification authority, use the guide rather than the source document on the "Derived from" line.

(2) If more than one security classification guide, source document, or combination of guide(s) and document(s) provided the derivative classification guidance, use the term "Multiple Sources" on the "Derived from" line. If "Multiple Sources" is placed on the "Derived from" line, a record of the sources will be maintained on or with the file or record copy of the document. Whenever feasible, this list should be included with all copies of the document. If the document has a bibliography or list of references, that can be used as the listing of sources as long as it is annotated to delineate the sources of classification from the other references. A document derivatively classified on the basis of a source document that is itself marked "Multiple Sources" will cite the source document on its "Derived from" line rather that the term "Multiple Sources". (For example, "Derived from: Headquarters, Department of the Army Report, Security 2001, an Army Odyssey, 10 February 1997, Office of the Deputy Chief of Staff for Intelligence (DAMI–CH).")

c. Combination of original and derivative classification. There can be situations in which some information in a document is originally classified at the time of preparation of the document and some information is derivatively classified. In those cases, mark the document with a "Classified by" line and place "Multiple Sources" on the line. For the information originally classified in the document, the OCA will be included in the list of sources required in paragraph 4-8b(2).

#### 4–9. Reason for original classification

Each originally classified document will bear a concise line that describes the reason for the decision to classify. This requirement applies only to originally classified documents and does not apply to derivatively classified documents. The "Reason" line will not be used on wholly derivatively classified documents. The "Reason" line is placed between the "Classified by" line and the "Declassify on" line. The reason(s) to classify relates to the categories of what can be classified, as specified in paragraph 2–8. The "Reason" line will either:

*a*. State one or more of the reasons listed in paragraph 2–8. For example: "Reason: Military plans, weapons systems, or operations"; or "Reason: Foreign government information"; or "Reasons: Military plans, weapons systems, or operations; and foreign government information".

b. State the reason in terms of listing the number "1.5" followed by the letter, in parentheses, that corresponds with the appropriate category or categories of information listed in section 1.5 of E.O 12958. This is the same list shown in paragraph 2–8 of this regulation. For example: If the information is classified because it concerns military plans, weapons systems, or operations, mark the document: "Reason: 1.5(a)". If the document is classified because it contains foreign government information, mark the document: "Reason: 1.5(b)". If the document is classified for both reasons, mark the document: "Reason: 1.5(b)".

c. For those cases in which the document contains both originally classified and derivatively classified information,

state the reason(s), as described in subparagraph a or b of this paragraph, and add the words, "and derivatively classified source" or "and derivatively classified sources", where more than one derivative source document is used.

#### 4-10. Declassification instructions—"Declassify on" line

Each classified document (except those containing RD and FRD) will be marked on the face of the document with a "Declassify on" line, with instructions for the declassification of the information. This applies for all classified documents, both originally and derivatively classified. The "Declassify on" line will be completed as follows:

a. Originally classified documents. If all the classified information is the product of original classification, the OCA will specify the instruction on the "Declassify on" line. The instruction will specify either a date for declassification, an event for declassification, or an indication that the information is exempt from declassification within ten years.

(1) If any information in the document has been exempted from declassification within ten years, referred to as the "Ten Year Rule," the "Declassify on" line will be completed with an "X" followed by a number or numbers which show the applicable exemption category or categories from paragraph 2–11c. There is no alternative to listing the category in this manner. For example, a document containing information requiring classification beyond ten years because it would reveal information that would impair the development or use of technology within a U.S. weapon system (category number 3 under paragraph 2–11c) would be marked: "Declassify on: X3". As another example, a document containing information requiring classification beyond ten years because it reveals U.S. military plans (category number 4 under paragraph 2–11c) would be marked "Declassify on: X4". Note: Listing the exemption category number rather than the words describing the category is the preferred method of citing declassification exemption instructions within DOD.

(2) For cases in which it is possible for the Original Classification Authority to select a date or event for declassification at a point occurring more than ten years in the future, the date or event would follow the exemption category number. An example is "Declassify on: X3, 11 November 2011". There can be cases in which it is not possible for the OCA to select a future date or event for declassification. In those cases, only the exemption category will be listed. Examples for those cases, such as "Declassify on: X3", are shown in subparagraph (1), above.

(3) If more than one exemption applies, the OCA will list each exemption. For example, a document originally classified beyond ten years because it would reveal information that would impair the development or use of technology within a U.S. weapons system (exemption category 3) and that also contains foreign government information (exemption category 5) would be marked, "Declassify on: X3,5".

(4) Regardless of the exemption category used, or whether or not a date or event for declassification has been selected, the information can be subject to the automatic declassification provisions of the current EO or statute on classification. EO 13142, the most recent amendment to EO 12958, section 3.4, requires all classified information contained in records that will be more than 25 years old on 17 October 2001, and have been determined to have permanent historical value under Title 44, USC, to be declassified on 17 October 2001, unless the information has been exempted. The current criteria for exemption for 25 year old information is contained in chapter 3 of this regulation and in section 3.4 of EO 12958. It is impossible to predict what criteria will apply to any future automatic declassification programs. It is important for OCAs to carefully select the appropriate exemption category (with or without a declassification date or event) for currently classified information. Future automatic declassification programs might use a formula to convert current exemptions to future criteria used in reviewing old classified documents. If more than one exemption applies, it is important to list each exemption category.

b. Derivatively classified documents. In a derivatively classified document there may be one source from which the classification is derived, or there may be several sources. The source may have been classified after 14 October 1995 (the date the requirements of EO 12958 went into effect) and reflects the current system of conveying declassification instructions. The source may have been classified prior to 14 October 95 under the former system in which the use of the term Originating Agency Determination Required (OADR) was often used. Or, the source may have been classified after 14 October 95 but still reflects the former system of conveying declassification instructions. Even in cases in which only one source document is used, and often in cases in which several sources are used, different declassification instructions may apply to the various items of information in the document being created. To ensure that all the information in the document is protected for as long as necessary, the most restrictive declassification instruction that applies to any of the information in the document will be placed on the "Declassify on" line. The term "most restrictive" means the latest date or event, or the date or event furthest in the future. Throughout this regulation the term "OADR" is used strictly because there are documents out there with this term. The term "OADR" is no longer authorized.

(1) If all the information in the document has the same declassification instruction (i.e. same date, event, or exemption category or categories), and that instruction is an allowable option under the new policy contained in EO 12958 as stated in this regulation, place that instruction on the "Declassify on" line. The allowable options are:

(a) A date or event for declassification within 10 years from original classification.

(b) An exemption category for information classified beyond 10 years (see paragraph 4–10a) such as "X4." When an exemption category is used, it may or may not be followed by a declassification date or event, depending upon whether the original classification authority has selected a declassification date or event.

(c) For documents that will be over 25 years old on 17 October 2001, and are contained in records that have been determined to have permanent historical value under Title 44 of the USC, an exemption category or categories as shown in chapter 3.

(d) The source may be marked with one of the indefinite markings used before the term "OADR" was authorized. In such cases, any information with indefinite declassification instructions will be treated as though it were marked as "OADR".

(e) There are two different lists of exemption categories. One list applies to information that requires classification for more than 10 years. This list is contained in paragraph 2–11c. The other list applies to information contained in the exempted file series list and that will be more than 25 years old by 17 October, 2001. That list is contained in paragraph 3–6e.

(2) If all the information in the document has been extracted from a document created before 14 October 1995 (the effective date of EO 12958) and was marked "OADR", place the statement "Source marked OADR" on the "Declassify on" line, followed by the date of the document after the words "Date of Source". For example, a derivative classifier extracts classified information from a document dated 3 June 1992 and marked "OADR". The newly created document containing that extract will be marked, "Declassify on: Source marked OADR; Date of Source: 3 June 1992." When using several sources of information marked "OADR", the "Date of Source" line will reflect the most recent date (the document with the latest date). For example, one source is dated 2 August 1993 and one is dated 1 September 1995. In this case, the newly created derivatively classified document will be marked:

Derived from: Multiple Sources

Declassify on: Source marked OADR;

Date of Source: 1 September 1995

(3) No matter what combination of indefinite declassification instructions and document dates used as sources to derivatively classify the document, the document originator will only select the source document with the most recent date and this will determine the date to place on the "Date of Source" line. Follow this policy for all cases involving information classified under previous Executive Orders that contain indefinite declassification instructions. Follow this policy for all cases involving information extracted from a document created after 14 October 1995 that was mistakenly marked as OADR. Where practical and feasible, notify the originator of that mistakenly marked document of the outdated declassification instructions and obtain the current correct markings.

(4) If the document is classified by more than one source ("multiple sources") and different declassification instructions apply, the derivative classifier will place the most restrictive declassification instruction on the "Declassify on" line. The most restrictive declassification instruction is the date or event that will occur farthest in the future (the longest date from now). The following applies:

(a) If declassification dates are specified for all of the sources of information used in the document, place the latest date (date farthest in the future) on the "Declassify on" line. For example, in creating a new document, information is extracted from documents marked for declassification on 20 March 1998, 1 June 2002, and 3 April 2009. The newly created document will be marked: "Declassify on: 3 April 2009".

(b) If the sources of classification are a combination of a date or dates with an event or events, the declassification instruction will reflect whichever date and event occurs later (date or event farthest in the future). If the date of the event(s) is unknown, the declassification instruction will reflect the most restrictive date and latest occurrence of the event(s). For example, one source specifies a declassification date of 11 November 2011, and the other a declassification event upon execution of operations. In this case, the document will be marked: "Declassify on: 11 November 2011 or execution of operations, whichever is later".

(c) If any of the information in the document does not have a specific date or event for declassification, the originator of the derivatively classified document will apply the most restrictive declassification instruction, according to the following:

1. When using information classified under a previous EO, any information with an indefinite declassification (such as Group 3 or OADR) is treated as if it were marked "OADR" and marked as specified in paragraph 4–9b.

2. When using information from sources marked with the current EO 12958 exemption markings (X1 through X8), the "Declassify on" line will be marked with all exemptions that apply to all sources used. For example, if one source cited "X2", another cited "X3" and the third cited "X5", the Declassify on line would read: "Declassify on: X2,3,5". The most recent date will be used on the "Date of Source" line. For example, a derivatively classified document that uses three sources with the latest source dated 10 February 1996, will be marked:

Derived from: Multiple Sources

Declassify on: Sources marked X2,3,5

Date of Source: 10 February 1996

3. When using one or more sources marked with an indefinite declassification from a previous Executive Order (such as OADR) as well as one or more sources marked with the current EO 12958 exemption markings (X1 through X8), the "Declassify on" line will cite the exemption category or exemption categories as well as "source marked OADR". The "Date of Source" line will cite the date of the most recent source. For example, a derivatively classified document that uses the three sources mentioned in subparagraph (2), above, and also uses a source dated 1 September

1995 and marked OADR will be marked: Derived from: Multiple Sources Declassify on: Sources marked X2,3,5 and OADR Date of Source: 10 February 1996

4. The above rules apply to derivatively classified documents when a combination of original classification and derivative sources are used. The term "sources" as used above also includes the classification guides or guidance supplied by the original classifier.

5. With sources having a combination of differing declassification instructions, it is important to determine which is the most restrictive. The most restrictive marking will always be used. This rule applies for all derivative classifications including those in which there is a combination of derivative sources and original classification. A marking that does not provide a definite declassification date will always be considered more restrictive than one with a specific date. For instance, a document that is classified by two sources, one dated 19 August 1994 and marked "OADR" and the other dated 10 December 1995 and marked "Declassify on: 24 May 2004", will be marked: "Declassify on: Source marked "OADR", Date of Source: 19 August 1994". See subportion (3) directly above for an example of a case in which one source is marked OADR and the other is marked with one or more of the exemption categories (X1 through X8) of Executive Order 12958.

#### 4-11. Sources that were created prior to 1976

Chapter 3 provides the policy for marking information contained in records that will be more than 25 years old on 17 October 2001, and have been determined to have permanent historical value under title 44, USC. In summary, under EO 13142, amendment to EO 12958, section 3.4, information more than 25 years old by 17 October 2001, and that is contained in records that have been determined to have permanent historical value under title 44, USC will be automatically declassified starting on 17 October 2001, unless that information is exempted from declassification. The exemption categories, required markings, and the DA policy for handling this program are discussed in chapter 3 of this regulation. This section is not intended to prescribe the policy for addressing the review of that information. That policy is contained in chapter 3. This section prescribes the policy to follow when material, that will be over 25 years old by 17 October 2001, is used as the source for derivatively classifying a newly created document. Commands will consult AR 25–400–2 and local records managers for advice on what constitutes a file determined to have permanent historical value under Title 44, USC. In creating new documents using the old sources that will be over 25 years on 17 October 2001, it will make a difference whether or not the information has already been reviewed to determine if it is in a record that has been determined to have permanent historical value and whether or not it has been reviewed to determine if it will be declassified or exempted from automatic declassification. There are three possible options:

*a.* The information is determined to be of permanent historical value under title 44, USC, has been reviewed for continued classification, and qualifies under one or more of the exemptions listed in paragraph 3–6e of this regulation (section 3.4 of EO 12958). If it qualifies for exemption, the exemption category and the future date or event for declassification (if one applies) will be shown on the document, file, or record. When one of these documents is used as a source in classifying a derivatively classified newly created document, use the term shown on the document or record that was applied when the information was reviewed. That term will be "25X" followed by the appropriate exemption category that pertains to information exempted from declassification at 25 years and state the new declassification date or event, if one has been determined. For example, "25X3(31 December 2015)" if the information is exempted because it reveals information that would impair U.S. cryptologic systems and now has been determined to be declassification. For example, "25X1" if the information would reveal the identity of a human intelligence source.

*b*. The information is contained in a record that has been determined to have permanent historical value under title 44, USC, has been reviewed, and has been determined to not qualify for exemption. This information will have been marked with a declassification date or event on or before 17 October 2001. This date or event will be used as declassification instructions.

c. The information is either in a record that has been determined to not have permanent historical value under title 44 USC; or is in a record that has been determined to have permanent historical value under title 44 USC but has not yet been reviewed for declassification. This information would be subject to declassification 25 years from the date of its origin. Thus, the date of the source document will be placed, as the following, for declassification instructions: Source marked OADR

Date of Source:(fill in applicable date)

#### 4-12. Warning notices

In certain circumstances, warning notices will be required if the document contains certain categories of information for which the notice applies. In addition to the notices listed below, other notices may be required by other DA regulations. Unless another regulation or authorized administrative publication prescribes different placement, these notices will be placed on the cover (or first page where there is no cover) of the document.

a. Restricted Data (RD). Documents containing RD will be marked: "RESTRICTED DATA" THIS MATERIAL

CONTAINS RESTRICTED DATA AS DEFINED IN THE ATOMIC ENERGY ACT OF 1954. UNAUTHORIZED DISCLOSURE SUBJECT TO ADMINISTRATIVE AND CRIMINAL SANCTIONS.

b. Formerly Restricted Data (FRD). Documents containing FRD, but no Restricted Data, will be marked: "FORMERLY RESTRICTED DATA" "Unauthorized disclosure subject to administrative and criminal sanctions. Handle as Restricted Data in foreign dissemination. Section 144.b, Atomic Energy Act, 1954."

c. Critical Nuclear Weapons Design Information (CNWDI). Messages containing CNWDI will be marked at the beginning of the text as "RD CNWDI." Documents containing CNWDI will be marked: "Critical Nuclear Weapons Design Information DOD Directive 5210.2 applies"

*d. Intelligence Information.* The policy on the control, dissemination, and marking of warning notices concerning intelligence information is contained in appendix D. Placement of these intelligence control markings will follow the same policy as stated in appendix D.

e. COMSEC Material. The following marking will be placed on COMSEC documents before release to contractors: "COMSEC Material – Access by Contractor Personnel Restricted to U.S. Citizens Holding Final Government Clearance."

*f*. Reproduction Notices. Classified information that is subject to dissemination or reproduction limitations will be marked with notices that say, in essence, the following: "Reproduction requires approval of originator or higher DOD authority of the originator". "Further dissemination only as directed by (insert appropriate office or official) or higher DOD authority"

g. Special Access Programs (SAPs) Documents. Special Access Programs documents may be identified with the phrase "Special Access Required" and the assigned nickname, codeword, trigraph, or digraph. AR 380–381 contains the Department of the Army policy on marking SAPs material. See appendix I for further information.

*h.* DODD 5230.24 requires distribution statements to be placed on technical documents, both classified and unclassified. These statements facilitate control, distribution and release of these documents without the need to repeatedly refer questions to the originating activity. The originating office may, of course, make case–by–case exceptions to distribution limitations imposed by the statements. Distribution statements on technical documents will be marked with notices that say, in essence, the following:

(1) Distribution Statement A — Approved for public release; distribution is unlimited.

(2) Distribution Statement B — Distribution authorized to U.S. Government agencies only; [reason]; [date]. Other requests for this document shall be referred to [controlling DOD office].

(3) Distribution Statement C — Distribution authorized to US Government agencies and their contractors; [reason]; [date]. Other requests for this document shall be referred to [controlling DOD office].

(4) Distribution Statement D — Distribution authorized to the DOD and US DOD contractors only; [reason]; [date]. Other requests for this document shall be referred to [controlling DOD office].

(5) Distribution Statement E — Distribution authorized to DOD Components only; [reason]; [date]. Other requests for this document shall be referred to [controlling DOD office].

(6) Distribution Statement F — Further distribution only as directed by [controlling DOD office] or higher DoD authority; [date].

(7) Distribution Statement X — Distribution authorized to US Government agencies and private individuals or enterprises eligible to obtain export-controlled technical data in accordance with regulations implementing 10 USC 140c; [date]. Other requests must be referred to [controlling DOD office].

*i.* Documents containing information provided by a foreign government. See section VI of this chapter for complete policy on marking foreign government information in classified DA documents. U.S. classified documents that contain extracts of information provided by a foreign government will be marked with the following warning notice: "FOREIGN GOVERNMENT INFORMATION"

*j.* Documents containing information provided by a foreign government or international organization. See section VII of this chapter for complete policy on marking information provided by a foreign government or international organization. Examples of an international organization are the United Nations (UN) and the North Atlantic Treaty Organization (NATO). The following example pertains to NATO. The same policy applies to any other international organization by replacing the word "NATO" with the appropriate name or abbreviation for that organization. DA classified documents that contain extracts of NATO classified information will bear a marking substantially as follows: "THIS DOCUMENT CONTAINS NATO CLASSIFIED INFORMATION"

*k*. The following warning notice must appear on all U.S. Government owned or operated automated information systems:

"THIS IS A DOD COMPUTER SYSTEM. BEFORE PROCESSING CLASSIFIED INFORMATION, CHECK THE SECURITY ACCREDITATION LEVEL OF THIS SYSTEM. DO NOT PROCESS, STORE, OR TRANSMIT INFOR-MATION CLASSIFIED ABOVE THE ACCREDITATION LEVEL OF THIS SYSTEM. THIS COMPUTER SYS-TEM, INCLUDING ALL RELATED EQUIPMENT, NETWORKS, AND NETWORK DEVICES (INCLUDES INTERNET ACCESS), ARE PROVIDED ONLY FOR AUTHORIZED U.S. GOVERNMENT USE. DOD COM-PUTER SYSTEMS MAY BE MONITORED, FOR ALL LAWFUL PURPOSES, INCLUDING TO ENSURE THEIR USE IS AUTHORIZED, FOR MANAGEMENT OF THE SYSTEM, TO FACILITATE PROTECTION AGAINST UNAUTHORIZED ACCESS, AND TO VERIFY SECURITY PROCEDURES, SURVIVABILITY, AND OPERA-TIONAL SECURITY. MONITORING INCLUDES, BUT IS NOT LIMITED TO, ACTIVE ATTACKS BY AU-THORIZED DOD ENTITIES TO TEST OR VERIFY THE SECURITY OF THIS SYSTEM, DURING MONITORING, INFORMATION MAY BE EXAMINED, RECORDED, COPIED AND USED FOR AUTHORIZED PURPOSES. ALL INFORMATION, INCLUDING PERSONAL INFORMATION, PLACED OR SENT OVER THIS SYSTEM MAY BE MONITORED. USE OF THIS DOD SYSTEM, AUTHORIZED OR UNAUTHORIZED, CON-STITUTES CONSENT TO MONITORING. UNAUTHORIZED USE OF THIS DOD COMPUTER SYSTEM MAY SUBJECT YOU TO CRIMINAL PROSECUTION. EVIDENCE OF UNAUTHORIZED USE COLLECTED DURING MONITORING MAY BE USED FOR ADMINISTRATIVE, CRIMINAL, OR OTHER ADVERSE ACTION. USE OF THIS SYSTEM CONSTITUTES CONSENT TO MONITORING FOR ALL LAWFUL PURPOSES."

*l.* Other Warning Notices. Subparagraphs a through k above represent the most commonly used warning notices. They do not necessarily represent the only warning notices. There is nothing in this regulation that prohibits other authorized warning notices from being applied to classified documents. Where other regulations authorize and require special warning notices, they may be applied to DA classified documents.

#### 4–13. Obsolete Restrictions and Control Markings

*a.* The following control markings are obsolete and will not be used, in accordance with the following guidelines: (1) *WNINTEL and NOCONTRACT*. The control markings, Warning Notice – Intelligence Sources or Methods Involved (WNINTEL), and NOT RELEASABLE TO CONTRACTORS/CONSULTANTS (abbreviated NOCONTRACT or NC) were rendered obsolete effective 12 April 1995. No permission of the originator is required to release, in accordance with this directive, material marked WNINTEL. Holders of documents prior to 12 April 1995 bearing the NOCONTRACT marking should apply the policies and procedures contained in DCID 1/7, section 6.1, for possible release of such documents.

(2) *Remarking*. Remarking of material bearing the WNINTEL, or NOCONTRACT, control marking is not required; however, holders of material bearing these markings may line through or otherwise remove the marking(s) from documents or other material.

(3) Obsolete markings. Other obsolete markings include: WARNING NOTICE–INTELLIGENCE SOURCES OR METHODS INVOLVED, WARNING NOTICE–SENSITIVE SOURCES AND METHODS INVOLVED, WARNING NOTICE–INTELLIGENCE SOURCES AND METHODS INVOLVED, WARNING NOTICE–SENSITIVE INTELLI-GENCE SOURCES AND METHODS INVOLVED, CONTROLLED DISSEM, NSC PARTICIPATING AGENCIES ONLY, INTEL COMPONENTS ONLY, LIMITED, CONTINUED CONTROL, NO DISSEM ABROAD, BACK-GROUND USE ONLY, USIB ONLY, NFIB ONLY.

*b*. Questions with respect to current applications of all control markings authorized by earlier directives on the dissemination and control of intelligence and used on documents issued prior to the effective date of DCID 1/7, 30 June 1998, should be referred to the agency or department originating the intelligence so marked.

#### 4–14. Downgrading instructions

Downgrading instructions are not required for every classified document, but they must be placed on the face of each document to which they apply. When the original classification authority has determined that a document will be downgraded to a lower classification upon the passage of a date or event, the document will be marked: "Downgrade to SECRET on..." followed by the date or event, and/or "Downgrade to CONFIDENTIAL on..." followed by the date or event. This marking is placed immediately before the "Declassify on" line and is used in addition to, and not as a substitute for, declassification instructions.

#### 4–15. The Modern Army Recordkeeping System

*a. Purpose.* The purpose of Army recordkeeping is to properly manage information, from its creation through final disposition, according to federal laws and Army recordkeeping requirements. AR 25–400–2:

(1) Establishes the Modern Army Recordkeeping System (MARKS) as a portion of the Army Information Resources Management Program (AIRMP).

(2) Furnishes the only legal authority for destroying nonpermanent Army information.

(3) Provides life-cycle management instructions for the systematic identification, maintenance, storage, retirement, and destruction of Army information recorded on any medium (paper, microforms, electronic, or any other).

(4) Ensures that the commander and staff have the information needed to accomplish the mission; that they have it when and where they need it; that they have it in usable format; and that it is created, maintained, used, and disposed of at the least possible cost.

(5) Preserves those records needed to protect the rights and interests of the Army and its members and former members, and those that are of permanent value.

(6) Ensures records related to matters involved in administrative or legal proceedings will be retained until the staff judge advocate or legal adviser authorizes resumption of normal disposition.

(7) Provides for the systematic removal of less active records from office space to low-cost storage space.

b. Application. MARKS applies to-

(1) All unclassified Army records, including For Official Use Only (FOUO) and sensitive, regardless of media.

(2) All classified Army records through SECRET. Records that are TOP SECRET may be set up under MARKS, or in any manner that will make accountability and control easier. Regardless of the arrangement used, however, the disposition instructions in this regulation, and under MARKS, will be applied to TOP SECRET records.

## c. Principles.

(1) Within the MARKS system, records are identified and filed under the number of the primary directive that prescribes those records be created, maintained, and used.

(2) The file number is the key to MARKS. It identifies the records for filing and retrieval. MARKS numbers are made up by the prescribing directive number followed by an alpha suffix. See section II, appendix F, for the recordkeeping requirements of file titles and dispositions for records created and maintained under the purview of this regulation.

*d. Further guidance.* Further guidance on the management and disposition of files and records can be found in AR 25–400–2.

## Section II Marking Special Types of Documents

## 4-16. Documents with component parts

If a classified and/or sensitive document has components likely to be removed and used or maintained separately, each component will be marked as a separate document. Examples of components are annexes, appendices, major parts of a report, or reference charts. If the entire major component is UNCLASSIFIED, it can be marked on its face, top and bottom: "UNCLASSIFIED", and a statement added: "All portions of this (annex, appendix, etc.) are UNCLASSIFIED." No further markings are required on this type of component.

## 4–17. Transmittal documents

Transmittals are documents that have classified and/or sensitive documents enclosed with or attached to them. An example is a letter with classified enclosures or a document that is used to describe the transmission of classified equipment, documents, or other material. The transmittal document itself may contain information classified and/or sensitive the same or higher than the material transmitted. Often the transmittal document itself is UNCLASSIFIED or classified at a lower level than the material being transmitted or enclosed.

*a.* If the transmittal contains information classified and/or sensitive the same or higher than the documents being transmitted, the transmittal will be marked the same as any other classified and/or sensitive document.

*b*. If the information in the transmittal is UNCLASSIFIED or classified at a lower level than one or more of the documents being transmitted, the transmittal will be marked as follows:

(1) Mark the face of the transmittal conspicuously, top and bottom, in letters larger than the rest of the text, with the highest classification found in any of the enclosed documents being transmitted. For example, an UNCLASSIFIED transmittal that has one SECRET and two CONFIDENTIAL enclosures or attachments will be marked "SECRET".

(2) Mark the face of the transmittal to show its classification status when separated from the material being transmitted. For example, the following or similar statements apply: "UNCLASSIFIED WHEN SEPARATED FROM CLASSIFIED ENCLOSURES," "UNCLASSIFIED WHEN ATTACHMENT 3 IS REMOVED," "CONFIDENTIAL UPON REMOVAL OF ENCLOSURES," "REGRADED CONFIDENTIAL WHEN SEPARATED FROM ENCLO-SURES," etc.

c. Any special warning notices that apply to the transmittal or to the documents being transmitted will be placed on the face of the transmittal document. Transmittals that are classified standing alone will be marked the same as other classified documents. UNCLASSIFIED transmittals will not be portion marked. The marking of classification at the top and bottom of interior pages of an UNCLASSIFIED transmittal is not required, but is encouraged.

#### 4–18. Classification by compilation

When a document is classified and/or sensitive by compilation as discussed in paragraph 2–13, it will be marked as specified in paragraph 4–6d.

#### 4-19. Translations

Translations of U.S. classified and/or sensitive information into a foreign language, will be marked with the appropriate U.S. classification/sensitivity markings and the foreign language equivalent. Section VIII, of this chapter, contains a list of foreign language classifications. The translations will clearly show the United States as the country of origin.

## 4-20. Electronically transmitted messages

This section does not pertain to documents transmitted by facsimile (FAX) transmission. Classified and/or sensitive

electronically transmitted messages will be marked the same as any other classified and/or sensitive document, with the following special provisions:

a. The first item in the text of the message will be the overall classification/sensitivity.

*b*. For messages printed by an automated system, overall and page markings will be applied by that system, provided they stand out conspicuously from the text. In older systems, this may be achieved by surrounding the markings with asterisks, stars, or other symbols. See appendix E of this regulation and AR 380–19 for more guidance.

c. A properly completed "Classified by" or "Derived from" line, reason, declassification instructions, and downgrading instructions (when applicable), will be included in the last lines of the message. Inclusion of a "Classified by" or "Derived from" line in the message is a new requirement. Declassification and downgrading instructions will not be used for messages containing Restricted Data or Formerly Restricted Data (RD or FRD). The abbreviations "CLAS" for "Classified by", "DECL" for "Declassify on", "DERV" for "Derived from" and "DNG" for "Downgrade to" may be used on messages.

#### 4-21. Documents marked for training purposes

Documents that contain no classified and/or sensitive information, but are marked with classification/sensitivity markings for training purposes, will be marked to clearly show that they are actually UNCLASSIFIED. An appropriate statement will be placed on each page of the document, for example, "CLASSIFIED FOR TRAINING ONLY", "UNCLASSIFIED SAMPLE", or "CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY". The term "training purposes only" does not mean the sending of mock or fake classified and/or sensitive messages during field exercises and subsequently marking them for automatic declassification/destruction on ENDEX (end–of–exercise). It is used for the purpose of providing examples of how classified and/or sensitive markings are properly applied, such as handbooks and other like publications.

## 4-22. Files, folders, and groups of documents

Files, folders, and similar groups of documents containing classified and/or sensitive information will be clearly marked as to the highest classification/sensitivity of information contained therein. The classification/sensitivity marking will be on the outside, front and back, and top and bottom, of the file or folder. Attaching a document cover sheet to the outside of the file or folder is acceptable in satisfying this requirement. When cover sheets are used, they will not be attached when the file is in a secure storage container. When cover sheets are removed, when the item is in secure storage, the file or folder must be marked to indicate the highest level of classified and/or sensitive information contained in the file.

#### 4-23. Printed documents produced by AIS equipment

There are no special provisions for documents produced by Automated Information Systems (AIS) which function as word processing systems (see appendix E of this regulation for further guidance). Documents produced on an AIS will be marked like other documents. For other AIS–generated documents, special exceptions may apply where the application of the marking requirements of this chapter are not feasible. These exceptions are:

*a.* Classification/sensitivity markings on interior pages of fan-folded printouts are required. These markings may be applied by the AIS equipment even though they may not meet the standard requirement of being conspicuous. At a minimum, the first page of the document and the back of the last page will be over-stamped, with the classification/ sensitivity, in letters larger than the print.

*b*. Special warning notices, identification of classification sources, and declassification (and downgrading, where applicable) instructions will either be marked on the cover of the document, or on the first page if there is no cover, or will be placed on a separate notice attached to the front of the document.

c. Pages or other portions of AIS printouts removed for separate use or maintenance, will be marked in the standard manner as individual documents.

#### Section III

## Marking Special Types of Material

#### 4–24. General policy

When classified and/or sensitive information is contained in AIS equipment, hardware, AIS media, or on film, tape, or other audio/visual media, or in another form not commonly thought of as a document, the marking provisions of this and other applicable regulations will be met in a way that is compatible with the type of material. The main concern is that holders and users of the material are clearly warned of the presence of classified and/or sensitive information needing protection. The information provided by the other markings required by this regulation will also be made available, either on the material or in a document or notice that accompanies it. Particular exceptions are noted below. The requirements of this chapter do not apply to the marking of security containers. The only markings allowed on security containers are those outlined in chapter 7 of this regulation.

## 4–25. Telephone or communications directories

Telephone or communications directory notice. Official U.S. Army telephone or communications directories will display the following notice on the front cover or prominently within the general information section: ATTENTION! DO NOT PROCESS, STORE, OR TRANSMIT CLASSIFIED INFORMATION ON NONSECURE TELECOMMUNICATIONS SYSTEMS. OFFICIAL DOD TELECOMMUNICATIONS SYSTEMS—INCLUDING TELEPHONES, FACSIMILE MACHINES, COMPUTER NETWORKS, AND MODEMS—ARE SUBJECT TO MONITORING FOR TELECOMMUNICATIONS SYSTEMS CONSTITUTES CONSENT TO INFORMATION SYSTEMS SECURITY MONITORING.

## 4-26. Blueprints, schematics, maps, and charts

Blueprints, engineering drawings, charts, maps, and similar items not contained in a classified and/or sensitive document will be marked with the overall highest classification/sensitivity of information contained therein. The classification/sensitivity marking will not be abbreviated, and will be conspicuous. The classification/sensitivity marking should be applied to the top and bottom of the material, if possible, or in some manner as to ensure the classification/sensitivity is readily known. The legend or title must also be marked to show its classification/sensitivity. An abbreviated marking in parentheses following the legend or title may be used. If the item is large enough that it is likely to be rolled or folded, the classification/sensitivity markings will be placed to be visible when the item is rolled or folded.

## 4-27. Photographs, negatives, and unprocessed film

*a.* Photographs and negatives will be marked with the overall highest classification/sensitivity of information contained thereon. Photographs will be marked on the face, if possible. If not possible to mark the face, the classification/sensitivity marking will be placed on the reverse side of the photograph. Other markings required by this regulation will be placed on photographs along with the classification/sensitivity marking, or will be included in accompanying documentation.

*b*. Roll negatives, positives, and unprocessed film, containing classified and/or sensitive information, will be marked with the overall highest classification/sensitivity of the information contained on the film. This marking will be placed either on the film itself, or on the canister, if one is used. If placed on the film itself, the marking will be placed at the beginning and end of the roll.

## 4-28. Slides and transparencies

*a.* Each slide or transparency will be marked with the highest level of classification/sensitivity contained on the slide or transparency, so that the marking is visible to personnel seeing the item projected, and to personnel physically holding the item. Each slide or transparency will have the classification/sensitivity and special warning notices (if any) marked on both the image area of the item and on the border, holder, or frame. Other required security markings (for example, classification authority and declassification instructions) may be placed in the image area; on the border, holder, or frame; or in a document or notice accompanying the item.

b. When a group of slides or transparencies is used together and maintained together as a set, for example, in the case of a set of briefing slides, each slide or transparency will have the classification/sensitivity marking and special warning notices (if any) on it, applied as described in subparagraph a, above. The other required security markings may be placed on the first slide or transparency (or the border, holder, or frame for the first item) in the set; these markings are not required to be placed on the other slides or transparencies in the set.

## 4-29. Motion picture films and videotapes

Classified and/or sensitive motion picture films and videotapes must be marked with their classification/sensitivity and warning notices (if any) at the beginning and end of the played or projected portion. Other required security markings will be placed at the beginning of the projected or played portion. Reels and cassettes will be marked with the overall classification/sensitivity of the item and kept in containers marked with the classification/sensitivity and other required security markings.

## 4–30. Sound recordings

Sound recordings containing classified and/or sensitive information will have an audible statement of their classification/sensitivity and warning notices (if any) at the beginning and end of the recording. Reels or cassettes will be marked with the overall classification/sensitivity of the item and kept in containers marked with the classification/ sensitivity and other required security markings of the item. Where this is not possible, this information will be recorded on documentation accompanying the item.

## 4-31. Microfilms and microfiche

Microfilm, microfiche, and similar media, will be marked so that the overall classification/sensitivity and warning notices (if any) are shown in the image area and can be read or copied as part of the item. Such items also will be marked with the classification/sensitivity markings and at least an abbreviation of any warning notices applied in such

a way as to be visible to the unaided (naked) eye. Other required security markings will be either placed on the item or included in an accompanying document or notice.

#### 4-32. Removable AIS storage media

*a.* Further details on the policy to protect and mark classified and/or sensitive information stored on AIS media is contained in AR 380–19, and appendix E, of this regulation. The following minimum standards are required for removable AIS storage media. Removable AIS storage media include magnetic tape reels, disk packs, diskettes, CD–ROMs, removable hard drives, disk cartridges, optical disks, paper tape reels, magnetic cards, tape cassettes and micro–cassettes, and any other device on which data is stored, and which normally is removable from the system by the user or operator. All such devices, containing classified and/or sensitive information, will be conspicuously marked with the highest level of classification/sensitivity stored on the device, and with any warning notices that may apply to the information. Other required markings, for example, classification authority and declassification instructions, will be marked on the outside of the device. An exception is, if classified and/or sensitive documents or files are prepared on a word processor and are stored on a floppy disk, and each document or file bears its own classification authority and declassification instructions, as entered with the word processor, the disk does not need to be marked with this information. If the required information is not stored in readily accessible format on the media, it must be marked on the outside of the media (for example, with a sticker or tag) or placed on documentation kept with the media.

b. One of the misconceptions many AIS users have, is in the use of media and AIS equipment of differing classifications. If a user places a removable medium, such as a diskette, that is marked and contains only unclassified data, into a classified AIS, they assume that as long as they don't save classified information on it, then their diskette is still unclassified. By the AIS merely accessing the diskette there arises the possibility of classified information being written to the diskette. Classified information is most often unknowingly transferred to the diskette in this manner, in one of the following three methods: "lost clusters;" "unallocated space;" and "slack space."

(1) A cluster is a group of disk sectors. The operating system assigns a unique number to each cluster and then keeps track of files according to which clusters they use. Occasionally, the operating system marks a cluster as being used even though it is not assigned to any file. This is called a lost cluster. You can free up disk space by reassigning lost clusters, but you should first make sure that the clusters do not, in fact, contain valuable data. In DOS and Windows, you can find lost clusters with the ScanDisk utility. DOS and Windows keep track of clusters with the File Allocation Table (FAT). The size of each cluster depends on the disk's partition size.

(2) Unallocated space is space on a storage medium that the Disk Operating System (DOS) regards as available for use when needed. As far as DOS is concerned, the space is empty. In actuality the space may contain deleted files, which are not gone until they are overwritten, or partial fragments of old files to include old file slack.

(3) The DOS and Windows file systems use fixed-size clusters. DOS and older Windows systems use a 16-bit file allocation table (FAT), which results in very large cluster sizes for large partitions. Even if the actual data being stored requires less storage than the cluster size, an entire cluster is reserved for the file. Unless a file is EXACTLY one or more clusters in length, which is unlikely, there will always be space between the file's "End of File" marker and the end of the cluster associated with that file. Any data that is located in that space, or the empty space itself, is referred to as file slack. An unintended consequence of the way that DOS handles its buffers, is that the buffer's contents may be "dumped" into file slack or "slack space". When files are written over a used diskette, one that had files on it but they were erased but not fully reformatted, the new files do not overwrite all the preceding data, therefore there is still readable data in the files slack space area. Also, whenever a file is used on an AIS, it acquires slack space from the hard drive. This can inadvertently be transferred back onto the diskette. Normal text editing software cannot read this area, however there are programs available to accomplish this. For example, if the partition size is 2 GB, each cluster will be 32 K. Even if a file requires only 4 K, the entire 32 K will be allocated, resulting in 28 K of slack space. Windows 95 (OSR 2), Windows 98, and Windows NT 4.0 resolve this problem by using a 32-bit FAT (FAT32) that supports cluster sizes as small as 4 K for very large partitions, however this still does not preclude the possibility of data in slack space.

c. This area has always been a sore spot for DA organizations. Material used or produced on, for, and by, an AIS, comprises the greater majority of an organization's day-to-day operations. For details on the security procedures for documents created for and on an AIS, web-based display, and common AIS security related misconceptions, see appendix E of this regulation.

#### 4–33. Fixed and internal AIS storage media

Systems Managers will make sure that AIS, including word processing systems, provide for classification/sensitivity designation of data stored in internal memory or maintained on fixed storage media.

#### 4–34. Standard form (SF) labels

a. If not marked otherwise, items covered by this section will be marked with the following labels:

- (1) SF 706 (TOP SECRET Label for ADP Media).
- (2) SF 707 (SECRET Label for ADP Media).
- (3) SF 708 (CONFIDENTIAL Label for ADP Media).
(4) SF 710 - (UNCLASSIFIED Label for ADP Media).

- (5) SF 711 (Data Descriptor Label for ADP Media).
- (6) SF 712 (CLASSIFIED SCI Label).

*b.* SF 710 is not required to be used in environments where there is no classified information created or used. In environments where classified information is created or used, SF 710 labels will be used to identify unclassified media from the classified AIS removable storage media. Media that is classified at the SCI level will have both the SF 711 and the SF 712 labels attached. The SF 711 will indicate the security classification of the data.

#### Section IV Changes in Markings

#### 4-35. Downgrading and declassification in accordance with markings

*a.* When a document is marked for downgrading or declassification on a date or event, the downgrading or declassification is automatic at the specified time unless notification to the contrary is received from the originator, the original classification authority, or other appropriate authority. There is no requirement to refer the document to the originator on or before the date or event for a downgrading or declassification decision. If a holder of the document has reason to believe it should not be downgraded or declassified, the holder will notify the originator and OCA (if known) of the information.

b. When a document is declassified in accordance with its markings, the overall and page markings will be canceled, if practical. If it is not practical to cancel the marking on each page on a bulky document, the page marking will be canceled, as a minimum, on the first page, on the cover (if there is one), title page (if there is one), and on any interior page that is copied or removed from the document.

c. If a document is downgraded (assigned a lower level of classification/sensitivity) in accordance with its markings, the old classification/sensitivity markings will be canceled and substituted with the new, lower classification/sensitivity markings will be applied under the same policy as stated in subparagraph b, above, for declassification markings.

#### 4-36. Downgrading and declassification earlier than scheduled

If a document is declassified or downgraded earlier than indicated by its markings, the rules for remarking as stated in paragraph 4–35 will be followed. In addition, the date of remarking and the authority for the action will be placed on the face of the document. The date of the remarking is considered the date that the remarking was authorized (for instance, date of the notice to remark) or, if no date is specified, the date that the material was physically remarked. The authority for the action is the identity of the OCA, or the designated declassification authority, who directed the action and the identification of the correspondence, classification guide, or other instruction or notification which required it.

#### 4-37. Upgrading

If a document is upgraded (assigned a higher level of classification/sensitivity), all classification/sensitivity markings affected by the upgrading will be changed to the new markings, without exception. In addition, the date of the remarking and the authority for the action will be placed on the face of the document. The date of the remarking is considered the date that the remarking was authorized (for instance, "Date of Notice to Remark") or, if no date is specified, the date that the material was physically remarked. The authority for the action is the identity of the OCA who directed and action and the identification of the correspondence, classification guide, or other instruction or notification which required it.

#### 4-38. Posted notice on bulk quantities of material

When the volume of material involved in a remarking action is so large that individually remarking each document is determined by the commander to cause serious interference with operations, the custodian will attach a notice to the storage unit providing the required information as stated in paragraphs 4-35, 4-36, and 4-37. When individual documents are permanently removed from the storage unit, they must be individually marked as required in paragraphs 4-35, 4-36, and 4-37. If documents are removed to be transferred in bulk to another storage unit, they need not be remarked if the new storage unit also has a proper notice posted.

#### 4-39. Extensions of duration of classification

If information has been marked for declassification on a specific date or event and the duration of classification is subsequently extended, the "Declassify on" line will be changed to show the new declassification instructions, the identity of the OCA authorizing the extension, and the date of the authorizing action. For example, "Declassify on: Classification extended on 1 Dec 2005 to 1 Dec 2015 by LTG Soldier, Commanding General, US Army Classification Command."

#### Section V Remarking and Using Old Classified Material

#### 4-40. Old markings

Some classified and/or sensitive documents are still in use which were marked, as specified by earlier versions of this regulation, based upon an earlier EO. There is no requirement to remark this material with the new markings specified by this regulation and EO 12958. This material will not be remarked unless specific instructions are received from the original classification authority. If the material is marked for automatic downgrading or declassification on a specific date or event, it will be remarked as specified in paragraph 4–34. If the document does not specify a specific date or event for downgrading or declassification (for example, if it is marked "Declassify on: OADR"), it will not be remarked until it reaches 25 years. Chapter 3 contains the policy on the marking of information over 25 years old.

#### 4-41. Earlier declassification and extension of classification

The requirements for declassification, exemptions from automatic declassification, and extensions of original classification dates apply to all classified information, including that classified under previous Executive Orders. Specific policy on classification and marking of information classified by previous Executive Orders is addressed in this regulation.

#### Section VI

#### Safeguarding Joint Chiefs of Staff Papers

#### 4-42. General

This section prescribes responsibilities and establishes procedures to secure and distribute Joint Chiefs of Staff (JCS) papers within the Army.

#### 4-43. References

a. AR 380-10, Technology Transfer, Disclosure of Information and Contacts with Foreign Representatives.

b. JCS Policy Memorandum 39, Release Procedures for JCS Papers.

c. SF 135 (Records Transmittal and Receipt) and SF 135A (Records Transmittal and Receipt (Continuation).

#### 4-44. Responsibilities

a. In accordance with JCS Memorandum of Policy, the Chief of Staff (CSA), Army will distribute JCS papers or extracts of these papers:

(1) Within Department of the Army.

(2) To those agencies operating under the JCS for whom the Army is Executive Agent.

*b*. The Deputy Chief of Staff for Operations and Plans (DCSOPS) will ensure that the Joint Action Control Office, Office of the Deputy Chief of Staff for Operations and Plans (ODCSOPS) performs the following functions:

(1) Control and distribution of JCS papers within DA.

- (2) Response to inquiries regarding distribution of JCS documents from:
- (a) Agencies or commands.
- (b) Organizations for which the Army is Executive Agent.
- c. MACOM commanders and heads of headquarters staff agencies will ensure that:
- (1) JCS papers are properly safeguarded.
- (2) Requests for JCS papers are forwarded to HQDA (DAMO-ZJC), WASH DC 20310-0421.

#### 4-45. Requirements

a. JCS papers, including extractions from such papers, will be safeguarded in accordance with this regulation. b. JCS papers will be safeguarded to ensure that release is not granted recipients not authorized as outlined in paragraph 4-46.

#### 4-46. Access

Access to JCS papers will be limited to persons who have:

a. Appropriate security clearances, and

b. Official duties that require knowledge or possession of the JCS papers (i.e. need-to-know).

#### 4-47. Familiarization requirements

- a. The following personnel will become familiar with the provisions of this chapter:
- (1) Those assigned to or employed by DA or any organization for which the Army is Executive Agent.
- (2) Those who have access to JCS papers.

b. Personnel, who have actual or potential access to JCS papers, will be briefed on their responsibilities regarding JCS papers at the time of initial assignment, and annually thereafter.

#### 4–48. Distribution of JCS documents

*a.* JCS papers will be distributed only within the Army staff and to commanders of Army field and component commands and agencies. No other distribution will be made unless approval has been granted by the Joint Secretariat, Organization of the Joint Chiefs of Staff.

*b.* Other Army activities that require information from JCS papers will be furnished abstracts when possible rather than complete documents. Such information will be phrased so that it can be clearly understood. For example, a decision of JCS should be referred to by such phrases as "On 20 August (YYYY), the Joint Chiefs of Staff approved/ requested/directed (fill–in the information)."

c. JCS papers that require a decision by the JCS will not be distributed outside the Army staff until a decision has been published.

*d.* JCS papers, that must be approved by the President or Secretary of Defense, will not be distributed outside the Army Staff until approval is obtained. After these papers have been approved, the Joint Chief Secretariat (OJCS) will officially notify the holders.

#### 4-49. Release and Distribution of Joint Strategic Planning System Documents

Release and distribution of Joint Strategic Planning System (JSPS) documents will be the same as for other JCS papers except for release to Service schools and colleges. However, JSPS documents are subject to the following additional controls:

*a*. The Joint Action Control Office, ODCSOPS, will request a semiannual sighting report on JSPS documents. The report will include outstanding copies or sections of the current edition of separately bound portions classified SECRET or above.

*b.* Sections or extracts of JSPS documents may be reproduced or distributed to Army activities that require this information. Information should be issued in this form when possible, rather than in the form of entire documents. *c.* JSPS documents will be accounted for by a continuous chain of receipts.

*d.* The CSA may distribute JSPS documents, except Joint Strategic Capabilities Plan (JSCP), to service schools and

colleges for the following purposes:

(1) To support the curriculum through controlled classroom use.

(2) For use in curriculum-related, directed research by U.S. personnel from organizations responsive to the JCS or agencies that have received the documents.

*e.* JSPS documents may not be reproduced or automatically distributed to faculty or students of service schools and colleges. Access to the JSCP and Joint Strategic Planning Document Supporting Analyses (JSPDSA) will be further restricted to those members of the faculty and U.S. student body with an official duty requirement. Faculty or students in service schools and colleges conducting independent work, not in response to a JCS tasking, are not considered to have an official duty requirement.

#### 4-50. Release and Distribution of Joint Operation Planning System Documents

Release and distribution of Joint Operation Planning System (JOPS) documents will be as follows:

a. Distribution or circulation will be limited to Army agencies directly concerned in supporting:

(1) Operations plans prepared by the commanders of unified and specified commands.

- (2) Plans written to support these commands.
- b. Distribution will not be made to Army service schools for:
- (1) Current and superseded operations plans.
- (2) Related documents prepared by supported, supporting, and subordinate commanders.

#### 4–51. Release of JCS information to Army Service Schools

a. JCS papers are not normally distributed to schools. However, documents may be requested on a case-by-case basis:

(1) To support the curriculum of U.S. students (controlled classroom use).

(2) For use in directed-institutional research.

b. Fully justified requests for release of information will be submitted through command channels to HQDA (DAMO-ZCJ), WASH DC 20310-0421.

c. Documents or information furnished to the schools will be controlled to ensure that access is limited to U.S. personnel with proper security clearances and need-to-know. Foreign nationals attending the schools may require access. If so, this fact will be specified in the request for release, along with full justification, as outlined in AR 380-10.

d. Release of JSPS and JOPS documents will be as outlined in paragraphs 4-49 and 4-50, above.

#### 4-52. Release of information to organizations outside DA

JCS papers or extracts thereof will not be distributed outside of DA. Exceptions to this policy will be processed as follows:

- a. Release of JCS documents or information extracted therefrom must be approved beforehand by JCS.
- b. Each request will be considered on a case-by-case basis.
- c. Requesting organizations will submit full justification to:
- (1) The Army agency with which they normally maintain contact;
- (2) The nearest Army area command or agency; or
- (3) The cognizant Army staff agency for validation.

d. These requests will be forwarded with recommendations to HQDA (DAMO-ZCJ), WASH DC 20310-0421 for action.

*e*. The numbers of JCS green papers (JCS 0000/000) less than 10 years old will not be referenced in the text of any extract for release to agencies outside DA.

*f*. Nonconcurrent JCS documents interfiled in nonconcurrent DA records may be transferred to records centers according to established requirements. In this case, SF 135 will stipulate that access to JCS documents attached by individuals or agencies not under the jurisdiction of the JCS or DA will be permitted only with the approval of the JCS.

#### 4-53. Reproduction of JCS documents

JCS documents will not be reproduced except as authorized under paragraph 4-49b.

#### Section VII

#### **Foreign Government Information**

#### 4-54. Policy

Each foreign government has its own policy on what information is classified and/or sensitive and for how long it will remain classified and/or sensitive. The classification/sensitivity and declassification policy of another government cannot and, in many cases, does not parallel that of the United States. When Foreign Government Information (FGI) is disclosed to the United States, it is done so with the understanding that the information will be protected and will not be declassified or released to another nation or to the public without the express permission of the originating government. This applies to both foreign government documents as well as in situations in which foreign government information is incorporated in a classified U.S. document. It is, therefore, important to identify foreign government information that is contained in U.S. classified documents. Throughout this regulation, when the term "foreign government" is used, the policy also applies to international pact organizations (for instance NATO), unless otherwise specified.

#### 4-55. Equivalent U.S. classification designations

Foreign classification designations generally parallel U.S. classification designations. The exception is that many foreign governments have a fourth (lowest) classification level called "RESTRICTED". A table of the equivalent foreign and international pact organization security classifications is contained in section VIII.

#### 4–56. Marking NATO documents

Classified documents originated by NATO, if not already marked with the appropriate classification in English, will be so marked. Other markings, such as the "Classified by/Derived from" line and declassification instructions will not be placed on documents originated by NATO. Documents originated by NATO that are marked "RESTRICTED" will be marked with the following additional notation: "TO BE SAFEGUARDED IN ACCORDANCE WITH USSAN INSTRUCTION 1–69". The USSAN Instruction 1–69 is implemented within the Department of the Army in AR 380–15 (AR 380–15 is classified NATO CONFIDENTIAL).

#### 4-57. Marking Other Foreign Government Documents

*a.* If the security classification designation of the foreign government document is shown in English, no other classification marking will be applied. If the foreign classification designation is not shown in English, the equivalent overall U.S. classification designation (TOP SECRET, SECRET, or CONFIDENTIAL) will be marked conspicuously on the document. See figure 4–14 for English terms to use in marking classified NATO documents. When a foreign government document is marked with a classification designation having no U.S. equivalent, such as RESTRICTED, it will be marked as specified in paragraph b, below.

*b.* Most foreign governments use a fourth (lowest) classification designation. Such designations are or equate to the foreign classification RESTRICTED. If foreign government documents are marked with any of these classification designations, no other classification marking will be applied. For such cases, the notation: "THIS CLASSIFIED MATERIAL IS TO BE SAFEGUARDED IN ACCORDANCE WITH AR 380–5, DOD Directive 5200.1–R, OR THE

NISPOM". An alternative authorized marking is "THIS CLASSIFIED MATERIAL IS TO BE SAFEGUARDED IN ACCORDANCE WITH DOD 5200.1-R OR THE NISPOM". When that is used, it will be understood that the governing regulation for DA commands is still AR 380-5.

c. Other marking requirements prescribed by this regulation for U.S. classified documents are not applicable to documents of foreign governments or international organizations of governments.

#### 4-58. Marking Foreign Government Information Provided in Confidence

Foreign documents containing FGI not classified by the foreign government but provided in confidence to the Department of the Army or any other element of the U.S. government or its contractors, will be classified if the information is covered under one or more or the reasons for classification listed in chapter 2. If it is deemed classified, it will be marked with the appropriate U.S. classification. If not, classification markings will not be applied to the document. If it is not otherwise obvious that the document was provided in confidence, DA commands can place the notation "FOREIGN GOVERNMENT (or list the name of the country) INFORMATION PROVIDED IN CONFIDENCE". That notation is not a requirement but is a consideration for cases in which the document, or copies of the document, can leave the control of the personnel who were aware of the confidentiality under which the information was provided.

#### 4-59. Marking of Foreign Government Information in Department of the Army Documents

In addition to the other markings required in this regulation, the following markings will be used in classified DA documents containing FGI (see definition of "Foreign Government Information" in appendix J).

a. When used in a classified DA document, FGI must be marked to prevent premature declassification or unauthorized access by third country nationals. A DA document that contains FGI will be marked on the face of the document to indicate that FGI is contained therein. In most situations the document will be marked "THIS DOCU-MENT CONTAINS (insert name of country) INFORMATION". As an alternative-and to be used in cases in which identification of the country would provide additional classified information or in cases in which the foreign country does not wish to be identified-the document will be marked: "THIS DOCUMENT CONTAINS FOREIGN GOV-ERNMENT INFORMATION" or "FOREIGN GOVERNMENT INFORMATION". In those situations the record file copy of the document will contain the identification of the foreign government providing the information. A DA document containing NATO classified information will be marked on the face of the document, "THIS DOCUMENT CONTAINS NATO CLASSIFIED INFORMATION". The face of the document is considered to be the cover and title page, or the first page where there is no cover or title page. In addition, the portions will be marked to identify the classification level, including UNCLASSIFIED, and the country of origin. Examples of portion marking are as follows: TOP SECRET information from the United Kingdom (UK-TS), SECRET information from Germany (GE-S), NATO SECRET information (NATO-S) or (NS), CONFIDENTIAL information from France (FR-C), and UNCLASSIFIED information from Costa Rica (CR-U). In unusual situations in which the foreign government does not wish to be specifically identified, the portions will be marked "FGI" together with the appropriate classification; for example, (FGI-S). In those cases the originator will maintain the identity of the foreign country furnishing the information with the record file copy of the document. Except in those unusual situations in which the foreign country does not wish to be identified, the "Classified by" or "Derived from" line will identify the U.S. as well as foreign classification sources. In cases in which multiple U.S. and one or more foreign sources are used, the term "Multiple sources and the Government of (insert name of foreign country or countries)" will be used on the "Classified by" or "Derived from" line. In those cases in which identification of the foreign country furnishing the information would reveal additional classified information, or when the foreign country does not wish to be identified, the foreign sources of classification will be maintained with the record file copy of the document and the term "Foreign Government" will be shown in addition to the U.S. sources on the "Classified by" or "Derived from" line.

b. When foreign government RESTRICTED or NATO RESTRICTED information is included in an otherwise unclassified DA document, the document will be marked NATO RESTRICTED. All marking requirements for classified documents will apply, to include portion markings which will show the letter "R" after the abbreviation for the country that furnished the information (for example, RESTRICTED information furnished by Canada (CA–R); or (NATO–R) or (NR) for NATO RESTRICTED. Each page containing the RESTRICTED information would be so marked as "(insert name of country or international organization) RESTRICTED" or "THIS PAGE CONTAINS (insert name of country or international organization) RESTRICTED INFORMATION". In addition, the applicable notice from paragraph a., above, will be included on the face of the document.

c. The "Classified by" or "Derived from" line of a DA document that is classified only because it contains classified foreign government information will be completed as described in this chapter for derivative classifications and, therefore, the term "Derived from" will be used. The "Declassify on" line will read "Source not marked".

Sample of Marking an Originally Classified Document

# CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET

Subject: Sample of Marking an Originally Classified Document (U)

1. (C) This page is UNCLASSIFIED and is marked SECRET for training purposes only. IMPORTANT: You must be an Original Classification Authority to originally classify a document. A majority of cleared personnel who generate classified documents are NOT Original Classification Authorities (OCAs). To be an OCA you must be appointed by the Secretary of the Army or the DCSINT. In most cases, documents are classified based on a Security Classification Guide or guidance (that was approved by an OCA) or upon information taken out of a source document (or both). This sample memo only pertains to those relatively few cases in which an OCA is generating a document that the OCA is originally classifying.

2. (S) In this example, the OCA is an official that has been delegated Original Classification Authority for a particular program. The OCA must be identified (see paragraph 4-8a), the reason that the information is classified must be stated (see paragraph 4-9), and the OCA must indicate a date, event, or exemption category for declassification (see paragraph 4-9a). If there is more than one reason or exemption, then all must be listed. In this example, the OCA has selected reason 1.5(a) (military plans, weapons systems, or operations). Instead of a date or event for declassification within 10 years, the OCA has authorized exemption category X2 (reveal information that could assist in the development or use of weapons of mass destruction) that must be protected beyond 10 years.

I.M. Trying Director, Security Awareness

Classified by: LTG A. SECRET, Chief of Weapons Programs, Army Security Is Important Command Reason: 1.5(a) Declassify on: X2 Date of source: 26 March 1999

SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-1. Sample of Marking an Originally Classified Document

Sample of Marking a Classified Document that is Exempt from the 25-Year Automatic Declassification

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET

Subject: Sample of Marking a Classified Document that is Exempt from the 25 Year Automatic Declassification (U)

1. (C) This page is UNCLASSIFIED and is marked SECRET for training purposes only. Chapter 3 provides the policy for marking information contained in records that will be more than 25 years old on 17 April 2000 and have been determined to have permanent historical value under title 44, USC. In summary, under EO 12958 (section 3.4) information more than 25 years old by 17 April 2000 and that is contained in records that have been determined to have permanent historical value under title 44, USC. In summary, under EO 12958 (section 3.4) information more than 25 years old by 17 April 2000 and that is contained in records that have been determined to have permanent historical value under title 44, USC will be automatically declassified starting on 17 April 2000 *unless* that information is exempted from declassification.

2. (S) In this example, the OCA is an official that has been delegated Original Classification Authority for a particular program. The OCA must be identified (see paragraph 4-8a) and the reason that the information is classified must be stated (see paragraph 4-9) The OCA has indicated the information is exempt from automatic declassification within 25 years (see paragraph 4-11). In this example, the OCA has selected reason 5 (reveal U.S. military war plans that remain in effect). Instead of a date or event for automatic declassification, the OCA has used this exemption category and included a date outside the 25-year window.

R.U. Real Director, Security

Classified by: BG I. INCHARGE, Chief, Army Program Command Reason: 1.5(a) Declassify on: 25X5(December 31, 2026) Date of source: 22 April 1998

SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-2. Sample of Marking a Classified Document that is Exempt from the 25-Year Automatic Declassification

Sample of Marking an Originally and Derivatively Classified Document

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET

Subject: Sample of Marking a Classified Document That Contains Some Information Originally Classified and Some Information Derivatively Classified From A Source Document (U)

1. (U) This page is UNCLASSIFIED and is marked SECRET for training purposes only in order to illustrate classification markings. In the case of either derivatively or originally classified documents, the classification/declassification markings are usually placed at the bottom of the face of the document on either the left, right, or center.

2. (S) In this example, the Original Classification Authority (OCA) is generating a document that has information that the OCA originally classified on 21 August 1996 but also contains information classified by another OCA that was taken from a source document dated 11 November 1995. In this case, the document is considered to be a derivatively classified document even though some of the information is originally classified. If there is any derivatively classified information, the document *must* be marked under the rules for derivative classification. In this example, the OCA classified information under two exemption categories, X3 and X6, because the information could not be declassified within ten years. The source document shows a declassification date of 10 December 2001 (the other OCA did not select an exemption category). The declassification date is always the most restrictive date or event (one that occurs farthest in the future). Since the X3 and X6 is an indefinite declassification date, it is always selected over a specific date. The date of the source is always the date of the most recent source used (in this case, the date of original classification, 21 August, 1996).

R. U. Confused Chief, I.M. Ready Command

Classified by: Multiple Sources Reasons: 1.5e, 1.5d, and derivatively classified source Declassify on: Source marked X3,6 Date of Source: 21 August 1996

SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-3. Sample of Marking an Originally and Derivatively Classified Document

Sample of Marking a Document Derivatively Classified from Information in Old Document

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET

Subject: Sample of How to Mark a Classified Document Derivatively Classified from Information in Old Document (U)

1. (C) This page is totally UNCLASSIFIED, if this paragraph were really CONFIDENTIAL, it would be marked "C" in parentheses just after the paragraph number, or just before the first word if there is no paragraph numbering used. Mark the classification of the subject or title at the end of the subject or title. This makes it easier to refer to the memo or other document if the subject or title is kept UNCLASSIFIED. If the subject or title is classified, then use the appropriate abbreviation of (C), (S) or (TS) for CONFIDENTIAL, SECRET, and TOP SECRET, respectively.

2. (U) In this example the lead-in part of the paragraph is UNCLASSIFIED, but the subparagraphs are classified at different levels. Each portion of the paragraph stands on its own and is marked according to the information in that particular section.

a. (S) The new way to mark classified documents contains some changes. Portion (paragraph), page, and overall markings as well as where to place warning notices has essentially not changed (see paragraphs 4-3 through 4-6 for overall, page, and portion markings, and 4-12 for warning notices). The main point to remember is: could the reader figure out what is classified and at what level by the markings? The answer must be yes.

b. (C) What has changed? Quite a lot.

3. (S) In this example, the memo is derivatively classified based solely on information classified under the old system (EO 12356) and marked "Declassify on OADR". Remember, OADR is no longer used.

H. P. CULVER Chief, How to Classify Division

Derived from: HQDA Memo, 10 Feb 94 Subj: "Security is a FORCE MULTIPLIER" Declassify on: Source marked OADR Date of Source: 10 February 1994 SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-4. Sample of Marking a Document Derivatively Classified from Information in Old Document

Sample of Marking a Document When each Portion is Unclassified but Together are classified by Compilation

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY CONFIDENTIAL

Subject: Sample of Marking a Document When each Portion is Unclassified but Together are Classified by Compilation (U)

1. (U) This page is totally UNCLASSIFIED, and is marked classified for training purposes only. Mark the classification of the subject or title at the end of the subject or title, keeping the title UNCLASSIFIED, if possible. This makes it easier to refer to the memo or other document if the subject or title is kept UNCLASSIFIED.

2. (U) In this example the each of the paragraphs are UNCLASSIFIED, but the whole page is classified at the assigned level. Each portion of the paragraph stands on its own as UNCLASSIFIED and is marked according to the information in that particular section. The problem arises when the information is presented together.

3. (U) A classification guide, dated 28 May 1996, that concerns a very sensitive military operation, states that the type of information, when compiled, revealed in this memo, is classified as CONFIDENTIAL, and exempted by category number 4, section 1.6 of EO 12958. Exemption 4 concerns information on U.S. military plans, or national security emergency preparedness plans.

4. (U) Therefore, a statement, such as the one in paragraph 5, will be included and all appropriate page and portion markings included.

5. (U) This portion is UNCLASSIFIED when extracted alone. When it is combined with one or more of the above paragraphs, this compilation is CONFIDENTIAL based on Section 1.5a, Executive Order 12958.

H. P. CULVER Chief, How to Classify Division

Derived from: BO Missile Classification Guide and multiple sources Declassify on: Source marked X4 Date of Source: 28 May 1996

## CONFIDENTIAL CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-5. Sample of Marking a Document When each Portion is Unclassified but Together are classified by Compilation

Sample of Marking a Document Derivatively Classified from One Source Classified under the Current System

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET

Subject: Sample of How to Mark a Classified Document that is Derivatively Classified from One Source Classified under the Current System (C)

1. (S) This page is UNCLASSIFIED and is marked SECRET for training purposes only to illustrate classification markings. Note that in this example, the subject of the memo is classified CONFIDENTIAL. It is helpful to choose an UNCLASSIFIED subject or title, but sometimes use of a classified subject or title cannot be avoided.

2. (C) In this example, the memo is classified because it concerns a weapons system and that weapons system has a Security Classification Guide. In the classification guide, it states that the type of information in this memo is classified and at the levels of classification indicated. The classification guide reflects the new system of classifying and marking required as a result of EO 12958. The classification guide says that the information contained in this memo will be declassified "X3". That means it requires protection for more than 10 years based on exemption category number 3, found in section 1.6 of EO 12958. Exemption category 3 concerns information if released would impair the development or use of technology within a U.S. weapons system. The Original Classification Authority does not have to select a specific date or event for declassification (but can if deemed feasible) if the information qualifies under one of the exemption categories. In this case, no date or event for declassification Authority. The classification guide is dated 21 August 1996.

I. M. MOOSE Chief, Army Needs Security Office

Derived from: Oxnard Missile (OM-1) Classification Guide: Dated 21 August 1996 Declassify on: Source marked X3 Date of Source: 21 August 1996

SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-6. Sample of Marking a Document Derivatively Classified from One Source Classified under the Current System

Sample of Marking a Document Derivatively Classified from Source Classified under the Old System and a Source Classified under the Current System

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

## CONFIDENTIAL

Subject: Sample of How to Mark a Classified Document Derivatively Classified from a Source Classified under the Old System and a Source Classified under the Current System (U)

1. (C) This page is totally UNCLASSIFIED and is only marked CONFIDENTIAL for training purposes to illustrate classification markings.

2. (C) There have been many changes in the area of classification and declassification instructions that go on the front of the document. This may include: all information in the document is originally classified, all derivatively classified, a combination of original and derivative classifications, some of the information classified was marked under the old system, all of the information classified was marked under the old system, some or all originally classified information was marked under the old system but marked different ways, some or all RD, FRD or FGI in the document, etc.

a. (U) The new marking system is national policy and is implemented in this regulation.

b. (C) In this example, this memo is derivatively classified using two sources. One source (dated 19 August 1995) was marked under the old system as "Declassify on: OADR". The other source (dated 24 May 1996) is marked under the current system as "Declassify on X3 (meaning that it requires classification beyond ten years based on exemption category 3, as defined in Sec. 1.6, EO 12958). The "date of source" is date of the most recent source.

I. Wanna Learn Chief, Hooah Division

Derived from: Multiple Sources Declassify on: Sources Marked X3 and OADR Date of Source: 24 May 1996 CONFIDENTIAL CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-7. Sample of Marking a Document Derivatively Classified from Source Classified under the Old System and a Source Classified under the Current System

Sample of a Document Derivatively Classified from Multiple Sources

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY TOP SECRET

Subject: Sample of How to Mark a Classified Document Derivatively Classified from Multiple Sources (U)

1. (TS) This page is UNCLASSIFIED and is marked TOP SECRET for training purposes only. While there are many possible situations for derivative classification, only the more common ones are shown in these figures.

2. (S) In this example, this memo is derivatively classified based on data on several subjects and those subjects are covered by two different Security Classification Guides.

a. (TS) One classification guide is dated 28 May 1996 and concerns a very sensitive military operation. That Security Classification Guide states that the type of information revealed in this memo is classified as TOP SECRET. A declassification date or event was not selected by the Original Classification Authority and the information was designated as exemption category number 4, section 1.6 of EO 12958. Exemption 4 concerns information on U.S. military plans, or national security emergency preparedness plans.

b. (C) The other classification guide is dated 10 December 1996 and concerns a weapons system. That guide states that the type of information revealed in this memo is classified CONFIDENTIAL. A declassification date or event was not selected by the OCA and the information was designated as exemption category 3, which concerns information that if revealed would impair the development or use of technology within a U.S. weapons system. Even though the first guide pertains to information classified at the higher level (TOP SECRET), the "Date of Source" line is always the most *recent* date of all the sources used--in this case, the date of the guide pertaining to the CONFIDENTIAL data in this memo.

Helen P. Security Chief, Project Moose Radar

Derived from: Multiple Sources Declassify on: Sources marked X3 and X4 Date of Sources: 10 December 1996

TOP SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-8. Sample of a Document Derivatively Classified from Multiple Sources

Sample of Marking a Document Where the Cover Memo is Unclassified but the Attachments are Classified

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY CONFIDENTIAL

Subject: Sample of Marking a Document Where the Cover Memo is Unclassified but the Attachments are Classified (U)

1. (U) This page is totally UNCLASSIFIED, and is marked classified for training purposes only. Mark the classification of the subject or title at the end of the subject or title, keeping the title UNCLASSIFIED, if possible. This makes it easier to refer to the memo or other document if the subject or title is kept UNCLASSIFIED.

2. (U) In this example this cover memorandum is UNCLASSIFIED, but the attachments are classified at the assigned level. Each portion of the paragraph, on this memorandum, stands on its own as UNCLASSIFIED and will be marked according to the information in that particular section. The cover memorandum will be marked with the highest overall classification of the complete package.

3. (U) When it is expected that the memorandum may be separated from the rest of the package, a statement, similar to the one in the lower right area, will be affixed.

4. (U) When this memorandum is separated from the rest of the package, be sure to line out the classification and remark it as UNCLASSIFIED (see paragraph 4-35 for more details).

I. M. ANOCA Chief, I Can Classify Group

Derived from: BOOM Project Classification Guide Declassify on: Source marked X2 Date of Source: 31 October 1998

Regrade UNCLASSIFIED when separated from CLASSIFIED attachment

## CONFIDENTIAL CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-9. Sample of Marking a Document Where the Cover Memo is Unclassified but the Attachments are Classified

Sample of Restricted Data (RD) and Formerly Restricted Data (FRD) with Warning Notice

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET SUBJECT: Sample of Postricted Data (PD) and Formerly Postricted Data (

SUBJECT: Sample of Restricted Data (RD) and Formerly Restricted Data (FRD) with Warning Notice (U)

1. (U) Restricted Data (RD) is all data concerning: (a) design, manufacture or utilization of atomic weapons; (b) the production of special nuclear material; or (c) the use of special nuclear material in the production of energy, but will not include data declassified or removed from the Restricted Data category under section 142 of the Atomic Energy Act of 1954, as amended. Formerly Restricted Data (FRD) is information removed from the Restricted Data category upon a joint determination by the Department of Energy (or antecedent agencies) and the Department of Defense that such information relates primarily to the military utilization of atomic weapons and that such information can be safeguarded adequately as classified defense information. For purposes of foreign dissemination, this information is treated in the same manner as Restricted Data.

2. (S-RD) Documents containing RD information are portion marked in the same manner as other classified documents. Paragraphs containing RD are further indicated by the additional abbreviation of "RD" following the collateral level, as shown at the beginning of this paragraph.

3. (C-FRD) A document containing RD also has the warning notice shown below. When documents contain both RD and FRD, the RD warning notice takes precedence and the FRD notice is omitted.

4. (S) Do not include downgrading or declassification instructions on a document containing RD or FRD. These documents are exempt from predetermined downgrading or declassification action. The DOD does not originate RD or FRD, therefore, all documents should reflect the "Derived from" byline.

Derived from: DOE CG-W5

RESTRICTED DATA This material contains Restricted Data as defined in the Atomic Energy Act of 1954. Unauthorized disclosure subject to administrative and criminal sanctions.

## SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-10. Sample of Marking a Document Where the Cover Memo is Unclassified but the Attachments are Classified

Sample of Marking Foreign Government Information

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET

SUBJECT: Sample of Marking Foreign Government Information (FGI)(U)

1. (U) The U.S. Government affords protection to information provided by foreign governments. Care must be taken to identify the source of the information.

2. (FGI-C) Mark the portions that contain the foreign government information to indicate the country of origin and the classification level. Substitute the words "Foreign Government Information" or "FGI" where the identity of the specific government must be concealed. The identity of the concealed foreign source in this example must be maintained with the record copy and adequately protected.

3. (UK-S) This paragraph contains information marked "Secret" by the government of the United Kingdom. The "Derived From" citation should cite the title of the document provided. Declassification date, event, or exemption category is carried forward, if known.

Derived From: FGI Source Document or identify the foreign government source document dated *(fill in the date)* Declassify On: X5, FGI

SECRET CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-11. Sample of Marking Foreign Government Information

Sample of Marking Working Papers

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY CONFIDENTIAL WORKING PAPERS Created: 1 April 1999

Subject: Sample of Marking Working Papers (U)

1. (C) This page is UNCLASSIFIED and is marked CONFIDENTIAL for training purposes only. IMPORTANT: You must be an Original Classification Authority to originally classify a document. Working papers are documents and material accumulated or created in the preparation of finished documents and material. Working papers containing classified information will be: Dated when created; conspicuously marked as "Working Papers" or "DRAFT" on the first page of the document in letters larger than the text; marked with the highest classification of any information contained in the material; protected in accordance with the assigned classification; destroyed when no longer needed; and accounted for, controlled, and marked in the manner prescribed for a finished document of the same classification. See paragraph 6-24, of this regulation for further information. This sample memo only pertains to those relatively few cases in which an OCA is generating a document that the OCA is originally classifying and is considered "Working Papers" until completed.

2. (C) In this example, the OCA is an official that has been delegated Original Classification Authority for a particular intelligence program. The OCA has selected reason 1.5(e) (scientific, technological, or economic matters relating to the national security). Instead of a date or event for declassification within 10 years, the OCA has authorized exemption category X8 (Violate a statute, treaty, or international agreement) for protection beyond 10 years.

Y.I. Never Director, Security Awareness

Classified by: BG I. Didit, Chief of Intelligence Programs, Army Security Command Reason: 1.5(c) Declassify on: X8 Date of source: 26 March 1999 WORKING PAPERS CONFIDENTIAL CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure 4-12. Sample of Marking Working Papers

Equivalent Foreign Security Classification

\*\*This is a direct reprint from DODD 5200.1-R.\*\*

COUNTRY	TOP SECRET	SECRE	T CONFIDEN	TIAL	OTHE	R
Albania	TEPER SEKRET	SEKRE	T IMIREBESU	ESHE	M	I KUFIZUAR
Argentina RESERVAI	ESTRICTAMENTE	SECRET	TO SECRETO	CON	FIDENC	IAL
Australia	TOP SECRET	SECRE	T CONFIDEN	TIAL	REST	RICTED
Austria	STRENG GEHEIM	GEHEIM	M VERSCHLU	SS		
Balkans State SE DRZAVA Belgium	STROGO POVERL CRET Military SEC TAJNA VOJN	-JIVO T RET IA TAJNA	TAJNO POVE	ERLJIV	0	
(French)	TRES SECRET RESTREINT	SECRE <sup>®</sup> IS	T CONFIDEN	CONFIDENTIEL		SION
(Flemish)	ZEER GEHEIM VERSPREIE	gehein Ding	M VERTROUV	VELIJK	BEPE	RTKE
BoliviaSUPE MUY SECF	ERSECRETO or RETO	SECRE		CIAL	RESE	RVADO
Brazil ULTF	RA SECRETO SECF	RETO C	CONFIDENCIAL	RESE	ERVAD	)
Bulgaria	STROGO SEKREN	NTO S	SEKRETEN or	POVE	ERITELI	EN or
SEKF	RETNO POVERITELNO NEPOZVOLEN (Illicit) ZABRANEN (Forbidden)		as in Limited)			

COUNTRY	TOP SECR	ET	SECRET		CONFIDE	NTIAL	OTHE	2
Cambodia	TRES SECF	RET	SECRET		SECRET/0	CONFIDE	ENTIEL	
Canada	TOP SECRE	ET	SECRET		CONFIDE	NTIAL	REST	RICTED
Chile SECR	ETO SECF	RETO	RESERV	AD	O RES	SERVAD	0	
Columbia	ULTRASEC REST	reto Fringid	SECRET OO	0	RESERVA	DO	CONFI	DENCIAL
Costa Rica	ALTO SECF	RETO	SECRET	0	CONFIDE	NCIAL		
Croatia	NAJVECI TA	AJNITAJ	NI TA	JNI	POVERLJ	IV OGR/	ANCIEN	
Denmark	YDERST HE TJEN	EMMELI ESTEBF	gt he Rug	EMN	<b>IELIGTFO</b>	RTROLIC	θT	TIL
Ecuador	SECRETISI	MO	SECRET	0	CONFIDE	NCIAL	RESER	RVADO
El Salvador	ULTRA SEC	RETO	SECRET	0	CONFIDE	NCIAL	RESER	RVADO
Ethiopia	YEMIAZ BIRTOU MISTIR MISTIR KILKIL							
Finland	ERITTAIN SALAINEN							
France	TRES SECRET SECRET DE			DE	FENSE	CONF	IDENTI	EL
	RESTREINTE							
Germany	STRENG G	EHEIM	GEHEIM		VS-VERT	RAULICH	l	
Greece APORRHT(	AKRWS DN	APOR	RHTON CRHSEW	VS	EMPISTE	JTIKON	PERIW	RISMENHS

COUNTRY TOP SECRET SECRET CONFIDENTIAL OTHER Guatemala ALTO SECRETO SECRETO CONFIDENCIAL RESERVADO Haiti SECRET CONFIDENTIAL Honduras SUPER SECRETO SECRETO CONFIDENCIAL RESERVADO Hong Kong TOP SECRET RESTRICTED SECRET CONFIDENTIAL Hungary SZIGOR'UAN TITKOS TITKOS BIZALMAS Iceland ALGJORTI TRUNADARMAL India TOP SECRET CONFIDENTIAL SECRET RESTRICTED Indonesia SANGAT RAHASIA RAHASIA TERBATAS BENKOLI SERRI SERRI KHEILI MAHRAMANEH MAHRAMANEH Iran Iraq ABSOLUTELY SECRET SECRET LIMITED (English Translation) Ireland AN-SICREIDEACH SICREIDEACH RUNDA SRIANTA (Gaelic) Israel SODI BEYOTER SODI SHAMUR MUGBAL SEGRETO RISERVATISSIMO RISERVATO Italy SEGRETISSIMO Japan KIMITSU GOKUHI HI TORIATSUKAICHUI Jordan MAKTUM JIDDAN MAKTUM SIRRI MAHDUD USE RUSSIAN EQUIVALENT Kazakstan

COUNTRY	TOP SECRET	SECRET	CONFIDENTIAL	OTHER				
Korea    KUF	P PI MIL ∥∥ KI	JP PI MIL	🛛 🗧 KUP PI MIL					
Kyrgystan	USE RUSSIAN EQUIVALENT							
Laos TRES	SECRET SECR RESTREINT	ET SECR	ET/CONFIDENTIEL	DIFFUSION				
Lebanon	TRES SECRET	SECRET	CONFIDENTIEL					
Moldavian <i>(May also u</i>	ULTRASECRET use Russian Equivale	SECRET ent)	CONFIDENTIAL OR SECRET	RETRINS				
Mexico	ALTO SECRETO	SECRETO	CONFIDENCIAL	RESTRINGIDO				
Netherlands DIENSTGE	ZEER GEHEIM HEIM VERTROUWELIJK	GEHEIM	CONFIDENTIEEL o	pr				
New Zealand	TOP SECRET	SECRET	CONFIDENTIAL	RESTRICTED				
Nicaragua	ALTO SECRETO	SECRETO	CONFIDENCIAL	RESERVADO				
Norway STRENGT HEMMELIG HEMMELIG KONFIDENSIELL BEGRENSET								
Pakistan	TOP SECRET	SECRET	CONFIDENTIAL	RESTRICTED				
Paraguay	SECRETO SECR	ETO CONF	IDENCIAL RESE	RVADO				
Peru ESTRICTAMENTE SECRETO CONFIDENCIAL RESERVADO SECRETO								

COUNTRY	TOP SECRET	SECRET	CONFIDENTIAL	OTHER			
Philippines	TOP SECRET	SECRET	CONFIDENTIAL	RESTRICTED			
Poland	TAIJNY SPECJAL	NEGO TAHJ	Y POUFNY				
Portugal	MUITO SECRETO	SECRETO	CONFIDENCIAL	RESERVADO			
Romania	ULTRASECRET SECRET	SECRET	CONFIDENTIAL or	RESTRINS			
Russian	COBEOWEHHO	CEKPETHO	)				
Saudi Arabia SAUDI TOP SECRET SAUDI VERY SAUDI SECRET SAUDI SECRET RESTRICTED							
Spain MEXI	MO SECRETO LIMITADA	SECRETO	CONFIDENCIAL	DIFFUSION			
Sweden HEMLIG HEMLIG (Red Borders)							
Switzerland (Three languages. TOP SECRET has a registration number to distinguish it from SECRET and CONFIDENTIAL)							
French	TRES SECRET SECRET DEFENSE CONFIDENTIEL						
DIFFUSION	DEFENSE RESTREINTE						
German	STRENG GEHEIM	GEHEIM	VERTRAULICH				
Italian	SEGRETISSIMO	SEGRETO	RISERVATISSIMO	RESERVATO			
Taiwan	(No translation in E	nglish charac	ters)				
Tajikstan	USE RUSSIAN EQ	UIVALENT					

COUNTRY	TOP SECRE	T	SECF	RET	CONF	IDENT	IAL	OTHER	
Thailand	LUP TISUD	LUPN	IAAG	LUP	POK I	PID			
Turkey	COK GIZLI GIZLI OZEL HIZMET OZEL								
Turkmenistan USE RUSSIAN EQUIVALENT									
Ukraine	TSILKOM SI	EKRET	NE	SEKR	RETNO	KONF	DENT	SIAL'NO	DYLA
Union of South Africa	TOP SECRE	T	SECR	RET	CONF	IDENT	AL	RESTRICT	ED
Afrikaans	UITERS GEI	HEIM	GEHE	IM	VERT	ROULI	<	BEPERK	
Unit Arab Republic (Egypt)	TOP SECRE	T	VERY	SECF	RET	SECR	ΞT	OFFICIAL	
United Kingdom TOP SECRET SECRET CONFIDENTIAL RESTRICTED									
Uruguay	ULTRA SEC	RETO	SECR	RETO	CONF	IDENC	IAL	RESERVAD	00
Uzbekistan USE RUSSIAN EQUIVALENT									
Viet Nam	TRES SECRET SECR			RET DEFENSE CONFIDENTIEL					
(French)	4	DEFE	NSE	REST	REINT	E			
(Vietnamese	KIN	TU M	AT						

NOTE: The classifications given above represent the nearest comparable designation that are used to signify degrees of protection and control similar to those prescribed for the equivalent U.S. classification.

#### Chapter 5 Controlled Unclassified Information

#### Section I For Official Use Only Information

#### 5-1. General

*a.* The requirements of the information security program apply only to information that requires protection in order to prevent damage to the national security and has been classified in accordance with EO 12958 or its predecessors. There are other types of information that require application of controls and protective measures for a variety of reasons. In accordance with DODD 5200.1–R this information is known as Controlled Unclassified Information (CUI). Since classified information and CUI exist side by side in the work environment, often in the same documents, this chapter is provided as an attempt to avoid confusion and promote proper handling. It covers several types of CUI, and provides basic information about the nature of this information and the procedures for identifying and controlling it. In some cases, the chapter refers to other DOD directives that provide more detailed guidance.

b. The types of information covered in this chapter include "For Official Use Only" information, "Sensitive But Unclassified" (formerly "Limited Official Use") information, "DEA Sensitive Information," "DOD Controlled Unclassified Nuclear Information," "Sensitive Information" as defined in the Computer Security Act of 1987, and information contained in technical documents.

#### 5-2. Description

*a*. For Official Use Only (FOUO) is a designation that is applied to unclassified information which is exempt from mandatory release to the public under the Freedom of Information Act (FOIA) (see AR 25–55 for more details). The FOIA specifies nine categories of information which can be withheld from release if requested by a member of the public. They are:

(1) Information which is currently and properly classified.

(2) Information which pertains solely to the internal rules and practices of the agency. This exemption has two profiles, "high" and "low." The "high" profile permits withholding of a document which, if released, would allow circumvention of an agency rule, policy, or statute, thereby impeding the agency in the conduct of its mission. The "low" profile permits withholding, if there is no public interest in the document, and it would be an administrative burden to process the request.

(3) Information specifically exempted by a statute establishing particular criteria for withholding. The language of the statute must clearly state that the information will not be disclosed.

(4) Information, such as trade secrets and commercial or financial information obtained from a company on a privileged or confidential basis, which, if released, would result in competitive harm to the company, impair the government's ability to obtain like information in the future, or protect the government's interest in compliance with program effectiveness.

(5) Intra-agency memoranda which are deliberative in nature; this exemption is appropriate for internal documents which are part of the decision making process and contain subjective evaluations, opinions and recommendations.

(6) Information, the release of which could reasonably be expected to constitute a clearly unwarranted invasion of the personal privacy of individuals.

- (7) Records or information compiled for law enforcement purposes that:
- (a) Could reasonably be expected to interfere with law enforcement proceedings.
- (b) Would deprive a person of a right to a fair trial or impartial adjudication.
- (c) Could reasonably be expected to constitute an unwarranted invasion of personal privacy of others.
- (d) Disclose the identity of a confidential source.
- (e) Disclose investigative techniques and procedures.
- (f) Could reasonably be expected to endanger the life or physical safety of any individual.
- (8) Certain records of agencies responsible for supervision of financial institutions.
- (9) Geological and geophysical information concerning wells.

b. Information which is currently and properly classified can be withheld from mandatory release under the first exemption category (subparagraph (1) above). FOUO is applied to information which is exempt under one of the other eight categories (subparagraphs (2) through (9) above). So, by definition, information must be unclassified in order to be designated FOUO. If an item of information is declassified, it can be designated FOUO if it qualifies under one of those other eight categories. This means that:

(1) Information cannot be classified and FOUO at the same time; and

(2) Information which is declassified can be designated FOUO, but only if it fits into one of the last eight exemption categories (2) through (9) above).

c. The FOIA provides that, for information to be exempt from mandatory release, it must fit into one of the qualifying categories and there must be a legitimate government purpose served by withholding it. Simply because information is marked FOUO does not mean it automatically qualifies for exemption. If a request for a record is received, the information must be reviewed to see if it meets this dual test. On the other hand, the absence of the FOUO marking does not automatically mean the information must be released. Some types of records (for example, personnel records) are not normally marked FOUO, but can still qualify for withholding under the FOIA. Only personnel officially appointed as a Release Authority can release Army information.

#### 5-3. Marking

*a.* Information which has been determined to qualify for FOUO status should be indicated, by markings, when included in documents and similar material. Markings should be applied at the time documents are drafted, whenever possible, to promote proper protection of the information.

b. Wholly unclassified documents and material containing FOUO information will be marked as follows:

(1) Documents will be marked "FOR OFFICIAL USE ONLY," in letters larger than the rest of the text, where practical, at the bottom of the front cover (if there is one), the title page (if there is one), the first page, and the outside of the back cover (if there is one).

(2) Pages of the document which contain FOUO information will be marked "FOR OFFICIAL USE ONLY" at the bottom.

(3) Material other than paper documents, for example, slides, computer media, films, etc., will bear markings which alert the holder or viewer that the material contains FOUO information.

(4) FOUO documents and material, transmitted outside the Department of Defense, must bear an expanded marking on the face of the document so that non–DOD holders understand the status of the information. A statement similar to this one should be used: This document contains information Exempt from mandatory disclosure under the FOIA. Exemption(s) (indicate the exemption(s)) apply.

c. Classified documents and material containing FOUO information will be marked as required by chapter 4 of this regulation, with FOUO information identified as follows:

(1) Overall markings on the document will follow the procedures in chapter 4. No special markings are required on the face of the document because it contains FOUO information.

(2) Portions of the document will be marked with their classification as required by chapter 4. If there are unclassified portions which contain FOUO information, they can be marked with "FOUO" in parentheses at the beginning of the portion. Since FOUO information is, by definition, unclassified, the "FOUO" is an acceptable substitute for the normal "U."

(3) Pages of the document which contain classified information will be marked as required by chapter 4 of this regulation. Pages which contain FOUO information but no classified information will be marked "FOR OFFICIAL USE ONLY" at the top and bottom.

d. Transmittal documents which have no classified material attached, but do have FOUO attachments, will be marked with a statement similar to this one: "FOR OFFICIAL USE ONLY ATTACHMENT."

*e*. Each part of electronically transmitted messages containing FOUO information will be marked appropriately. Unclassified messages containing FOUO information will contain the abbreviation "FOUO" before the beginning of the text.

#### 5-4. Access to FOUO information

FOUO information can be disseminated within DOD components and between officials of Army components and Army contractors, consultants, and grantees, as necessary, in the conduct of official business. FOUO information can also be released to officials in other departments and agencies of the Executive and Judicial Branches in performance of a valid government function. Special restrictions can apply to information covered by the Privacy Act. Release of FOUO information to members of Congress is covered by DODD 5400.4, and to the General Accounting Office by DODD 7650.1.

#### 5-5. Protection of FOUO information

*a.* During working hours, reasonable steps should be taken to minimize risk of access by unauthorized personnel. After working hours, FOUO information can be stored in unlocked containers, desks or cabinets if U.S. Government or U.S. Government–contract building security is provided, or in locked desks, file cabinets, bookcases, or similar items.

*b.* FOUO documents and material can be transmitted via first class mail, parcel post, or, for bulk shipments, fourth class mail. Electronic transmission of FOUO information by voice, data, facsimile or similar means, should be by approved secure communications systems whenever possible.

c. Record copies of FOUO documents will be disposed of in accordance with AR 25–400–2. Non-record FOUO documents can be destroyed by shredding or tearing into pieces and discarding the pieces in regular trash containers.

#### 5–6. Further guidance

Further guidance on safeguarding personal information is contained in DOD 5400.11-R.

#### Section II

#### Sensitive But Unclassified and Limited Official Use Information

#### 5–7. Description

Sensitive But Unclassified (SBU) information is information originated within the Department of State which warrants a degree of protection and administrative control and meets the criteria for exemption from mandatory public disclosure under the Freedom of Information Act. Prior to 26 January 1995, this information was designated and marked Limited Official Use (LOU). The LOU designation will no longer be used.

#### 5–8. Marking

The Department of State does not require that SBU information be specifically marked, but does require that holders be made aware of the need for controls. When SBU information is included in DOD documents, the documents will be marked as if the information were FOUO. There is no requirement to remark existing material containing LOU information.

#### 5-9. Access to SBU information

Within the Department of the Army, the criteria for allowing access to SBU information are the same as those required for FOUO information (see paragraph 5–4).

#### 5–10. Protection of SBU information

Within the Department of the Army, SBU information will be afforded the same protection as that required for FOUO information (see paragraph 5–5).

#### Section III

#### Drug Enforcement Administration Sensitive Information

#### 5–11. Description

Drug Enforcement Administration (DEA) sensitive information is unclassified information which is originated by DEA and requires protection against unauthorized disclosure in order to protect sources and methods of investigative activity, evidence, and the integrity of pretrial investigative reports. The administrator, and certain other officials, of the DEA have been authorized to designate information as "DEA SENSITIVE". The Department of Defense has agreed to implement protective measures for the following DEA sensitive information in its possession.

- a. Information and material that is investigative in nature.
- b. Information and material to which access is restricted by law.
- c. Information and material which is critical to the operation and mission of the DEA.
- d. Information and material in which the disclosure of such would violate a privileged relationship.

#### 5-12. Marking

*a.* Unclassified documents containing DEA sensitive information will be marked "DEA SENSITIVE," in letters larger than the rest of the text, where practical, at the top and bottom of the front cover (if there is one), the title page (if there is one), and the outside of the back cover (if there is one).

*b*. In unclassified documents, each page containing DEA sensitive information will be marked "DEA SENSITIVE" top and bottom. Classified documents containing DEA sensitive information will be marked as required by chapter 4, except that pages containing DEA sensitive information, but no classified information, will be marked "DEA SENSI-TIVE" top and bottom.

c. Portions of DA documents which contain DEA sensitive information will be marked "(DEA)" at the beginning of the portion. This applies to classified, as well as unclassified documents. If a portion of a classified document contains both classified and DEA sensitive information, the "DEA" marking will be included along with the parenthetical classification marking. For example, a document containing DEA sensitive information along with SECRET information will be marked "(S)(DEA)."

#### 5–13. Access to DEA sensitive information

Access to DEA sensitive information will be granted only to persons who have a valid need-to-know for the

information. A security clearance is not required. DEA sensitive information in the possession of the Department of Defense, cannot be released outside the DOD without prior authorization by the DEA.

#### 5–14. Protection of DEA sensitive information

*a.* DEA sensitive material can be transmitted within Continental U.S. (CONUS) by first class mail. Transmission outside CONUS must be by a means approved for transmission of SECRET material. Non–U.S. Government package delivery and courier services will not be used. The material will be enclosed in two opaque envelopes or containers, the inner one marked "DEA SENSITIVE" on both sides. Electronic transmission of DEA sensitive information within CONUS should be over secure communications circuits whenever possible; transmission outside CONUS must be over approved secure communications circuits.

b. Reproduction of DEA sensitive information and material will be limited to that required for operational needs.

c. DEA sensitive material will be destroyed by a means approved for the destruction of CONFIDENTIAL material.

#### Section IV

### **DOD Unclassified Controlled Nuclear Information**

#### 5–15. Description

DOD Unclassified Controlled Nuclear Information (UCNI) is unclassified information on security measures, including security plans, procedures, and equipment, for the physical protection of DOD Special Nuclear Material (SNM), equipment, and facilities. Information is designated DOD UCNI only when it is determined that its unauthorized disclosure could reasonably be expected to have a significant adverse effect on the health and safety of the public or the common defense and security, by increasing significantly, the likelihood of the illegal production of nuclear weapons or the theft, diversion, or sabotage of DOD SNM, equipment, or facilities. Information can be designated DOD UCNI by the SECARMY and individuals to whom they have delegated the authority.

#### 5-16. Marking

a. Unclassified documents and material containing DOD UCNI will be marked as follows:

(1) The face of the document and the outside of the back cover, if there is one, will be marked "DOD CON-TROLLED UNCLASSIFIED NUCLEAR INFORMATION" in letters larger than the rest of the text, where practical.

(2) Portions of the document which contain DOD UCNI will be marked with "(DOD UCNI)" at the beginning of the portion.

b. Classified documents and material containing DOD UCNI will be marked in accordance with chapter 4, except that:

(1) Pages with no classified information but containing DOD UCNI will be marked "DOD CONTROLLED UNCLASSIFIED NUCLEAR INFORMATION" at the top and bottom.

(2) Portions of the document which contain DOD UCNI will be marked with "(DOD UCNI)" at the beginning of the portion in addition to the classified marking, where appropriate.

c. Material other than paper documents, for example, slides computer media, films, etc., will bear markings which alert the holder or viewer that the material contains DOD UCNI.

*d.* Documents and material containing DOD UCNI and transmitted outside the Department of Defense must bear an expanded marking on the face of the document so that non–DOD holders understand the status of the information. The following statement will be used: Department of Defense — Controlled Unclassified Nuclear — Information exempt from mandatory disclosure (5 USC 552(b)(3), as authorized by 10 USC 128)

e. Transmittal documents which have DOD UCNI attachments will bear a statement: The attached document contains DOD Unclassified — Controlled Nuclear Information (DOD UCNI).

#### 5–17. Access to DOD UCNI

Access to DOD UCNI will be granted only to persons who have a valid need-to-know for the information and are specifically eligible for access under the provisions of DODD 5210.83.

#### 5-18. Protection of DOD UCNI

*a.* During working hours, reasonable steps should be taken to minimize risk of access by unauthorized personnel. After working hours, DOD UCNI can be stored in unlocked containers, desks or cabinets if U.S. Government or U.S. Government–contract building security is provided, or in locked buildings, rooms, desks, file cabinets, bookcases, or similar items.

b. DOD UCNI can be transmitted by first class mail in a single, opaque envelope or wrapping. Except in emergencies, electronic transmission of DOD UCNI will be over approved secure communications circuits.

c. Record copies of DOD UCNI documents will be disposed of in accordance with the Federal Records Act (44

USC 33) and component records management directives. Non-record DOD UCNI documents can be destroyed by shredding or tearing into pieces and discarding the pieces in regular trash containers.

#### Section V

#### Sensitive Information (Computer Security Act of 1987)

#### 5–19. Description

*a.* The Computer Security Act of 1987 established requirements for protection of certain information in federal government Automated Information Systems (AIS). This information is referred to as "sensitive" information, defined in the Act as: "Any information, the loss, misuse, or unauthorized access to or modification of which could adversely affect the national interest or the conduct of federal programs, or the privacy to which individuals are entitled under section 552a of Title 5, USC (the Privacy Act), but which has not been specifically authorized under criteria established by an Executive Order or an Act of Congress to be kept secret in the interest of national defense or foreign policy."

*b.* Two aspects of this definition deserve attention. First, the Computer Security Act of 1987 applies only to unclassified information which deserves protection. Second, unlike most other programs for protection of information, the Computer Security Act of 1987 is concerned with protecting the availability and integrity, as well as, the confidentiality of information. Much of the information which fits the Computer Security Act of 1987's definition of "sensitive" falls within the other categories of information discussed in this chapter.

#### 5-20. Marking

There is no specific marking authorized for the designation of "sensitive" information. If the information fits within one of the other categories of information described in this chapter, the appropriate marking requirements apply.

#### 5–21. Access to sensitive information

If sensitive information falls within one of the other categories of information described in this chapter, the specific limitations on access for the appropriate category will be applied. If it does not, access to the information will be limited only to those with a valid need for such access in order to perform a legitimate organizational function, as dictated by common sense principles of security management, learned through a proper and thorough security education program.

#### 5-22. Protection of sensitive information

Information on a DA AIS, which is determined to be "sensitive," within the meaning of the Computer Security Act of 1987, will be provided protection which is:

*a*. Determined after thorough consideration of the value and sensitivity of the information and the probable adverse impact of loss of its availability, integrity or confidentiality.

b. In compliance with applicable DA policy and requirements for security of information within automated systems.

c. Commensurate with the degree of protection required for the category of information described in this chapter to which it belongs (if any).

d. Based on sound application of risk management techniques and procedures.

#### 5-23. Further guidance

Further guidance is found in appendix E of this regulation, AR 380–19, DODD 5200.28, and other related publications.

#### 5–24. Technical documents

DODD 5230.24 and AR 70–11 require distribution statements to be placed on technical documents no matter if they are classified or unclassified (See figure 5–1). These statements facilitate control, distribution and release of these documents without the need to repeatedly refer questions to the originating activity. The originating office can, of course, make case–by–case exceptions to distribution limitations imposed by the statements.

Distribution Statements for Technical Documents

Distribution Statement A Approved for public release; distribution is unlimited.

Distribution Statement B Distribution authorized to U.S. Government Agencies only; *[reason]*; *[date]*. Other requests for this document will be referred to *[controlling DOD office]*.

Distribution Statement C Distribution authorized to U.S. Government Agencies and their contractors; *[reason]*; *[date]*. Other requests for this document will be referred to *[controlling DOD office]*.

Distribution Statement D Distribution authorized to the DoD and U.S. DoD contractors only; *[reason]*; *[date]*. Other requests for this document will be referred to The*[controlling DOD office]*.

Distribution Statement E Distribution authorized to DoD Components only; [reason]; [date] Other requests for this document will be referred to The[controlling DOD office].

Distribution Statement F Further distribution only as directed by *[controlling DOD office]* or higher DoD authority; *[date]* 

Distribution Statement X Distribution authorized to U.S. Government Agencies and private individuals or enterprises eligible to obtain export-controlled technical data in accordance with DoD Directive 5230.24; [date]. Controlling DoD office is [controlling DOD office].

Figure 5-1. Distribution Statements for Technical Documents

## Chapter 6 Access, Control, Safeguarding, and Visits

Section I Access

#### 6-1. Responsibilities

DA personnel are responsible, both personally and officially, for safeguarding classified information for which they have access. This responsibility includes ensuring they do not permit access, to sensitive or classified information, by unauthorized personnel. Any person who does not have a need-to-know and who is not cleared or granted access to information at that level, in accordance with the policy established in AR 380–67, is considered unauthorized personnel. Both the clearance/access authorization and the need-to-know must be present before access is authorized. The holder of the information, not the potential recipient, must confirm valid need-to-know and must verify the level of security clearance or access, including auditory and visual means. Care will be exercised to make sure that classified conversations are not made within hearing distance of unauthorized personnel. Collecting, obtaining, recording, or removing, for any personal use whatsoever, of any material or information classified in the interest of national security, is prohibited.

#### 6-2. Nondisclosure Agreement

*a.* Prior to granting access to classified information, DA personnel will receive a briefing outlining their responsibility to protect classified information and will sign the SF 312 (Classified Information Nondisclosure Agreement (NDA)). Cleared personnel who have signed an earlier nondisclosure agreement, the SF 189 (Classified Information Nondisclosure Agreement) (replaced by SF 312), and have not already signed the SF 312, do not need to sign the SF 312. They may, however, elect to replace the old SF 189 with a newly signed SF 312. National policy requires that SF 312 and SF 189 NDAs be retained for 50 years from the date of signature. Execution of the NDA is mandatory for all personnel as a condition of access to classified information. It will be signed once unless verification of previous execution of the form indicates that the form cannot be located, in which case the form will be signed again and filed as if it were an original. In order to preclude duplicate NDAs, reasonable effort will be made to verify an existing NDA prior to asking for a second form to be signed. The purpose of the NDA is to make sure that personnel requiring access to classified information are advised of their responsibility to protect that classified information.

*b*. Contractor personnel will execute the NDA through their company and not through the sponsoring DA command. Non–U.S. Government personnel, who have been hired under Civil Service procedures as consultants to the Department of the Army, and granted a DA security clearance or access authorization, in accordance with AR 380–67, will follow the same procedure, for execution of the SF 312, as civilian personnel. When in exceptional situations in which access to specific classified information is approved for uncleared non–government personnel, under the provisions of AR 380–67, the execution of the NDA will follow the same policy as stated for civilian personnel, as amended.

#### 6-3. Signing and filing the NDA

The command will have proof of clearance prior to execution of the NDA. Proof of security clearance will be the receipt of the completed DA Form 873 (Certificate of Clearance and/or Security Determination). Other means of verifying the clearance can come from the Department of the Army Central Clearance Facility (CCF), a review of individual's personnel file and verification that it contains the DA Form 873, or the issuance of an interim security clearance. Upon proof of clearance, a command official, typically the command security manager, will brief the individual on the responsibilities to protect classified information. After verification from a picture identification card, the individual will read, date, and sign the NDA. The command official will witness the execution of the NDA by signing and dating the form immediately after the individual's signature. The same official, or another official in the command, that witnesses the form, can serve as the accepting official.

- a. Civilian Personnel.
- (1) Intelligence Related Position:

Copies of nondisclosure agreements, such as SF 312 or SF 189 or similar forms, signed by civilian personnel, including employees of contractors, licensees, or grantees, with access to information that is classified under standards put forth by Executive Orders governing security classification, fall into this category. These forms should be maintained separately from personnel security clearance files. Agreements for civilian employees working for elements of the intelligence community must be maintained separately from the official personnel folder. These forms, that are maintained separately from the individual's official personnel folder, will be destroyed when 70 years old.

(2) All Others. The accepting official will forward the original NDA to the supporting local or regional Civilian

Personnel Office (CPO), to be filed in the individual's Official Personnel File (OPF). The form will be filed on the permanent side of the OPF as an adjunct to the DA Form 873. NDAs for civilian employees transferring from one duty station to another, to include transferring to another U.S. Government agency, will transfer as part of their OPF. The NDA will not be removed from the OPF. If a command receives an NDA executed by a current employee while that employee was assigned to another command or agency, the command official receiving the form will forward the NDA to the applicable local or regional supporting CPO for insertion into the OPF as an adjunct to the DA Form 873. When a civilian employee transfers from one duty station to another, the designated command civilian personnel official will ensure that the Standard Form 75 (Request for Preliminary Employment Data) verifies that a completed NDA is on file. These forms, that are maintained in the individual's official personnel folder, will apply the disposition instructions for the official personnel folder.

b. Military Personnel. For military personnel, a copy of the NDA will be kept on file by the command security manager, or other designated command official, for verification that the individual has executed the NDA. Copies of nondisclosure agreements, such as SF 312 or SF 189 or similar forms, signed by military personnel, with access to information that is classified under standards put forth by Executive Orders governing security classification, should be maintained separately from personnel security clearance files. This copy will remain in the command file until the individual transfers or is separated from the U.S. Army. Upon the soldier's arrival at the new duty station, the command security manager will maintain a copy of the original, newly signed NDA on file pending the next transfer. The accepting official will forward the original NDA to the address below, where it will be converted to microfiche and filed with the soldier's official records. Upon notification of transfer, the command security manager, or other designated command official, will send the copy of the NDA to the gaining organization's command security manager either by mail or in the possession of the transferring individual.

(1) Active Army Commissioned and Warrant Officers: Commander, U.S. Total Army Personnel Command, ATTN: TAPC-MSR Alexandria, VA 22332-0400.

(2) Active Army Enlisted Personnel: Commander, U.S. Army Enlisted Records and Evaluation Center, ATTN: PCRE-FS, 8899 East 56th Street, Fort Benjamin Harrison, IN 46249–5301.

(3) Reservists: Commander, U.S. Army Reserve Personnel Center, ATTN: DARP-PRD-MP, 9700 Page Avenue, St. Louis, MO 63132–5200. When a cleared Individual Ready Reserve (IRR) member is ordered to active duty for training that will involve access to classified information and previous execution of the NDA cannot be verified, an NDA will be completed at the training site and the original forwarded to the U.S. Army Reserve Personnel Center.

(4) National Guard Commissioned and Warrant Officers: Army National Guard Personnel Division, ATTN: NGB-ARD-C, 111 South George Mason Drive, Arlington, VA 22204–1382.

(5) For National Guard enlisted soldiers: Forward to the soldier's State Adjutant General, ATTN: POMSO.

c. Department of the Army Consultants and Other Non–U.S. Government Personnel. If a consultant to the Department of the Army is hired under Civil Service procedures, as opposed to contracting with a company for consultant services, the NDA will be executed and filed with the DA Form 873. If the consultant's OPF is not retired, the command is obligated to retain the NDA for the required 50–year retention period. Consultant NDAs cannot be used by or transferred to another activity. They only authorize access to classified information under a specific agreement and an access termination form must be executed when the agreement has ceased or when classified access is no longer required, whichever occurs first. In special situations where non–U.S. Government uncleared personnel have been granted classified access to specific information in accordance with the policy established in AR 380–67, the NDA will be attached to the exception to policy memorandum or other appropriate written authorization which authorized the individual's access to classified information and will be retained in the command's files for 50 years.

#### 6-4. Refusal to execute the NDA

If a person refuses to sign the NDA, the individual will be advised of the applicable portions of the NDA, SF 312. The individual will be given five calendar days to reconsider and will not be permitted access to classified information during that time. At the end of the five-day period, the individual will again be requested to sign the NDA. If at that point the individual still refuses to sign the NDA, their classified access, if it had been previously granted, will be formally suspended, the individual will not be permitted any access to classified information, the Department of the Army's Central Clearance Facility will be notified concerning clearance revocation or denial action, and the matter will be reported as required by AR 380–67.

#### 6-5. Debriefing and termination of classified access

*a.* Classified information is not the personal possession of any DA personnel, regardless of rank, title, or position. Classified information will not be removed to nonofficial or unapproved locations, such as personal residences, upon the termination of employment or military service of any person, including the custodian of that material.

b. All DA personnel who are retiring, resigning, being discharged, or will no longer have access to classified information, will out-process through the command security manager's office or other designated command office. During this out-processing the individual will be informed that security clearance and access to classified information has terminated and that the individual still has an obligation to protect any knowledge they have of classified information. DA personnel will sign a debriefing statement during out-processing. The debriefing statement will either

be the NDA Security Debriefing Acknowledgement section of the SF 312, or DA Form 2962 (Security Termination Statement). The debriefing, as a minimum, will consist of informing the individual of the continuing obligation to protect classified information accessed, the admonition that discussion or other revelation of classified information to unauthorized persons is prohibited, provide instructions for reporting any unauthorized attempt to gain access to classified information, advise the individual of the prohibition against retaining classified material when leaving the command, and remind the individual of the potential civil and criminal penalties for failure to fulfill these continuing responsibilities. The same procedures will be followed for DA personnel still employed and still in service whose security clearance has been withdrawn, denied (after interim access was granted), or revoked either for cause or for administrative reasons due to lack of need for future access to classified information. In these cases both civilian and military DA personnel will execute the debriefing statement.

c. Unless exempted by the senior security official at the MACOM, security out-processing is required for all cleared personnel transferring to another DA command or to a Federal Government agency. Transfers will not require the execution of the type of debriefing statement described in subparagraph b, above. This does not preclude the command from requesting the transferring individual sign or initial a form or statement indicating, in substance, that the individual has been advised of the continuing responsibility to protect classified information and/or has completed the security out-processing. Personnel transferring will be briefed on the responsibilities stated in subparagraph b, above. Additionally, personnel transferring will be advised that classified information previously created, or in the custody of, the individual, including that gained while attending training or conferences, does not belong to the individual and does not transfer to the gaining command without appropriate approval by both the gaining and losing commands. Such approval will be based upon the losing command's assessment of the need-to-know for the information by the gaining command. Out-processing can also be used as a means to ensure that the appropriate command security officials are aware of the departure of personnel to ensure combinations and passwords are changed, keys are returned, accountable documents and property are under new custody, etc. Where out-processing is not required for transfers, the command will establish procedures to ensure that the command security manager is advised of such transfers.

*d*. For all DA military personnel, retiring, resigning, or separating from military service, the DA Form 2962, or the termination portion of the NDA, will be executed and maintained on file by the command security manager, or other designated command official, at the soldier's last duty station, for a period of two years, in accordance with AR 25-400-2.

*e*. All Army civilian personnel who are retiring or resigning from government service, must out-process through the activity's security office. The security official will debrief the civilian employee about the continuing obligation to protect the classified information accessed during government service. The civilian employee should sign a DA Form 2962 or the NDA Debriefing Acknowledgement, which will be retained by the activity. Signing the NDA Debriefing Acknowledgement is the individual's option upon final separation from the government service, however, the individual will be informed that security clearance and access to classified information has terminated and that the individual still has a legal obligation to protect classified information. The original NDA, for civilian employees, who retire or resign from government service, will remain in the employee's OPF and will be retired as part of the OPF. The NDA (SF 189 or SF 312) for civilian employees who retired or resigned prior to 1993 and are currently filed in an inactive file will be forwarded to: National Personnel Records Center, Civilian Personnel Records, 111 Winnebago Street, St. Louis, MO 63118.

*f*. Refusal to sign the DA Form 2962 or the termination portion of the NDA, SF 312, will be considered a lack of personal commitment to protect classified information. Personnel who refuse to sign a termination statement will not be granted further access to classified information and their security clearance may be revoked or denied in accordance with AR 380–67.

#### 6-6. Communication and cooperation between command officials

Commanders will establish policy and procedures to ensure that other command officials and personnel advise the command security manager of any information affecting an individual's access to classified information. Personnel officials will make sure that transfer and recruitment documents, including vacancy announcements, indicate if a security clearance is required for the position.

## 6-7. Access to restricted data, formerly restricted data, and critical nuclear weapons design information

*a.* Access to RD (less CNWDI) and FRD by DA personnel, at Army facilities, will be under the same conditions as for all other classified information, based on the appropriate security clearance and access, need-to-know for the information, and in accordance with DODD 5210.2. See paragraph 6–17 for the requirement for DA certification to access classified information, including RD and FRD, held by Department of Energy (DOE) personnel and for classified visits to DOE certified facilities. Because of the sensitivity of nuclear information, the need-to-know criteria will be strictly enforced for all access to RD and FRD information.

b. Critical Nuclear Weapons Design Information (CNWDI) is a category of SECRET and TOP SECRET restricted data. Access to and dissemination of CNWDI is of particular concern to national security. Access to CNWDI will be limited to U.S. citizens with final TOP SECRET or SECRET, as appropriate to the information being accessed,

security clearance, and will be limited to the minimum number of personnel who require such access to accomplish assigned duties. Access to CNWDI will be limited to personnel whose need-to-know has been justified to, and verified by, an official authorized to sign DOE Form 5631.20 (Request for Visit or Access Approval), or by a representative appointed by that official. Once the need-to-know for CNWDI access has been justified and verified, personnel who have a need for access to CNWDI will be briefed on its sensitivity before access is granted. See paragraph 9–13, for suggested CNWDI briefing. CNWDI access authorizations will be reflected in appropriate security records. Records of CNWDI briefings and access authorizations will be maintained in a manner that will ease verification by certifying officials.

*c*. Access to CNWDI is strictly limited to U.S. citizens. In rare cases an exception to the U.S. citizenship requirement will be made if a non–U.S. citizen possesses a unique or very unusual talent or skill that is essential to the U.S. Government, and it is not possessed, to a comparable degree, by an available U.S. citizen. In such cases, the determination can be made that it is in the overall best interest of the United States to permit access to CNWDI. This determination will be made by the Secretary of Defense based upon the recommendation of the Secretary of the Army. Such requests will be forwarded through command channels to DAMI–CH.

#### 6-8. Access by persons outside the Executive Branch

Classified information can be made available to individuals or agencies outside the Executive Branch provided that such information is necessary for performance of a function from which the U.S. Government will derive a benefit or advantage, and that such release is not prohibited by the originating department or agency. MACOM Commanders and the Administrative Assistant to the Secretary of the Army are designated as Department of the Army Release Authorities. They are authorized to determine, subject to OCA approval and before the release of classified information, the propriety of such action in the interest of national security and the assurance of the recipient's trustworthiness and need-to-know. This authority can be further delegated, if required.

a. Congress

(1) Congressional staff members requiring access to DOD classified information will be processed for a security clearance in accordance with DODD 5142.1 and the provisions of AR 380–67. The Director, Washington Headquarters Services (WHS), will initiate the required investigation (initial or reinvestigation) to DIS, adjudicate the results and grant, deny or revoke the security clearance, as appropriate. The Assistant Secretary of Defense (Legislative Affairs) will be notified by WHS of the completed clearance action.

(2) The Assistant Secretary of Defense (Legislative Affairs) as the principal staff assistant to the Secretary of Defense for DOD relations with the members of Congress, will provide for DOD processing of personal security clearances for members of Congressional staffs.

(3) Personnel testifying before a Congressional committee, in executive session, in relation to a classified matter, will obtain the assurance of the committee that individuals present have a security clearance commensurate with the highest classification of information that is to be presented.

b. Government Printing Office (GPO) Documents. Governmentt Printing Office (GPO) Documents and material of all classification may be processed by the GPO, which protects the information in accordance with the DOD/GPO Security Agreement of February 20, 1981.

*c*. General Accounting Office representative. Representatives of the General Accounting Office (GAO) Representatives of the GAO can be granted access to classified information, originated by and in the possession of the Department of the Army and DOD, when such information is relevant to the performance of the statutory responsibilities of that office, as set forth in DODD 7650.1. Certifications of security clearance, and the basis thereof, will be accomplished pursuant to arrangements between GAO and the concerned command. Personal recognition or presentation of official GAO credential cards are acceptable for identification purposes.

*d.* Historical Researchers and Former Presidential Appointees (See DODD 5200.1–R, Chapter 6) MACOM commanders, the Administrative Assistant to the Secretary of the Army, and the DCSINT are authorized to execute the provisions of DODD 5200.1–R pertaining to historical researchers and former presidential appointees. This authority cannot be further delegated. Investigative personnel security requirements contained in AR 380–67 will be followed and the NDA must be executed and maintained for the required period of time.

e. Judicial Proceedings DODD 5405.2 governs the release of classified information in litigation.

*f.* Other Situations When necessary, in the interests of national security, MACOM commanders and the Administrative Assistant to the Secretary of the Army can authorize access by persons outside the federal government, other than those identified above, to classified information. This is accomplished only upon determining that the recipient is trustworthy, as determined by AR 380–67 requirements, for the purpose of accomplishing a national security objective, and that the recipient can and will safeguard the information from unauthorized access. This authority will not be further delegated. Once the approval official has determined need–to–know, and that the recipient can and will safeguard the information of AR 380–67 will be followed, regarding the personnel security investigative requirements to be met, prior to granting access to classified information. The approval authority will ensure that an NDA is executed prior to access and maintained for the required period of time.

#### Section II Control Measures and Visits

#### 6-9. Responsibilities for maintaining classified information.

*a*. Commands will maintain a system of control measures that ensures that access to classified information is limited only to authorized persons. The control measures will be appropriate to the environment in which the access occurs and the nature and volume of the information. The system will include technical, where appropriate, physical, administrative, personal, and personnel control measures.

b. DA personnel granted access to classified information are responsible for protecting classified information of which they have knowledge or that is in their possession or control. DA personnel are personally responsible for taking proper precautions to ensure that unauthorized persons do not gain access to classified information. Classified information will be protected at all times, either by storage in an approved security container, or having it under the personal observation and physical control of an authorized individual.

#### 6-10. Care during working hours

a. Classified material removed from storage will be kept under constant surveillance and control by authorized personnel. Classified document cover sheets, Standard Forms 703 (TOP SECRET Cover Sheet), 704 (SECRET Cover Sheet), and 705 (CONFIDENTIAL Cover Sheet), will be placed on classified documents or files not in security storage. All items containing classified information, such as drafts, carbons, notes, floppy disks, typewriter and printer ribbons, plates, stencils, worksheets, etc., will be destroyed immediately after they have served their purpose, or protected as required for the level of classified information they contain.

*b.* SF 702 (Security Container Check Sheet) will be displayed conspicuously on each piece of equipment used to store classified material. SF 702 need not be used for facilities secured by high-security locks, provided the key and lock control register provides an audit capability in the event of unsecured facilities. SF 702 is used to record the date and time of each instance when a security container is opened and closed. The following procedures apply:

(1) Properly cleared personnel will record the date and time whenever they unlock or lock the security equipment during the day followed by their initials.

(2) If a security container is locked, and the room in which it is located is to be left unattended, whenever possible, a person, other than the person who locked the safe, will check the container to make sure it is properly secured. The person doing the checking will record the time the container was checked and initial the form. The person who locked the safe will see that the check is made.

(3) Containers not opened during a workday will be checked and the action recorded as in subparagraph (2) above.

(4) Notations will also be made on SF 702 if containers are opened after hours, on weekends, and on holidays, as provided above.

(5) The SF 702 will be retained at least 24 hours following the last entry.

c. Reversible "OPEN-CLOSED" or "OPEN-LOCKED" signs will be used on each security container or vault in which classified information is stored. Signs are available through normal supply channels.

d. A person discovering a security container or security storage area open and unattended will-

(1) Keep the container or area under guard or surveillance.

(2) Notify one of the persons listed on part 1, SF 700 (Security Container Information), affixed to the inside of the security container lock drawer. If one of these individuals cannot be contacted, the duty officer, security manager, or other appropriate official will be notified.

e. Individuals contacted when a container or area is found open or unattended will-

(1) Report personally to the location; check the contents of the container or area for visible indications or evidence of tampering, theft, or compromise. If any evidence of tampering, theft, or compromise is noted:

(a) Installation or activity security personnel (if not at the scene) will be immediately notified so that a preliminary investigation can be initiated.

(b) The custodian will cease examination of the container and its contents (to prevent destruction of physical evidence) unless otherwise instructed by security personnel.

(c) A lock technician will be called to determine the nature of the tampering, and whether the security container is operating properly.

(2) Change the combination and lock the container. If the combination cannot be changed immediately, the security container will be locked and placed under guard until the combination can be changed; or the classified contents will be transferred to another container or secure area.

(3) If not previously accomplished, report the incident to the commander or security manager immediately for action relative to compromise or possible compromise.

#### 6-11. End-of-Day security checks

a. Commands that access, process, or store-classified information will establish a system of security checks at the close of each working day to ensure that all classified material is properly secured. Standard Form 701 (Activity

Security Checklist), will be used to record these checks. An integral part of the security check system will be the securing of all vaults, secure rooms, and containers used for the storage of classified material; SF 702 will be used to record such actions. In addition, Standard Forms 701 and 702 will be annotated to reflect after-hours, weekend, and holiday activity.

b. After-duty-hours security checks of desks may be conducted, provided:

(1) Each military member and civilian employee is notified of local policy and procedures pertaining to after-hours inspections, locking of desks, and maintenance of duplicate keys or combinations. Notification must be in writing, and in advance of any after-hours inspection program.

(2) After-duty-hours inspections are conducted only by military or civilian security personnel, and for the sole purpose of detecting improperly secured classified information.

#### 6–12. Emergency planning

Commands will develop plans for the protection, removal, and destruction of classified material in case of fire, flood, earthquake, other natural disasters, civil disturbance, terrorist activities, or enemy action, to minimize the risk of its compromise. The level of detail in the plan and the amount and frequency of testing of the plan is at the command option, subject to MACOM approval, and should be based upon an assessment of the risk which might place the information in jeopardy. In this regard, special concern will be given for locations outside the United States. In preparing emergency plans, consideration must be given to reducing the amount of classified material on hand, including the transfer of information to microforms or removable computer media to reduce bulk, and the storage of less frequently used material at more secure locations. AR 380–40 contains policy for the emergency protection, including emergency destruction under no–notice conditions, of COMSEC material.

#### 6–13. Telephone conversations

*a.* Classified discussions are not permitted in personal residences, in public, in public transportation conveyances (airplane, taxi, etc.), or in any area outside approved spaces on a U.S. Government or cleared contractor facility. Classified information will only be discussed, in telephone conversations, over secure communication equipment, such as a STU–III, and circuits approved for transmission of information at the level of classification being discussed. When discussing classified information, the ability of others in the area, who are not appropriately cleared or do not have a need–to–know, will be taken into consideration to make sure that classified information is not compromised by being heard or otherwise accessed by unauthorized personnel. This includes instances where the installation of STU–III telephones are authorized in personal residences. Non–secure telephones will have DD Form 2056 (Telephone Monitoring Notification Decal) affixed, advising the user that the telephone is subject to monitoring at all times and that use constitutes consent to this. Further guidance on monitoring can be found in AR 380–53.

b. As an exception to the policy on classified discussions in certain situations requiring immediate contact and discussion of classified information in off-duty hours, the installation of a secure telephone unit (such as STU-III) can be authorized in personal residences to the extent that MACOM policy permits, up to, and including, the SECRET level. Only the SECARMY is authorized to permit TOP SECRET communications, via approved secure methods, and document storage, in personal residences. The MACOM commander is authorized to permit SECRET communications, via approved secure methods, and document storage in personal residences. This will not be authorized for personal convenience. Where such communications units are permitted, care must be exercised in ensuring that unauthorized personnel, to include family members, are not within hearing distance when classified discussions take place, and that the control key for the communications unit is either personally retained or stored in a discrete location separate from the unit. In such cases, it can be necessary for the custodian of the unit to make notes regarding the classified discussion that occurs over the security telephone. Where this occurs, such classified notes can be retained in the personal residence only until the next duty day. If the next duty day falls during a period of more than one day, leave, Temporary Duty (TDY), or other absence, the material will be delivered for storage to a U.S. Government or cleared contractor facility prior to such absence. While in a personal residence, such classified notes will be safeguarded and under the personal, physical control of the authorized, cleared holder of the notes, at all times.

#### 6–14. Speakerphone guidance

*a*. There has been a lot of questions and debates over the use of speakerphones in Sensitive Compartmented Information Facilities (SCIF) and other open-storage areas. According to Director of Central Intelligence Directives (DCID)1/21 speakerphones are restricted from common-use areas where sensitive conversations might be picked up inadvertently.

*b.* NSA S412 approves the installation/enablement of speakerphones on National Secure Telephone Systems (NSTS) and Secure Telephone Unit (STU)–III instruments. These systems will only be used in sole–use offices, conference rooms, and similar areas, and all room occupants are required to be aware of the conversations taking place, such as rooms used for contingency planning. The intent of speakerphone approval, rests with the room occupant assuming responsibility for taking the necessary precautions to ensure that the classified discussion is not overheard. STU–IIIs must be configured in such a manner as to prevent speaker enablement in the non–secure mode. Approval for use of non–secure speakerphones on NSTS and STU–III instruments will be granted by NSA S412 on a case–by–case basis.

In some configurations, Integrated Services Digital Network (ISDN) technology allows outsiders the capability to activate the speaker on the telephone without the individual's knowledge.

c. NSA S412 representatives conduct random reviews as part of their SCIF inspections to ensure speakerphones are being utilized IAW prescribed policy. NSA also maintains records of current speakerphone locations, which may be verified as required. Therefore, after NSA grants approval for use of a speakerphone in a selected sole–office or conference room, you should not move the speakerphone unless you submit to NSA S412 a new request for speakerphone use.

d. NSA speakerphone security guidance for SCIF areas include the following:

(1) Sole-use offices.

(a) The telephone instrument must be located in a sole-use office that affords sound attenuation.

(b) The office door must be closed prior to engaging in a speakerphone conversation.

(c) The user must be aware of any uncleared individuals in the vicinity and exercise sound judgement in determining when it is appropriate to engage in a speakerphone discussion.

(2) Conference rooms.

(a) The telephone instrument must be used in a conference room that affords sound attenuation.

(b) All conference room occupants must be appropriately cleared and have a "need-to-know" regarding the conversation.

(c) Entrance into the conference room must be controlled to ensure that only appropriately cleared individuals are present.

*e*. Any speakerphone mishap involving a possible security compromise or violation must be reported to NSA S412. See DCID 1/21 for further guidance.

#### 6-15. Removal of Classified Storage and Information Processing Equipment

Storage containers and information processing equipment, which had been used to store or process classified information, will be inspected by cleared personnel, before removal from protected areas, and/or before unauthorized persons are allowed unescorted access to them. The inspection will ensure that no classified information remains within or on the equipment. Items to be inspected include security containers, reproduction equipment, facsimile machines, micrographic readers and printers, AIS equipment and components, equipment used to destroy classified material, and other equipment used for safeguarding or processing classified information. A written record of the inspection will be completed and maintained in accordance with paragraph 7-11.

#### 6–16. Visits

Commands will establish procedures to control access to classified information by visitors. See AR 380–10 for the policy on foreign visitors.

*a.* Except when a continuing, frequent working relationship is established, through which a current security clearance and need-to-know are determined, DA personnel visiting other Army commands, other U.S. Government agencies, and U.S. Government contractors, will provide advance notification of any pending visit that is anticipated to involve access to classified information. The visit request will provide certification of the visitor's security clearance and the date(s) and purpose of the visit. Visit requests will be signed by an official other than the visitor and that official will be in a position to verify the visitor's security clearance. It can be approved, denied, or rescheduled at the option of the command being visited. As a general rule, unless otherwise indicated or contrary to command policy, visit requests sent to DA commands will be considered to be approved unless notification to the contrary is received. It is a recommended procedure to verify that the request was received and accepted. Visit requests can remain valid for up to one year.

*b*. Unless informed to the contrary, by the activity to be visited, visit requests involving DA personnel will include the following:

(1) Visitor's full name, date and place of birth, social security number, and rank or grade.

(2) Certification of visitor's security clearance and any special access authorizations required for the visit.

(3) Full address of visitor's command, and telephone number of a point of contact at that command. Point of contact is generally the person signing the visit request or the command security manager or other official who can verify the clearance status of the visitor.

(4) Name and address of the activity to be visited and name of person(s) to be contacted at the visited activity.

(5) Purpose of the visit, in sufficient detail to establish an assessment of need-to-know, and necessity of the visit.

(6) Date and duration of proposed visit. Intermittent visits on the same visit request can be authorized, where clearly stated on the request and approved by the command being visited, for up to one year.

c. Each agency outside of the Department of the Army has its own criteria for what constitutes the required elements on a visit request. Most are similar to, but may not be exactly the same as, the above. For example, contractors, under the terms of the National Industrial Security Program (NISP) Operating Manual (NISPOM), are not required to furnish the visitor's social security number. In such cases in which personnel, particularly non–DA personnel, are visiting an Army command, the command can decide whether or not the visit request is adequate and/or
request more information if necessary. Care is to be exercised in requiring additional personal information if it is not specifically relevant. For instance, if it is the policy of another agency to exclude the social security number, or the rank/grade, or the date/place of birth of the visitor, that information cannot be relevant, at least to that agency, if all the other elements of the visit request are satisfied. However, the visited Command will make the final determination as to what is required on the visit request.

### 6-17. Classified visits by Department of Energy personnel and to DOE facilities

a. Certification of security clearances. DA commands will accept certification of security clearances granted by other components of DOD and all other Federal Government departments and agencies, including clearances granted to contractors of the Federal Government. The DOE and Nuclear Regulatory Commission (NRC) use a different terminology in granting clearances. DA commands will accept DOE and NRC clearances for access to classified information as shown below.

(1) DOE clearances.

(a) Q sensitive. TOP SECRET Restricted Data/Formerly Restricted Data, and all other TOP SECRET information (exclusive of RD and FRD).

(b) Q non-sensitive. SECRET Restricted Data, TOP SECRET Formerly Restricted Data, and all other TOP SE-CRET information (exclusive of RD).

(c) TOP SECRET. No access to RD. Access authorized to TOP SECRET FRD and all other TOP SECRET information (exclusive of RD).

(d) L CONFIDENTIAL RD, SECRET FRD. L CONFIDENTIAL RD, SECRET FRD, and all other SECRET information (exclusive of RD).

(e) SECRET. No access to RD. Access authorized to SECRET FRD, and all other SECRET information (exclusive of RD).

(2) NRC clearances.

(a) Q (F) and (0). TOP SECRET RD/FRD, and all other TOP SECRET information.

(b) L (contractor). SECRET information (exclusive of RD and FRD), and CONFIDENTIAL FRD.

(c) L (employee). No access to RD or FRD. Access authorized to SECRET information (exclusive of RD and FRD).

(3) Company CONFIDENTIAL clearances. DOD or NISP approved Company CONFIDENTIAL clearances are not valid for access to RD.

b. Visits to DOE facilities and other requests for access to RD held by DOE.

(1) Requests for access to RD held by DOE or other U.S. federal agencies outside of DOD that are designated by DOE, will be made on DOE Form 5631.20, which replaced DOE Form DP–277. The form will be signed by an official that is authorized, no lower than brigade commander level or equivalent, or that official's representative, designated in writing for security matters. The list of authorized officials will be approved and compiled by DAMI–CH and forwarded to the DOE. An information copy of the list will be forwarded, by DAMI–CH, to, and held at, the office of the senior security official at each MACOM, and for the HQDA activities, the office of the senior security official in the office of the Administrative Assistant to the Secretary of the Army. Additions, changes, or deletions to the list of certifying officials will be forwarded through command channels to DAMI–CH, and will include the complete name and address of the command, title of the command certifying official, phone number of the office that will make the certification, and the justification.

(2) The request will state that the individual requires repeated access to the same type of information, or continuing visits to a facility, under the cognizance of the same approving authority, if applicable. Generally, arrangements for continuing access can be made for a period not to exceed one year, subject to approval by DOE. In such cases, the DOE facility requires advance notification of each visit. This notification procedure will be verified with the approving office. Access by military personnel can be arranged for the specified period of their assignment. If access to CNWDI is required, a statement to that affect will be listed on the form. It is recommended that the point of contact, at the site to be visited, be contacted to obtain the correct office location for submission of the form. Without any specific information to the contrary, the DOE Form 5631.20 will be submitted as follows:

(a) Direct to DOE, Director of Safeguards and Security, Washington, D.C. 20585, for information at DOE.

(b) To the U.S. Nuclear Regulatory Commission, Division of Security, Washington, D.C. 20555, for requests that pertain to the Army and Navy Research Reactor Program and when the RD is held by federal government agencies other than DOD.

(c) To the managers of the Albuquerque or San Francisco DOE operations offices, or officials designated by them, for data that pertains to weapons programs.

(d) To the DOE headquarters division that has responsibility for the subject matter involved, regarding information in the custody of DOE personnel in situations other than those described above.

(3) DOE personnel will accept oral requests for access to RD in emergencies. In such cases, the same information that is requested on DOE Form 5631.20 will be provided and an appropriate written confirmation will be forward as

soon as possible. DOE personnel authorized to approve requests for access to, or release of, RD will make determinations of emergency situations.

## 6-18. Classified meetings and conferences

Meetings, conferences, classes, seminars, symposia, and similar activities, at which classified information is to be presented or discussed, are considered "classified meetings." The classified portions of these meetings present special vulnerabilities to unauthorized disclosure and will be limited to persons possessing an appropriate clearance and access and the need-to-know for the specific information involved. Security requirements contained elsewhere in this regulation and other applicable security regulations apply, without exception, to classified meetings.

*a*. For purposes of this regulation, classified meetings are divided into two categories: in-house classified meetings, and association-related classified meetings.

(1) In-House Classified Meetings. In-house classified meetings involve routine gatherings of U.S. Government officials, classes conducted at U.S. Government schools, meetings between U.S. Government personnel and current or potential contractor personnel, other than that which is association-related, on a matter related to a specific government contract, program, or project, or gatherings of U.S. Government personnel and foreign government and/or foreign contractor representatives on a matter related to a specific government contract, program, or project. There are no special requirements for in-house classified meetings. Security and foreign disclosure procedures specified elsewhere in this and other applicable regulations, such as AR 380–10, apply, including the requirement to hold the classified portions of the meeting in approved spaces on a U.S. Government installation or cleared contractor facility.

(2) Association-Related Classified Meetings. A meeting which involves a non-U.S. Government association or organization is considered, for the purposes of this regulation, to be an association-related meeting. Examples of association-related meetings are conferences, symposia, seminars, or other meetings of organizations such as the American Society of Military Engineers, the Association of Old Crows, Association of the U.S. Army, American Defense Preparedness Association, Aerospace Industries Association, etc. The need-to-know concept must be strictly adhered to when considering classified sessions at an association-related meeting. The mere fact that an individual is cleared and is a member of the association does not in any way constitute a need-to-know for the classified information to be presented at the meeting. Wide dissemination of classified information at the type of general meeting, represented in association-related meetings, increases the potential risk of unauthorized access to, and compromise of, classified information. Presentation of classified information at such meetings requires complete justification to ensure that dissemination of the information, to the association membership, is in the U.S. Government's interests, and, where presentation is justified, requires careful attention to security requirements. When an association wishes to hold a classified meeting or classified sessions at its general meeting for its members and other invited guests, it must obtain the agreement of a U.S. Government agency to act as the U.S. Government security control for the meeting. DA Commands will not accept security control for association-related classified meetings until approval has been obtained from the DCSINT, through DAMI-CH.

*b.* Approval authority. Security control by a DA command for any association-related classified meeting, and any meeting which involves foreign participation, and does not fall within the criteria of "in-house" classified meeting, as stated in subparagraph a above, will require the approval of the DCSINT. Once approved, a specific, detailed security plan must be put in place to ensure that classified information is safeguarded and is not accessed by unauthorized persons, those without a clearance, or cleared but without a need-to-know for the specific information. The plan will be developed by the command and approved by the MACOM. HQDA approval for security control will be considered only when all of the following conditions are met:

(1) The meeting will serve a specific U.S. Government purpose.

(2) The use of other appropriate channels for dissemination of classified information will not serve the purpose.

(3) The meeting location will be under the security control of a DA command, other U.S. Government activity, or a U.S. contractor with an appropriate facility security clearance.

(4) Adequate security procedures have been developed and will be implemented to minimize risk to the classified information involved.

(5) Screening for verification of security clearance and need-to-know of potential attendees is specifically addressed and followed.

(6) Classified sessions will be segregated from unclassified sessions whenever possible.

(7) Any participation by foreign nationals or foreign representatives complies with the requirements of DODI 5230.20 and DODD 5230.11. For example, assurance is obtained, in writing, from the responsible U.S. Government foreign disclosure office(s) that the information to be presented has been cleared for foreign disclosure.

(8) Announcement of the classified meeting will be unclassified and limited to a general description of topics expected to be presented, names of speakers, logistical information, and administrative and security instructions.

(9) Non-government organizations may assist in organizing and provide administrative support for a classified meeting, but all security requirements remain the specific responsibility of the DA MACOM sponsoring the meeting.

(10) Procedures must ensure that classified documents, recordings, audiovisual material, notes, and other materials created, distributed, or used during the meeting are controlled, safeguarded, and transported as required by other

provisions of this regulation. Note taking or electronic recording during classified sessions will be permitted only when it is determined that such action is necessary to fulfill the U.S. Government purpose for the meeting.

*c. Obtaining authorization.* When a DA command wishes to be authorized to serve as the security sponsor for an association-related classified meeting, they will make the request, through command channels, to DAMI-CH. It must be received at least 120 days in advance of the meeting. The request will contain adequate justification or explanation for all points in subparagraph b above and an overall description of the classified information to be presented. In addition, if the association is intending to invite citizens of foreign countries, the command will request approval for the disclosure of specified classified information to the countries or foreign nationals expected to attend (see AR 380–10). Foreign nationals, of approved countries, can then submit visit requests, through their embassies, as provided in AR 380–10.

*d. Command and MACOM responsibilities.* The decision on the approval or disapproval of security sponsorship, of an association–related classified meeting, will be communicated by DAMI–CH, to the command and its MACOM. If HQDA approval has been obtained, the command authorized as the security sponsor can inform the association that the announcements for the location of the meeting can be sent to its members. Announcements mentioning the location of the meeting or stating that a DA command has accepted security sponsorship can be made only after HQDA approval has been obtained. Announcements of classified meetings will be unclassified and limited to a general description of topics expected to be presented, names of speakers, logistical information, and administrative and security instructions. The command sponsoring the meeting will take the following actions:

(1) Appoint a security manager for this project. Only DA personnel can be appointed as the security manager for the meeting. Other U.S. Government or cleared contractor personnel can assist with the implementation of security requirements under the direction of the appointed security manager. Other association personnel can assist in the organization of, and administrative support to, the meeting.

(2) Develop a security plan. The command's MACOM must approve the final security plan to be implemented at the meeting. The security plan will describe the procedures that will be used to ensure that proper security measures for the access, control, storage, and dissemination of the classified information have been developed and will be implemented.

(3) Make sure that all personnel attending have the appropriate security clearance, access, and need-to-know for the specific information to be presented. The mere fact that an individual is cleared and is a member of the association does not in any way constitute a need-to-know for the classified information to be presented at the meeting. A visit request will be required for all U.S. Government personnel. If the U.S. Government personnel attending the meeting are in a duty status, as opposed to being in a leave status, the visit request can serve as certification of need-to-know will not be required. A visit request with U.S. Government contracting officer certification of need-to-know, will be required for contractor personnel. The certification of need-to-know for contractor personnel will be in writing, will specify the applicable contract number, or project or program if pre-contract activity serves as justification, and will be made by the U.S. Government contracting officer for the particular contract pertaining to the need-to-know for the information presented or discussed at the meeting.

(4) Make sure that participation or attendance by foreign nationals or foreign representatives complies with the requirements of AR 380–10. Foreign personnel can be invited to participate only after HQDA approval and the requirements of AR 380–10 are met.

(5) Make sure that all classified papers to be used, and all classified presentations, have been approved, in writing, by the originators of the information, for release to the cleared association members. Written foreign disclosure authorization will be required if there are to be foreign attendees at the meeting (see AR 380–10).

(6) Make sure that the classified sessions are segregated from the unclassified sessions at all times and that the classified sessions are held only at a U.S. Government installation or a cleared contractor facility. The location must allow proper control of physical, auditory and visual access to the classified information.

Note: No Department of the Army official is authorized to waive the requirement to hold such classified meetings at a U.S. Government installation or cleared contractor facility.

(7) Make sure that all classified material created, distributed, or used during the meeting are controlled, accounted for, safeguarded, and transmitted as required by other provisions of this regulation. Note-taking or electronic recording during the classified sessions will be permitted only if, in approving the plan, the MACOM approving official makes a specific determination that it is necessary to do so in order to fulfill the U.S. Government purpose for the meeting. If such approval is requested, the command will specifically state such in the security plan, or in the cover letter, for the MACOM request for approval of the plan.

### 6–19. Information processing equipment

There are a variety of non-COMSEC-approved equipment that are used to process classified information. This includes copiers, facsimile machines, computers, notebooks and other AIS equipment and peripherals, electronic typewriters, word processing systems, hand-held personal data managers, etc. Commands will identify those features, parts, or functions of equipment used to process classified information that can retain all or part of the information. Command security procedures will prescribe the appropriate safeguards to prevent unauthorized access to that information, and replace, control, and/or destroy equipment parts, pursuant to the level of the classified material contained

therein, when the information cannot be removed from them. Alternatively, the equipment can be designated as classified and appropriately protected at the retained information's classification level (for instance, by being installed in a vault approved for the storage of classified information at that classification level).

### 6-20. Receipt of classified material

Commands will develop procedures to protect incoming mail, bulk shipments, and items delivered by messenger, until a determination is made whether classified information is contained in the mail. Screening points will be established to limit access to classified information.

### Section III

### Accountability and Administrative Procedures

### 6-21. TOP SECRET information

Material containing TOP SECRET information will be provided continuous control and accountability. Commands will establish procedures, tailored to the individual situation, in accordance with the principles of risk management, for the control and accountability of the TOP SECRET material they hold. These procedures will provide the means of facilitating oversight and management of TOP SECRET access controls, assessment and management of holdings, and identification of material at risk, in cases of potential unauthorized disclosure. In developing these procedures, the following minimum requirements will be met.

*a.* TOP SECRET Control Officers (TSCO) will be designated within offices which handle or maintain TOP SECRET material. They will be responsible for receiving, dispatching, and maintaining accountability and access records for TOP SECRET material. Such individuals will be selected on the basis of experience and reliability, and as a general rule, will already possess the appropriate clearance and access level equal to or higher than the information to be handled, and be a minimum grade of GS–07 or rank of E–7. Grade/rank can be waived by the commander, subject to MACOM policy on this matter. One or more alternate TSCOs will also be designated. There is no grade/rank preference for alternate TSCOs. TSCOs need not be appointed in those instances where there is no likelihood of processing TOP SECRET documentation. In such circumstances, the command will record the fact that a TSCO has not been appointed. TSCOs will maintain a current, accurate system of accountability within the command for all TOP SECRET documents and other material. TSCOs will record the receipt, dispatch, downgrading, movement from one command element to another, current custodian, and destruction of all TOP SECRET material. Automated information systems can be used to maintain these records.

*b.* TOP SECRET material will be accounted for by a continuous chain of receipts. Receipts will be maintained for five years. TOP SECRET registers (for example, DA Form 455 (Mail and Document Register) or equivalent) and TOP SECRET accountability record forms (for example, DA Form 3964 (Classified Document Accountability Record) or equivalent) will reflect sufficient information to identify adequately the TOP SECRET document or material. As a minimum, it will include the title or short title, date of the document, identification of the originator, copy number, and disposition. TOP SECRET material will be numbered serially and marked to indicate its copy number (for example, copy 1 of 2 copies) and accounted for accordingly.

c. TOP SECRET material will be inventoried at least once annually. The inventory will reconcile the TOP SECRET accountability register and records with 100 percent of the TOP SECRET material held. The inventory will be conducted by two properly cleared individuals. One will be the TSCO or alternate, and the other will be a properly cleared, disinterested party, that is neither a TSCO, alternate, or subordinate to either official. The inventory will consist of a physical sighting of the material or written evidence of authorized disposition, such as certificate of destruction or receipt of transfer. At the time of the inventory, each TOP SECRET document or material will be physically examined for completeness and the TSCO will ensure that the accountability record accurately reflects the material held. Discrepancies found during the inventory will be resolved immediately, or, where they cannot be immediately resolved, referred to the command security manager for further investigation.

*d.* In activities that store exceptionally large volumes of TOP SECRET material, MACOMs can authorize the inventory of TOP SECRET material to be limited to documents and material to which access has been granted within the past year (use the TSCO's accountability records), and 10 percent of the remaining inventory. The 10 percent will be randomly selected. MACOMs will document these authorizations and retain such documentation for as long as the authorization remains. In such cases, MACOM oversight will include a spot inventory of randomly selected TOP SECRET material during Advice and Assistance Visits (AAV), inspections, or other security reviews. Note: MACOMs cannot authorize the exception to the 100 percent inventory for TOP SECRET Special Access Programs information (see DODD 5220.22–M). Requests for an exception to the 100 percent annual inventory for TOP SECRET Special Access Programs information will be completely justified and submitted through command channels to DAMI–CH (SAPs).

*e*. Before leaving the command, the TSCO or alternate will conduct a joint inventory with the new TSCO or alternate of all TOP SECRET material for which they have custodial responsibility. In addition, a 100 percent inventory of all TOP SECRET material held by the command is advised, but not required. However, the new TSCO or alternate will be held accountable for all TOP SECRET material for which they have custodial responsibility.

*f*. As stated above, commands will establish procedures that provide a means of making easier the oversight and management of TOP SECRET access controls, assessment and management of holdings, and the identification of material at risk, in cases of potential unauthorized disclosure. Disclosure records can be used as a means to help meet these requirements. Disclosure records, such as DA Form 969 (Top Secret Document Record) or a similar form, provide a means to record the name of the individual(s) to whom the information/material has been disclosed, and the date(s) of the disclosure. They are a means that commands can choose to use in the execution of the above–cited responsibilities.

## 6-22. SECRET and CONFIDENTIAL information

Commands will establish procedures to control all SECRET and CONFIDENTIAL information and material originated, received, distributed, or routed to sub-elements within the command, and all information disposed of by the command by transfer of custody or destruction. The control system for SECRET and CONFIDENTIAL information is to be determined by a practical balance of security and operating efficiency.

## 6-23. NATO and Foreign Government material

Accountability requirements for NATO material are contained in AR 380-15. See paragraph 6-29 for recording the destruction of foreign government and NATO material.

## 6-24. Working papers

*a*. Working papers are documents and material accumulated or created in the preparation of finished documents and material. Working papers containing classified information will be:

(1) Dated when created.

(2) Conspicuously marked as "DRAFT" or "WORKING PAPERS" on the first page of the document in letters larger than the text.

(3) Marked with the highest classification of any information contained in the material.

- (4) Protected in accordance with the assigned classification.
- (5) Destroyed when no longer needed.

(6) Accounted for, controlled, and marked in the manner prescribed for a finished document of the same classification when:

(a) Released by the originator outside the command or transmitted electronically or through message center channels within the activity (exclusive or through a local area network or other automated system when the transmission does not go beyond the command).

(b) Retained for more than 180 days from the date of origin.

(c) Filed permanently.

(d) TOP SECRET information is contained therein.

*b*. MACOMs can grant exceptions for accountability, control, and marking requirements for working papers containing TOP SECRET information on a case-by-case basis provided a determination is made that:

(1) The conditions set forth in subparagraphs a6(a), (b), or (c), above, will remain in effect.

(2) The Command seeking an exception routinely handles large volumes of TOP SECRET working papers and compliance with prescribed accountability, control, and marking requirements would have an adverse affect on the command's mission or operations.

(3) Access to areas where TOP SECRET working papers are handled is restricted to personnel who have a TOP SECRET clearance, and other safeguarding measures are adequate to prevent the possibility of unauthorized disclosure.

## Section IV

## **Reproduction of Classified Material**

## 6–25. Policy

Documents and other material containing classified information will be reproduced only when necessary for the accomplishment of the command's mission or for compliance with applicable statutes or directives. Reproduction equipment and the reproduction process involve substantial risk. Therefore, commands will establish and enforce procedures for the reproduction of classified material which limit reproduction to that which is mission essential and will make sure that appropriate countermeasures are taken to negate or minimize any risk. All copies of classified documents reproduced for any purpose, including those incorporated in working papers, are subject to the same safeguards and controls prescribed for the document from which the reproduction is made. Reproduced material will be clearly identified as classified at the applicable level. TOP SECRET material will be numbered serially and marked to indicate its copy number (for example, copy 1 of 2 copies) and accounted for accordingly. Waste products generated during reproduction will be properly safeguarded as appropriate to the level of classification contained within, and destroyed in a manner approved for the destruction of classified information at that classification level.

a. Stated prohibition against reproduction of information at any classification level will be prominently displayed and strictly observed (for example, notices stating "Reproduction Only by Permission of Originator").

*b.* Except for the controlled initial distribution of information processed or received electronically, or that containing COMSEC or SCI which are governed by separate requirements, reproduction of TOP SECRET information, either portions of documents or whole documents, will be strictly controlled. Written authorization, by the TSCO or other command official who has been designated in writing as having this authority (TOP SECRET reproduction control official) is required. The command official can only authorize the local reproduction of TOP SECRET material with permission of the originator or higher authority (see paragraph 6–26). Once copies are made, the TSCO will be advised of the reproduction and will place the copied material under the command accountability system.

c. Specific equipment will be designated for the reproduction of classified information. Such equipment cannot be designated for classified reproduction if it leaves latent images in the equipment or on other material. Exceptions are that the equipment is in a vault or other area approved for the storage of classified information, the equipment is protected as classified material, and the material on which the image resides is destroyed as classified waste. Rules for reproduction equipment will be made aware of the risks involved with the specific equipment, the command procedures concerning the protection, control and accountability of reproduced information as well as the destruction of classified waste products. Information on security hazards associated with various types of reproduction equipment can be obtained from the Intelligence Materiel Activity (IMA), Intelligence Materiel Management Center, Fort Meade, MD, 20755–5315. Notices prohibiting reproduction of classified information.

#### 6–26. Approval for reproduction

*a*. Commands will establish procedures which ensure that appropriate approval is granted before classified material is reproduced. As a minimum, these procedures will—

(1) Require approval of the originator or higher authority before reproducing TOP SECRET documents and material.

(2) Ensure compliance with reproduction limitations placed on documents by originators and special controls applicable to Special Access Programs and other special categories of information.

(3) Make easier the oversight and control of reproduction of classified material.

b. A written record will be maintained verifying the approval authorization for reproduction of TOP SECRET information and, where required, for information addressed in subparagraph a(2), above. This record will be maintained for the life of the copy and filed with the destruction certificate when the copy is destroyed.

c. The provisions of subparagraphs a(1) and a(2), above, will not restrict the reproduction of documents for the purpose of facilitating declassification review. After review for declassification, those reproduced documents that remain classified must be destroyed in accordance with section V of this chapter.

## Section V

## **Disposition and Destruction of Classified Material**

#### 6–27. Policy

*a.* Classified documents and other material will be retained only if they are required for effective and efficient operation of the command or if their retention is required by law or regulation. Requests from contractors for retention of classified material will only be approved if they meet the same criteria and approvals are in the best interests of the government. See AR 380–49 for more guidance on contractor retention.

*b.* Documents which are no longer required for operational purposes will be disposed of in accordance with the provisions of the Federal Records Act (44 USC chapters 21 and 33) as implemented by AR 25–400–2. Classified information is subject to the same retention criteria as unclassified information. Special care will be exercised in the placing of classified information in files designated under AR 25–400–2 as "permanent". Such files can, and many are, eventually accessioned into the National Archives. These files are subject to any automatic, systematic, and mandatory declassification systems that exist now or will in the future. Resources, at present, cannot permit a careful review of all of the material prior to declassification. In order to conserve resources, the declassification review personnel can rely heavily on the markings on the front of the document and on the SF 135 for those cases in which the boxes cannot be opened prior to a decision to employ bulk declassification based upon a description of the contents of the file box. Resources cannot permit a careful review for unclassified, "For Official Use Only (FOUO)," information unless the FOUO markings are conspicuous on the front of the document and on the portions to which they apply, and are included in the SF 135 description. Command personnel must be aware that the current policy, as stated in the recent EO 12958, with amendments, refers to the declassification of information contained in permanent records that have been determined to have permanent historical value, by the Archivist of the U.S., under Title 44, U.S. Code. Therefore, once the classified material has been placed in a file designated under MARKS (AR 25–400–2) as "permanent," the

information in the files will be subject to the automatic declassification provisions of the prevailing EO, whether or not reviewed for declassification by DA personnel.

*c*. Commands will review classified files designated as "permanent," under AR 25–400–2, prior to forwarding to a Federal Records Center, where the files are maintained pending ultimate destruction or accession into the National Archives. Each classified document in the files will be reviewed to ensure that:

(1) The classified material is a necessary part of the file as described in AR 25-400-2.

(2) That only the record copy is placed in the file and that duplicate copies are destroyed.

(3) That the classified material has been reviewed for downgrading and declassification instructions and is properly remarked if downgraded or declassified.

(4) That any For Official Use Only information, that is contained in the document, is properly marked and a notice, that the document contains FOUO information, is displayed on the front cover and title page, or the first page when there is no cover or title page. Also, that it is marked on the applicable portions of the document. It is recommended that unclassified documents in the file that contain FOUO information be checked at the same time to make sure they are properly identified on the documents, on the file, and on the SF 135. This is because the National Archives can permit access to the public of unclassified information in permanent files if it is not clearly apparent that the information contains FOUO information.

(5) That the subject of the classified information is adequately described on the file label.

(6) That Restricted Data and Formerly Restricted Data and foreign government information are not intermingled with other information, and it is clearly marked on the file and accompanying forms (see AR 25–400–2, paragraph 9–2).

(7) That TOP SECRET information is not included unless it meets the criteria stated in AR 25-400-2, paragraph 9-2.

(8) That the subject of the classified information is adequately and completely described in the accompanying documentation, the SF 135 and SF 135–A (Continuation) as required by AR 25–400–2. Paragraph 9–5 of AR 25–400–2 requires that the SF 135, or SF 135–A, describe records in enough detail to permit quick retrieval of specific documents. This is true for all files whether classified or unclassified. However, particular attention to this requirement must be paid when the file is designated as "permanent" and contains classified material.

d. Commanders will make sure that the management of the retention of classified material is included in oversight and evaluation of program effectiveness.

*e*. Material which has been identified for destruction will continue to be protected as appropriate for its classification until it is actually destroyed. Destruction of classified documents and material will be accomplished by a means which eliminate risk of reconstruction of the classified information and will follow the criteria stated in this regulation.

### 6-28. Methods and standards for destruction

*a.* Classified documents and materials will be destroyed by burning, or, when meeting the standards contained in chapter 3, of this regulation, by melting, chemical decomposition, pulping, pulverizing, cross-cut shredding, or mutilation, sufficient to preclude recognition or reconstruction of the classified information. Strip shredders, those that do not have a half-inch cross cut feature, do not sufficiently destroy the information and are not authorized for use. MACOMs can approve the use of strip shredders, in exceptional cases, for use in the destruction of classified material at the SECRET level and below, under the following conditions: if the equipment was purchased prior to June, 1986, there are no other means of destruction available to the command, and additional precautions, such as shredding at least 20 pages of similar material, not blank paper, at the same time, are utilized to minimize risk of reconstruction of the material. MACOMs must consider efforts to replace the strip shredders as soon as possible.

b. Technical assistance and guidance on the application of the standards described above can be obtained from the Intelligence Materiel Activity, Intelligence Materiel Management Center, Fort Meade, MD 20755-5315.

c. Systems which involve the collection of classified material for later destruction, for example, the use of burn bags, will include provisions for minimizing the risk of compromise of the material while it awaits destruction.

#### 6-29. Records of destruction

*a.* Records of destruction are required for TOP SECRET documents and material. The record will be executed when the material is actually destroyed, or when it is torn and placed in a burn bag or similar container. Two persons will sign the destruction record as witnessing the destruction. DA Form 3964 (Classified Document Accountability Record) may be used for this purpose. Destruction records are not required for waste materials (scratch notes, typewriter and printer ribbons, carbon paper, etc.) containing TOP SECRET information, unless that material has been placed on an accountability record.

b. Records of destruction are not required for SECRET material, except for NATO and foreign government documents. For NATO or foreign government SECRET material, two signatures are required on the record of destruction. Records of destruction are not necessary for CONFIDENTIAL material unless required by the originator.

c. Records of destruction will be maintained for 5 years from the date of destruction. For guidance on requirements for NATO classified material, to include retention standards, see AR 380–15.

## Section VI Waivers

## 6-30. General

Waivers to the requirements in sections III through V of this Chapter can be authorized by MACOM commanders, and for HQDA activities, by the Administrative Assistant to the Secretary of the Army. This authority will not be further delegated and does not apply to any other section, Chapter, or appendix of this regulation. Waiver approval will follow the policy discussed in this Chapter. Waivers pertaining to SAPs will be submitted in accordance with appendix I, of this regulation, and AR 380–381.

### 6-31. Unique situation and compensatory measures

- *a*. A waiver will be granted only on a case–by–case basis and when the approval authority has determined that: (1) A unique or an unusual situation or factor exists requiring deviation from the established policy; and
- (2) A system of alternative compensatory measures adequately addresses the protection of classified information.

b. The alternative compensatory measures will be tailored to make sure that the intent of the protection requirement has been fulfilled by the application of other measures not addressed in the established policy. The alternative compensatory measures must show the protection afforded classified information is sufficient to reasonably deter and detect the loss or compromise of the classified information. Deviations to established requirements will be based on the consideration of risk management factors such as: criticality, sensitivity, and value of the information, analysis of the threat both known and anticipated, vulnerability to exploitation, and countermeasures benefits versus cost (both monetary and cost to national security). Waivers to requirements for records that must be maintained, concerning foreign government information, are not authorized.

### 6-32. Duration

Waivers are normally granted for a limited, specific duration, but can be approved for an indefinite period, if deemed appropriate, by the approval authority. In either case, a waiver must be revalidated no less than every five years. The revalidation will require rejustification of the unique or unusual circumstances that supports the request for a waiver and a current assessment to make sure the alternative compensatory measures do afford the protection to the classified information and that they are sufficient to reasonably deter and detect loss or compromise and meet the intent of the established policy being waived.

### 6–33. Documentation

Waivers will be documented and furnished upon request to other agencies with whom classified information or secure facilities are shared. The waiver documentation will describe the alternative compensatory measures and contain an assessment of how those alternative compensatory measures fulfill the intent of the protection requirement of the policy being waived. The continuing need for waivers will be a factor in the command inspection program. The record of waiver will be made available to inspection personnel and will be maintained for as long as the waiver is in effect.

### 6-34. Prior waivers

Waivers granted prior to the effective date of this regulation are canceled no later than one year after the effective date of this regulation. See paragraph 1–19 for waivers pertaining to any requirement beyond those contained in sections III through V of this Chapter.

### Section VII Inspections

### 6-35. Self Inspection

Heads of DA MACOMs, units, activities, and agencies will establish and maintain a self-inspection program based on program needs and the degree of involvement with classified information. The purpose of the program shall be to evaluate and assess the effectiveness and efficiency of the Command's implementation of the Army Information Security Program. Commands that originate significant amounts of classified information should be inspected at least annually.

## 6-36. Entry Exit Inspection Program and Two Person Integrity for TOP SECRET Information

The previous edition of this regulation required all commands to establish a program to inspect for the unauthorized removal of classified information. Although this program, known as the Entry/Exit Inspection Program (EEIP) is, effective by this regulation, no longer a Department of the Army-wide requirement, it does remain an effective tool that can be used in command security programs to deter and detect the unauthorized removal of classified information.

When effectively implemented the EEIP provides visibility and emphasis to the command security program. Its use is a command option. The Two Person Integrity (TPI) Program is, effective by this regulation, no longer a Department of the Army–wide requirement. Personnel are reminded that the unauthorized disclosure of TOP SECRET information can result in exceptionally grave damage to national security. TPI is a tool that can be used to better protect this high level of classification and should be considered for inclusion in command security programs. Its use is also a command option. Two persons are required, however, for the destruction of TOP SECRET material as stated in paragraph 6–29, and may be required for SAPs (see AR 380–381).

# Chapter 7 Storage and Physical Security Standards

Section I General

## 7–1. Policy

Classified information will be secured under conditions adequate to prevent access by unauthorized persons and meeting the minimum standards specified in this regulation. An assessment of the threat to the material, the location of the command, and the sensitivity of the information, will be considered when determining if the minimum requirements of this Chapter require enhancement, as determined by the local command. Based upon an assessment of the threat, the command will institute appropriate security measures designed to make unauthorized access so difficult that an intruder will hesitate to attempt to try to gain access or enhance the likelihood of discovery and apprehension if an unauthorized access is attempted.

## 7-2. Physical security policy

*a.* Physical security is intended to be built upon a system of defense, or security in depth, to provide accumulated delay time. AR 190–13, AR 190–16, and Field Manual (FM) 19–30, provide additional information on the principals of physical security. For technical assistance concerning classified material physical security storage standards, commands can contact the Army Intelligence Materiel Activity (IMA), Intelligence Materiel Management Center, Fort George G. Meade, MD 20755–5315.

b. AR 190–13 prescribes minimum uniform standards and procedures in the use of security identification cards and badges to control personnel movement into, and movement within, restricted areas. These standards and procedures are established to safeguard facilities against espionage, sabotage, damage, and theft. Security identification cards and badges may be used to control access to installations and activities. They will be used in addition to other required identification cards to military personnel, civilian DOD and contractor employees, and visitors entering installations, activities, or restricted areas, as determined by the commander concerned.

## Section II Storage Standards

### 7-3. Standards for storage equipment

General Services Administration (GSA) establishes and publishes minimum standards, specifications, and supply schedules for containers, vault doors, modular vaults, alarm systems, and associated security devices suitable for the storage and protection of classified information.

## 7–4. Storage of classified information

*a.* Classified information that is not under the personal control and observation of an authorized person, is to be guarded or stored in a locked security container, vault, room, or area, pursuant to the level of classification and this regulation by one or more of the following methods:

(1) TOP SECRET information will be stored as identified below:

(a) A GSA-approved security container with one of the following supplemental controls:

1. The location that houses the security container will be subject to continuous protection by cleared guard or duty personnel.

2. Cleared guard or duty personnel will inspect the security container once every two hours, but not in a way that indicates a pattern.

3. An Intrusion Detection System (IDS), meeting the requirements of section III of this Chapter, with personnel responding to the alarm, arriving within 15 minutes of the alarm annunciation.

4. Security-in-depth when the GSA-approved container is equipped with a lock meeting Federal Specification FF-L-2740A. See appendix J for a definition of security-in-depth.

(b) A vault, modular vault, or security room constructed in accordance with section III of this Chapter, and equipped

with an IDS with the personnel responding to the alarm within 15 minutes of the alarm annunciation if the area is covered by security-in-depth, or a 5 minute alarm response time if it is not. Other rooms that were approved under former policy for the storage of TOP SECRET in the U.S. can continue to be used.

(c) New purchases of combination locks for GSA-approved security containers, vault doors and secure rooms will conform to Federal Specification FF-L-2740A. Existing, non-FF-L-2740A mechanical combination locks will not be repaired. If they should fail, they will be replaced with locks meeting FF-L-2740A. See section IV for information on retrofitting locks (replacing locks with those meeting Federal Specification FF-L-2740A) on existing containers where the lock is not in need of repair.

(d) Under field conditions, during military operations, commanders can prescribe the measures deemed adequate to meet the storage standard contained in subparagraphs 1 and 2 above.

(2) SECRET information will be stored—

(a) In the same manner as prescribed for TOP SECRET.

(b) In a GSA-approved security container or vault without supplemental controls.

(c) In secure rooms that were approved for the storage of SECRET or CONFIDENTIAL information by the 28 February 1988 edition of this regulation, provided that the approval for storage occurred prior to 1 October 1995.

(d) Until 1 October 2002, in a non-GSA-approved container having a built-in combination lock, or in a non-GSA-approved container secured with a rigid metal lock-bar and a GSA-approved padlock with one or more of the following supplemental controls.

1. The location that houses the container is subject to continuous protection by cleared guard or duty personnel.

2. Cleared guard or duty personnel will inspect the security container once every four hours, using random times.

3. An IDS with the personnel responding to the alarm arriving within 30 minutes of the alarm. In order to reduce the risk of the lock being swapped while the container is opened, the padlock will be secured to the hasp in the locked position, or the padlock will be locked and placed inside the cabinet. Commands are encouraged to replace the non–GSA–approved cabinets with GSA–approved security containers as soon as feasible, prior to the mandatory replacement date of 1 October 2002. New lock–bar cabinets will not be fabricated from either existing or new containers, nor will any existing lock–bar container, that was not previously used for the protection of classified information, be put into use for that purpose.

(3) CONFIDENTIAL information will be stored in the same manner as prescribed for TOP SECRET and SECRET information except that supplemental controls are not required. Where lock-bar cabinets are used, in order to reduce the risk of the lock being swapped while the container is open, the padlock will be secured to the hasp in the locked position, or the padlock will be locked and placed inside the cabinet. Commands are encouraged to replace the non-GSA-approved cabinets with GSA-approved security containers as soon as feasible prior to the mandatory replacement date of 1 October 2002. New lock-bar cabinets will not be fabricated from either existing or new containers, nor will any existing lock-bar container, that was not previously used for the protection of classified information, be put into use for that purpose.

b. Specialized security equipment.

(1) GSA-approved field safes and special purpose, one and two drawer, light-weight, security containers, approved by the GSA, are used primarily for storage of classified information in the field and in military platforms, and will be used only for those or similar purposes. Such containers will be securely fastened to the structure or under sufficient surveillance to prevent their theft or compromise.

(2) GSA-approved map and plan files are available for storage of odd-sized items such as computer media, maps, charts, and classified equipment.

(3) GSA-approved modular vaults, meeting Federal Specification AA-V-2737, can be used to store classified information as an alternative to vault requirements described in section III of this Chapter.

c. Replacement of combination locks. The mission and location of the command, the classification level and sensitivity of the information, and the overall security posture of the activity, are factors used in determining the priority for replacement of existing combination locks. All system components and supplemental security measures, including electronic security systems (e.g., intrusion detection systems, automated entry control subsystems, and video assessment subsystems), and level of operations, must be evaluated by the command when determining the priority for replacement of security equipment. Section IV of this Chapter provides a matrix illustrating a prioritization scheme for the replacement of existing combination locks on GSA–approved security containers and vault doors, and can be used as a guide for this purpose. The prioritization scheme can be tailored to specific environments and sensitivity of information stored. Priority 1 requires immediate replacement. Replacement is generally considered to be accomplished when the equipment is obtained and installed within the framework of the command budget constraints, but in no event will exceed two years from the effective date of this regulation.

*d. Storage areas.* Storage areas, for bulky material containing SECRET or CONFIDENTIAL information, can have access openings secured by GSA–approved, changeable, combination padlocks (Federal Specification FF–P–110 series) or high security, key–operated padlocks (Military Specification MIL–P–43607). Other security measures are required, in accordance with paragraph 7–4a(1), above, for TOP SECRET material, and are strongly recommended for all other levels of classified material.

(1) Commands will establish administrative procedures for the control and accountability of keys and locks whenever key-operated, high-security padlocks are utilized. The level of protection provided such keys will be equivalent to that afforded the classified information being protected by the padlock. As a minimum, the following procedures will be implemented.

(a) A key and lock custodian will be appointed in writing to ensure proper custody and handling of keys and locks. (b) A key and lock control register will be maintained to identify keys for each lock and their current location and custody.

(c) Keys and locks will be audited at least quarterly.

(d) Keys will be inventoried with each change of custodian. Keys will not be removed from the premises.

(e) Keys and spare locks will be protected in a security container or other secure container;

(f) In order to reduce the risk of the padlock being swapped while the container is opened, the padlock and the key will be either placed in the security container, or the padlock will be locked to the hasp and the key either personally retained, retained at a central location, or placed inside the unlocked container.

(g) Since there is a lesser degree of risk of compromise with key operated locks, they will be changed or rotated at a minimum of once every two years, and will be immediately replaced upon loss or compromise of their keys.

(2) Section 1386 of Title 18, United States Code, makes unauthorized possession of keys, key-blanks, key-ways or locks adopted by any part of the Department of Defense for use in the protection of conventional arms, ammunition, or explosives, special weapons, and classified equipment, a criminal offense punishable by fine or imprisonment for up to 10 years, or both.

#### 7-5. Procurement of New Storage Equipment

*a*. New security storage equipment will be procured from those items listed on the GSA Federal Supply Schedule. Exceptions can be made by the MACOM commander, will be fully justified, and will be reported to DAMI–CH, who must notify the Office of the Secretary of Defense (ASD(C31)) of the details of the exception.

b. As stated in paragraph 7–4a(3) above, new lock-bar containers used to store classified material will not be fabricated from either existing or new cabinets, and existing lock-bar containers will be phased out and no longer authorized for use after 1 October 2002.

c. Nothing in this Chapter will be construed to modify existing federal supply class management assignments made under DODD 5030.47.

#### 7-6. Residential storage

Classified information will not be stored in a personal residence, on or off a military installation. Classified information will not be stored in any location outside an approved location at a U.S. Government or cleared contractor facility. Exceptions are:

*a.* In extreme and exceptional situations, a MACOM commander, or the Administrative Assistant to the Secretary of the Army for HQDA activities, can approve the temporary storage of SECRET and CONFIDENTIAL material only, in a personal residence, either on or off a military installation, or in another location that is not a U.S. Government or cleared contractor facility. This authority will not be further delegated. A validated operational requirement must exist for consideration of such requests and requests will not be approved for personal convenience. Authorization for such temporary storage must be in writing and will include written procedures for the protection of the information. The material will be stored in a GSA–approved security container and protected with an intrusion detection (alarm) system (IDS). Other methods of supplemental control can be used in place of an IDS, where the other methods provide substantially the same assurance of protection. Physical security standards, beyond the requirement for storage in a GSA–approved security container protected with an IDS, will be determined by the approving official.

b. The Secretary of the Army is the only DA official that can authorize the removal of TOP SECRET information and/or material from designated work areas for temporary storage outside a government or cleared contractor facility, to include the storage at a personal residence on a government facility. MACOM commanders can authorize the removal SECRET, and below, information and/or material from designated work areas for temporary storage outside a government or cleared contractor facility, to include the storage at a personal residence on a government facility. Where such approval is granted, to temporarily store classified information and/or material outside a designated work area at a government or cleared contractor facility, a GSA-approved security container will be furnished for storage. The container will be protected by an (IDS) as prescribed in section III of this Chapter, and written procedures addressing the appropriate protection of the information will be provided to the holder of the material. Other methods of supplemental control can be used in place of an IDS where the other methods provide the same assurance of protection. As a minimum, the written procedures concerning the storage of any level of classified information, will require the material to be under personal control, of the authorized individual, at all times when it is not secured in a GSA-approved security container. Also included will be the identification and signature receipt of the material temporarily stored, the reconciliation of the material upon its return, and the requirement that the material be returned as soon as possible after the operational requirement has ended. All authorizations, irrespective of classification level of material involved, will specify a specific expiration date.

c. Classified discussions are not permitted in personal residences, in public, in public transportation conveyances (airplane, taxi, etc.), or in any area outside approved spaces on a U.S. Government or cleared contractor facility. As an exception to this policy, and in certain situations requiring immediate contact and discussion of classified information in off-duty hours, the installation of a secure telephone unit (such as STU–III) can be authorized in personal residences to the extent that MACOM policy permits, up to, and including, the SECRET level. Only the SECARMY is authorized to permit TOP SECRET communications and document storage in personal residences. This will not be authorized for personal convenience. Where such units are permitted, care must be exercised in ensuring that unauthorized personnel, to include family members, are not within hearing distance when classified discussions take place, and that the control key for the unit is either personally retained or stored in a discrete location separate from the unit. In such cases, it can be necessary for the custodian of the unit to make notes regarding the classified discussion that occurs over the security telephone. Where this occurs, such classified notes can be retained in the personal residence only until the next duty day. If the next duty day falls during a period of more than one day, leave, Temporary Duty (TDY), or other absence, the material will be delivered for storage to a U.S. Government or cleared contractor facility prior to such absence. While in a personal residence, such classified notes will be safeguarded and under the personal, physical control of the authorized, cleared holder of the notes, at all times.

#### 7–7. Safeguarding of U.S. Classified Information Located in Foreign Countries

Except for classified information released to a foreign government or international organization, and under the safeguarding of that country or organization, U.S. classified material will be retained in foreign countries only when necessary to satisfy specific U.S. Government requirements. Commanders will take into consideration the additional risk associated with storing, discussing, and processing classified information outside the United States in establishing procedures to implement this regulation. Particular attention will be paid to the foreign release requirements of AR 380–10, making sure that classified material is not accessed by foreign personnel not authorized access to the information, keeping classified holdings to the minimum required, making sure that classified material no longer required is frequently and completely destroyed, making sure that classified discussions and processing are protected from unauthorized access from personnel working in the area, that classified discussions are conducted on secure communications equipment, and requiring that the emergency destruction plan is rehearsed and is practical for execution. U.S. classified material in foreign countries will be stored at:

a. A U.S. military installation, or a location where the United States enjoys extraterritorial status, such as an embassy or consulate.

b. A U.S. Government activity located in a building used exclusively by U.S. Government tenants, provided the building is under 24-hour control by U.S. Government and U.S. citizen personnel.

c. A U.S. Government activity located in a building not used exclusively by U.S. Government tenants nor under host government control, provided the classified material is stored in security containers approved by the GSA and is placed under 24 hour control by U.S. Government and U.S. citizen personnel.

*d.* A U.S. Government activity located in a building not used exclusively by U.S. Government tenants but which is under host government control, provided the classified material is stored in GSA–approved security containers, which are further secured in a locked room or area, to which only authorized U.S. personnel have access. The room or area will be secured with a 3–position dial combination lock meeting Federal Specification FF–L–2740A (electro–mechanical lock). MACOMs can approve the use of an existing non–FF–L–2740A lock until the lock meeting Federal Specification FF–L–2740A is installed.

### 7-8. Equipment Designations and Combinations

*a.* There will be no external mark revealing the level of classified information authorized to be stored in a given container or vault. Priorities for emergency evacuation and destruction will not be marked or posted on the exterior of storage containers, vaults, or secure rooms. For identification and/or inventory purposes, each vault or container will bear, externally, an assigned number or symbol not relating to any known security markings. This, along with the SF 702 and the "OPEN–CLOSED" or "OPEN–LOCKED" signs, are the only items permitted on the exterior of the security container. The top of the security container will not be used as a "bookshelf" or paper storage area. Storage of various non–authorized items on the top of storage containers, could lead to classified material being inadvertently left unsecured and/or mixed in with other miscellaneous material.

*b*. Combinations to security containers, vaults, and secure rooms will be changed only by individuals assigned that responsibility in writing (for example, the command security manager) and the appropriate security clearance. Combinations will be changed:

(1) When placed in use.

- (2) Whenever an individual knowing the combination no longer requires access.
- (3) When the combination has been subject to possible compromise.

(4) At least once annually.

(5) When taken out of service. When taken out of service, built-in combination locks will be reset to the standard combination 50-25-50; combination padlocks will be reset to the standard combination 10-20-30.

(6) Annually, per U.S. Central Registry, when NATO information is stored in the security container, vault, or secure room.

c. A record will be maintained for each vault, secure room, or container used for storing classified information, showing location of the container, the names, home addresses, and home telephone numbers of the individuals having knowledge of the combination. Standard Form 700 (Security Container Information) will be used for this purpose. A current record for all security containers, vault doors, and padlock combinations will be kept on SF 700.

(1) Complete part 1 and part 2A, SF 700. Include the name and signature of the person making the combination change in item 9, part 1.

(2) Part 1, SF 700 will be posted on the inside of the lock drawer of the security container.

(3) Parts 2 and 2A, SF 700 will be marked with the highest classification of material stored in the container. (4) Part 2A, SF 700 will be detached and inserted in the envelope. Part 2A, SF 700, used to record a TOP SECRET combination, will be accounted for in the same manner as other TOP SECRET documents, except that a DA Form 969 is not required. Because of the design of the SF 700, the TOP SECRET information would not be disclosed to personnel handling the sealed envelope. Upon change of a TOP SECRET combination, the old Part 2A is automatically declassified, and may be deleted from the TOP SECRET register (or DA Form 3964).

(5) Only part 1, SF 700 need be completed for security containers storing two-person control material. Parts 2 and 2A need be used only if there is a specific need for recording the combination.

*d.* The combination of a container, vault or secure room used for the storage of classified information will be treated as information having a classification equal to the highest classification level of the classified information to be stored inside. Such written records are classified and will be stored in containers approved for the storage of classified information, at the appropriate classification level, at the next higher headquarters. Written records of combinations will not be personally retained in wallets, purses, briefcases, desk drawers, on calendars or note pads, or written "in code" or foreign languages and stored in unapproved locations.

e. Access to the combination of a vault or container used for the storage of classified information will be granted only to those individuals who are authorized access to the classified information that is to be stored inside.

*f*. Entrances to secure rooms or areas, will be either under visual control at all times during duty hours, to preclude entry by unauthorized personnel, or the entry will be equipped with electric, mechanical, or electro-mechanical access control devices to limit access during duty hours. Section III, of this Chapter, provides standards for these access control devices. Electronically actuated locks (for example, cipher and magnetic strip card locks) and other such locking devices used primarily for duty-hours access control do not afford by themselves the required degree of protection for classified information and must not be used either during or after duty hours as a substitute for the locks prescribed in paragraph 7–4b.

### 7-9. Repair of Damaged Security Containers

Neutralization of lock-outs, or repair of any damage, that affects the integrity of a security container approved for storage of classified information, will be accomplished only by authorized persons who have been the subject of a trustworthiness determination, in accordance with AR 380-67, or are continuously escorted while so engaged.

*a.* With the exception of frames bent through application of extraordinary stress, a GSA-approved security container manufactured prior to October 1991 (identified by a silver GSA label with black lettering affixed to the exterior of the container) is considered to have been restored to its original state of security integrity as follows:

(1) All damaged or altered parts, for example, the locking drawer, drawer head, or lock, are replaced.

(2) The safe has been drilled immediately adjacent to or through the dial ring to neutralize a lock-out, a replacement lock meeting FF-L-2740A is used, and the drilled hole is repaired with a tapered, hardened tool-steel pin, or a steel dowel, drill bit, or bearing, with a diameter slightly larger than the hole, and of such length that when driven into the hole there will remain at each end of the rod a willow recess not less than <sup>1</sup>8-inch nor more than &frac316;-inch deep to permit the acceptance of substantial welds, and the rod is welded both on the inside and outside surfaces. The outside of the drawer head must then be puttied, sanded, and repainted in such a way that no visible evidence of the hole or its repair remains on the outer surface.

*b*. In the interests of cost efficiency, the procedures identified in subparagraph a(2) above, should not be used for GSA-approved security containers purchased after October 1991, distinguished by a silver GSA label with red lettering affixed to the outside of the container control drawer, until it is first determined whether warranty protection still applies. To make this determination, it will be necessary to contact the manufacturer and provide the serial number and date of manufacture of the container. If the container is under warranty, a lockout will be neutralized using the procedures described in the Federal Standard FED–STD–809 (Neutralization and Repair of GSA Approved Containers), dated 1 April 1998.

c. Unapproved modification or repair of security containers and vault doors is considered a violation of the container or door's integrity and the GSA label will be removed. Thereafter, these safes will not be used to protect classified information except as otherwise authorized in this regulation.

d. For technical assistance concerning classified material physical security storage standards, commands can contact the Interagency Advisory Committee on Security Equipment (IACSE). The designated DA representatives to the

IACSE, Security Equipment and Locking Systems (SEALS) subcommittee can be reached through the Army Intelligence Materiel Activity, Intelligence Materiel Management Center, Fort George G. Meade, MD 20755–5315.

#### 7–10. Maintenance and Operating Inspections

MACOMs will establish procedures concerning repair and maintenance of classified material security containers, vaults, and secure rooms, to include a schedule for periodic maintenance. The following guidelines pertain to spotting repair and maintenance problems that will be addressed outside the regular maintenance schedule.

*a*. Security containers are usually serviceable for at least 25 years, if properly maintained. The life span of the container is often cut short by lock or locking bolt linkage malfunctions that require neutralization of the container. Most of these problems can be detected in their early stages, and definite symptoms can warn of a developing problem. Users should be alert for these symptoms, and if any of them are detected, the users should immediately contact their supporting maintenance activity for help. It is important to never use force to try to correct the problem. Critically needed material should not be stored in containers showing any of these symptoms, since they cannot be depended upon to open again. Should that occur, the user can be faced with a lockout.

b. Users should watch for the following signs of trouble:

(1) A dial that is unusually loose or difficult to turn.

- (2) Any jiggling movement in the dial ring. This is often detected when a twist motion is applied to the dial.
- (3) Difficulty in dialing the combination or opening the container. Examples are:
- (a) The need to dial the combination more than once, when human error is not at fault.
- (b) The need to dial on numbers that are slightly above or below the correct number in the combination.
- (4) Difficulty with the control drawer or other drawers. Examples are:

(a) Drawers rubbing against container walls. This can be caused if the container is not leveled, or the tracks or cradles are not properly aligned.

(b) Problems with opening or closing drawers because the tracks or cradles need lubricant, material is jammed in behind the drawer, or the internal locking mechanism is tripped.

(5) Difficulty in locking the control drawer. Examples are:

(a) The control drawer handle or latch will not return to the locking position when the drawer is shut.

(b) On Sargent and Greenleaf (S&G) or other similar locks, the butterfly in the center of the dial will not turn after the control drawer is shut and the dial has been turned to zero.

(c) The locking bolts move roughly, slip, or drag, or the linkage is burred or deformed.

c. Commands will periodically remind users of containers about the above guidelines.

### 7-11. Turn-in or Transfer of Security Equipment

In addition to having combinations reset before turn-in (see paragraph 7–8b(5)), security equipment will be inspected before turn-in or transfer to ensure that classified material is not left in the container. The turn-in procedure will include removal of each container drawer and inspection of the interior to make sure that all papers and other material are removed and that the container is completely empty. Vaults, secure rooms, incinerators, shredders, or other classified material destruction devices, as well as the rooms in which they are located, will be thoroughly inspected to make sure that no classified material remains. A written, signed record certifying that this inspection has been accomplished and that no classified material remains, will be furnished to the command security manager and filed for two years.

### Section III Physical Security Standards

#### 7–12. General

This section provides the general construction standards for areas approved for the open storage of classified information, general standards for intrusion detection (alarm) systems (IDS) used in areas in which classified information is stored, access control standards, and priorities for the replacement of locks on security containers. Classified material will be stored to the maximum extent feasible in GSA-approved security containers. Open storage areas will only be approved when storage in other approved security containers is not feasible due to the size, shape, or volume of material stored.

## 7-13. Vault and Secure Room (Open Storage Area) Construction Standards

#### a. Vault.

(1) Floor and Walls Eight inches of concrete reinforced to meet current standards. Walls are to extend to the underside of the roof slab above.

(2) Roof Monolithic reinforced concrete slab of thickness to be determined by structural requirements, but not less than the floors and walls.

(3) Ceiling The roof or ceiling must be reinforced concrete of a thickness to be determined by structural requirements, but not less than the floors and walls.

(4) Door and Frame Vault door and frame unit will conform to Federal Specification AA-D-2757 Class 8-vault door, or Federal Specification AA-D-600 Class 5 vault door.

#### b. Secure room

(1) *Floor, Walls, and Roof.* The walls, floor, and roof construction of secure rooms must be of permanent construction materials, i.e., plaster, gypsum wallboard, metal panels, hardboard, wood, plywood, or other materials offering resistance to, and evidence of, unauthorized entry into the area. Walls will be extended to the true ceiling and attached with permanent construction materials, with mesh or 18 gauge expanded steel screen.

(2) Ceiling. The ceiling will be constructed of plaster, gypsum, wallboard material, hardware, or other similar material that the command security manager judges to be of equivalent strength.

(3) *Doors.* The access door to the room will be substantially constructed of wood or metal. The hinge pins of out-swing doors will be pinned, brazed, or spot-welded to prevent removal. The access door will be equipped with a built-in GSA-approved combination lock meeting Federal Specification FF-L-2740A. For open storage areas approved under previous standards, the lock can be the previously approved GSA combination lock. However, upon retrofit, the door must be fitted with a lock meeting Federal Specification FF-L-2740A (See paragraph 7–21 for priorities for replacement of such locks). Doors, other than the access door, will be secured from the inside. For example, by using a dead bolt lock, panic dead bolt lock, or rigid wood or metal bar which extends across the width of the door, or by any other means that will prevent entry from the outside. Key operated locks that can be accessed from the exterior side of the door are not authorized.

(4) *Windows*. Windows which are less than 18 feet above the ground when measured from the bottom of the window, or are easily accessible by means of objects directly beneath the windows, will be constructed from or covered with materials which will provide protection from forced entry. The protection provided to the windows need be no stronger than the strength of the contiguous walls.

(5) *Openings*. Utility openings, such as ducts and vents, will be kept at less than a person-passable, 96 square inches, opening. Openings larger than 96 square inches will be hardened in accordance with Military Handbook 1013/1A, which provides guidance to ensure that appropriate physical security considerations are included in the design of facilities.

#### 7–14. Intrusion Detection System Standards

a. An Intrusion Detection System (IDS), often referred to as an alarm, must detect an unauthorized penetration in the secured area. An IDS complements other physical security measures and consists of the following:

- (1) Intrusion Detection Equipment (IDE).
- (2) Security forces.
- (3) Operating procedures.

b. System functions. IDS components operate as a system with the following four distinct phases:

- (1) Detection.
- (2) Communications.
- (3) Assessment
- (4) Response.

c. These elements are equally important, and none can be eliminated if an IDS is to provide an acceptable degree of protection.

(1) Detection. The detection phase begins as soon as a detector or sensor reacts to stimuli it is designed to detect. The sensor alarm condition is then transmitted over cabling located within the protected area to the Premise Control Unit (PCU). The PCU may service many sensors. The PCU, and the sensors it serves, comprise a "zone" at the monitor station. This will be used as the definition of an alarmed zone for purposes of this regulation.

(2) Reporting. The PCU receives signals from all sensors in a protected area and incorporates these signals into a communication scheme. Another signal is added to the communication for supervision, to prevent compromise of the communication scheme by tampering or injection of false information by an intruder. The supervised signal is sent by the PCU through the transmission link to the monitor station. Inside the monitor station either a dedicated panel or central processor monitors information from the PCU signals. When an alarm occurs, an annunciator generates an audible and visible alert to security personnel. Alarms result normally from intrusion, tampering, component failure, or system power failure.

(3) Assessment The assessment period is the first phase that requires human interaction. When alarm conditions occur, the operator assesses the situation and dispatches the response force, as necessary.

(4) Response The response phase begins as soon as the operator assesses an alarm condition. A response force must immediately respond to all alarms. The response phase must also determine the precise nature of the alarm and take all measures necessary to safeguard the secure area.

#### 7-15. Selection of equipment

*a*. As determined by the commander, and in accordance with the minimum standards established by this regulation, all areas that reasonably afford access to the container, or where classified data is stored, are to be protected by IDS unless continually occupied. Prior to the installation of an IDS, commanders, or their designated personnel, will consider the threat, the vulnerabilities, and any in-depth security measures, and will perform a risk analysis to determine if IDS is appropriate to the situation.

b. Acceptability of Equipment. All IDE must be UL-listed, or equivalent, and approved by the Department of the Army or authorized U.S. Government contractor. Government installed, maintained, or furnished systems are acceptable.

## 7-16. IDS Transmission

a. Transmission Line Security. When the transmission line leaves the facility and traverses an uncontrolled area, Class I or Class II line supervision will be used.

(1) *Class I.* Class I line security is achieved through the use of Data Encryption Standard (DES) or an algorithm based on the cipher feedback or cipher block chaining mode of encryption. Certification by the National Institute of Standards (NIST) or another independent testing laboratory is required.

(2) *Class II.* Class II line supervision refers to systems in which the transmission is based on pseudo random generated tones or digital encoding using an interrogation and response scheme throughout the entire communication, or UL Class AA line supervision. The signal will not repeat itself within a minimum six-month period. Class II security will be impervious to compromise using resistance, voltage, current, or signal substitution techniques.

*b. Internal Cabling.* The cabling between the sensors and the PCU must be dedicated to the IDS and must comply with national and local code standards.

c. Entry Control Systems. If an entry control system is integrated into an IDS, reports from the automated entry control system must be subordinate in priority to reports from intrusion alarms.

*d. Maintenance Mode.* When an alarm zone is placed in the maintenance mode, the condition will be signaled automatically to the monitor station. The signal must appear as an alarm or maintenance message at the monitor station and the IDS will not be securable while in the maintenance mode. The alarm or message must be continually visible at the monitor station throughout the period of maintenance. A standard operating procedure will be established to address appropriate actions when maintenance access is indicated at the panel. All maintenance periods will be archived in the system. A self-test feature will be limited to one second per occurrence.

e. Annunciation of Shunting or Masking Condition. Shunting or masking of any internal zone or sensor must be appropriately logged or recorded in archive. A shunted or masked internal zone or sensor must be displayed as such at the monitor station throughout the period the condition exists whenever there is a survey of zones or sensors.

*f. Indications.* Indications of alarm status will be revealed at the monitoring station and optionally within the confines of the secure area.

g. Power Supplies Primary power for all IDE will be commercial AC or DC power. In the event of commercial power failure at the protected area or monitor station, the equipment will change power sources without causing an alarm indication.

(1) Emergency Power Emergency power will consist of a protected independent backup power source that provides a minimum of 4 hours operating power battery and/or generator power. When batteries are used for emergency power, they will be maintained at full charge by automatic charging circuits. The manufacturers periodic maintenance schedule will be followed and results documented.

(2) Power Source and Failure Indication An illuminated indication will exist at the PCU of the power source in use (AC or DC). Equipment at the monitor station will indicate a failure in power source, a change in power source, and the location of the failure or change.

*h*. Component Tamper Protection IDE components located inside or outside the secure area will be evaluated for a tamper protection requirement. If access to a junction box or controller will enable an unauthorized modification, tamper protection will be provided.

#### 7–17. System Requirements

*a.* Independent Equipment When many alarmed areas are protected by one monitor station, secure room zones, areas in which classified information is stored, must be clearly distinguishable from the other zones to ensure a priority response. All sensors will be installed within the protected area.

b. Access and/or Secure Switch and PCU No capability is to exist to allow changing the access status of the IDS from a location outside the protected area. All PCUs must be located inside the secure area and are to be located near the entrance. Only assigned personnel will initiate changes in access and secure status. Operation of the PCU can be restricted by the use of a device or procedure that verifies authorized use. In the secure mode, any unauthorized entry into the space will cause an alarm to be transmitted to the monitor station.

c. Motion Detection Protection Secure areas that reasonably afford access to the container or where classified data is stored, are to be protected with motion detection sensors, e.g., ultrasonic and/or passive infrared. Use of dual

technology is authorized when one technology transmits an alarm condition independently from the other technology. A failed detector will cause an immediate and continuous alarm condition.

d. Protection of Perimeter Doors Each perimeter door will be protected by a Balanced Magnetic Switch (BMS) that meets the standards of UL 634.

e. Windows All readily accessible windows (within 18 feet of ground level) will be protected by an IDS, either independently or by the motion detection sensors in the space.

*f.* IDS Requirements for Continuous Operations Facility A continuous operations facility may not require an IDS. This type of secure area should be equipped with an alerting system if the occupants cannot observe all potential entrances into the room. Duress devices could also be required depending upon the situation.

g. False and/or Nuisance Alarm Any alarm signal transmitted in the absence of detected intrusion, or identified as a nuisance alarm, is a false alarm. A nuisance alarm is the activation of an alarm sensor by some influence for which the sensor was designed, but which is not related to an intrusion attempt. All alarms will be investigated and the results documented. The maintenance program for the IDS must make sure that incidents of false alarms do not exceed one in a period of 30 days per zone.

### 7-18. Installation, Maintenance and Monitoring

*a.* IDS Installation and Maintenance Personnel Alarm installation and maintenance will be accomplished by U.S. citizens who have been subjected to a trustworthiness determination, in accordance with AR 380–67 (See also DODD 5200.2–R).

b. Monitor Station Staffing The monitor station is to be supervised continuously by U.S. citizens who have been subjected to a trustworthiness determination in accordance with the regulations noted in subparagraph a, above.

## 7-19. Access Controls While Material is Not Secured in Security Containers

This section applies to open storage areas such as vaults and secure rooms. It can also apply, at the MACOM's option, to other areas of security interest, such as areas in which significant amounts of classified material or especially sensitive material are routinely accessed. This section does not apply to open storage of SAPs material. See appendix I and AR 380–381 regarding open storage of SAPs material and information.

*a.* The perimeter entrance will be under visual control at all times during working hours so as to deny entry to unauthorized personnel. This can be accomplished by several methods, such as an employee work station, guard, closed circuit television (CCTV), or a day access combination lock when there are persons present in the area but the entrance is not under visual control. Regardless of the method used, an access control system will be used on the entrance. Note: Uncleared persons will be escorted within the facility by a cleared person, who is familiar with the security procedures at the facility, and an announcement, either auditory or visual, will be used to alert others of the uncleared person's presence.

*b*. An automated entry control system (AECS) can be used to control admittance during working hours instead of visual or other methods of control. That AECS must meet the criteria stated below. Further guidance can be obtained from IACSE, SEALS subcommittee, at Fort Meade, MD (see paragraph 7–9d for contact information). The automated entry control system must identify an individual and authenticate the person's authority to enter the area through the use of one of the following:

(1) ID Badges or Key Cards The identification (ID) badge or key card must use embedded sensors, integrated circuits, magnetic stripes, or other means of encoding data that identifies the facility and the individual to whom the card is issued.

(2) Personal Identity Verification Personal identity verification (biometrics devices) identifies the individual requesting access by some unique personal characteristic, such as:

- (a) Fingerprinting.
- (b) Hand geometry.
- (c) Handwriting.
- (d) Retina scans.

(e) Voice recognition. A biometrics device can be particularly appropriate for access to areas in which highly sensitive information is located.

c. In conjunction with subparagraph (2), above, a personal identification number (PIN) can be required. The PIN must be separately entered into the system by each individual using a keypad device and will consist of four or more digits, randomly selected, with no known or logical association with the individual. The PIN must be changed or discontinued when it is believed to have been compromised, subjected to compromise, or the individual no longer requires access.

*d*. Authentication of the individual's authorization to enter the area must be accomplished within the system by inputs from the ID badge/card, the personal identity verification device, or the keypad with an electronic database of individuals authorized to enter the area. Procedures will be established, in writing, for removal of the individual's authorization to enter the area upon reassignment, transfer, or termination, or when the individual's access is suspended, revoked, or downgraded to a level lower than the required access level.

*e*. Protection must be established and maintained for all devices or equipment which constitute the entry control system. The level of protection can vary depending upon the type of device or equipment being protected. This can be accomplished by the following:

(1) Location where authorization data and personal identification or verification data is entered or inputted, stored, or recorded, is protected.

(2) Card readers, keypads, communication or interface devices located outside the entrance to a controlled area will have tamper resistant enclosures and be securely fastened to the wall or other permanent structure. Control panels located within a controlled area will require only a minimal degree of physical security protection sufficient to preclude unauthorized access to the mechanism.

(3) Keypad devices will be designed or installed in such a manner that an unauthorized person in the immediate vicinity cannot observe the selection of input numbers.

(4) Systems that use transmission lines to carry access authorizations, personal identification data, or verification data between devices or equipment located outside the controlled area will have line supervision (see paragraph 7-16 for explanation of line supervision).

(5) Electric strikes used in access control systems will be of heavy duty, industrial grade.

*f.* Access to records and information concerning encoded ID data and PINs will be restricted. Access to identification or authorizing data, operating system software or any identifying data associated with the entry control system will be limited to the fewest number of personnel as possible. Such data or software will be kept secure when unattended.

g. Records will be maintained reflecting active assignment of ID badge/card, PIN, level of access, and similar system–related records. Records concerning personnel removed from the system will be retained for 90 days. Records of entries will be retained for at least 90 days or until any investigations of system violations and incidents have been investigated, resolved and recorded.

*h.* Personnel who are the first to enter or last to leave an area will be required to secure the entrance or exit point. Authorized personnel who permit another individual to enter the area are responsible for confirmation of clearance, need-to-know, and access. Commanders can approve the use of standardized AECS which meet the criteria specified below. MACOMs and the Administrative Assistant to the Secretary of the Army, for HQDA activities, can approve deviations to these standards, based upon compensatory measures that provide a commensurate level of assurance of access control. Criteria for standardized AECS is as follows:

(1) For a Level 1 key card system, the AECS must provide a .95 probability of granting access to an authorized user providing the proper identifying information within three attempts. In addition, the system must ensure an unauthorized user is granted access with less than 0.05 probability after three attempts to gain entry have been made.

(2) For a Level 2 key card and PIN system, the AECS must provide a .97 probability of granting access to an authorized user providing the proper identifying information within three attempts. In addition, the system must ensure an unauthorized user is granted access with less than .010 probability after three attempts to gain entry have been made.

(3) For a Level 3 key card and PIN and biometrics identifier system, the AECS must provide a .97 probability of granting access to an unauthorized user providing the proper identifying information within three attempts. In addition, the system must ensure an authorized user is granted access with less than .005 probability after three attempts to gain entry have been made.

*i*. Electric, Mechanical, or Electromechanical Access Control Devices. Electric, mechanical, or electromechanical devices which meet the criteria stated below, may be used to control admittance to secure areas during duty hours, if the entrance is under visual or other command approved system of control by cleared authorized personnel located in the area. These devices are also acceptable to control access to selected or otherwise compartmented areas within a secure area. Nothing in this statement is intended to modify the policy stated in AR 380–28 for the protection of sensitive compartmented information. Access control devices will be installed in the following manner:

(1) The electronic control panel containing the mechanical mechanism by which the combination is set is to be located inside the area. The control panel, located within the area, will require only minimal degree of physical security designed to preclude unauthorized access to the mechanism.

(2) The control panel will be installed in such a manner, or have a shielding device mounted, so that an unauthorized person in the immediate vicinity cannot observe the setting or changing of the combination.

(3) The selection and setting of the combination will be accomplished by an individual cleared at the same level as the highest classified information controlled within.

(4) Electrical components, wiring included, or mechanical links (cables, rods and so on) should be accessible only from inside the area, or, if they traverse an uncontrolled area they should be secured within protecting covering to preclude surreptitious manipulation of components.

#### 7–20. Minimum standards for deviations to construction standards for open storage areas

Where deviations to the general open storage areas, vaults and strong rooms, are approved by the MACOM or the Administrative Assistant to the Secretary of the Army, for HQDA activities, the following standards, as a minimum,

will be satisfied and the areas will be certified as being designed to sufficiently deter, detect, or delay entry, to unauthorized persons from gaining access to the classified information stored therein.

*a. Construction:* The perimeter walls, floors, and ceiling will be permanently constructed and attached to each other. All construction will be done in a manner as to provide visual evidence of unauthorized penetration.

*b. Doors:* Doors will be constructed of wood, metal, or other solid material. Entrance doors will be secured with a built-in GSA approved three-position combination lock. Other doors will be secured from the inside with either a combination lock, panic hardware, a dead bolt, or a rigid wood or metal bar which extends across the width of the door. Any door that permits entry from the outside must be secured with a GSA approved three position built-in combination lock. In unusual circumstances in which a built-in combination lock is not feasible, the designated official can approve a GSA approved three-position combination padlock. Where such padlocks are used, they will be secured in a security container or approved open storage area or kept in the locked position on the hasp during hours in which the area is opened. Key operated or cipher locks are not permitted on any door that provides access from the exterior of the open storage area.

c. Vents, Ducts, and Miscellaneous Openings: All vents, ducts, and similar openings in excess of 96 square inches, and/or over six inches in its smallest dimension, that enter or pass through an open storage area, will be protected with either bars, expanded metal grills, or an intrusion detection system. Sound baffles will be used if classified discussions occur in an area in which the sound carries outside the range of authorized personnel.

d. Windows:

(1) All windows which might reasonably afford visual observation of classified activities within the facility will be made opaque or equipped with blinds, drapes, or other coverings.

(2) Windows at ground level will be constructed from or covered with materials which provide protection from forced entry. The protection provided to the windows will be as strong as the strength of the contiguous walls.

(3) Approved open storage areas which are located within an Army command, located on an Army installation or other Army controlled compound, may eliminate the requirement for forced entry protection if the windows are made inoperable either by permanently sealing them or equipping them on the inside with a locking mechanism. In either situation, sealing or locking, the windows must be covered by an IDS, either independently or by the motion detection sensors within the area.

e. IDS. An IDS must be used (see standards in paragraph 7–12) to provide complete coverage of the entire open storage area. Minimum response time for any area storing classified information is contained in paragraph 7–4a. Locations outside the United States and all areas that store information which is of special risk of theft or espionage, or in areas of high risk, must seriously consider reducing the minimum response time.

## Section IV

## Lock Replacement Priorities

### 7-21. Priorities for Replacement of Locks

All newly purchased GSA approved security containers are equipped with the electromechanical GSA–approved combination lock meeting Federal Specification FF–L–2740A. New purchases of combination locks for GSA–approved security containers, vault doors and secure rooms will conform to Federal Specification FF–L–2740A. Existing mechanical combination locks that do not meet this specification will not be repaired. If they do fail, they will be replaced with locks meeting FF–L–2740A. Army commands will be advised by HQDA of the policy concerning the retrofitting of existing security containers with the new electromechanical locks meeting Federal Specification, FF–L–2740A. This section contains the recommended general priorities for the lock retrofit program (See figure 7–1). In accordance with the Army Intelligence Materiel Activity (AIMA), it is to be implemented upon notification, or as command funds become available and upgrading of locks is assessed as a component of the command security program. Where individual situations are assessed as requiring a modification to the general priorities, they can be made at command option, unless otherwise instructed by HQDA. An individual situation could include a risk assessment of such factors as amount of material held, sensitivity of the information, threat to the information, environment in which the container is located, and depth of other security features that control access to the container or area. Priorities range from 1 to 4, with 1 being the highest and 4 the lowest.

# Lock Replacement Priorities

In the United States and Its Territories

ITEM TS/SA	Ps	TS	S/SA	Ps	S-C
Vault Doors	1	1	3 <sup>1</sup>	4 <sup>1</sup>	
Containers (/	۹)*	3 <sup>2</sup>	4 <sup>2</sup>	4 <sup>2</sup>	4 <sup>2</sup>
Containers (E	3)**	1	1	1	2
Crypto1	1	2	2		

Outside the United States and Its Territories

ITEM TS/SAPs	TS	S/SAPs		S-C
Vault Doors 1	1	2 <sup>1</sup>	2 <sup>1</sup>	
Containers (A)*	2	2	3 <sup>2</sup>	3
Containers (B)**	1	1	1	2 <sup>2</sup>
Crypto1 1	2 <sup>2</sup>	2 <sup>2</sup>		
High Risk Areas	1	1	1	1

\*A - Located in a controlled environment where the Department of the Army, or other element of the Department of Defense, has the authority to prevent unauthorized disclosure of classified information. The command can control or deny access to the space, post guards, require identification, challenge presence, inspect packages, program elevators, or take other reasonable measures necessary to deny unauthorized access.

\*\*B - Located in an uncontrolled area without the perimeter security measures identified in \*A, above.

<sup>1</sup> If the vault door lock is the only GSA approved three-position combination lock that secures the material, the priority will be designated as 1. A lower priority is assigned based on command assessment in situations in which the material is secured in an approved container with a GSA approved three position combination lock that is located within the vault.

 $^{2}$  The higher priority will be used at command option based upon a weighing of the factors involved, such as: the sensitivity of the material stored, volume of material, threat to the information, and environment in which the container is located.

Figure 7-1. Lock Replacement Priorities

# Chapter 8 Transmission and Transportation

Section I

## Methods of Transmission and Transportation

## 8–1. Policy

Classified information will be transmitted and transported only as specified in this Chapter. COMSEC information will be transmitted in accordance with AR 380–40. Special Access Programs material will be transmitted and transported in accordance with appendix I of this regulation, AR 380–381, and applicable SAPs procedure guides. Commands will establish local procedures to meet the minimum requirements to minimize risk of compromise while permitting use of the most effective transmission or transportation means. External, street side, collection boxes, for instance, U.S. Mail boxes, will not be used for the dispatch of classified information. Commands will develop procedures to protect incoming mail, bulk shipments, and items delivered by messenger, until a determination is made whether classified information is contained therein. Screening points will be established to limit access of classified information to only cleared personnel.

## 8-2. TOP SECRET Information

TOP SECRET information will be transmitted only by:

a. A cryptographic system authorized by the Director, NSA, or a protected distribution system designed and installed to meet approved NSA standards. This applies to voice, data, message, and facsimile transmissions.

b. The Defense Courier Service (DCS) (see DODD 5200.33-R).

c. Authorized command courier or messenger services.

d. The Department of State Diplomatic Courier Service.

e. Cleared U.S. military personnel and U.S. Government civilian employees, traveling by surface transportation, or traveling on a conveyance owned, controlled, or chartered by the U.S. Government or DOD contractors.

f. Cleared U.S. military personnel and U.S. Government civilian employees on scheduled commercial passenger aircraft.

g. Cleared DOD contractor employees within and between the United States and its territories, when the transmission has been authorized, in writing, by the appropriate Cognizant Security Agency (CSA), or a designated representative. For DA contractors, the CSA is generally the Defense Security Service (DSS).

## 8-3. SECRET information

SECRET information can be transmitted by:

a. Any of the means approved for the transmission of TOP SECRET information.

b. U.S. Postal Service registered mail, within and between the 50 States, the District of Columbia, and the Commonwealth of Puerto Rico.

c. U.S. Postal Service express mail, within and between the 50 States, the District of Columbia, and the Commonwealth of Puerto Rico. The "Waiver of Signature and Indemnity" block on the U.S. Postal Service express mail label 11–B, will not be executed under any circumstances. The use of external, street side, express mail collection boxes is prohibited.

*d.* U.S. Postal Service registered mail, through Army, Navy, or Air Force Postal Service facilities, for instance, APO/FPO, outside the United States and its territories, so that the information does not, at any time, pass out of U.S. citizen control, and does not pass through a foreign postal system or any foreign inspection.

e. United States Postal Service and Canadian registered mail, with registered mail receipt between U.S. Government and Canadian Government installations in the U.S. and Canada.

f. As an exception, in urgent situations requiring next-day delivery, an overnight or next-day delivery service, that is a current holder of a GSA contract for overnight delivery of material for the Executive Branch, provided that the delivery service is U.S. owned and operated and provides automated in-transit tracking of the material. These companies are not required to be cleared and generally are to be considered uncleared. Their employees are not cleared and are not required to be U.S. citizens, and the companies are not required to meet the storage requirements contained in this regulation. For the purpose of this section of the regulation, an urgent situation exists when the classified material must be received by the next day, there is no other authorized means to make the delivery, excluding the handcarrying by authorized personnel, the delivery company assures delivery by the required date, and the transmission complies with the provision of Title 39, U.S. Code (USC), section 320.6, Postal Services, as amended. The sender will comply with the requirement contained in paragraph 8–10, to address the package to the command or activity, and not address it to an individual. Since delivery services usually require the building number and name of recipient, the sender will contact the recipient to ensure that an authorized and appropriately cleared person will be available to sign for the material, and they will verify the authorized address to make sure that it is displayed correctly on the package label. Unless it is not possible, for example, if the material is needed on a weekend and the mailroom is not in operation then, the package label will be addressed to a supporting mailroom. The release signature block on the receipt label will not be executed under any circumstances. Executing the release signature block, ensures that someone, but not necessarily the addressee, if the addressee is unavailable when the package is delivered, signs for the package. These precautions are required because uncleared, commercial overnight delivery services can deliver the package directly to the person named, and building identified, on the label, or to whomever signs for the material, if the addressee is unavailable when the material is delivered. U.S. Postal Service mail is delivered or picked-up by a centralized command mailroom where personnel who open mail are cleared and the material is properly safeguarded until opened. The use of external, street side, commercial delivery service collection boxes is prohibited. Note: In many situations, the United States Postal Service express mail can meet the next day delivery standards and should be used, as noted in subparagraph c, above.

g. Carriers authorized to transport SECRET information by way of a Protective Security Service (PSS) under the National Industrial Security Program (NISP). This method is authorized only within the United States boundaries when other methods are impractical.

*h*. Appropriately cleared contractor employees, provided that the transmission meets the requirements specified in DODD 5220.22–R and DODD 5220.22–M (NISPOM).

*i*. U.S. Government and U.S. Government contract vehicles, including aircraft, ships of the U.S. Navy, civil service-operated U.S. Navy ships, and ships of United States registry. Appropriately cleared operators of vehicles, officers of ships, or pilots of aircraft, who are U.S. citizens, may be designated as escorts, provided the control of the carrier is maintained on a 24-hour basis. The escort will protect the shipment at all times, through personal observation or authorized storage, to prevent inspection, tampering, pilferage, or unauthorized access. Observation of the shipment is not required during flight or sea transit, provided it is loaded into a compartment that is not accessible, to any unauthorized persons, or in a specialized secure, safe-like container. The escort will, if possible, observe the loading of the shipment.

## 8–4. CONFIDENTIAL information

CONFIDENTIAL information may be transmitted by:

*a.* Means approved for the transmission of SECRET information. However, U.S. Postal Service registered mail will be used for CONFIDENTIAL material only as indicated below:

(1) NATO CONFIDENTIAL information. If NATO CONFIDENTIAL material is sent between U.S. Government activities, within the Continental United States, its territories, and the District of Columbia, it can be sent by first class mail. The caveat, "POSTMASTER: RETURN SERVICE REQUESTED" will be affixed to the outer wrapper.

(2) Other CONFIDENTIAL material sent to and from FPO or APO addressees, located outside the U.S. and its territories.

(3) Other CONFIDENTIAL material when the originator is uncertain that the addressee's location is within U.S. boundaries or knows the addressee's location is outside U.S. boundaries.

b. United States Postal Service certified mail (or registered mail, if required above) for material addressed to DOD contractors or non-DOD agencies.

c. United States Postal Service first class mail between DOD component locations anywhere in the U.S., its territories, and the District of Columbia. The use of external, street side, postal collection mailboxes is prohibited. The outer envelope or wrappers will be endorsed, where possible, in letters larger than the text on the address of the envelope: "POSTMASTER: RETURN SERVICE REQUESTED."

d. Within United States boundaries, commercial carriers that provide a Constant Surveillance Service (CSS).

*e*. In the custody of commanders or masters of ships of United States registry, who are United States citizens. CONFIDENTIAL information shipped on ships of U.S. registry, cannot pass out of United States control. The commanders or masters must sign a receipt for the material and agree to:

(1) Deny access to the CONFIDENTIAL material by unauthorized persons, including customs inspectors, with the understanding that CONFIDENTIAL cargo, that would be subject to customs inspection, will not be unloaded.

(2) Maintain control of the cargo until a receipt is obtained from an authorized representative of the consignee.

## 8-5. NATO restricted material

NATO restricted material can be transmitted within the United States and to designated APO/FPO addresses, by U.S. first class mail, single wrapped with a notation on the envelope, "POSTMASTER: RETURN SERVICE RE-QUESTED." When to or from areas outside the United States and APO/FPO addresses, NATO restricted material will be sent, double-wrapped, to NATO addressees through United States or NATO member postal systems.

## Section II

### Transmission of Classified Material to Foreign Governments

### 8-6. General

Classified information or material approved for release to a foreign government, in accordance with AR 380–10, will be transferred between authorized representatives of each government in compliance with the provisions of this Chapter. Each contract, agreement, or other arrangement, that involves the release of classified material to foreign entities, will either contain detailed transmission instructions, or require that a separate transportation plan be approved, by the appropriate security and transportation officials and the recipient government, prior to release of the material. Transportation plan requirements are outlined in paragraph 8–7h. (See DOD TS–5105.21–M–2 for further guidance regarding SCI.)

## 8–7. Procedures

*a.* Classified information or material to be released directly to a foreign government representative will be delivered or transmitted only to a person who has been designated in writing by the recipient government to sign for and assume custody and responsibility on behalf of said government (referred to as the designated government representative). This written designation will contain assurances that such person has a security clearance at the appropriate level, and that the person will assume full security responsibility for the material on behalf of the foreign government. The recipient will be required to execute a receipt for the material, regardless of the level of classification. The foreign government can designate a freight forwarder as their representative for receipt of freight.

b. Classified material that is suitable for transfer by courier or postal service, in accordance with this regulation, and which cannot be transferred directly to a foreign government's designated representative, will be transmitted to:

(1) An embassy, consulate, or other official agency of the recipient government having extraterritorial status in the United States.

(2) A U.S. embassy or a U.S. military organization in the recipient country or in a third-party country for delivery to a designated representative of the recipient government.

c. The shipment of classified material as freight by truck, rail, aircraft, or ship, will be in compliance with the following:

(1) Army officials authorized to approve a Foreign Military Sales (FMS) transaction that involves the delivery of U.S. classified material to a foreign purchaser will, at the beginning of negotiations or consideration of a proposal, consult with DOD transportation authorities (Military Traffic Management Command, Military Sealift Command, Air Mobility Command, as appropriate) to determine whether secure shipment from the CONUS point of origin to the ultimate foreign destination is feasible. Normally, the U.S. will use the Defense Transportation System (DTS) to deliver classified material to the recipient government. A transportation plan will be developed by the shipper that prepares the Letter of Offer in coordination with the purchasing government. MACOM security officials, of the MSC or agency that prepares the Letter of Offer, will evaluate and approve the transportation plan, and may delegate this authority as they deem necessary. This does not, however, relieve the MACOM of the ultimate responsibility for oversight.

(2) Classified shipments resulting from direct commercial sales must comply with the same security standards that apply to FMS shipments. To develop and obtain approval of the required transportation plan, defense contractors will consult with the purchasing government and the Defense Security Service Regional Security Office prior to signing of a commercial contract that will result in the shipment of classified material.

(3) Delivery of classified material to a foreign government at a point within the U.S., its territories, or its possessions, will be accomplished at:

(a) An embassy, consulate, or other official agency under the control of the recipient government.

(b) The point of origin. When a designated representative of the recipient government accepts delivery of classified U.S. material at the point of origin (for example, a manufacturing facility or depot), the agency official who transfers custody will make sure the recipient is aware of secure means of onward movement of the classified material to its final destination, consistent with the approved transportation plan.

(c) A military or commercial port of embarkation (POE) that is a recognized point of departure from the U.S., its territories or possessions, for on-loading aboard a ship, aircraft, or other carrier. In these cases, the transportation plan will provide for U.S. controlled secure shipment to the CONUS transshipment point and the identification of a secure storage facility, either government or commercial, at or near the POE. An agency official authorized to transfer custody is to supervise or observe the on-loading of the FMS material being transported, when physical and security custody of the material has yet to be transferred formally to the foreign recipient. In the event that the transfer of physical security

and custody cannot be accomplished promptly, the agency official will make sure that the classified material is either returned to a secure storage facility of the U.S. shipper, segregated and placed under constant surveillance of a duly cleared U.S. security force at the POE, or held in the secure storage facility designated in the transportation plan.

(d) An appropriately cleared freight forwarder facility identified by the recipient government as its designated representative. In these cases, a person identified as a designated representative must be present to accept delivery of the classified material and receipt for it, to include full acceptance of security responsibility.

d. Delivery outside the United States, its territories, or possessions:

(1) Classified U.S. material to be delivered to a foreign government within the recipient country, will be delivered on arrival in the recipient country to a U.S. Government representative, who will arrange for its transfer to a designated representative of the recipient government. If the shipment is escorted by a U.S. Government official authorized to accomplish the transfer of custody, the material can be delivered directly to the recipient government's designated representative upon arrival.

(2) Classified material to be delivered to a foreign government representative within a third country will be delivered to an agency or installation of the U.S., or of the recipient government, that has extraterritorial status or otherwise is exempt from the jurisdiction of the third country. Unless the material is accompanied by a U.S. Government official authorized to accomplish the transfer of custody, a U.S. Government official will be designated locally to receive the shipment upon arrival and deliver it to the recipient government's designated representative.

*e*. Overseas shipments of U.S. classified material will be made only via ships, aircraft, or other carriers that are owned, or chartered by the U.S. Government or under U.S. registry, owned or chartered by, or under the registry of, the recipient government, or otherwise authorized by the head of the command or agency having classification jurisdiction over the material involved. Overseas shipment of classified material will be escorted, prepared for shipment, packaged, and stored, onboard as prescribed elsewhere in this regulation, DODD 5220.22–R, and DODD 5220.22–M.

*f*. Only freight forwarders that have been granted an appropriate security clearance by the DOD or the recipient government are eligible to receive, process related security documents, and store U.S. classified material authorized for release to foreign governments. However, a freight forwarder that does not have access to or custody of classified material, and is not required to perform security–related functions, will not be cleared.

g. Foreign governments can return classified material to a U.S. contractor for repair, modification, or maintenance. At the time the material is initially released to the foreign government the approved methods of return shipment will be specified in the Letter of Offer and Acceptance (LOA) for Foreign Military Sales, the security requirements section of a direct commercial sales contract, or in the original transportation plan. The contractor, upon notification of a return shipment, will give advance notice of arrival to the applicable user agency or the Defense Security Service, and arrange for secure inland shipment within the U.S. if such shipment has not been prearranged.

h. Transportation plan requirements are as follows:

(1) Preparation and coordination is as follows:

(a) Foreign Military Sales. U.S. classified material to be furnished to a foreign government or international organization under FMS transactions, will normally be shipped via the Defense Transportation System and delivered to the foreign government within its own territory. The U.S. Government can permit other arrangements for such shipments when it determines that the recipient foreign government has its own secure facilities and means of shipment from the point of receipt to ultimate destination. In any FMS case, the agency or command having security cognizance over the classified material involved is responsible, in coordination with the foreign recipient, for developing a transportation plan. When the point of origin is a U.S. contractor facility, the contractor and Defense Security Service will be provided a copy of the plan by the agency or command.

(b) Commercial transactions. The contractor will prepare a transportation plan for each commercial contract, subcontract, or other legally binding arrangement providing for the transfer of classified freight to foreign governments, to be moved by truck, rail, aircraft, or ship. The requirement for a transportation plan applies to U.S. and foreign contracts.

(c) The transportation plan will describe arrangements for secure shipment of the material from the point of origin to the ultimate destination. It must identify recognized points of embarkation from the U.S., its territories, or possessions for transfer to a specified ship, aircraft, or other authorized carrier. It must identify a government or commercial secure facility in the vicinity of the points of embarkation and debarkation that can be used for storage if transfer or onward movement cannot take place immediately. Except as described in paragraph 8–7d, a U.S. Government official authorized to transfer custody and control, must supervise the on–loading of classified material when the material has yet to be officially transferred. The plan must provide for security arrangements in the event custody cannot be transferred promptly.

(d) Upon transfer of title to the purchasing foreign government, classified material can be delivered to a freight forwarder that is designated, in writing, by the foreign government, as its representative for that shipment, and is cleared to the level of the classified material to be received. The freight forwarder will be provided a copy of the transportation plan and agree to comply.

(2) The transportation plan will, as a minimum, will include-

(a) A description of the material to be shipped and a brief narrative description indicating where and under what circumstances transfer of custody will occur.

(b) Identification, by name and title, of the designated representative (or alternate) of the recipient government or international organization who will receipt for and assume security responsibility for the classified material.

(c) Identification and specific location(s) of delivery point(s) and security arrangements while the material is located at the delivery points.

(d) Identification of commercial carriers, freight forwarders, and/or transportation agents who will be involved in the shipping process, the extent of their involvement, and their clearance.

(e) Identification of any storage or processing facilities and transfer points to be used, certification that such facilities are authorized by competent government authority to receive, store, or process the level of classified material to be shipped, and a description of security arrangements while the material is located at the facilities.

(f) Routes and, if applicable, security arrangements for overnight stops or delays enroute.

(g) Arrangements for dealing with port security and customs officials.

(h) The identification, by name or title, of couriers, escorts, or other responsible officials (e.g. Captain or Crew Chief) to be used, including social security number, government identification or passport number, security clearance, and details concerning their responsibilities.

(i) Description of the shipping methods to be used and the identification of the foreign or domestic carriers.

(j) Description of packaging requirements, seals and storage during shipment.

(k) A requirement for the recipient government or international organization to examine shipping documents upon receipt of the classified material in its own territory and notify DSS or the agency or command having security cognizance over the classified material if the material has been transferred enroute to any carrier not authorized by the transportation plan.

(1) Requirement for the recipient government or international organization to inform DSS or the agency or command having security cognizance over the classified material promptly and fully of any known or suspected compromise of classified material.

(m) Arrangements for return shipments if necessary for repair, modification or maintenance.

#### 8–8. Shipment of freight

Where applicable, commands will establish procedures for shipment of bulk classified material as freight, to include provisions for shipment in closed vehicles, when required, appropriate notice to the consignee concerning the shipment, procedures at transshipment activities, and action to be taken in case of non-delivery or unexpected delay in delivery. DA Form 1965 (Delivery and Pick Up Service) may be used as a manifest for delivery by courier or messenger of sealed containers. The Top Secret and Secret contents will have an attached receipt form to be completed by the recipient and returned to the originator.

#### Section III

#### Preparation of Material for Transmission

#### 8–9. Envelopes or containers

*a*. When classified information is transmitted, it will be enclosed in two opaque, sealed envelopes, wrappings, or containers, durable enough to properly protect the material from accidental exposure and to ease in detecting tampering. The following exceptions apply:

(1) If the classified material is an internal component of a packageable item of equipment, the outside shell or body can be considered as the inner enclosure provided it does not reveal classified information.

(2) If the classified material is an inaccessible internal component of a bulky item of equipment, the outside or body of the item can be considered to be a sufficient enclosure provided observation of it does not reveal classified information.

(3) If the classified material is an item or piece of equipment that is not easily packageable and the shell or body is classified, it will be concealed with an opaque covering that will hide all classified features.

(4) Specialized shipping containers, including closed cargo transporters, can be considered the outer wrapping or cover when used.

(5) When classified material is handcarried outside an activity, a locked briefcase can serve as the outer wrapper. In such cases, the addressing requirements for the outer wrapper, as stated in paragraph 8–10a, below, do not apply.

(6) NATO restricted material does not have to be double-wrapped when it is transmitted within the United States. The marking "NATO RESTRICTED" will not appear on the outermost wrapper.

b. Classified material will be prepared for shipment, packaged, and sealed in ways that minimize the risk of accidental exposure or undetected deliberate compromise. Documents will be packaged so that the classified text is not in direct contact with the inner envelope or container. For documents that do not have an unclassified cover or cover

transmittal letter or form, this can be accomplished by inserting an opaque sheet or cardboard sheet on top of the classified text in the inner envelope.

### 8–10. Addressing

*a.* The outer envelope or container for classified material will be addressed to an official government activity or to a DOD contractor with a facility clearance and appropriate storage capability. It will show the complete return address of the sender. The outer envelope will not be addressed to an individual. Office codes or phrases such as "Attention: Research Department" may be used.

b. The inner envelope or container will show the address of the receiving activity, the address of the sender, the highest classification of the contents, including, where appropriate, any special markings such as "RESTRICTED DATA" or "NATO," and any other special instructions. The inner envelope may have an "attention line" with a person's name.

c. The outer envelope or single container will not bear a classification marking or any other unusual marks that might invite special attention to the fact that the contents are classified.

d. Classified information intended only for U.S. elements of international staffs or other organizations, must be addressed specifically to those elements.

### 8-11. Mail channels with the Department of Energy

Other federal government agencies can require special certification or special procedures before forwarding classified information to another agency. Where that is the case, DA commands will comply with the requirements of those agencies. Specifically, the Department of Energy (DOE) requires that a "mail channel" be established prior to the transmission of certain classified information from a DOE facility to another activity. The mail channel, or material channel for transmission of material other than mail, will be certified by a designated DA certification official, will be made on DOE Form 5631.20 and will include the certified classified mailing address. The certification official will be one of the officials authorized to sign DOE Form 5631.20. See paragraph 6–17, of this regulation, for policy on personnel authorized to sign the DOE Form 5631.20. The DOE Form 5631.20 replaced DOE Form DP–277. It is recommended that the DOE facility that holds the material be contacted for the proper address and information to be completed on the form. Unless notified to the contrary by the DOE facility, the mail or material channel may not exceed one year, subject to renewal of the form.

### Section IV

## Escort or Handcarrying of Classified Material

#### 8–12. General provisions

*a.* Appropriately cleared personnel may be authorized to escort or handcarry classified material between locations when other means of transmission or transportation cannot be used. Handcarrying of classified material will be limited to situations of absolute necessity and will be carried out to make sure it does not pose an unacceptable risk to the information. Generally, two–way handcarrying, carrying the material both to and from the destination, is not authorized unless specific justification has been provided and both situations involving the handcarrying meet the requirements stated in this section. Handcarrying will be authorized only when:

(1) The information is not available at the destination and is required by operational necessity or a contractual requirement.

(2) The information cannot be sent by a secure facsimile transmission or by other secure means, for example, U.S. Postal Service express mail.

(3) The handcarry has been authorized by the appropriate official. For handcarrying within and between the United States, its territories, and Canada, the authorizing official will be determined by the commander, subject to MACOM approval. For all other areas, approval is at the MACOM level and can be further delegated in writing by the MACOM. Where delegated, the MACOM will exercise oversight, during inspections and/or assistance visits, by requiring copies of approvals, or by other means, to ensure the requirements of this section are met.

(4) The handcarry is accomplished aboard a U.S. carrier, or a foreign carrier if no U.S. carrier is available, and the information will remain in the custody and physical control of the U.S. escort at all times.

(5) Arrangements have been made for secure storage during overnight stops and similar periods. The material will not be kept in hotels, personal residences, vehicles, or any other unapproved storage location.

(6) A receipt for the material, for all classification levels, is obtained from an appropriate official at the destination and the receipt is returned to the appropriate official at the traveler's command.

*b*. Many of the principles contained in paragraph 8–14, of this regulation, apply to all situations involving the handcarrying of classified information and are not restricted to those situations involving classified material handcarried outside the United States. Commands will consider the principles stated in paragraph 8–13 in developing command

procedures concerning the handcarrying of classified material and incorporate those that are deemed applicable to the handcarrying of classified material within the United States.

### 8–13. Documentation

a. Responsible officials will provide a written statement to all individuals escorting or carrying classified material authorizing such transmission.

b. The DD Form 2501 (Courier Authorization Card) may be used to identify appropriately cleared DA personnel who have been approved to handcarry classified material in accordance with the following, except that in the case of travel aboard commercial aircraft, the provisions of paragraph 8–15 of this regulation apply:

(1) The individual has a recurrent need to handcarry classified information;

(2) The form is signed by an appropriate official in the individual's servicing security office;

(3) Stocks of the form are controlled to preclude unauthorized use.

(4) The form is issued for no more than two years at a time. The requirement for authorization to handcarry will be reevaluated and/or revalidated on at least an biennial basis, and a new form issued, if appropriate.

(5) The use of the DD Form 2501 for identification and/or verification of authorization to handcarry Sensitive Compartmented Information or Special Access Programs information, will be in accordance with policies and procedures, established by the official having security responsibility for such information or programs.

## 8–14. Security requirements for temporary duty travel outside the United States

*a.* As stated above, the handcarrying of classified information is not a routine method of transmission and will only be approved when fully justified. Handcarrying classified material outside the United States subjects the information to increased risk. When classified material is handcarried, for delivery to a foreign government representative, or when classified information is discussed with, or otherwise disclosed to, foreign national personnel, the requirements of AR 380–10 will be strictly followed.

*b*. The DOD requires that a request for travel outside the United States contain a written statement by the traveler that classified information will or will not, as applicable, be disclosed during the trip. If the foreign disclosure of classified information is involved, there will be an additional written statement that disclosure authorization has been obtained in accordance with DODD 5230.11. For DA commands, AR 380–10 applies. The statement also will specify whether authorization has been obtained to carry classified material, in compliance with the provisions of this regulation.

c. If the traveler has been authorized to carry classified material, a copy of the written authorization will accompany the justification for the temporary duty travel (TDY). This authorization, the courier orders, will be provided by the traveler's Special Security Officer (SSO) or Command Security Manager (CSM). They will be kept in a secure place and will not be presented unless circumstances dictate.

*d.* Because of Operations Security (OPSEC) concerns, Block 16 of DD Form 1610 (Request and Authorization for TDY Travel of DOD Personnel) will not contain statements that identify the traveler as carrying classified information.

*e*. Travelers who are authorized to carry classified material on international flights, or by surface conveyance if crossing international borders, must have courier orders. The DD Form 2501 is not a valid form of courier authorization for travel overseas. A memorandum on command letterhead is required and will, as a minimum, provide the information specified in subparagraph (11), below. Travelers will be informed of, and acknowledge, their security responsibilities. The latter requirement may be satisfied by a briefing or by requiring the traveler to read written instructions that, as a minimum, contain the information listed below.

(1) The traveler is liable and responsible for the material being escorted.

(2) Throughout the journey, the classified material will stay in the personal possession of the traveler, except when it is in authorized storage.

(3) The material will not be opened en route except in the circumstances described in subparagraph (9), below.

(4) The classified material is not to be discussed or disclosed in any public place.

(5) The classified material is not, under any circumstances, to be left unattended. During overnight stops, U.S. military facilities, embassies, or cleared contractor facilities will be used. Classified material will not be stored in vehicles, hotel rooms or safes, personal residences, or any other unauthorized storage facility or location.

(6) The traveler will not deviate from the authorized travel schedule, unless such deviation is beyond the traveler's control, such as cancellation of a flight. The traveler will immediately notify their command of any delays.

(7) In cases of emergency, the traveler will take appropriate measures to protect the classified material, and will notify their command as soon as possible.

(8) The traveler is responsible for ensuring that personal travel documentation, such as passport, courier authorization, and medical documents, etc., are complete, valid, and current.

(9) There is no assurance of immunity from search by the customs, police, and/or immigration officials of the various countries whose border the traveler may be crossing. Therefore, should such officials inquire into the contents of the consignment, the traveler will present the courier orders and ask to speak to the senior customs, police and/or immigration official. This action should normally suffice to pass the material through unopened. However, if the senior

customs, police, and/or immigration official demands to see the actual contents of the package, it should be opened only in his/her presence, and must be done in an area out of sight of the general public, if possible. If the traveler is permitted to pass, notification to his/her command will be done at the earliest possible time.

(a) Precautions must be taken to show officials only as much of the contents as will satisfy them that the package does not contain any other item. The traveler should ask the official to repack or assist in repackaging of the material immediately upon completion of the examination.

(b) The senior customs, police, and/or immigration official, should be requested to provide evidence of the opening and inspection of the package, by sealing and signing it when closed, and confirming on the shipping documents, if any, or courier certificate, that the package has been opened.

(c) If the package has been opened under such circumstances as those mentioned above, the traveler will inform, in writing, the addressee and the dispatching security officer of this fact.

(10) Prior to travel, classified material to be carried by a traveler will be inventoried and a copy of the inventory retained by the traveler's security office. A copy of the inventory will be placed inside the classified package.

(11) Travel orders (DD Form 1610) will identify the traveler by name, title, organization, and include the traveler's passport or identification number. The travel orders will describe the route to be taken by the traveler, the traveler's itinerary can be attached for this purpose; describe the package to be carried by size, weight, and configuration, but will not contain statements that identify the package as containing classified material; reflect a date of issue and expiration; and contain the name, title, and telephone number of an appropriate official, within the traveler's command, who may be contacted to verify the authorization to escort classified material. Courier orders will contain this same information, in addition to a complete description of the material that is to be carried and expiration of authorization to carry the material. Where possible, the courier authorization should show the phone number of the U.S. embassy or consulate, closest to the area that the traveler will enter the country, in case the assistance of the U.S. State Department is needed in clearing customs. As an alternative, the courier orders should show the name and phone number of a point of contact at the activity located in the foreign country that is to be visited. Courier orders will be signed by the official authorizing the handcarrying of classified material.

(12) Upon return, the traveler will return all classified material, in a sealed package, or produce a receipt, signed by the security officer of the addressee organization, for any material that is not returned.

f. For guidance on handcarrying NATO information, travelers who are authorized to carry NATO classified material on international flights will refer to AR 380–15.

### 8–15. Handcarrying or escorting classified material aboard commercial passenger aircraft

*a*. Airport and aircraft operating and security procedures make handcarrying and escorting classified material on commercial passenger aircraft a complex task. Advance coordination with appropriate authorities is essential. See figure 8–1 for listings of the Federal Aviation Administration (FAA) Air Transportation Security Field Offices. During this coordination, specific advice should be sought regarding the nature of documentation that is required. Generally, the following has been found to meet requirements:

(1) The individual designated as courier will be in possession of a DOD military ID card (DD Form 2, (Armed Forces of the United States Geneva Convention Identification Card )) or civilian ID card, or contractor-issued identification card, that includes a photograph, descriptive data, and signature of the individual. If the identification card does not contain date of birth, height, weight, and signature, these items must be included in the written authorization.

(2) The courier will have the original of the authorization letter. A reproduced copy is not acceptable. The traveler will have sufficient authenticated copies to provide a copy to each airline involved. The letter will be prepared on letterhead stationary of the agency authorizing the carrying of classified material and will—

(a) Give the full name of the individual and his or her employing agency or company.

(b) Describe the type of identification the individual will present, for example, Naval Research Laboratory Identification Card N0. 1234; ABC Corporation Identification Card No. 1234.

(c) Describe the material being carried, for example three sealed packages,  $9' \times 8' \times 24'$ , addressee and addresser.

(d) Identify the points of departure, destination, and known transfer(s).

(e) Carry a date of issue and expiration.

(f) Carry the name, title and signature, of the official issuing the letter. Each package or carton to be exempt will be signed on its face by the official who signed the letter.

(g) Carry the name of the person designated to confirm the letter of authorization, and that person's official U.S. Government telephone number.

*b*. The traveler should process through the airline ticketing and boarding procedure the same as other passengers. The package or the carry–on luggage containing it should be routinely offered for inspection for weapons.

## 8-16. Consignor/consignee responsibility for shipment of bulky material

The consignor of a bulk shipment will-

*a*. Select a carrier that will provide a single line service from the point of origin to destination, when such a service is available.

b. Ship packages weighing less that 200 pounds in closed vehicles only.

c. Notify the consignees and military transshipping activities of the nature of the shipment, including level of classification, the means of shipment, the serial number of the seals, if used, and the anticipated time and date of arrival by separate communication, at least 24 hours in advance of arrival of the shipment.

*d*. Advise the first military transshipping activity that, in the event the material does not move on the conveyance originally anticipated, the transshipping activity should advise the consignee with information of the firm date and estimated time of arrival. Upon receipt of the advance notice of a shipment of classified material, consignees and transshipping activities will take appropriate steps to receive the classified shipment and to protect it upon arrival.

*e*. Annotate the bills of lading to require the carrier to notify the consignor immediately, by the fastest means, if the shipment is unduly delayed in route. Such annotations will not under any circumstances disclose the classified nature of the commodity. When seals are used, annotate substantially as follows: "DO NOT BREAK SEALS EXCEPT IN EMERGENCY OR UPON AUTHORITY OF CONSIGNOR OR CONSIGNEE. IF BROKEN, APPLY CARRIER'S SEALS AS SOON AS POSIBLE AND IMMEDIATELY NOTIFY CONSIGNOR AND CONSIGNEE."

*f*. Require the consignee to advise the consignor of any shipment not received more than 48 hours after the estimated time of arrival furnished by the consignor or the transshipping activity. Upon receipt of such notice, the consignor will immediately trace the shipment. If there is evidence that the classified material was subjected to compromise, the procedures set forth in chapter 10 of this regulation for reporting compromises will apply.

City State		City State	 ?	
Anchorage	Alaska	New York (La Guardia) New York		
Atlanta	Georgia	Philadelphia	Pennsylvania	a
Baltimore	Maryland	Pittsburgh	Pennsylvania	a
Boston	Massachusetts	Portland	Oregon	
Chicago (O'H	lare) Illinois	Saint Louis	Missouri	
Cleveland	Ohio	San Antonio	Texas	
Dallas	Texas	San Diego	California	
Denver	Colorado	San Franciso	co Califo	rnia
Detroit	Michigan	San Juan	Puerto Rico	
Honolulu	Hawaii	Seattle	Washington	
Houston	Texas	Tampa	Florida	
Kansas City	Missouri	Tucson	Arizona	
Las Vegas	Nevada	Washington	(Dulles)	Washington, D.C.
Los Angeles	California	Washington	(National)	Washington, D.C.
Miami Florida	a			
Minneapolis	Minnesota			
Newark	New Jersey			
New Orleans	Louisiana			
New York (Jo	ohn F. Kennedy) Ne	w York		

Federal Aviation Administration (FAA) Air Transportation Security Field Offices

Figure 8-1. Federal Aviation Administration (FAA) Air Transportation Security Field Offices

# Chapter 9 Security Education

Section I Policy

## 9–1. General policy

Commanders will establish security education programs. These programs will be aimed at promoting quality performance of security responsibilities by command personnel, and will be tailored, as much as possible, to the specific involvement of individuals in the information security program and the command's mission. The programs will *a.* Provide necessary knowledge and information to enable quality performance of security functions.

*b.* Promote understanding of information security program policies and requirements, and their importance to the

b. Promote understanding of information security program policies and requirements, and their importance to the national security.

c. Instill and maintain continuing awareness of security requirements and the intelligence collection threat.

d. Assist in promoting a high degree of motivation to support program goals.

## 9-2. Methodology

Security education must be a continuous, rather than periodic, influence on individual security performance. Periodic briefings, training sessions, and other formal presentations will be supplemented with other informational and promotional efforts to ensure maintenance of continuous awareness and performance quality. The use of job performance aids and other substitutes for formal training, for example, video tapes or self–paced computer programs, can be used when they are determined to be the most effective means of achieving program goals. The circulation of directives or similar material on a "read–and–initial" basis will not be considered as fulfilling any of the specific requirements of this Chapter, because there is no basis to gauge effectiveness.

### Section II Briefings

## 9-3. Initial orientation

All DA personnel, especially those who could be expected to play a role in the information security program, will be given an initial orientation. The purpose of the orientation will be:

- a. To ensure personnel are aware of the roles they are expected to play in the information security program;
- b. The importance of their fulfilling their responsibilities; and
- c. That they have enough information to fulfill those responsibilities.

## 9–4. Cleared personnel

*a*. All personnel, especially those granted access or are expected to be granted access to classified information, will be provided an initial orientation to the information security program before being allowed access to any classified or sensitive information. The initial orientation is intended to:

(1) Produce a basic understanding of the nature of classified information and the importance of its protection to the national security;

(2) Place employees on notice of their responsibility to play a role in the security program; and

(3) Provide them enough information to ensure proper protection of classified/sensitive information in their possession. Security educators will, as a minimum, include the following points in their Security Education Programs:

(a) The nature of U.S. and foreign government classified and sensitive information, its importance to the national security, and the degree of damage associated with each level of classification/sensitivity.

(b) How to recognize U.S. and foreign government classified and sensitive information that personnel may encounter, including markings, etc.

(c) The individual's responsibility for protection of classified and sensitive information, and the consequences of failing to do so.

(d) Procedures and criteria for authorizing access to classified and sensitive information.

(e) Procedures for safeguarding and control of classified and sensitive information in the individual's work environment.

(f) Proper reaction to discovery of information believed to be classified/sensitive in the public media.

(g) The security management and support structure within the command, to include sources of help with security problems and questions and proper procedures for challenging classifications believed to be improper.

(h) Penalties associated with careless handling or compromise of classified/sensitive information. See chapter 10 for extracts from the espionage laws and federal statutes for details.

b. Before being granted access to classified information, employees must sign SF 312. See paragraph 6-2 of this regulation, for details regarding the use of the SF 312.

#### 9-5. Briefing upon refusal to sign the NDA, SF 312

Chapter 6 of this regulation contains the policy on the execution of the Classified Information Nondisclosure Agreement (NDA)(SF 312). Individuals who refuse to sign the form will be advised of the following:

a. Execution of the NDA is mandatory for all military and civilian personnel as a condition of access to classified information.

*b*. The purpose of the NDA is to make sure each individual authorized access to classified information is aware of the duties and responsibilities imposed by law and to ensure that the individual understands the personal commitment to protect classified information.

c. The NDA does not impose any obligation beyond those prescribed by law to protect classified information.

*d*. In the interests of national security, classified information will be disclosed only to those individuals who are committed personally to protecting such information, have a valid need-to-know, and who have been granted access to classified information. Any individual who shows a reluctance to sign an NDA will be considered to have a lack of personal commitment to protect classified information.

*e*. An individual who refuses to sign the NDA will not be granted access to classified information. That individual will be granted a period of time, not to exceed 5 calendar days, to re-evaluate their decision. If the individual refuses to sign the NDA at the end of that 5-day period, that individual will be advised that the foregoing action will require termination of their present and future access to classified information. In addition, the individual will be advised that the reluctance to sign the NDA may place their security clearance in jeopardy.

f. There is no prohibition against the individual consulting with an attorney in this matter. However, the individual will be advised that such consultation will be at no additional expense to the government nor will it extend the 5-day re-evaluation time limit.

#### 9–6. Briefing uncleared personnel

Personnel who are not cleared for access to classified information will be included in the security education program. Especially if they will be working in situations where inadvertent access to classified information might occur or they will have access to unclassified/sensitive information which might be of value to intelligence collectors. They will be provided with a brief explanation of the nature and importance of classified and sensitive information and actions they should take if they discover classified information unsecured, note an apparent security vulnerability, or believe they have been contacted by an intelligence collector or other unauthorized individual seeking to gain access to sensitive government information. Security training for all DA personnel is the command's responsibility. Security education training is useful in the understanding of why official information must be protected and, therefore, inclusion of uncleared personnel in certain aspects of the security education program is essential.

#### 9–7. Refresher briefing

Security education programs will include efforts to maintain and reinforce quality performance of security responsibilities. As a minimum, all DA employees, especially those who have access to, create, process, or handle classified/ sensitive information, will be provided refresher training in their responsibilities at least once a year. The actual frequency and nature of continuing security education must be determined by the needs of, and outlined in, the command's information security program and the nature of the command personnel involvement in the program. As a minimum, all personnel will receive annual refresher training that reinforces the policies, principles, and procedures, covered in initial and specialized training. Refresher training will also address the threat and the techniques employed by foreign intelligence activities attempting to obtain classified information, and advise personnel of penalties for engaging in espionage activities. Refresher training should also address issues or concerns identified during unit self-inspections. Whenever security policies and procedures change, personnel, whose duties would be impacted by these changes, must be briefed as soon as possible.

#### 9-8. Foreign travel briefing

*a*. Although foreign travel (personal or official) may be briefly discussed during annual refresher briefings, it is also a requirement to attend a separate foreign travel briefing for all active and reserve soldiers pending travel outside the U.S. and its territories. This is especially true for those that travel frequently. It is in the best interest of the traveler, as well as the command, to ensure the traveler is fully prepared for any particular security or safety concerns that their travel to foreign areas may introduce.

b. A foreign travel briefing used to be only offered to those individuals who had access to classified information. AR 525–13 requires that all DA military and civilian personnel pending travel outside the U.S. and its territories or possessions must attend the Antiterrorism/Force Protection (AT/FP) Level I awareness training, prior to departure from their current duty station. Training must be received within 6 months of departure date to the overseas area. It is the

commander's responsibility to ensure all DA military and civilain personnel are scheduled for and receive this training prior to their departure. All DA military and civilian personnel will not outprocess or depart on PCS, TDY, TCS, leave, or pass to an overseas area without AT/FP training.

c. Upon request, an unclassified version may be given to dependents or others who do not have access. For updated information, regarding foreign travel, contact your command security manager or the U.S. State Department. Examples of briefings and country specific information are available on the State Department's unclassified Internet website at http://www.state.gov/. Individuals with SCI access should be referred to their SSO for foreign travel briefing requirements.

*d*. Upon return, the traveler should be provided the opportunity to report any incident to their command security manager, no matter how insignificant, that could have security implications.

## Section III Special Requirements

## 9-9. General policy

DA personnel in positions which require performance of specified roles in the information security program will be provided security education sufficient to permit quality performance of those duties. The training will be provided before, concurrent with, or not later than six months following assumption of those positions.

## 9–10. Original classifiers

*a*. Officials who have been granted original classification authority will be educated in their responsibilities before they exercise the delegated authority. They will be provided an understanding of:

(1) The difference between original and derivative classification and the circumstances in which each is appropriate.

(2) Requirements, standards and criteria for original classification of information.

- (3) Prohibitions and limitations on classification.
- (4) The process and standards for determining duration of classification.
- (5) Requirements for creation and maintenance of classification guides.

(6) Agency procedures for challenging classification, and the responsibilities of the original classification authority in responding to challenges.

(7) Declassification decision-making regarding information classified under previous Executive Orders, EO 12958 and amendments, DODD 5200.1-R, and previous editions of this regulation, including mandatory and systematic declassification review.

(8) Aspects of marking, controlling and the safeguarding of classified information (including requirements for automated systems) which could affect classification and declassification decision-making.

(9) Requirements and procedures for safeguarding foreign government information.

*b*. Security educators may consider the use of job aids or similar techniques, for example, video tapes or self-paced computer training programs, to replace or supplement traditional educational techniques, due to the relatively low frequency of training for original classification authorities. The Defense Security Service Academy may be contacted for advice in obtaining training aids. They can be found on the Internet at http://www.dss.mil/training.

c. DOD Handbook 5200.1–PH can be used as a reference by original classification authorities and personnel that assist these officials. Many of the basic principles of making an original classification decision have not changed since the publication of the handbook, and it may be useful both as a training aid for newly designated original classification authorities, as a reference in making original classification decisions, and developing command level security classification guides. Excerpts from this handbook appear as appendix G of this regulation.

## 9-11. Derivative classifiers

DA personnel whose responsibilities include derivative classification, will be trained in requirements and procedures appropriate to the information and material they will be classifying, including the proper use of classification guides and source documents. As a minimum, the training will address the following questions:

a. What are the original and derivative classification processes and the standards applicable to each?

b. What are the proper and complete classification markings to be applied to classified information?

c. What are the authorities, methods and processes for downgrading and declassifying information?

d. What are the methods for the proper use, storage, reproduction, transmission, dissemination and destruction of classified information?

e. What are the requirements for creating and updating classification and declassification guides?

f. What are the requirements for controlling access to classified information?

g. What are the procedures for investigating and reporting instances of actual or potential compromise of classified information and the penalties that may be associated with violation of established security policies and procedures?

*h*. What are the requirements for creating, maintaining, and terminating special access programs, and the mechanisms for monitoring such programs?

*i*. What are the procedures for the secure use, certification and accreditation of automated information systems and networks which use, process, store, reproduce, or transmit classified information?

j. What are the requirements for oversight of the security classification program, including self-inspections?

#### 9–12. Security Program Management personnel

Security managers, security staff members, and others with significant responsibility for management of the information security program, will be trained/educated to fulfill their roles. The Defense Security Service Academy, formerly the Defense Security Institute (DSI), Baltimore Maryland, should be contacted for advice in obtaining training and/or training aids. They can be found on the Internet at http://www.dss.mil/training/. The training and education should be tailored to suit their expected contributions to the program, and will include at a minimum:

*a.* Procedures, standards, and processes for original and derivative classification, and for downgrading and declassifying information.

b. Proper marking of classified/sensitive documents and material.

c. Requirements and techniques for controlling access to classified/sensitive information, and for the proper use, storage, reproduction, transmission, transportation, dissemination and destruction of classified and sensitive material.

d. Procedures and requirements for responding appropriately to security violations and other security incidents.

*e*. Requirements and procedures for securing classified/sensitive information processed, maintained, or stored on automated information systems. It should be noted that command–wide responsibilities for the Information Systems Security (ISS) Program can be delegated to another official as specified in AR 380–19. The security manager and select security staff personnel will be educated in at least the basics of ISS to promote a seamless, integrated security program.

f. Requirements and methods for security education, program oversight, and program management.

### 9-13. Critical Nuclear Weapons Design Information Briefing

As stated in Chapter 6, DA personnel will be briefed on the sensitivity of Critical Nuclear Weapons Design Information (CNWDI) before access is granted. The following is an example of a sample briefing for CNWDI access, and it, or a version thereof, is suggested for use.

*a.* CNWDI is that TOP SECRET or SECRET Restricted Data that reveals the theory of operation or design of the components of a thermonuclear or implosion type fission bomb, warhead, demolition munition, or test device. Access to and dissemination of CNWDI is of particular concern to the Army. Because of its extreme sensitivity, access must be limited to the minimum number of persons.

*b.* You have been nominated for access to CNWDI, certified by the appropriate official as having the required "need–to–know" for that category of information, and determined to have an appropriate clearance for access to be granted.

c. Access to CNWDI entails additional responsibilities as well as special procedures. These are contained in AR 380–5, and include special access, dissemination, and marking requirements. It is particularly emphasized that CNWDI can only be disseminated to those individuals who have been fully authorized for access and have a verified need–to–know. If in doubt, check with (fill in appropriate name/office/telephone number).

d. Any questions regarding procedures governing access to, dissemination of, or safeguarding CNWDI are to be referred to (fill in appropriate name/office/telephone number).

### 9-14. Others

Commands will include in their security education programs, either in the general program or as part of special briefings to select personnel affected, provisions regarding special education and training for personnel who:

a. Use automated information systems to store, process, or transmit classified/sensitive information (see appendix E of this regulation and AR 380-19).

b. Will be traveling to foreign countries where special concerns about possible exploitation exist or will be attending professional meetings or conferences where foreign attendance is likely. The military intelligence unit providing counterintelligence support to the command, should be contacted for assistance and/or information in this regard.

c. Will be escorting, handcarrying, or serving as a courier for classified/sensitive material.

d. Are authorized access to classified and/or sensitive information requiring special control or safeguarding measures.

e. Are involved with international programs.

*f.* Regardless of clearance and/or access level held, all DA personnel will receive Subversion and Espionage Directed Against the U.S. Army (SAEDA) training, at a minimum of every two years, pursuant to Interim Change I01, AR 381–12.

## Section IV Termination Briefings

## 9-15. General Policy

DA commands will establish procedures to make sure that cleared employees, who leave the command or whose clearance is terminated, receive a termination briefing. See paragraph 6–5 of this regulation for more detailed policy on termination briefings. This briefing will—

a. Emphasize the individual's continued responsibility for protection of classified/sensitive information to which they have had access;

b. Provide instructions for reporting any unauthorized attempt to gain access to such information;

c. Advise the individual of the prohibition against retaining classified/sensitive material when leaving the organization; and

*d*. Remind them of the potential civil and criminal penalties for failure to fulfill their continuing responsibilities (see chap 10).

## Section V Program Oversight

## 9-16. General policy

DA commanders will ensure that their security education programs are appropriately evaluated during self-inspections and during oversight activities of subordinate commands or organizational units. This evaluation will include assessment of the quality and effectiveness of security education efforts, as well as ensuring appropriate coverage of the target populations. Commands will maintain a record of the programs offered and of the personnel that participated. These records will be maintained for two years and will be available for review during oversight inspections and assistance visits. These evaluations will also be included in block 9 of the SF 311 annual report.

# Chapter 10 Unauthorized Disclosure and Other Security Incidents

Section I Policy

### 10–1. General policy

*a.* The compromise of classified information can cause damage to our national security. Loss of classified material is just as serious. If classified material is lost, it cannot be determined if the information has been compromised. When loss or compromise of classified information happens, immediate action is required to minimize any damage and eliminate any conditions that might cause further compromises. To do this, prompt and effective investigation of the situation and prompt reporting of results are critical. Each incident in which classified information or material may have been lost or compromised must be the subject of a preliminary inquiry as described in this Chapter (See figure 10–1 for a sample preliminary inquiry). The purposes of this preliminary inquiry will be to:

(1) Determine whether classified information was compromised and, if so, whether there is damage to the national security.

(2) Determine what persons, situations, and/or conditions were responsible for or contributed to the incident.

*b*. The provisions of other Department of the Army regulations that require the investigation and reporting of counterintelligence (CI), criminal, or other serious incidents can also apply to incidents discussed in this Chapter. For example, AR 190–40, AR 195–2, AR 381–10, AR 381–12, AR 381–20, and any other Army regulations that implement DODI 5240.4. All applicable Army regulations will be followed in reporting and investigating these matters.

*c*. Incidents involving cryptographic information will be handled in accordance with National Communications Security Instruction (NACSI) 4006 (See AR 380–40). Details involving the use of DA Form 2134 (Security Violation Report(s)) for reporting COMSEC security violations, discussed in previous editions of this regulation, are covered in AR 380–40. Incidents involving SCI will be handled in accordance with DOD Manual 5105.21–M–1 (see AR 380–28). Incidents involving SAPs information will be handled in accordance with AR 380–381 (See appendix I for more information).

*d*. The Office of the Under Secretary of Defense (Policy), (USD(P)), will be notified of all losses or compromises of foreign government information. These reports will be made through command channels and DAMI–CH.

#### 10-2. Reaction to discovery of incident

*a*. Anyone finding classified material out of proper control, will take custody of and safeguard the material, if possible, and immediately notify the appropriate security authorities. In all cases, the individual's immediate supervisor is to be notified.

b. Any person who becomes aware of the possible loss or compromise of classified information will immediately report it to the commander, the command security manager, or other official the commander may direct, for instance, a department head or office chief if indicated by the command security manual or other command directive. If the person believes that the commander, security manager, or other official designated to receive such reports may have been involved in the incident, the person making the discovery will report it to the security authorities at the next higher level of command or supervision.

c. If classified information appears in the public media, personnel will not make any statement or comment that would confirm the accuracy or verify the classified status of the information. See paragraph 4–2, of this regulation, for further guidance.

(1) It is essential that Army personnel are careful to neither confirm nor deny the existence of classified information or the accuracy of that information in the public media. Personnel will report such matters to the command security manager, or other official designated by the commander. The matter will then be reported to the original classification authority for the information.

(2) The news article or other medium will not be marked as classified, however, the written report detailing the discovery of the information in the public media will be classified to the level of the information believed to have been compromised. Personnel will not discuss the matter with anyone without the expressed consent of the command security manager, or an individual so designated by the command security manager or commander. An appropriate security clearance and need-to-know is required. No discussions will be made over non-secure circuits.

(3) If approached by a representative of the media who wishes to discuss information, personnel will neither confirm nor deny the accuracy of or the classification of the information, and will report the approach immediately to the appropriate command security and public affairs authorities.

*d*. Any incident in which the deliberate compromise of classified information or involvement of foreign intelligence agencies is suspected, will be reported to the command security manager and the supporting counterintelligence organization.

#### 10–3. The preliminary inquiry

When an incident of possible loss or compromise of classified information is reported, the command will immediately initiate a preliminary inquiry into the incident. If the information was in the custody of another activity at the time of the possible compromise, that activity will be notified and will assume responsibility for the preliminary inquiry. This preliminary inquiry will be conducted according to these guidelines:

*a.* The person appointed to conduct the preliminary inquiry will have the appropriate security clearance, the ability and available resources to conduct an effective preliminary inquiry, and will not be likely to have been involved, directly or indirectly, in the incident. Except in unusual circumstances, the command security manager will not be appointed to conduct the preliminary inquiry. It is typically the responsibility of the security manager, unless command policy states otherwise, to make sure that an official is appointed to conduct the preliminary inquiry is completed in accordance with this regulation and command policy and procedure. Advice and assistance may be requested from the supporting counterintelligence organization.

b. In cases of apparent loss of classified material, the person conducting the preliminary inquiry will ensure that a thorough search for the material has been conducted. Document the steps taken to locate the material.

c. The preliminary inquiry will focus on answering the following questions:

(1) When, where, and how did the incident occur? Exactly what happened?

(2) What specific classified information and/or material (to include foreign government information) was involved? What was the level of classification of the information?

(3) What persons, situations, or conditions caused or contributed to the incident?

*d*. Every preliminary inquiry into possible loss or compromise of classified information or material will include a judgment about whether any compromise did occur and what, if any, potential damage to national security has occurred. One of the following alternatives will be chosen:

(1) Compromise of classified information did not occur.

(2) Compromise of classified information may have occurred.

(3) Compromise of classified information did occur, but there is no reasonable possibility of damage to the national security. Note, the determinations about damage here, and in the following paragraph, are not "damage assessments" as discussed in paragraph 10–5. The determination to be made here is whether the circumstances of the incident are such, that the possibility of damage to the national security can be discounted.

(4) Compromise of classified information did occur and damage to the national security may result.

*e*. In cases of apparent unauthorized disclosure of classified information to the public media, the preliminary inquiry will include the following additional questions in order to determine if a leak investigation is warranted:
(1) What is the date and identity of the article disclosing classified information?

(2) What specific statements in the article are considered classified and whether the data was properly classified?(3) If the data came from a specific document, what is the source document's origin, and the identity of the individual responsible for the security of the classified information disclosed?

- (4) What is the extent of dissemination of the data?
- (5) Has the information been previously officially released?
- (6) Was prepublication clearance sought from proper authorities?

(7) Have portions of, or background data on, the material, been published officially or in the open press from which an educated speculation on the consolidated data is derived?

(8) Can the data be declassified or otherwise made available for prosecution and, if so, what is the identity of the person competent to testify concerning its classification?

(9) Had declassification been decided upon before the data was published?

- (10) What is the effect the disclosure of the classified data would have on the national security?
- (11) Is the disclosed classified data accurate?

*f*. If at any time, during the preliminary inquiry, it appears that deliberate compromise of classified information may have occurred, the situation will be immediately reported to the chain of command and supporting counterintelligence unit. Apparent violations of other criminal law will be reported to the supporting criminal investigative activity. Coordination with the command's legal counsel is recommended whenever it seems likely that administrative or other sanctions may be taken against someone because of the incident.

#### 10–4. Reporting results of the preliminary inquiry

*a.* If the conclusion of the preliminary inquiry is as stated in paragraph 10–3d(2) or (4), (compromise could have occurred, or compromise did occur and damage to the national security can result) the official initiating the preliminary inquiry will immediately notify the originator of the information or material involved. If the originator was not the original classification authority, the OCA will also be immediately notified (see paragraph 10–5a, below). If the originator cannot be determined, the command's MACOM will be contacted for guidance. The MACOM will contact DAMI–CH, for those cases in which the MACOM cannot direct the command to the appropriate activity. Notification of the originator and original classification authority will not be delayed pending completion of any additional inquiry or resolution of other related issues.

*b.* If the conclusion of the preliminary inquiry is as stated in paragraph 10–3d(2) or (4), the command will report the matter through command channels to its MACOM, or to the Administrative Assistant to the Secretary of the Army (AASA) for offices and activities under HQDA. The MACOM or the AASA will review the report for completeness and adequacy of investigation and for the appropriateness of the corrective action/sanctions taken. Such reports will be filed and retained for a period no less than two years and are subject to HQDA or other appropriate agency oversight. MACOMs and the AASA will establish policy and procedures concerning whether or not there will be a forwarding of the reports of preliminary inquiry when the conclusion is other than stated in paragraph 10–3d(2) or (4). Reports of preliminary inquiry will be included in the Command management control review and oversight. If analysis shows that defects in the procedures and requirements of this regulation, or another Army regulation or DOD directive, contributed to the incident, MACOM, and the AASA officials will so advise DAMI–CH. DAMI–CH officials will evaluate the incident and report the conclusions, where deemed warranted, to DOD officials, if the problem concerns a DOD requirement. Report defects in the procedures and requirements regarding Army or other DOD SAPs directives, regulations, instructions, or other regulatory guidance through command channels to DAMI–CH (SAP) and DACS–DMP. If the problem concerns a DOD SAPs Directive, Instruction, or other regulatory guidance, HQDA will report to the Director, Special Programs, ODUSD(P).

c. If the conclusion of the preliminary inquiry is as stated in paragraph 10-3d(2) or (4), and foreign government information is involved, the incident will be reported through command channels and DAMI–CH to the Director of International Security Programs, ODUSD(P), who will notify the foreign government.

d. If the preliminary inquiry concludes that violations of the provisions of this regulation or criminal statutes did occur, see Chapter I, section VI, for other reporting requirements that may apply.

*e*. Commands will forward, through command channels to DAMI–CH, a copy or summary of the preliminary inquiry or investigation conducted, as a result of the unauthorized disclosure of classified information to the public media. An example of a preliminary report format can be found at figure 10–1, of this Chapter. DAMI–CH will forward such preliminary inquiry reports to the Director, Counterintelligence and Security Programs, OASD(C3I). SAPs leak inquiries or investigations will be provided directly to DAMI–CH (SAP) and DACS–DMP, for forwarding to the Director, Special Programs, ODUSD(P)SP (see appendix I of this regulation and AR 380–381 for more details).

#### 10-5. Reevaluation and damage assessment

*a*. When notified of possible or actual compromise, the holder of the information or material will ensure that the original classification authority, responsible for each item of the information, is notified of the incident. The OCA will verify and reevaluate the classification of the information and will conduct a damage assessment.

b. When classified information under the control of more than one command or agency is involved, the affected activities are responsible for coordinating their efforts in damage assessment and reevaluation. When participation by foreign governments or international organizations in damage assessment and reevaluation is required, contacts will be made through established intergovernmental liaison channels.

c. The first step in the reevaluation and damage assessment process is for the OCA to verify the actual, current classification of the information involved. The OCA determines whether the information currently is classified and the level and duration of classification that applies.

*d*. The second step is to reevaluate the classification of the information to see whether the classification should be continued or changed. This review will consider the following possibilities:

(1) The information has lost all or some of its sensitivity since it was classified, and will be downgraded or declassified. In rare cases, it might also be discovered that the information has gained in sensitivity and must be upgraded.

(2) The information has been so compromised by this incident that attempting to protect it further is unrealistic, or inadvisable, and it is be declassified.

(3) The information must continue to be classified at the same level.

*e*. The third step is to determine whether there are countermeasures that can be taken to minimize or eliminate the damage to the national security that could result from the compromise. These countermeasures might include changing plans or system design features, revising operating procedures, providing increased protection to related information, through classification or upgrading, etc. The OCA performing this function is responsible for initiating or recommending the appropriate countermeasures.

*f*. The final step is performing the damage assessment. The OCA will determine, given the nature of the information and the countermeasures, if any, that will be employed, what the probable impact of the compromise will be on our national security. In contrast to the first three steps in this process, which must be completed quickly, this step is sometimes a long-term, multi-disciplinary analysis of the adverse effects of the compromise on systems, plans, operations, and/or intelligence.

#### 10-6. Debriefings in cases of unauthorized access

In cases where a person has had unauthorized access to classified information, it is advisable to discuss the situation with the individual to enhance the probability that they will properly protect it. Whether such a discussion, commonly called a "debriefing," is held, is to be decided by the commander, security manager, or other designated official. This decision must be based on the circumstances of the incident, what is known about the person or persons involved, and the nature of the classified information. The following general guidelines apply:

*a*. If the unauthorized access was by a person with the appropriate security clearance but no need-to-know, debriefing is usually unnecessary. Debriefing is required if the individual is not aware the information is classified and that it needs protection. Inform the person that the information is classified and it requires protection. In these cases, the signing of a debriefing statement (see subparagraph e, below) is usually not necessary.

b. If the unauthorized access was by U.S. government personnel, civilian or military, without the appropriate security clearance, debriefing will be accomplished. Personnel will be advised of their responsibility to prevent further dissemination of the information and of the administrative sanctions and criminal penalties which might follow if they fail to do so. The debriefing official will make sure the individual understands what classified information is and why its protection is important.

c. If the person who had unauthorized access is an employee of a cleared contractor participating in the national industrial security program, the same guidelines apply as for U.S. Government personnel. Coordination with the employing firm's facility security officer/manager is recommended unless such coordination would place the information at increased risk.

*d.* If the person involved is neither U.S. government personnel, nor an employee of a cleared contractor, the decision will be made by the commander. The key question to be decided is whether the debriefing will have any likely positive effect on the person's ability and/or willingness to protect the information. As a general rule, it is often more effective in the long run to explain that a mistake occurred and that the person had unauthorized access to certain sensitive U.S. government information. Also, that such access should not have happened and that the U.S. Army needs the individual to understand that the information must be protected and never further discussed or otherwise revealed to other unauthorized personnel.

*e*. It is useful to have the person being debriefed sign a statement acknowledging the debriefing and their understanding of its contents. This may have a significant psychological effect in emphasizing the seriousness of the situation. If the person refuses to sign a debriefing statement, when asked, this fact, and their stated reasons for refusing, will be made a matter of record in the preliminary inquiry. The nearest counterintelligence unit will immediately be notified so that a trained CI investigator can explain the reason for the debriefing and advise the individual that a refusal to sign could indicate an unwillingness to protect classified information and could place their clearance, if held at the time, in jeopardy. f. In any case where the person to be debriefed may be the subject of criminal prosecution or disciplinary action, command officials are advised to consult with legal counsel before attempting to debrief the individual.

### 10–7. Management and oversight

*a.* Department of the Army commands, and especially MACOMs, will establish necessary reporting and oversight mechanisms, to ensure that inquiries are conducted when required, that they are done in a timely and effective manner, and that appropriate management action is taken to correct identified problem areas. Inquiries and management analyses of security incidents will consider possible systemic shortcomings which could have caused or contributed to the incident. The effectiveness of command security procedures, security education, supervisory oversight of security practices, command management emphasis on security, etc., will be considered when determining causes and contributing factors. The focus of management's response to security incidents will be to eliminate, or minimize, the possibility of further incidents occurring. Appropriate disciplinary action or legal prosecution, discussed in Chapter 1, section VII, of this regulation, is sometimes one means of doing this, but the broader focus on prevention must not be lost. Disciplinary action will not be the sole command reaction to a security incident, unless there has been a consideration of what other factors may have contributed to the situation.

*b*. Commands, and especially MACOMs and MSCs, will establish a system of controls and procedures to make sure that reports of security inquiries and damage assessments are conducted, when required, and that their results are available as needed. Such reports will be available for review during inspections and oversight reviews. MACOMs, and the AASA for HQDA activities, can establish reporting requirements for such inquiries and assessments. Reports of the results of security inquiries will be reported to DAMI–CH, only for those situations stated in paragraph 10–4.

## 10-8. Additional investigation

Additional investigation, beyond what is required by this Chapter, such as an AR 15–6 investigation, may be needed to permit application of appropriate sanctions for violation of regulations, criminal prosecution, or determination of effective remedies for discovered vulnerabilities. The preliminary inquiry required by this Chapter, serves as a part of these investigations, but notification of originators will not be delayed pending the completion of these investigations.

## 10-9. Unauthorized absences, suicides, or incapacitation

When an individual, who has had access to classified information, is absent without authorization, commits or attempts to commit suicide, or is temporarily or permanently incapacitated, the command will inquire into the situation to see if there are indications of activities, behavior, or associations, that could indicate classified information might be at risk. If so, the supporting counterintelligence organization will be notified. The scope and depth of this preliminary inquiry will depend on the length of the absence, factors leading to the actual or attempted suicide, or reasons and causes for the incapacitation, and the sensitivity of the classified information involved. See AR 190–40 for further details.

#### 10–10. Negligence

DOD military and civilian personnel are subject to administrative sanctions if they negligently disclose, to unauthorized persons, information properly classified under EO 12958 or any prior or subsequent order. Administrative action against U.S. military personnel, under the Uniform Code of Military Justice (UCMJ), can be pursued, but is not required. Administrative action against civilian personnel can be pursued under U.S. Army civilian personnel regulations, but is not required. No action is to be taken until a full inquiry has been completed to determine the seriousness of the incident.

#### Section II

# Extracts of Espionage Laws and Federal Statutes

# 10-11. United States Code, Title 18, Section 641 - Public Money, Property Or Records

Whoever embezzles, steals, purloins, or knowingly converts to his use or the use of another, or without authority, sells, conveys or disposes of any record, voucher, money, or thing of value of the United States or of any department or agency thereof, or any property made or being made under contract for the United States or any department or agency thereof; or Whoever receives, conceals, or retains the same with intent to convert it to his use or gain, knowing it to have been embezzled, stolen, purloined or converted – Shall be fined under this title or imprisoned not more than ten years, or both; but if the value of such property does not exceed the sum of \$1,000, he shall be fined under this title or imprisoned not more than one year, or both. The word "value" means face, par, or market value, or cost price, either wholesale or retail, whichever is greater.

# 10–12. United States Code, Title 18, Section 793 – Gathering, Transmitting, Or Losing Defense Information

*a.* Whoever, for the purpose of obtaining information respecting the national defense with intent or reason to believe that the information is to be used to the injury of the United States, or to the advantage of any foreign nation goes upon, enters, flies over, or otherwise obtains information concerning any vehicle, aircraft, work or defense, facilities,

fueling station, fort, battery, station, dockyard, engineering facility, railroad, arsenal, camp, factory, mine, telegraph, telephone, wireless, or signal station, vessel, navy yard, naval station, submarine base, torpedo station, or other place connected with the national defense owned or constructed, or in progress of construction by the United States or under the control of the United States, or of any of its officers, departments, or agencies, or within the exclusive jurisdiction of the United States, or any place in which any vessel, aircraft, arms, munitions, or other materials or instruments for use in time of war are being made, prepared, repaired, stored, or are the subject of research or development, under any contract or agreement with the United States, or any department or agency thereof, or with any person on behalf of the United States, or any prohibited place so designated by the President by proclamation in time of war or in case of national emergency in which anything for the use of the Army, Navy, or Air Force is being prepared or constructed or stored, information as to which prohibited place the President has determined would be prejudicial to the national defense; or

*b*. Whoever, for the purpose aforesaid, and with like intent or reason to believe, copies, takes, makes, or obtains, or attempts to copy, take, make, or obtain any sketch, photograph, photographic negative, blueprint, plan, map model, instrument, appliance, document, writing, or note of anything connected with the national defense; or

c. Whoever, for the purpose aforesaid, receives or obtains or agrees or attempts to receive or obtain from any person, or from any source whatever, any document, writing, code book, signal book, sketch, photograph, photographic negative, blueprint, plan, map, model, instrument, appliance, or note of anything connected with the national defense, knowing or having reason to believe, at the time he receives or obtains, or agrees or attempts to receive or obtain it, that it has been or will be obtained, taken, made or disposed of by any person contrary to the provisions of this Chapter; or

*d.* Whoever, lawfully having possession of, access to, control over, or being entrusted with any document, writing, code book, signal book, sketch, photograph, photographic negative, blueprint, plan, map, model, instrument, appliance, or note relating to the national defense, or information relating to the national defense, when information the possessor has reason to believe could be used to the injury of the United States or the advantage of any foreign nation, willfully communicates, delivers, transmits, or causes to be communicated, delivered, or transmitted, or attempts to communicate, deliver, transmit or cause to be communicated, delivered or transmitted the same to any person not entitled to receive it, or willfully retains the same and fails to deliver it on demand to the officer or employee of the United States entitled to receive it; or

*e*. Whoever, having unauthorized possession of, access to, or control over any document, writing, code book, signal book, sketch, photograph, photographic negative, blueprint, plan, map, model, instrument, appliance, or note relating to the national defense, or information relating to the national defense which information the possessor has reason to believe could be used to the injury of the United States or to the advantage of any foreign nation, willfully communicates, delivers, transmits or causes to be communicated, delivered, or transmitted, or attempts to communicate, deliver, transmit or cause to be communicated, delivered, or transmitted the same to any person not entitled to receive it, or willfully retains the same and fails to deliver it to the office or employee of the United States entitled to receive it; or

*f*. Whoever, being entrusted with or having lawful possession or control of any document, writing, code book, signal book, sketch, photograph, photographic negative, blueprint, plan, map model, instrument, appliance, note, or information relating to the national defense,

(1) Through gross negligence permits the same to be removed from its proper place of custody or delivered to anyone in violation of his trust, or to be lost, stolen, abstracted, or destroyed, or

(2) Having knowledge that the same has been illegally removed from its proper place of custody or delivered to anyone in violation of his trust, or lost, or stolen, abstracted, or destroyed, and fails to make prompt report of such loss, theft, abstraction, or destruction to his superior officer, will be fined not more than \$10,000 or imprisoned not more than ten(10) years, or both.

g. If two or more persons conspire to violate any of the foregoing provisions of this section, and one or more of such persons do any act to effect the object of the conspiracy, each of the parties to such conspiracy will be subject to the punishment provided for the offense which is the object of such conspiracy. June 25, 1948, c. 645, section 1, 62 Stat. 736, amended Sept. 23, 1950, c. 1024, section 18, 64 Stat.

# 10–13. United States Code Title 18, Section 794 – Gathering Or Delivering Defense Information To Aid Foreign Government

*a.* Whoever, with intent or reason to believe that it is to be used to the injury of the United States or to the advantage of a foreign nation, communicates, delivers, or transmits, or attempts to communicate, deliver, or transmit, to any foreign government, or to any faction or party or military or naval force within a foreign country, whether recognized or unrecognized by the United States, or to any representative, officer, agent, employee, subject, or citizen thereof, either directly or indirectly, any document, writing, code book, signal book, sketch, photograph, photographic negative, blueprint, plan, map, model, note, instrument, appliance, or information relating to the national defense, will be punished by death or by imprisonment for any term of years or for life.

b. Whoever, in time of war, with intent that same will be communicated to the enemy, collects, records, publishes, or communicates, or attempts to elicit any information with respect to the movement, numbers, description, condition,

or disposition of any of the Armed Forces, ships, aircraft, or war materials of the United States, or with respect to the plans or conduct, or supposed plans or conduct of any naval or military operations, or with respect to any works or measures undertaken for or connected with, or intended for the fortification or defense of any place, or any other information relating to the public defense, which might be useful to the enemy, will be punished by death or by imprisonment for any term of years or for life.

c. If two or more persons conspire to violate this section, and one or more of such persons do any act to effect the object of the conspiracy each of the parties to such conspiracy will be subject to the punishment provided for the offense which is the object of such conspiracy. As amended Sept. 3, 1954, c. 1261, Title II, section 201, 68 Stat. 1219.

# 10–14. United States Code, Title 18, Section 795 – Photographing And Sketching Defense Installations

*a.* Whenever, in the interest of national defense, the President defines certain vital military and naval installations or equipment as requiring protection against the general dissemination of information relative, thereto, it will be unlawful to make any photograph, sketch, picture, drawing, map, or graphical representation of such vital military and naval installations or equipment without first obtaining permission of the commanding officer of the military or naval post, camp, or station, or naval vessels, military or naval command concerned, or higher authority, and promptly submitting the product obtained to such command officer or higher authority for censorship or such other action as he may deem necessary.

b. Whoever violates this section will be fined not more than \$1,000 or imprisoned not more than one year, or both. (June 25, 1948, ch. 645, 62 Stat. 737.)

# 10-15. United States Code, Title 18, Section 796 - Use Of Aircraft For Photographs Of Defense Installations

Whoever uses or permits the use of an aircraft or any contrivance used, or designed for navigation or flight in the air, for the purpose of making a photograph, sketch, picture, drawing, map, or graphical representation of vital military or naval installations or equipment, in violation of section 795 of this title, will be fined not more than \$1,000 or imprisoned not more than one year, or both. (June 25, 1948, ch. 645,62 Stat. 738.)

# 10–16. United States Code, Title 18, Section 797 – Publication And Sale Of Photographs Of Defense Installations

On and after thirty days from the date upon which the President defines any vital military or naval installation or equipment as being within the category contemplated under section 795 of this title, whoever reproduces, publishes, sells, or gives away any photograph, sketch, picture, drawing, map, or graphical representation of the vital military or naval installations or equipment so defined, without first obtaining permission of the commanding officer of the military or naval post, camp, or station concerned, or higher authority, unless such photograph, sketch, picture, drawing, map, or graphical representation has clearly indicated thereon that it has been censored by the proper military or naval authority, will be fined not more than \$1,000 or imprisoned not more than one year, or both. (June 25, 1948, ch. 645,62 Stat. 738.)

# 10-17. United States Code, Title 18, Section 798 - Disclosure Of Classified Information

Whoever, knowingly, and willfully communicates, furnishes, transmits, or otherwise makes available to an unauthorized person, or publishes, or uses in any manner prejudicial to the safety or interest of the United States or for the benefit of any foreign government to the detriment of the United States any classified information: (1) concerning the nature, preparation, or use of any code, cipher, or cryptographic system of the United States or any foreign government; or (2) concerning the design, construction, use, maintenance, or repair or any device, apparatus, or appliance used or prepared or planned for use by the United States or any foreign government for cryptographic or communication intelligence purposes; or (3) concerning the communication intelligence from the communications of any foreign government, knowing the same to have been obtained by such processes – will be fined not more than \$10, 000 or imprisoned not more than ten (10) years, or both. (Added Oct 31, 1951, ch. 655, section 24(a), 65 Stat. 719.)

# 10-18. United States Code, Title 50, Section 797 - Violate Regulations And Aiding And Abetting

a. Whoever willfully will violate any such regulation or order as, pursuant to lawful authority, will be or has been promulgated or approved by the Secretary of Defense, or by any military commander designated by the Secretary of Defense, or by the Director of the National Advisory Committee for Aeronautics, for the protection or security of military or naval aircraft, airports, airport facilities, vessels, harbors, ports, piers, water-front facilities, bases, forts, posts, laboratories, stations, vehicles, equipment, explosives, or other property or places subject to the jurisdiction, administration, or in the custody of the Department of Defense, any Department or agency of which said Department consists, or any officer or employee thereof, relating to fire hazards, fire protection, lighting, machinery, guard service, disrepair, disuse or other re-satisfactory conditions thereon, or the ingress thereto or egress or removal of persons therefrom, or otherwise providing for safeguarding the same against destruction, loss, or injury by accident or

by enemy action, sabotage, or other subversive actions, will be guilty of a misdemeanor and upon conviction thereof will be liable to a fine of not to exceed \$5,000 or to imprisonment for not more than one year, or both.

b. Every such regulation or order will be posted in conspicuous and appropriate places. (Sept 23, 1950, ch. 1024, Title I, section 21, 64 Stat. 1005.)

#### 10-19. United States Code, Title 18, Section 952 - Diplomatic Codes And Correspondence

Whoever, by virtue of his employment by the United States, obtains from another or has or has had custody of or access to, any official diplomatic code or any matter prepared in any such code, or which purports to have been prepared in any such code, and without authorization or competent authority, willfully publishes or furnishes to another any such code or matter, or any matter which was obtained while in the process of transmission between any foreign government and its diplomatic mission in the United States, shall be fined under this title or imprisoned not more than ten years, or both.

#### 10-20. United States Code, Title 18, Section 1001 - False And Fraudulent Statements

Whoever, in any matter within the jurisdiction of any department or agency of the United States, knowingly and willfully falsifies, conceals, or covers up by any trick, scheme or device, a material fact or makes any false fictitious or fraudulent statements or representations or makes or uses any false writing or document knowing the same to contain any false, fictitious or fraudulent statement or entry, will be fined not more than \$10,000 or imprisoned not more than five (5) years, or both.

# 10-21. United States Code, Title 18, Section 1924 - Unauthorized Removal And Retention Of Classified Documents Or Material

*a.* Whoever, being an officer, employee, contractor, or consultant of the United States, and, by virtue of his office, employment, position, or contract, becomes possessed of documents or materials containing classified information of the United States, knowingly removes such documents or materials without authority and with the intent to retain such documents or materials at an unauthorized location shall be fined not more than \$1,000, or imprisoned for not more than one year, or both.

b. For purposes of this section, the provision of documents and materials to the Congress shall not constitute an offense under subsection a.

c. In this section, the term "classified information of the United States" means information originated, owned, or possessed by the United States Government concerning the national defense or foreign relations of the United States that has been determined pursuant to law or Executive Order to require protection against unauthorized disclosure in the interest of national security.

## 10-22. United States Code, Title 50, Sections 783 (B) And (D)

*a.* Section 783 (b). It will be unlawful for any officer or employee of the United States or any department or agency thereof, or of any corporation the stock of which is owned in whole or in major part by the United States or any department or agency thereof, to communicate in any manner or by any means, to any other person whom such officer or employee knows or has reason to believe to be an agent or representative of any foreign government or an officer or member of any Communist organization as defined in paragraph (5) of section 782 of this title, any information of a kind which will have been classified by the President (or by the head of any such department, agency, or corporation with the approval of the President) as affecting the security of the United States, knowing or having reason to know that such information has been so classified, unless such officer or employee will have been specifically authorized by the President or by the head of the department, agency or corporation by which this officer or employee is employed, to make such disclosure of such information.

b. Section 783 (d). Any person who violates any provision of this section will, upon conviction thereof, be punished by a fine of not more than \$10,000, or imprisonment for not more than ten (10) years, or by both such fine and such imprisonment, and will moreover, be thereafter ineligible to hold any office, or place of honor, profit, or trust created by the Constitution or laws of the United States.

#### 10-23. UNIFORM CODE OF MILITARY JUSTICE Article 106a ESPIONAGE

*a.* Any person subject to this Chapter who, with intent or reason to believe that it is to be used to the injury of the United States or to the advantage of a foreign nation, communicates, delivers, or transmits, or attempts to communicate, deliver, or transmit, to any entity described in paragraph (2), either directly or indirectly, any thing described in paragraph (3) will be punished as a court-martial may direct, except that if the accused is found guilty of an offense that directly concerns (A) nuclear weaponry, military spacecraft or satellites, early warning systems, or other means of defense or retaliation against large scale attack, (B) war plans, (C) communications intelligence or cryptographic information, or (D) any other major weapons system or major element of defense strategy, the accused will be punished by death or such other punishment as a court-martial may direct.

- b. An entity referred to in paragraph (1) is-
- (1) A foreign government;

(2) A faction or party or military or naval force within a foreign country, whether recognized or unrecognized by the United States; or

(3) A representative, officer, agent, employee, subject or citizen of such a government, faction, party, or force.

(4) A thing referred to in paragraph (1) is a document, writing, code book, signal book, sketch, photograph, photographic negative, blueprint, plan, map, model, note, instrument, appliance, or information relating to the national defense.

c. No person may be sentenced by court-martial to suffer death for an offense under this section (article) unless—(1) The members of the court-martial unanimously find at least one of the aggravating factors set out in subsection c; and

(2) The members unanimously determine that any extenuating or mitigating circumstances are substantially outweighed by any aggravating circumstances, including the aggravating factors set out under subsection (c).

(3) Findings under this subsection may be based on (A) evidence introduced on the issue of guilt or innocence; (B) evidence introduced during the sentencing proceeding; or (C) all such evidence.

(4) The accused will be given broad latitude to present matters in extenuation and mitigation.

*d*. A sentence of death may be adjudged by a court-martial for an offense under this section (article) only if the members unanimously find, beyond a reasonable doubt, one of more of the following aggravating factors:

(1) The accused has been convicted of another offense involving espionage or treason for which either a sentence of death or imprisonment for life was authorized by statute.

(2) In the commission of the offense, the accused knowingly created a grave risk of substantial damage to the national security.

(3) In the commission of the offense, the accused knowingly created a grave risk of death to another person.

(4) Any other factor that may be prescribed by the President by regulations under section 836 of this title (Article 36).

Sample Preliminary Inquiry Report

(Appropriate office symbol) (380-5s)

MEMORANDUM THRU (Commander at the next level up. May require more than one.)

MEMORANDUM FOR (See paragraph 10-4 for appropriate addressee.)

SUBJECT: Preliminary Inquiry - Possible Compromise of Classified Information

1. In compliance with AR 380-5, the following is my Report of Preliminary Inquiry.

2. Facts and Circumstances. (The investigating officer will be completely objective and consider all facts and circumstances to answer the following questions. When the facts are lengthy or complicated, a separate paragraph containing a narrative summary of events, in chronological order, may also be necessary.)

a. Who? (Complete identity of everyone involved, including responsible officials, and how they are involved.)

b. What? (Exact description of the information/material involved, what happened to it, and, if lost, what steps were taken to locate the missing information.)

c. When? (Date and time the incident occurred (if known) and the date and time the situation was discovered and reported.)

d. Where? (Complete identification of unit, section, activity, building and room number, and/or geographic location.)

e. How? (Circumstances of the incident, chronologically, relating how the information/material was lost or compromised. Summarize the evidence supporting your conclusion and attach supporting enclosure(s) when appropriate.)

f. Why? (What are the applicable policies, regulations, etc., for controlling the information/material involved? Were they followed? Was anyone negligent or derelict in their duties? Was the unit/activity SOP adequate to ensure compliance with applicable regulations/directives of higher headquarters and/or for ensuring the proper protection of the information/material concerned under the circumstances? Was the unit/activity's security education program sufficient enough to cover all information security aspects required by AR 380-5?)

3. Findings. When all of the above questions have been answered, the Investigating Officer will review the facts to reach findings on the following matters (*In cases of a media leak, use the eleven questions in paragraph 10-3e*):

a. Did a loss of classified information/material occur?

b. Did a compromise occur, or under the circumstances, what is the probability of compromise? (Or state that a compromise did not occur, or that there is minimal risk of damage to the national security.)

c. If a loss or compromise occurred, has a damage assessment been done? (*Outline the damage assessment procedures taken.*)

d. Is there any indication of significant security weaknesses in the unit/activity? If so, state them. (Were there any deficiencies in procedures for safeguarding classified information? Were there any violations of established procedures? If so, were they significant or contributory to the loss or compromise? How so?)

e. Is disciplinary action appropriate? Have administrative procedures as required by Chapter 1, section VI, of this regulation, been initiated? If so, what were they?

4. Recommendations. (The investigating officer will make specific recommendations based upon their findings. The recommendation(s) will include any relevant corrective action(s) or administrative sanctions consistent with the findings, but, as a minimum, will address the following:)

a. (A compromise of classified information did not occur.)

b. (If the findings are that a loss occurred and the probability of damage to the national security cannot be discounted, or it is determined that further investigation is likely to be productive, a recommendation that an investigation under AR 15-6, a counterintelligence investigation, or a criminal investigation, be conducted, may be included.)

c. (If there was a significant security weakness, the recommendation will include specific changes that should be made to correct the deficiency. In addition, further investigation, under AR 15-6, may be appropriate if the weakness resulted from conscious noncompliance with applicable regulations and directives.)

d. (If administrative sanction(s) are warranted, the recommendation(s) should identify the specific violation committed, by whom it was committed, and by whom the sanction(s) should be administered. If further investigation under AR 15-6 is not recommended for one of the reasons listed above, there need not be a recommendation for further investigation under AR 15-6 solely to impose administrative sanctions. The investigative officer should consult with the Staff Judge Advocate to determine whether

Figure 10-1. Sample Preliminary Inquiry Report—Continued

to recommend additional investigation or whether normal channels under the Uniform Code of Military Justice will suffice.)

e. (If further investigation under AR 15-6 is recommended, the recommendation will identify any person(s) who should be designated as the respondent and make a recommendation as to whether formal or informal investigation should be conducted, pursuant to AR 15-6, paragraph 1-2b.)

5. Comments. (Use this area for any additional comments not covered by the preceding paragraphs. If there are none, do not include this paragraph.)

6. Point of Contact. (*Give full identification of the investigating officer to include, full name, rank, SSAN, full unit designation/location, and phone number(s).*)

(Signature and Signature Block Of investigating officer)

Figure 10-1. Sample Preliminary Inquiry Report—Continued

# Appendix A References

Section I Required Publications

**DOD Directive 5200.1–R** Information Security Program Regulation (Cited in para 1–1)

**Executive Order 12958** Classified National Security Information (Cited in para 1–1)

Executive Order 12972 Amendment to Executive Order No. 12958 (Cited in para 3–5)

Executive Order 13142 Amendment to Executive Order No. 12958 (Cited in para 4–10)

Section II Related Publications

A related publication is merely a source of additional information. The user does not have to read it to understand this regulation.

AR 15-6 Procedures for Investigating Officers and Boards of Officers

AR 25–55 Freedom of Information Act Program

AR 25-400-2 The Modern Army Recordkeeping System (MARKS)

AR 190–13 The Army Physical Security Program

AR 190–16 Physical Security

AR 190–40 Serious Incident Report

AR 195–2 Criminal Investigation Activities

AR 380-10 Technology Transfer, Disclosure of Information and Contacts with Foreign Representatives

AR 380–15 Safeguarding Classified NATO Information

AR 380–19 Information Systems Security

AR 380–28 Department of the Army Special Security System

AR 380-40 Policy for Safeguarding and Controlling Communications Security (COMSEC) Material AR 380–49 Department of the Army Industrial Security Program

AR 380–53

Information Systems Security Monitoring

AR 380–67 Personnel Security Program

AR 380–381 Special Access Programs

AR 381-12 Subversion and Espionage Directed Against the U.S. Army (SAEDA)

AR 381–20 The Army Counterintelligence Program

AR 525–13 Antiterrorism Force Protection: Security of Personnel, Information, and Critical Resources

DCID 1/19

Security Policy for Sensitive Compartmented Information (SCI)

DCID 1/21

Physical Security Standards for SCIFs

DCID 1/7

Security Controls on the Dissemination of Intelligence Information

DCID 5/6

Intelligence Disclosure Policy, and the National Policy and Procedures for the Disclosure of Classified Military Information to Foreign Governments and International Organizations (abbreviated title: National Disclosure Policy 1 or NDP 1)

DOD Directive 5000.1

Defense Acquisition

DOD Directive 5030.47

National Supply System

**DOD Directive 5142.1** Assistant Secretary of Defense (Legislative Affairs)

**DOD 5200.1–H** DOD Handbook for Writing Security Classification Guidance

DOD O 5200.1-1

DOD Index of Security Classification Guides

DOD 5200.2-R Department of Defense Personnel Security Program (DODPSP)

**DOD Directive 5200.28** Security Requirements for Automated Data Processing (ADP) Systems

**DOD Directive 5210.2** Defense Courier Service **DOD Directive 5210.2** Access to and Dissemination of Restricted Data

**DOD Directive 5210.83** Department of Defense Controlled Unclassified Nuclear Information (DOD UCNI)

DOD 5220.22-M National Industrial Security Program (NISP) Operating Manual

**DOD 5220.22–R** Industrial Security Regulation

**DOD Directive 5230.11** Disclosure of Classified Military Information to Foreign Governments and International Organizations

**DOD Directive 5230.20** Visits, Assignments, and Exchanges of Foreign Nationals

**DOD Directive 5230.24** Distribution Statements on Technical Documents

**DOD Directive 5400.4** Provision of Information to Congress

DOD 5400.11-R Department of Defense Privacy Program

**DOD Directive 5405.2** Release of Official Information in Litigation and Testimony by DOD Personnel As Witnesses

**DOD Directive 7650.1** General Accounting Office (GAO) and Comptroller General Access to Records

DOD Handbook 5200.1–PH Department of Defense Handbook for Writing Security Classification Guidance

**DOD Instruction 5240.4** Reporting of Counterintelligence and Criminal Violations

**DOD Instruction 5505.2** Criminal Investigations of Fraud Offenses

DOD Manual 5105.21–M–1 Sensitive Compartmented Information Administrative Security Manual

Federal Standard FED-STD-809 Neutralization and Repair of GSA Approved Containers

FM 19–30 Physical Security

JCS Policy Memorandum 39 Release Procedures for JCS Papers

Joint Army-Navy-Air Force Publication (JANAP) 299 US Joint Code Word Index

Military Handbook 1013/1A Design Guidance for Physical Security of Facilities Office of Management and Budget, Information Security Oversight Office, Directive No. 1, Classified National Security Information (32 CFR Part 2001)

#### Title 18, United States Code

U.S. Central Registry

## Section III

#### **Prescribed Forms**

Except where indicated below, the following forms are available on the Army Electronic Library (AEL) CD-ROM (EM 0001) and the USAPA web site (www.usapa.army.mil).

#### DA Form 455

Mail and Document Register. (Prescribed in para 6-21b.)

## DA Form 969

Top Secret Document Record. (Prescribed in para 6–21f.)

#### **DA Form 1575**

Request for/or Notification of Regrading Action (Prescribed in para 2-22a.)

#### DA Form 1965

Delivery and Pick Up Service. (Prescribed in para 8-8.) Form is available in paper through normal forms supply channels.

#### **DD Form 2501**

Courier Authorization Card. (Prescribed in para 8–13b.) This form is available in paper through normal forms supply channels.

#### **DA Form 2962**

Security Termination Statement. (Prescribed in 6-5b.)

#### DA Form 3964

Classified Document Accountability Record. (Prescribed in para 6-29a.)

#### DOE Form 5631.20

Request for Visit or Access Approval. (Prescribed in para 6-7b and 6-17b.) This form is available at www.explorer.doe.gov:1776/htmls/DOEFORMS.html.

# SF 311

Agency Information Security Program Data. (Prescribed in para 1–7h.)

#### SF 312

Classified Information Nondisclosure Agreement (NDA). (Prescribed in para 6-2.)

#### SF 700

Security Container Information. (Prescribed in para 7–8c.) This form is available in paper through the Federal Supply Service, Ft. Worth, TX Phone: 817-978-2051.

#### SF 701

Activity Security Checklist. (Prescribed in para 6–11.) This form is available in paper through the Federal Supply Service, Ft. Worth, TX Phone: 817-978-2051.

#### SF 702

Security Container Check Sheet. (Prescribed in para 6–10b.) This form is available in paper through the Federal Supply Service, Ft. Worth, TX Phone: 817-978-2051. The electronic form is available at http://web1.osd.mil/icdhome/ sfeforms.htm

# SF 703

Top Secret Cover Sheet. (Prescribed in para 6–10a.) (This form is available in paper through normal forms supply channels.)

# SF 704

Secret Cover Sheet. (Prescribed in para 6-10a.) (This form is available in paper through normal forms supply channels.)

# SF 705

Confidential Cover Sheet. (Prescribed in para 6–10a.) (This form is available in paper through normal forms supply channels.)

# SF 706

Orange Top Secret Label. (Prescribed in para 4–34.) (This form is available in paper through normal forms supply channels.)

# SF 707

Red Secret Label. (Prescribed in para 4-34.) (This form is available in paper through normal forms supply channels.)

# SF 708

Blue Confidential Label. (Prescribed in para 4–34.) (This form is available in paper through normal forms supply channels.)

# SF 710

Green Unclassified Label. (Prescribed in para 4–34.) (This form is available in paper through normal forms supply channels.)

# SF 711

Data Descriptor Label. (Prescribed in para 4–34.) (This form is available in paper through normal forms supply channels.)

# SF 712

Classified SCI Label. (Prescribed in para 4–34.) This form is available in paper through the Federal Supply Service, Ft. Worth, TX Phone: 817-978-2051.

## Section IV Referenced Forms

DA Form 11-2-R

Management Control Evaluation Certification Statement

# DA Form 873

Certificate of Clearance and/or Security Determination

# DD Form 2

Armed Forces of the United States Geneva Convention Identification Card

# **DD Form 173/1**

Joint Message Form. (This form is available in paper through normal forms supply channels.

# **DD Form 1610**

Request and Authorization for TDY Travel of DOD Personnel

# **DD Form 1847–1**

Sensitive Compartmented Information Nondisclosure Statement. (This form is available in paper through normal forms supply channels.)

# DD Form 2024

DOD Security Classification Guide Data Elements

# **DD** Form 2056

Telephone Monitoring Notification Decal. (This form is available in paper through normal forms supply channels.)

# SF 75

Request for Preliminary Employment Data. (This form is available in paper through normal supply channels.)

# SF 135

Records Transmittal and Receipt

### SF 135-A

Records Transmittal and Receipt Continuation

### Appendix B Presidential Executive Orders (EO) 12958, EO 12972 and EO 13142

This regulation implements the policy set forth in Executive Order (E.O.) 12958, Classified National Security Information, April 17, 1995, with amendments, and Department of Defense Directive 5200.1–R, Information Security Program, January 14, 1997. The following are reprints of Executive Order (EO) 12958 and its supplements, EO 12972 and EO 13142."

## 1.0. — Section 1.0 Executive Order 12958 of April 17, 1995 The President

## **Classified National Security Information**

This Order prescribes a uniform system for classifying, safeguarding, and declassifying national security information. Our democratic principles require that the American people be informed of the activities of their Government. Also, our nation's progress depends on the free flow of information. Nevertheless, throughout our history, the national interest has required that certain information be maintained in confidence in Order to protect our citizens, our democratic institutions, and our participation within the community of nations. Protecting information critical to our Nation's security remains a priority. In recent years, however, dramatic changes have altered, although not eliminated, the national security threats that we confront. These changes provide a greater opportunity to emphasize our commitment to open Government.

NOW, THEREFORE, by the authority vested in me as President by the Constitution and the laws of the United States of America, it is hereby Ordered as follows:

#### Part 1 – Original Classification

# 1.1. — Section 1.1 Definitions

For purposes of this Order:

a. "National security" means the national defense or foreign relations of the United States.

b. "Information" means any knowledge that can be communicated or documentary material, regardless of its physical form or characteristics, that is owned by, produced by or for, or is under the control of the United States Government. "Control" means the authority of the agency that originates information, or its successor in function, to regulate access to the information.

c. "Classified national security information" (hereafter "classified information") means information that has been determined pursuant to this Order or any predecessor Order to require protection against unauthorized disclosure and is marked to indicate its classified status when in documentary form.

d. "Foreign Government Information" means:

(1) Information provided to the United States Government by a foreign government or governments, an international organization of governments, or any element thereof, with the expectation that the information, the source of the information, or both, are to be held in confidence.

(2) Information produced by the United States pursuant to or as a result of a joint arrangement with a foreign government or governments, or an international organization of governments, or any element thereof, requiring that the information, the arrangement, or both, are to be held in confidence.

(3) Information received and treated as "Foreign Government Information" under the terms of a predecessor Order. *e.* "Classification" means the act or process by which information is determined to be classified information.

*f.* "Original classification" means an initial determination that information requires, in the interest of national security, protection against unauthorized disclosure.

g. "Original classification authority" means an individual authorized in writing, either by the President, or by agency heads or other officials designated by the President, to classify information in the first instance.

h. "Unauthorized disclosure" means a communication or physical transfer of classified information to an unauthorized recipient.

*i.* "Agency" means any "Executive agency," as defined in 5 USC 105, and any other entity within the executive branch that comes into the possession of classified information.

*j.* "Senior agency official" means the official designated by the agency head under section 5.6(c) of this Order to direct and administer the agency's program under which information is classified, safeguarded and declassified.

k. "Confidential source" means any individual or organization that has provided, or that may reasonably be expected to provide, information to the United States on matters pertaining to the national security with the expectation that the information or relationship, or both, are to be held in confidence.

*l.* "Damage to the national security" means harm to the national defense or foreign relations of the United States from the unauthorized disclosure of information, to include the sensitivity, value, and utility of that information.

# 1.2. — Section 1.2

## **Classification Standards**

*a*. Information may be originally classified under the terms of this Order only if all of the following conditions are met:

(1) An original classification authority is classifying the information.

(2) The information is owned by, produced by or for, or is under the control of the United States Government.

(3) The information falls within one or more of the categories of information listed in section 1.5 of this Order.

(4) The original classification authority determines that the unauthorized disclosure of the information reasonably could be expected to result in damage to the national security and the original classification authority is able to identify or describe the damage.

*b*. If there is significant doubt about the need to classify information, it will not be classified. This provision does not:

(1) Amplify or modify the substantive criteria or procedures for classification.

(2) Create any substantive or procedural rights subject to judicial review.

c. Classified information will not be declassified automatically as a result of any unauthorized disclosure of identical or similar information.

# 1.3. — Section 1.3

# Classification Levels

a. Information may be classified at one of the following three levels:

(1) "TOP SECRET" will be applied to information, the unauthorized disclosure of which reasonably could be expected to cause exceptionally grave damage to the national security that the original classification authority is able to identify or describe.

(2) "SECRET" will be applied to information, the unauthorized disclosure of which reasonably could be expected to cause serious damage to the national security that the original classification authority is able to identify or describe.

(3) "CONFIDENTIAL" will be applied to information, the unauthorized disclosure of which reasonably could be expected to cause damage to the national security that the original classification authority is able to identify or describe.

b. Except as otherwise provided by statute, no other terms will be used to identify United States classified information.

c. If there is significant doubt about the appropriate level of classification, it will be classified at the lower level.

# 1.4. — Section 1.4

#### **Classification Authority**

a. The authority to classify information originally may be exercised only by:

(1) The President.

(2) Agency heads and officials designated by the President in the Federal Register.

(3) United States Government officials delegated this authority pursuant to paragraph (c), below.

b. Officials authorized to classify information at a specified level are also authorized to classify information at a lower level.

c. Delegation of original classification authority.

(1) Delegation of original classification authority will be limited to the minimum required to administer this Order. Agency heads are responsible for ensuring that designated subordinate officials have a demonstrable and continuing need to exercise this authority.

(2) "TOP SECRET" original classification authority may be delegated only by the President or by an agency head or official designated pursuant to paragraph (a)(2), above.

(3) "SECRET" or "CONFIDENTIAL" original classification authority may be delegated only by the President; an

agency head or official designated pursuant to paragraph a.2, above; or the senior agency official, provided that official has been delegated "TOP SECRET" original classification authority by the agency head.

(4) Each delegation of original classification authority will be in writing and the authority will not be re-delegated except as provided in this Order. Each delegation will identify the official by name or position title.

*d*. Original classification authorities must receive training in original classification as provided in this Order and its implementing directives.

*e*. Exceptional Cases. When an employee, contractor, licensee, certificate holder, or grantee of an agency that does not have original classification authority originates information believed by that person to require classification, the information will be protected in a manner consistent with this Order and its implementing directives. The information will be transmitted promptly as provided under this Order or its implementing directives to the agency that has appropriate subject matter interest and classification authority with respect to this information. That agency will decide within 30 days whether to classify this information. If it is not clear which agency has classification responsibility for this information, it will be sent to the Director of the Information Security Oversight Office (ISOO). The Director will determine the agency having primary subject matter interest and forward the information, with appropriate recommendations, to that agency for a classification determination.

# 1.5. — Section 1.5

## **Classification Categories**

Information may not be considered for classification unless it concerns:

- a. Military plans, weapons systems, or operations.
- b. Foreign government information.
- c. Intelligence activities (including special activities), intelligence sources or methods, or cryptology.
- d. Foreign relations or foreign activities of the United States, including confidential sources.
- e. Scientific, technological, or economic matters relating to the national security.
- f. United States Government programs for safeguarding nuclear materials or facilities; or
- g. Vulnerabilities or capabilities of systems, installations, projects or plans relating to the national security.

# 1.6. — Section 1.6

#### Duration of Classification

*a*. At the time of original classification, the original classification authority will attempt to establish a specific date or event for declassification based upon the duration of the national security sensitivity of the information. The date or event will not exceed the time frame in paragraph b, below.

b. If the original classification authority cannot determine an earlier specific date or event for declassification, information will be marked for declassification 10 years from the date of the original decision, except as provided in paragraph d, below.

c. An original classification authority may extend the duration of classification or reclassify specific information for successive periods not to exceed 10 years at a time if such action is consistent with the standards and procedures established under this Order. This provision does not apply to information contained in records that are more than 25 years old and have been determined to have permanent historical value under Title 44, United States Code.

*d*. At the time of original classification, the original classification authority may exempt from declassification within 10 years specific information, the unauthorized disclosure of which could reasonably be expected to cause damage to the national security, for a period greater than that provided in paragraph b, above, and the release of which could reasonably be expected to:

(1) Reveal an intelligence source, method, or activity, or a cryptologic system or activity.

(2) Reveal information that would assist in the development or use of weapons of mass destruction.

(3) Reveal information that would impair the development or use of technology within a United States weapon system.

(4) Reveal United States military plans, or national security emergency preparedness plans.

(5) Reveal foreign government information.

(6) Damage relations between the United States and a foreign government, reveal a confidential source, or seriously undermine diplomatic activities that are reasonably expected to be ongoing for a period greater than that provided in paragraph b, above.

(7) Impair the ability of responsible United States Government officials to protect the President, the Vice President, and other individuals for whom protection services, in the interest of national security, are authorized.

(8) Violate a statute, treaty, or international agreement.

*e*. Information marked for an indefinite duration of classification under predecessor Orders, for example, "Originating Agency's Determination Required" (OADR), or information classified under predecessor Orders that contains no declassification instructions will be declassified in accordance with part 3 of this Order.

#### 1.7. — Section 1.7 Identification and Markings

*a*. At the time of original classification, the following will appear on the face of each classified document, or will be applied to other classified media in an appropriate manner:

(1) One of the three classification levels defined in Section 1.3 of this Order.

- (2) The identity, by name or personal identifier and position, of the original classification authority.
- (3) The agency and office of origin, if not otherwise evident.

(4) Declassification instructions, which will indicate one of the following:

- (a) The date or event for declassification, as prescribed in Section 1.6.a or Section 1.6.c.
- (b) The date that is ten years from the date of original classification, as prescribed in Section 1.6.b.

(c) The exemption category from automatic declassification, as prescribed in Section 1.6.d.

(5) A concise reason for classification which, at a minimum, cites the applicable classification categories in Section 1.5 of this Order.

b. Specific information contained in paragraph a, above, may be excluded if it would reveal additional classified information.

c. Each classified document will, by marking or other means, indicate which portions are classified, with the applicable classification level, which portions are exempt from declassification under section 1.6.d of this Order, and which portions are unclassified. In accordance with standards prescribed in directives issued under this Order, the Director of the Information Security Oversight Office may grant waivers of this requirement for specified classes of documents or information. The Director will revoke any waiver upon a finding of abuse.

*d*. Markings implementing the provisions of this Order, including abbreviations and requirements to safeguard classified working papers, will conform to the standards prescribed in implementing directives issued pursuant to this Order.

*e*. Foreign government information will retain its original classification markings or will be assigned a U.S. classification that provides a degree of protection at least equivalent to that required by the entity that furnished the information.

*f*. Information assigned a level of classification under this or predecessor Orders will be considered as classified at that level of classification despite the omission of other required markings. Whenever such information is used in the derivative classification process or is reviewed for possible declassification, holders of such information will coordinate with an appropriate classification authority for the application of omitted markings.

g. The classification authority will, whenever practicable, use a classified addendum whenever classified information constitutes a small portion of an otherwise unclassified document.

# 1.8. — Section 1.8

# **Classification Prohibitions and Limitations**

a. In no case will information be classified in Order to:

(1) Conceal violations of law, inefficiency, or administrative error.

(2) Prevent embarrassment to a person, organization, or agency.

(3) Restrain competition.

(4) Prevent or delay the release of information that does not require protection in the interest of national security.

b. Basic scientific research information not clearly related to the national security may not be classified.

c. Information may not be reclassified after it has been declassified and released to the public under proper authority.

*d.* Information that has not previously been disclosed to the public under proper authority may be classified or reclassified after an agency has received a request for it under the Freedom of Information Act (5 USC 552) or the Privacy Act of 1974 (5 USC 552a), or the mandatory review provisions of section 3.6 of this Order only if such classification meets the requirements of this Order and is accomplished on a document–by–document basis with the personal participation or under the direction of the agency head, the deputy agency head, or the senior agency official designated under section 5.6 of this Order. This provision does not apply to classified information contained in records that are more than 25 years old and have been determined to have permanent historical value under title 44, United States Code.

*e*. Compilations of items of information which are individually unclassified may be classified if the compiled information reveals an additional association or relationship that:

(1) Meets the standards for classification under this Order.

(2) Is not otherwise revealed in the individual items of information. As used in this Order, "compilation" means an aggregation of pre-existing unclassified items of information.

#### 1.9. — Section 1.9 Classification Challenges

*a*. Authorized holders of information who, in good faith, believe that its classification status is improper are encouraged and expected to challenge the classification status of the information in accordance with agency procedures established under paragraph b, below.

b. In accordance with implementing directives issued pursuant to this Order, an agency head or senior agency official will establish procedures under which authorized holders of information are encouraged and expected to challenge the classification of information that they believe is improperly classified or unclassified. These procedures will assure that:

(1) Individuals are not subject to retribution for bringing such actions.

(2) An opportunity is provided for review by an impartial official or panel.

(3) Individuals are advised of their right to appeal agency decisions to the Interagency Security Classification Appeals Panel established by Section 5.4 of this Order.

#### Part 2 – Derivative Classification

2.1. — Section 2.1 Definitions

#### For purposes of this Order:

*a.* "Derivative classification" means the incorporating, paraphrasing, restating or generating in new form information that is already classified, and marking the newly developed material consistent with the classification markings that apply to the source information. Derivative classification includes the classification of information based on classification guidance. The duplication or reproduction of existing classified information is not derivative classification.

b. "Classification guidance" means any instruction or source that prescribes the classification of specific information.

c. "Classification guide" means a documentary form of classification guidance issued by an original classification authority that identifies the elements of information regarding a specific subject that must be classified and establishes the level and duration of classification for each such element.

d. "Source document" means an existing document that contains classified information that is incorporated, paraphrased, restated, or generated in new form into a new document.

e. "Multiple sources" means two or more source documents, classification guides or a combination of both.

# 2.2. — Section 2.2

#### Use of Derivative Classification

*a*. Persons who only reproduce, extract, or summarize classified information, or who only apply classification markings derived from source material or as directed by a classification guide, need not possess original classification authority.

b. Persons who apply derivative classification markings will-

(1) Observe and respect original classification decisions.

(2) Carry forward to any newly created documents the pertinent classification markings. For information derivatively classified based on multiple sources, the derivative classifier will carry forward:

(a) The date or event for declassification that corresponds to the longest period of classification among the sources.

(b) Listing of these sources on or attached to the official file or record copy.

#### 2.3. — Section 2.3 Classification Guides

*a*. Agencies with original classification authority will prepare classification guides to facilitate the proper and uniform derivative classification of information. These guides will conform to standards contained in directives issued under this Order.

b. Each guide will be approved personally and in writing by an official who:

(1) Has program or supervisory responsibility over the information or is the senior agency official.

(2) Is authorized to classify information originally at the highest level of classification prescribed in the guide.

c. Agencies will establish procedures to assure that classification guides are reviewed and updated as provided in directives issued under this Order.

#### Part 3 – Declassification and Downgrading

# 3.1. — Section 3.1 Definitions

#### For purposes of this Order:

a. "Declassification" means the authorized change in the status of information from classified information to unclassified information.

b. "Automatic declassification" means the declassification of information based solely upon:

(1) The occurrence of a specific date or event as determined by the original classification authority.

(2) The expiration of a maximum time frame for duration of classification established under this Order.

c. "Declassification authority" means:

(1) The official who authorized the original classification, if that official is still serving in the same position.

(2) The originator's current successor in function.

(3) A supervisory official.

(4) Officials delegated declassification authority in writing by the agency head or the senior agency official.

d. "Mandatory declassification review" means the review for declassification of classified information in response to a request for declassification that meets the requirements under Section 3.6 of this Order.

e. "Systematic declassification review" means the review for declassification of classified information contained in records that have been determined by the Archivist of the United States ("Archivist") to have permanent historical value in accordance with Chapter 33 of Title 44, United States Code.

f. "Declassification guide" means written instructions issued by a declassification authority that describes the elements of information regarding a specific subject that may be declassified and the elements that must remain classified.

g. "Downgrading" means a determination by a declassification authority that information classified and safeguarded at a specified level will be classified and safeguarded at a lower level.

*h.* "File series" means documentary material, regardless of its physical form or characteristics, that is arranged in accordance with a filing system or maintained as a unit because it pertains to the same function or activity.

#### 3.2. — Section 3.2 Authority for Declassificati

# Authority for Declassification

a. Information will be declassified as soon as it no longer meets the standards for classification under this Order. b. It is presumed that information that continues to meet the classification requirements under this Order requires continued protection. In some exceptional cases, however, the need to protect such information may be outweighed by the public interest in disclosure of the information, and in these cases the information should be declassified. When such questions arise, they will be referred to the agency head or the senior agency official. That official will determine, as an exercise of discretion, whether the public interest in disclosure outweighs the damage to national security that might reasonably be expected from disclosure. This provision does not:

(1) Amplify or modify the substantive criteria or procedures for classification.

(2) Create any substantive or procedural rights subject to judicial review.

c. If the Director of the Information Security Oversight Office determines that information is classified in violation of this Order, the Director may require the information to be declassified by the agency that originated the classification. Any such decision by the Director may be appealed to the President through the Assistant to the President for National Security Affairs. The information will remain classified pending a prompt decision on the appeal.

d. The provisions of this section will also apply to agencies that, under the terms of this Order, do not have original classification authority, but had such authority under predecessor Orders.

# 3.3. — Section 3.3

#### **Transferred Information**

*a*. In the case of classified information transferred in conjunction with a transfer of functions, and not merely for storage purposes, the receiving agency will be deemed to be the originating agency for purposes of this Order.

b. In the case of classified information that is not officially transferred as described in paragraph a, above, but that originated in an agency that has ceased to exist and for which there is no successor agency, each agency in possession of such information will be deemed to be the originating agency for purposes of this Order. Such information may be declassified or downgraded by the agency in possession after consultation with any other agency that has an interest in the subject matter of the information.

c. Classified information accessioned into the National Archives and Records Administration ("National Archives")

as of the effective date of this Order will be declassified or downgraded by the Archivist in accordance with this Order, the directives issued pursuant to this Order, agency declassification guides, and any existing procedural agreement between the Archivist and the relevant agency head.

*d.* The originating agency will take all reasonable steps to declassify classified information contained in records determined to have permanent historical value before they are accessioned into the National Archives. However, the Archivist may require that records containing classified information be accessioned into the National Archives when necessary to comply with the provisions of the Federal Records Act. This provision does not apply to information being transferred to the Archivist pursuant to Section 2203 of Title 44, United States Code, or information for which the National Archives and Records Administration serves as the custodian of the records of an agency or organization that goes out of existence.

*e*. To the extent practicable, agencies will adopt a system of records management that will facilitate the public release of documents at the time such documents are declassified pursuant to the provisions for automatic declassification in sections 1.6 and 3.4 of this Order.

### 3.4. — Section 3.4

# Automatic Declassification

*a.* Subject to paragraph b, below, within five years from the date of this Order, all classified information contained in records that: (1) are more than 25 years old; and (2) have been determined to have permanent historical value under Title 44, United States Code, will be automatically declassified whether or not the records have been reviewed. Subsequently, all classified information in such records will be automatically declassified no longer than 25 years from the date of its original classification, except as provided in paragraph b, below.

b. An agency head may exempt from automatic declassification under paragraph a, above, specific information, the release of which should be expected to:

(1) Reveal the identity of a confidential human source, or reveal information about the application of an intelligence source or method, or reveal the identity of a human intelligence source when the unauthorized disclosure of that source would clearly and demonstrably damage the national security interests of the United States.

(2) Reveal information that would assist in the development or use of weapons of mass destruction.

(3) Reveal information that would impair U.S. cryptologic systems or activities.

(4) Reveal information that would impair the application of state of the art technology within a U.S. weapon system.

(5) Reveal actual U.S. military war plans that remain in effect.

(6) Reveal information that would seriously and demonstrably impair relations between the United States and a foreign government, or seriously and demonstrably undermine ongoing diplomatic activities of the United States.

(7) Reveal information that would clearly and demonstrably impair the current ability of United States Government officials to protect the President, Vice President, and other officials for whom protection services, in the interest of national security, are authorized.

(8) Reveal information that would seriously and demonstrably impair current national security emergency preparedness plans.

(9) Violate a statute, treaty, or international agreement.

c. No later than the effective date of this Order, an agency head will notify the President through the Assistant to the President for National Security Affairs of any specific file series of records for which a review or assessment has determined that the information within those file series almost invariably falls within one or more of the exemption categories listed in paragraph b, above, and which the agency proposes to exempt from automatic declassification. The notification will include:

(1) A description of the file series.

(2) An explanation of why the information within the file series is almost invariably exempt from automatic declassification and why the information must remain classified for a longer period of time.

(3) Except for the identity of a confidential human source or a human intelligence source, as provided in paragraph b, above, a specific date or event for declassification of the information. The President may direct the agency head not to exempt the file series or to declassify the information within that series at an earlier date than recommended.

*d.* At least 180 days before information is automatically declassified under this section, an agency head or senior agency official will notify the Director of the Information Security Oversight Office, serving as executive Secretary of the Interagency Security Classification Appeals Panel, of any specific information beyond that included in a notification to the President under paragraph c, above, that the agency proposes to exempt from automatic declassification. The notification will include:

(1) A description of the information.

(2) an explanation of why the information is exempt from automatic declassification and must remain classified for a longer period of time.

(3) Except for the identity of a confidential human source or a human intelligence source, as provided in paragraph b, above, a specific date or event for declassification of the information. The Panel may direct the agency not to exempt the information or to declassify it at an earlier date than recommended. The agency head may appeal such a

decision to the President through the Assistant to the President for National Security Affairs. The information will remain classified while such an appeal is pending.

*e*. No later than the effective date of this Order, the agency head or senior agency official will provide the Director of the Information Security Oversight Office with a plan for compliance with the requirements of this section, including the establishment of interim target dates. Each such plan will include the requirement that the agency declassify at least 15 percent of the records affected by this section no later than 1 year from the effective date of this Order, and similar commitments for subsequent years until the effective date for automatic declassification.

f. Information exempted from automatic declassification under this section will remain subject to the mandatory and systematic declassification review provisions of this Order.

g. The Secretary of State will determine when the United States should commence negotiations with the appropriate officials of a foreign government or international organization of governments to modify any treaty international agreement that requires the classification of information contained in records affected by this section for a period longer than 25 years from the date of its creation, unless the treaty or international agreement pertains to information that may otherwise remain classified beyond 25 years under this section.

## 3.5. — Section 3.5

#### Systematic Declassification Review

*a*. Each agency that has originated classified information under this Order or its predecessors will establish and conduct a program for systematic declassification review. This program will apply to historically valuable records exempted from automatic declassification under Section 3.4 of this Order. Agencies will prioritize the systematic review of records based upon:

(1) Recommendations of the Information Security Policy Advisory Council, established in Section 5.5 of this Order, on specific subject areas for systematic review concentration.

(2) The degree of researcher interest and the likelihood of declassification upon review.

b. The Archivist will conduct a systematic declassification review program for classified information 1), accessioned into the National Archives as of the effective date of this Order 2), information transferred to the Archivist pursuant to Section 2203 of Title 44, United States Code, and 3), information for which the National Archives and Records Administration serves as the custodian of the records of an agency or organization that has gone out of existence. This program will apply to pertinent records no later than 25 years from the date of their creation. The Archivist will establish priorities for the systematic review of these records based upon the recommendations of the Information Security Policy Advisory Council, or the degree of researcher interest and the likelihood of declassification upon review. These records will be reviewed in accordance with the standards of this Order, its implementing directives, and declassification guides provided to the Archivist by each agency that originated the records. The Director of the Information Security Oversight Office will assure that agencies provide the Archivist with adequate and current declassification guides.

c. After consultation with affected agencies, the Secretary of Defense may establish special procedures for systematic review for declassification of classified cryptologic information, and the Director of Central Intelligence may establish special procedures for systematic review for declassification of classified information pertaining to intelligence activities (including special activities), or intelligence sources or methods.

# 3.6. — Section 3.6

## Mandatory Declassification Review

a. Except as provided in paragraph b, below, all information classified under this Order or predecessor Orders will be subject to a review for declassification by the originating agency if:

(1) The request for a review describes the document or material containing the information with sufficient specificity to enable the agency to locate it with a reasonable amount of effort.

(2) The information is not exempted from search and review under the Central Intelligence Agency Information Act.

(3) The information has not been reviewed for declassification within the past two years. If the agency has reviewed the information within the past two years, or the information is the subject of pending litigation, the agency will inform the requester of this fact and of the requester's appeal rights.

- b. Information originated by:
- (1) The incumbent President.
- (2) The incumbent President's White House Staff.
- (3) Committees, commissions, or boards appointed by the incumbent President.

(4) Other entities within the Executive Office of the President that solely advise and assist the incumbent President is exempted from the provisions of paragraph a, above. However, the Archivist will have the authority to review, downgrade, and declassify information of former presidents under the control of the Archivist pursuant to Sections 2107, 2111, 2111 note, or 2203 of Title 44, United States Code. Review procedures developed by the Archivist will provide for consultation with agencies having primary subject matter interest and will be consistent with the provisions of applicable laws or lawful agreements that pertain to the respective presidential papers or records. Agencies with

primary subject matter interest will be notified promptly of the Archivist's decision. Any final decision by the Archivist may be appealed by the requester or an agency to the Interagency Security Classification Appeals Panel. The information will remain classified pending a prompt decision on the appeal.

c. Agencies conducting a mandatory review for declassification will declassify information that no longer meets the standards for classification under this Order. They will release this information unless withholding is otherwise authorized and warranted under applicable law.

*d*. In accordance with directives issued pursuant to this Order, agency heads will develop procedures to process requests for the mandatory review of classified information. These procedures will apply to information classified under this or predecessor Orders. They also will provide a means for administratively appealing a denial of a mandatory review request, and for notifying the requester of the right to appeal a final agency decision to the Interagency Security Classification Appeals Panel.

*e*. After consultation with affected agencies, the Secretary of Defense will develop special procedures for the review of cryptologic information, the Director of Central Intelligence will develop special procedures for the review of information pertaining to intelligence activities (including special activities), or intelligence sources or methods, and the Archivist will develop special procedures for the review of information accessioned into the National Archives.

#### 3.7. — Section 3.7

#### **Processing Requests and Reviews**

In response to a request for information under the Freedom of Information Act, the Privacy Act of 1974, or the mandatory review provisions of this Order, or pursuant to the automatic declassification or systematic review provisions of this Order:

a. An agency may refuse to confirm or deny the existence or non-existence of requested information whenever the fact of its existence or non-existence is itself classified under this Order.

b. When an agency receives any request for documents in its custody that contain information that was originally classified by another agency, or comes across such documents in the process of the automatic declassification or systematic review provisions of this Order, it will refer copies of any request and the pertinent documents to the originating agency for processing, and may, after consultation with the originating agency, inform any requester of the referral unless such association is itself classified under this Order. In cases in which the originating agency determines in writing that a response under paragraph (a), above, is required, the referring agency will respond to the requester in accordance with that paragraph.

#### 3.8. — Section 3.8 Declassification Database

*a*. The Archivist in conjunction with the Director of the Information Security Oversight Office and those agencies that originate classified information, will establish a Government–wide database of information that has been declassified. The Archivist will also explore other possible uses of technology to facilitate the declassification process.

b. Agency heads will fully cooperate with the Archivist in these efforts.

c. Except as otherwise authorized and warranted by law, all declassified information contained within the database established under paragraph (a), above, will be available to the public.

Part 4 – Safeguarding

4.1. — Section 4.1 Definitions

#### For purposes of this Order:

a. "Safeguarding" means measures and controls that are prescribed to protect classified information.

b. "Access" means the ability or opportunity to gain knowledge of classified information.

c. "Need-to-know" means a determination made by an authorized holder of classified information that a prospective recipient requires access to specific classified information in Order to perform or assist in a lawful and authorized governmental function.

d. "Automated information system" means an assembly of computer hardware, software, or firmware configured to collect, create, communicate, compute, disseminate, process, store, or control data or information.

e. "Integrity" means the state that exists when information is unchanged from its source and has not been accidentally or intentionally modified, altered, or destroyed.

f. "Network" means a system of two or more computers that can exchange data or information.

g. "Telecommunications" means the preparation, transmission, or communication of information by electronic means.

*h.* "Special access program" means a program established for a specific class of classified information that imposes safeguarding and access requirements that exceed those normally required for information at the same classification level.

#### 4.2. — Section 4.2 General Restrictions on Access

a. A person may have access to classified information provided that:

(1) A favorable determination of eligibility for access has been made by an agency head or the agency head's designee.

(2) The person has signed an approved nondisclosure agreement.

(3) The person has a need-to-know the information.

b. Classified information will remain under the control of the originating agency or its successor in function. An agency will not disclose information originally classified by another agency without its authorization. An official or employee leaving agency service may not remove classified information from the agency's control.

c. Classified information may not be removed from official premises without proper authorization.

*d*. Persons authorized to disseminate classified information outside the executive branch will assure the protection of the information in a manner equivalent to that provided within the executive branch.

*e*. Consistent with law, directives and regulation, an agency head or senior agency official will establish uniform procedures to ensure that automated information systems, including networks and telecommunications systems, that collect, create, communicate, compute, disseminate, process, or store classified information have controls that:

(1) Prevent access by unauthorized persons.

(2) Ensure the integrity of the information.

*f.* Consistent with law, directives and regulation, each agency head or senior agency official will establish controls to ensure that classified information is used, processed, stored, reproduced, transmitted, and destroyed under conditions that provide adequate protection and prevent access by unauthorized persons.

g. Consistent with directives issued pursuant to this Order, an agency will safeguard foreign government information under standards that provide a degree of protection at least equivalent to that required by the government or international organization of governments that furnished the information. When adequate to achieve equivalency, these standards may be less restrictive than the safeguarding standards that ordinarily apply to United States CONFIDEN-TIAL information, including allowing access to individuals with a need to know who have not otherwise been cleared for access to classified information or executed an approved nondisclosure agreement.

*h*. Except as provided by statute or directives issued pursuant to this Order, classified information originating in one agency may not be disseminated outside any other agency to which it has been made available without the consent of the originating agency. An agency head or senior agency official may waive this requirement for specific information originated within that agency. For purposes of this section, the Department of Defense will be considered one agency.

## 4.3. — Section 4.3 Distribution Controls

*a*. Each agency will establish controls over the distribution of classified information to assure that it is distributed only to organizations or individuals eligible for access who also have a need to know the information.

*b*. Each agency will update, at least annually, the automatic, routine, or recurring distribution of classified information that they distribute. Recipients will cooperate fully with distributors who are updating distribution lists and will notify distributors whenever a relevant change in status occurs.

# 4.4. — Section 4.4 Special Access Programs

*a.* Establishment of special access programs. Unless otherwise authorized by the President, only the Secretaries of State, Defense and Energy, and the Director of Central Intelligence, or the principal deputy of each, may create a special access program. For special access programs pertaining to intelligence activities (including special activities, but not including military operational, strategic and tactical programs), or intelligence sources or methods, this function will be exercised by the Director of Central Intelligence. These officials will keep the number of these programs at an absolute minimum, and will establish them only upon a specific finding that:

(1) The vulnerability of, or threat to, specific information is exceptional.

(2) The normal criteria for determining eligibility for access applicable to information classified at the same level are not deemed sufficient to protect the information from unauthorized disclosure.

(3) The program is required by statute.

b. Requirements and limitations.

(1) Special access programs will be limited to programs in which the number of persons who will have access ordinarily will be reasonably small and commensurate with the objective of providing enhanced protection for the information involved.

(2) Each agency head will establish and maintain a system of accounting for special access programs consistent with directives issued pursuant to this Order.

(3) Special access programs will be subject to the oversight program established under section 5.6(c) of this Order.

In addition, the Director of the Information Security Oversight Office will be afforded access to these programs, in accordance with the security requirements of each program, in Order to perform the functions assigned to the Information Security Oversight Office under this Order. An agency head may limit access to a special access program to the Director and no more than one other employee of the Information Security Oversight Office; or, for special access programs that are extraordinarily sensitive and vulnerable, to the Director only.

(4) The agency head or principal deputy will review annually each special access program to determine whether it continues to meet the requirements of this Order.

(5) Upon request, an agency will brief the Assistant to the President for National Security Affairs, or his or her designee, on any or all of the agency's special access programs.

c. Within 180 days after the effective date of this Order, each agency head or principal deputy will review all existing special access programs under the agency's jurisdiction. These officials will terminate any special access programs that do not clearly meet the provisions of this Order. Each existing special access program that an agency head or principal deputy validates will be treated as if it were established on the effective date of this Order.

d. Nothing in this Order will supersede any requirement made by or under 10 USC 119.

#### 4.5. - Section 4.5

#### Access by Historical Researchers and Former Presidential Appointees

*a*. The requirement in Section 4.2.a.3 of this Order that access to classified information may be granted only to individuals who have a need-to-know the information may be waived for persons who:

(1) Are engaged in historical research projects.

(2) Previously have occupied policy-making positions to which they were appointed by the President.

b. Waivers under this section may be granted only if the agency head or senior agency official of the originating agency:

(1) Determines in writing that access is consistent with the interest of national security.

(2) Takes appropriate steps to protect classified information from unauthorized disclosure or compromise, and ensures that the information is safeguarded in a manner consistent with this Order.

(3) Limits the access granted to former presidential appointees to items that the person originated, reviewed, signed, or received while serving as a presidential appointee.

#### Part 5 – Implementation and Review

5.1. — Section 5.1 Definitions

#### For purposes of this Order:

a. "Self-inspection" means the internal review and evaluation of individual agency activities and the agency as a whole with respect to the implementation of the program established under this Order and its implementing directives.

b. "Violation" means:

(1) Any knowing, willful or negligent action that could reasonably be expected to result in an unauthorized disclosure of classified information.

(2) Any knowing, willful or negligent action to classify or continue the classification of information contrary to the requirements of this Order or its implementing directives.

(3) Any knowing, willful or negligent action to create or continue a special access program contrary to the requirements of this Order.

c. "Infraction" means any knowing, willful or negligent action contrary to the requirements of this Order or its implementing directives that does not comprise a "violation," as defined above.

#### 5.2. — Section 5.2 Program Direction

*a.* The Director of the Office of Management and Budget, in consultation with the Assistant to the President for National Security Affairs and the co-chairs of the Security Policy Board, will issue such directives as are necessary to implement this Order. These directives will be binding upon the agencies. Directives issued by the Director of the Office of Management and Budget will establish standards for:

- (1) Classification and marking principles.
- (2) Agency security education and training programs.
- (3) Agency self-inspection programs.
- (4) Classification and declassification guides.

b. The Director of the Office of Management and Budget will delegate the implementation and monitorship functions of this program to the Director of the Information Security Oversight Office.

c. The Security Policy Board, established by a Presidential Decision Directive, will make a recommendation to the President through the Assistant to the President for National Security Affairs with respect to the issuance of a Presidential directive on safeguarding classified information. The Presidential directive will pertain to the handling, storage, distribution, transmittal and destruction of and accounting for classified information.

# 5.3. — Section 5.3

# Information Security Oversight Office

*a*. There is established within the Office of Management and Budget an Information Security Oversight Office. The Director of the Office of Management and Budget will appoint the Director of the Information Security Oversight Office, subject to the approval of the President.

b. Under the direction of the Director of the Office of Management and Budget acting in consultation with the Assistant to the President for National Security Affairs, the Director of the Information Security Oversight Office will—

(1) Develop directives for the implementation of this Order.

(2) Oversee agency actions to ensure compliance with this Order and its implementing directives.

(3) Review and approve agency implementing regulations and agency guides for systematic declassification review prior to their issuance by the agency.

(4) Have the authority to conduct on-site reviews of each agency's program established under this Order, and to require of each agency those reports, information, and other cooperation that may be necessary to fulfill its responsibilities. If granting access to specific categories of classified information would pose an exceptional national security risk, the affected agency head or the senior agency official will submit a written justification recommending the denial of access to the Director of the Office of Management and Budget within 60 days of the request for access. Access will be denied pending a prompt decision by the Director of the Office of Management and Budget, who will consult on this decision with the Assistant to the President for National Security Affairs.

(5) Review requests for original classification authority from agencies or officials not granted original classification authority and, if deemed appropriate, recommend presidential approval through the Director of the Office of Management and Budget.

(6) Consider and take action on complaints and suggestions from persons within or outside the government with respect to the administration of the program established under this Order.

(7) Have the authority to prescribe, after consultation with affected agencies, standardization of forms or procedures that will promote the implementation of the program established under this Order.

(8) Report at least annually to the President on the implementation of this Order.

(9) Convene and chair interagency meetings to discuss matters pertaining to the program established by this Order.

### 5.4. — Section 5.4

## Interagency Security Classification Appeals Panel

a. Establishment and Administration.

(1) There is established an Interagency Security Classification Appeals Panel ("Panel"). The Secretaries of State and Defense, the Attorney General, the Director of Central Intelligence, the Archivist of the United States, and the Assistant to the President for National Security Affairs will each appoint a senior level representative to serve as a member of the Panel. The President will select the Chair of the Panel from among the Panel members.

(2) A vacancy on the Panel will be filled as quickly as possible as provided in paragraph 1, above.

(3) The Director of the Information Security Oversight Office will serve as the Executive Secretary. The staff of the Information Security Oversight Office will provide program and administrative support for the Panel.

(4) The members and staff of the Panel will be required to meet eligibility for access standards in Order to fulfill the Panel's functions.

(5) The Panel will meet at the call of the Chair. The Chair will schedule meetings as may be necessary for the Panel to fulfill its functions in a timely manner.

(6) The Information Security Oversight Office will include in its reports to the President a summary of the Panel's activities.

b. Functions. The Panel will-

(1) Decide on appeals by persons who have filed classification challenges under Section 1.9 of this Order.

(2) Approve, deny or amend agency exemptions from automatic declassification as provided in Section 3.4 of this Order.

(3) Decide on appeals by persons or entities who have filed requests for mandatory declassification review under Section 3.6 of this Order.

c. Rules and Procedures. The Panel will issue bylaws, which will be published in the Federal Register no later than

120 days from the effective date of this Order. The bylaws will establish the rules and procedures that the Panel will follow in accepting, considering, and issuing decisions on appeals. The rules and procedures of the Panel will provide that the Panel will consider appeals only on actions in which: (1) the appellant has exhausted his or her administrative remedies within the responsible agency; (2) there is no current action pending on the issue within the federal courts; and (3) the information has not been the subject of review by the federal courts or the Panel within the past two years.

*d*. Agency heads will cooperate fully with the Panel so that it can fulfill its functions in a timely and fully informed manner. An agency head may appeal a decision of the Panel to the President through the Assistant to the President for National Security Affairs. The Panel will report to the President through the Assistant to the President for National Security Affairs any instance in which it believes that an agency head is not cooperating fully with the Panel.

*e*. The Appeals Panel is established for the sole purpose of advising and assisting the President in the discharge of his constitutional and discretionary authority to protect the national security of the United States. Panel decisions are committed to the discretion of the Panel, unless reversed by the President.

#### 5.5. — Section 5.5

#### Information Security Policy Advisory Council

*a.* Establishment. There is established an Information Security Policy Advisory Council ("Council"). The Council will be composed of seven members appointed by the President for staggered terms not to exceed four years, from among persons who have demonstrated interest and expertise in an area related to the subject matter of this Order and are not otherwise employees of the Federal Government. The President will appoint the Council chair from among the members. The Council will comply with the Federal Advisory Committee Act, as amended, 5 USC App. 2.

b. Functions. The Council will-

(1) Advise the President, the Assistant to the President for National Security Affairs, the Director of the Office of Management and Budget, or such other executive branch officials as it deems appropriate, on policies established under this Order or its implementing directives, including recommended changes to those policies.

(2) Provide recommendations to agency heads for specific subject areas for systematic declassification review.

(3) Serve as a forum to discuss policy issues in dispute.

c. Meetings. The Council will meet at least twice each calendar year, and as determined by the Assistant to the President for National Security Affairs or the Director of the Office of Management and Budget.

d. Administration.

(1) Each Council member may be compensated at a rate of pay not to exceed the daily equivalent of the annual rate of basic pay in effect for grade GS-18 of the general schedule under Section 5376 of Title 5, United States Code, for each day during which that member is engaged in the actual performance of the duties of the Council.

(2) While away from their homes or regular place of business in the actual performance of the duties of the Council, members may be allowed travel expenses, including per diem in lieu of subsistence, as authorized by law for persons serving intermittently in the Government service (5 USC 5703.b).

(3) To the extent permitted by law and subject to the availability of funds, the Information Security Oversight Office will provide the Council with administrative services, facilities, staff, and other support services necessary for the performance of its functions.

(4) Notwithstanding any other Executive Order, the functions of the President under the Federal Advisory Committee Act, as amended, that are applicable to the Council, except that of reporting to the Congress, will be performed by the Director of the Information Security Oversight Office in accordance with the guidelines and procedures established by the General Services Administration.

# 5.6. — Section 5.6

# General Responsibilities

Heads of agencies that originate or handle classified information will-

a. Demonstrate personal commitment and commit senior management to the successful implementation of the program established under this Order.

b. Commit necessary resources to the effective implementation of the program established under this Order.

c. Designate a senior agency official to direct and administer the program, whose responsibilities will include: (1) Overseeing the agency's program established under this Order, provided, an agency head may designate a separate official to oversee special access programs authorized under this Order. This official will provide a full accounting of the agency's special access programs at least annually.

(2) Promulgating implementing regulations, which will be published in the Federal Register to the extent that they affect members of the public.

(3) Establishing and maintaining security education and training programs.

(4) Establishing and maintaining an ongoing self-inspection program, which will include the periodic review and assessment of the agency's classified product.

(5) Establishing procedures to prevent unnecessary access to classified information, including procedures that i), require that a need for access to classified information is established before initiating administrative clearance

procedures, and ii), ensure that the number of persons granted access to classified information is limited to the minimum consistent with operational and security requirements and needs.

(6) Developing special contingency plans for the safeguarding of classified information used in or near hostile or potentially hostile areas.

(7) Assuring that the performance contract or other system used to rate civilian or military personnel performance includes the management of classified information as a critical element or item to be evaluated in the rating of: (i) original classification authorities; (ii) security managers or security specialists; and (iii) all other personnel whose duties significantly involve the creation or handling of classified information.

(8) Accounting for the costs associated with the implementation of this Order, which will be reported to the Director of the Information Security Oversight Office for publication.

(9) Assigning in a prompt manner agency personnel to respond to any request, appeal, challenge, complaint or suggestion arising out of this Order that pertains to classified information that originated in a component of the agency that no longer exists and for which there is no clear successor in function.

# 5.7. — Section 5.7 Sanctions

*a*. If the Director of the Information Security Oversight Office finds that a violation of this Order or its implementing directives may have occurred, the Director will make a report to the head of the agency or to the senior agency official so that corrective steps, if appropriate, may be taken.

b. Officers and employees of the United States Government, and its contractors, licensees, certificate holders, and grantees will be subject to appropriate sanctions if they knowingly, willfully, or negligently:

(1) Disclose to unauthorized persons information properly classified under this Order or predecessor Orders.

(2) Classify or continue the classification of information in violation of this Order or any implementing directive.

(3) Create or continue a special access program contrary to the requirements of this Order.

(4) Contravene any other provision of this Order or its implementing directives.

c. Sanctions may include reprimand, suspension without pay, removal, termination of classification authority, loss or denial of access to classified information, or other sanctions in accordance with applicable law and agency regulation.

d. The agency head, senior agency official, or other supervisory official will, at a minimum, promptly remove the classification authority of any individual who demonstrates reckless disregard or a pattern of error in applying the classification standards of this Order.

e. The agency head or senior agency official will-

(1) Take appropriate and prompt corrective action when a violation or infraction under paragraph b, above, occurs.

(2) Notify the Director of the Information Security Oversight Office when a violation under paragraph b.1, 2 or 3, above, occurs.

#### Part 6 – General Provisions

#### 6.1. — Section 6.1 General Provisions

*a*. Nothing in this Order will supersede any requirement made by or under the Atomic Energy Act of 1954, as amended, or the National Security Act of 1947, as amended. "Restricted Data" and "Formerly Restricted Data" will be handled, protected, classified, downgraded, and declassified in conformity with the provisions of the Atomic Energy Act of 1954, as amended, and regulations issued under that Act.

b. The Attorney General, upon request by the head of an agency or the Director of the Information Security Oversight Office, will render an interpretation of this Order with respect to any question arising in the course of its administration.

c. Nothing in this Order limits the protection afforded any information by other provisions of law, including the exemptions to the Freedom of Information Act, the Privacy Act, and the National Security Act of 1947, as amended. This Order is not intended, and should not be construed, to create any right or benefit, substantive or procedural, enforceable at law by a party against the United States, its agencies, its officers, or its employees. The foregoing is in addition to the specific provisos set forth in Sections 1.2.b, 3.2.b and 5.4.e of this Order.

d. Executive Order No. 12356 of April 6, 1982, is revoked as of the effective date of this Order.

### 6.2. — Section 6.2 Effective Date

This Order will become effective 180 days from the date of its issuance.

William J. Clinton

#### THE WHITE HOUSE

April 17, 1995

Executive Order 12972 of September 18, 1995

The President

Amendment to Executive Order No. 12958

By the authority vested in me as President by the Constitution and the laws of the United States of America, and in Order to amend Executive Order No. 12958, it is hereby Ordered that the definition of "agency" in Section 1.1.i of such Order is hereby amended to read as follows: (i)"Agency" means any "Executive agency" as defined in 5 USC 105, any "Military department" as defined in 5 USC 102, and "any other entity within the executive branch that comes into the possession of classified information."

William J. Clinton

THE WHITE HOUSE

September 18, 1995

Executive Order 13142 of November 19, 1999

The President

Amendment to Executive Order 12958

By the authority vested in me as President by the Constitution and the laws of the United States of America, and in order to extend and establish specific dates for the time within which all classified information contained in records more than 25 years old that have been determined to have historical value under title 44, United States Code, should be automatically declassified, and to establish the Information Security Oversight Office within the National Archives and Records Administration, it is hereby ordered that Executive Order 12958 is amended as follows:

Sec. 1. In the first sentence of section 3.4(a) of Executive Order 12958, the words "within five years from the date of this order" are deleted and the words "within six and one half years from the date of this order" are inserted in lieu thereof.

Sec. 2. The following new language is inserted at the end of section 3.4(a): "For records otherwise subject to this paragraph for which a review or assessment conducted by the agency and confirmed by the Information Security Oversight Office has determined that they: (1) contain information that was created by or is under the control of more than one agency, or (2) are within file series containing information that almost invariably pertains to intelligence sources or methods, all classified information in such records shall be automatically declassified, whether or not the records have been reviewed, within 8 years from the date of this order, except as provided in paragraph (b), below. For records that contain information that becomes subject to automatic declassification after the dates otherwise established in this paragraph, all classified information in such records shall be automatically declassified, whether or not the records have been reviewed on December 31 of the year that is 25 years from the origin of the information, except as provided in paragraph (b), below."

Sec. 3. Subsections (a) and (b) of section 5.2 are amended to read as follows:

"(a) The Director of the Information Security Oversight Office, under the direction of the Archivist of the United States and in consultation with the Assistant to the President for National Security Affairs and the co-chairs of the Security Policy Board, shall issue such directives as are necessary to implement this order. These directives shall be binding upon the agencies. Directives issued by the Director of the Information Security Oversight Office shall establish standards for:

- (1) classification and marking principles;
- (2) agency security education and training programs;
- (3) agency self-inspection programs; and
- (4) classification and declassification guides.

(b) The Archivist of the United States shall delegate the implementation and monitorship functions of this program to the Director of the Information Security Oversight Office."

Sec. 4. Subsection (a) and the introductory clause and item (4) of subsection (b) of section 5.3 are amended as follows: (a) Subsection (a) shall read "(a) There is established within the National Archives and Records Administration an Information Security Oversight Office. The Archivist of the United States shall appoint the Director of the Information Security Oversight Office, subject to the approval of the President."

(b) The introductory clause of subsection (b) shall read "Under the direction of the Archivist of the United States, acting in consultation with the Assistant to the President for National Security Affairs, the Director of the Information Security Oversight Office shall:

(c) Item (4) of subsection (b) shall read "(4) have the authority to conduct on-site reviews of each agency's program established under this order, and to require of each agency those reports, information, and other cooperation that may be necessary to fulfill its responsibilities. If granting access to specific categories of classified information would pose an exceptional national security risk, the affected agency head or the senior agency official shall submit a written justification recommending the denial of access to the President through the Assistant to the President for National Security Affairs within 60 days of the request for access. Access shall be denied pending the response,".

#### WILLIAM J. CLINTON

THE WHITE HOUSE,

November 19, 1999.

# Appendix C Special Procedures for Use in Systematic and Mandatory Review of Cryptologic Information

#### C–1. General guidelines

*a.* Cryptologic information uncovered in systematic or mandatory review for declassification of 25-year-old government records is not to be declassified by other than the National Security Agency. The information could concern or reveal the processes, techniques, operations, and scope of Signals Intelligence (SIGINT), which consists of Communications Intelligence (COMINT), Electronic Intelligence (ELINT), and Foreign Instrumentation Signals Intelligence (FISINT), or it could concern the components of Information Systems Security (INFOSEC), which consists of Communications Security (COMSEC) and Computer Security (COMPUSEC), including the communications portion of cover and deception plans. Much cryptologic information is also considered "Foreign Government Information" as defined in paragraph 1.1.d of Executive Order 12958.

b. Recognition of cryptologic information cannot always be an easy task. There are several broad classes of cryptologic information, as follows:

(1) Those that relate to INFOSEC. In documentary form, they provide COMSEC/COMPUSEC guidance or information. Many COMSEC/COMPUSEC documents and materials are accountable under the Communications Security Material Control System. Examples are items bearing Telecommunications Security (TSEC) nomenclature and crypto keying material for use in enciphering communications, other COMSEC/COMPUSEC documentation such as the National Telecommunications and Information Systems Security Committee (NTISSC) or its predecessor organization, COMSEC/COMPUSEC Resources Program documents, COMSEC Equipment Engineering Bulletins, COMSEC Equipment System Descriptions, and COMSEC Technical Bulletins.

(2) Those that relate to SIGINT. These appear as reports in various formats that bear security classifications, frequently followed by five letter code-words, ULTRA (from World War II) for example, and often carry warning caveats such as "This document contains code-word material" and "Utmost secrecy is necessary..." or "Handle Via COMINT Channels Only" "HVCCO" or "CCO." Formats can appear as messages having addresses, "from" and "to" sections, and as summaries with SIGINT content with or without other kinds of intelligence and comment.

(3) Research. Research, development, test, life cycle support, planning, and evaluation reports and information that relates to either COMSEC, COMPUSEC, or SIGINT.

#### C-2. Identification

a. Some commonly used words that help to identify cryptologic documents and materials are: "cipher," "code," "code-word," "Communications Intelligence," "special intelligence," "Communications Security," "Computer Security, " "cryptoanalysis," "crypto," "cryptography," "cryptosystem," "decipher," "decode," "decrypt," "direction finding," "Electronic Intelligence," "Electronic Security" (ELSEC), "encipher," "encode," "encrypt," "Foreign Instrumentation Signals Intelligence," "telemetry," "Information Systems Security," "intercept," and "Traffic Analysis" (TA). *b.* Special procedures apply to the review and declassification of classified cryptologic information. The following will be observed in the review of such information.

(1) INFOSEC (COMSEC/COMPUSEC) Documents and materials.

(2) If records or materials in this category are found in agency or department component files that are not under INFOSEC control, refer them to the senior COMSEC/COMPUSEC authority of the agency or department concerned or return them, by appropriate channels, to the address in paragraph C-2b(5).

(3) If the COMSEC/COMPUSEC information has been incorporated into other documents by the receiving agency, that information must be referred to the National Security Agency/Chief, Central Security Service (NSA/CSS), for review before declassification occurs.

(4) SIGINT (COMINT, ELINT, and FISINT) Information.

(a) If the SIGINT information is contained in a document or record originated by a U.S. Government cryptologic organization and is in the files of a non-cryptologic agency or department, such material will not be declassified. The material can be destroyed unless the holding agency's approved records disposition schedule requires its retention. If the material must be retained, it must be referred to the NSA/CSS for systematic review for declassification when it becomes 25 years old.

(b) If the SIGINT information has been incorporated by the receiving agency into documents it produces, referral of the SIGINT information to the NSA/CSS for review is necessary prior to any declassification action.

(5) COMSEC/COMPUSEC or SIGINT information which requires declassification by the NSA/CSA should be sent to: Director, National Security Agency, Chief, Central Security Service, ATTN: Information Policy Staff (N5P6), Fort George G. Meade, MD 20755–6000.

#### Appendix D Security Controls on Dissemination and Marking of Warning Notices on Intelligence Information

#### D-1. General

Intelligence information will be controlled and marked in accordance with Director of Central Intelligence Directive (DCID) 1/7, "Security Controls on the Dissemination of Intelligence Information", included as figure D–1 of this appendix, and future revisions. Control markings as well as all other policy stipulated in DCID 1/7 apply solely to intelligence information and not to other classified information. Except as specifically stipulated in this appendix, intelligence information will be safeguarded in the same manner as other types of classified information of the same classification level.

a. Department of the Army security education and awareness programs will include the provisions of DCID 1/7.

*b*. The challenge procedures regarding classification of information contained in this regulation will be followed in implementing the challenge procedures concerning intelligence information as stipulated in section 14.2 of DCID 1/7.

c. The procedures regarding the investigation and reporting of the compromise of classified information, contained in this regulation, will be followed for classified intelligence. If the compromise involves the intelligence of another (non–DA) agency, a report will be made to the Director of Central Intelligence, through command channels (DAMI–CH), and the agency that originated the intelligence.

*d.* These classification and control markings, and their authorized abbreviations, are compiled in the Authorized Classification and Control Markings Register, maintained by the Controlled Access Program Coordination Office (CAPCO). The Authorized Classification and Control Markings Register is available, on Intelink–TS at: http://www.cms.ic.gov/capco/ and on Intelink–S at: http://www.cms.cia.sgov.gov/capco/. Examples of proper marking of intelligence community documents can be found at figures D–2 and D–3.

#### D–2. Responsibilities

*a.* The HQDA Deputy Chief of Staff for Intelligence (DCSINT) is the principal DA proponent for intelligence policy matters, to include the security controls on dissemination of intelligence information. The DCSINT is the DA Senior Official of the Intelligence Community (SOIC).

*b*. The DCSINT has, effective by this regulation, delegated to the senior intelligence official at each MACOM, and to the Administrative Assistant to the Secretary of the Army for HQDA staff offices and activities, authority to execute all the provisions of DCID 1/7 that refer to the "designee" of the SOIC. Those designees may further delegate this authority, in writing, to the senior intelligence official at major subordinate commands. This authority includes the release of intelligence information to contractors and consultants, subject to the conditions of this regulation. This includes those situations requiring access in response to a formal solicitation, such as a request for proposal, invitation to bid, or similar solicitation request.

c. The DCSINT has, effective by this regulation, delegated to the Director, Defense Security Service (DSS), the authority to release intelligence information to appropriately cleared U.S. contractors supporting an Army mission, and

having a demonstrated need to know, subject to the controls of DCID 1/7, to include those in section 6. The Director, DSS can further delegate this authority to subordinate DSS officials.

*d*. Army personnel, to include those who are not original classification authorities, who generate classified intelligence products or incorporate intelligence information in primarily non-intelligence products, are authorized and will apply, effective by DCID 1/7, control markings specified in DCID 1/7, section 9.

#### D-3. General Guidance

*a*. Under this classification marking system, for documents containing intelligence information, the following rules apply to the classification line, used at the top and bottom of each page of a document, and portion marking:

(1) The full classification line must be conspicuously placed at both the top and bottom of each page of a classified document, or an unclassified document which either has other control markings or is being transmitted electronically. Note: It is optional to mark UNCLASSIFIED on the top and bottom of documents which are not classified and bear no other control markings, such as FOUO or PROPIN.

(2) The classification line and portion markings always use uppercase letters.

(3) The specified U.S. or non–U.S. classification portion of the classification line at the top and bottom of page must be spelled out in full and in English, and may not be abbreviated. Any other information included on the classification line may be spelled out or abbreviated unless otherwise directed by component policy. Note: All classified and unclassified information which bears any control markings, must bear either a U.S. or non–U.S. classification marking, but not both, at the top and bottom of each page.

(4) If the classification is TOP SECRET, SECRET, or CONFIDENTIAL, there must be a declassification date or exemption entry (for example, TOP SECRET//17 December 2001 or SECRET//X1). The full class/declass block, required by EO 12958, must still appear on the first page of a classified document.

(5) The pertinent caveats from each category in the Authorized Classification and Control Markings Register form a field in the classification line or portion marking. Fields are always separated by a double right slash (for example, (U// FOUO)).

(6) Only fields with entries applicable to the document are represented in the classification line. No slashes, hyphens, or spaces, are used to hold the place of fields with no entries applicable to the document.

(7) Only one entry may appear in the U.S. classification, non-U.S. classification, and declassification date fields.

(8) All information may only bear one classification marking, either U.S. or non-U.S..

(9) Multiple entries may be chosen from the SCI Control System, Dissemination Control, and Non-Intelligence Community Marking categories, if the entries are applicable to the document. If multiple entries are used within the field, they are listed in the order in which they appear in the Authorized Classification and Control Markings Register.

(a) For multiple SCI Control System entries, use a single right slash as the separator between the individual SCI Control System entries.

(b) Use a comma with no space interjected as the separator between multiple Dissemination Control or multiple Non–Intelligence Community Marking entries.

(10) Portion markings are to be included at the beginning of the respective portion and enclosed in parentheses. Standard marking separators (slashes, hyphens, commas, etc.) are to be used where necessary.

(11) The order for markings must follow the order in which the markings appear in the Authorized Classification and Control Markings Register.

b. Some of the dissemination control and non-intelligence community markings are restricted to use by certain agencies. Non-U.S. classification markings are restricted to the respective countries or international organizations. In addition, some of the markings in the register are prohibited from use with other markings. Further information is provided in the "Relationship to Other Markings" column of the Authorized Classification and Control Markings Register.

Director of Central Intelligence Directive (DCID) 1/7, Security Controls on the

**Dissemination of Intelligence Information** 

# DIRECTOR OF CENTRAL INTELLIGENCE DIRECTIVE 1/71

Security Controls on the Dissemination of Intelligence Information

(Effective 30 June 1998)

Introduction

Pursuant to the provisions of the National Security Act of 1947, as amended, Executive Order 12333, Executive Order 12958 and implementing directives thereto, policies, controls, and procedures for the dissemination and use of intelligence information and related materials are herewith established in this Director of Central Intelligence Directive (Directive or DCID). Nothing in this policy is intended to amend, modify, or derogate the authorities of the DCI contained in Statute or Executive Order.

Figure D-1. Director of Central Intelligence Directive (DCID) 1/7, Security Controls on the Dissemination of Intelligence Information

# 1.0. Policy

# 1.1.

It is the policy of the DCI that intelligence be produced in a way that balances the need for maximum utility of the information to the intended recipient with protection of intelligence sources and methods. The controls and procedures established by this directive should be applied uniformly in the dissemination and use of intelligence originated by all Intelligence Community components in accordance with the following principles:

# 1.1.1.

Originators of classified intelligence information should write for the consumer. This policy is intended to provide for the optimum dissemination of timely, tailored intelligence to consumers in a form that allows use of the information to support all need to know customers.

# 1.1.2.

The originator of intelligence is responsible for determining the appropriate level of protection prescribed by classification and dissemination policy. Originators shall take a risk management approach when preparing information for dissemination.

# 2.0. Purpose

# 2.1.

This directive establishes policies, controls, and procedures for the dissemination and use of intelligence information to ensure that, while facilitating its interchange for intelligence purposes, it will be adequately protected. This directive implements and amplifies applicable portions of the directives of the Information Security Oversight Office issued pursuant to Executive Order (EO) 12958 and directives of the Security Policy Board issued pursuant to EO 12958 and PDD–29.

# 2.2.

Additionally, this directive sets forth policies and procedures governing the release of intelligence to contractors and consultants, foreign governments, international organizations or coalition partners consisting of sovereign states, and to foreign nationals and immigrant aliens, including those employed by the US Government.

# 2.3.

Executive Order 12958 provides for the establishment of Special Access Programs, including Sensitive Compartmented Information. DCID 3/29 provides procedures for the establishment and review of Special Access Programs pertaining to intelligence activities and restricted collateral information. Intelligence Community components may establish and maintain dissemination controls on such information as approved under the policies and procedures contained in DCID 3/29, this DCID, and implementing guidance.

# 3.0. Definitions

# 3.1.

"Caveated" information is information subject to one of the authorized control markings under section 9.

# 3.2.

Intelligence Community (and agencies within the Intelligence Community) refers to the United States Government agencies and organizations and activities identified in section 3 of the National Security Act of 1947, as amended, 50 USC 401a(4), and section 3.4(f) (1 through 6) of Executive Order 12333.

# 3.3.

Intelligence information and related materials (hereinafter referred to as "Intelligence") include the following information, whether written or in any other medium, classified pursuant to EO 12958 or any predecessor or successor Executive Order:

# 3.3.1.

Foreign intelligence and counterintelligence defined in the National Security Act of 1947, as amended, and in Executive Order 12333;

# 3.3.2.

Information describing US foreign intelligence and counterintelligence activities, sources, methods, equipment, or methodology used for the acquisition, processing, or exploitation of such intelligence; foreign military hardware obtained through intelligence activities for exploitation and the results of the exploitation; and any other data resulting from US intelligence collection efforts; and,

# 3.3.3.

Information on Intelligence Community protective security programs (e.g., personnel, physical, technical, and information security).

# 3.4.

"Need-to-know" is the determination by an authorized holder of classified information that a prospective recipient requires access to specific classified information in order to perform or assist in a lawful and authorized governmental function. Such persons shall possess an appropriate security clearance and access approval granted pursuant to Executive Order 12968, Access to Classified Information.

# 3.5.

Senior Official of the Intelligence Community (SOIC) is the head of an agency, office, bureau, or other intelligence element as identified in section 3 of the National Security Act of 1947, as amended, 50 USC 401a(4), and section 3.4(f) (1 through 6) of Executive Order 12333.

# 3.6.

A "tear line" is the place on an intelligence report (usually denoted by a series of dashes) at which the sanitized version of a more highly classified and/or controlled report begins. The sanitized information below the tear line should contain the substance of the information above the tear line, but without identifying the sensitive sources and methods. This will permit wider dissemination, in accordance with the "need to know" principle and foreign disclosure guidelines, of the information below the tear line.

## 4.0. General Applicability

## 4.1.

In support of the policy statement in section 1.0, classifiers of intelligence information shall take a risk management approach when preparing information for dissemination. In the interest of the widest possible dissemination of information to consumers with a "need to know", classifiers shall carefully consider the needs of all appropriate intelligence consumers regarding sources and methods information or sensitive analytic comments and use control markings only when necessary and in accordance with this directive, using tear lines and other formats to meet consumer needs for intelligence.

#### 4.2.

In carrying out this policy, intelligence producers shall prepare their reports and products at the lowest classification level commensurate with expected damage that could be caused by unauthorized disclosure. When necessary, the material should be prepared in other formats (e.g. tear line form) to permit broader dissemination or release of information.

#### 4.3.

All material shall be portion marked to allow ready identification of information that cannot be broadly disseminated or released, except for material for which a waiver has been obtained under EO 12958.

# 4.4.

The substance of this directive shall be promulgated by each Intelligence Community component, and appropriate procedures permitting prompt interagency consultation established.

# 5.0. Use By and Dissemination Among Executive Branch Departments/Agencies of the US Government

# 5.1.

Executive Order 12958 provides that classified information originating in one US department or agency shall not be disseminated beyond any recipient agency without the consent of the originating agency. However, to facilitate use and dissemination of intelligence within and among Intelligence Community components and to provide for the timely flow of intelligence to consumers, the following controlled relief to the "third agency rule" is hereby established:
## 5.1.1.

Each Intelligence Community component consents to the use of its classified intelligence in classified intelligence products of other Intelligence Community components, including its contractors under section 6, and to the dissemination of those products within executive branch departments/agencies of the US Government, except as specifically restricted by controls defined in this directive or other DCI guidance.

## 5.1.2.

As provided in 5.1.1, classified intelligence that bears no restrictive control markings may be given secondary US dissemination in classified channels to any U.S. Executive Branch department/agency not on original distribution if: (a) the intelligence has first been sanitized by the removal of all references and inferences to intelligence sources and methods and the identity of the producing agency; or (b) if the product is not so sanitized, the consent of the originator has been obtained. If there is any doubt concerning a reference or inference to intelligence sources and methods, relevant intelligence documents should not be given secondary dissemination until the recipient has consulted with the originator.

## 5.1.3.

Any component disseminating intelligence beyond the Intelligence Community assumes responsibility for ensuring that recipient organizations agree to observe the need-to-know principle and the restrictions prescribed by this directive, and to maintain adequate safeguards.

## 6.0. Policy and Procedures Governing the Release of Intelligence to Contractors and Consultants

## 6.1.

SOICs, or their designees, may release intelligence to appropriately cleared or access-approved US contractors and consultants (hereinafter "contractor") having a demonstrated "need-to-know" without referral to the originating agency prior to release provided that:

## 6.1.1.

At the initiation of the contract, the SOIC or his/her designee specifies and certifies in writing that disclosure of the specified information does not create an unfair competitive advantage for the contractor or a conflict of interest with the contractor's obligation to protect the information. If, during the course of the contract, the contractor's requirements for information changes to require new or significantly different information, the SOIC or his/her designee shall make a new specification and certification. In cases where the designated official cannot or does not resolve the issue of unfair competitive advantage or conflict of interest, consent of the originator is required;

## 6.1.2.

Release is made only to contractors certified by the SOIC (or designee) of the sponsoring organization as performing classified services in support of a national security mission;

## 6.1.3.

The contractor has an approved safeguarding capability if retention of the intelligence is required;

## 6.1.4.

Contractors are not authorized to disclose further or release intelligence to any of their components or employees or to another contractor (including subcontractors) without the prior written notification and approval of the SOIC or his/her designee unless such disclosure or release is authorized in writing at the initiation of the contract as an operational requirement;

## 6.1.5.

Intelligence released to contractors, all reproductions thereof, and all other material generated based on, or incorporating data therefrom (including authorized reproductions), remain the property of the US Government. Final disposition of intelligence information shall be governed by the sponsoring agency;

## 6.1.6.

National Intelligence Estimates (NIEs), Special National Intelligence Estimates (SNIEs), and Interagency Intelligence Memoranda may be released to appropriately cleared contractors possessing an appropriate level facility clearance and need-to-know, except as regulated by provisions concerning proprietary information as defined in sections 6.1.7 and 9.3, below;

## 6.1.7.

Except as provided in section 6.3 below, intelligence that bears the control marking "CAUTION-PROPRIETARY

INFORMATION INVOLVED" (abbreviated "PROPIN" or "PR") may not be released to contractors, unless prior permission has been obtained from the originator and those providing the intelligence to the originator. Intelligence that bears the control marking, "DISSEMINATION AND EXTRACTION OF INFORMATION CONTROLLED BY ORIGINATOR" (abbreviated "ORCON") may only be released to contractors within Government facilities. These control markings are further described under sections 9.2 and 9.3, below; and

## 6.1.8.

Authorized release to foreign nationals or foreign contractors is undertaken through established channels in accordance with sections 7 and 8, and DCID 5/6, Intelligence Disclosure Policy, and the National Policy and Procedures for the Disclosure of Classified Military Information to Foreign Governments and International Organizations (abbreviated title: National Disclosure Policy 1 or NDP 1) to the extent consistent with DCIDs and other DCI guidance.

## 6.2. Policies and Procedures for Contractors Inside Government Owned or Controlled Facilities

## 6.2.1.

Contractors who perform duties inside a Government owned or controlled facility will follow the procedures and policies of that sponsoring Intelligence Community member in accordance with section 6.1 of this directive.

## 6.3. Policies and Procedures for Contractors Outside Government Owned or Controlled Facilities

## 6.3.1.

Contractors who perform duties outside of Government owned or controlled facilities will adhere to the following additional policies and procedures:

## 6.3.1.1.

The SOIC of the sponsoring agency, or his/her designee, is responsible for ensuring that releases to contractors of intelligence marked ORCON and/or PROPIN are made only with the consent of the originating agency pursuant to this directive and through established channels; (See sections 9.2 and 9.3);

## 6.3.1.2.

The sponsoring agency shall maintain a record of material released;

## 6.3.1.3.

Contractors shall establish procedures to control all intelligence received, produced, and held by them in accordance with the provisions of the National Industrial Security Program Operating Manual. This will not impose internal receipt and document accountability requirements for internal traceability and audit purposes;

## 6.3.1.4.

All reproductions and extractions of intelligence shall be classified, marked, and controlled in the same manner as the original(s);

## 6.3.1.5.

Sensitive Compartmented Information released to contractors shall be controlled pursuant to the provisions of DCID 1/ 19, Security Policy for Sensitive Compartmented Information (SCI); and,

## 6.3.1.6.

Sponsoring agencies shall delete any reference to the Central Intelligence Agency (CIA), the phrase "Directorate of Operations" and any of its components, the place acquired, the field number, the source description, and field dissemination from all CIA Directorate of Operations reports passed to contractors, unless prior approval to do otherwise is obtained from CIA.

## 7.0. Release to Foreign Governments, International Organizations, and Coalition Partners

## 7.1.

It is the policy of the DCI that intelligence may be shared with foreign governments, and international organizations or coalition partners consisting of sovereign states to the extent such sharing promotes the interests of the United States, is consistent with US law, does not pose unreasonable risk to US foreign policy or national defense, and is limited to a specific purpose and normally of limited duration. The release of intelligence to such entities is subject to this directive, DCID 5/6, Intelligence Disclosure Policy, and NDP 1 to the extent consistent with DCIDs and other DCI guidance.

## 7.1.1.

Intelligence Community elements shall restrict the information subject to control markings to the minimum necessary.

If it is not possible to prepare the entire report at the collateral, uncaveated level, IC elements shall organize their intelligence reports and products to identify clearly information not authorized for release to foreign entities.

## 7.2.

Intelligence information that bears no specific control marking may be released to foreign governments, international organizations, or coalition partners provided that:

## 7.2.1.

A positive foreign disclosure decision is made by a Designated Intelligence Disclosure Official in accordance with procedures in DCID 5/6;

## 7.2.2.

No reference is made to the originating agency or to the source of the documents on which the released product is based; and,

## 7.2.3.

The source or manner of acquisition of the intelligence (including analytic judgments or techniques), and/or the location where the intelligence was collected (if relevant to protect sources and methods) is not revealed and cannot be deduced in any manner.

## 7.3.

RESTRICTED DATA and FORMERLY RESTRICTED DATA may only be released to foreign governments pursuant to an agreement for cooperation as required by sections 123 and 144 of Public Law 585, Atomic Energy Act of 1954, as amended.

## 8.0. Dissemination to Non-Governmental Foreign Nationals or Foreign Contractors

## 8.1.

It is the policy of the DCI that no classified intelligence will be shared with foreign nationals, foreign contractors, or international organizations not consisting of sovereign states, except in accordance with the provisions of this section.

## 8.2.

Intelligence, even though it bears no restrictive control markings, will not be released in any form to foreign nationals or immigrant aliens (including those employed by, used by, or integrated into the US Government) without the permission of the originator. In such cases where permission of the originator has been granted, the release must be in accordance with DCID 5/6, and the NDP 1 to the extent consistent with DCIDs and other DCI guidance.

## 8.3.

Release of intelligence to a foreign contractor or company under contract to the US Government must be through the foreign government of the country which the contractor is representing, unless otherwise directed in government-to-government agreements or there is an appropriate US channel for release of the information. Provisions concerning release to foreign governments is contained in section 7.0, above.

## 9.0. Authorized Control Markings

#### 9.1.

DCI policy is that the authorized control markings for intelligence information in this section shall be individually assigned as prescribed by an Original Classification Authority (OCA) or by officials designated by a SOIC and used in conjunction with security classifications and other markings specified by Executive Order 12958 and its implementing directive(s). Unless originator consent is obtained, these markings shall be carried forward to any new format or medium in which the same information is incorporated.

#### 9.1.1.

To the maximum extent possible, information assigned an authorized control marking shall not be combined with uncaveated information in such a way as to render the uncaveated information subject to the control marking. To fulfill the requirements of paragraph 9.6.1 below, SOICs shall establish procedures in implementing directives to expedite further dissemination of essential intelligence. Whenever possible, caveated intelligence information reports should include the identity and contact instructions of the organization authorized to approve further dissemination on a case–by–case basis.

## 9.2. "DISSEMINATION AND EXTRACTION OF INFORMATION CONTROLLED BY ORIGINATOR" (ORCON)

## 9.2.1.

This marking (ORCON or abbreviated OC) may be used only on classified intelligence that clearly identifies or would reasonably permit ready identification of intelligence sources or methods that are particularly susceptible to countermeasures that would nullify or measurably reduce their effectiveness. It is used to enable the originator to maintain continuing knowledge and supervision of distribution of the intelligence beyond its original dissemination. This control marking may not be used when access to the intelligence information will reasonably be protected by use of its classification markings, i.e., CONFIDENTIAL, SECRET or TOP SECRET, or by use of any other control markings specified herein or in other DCIDs. Requests for further dissemination of intelligence bearing this marking shall be reviewed in a timely manner.

## 9.2.2.

Information bearing this marking may be disseminated within the headquarters and specified subordinate elements of recipient organizations, including their contractors within Government facilities. This information may also be incorporated in whole or in part into other briefings or products, provided the briefing or intelligence product is presented or distributed only to original recipients of the information. Dissemination beyond headquarters and specified subordinate elements or to agencies other than the original recipients requires advance permission from the originator.

## 9.2.3.

Information bearing this marking must not be used in taking investigative or legal action without the advance permission of the originator.

## 9.2.4.

As ORCON is the most restrictive marking herein, agencies that originate intelligence will follow the procedures established in the classified DCID 1/7 Supplement, "Guidelines for Use of ORCON Caveat."

## 9.3.

"CAUTION-PROPRIETARY INFORMATION INVOLVED" (PROPIN). This marking is used, with or without a security classification, to identify information provided by a commercial firm or private source under an express or implied understanding that the information will be protected as a proprietary trade secret or proprietary data believed to have actual or potential value. This marking may be used on government proprietary information only when the government proprietary information can provide a contractor(s) an unfair advantage, such as US Government budget or financial information. Information bearing this marking shall not be disseminated outside the Federal Government in any form without the express permission of the originator of the intelligence and provider of the proprietary information. This marking shall not contractors irrespective of their status to, or within, the US Government without the authorization of the originator of the intelligence and provider of the information. This marking shall be abbreviated "PROPIN" or "PR."

## 9.4.

"NOT RELEASABLE TO FOREIGN NATIONALS" – NOFORN (NF). This marking is used to identify intelligence which an originator has determined falls under the criteria of DCID 5/6, "Intelligence Which May Not Be Disclosed or Released," and may not be provided in any form to foreign governments, international organizations, coalition partners, foreign nationals, or immigrant aliens without originator approval.

## 9.5.

"AUTHORIZED FOR RELEASE TO..(name of country(ies)/international organization)" (REL TO). This control marking is used when a limited exception to the marking requirements in section 9.4 may be authorized to release the information beyond US recipients. This marking is authorized only when the originator has an intelligence sharing arrangement or relationship with a foreign government approved in accordance with DCI policies and procedures that permits the release of the specific intelligence information to that foreign government, but to no other in any form without originator consent.

## 9.6. Further Dissemination of Intelligence with Authorized Control Marking(s)

## 9.6.1.

This directive does not restrict an authorized recipient of intelligence at any level from directly contacting the originator of the intelligence to ask for relief from a specific control marking(s) in order to further disseminate intelligence material to additional users for which the authorized original recipient believes there is a valid need-to-know. Authorized recipients are encouraged to seek such further dissemination through normal liaison

channels for release to US Government agencies or contractors and through foreign disclosure channels for foreign release, on a case-by-case basis, in order to expedite further dissemination of essential intelligence.

## 9.6.2.

Authorized recipients may obtain information regarding points of contact at agencies that originate intelligence from their local dissemination authorities or from instructions issued periodically by these intelligence producers. Intelligence products often also carry a point of contact name/office and telephone number responsible for the product. If no other information is available, authorized recipients are encouraged to contact the producing agency of the document to identify the official or office authorized to provide relief from authorized control marking(s).

## 9.6.3.

If there are any questions about whom to contact for guidance, recipients are also encouraged to contact the Director of Central Intelligence (DCI) representative at the Commander–in–Chief (CINC) Headquarters, overseas mission, trade delegation, or treaty negotiating team under which they operate.

## 9.7.

A SOIC may authorize the use of additional security control markings for Sensitive Compartmented Information (SCI), Special Access Program (SAP) information, restricted collateral information, or other classified intelligence information, consistent with policies and procedures contained in DCID 3/29 and this directive. A uniform list of security control markings authorized for dissemination of classified information by components of the Intelligence Community, and the authorized abbreviated forms of such markings, shall be compiled in the central register maintained pursuant to DCID 3/29. The forms of the markings and abbreviations listed in this register shall be the only forms of those markings used for dissemination of classified information by components of the Intelligence Community, unless an exception is specifically authorized by a SOIC.

## 10.0. Dissemination and Disclosure Under Emergency Conditions

## 10.1.

Certain emergency situations4 that involve an imminent threat to life or mission warrant dissemination of intelligence to organizations and individuals not routinely included in such dissemination. When the National Command Authority (NCA) directs that an emergency situation exists, SOICs will ensure that intelligence support provided to the ongoing operations conforms with this directive, DCID 5/6, and NDP 1 to the maximum extent practical and consistent with the mission.

#### 10.1.1.

Dissemination of intelligence under this provision is authorized only if: (a) an authority designated by the military commander or civilian official determines that adherence to this DCID reasonably is expected to preclude timely dissemination to protect life or mission; (b) disseminations are for limited duration and narrowly limited to persons or entities that need the information within 72 hours to satisfy an imminent emergency need; and (c) there is insufficient time to obtain approval through normal intelligence disclosure channels.

#### 10.1.2.

The disclosing authority will report the dissemination through normal disclosure channels within 24 hours of the dissemination, or at the earliest opportunity thereafter as the emergency permits. For purposes of this provision, planning for contingency activities or operations not expected to occur within 72 hours does not constitute "imminent" need that warrants exercise of the emergency waiver to bypass the requirements of this DCID.

#### 10.1.3.

Military commanders and/or responsible civilian officials will ensure that written guidelines for emergency dissemination contain provisions for safeguarding disseminated intelligence and notifying producers of disclosures of information necessary to meet mission requirements.

#### 10.1.4.

The NCA, and/or major commands or responsible civilian officials will immediately advise intelligence producers when the emergency situation ends.

#### 11.0. Procedures Governing Use of Authorized Control Markings

## 11.1.

Any recipient desiring to disseminate intelligence in a manner contrary to the control markings established by this directive must obtain the advance permission of the agency that originated the intelligence. Such permission applies only to the specific purpose agreed to by the originator and does not automatically apply to all recipients. Producers of

intelligence will ensure that prompt consideration is given to recipients' requests with particular attention to reviewing and editing, if necessary, sanitized or paraphrased versions to derive a text suitable for release subject to lesser or no control marking(s).

## 11.2.

The control markings authorized above shall be shown on the title page, front cover, and other applicable pages of documents; incorporated in the text of electrical communications; shown on graphics; and associated (in full or abbreviated form) with data stored or processed in automated information systems. The control markings also shall be indicated by parenthetical use of the marking abbreviations at the beginning or end of the appropriate portions in accordance with EO 12958.

## 12.0. Obsolete Restrictions and Control Markings

## 12.1.

The following control markings are obsolete and will not be used in accordance with the following guidelines:

## 12.1.1.

WNINTEL and NOCONTRACT. The control markings, Warning Notice – Intelligence Sources or Methods Involved (WNINTEL), and NOT RELEASABLE TO CONTRACTORS/CONSULTANTS (abbreviated NOCONTRACT or NC) were rendered obsolete effective 12 April 1995. No permission of the originator is required to release, in accordance with this directive, material marked WNINTEL. Holders of documents prior to 12 April 1995 bearing the NOCONTRACT marking should apply the policies and procedures contained in section 6.1 for possible release of such documents.

## 12.1.2.

Remarking of material bearing the WNINTEL, or NOCONTRACT, control marking is not required; however, holders of material bearing these markings may line through or otherwise remove the marking(s) from documents or other material.

## 12.1.3.

Other obsolete markings include: WARNING NOTICE-INTELLIGENCE SOURCES OR METHODS INVOLVED, WARNING NOTICE-SENSITIVE SOURCES AND METHODS INVOLVED, WARNING NOTICE-INTELLIGENCE SOURCES AND METHODS INVOLVED, WARNING NOTICE-SENSITIVE INTELLIGENCE SOURCES AND METHODS INVOLVED, CONTROLLED DISSEM, NSC PARTICIPATING AGENCIES ONLY, INTEL COMPONENTS ONLY, LIMITED, CONTINUED CONTROL, NO DISSEM ABROAD, BACKGROUND USE ONLY, USIB ONLY, NFIB ONLY.

## 12.2.

Questions with respect to current applications of all control markings authorized by earlier directives on the dissemination and control of intelligence and used on documents issued prior to the effective date of this directive should be referred to the agency or department originating the intelligence so marked.

## 13.0. Reporting Unauthorized Disclosures

## 13.1.

Violations of the foregoing restrictions and control markings that result in unauthorized disclosure by one agency of the intelligence of another shall be reported to the Director of Central Intelligence through appropriate Intelligence Community channels.

## 14.0. Responsibilities of SOICs

#### 14.1.

SOICs shall be responsible for the implementation of internal controls and shall conduct training to ensure that the dissemination and release policies contained in this directive and the limitations on the use of control markings are followed. SOICs shall assure that agency personnel are accountable for the proper marking of classified information under this directive and section 5.6 of EO 12958.

## 14.2.

SOICs shall establish challenge procedures by which US consumers may register complaints about the misuse of control markings or the lack of use of tear line reporting or portion marking. Information concerning such challenges shall be provided to the Security Policy Board staff upon request or for the annual review.

## 15.0. Annual Report on the Use of Control Markings

## 15.1.

The Security Policy Board staff shall report to the DCI and Deputy Secretary of Defense on Intelligence Community compliance with this directive, including recommendations for further policies in this area. The report will include an in-depth evaluation of the use of control markings in intelligence reporting/production, including consumer evaluations and producer perspectives on implementation of the directive. The report shall also include information and statistics on challenges formally lodged pursuant to agency procedures under section 1.9 of Executive Order 12958 within and among intelligence agencies on the use of control markings, including their adjudication and the number of times the authority in section 10 was used and the documents provided. In order to inform the Security Policy Board staff of substantive detail in these areas for purposes of this review, Intelligence Community elements shall respond to requests for information from the Security Policy Board staff. Intelligence Community elements may build this program into their Self–Inspection programs under EO 12958. The Security Policy Board staff shall also obtain pertinent information on this subject from intelligence consumers as required.

## 15.2.

The report required by this section shall be conducted annually, unless otherwise directed by the DCI. The Staff Director, Security Policy Board shall establish the schedule for the report.

## 16.0. Interpretation

## 16.1.

Questions concerning the implementation of this policy and these procedures shall be referred to the Community Management Staff.

\_\_\_\_\_/S/\_\_\_\_ 30 June 1998 Director of Central Intelligence Date

#### ENDNOTES:

<sup>1</sup>This directive supersedes DCID 1/7, dated 12 April 1995

<sup>2</sup>Recipients will apprise originating agencies as to which components comprise the headquarters element and identify subordinate elements that may be included as direct recipients of intelligence information.

<sup>3</sup>This provision is a requirement of the Trade Secrets Act, as amended (18 USC 1905). The consent of the originator is required to permit release of material marked CAUTION–PROPRIETARY INFORMATION INVOLVED, PROPIN, or PR to other than federal government employees.

<sup>4</sup>For the purposes of implementing this portion of the DCID, "emergency situation" is defined as one of the following:

a) Declared Joint Chiefs of Staff (JCS) alert condition of defense emergency, air defense emergency or Defense Readiness Condition (DEFCON) 3;

b) Hostile action(s) being initiated against the United States or combined US/coalition/friendly forces;

c) US persons or facilities being immediately threatened by hostile forces;

d) US or combined US/coalition/friendly forces planning for or being deployed to protect or rescue US persons, or US/ coalition/friendly forces;

e) US civilian operations in response to US or international disasters/catastrophes of sufficient severity to warrant Presidential declared disaster assistance/relief.

Sample of Marking an Originally Classified Intelligence Community Document

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY SECRET//X1

Subject: Sample of Marking an Originally Classified Intelligence Community Document (U)

1. (S) This page is UNCLASSIFIED and is marked SECRET for training purposes only. IMPORTANT: You must be an Original Classification Authority to originally classify a document. In most cases, documents are classified based on a Security Classification Guide or guidance (that was approved by an OCA) or upon information taken out of a source document (or both). This sample memo only pertains to those relatively few cases in which an OCA is generating a document that the OCA is originally classifying.

2. (C) The OCA must be identified (see paragraph 4-8a), the reason that the information is classified must be stated (see paragraph 4-9), and the OCA must indicate a date, event, or exemption category for declassification (see paragraph 4-9a). If there is more than one reason or exemption, then all must be listed. In this example, the OCA has selected reason 1.5(c) (intelligence activities, intelligence sources or methods, or cryptology). Instead of a date or event for declassification within 10 years, the OCA has authorized exemption category X1 (information that would reveal an activity, intelligence source, or methods), or a cryptologic system or activity that must be protected beyond 10 years).

I.M. Trying

Director, Security Awareness

Classified by: LTG A. SECRET, Chief of Intelligence Programs, Army Security Is Important Command Reason: 1.5(c)

Declassify on: X1

Date of source: 26 March 1999 SECRET//X1 CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY Figure D-2. Sample of Marking an Originally Classified Intelligence Community Document Sample of Marking Foreign Government Intelligence Information CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY //FGI SECRET//X5

SUBJECT: Sample of Marking Foreign Government Intelligence (FGI) Information (U)

1. (U) The U.S. Government affords protection to information provided by foreign governments. Care must be taken to identify the source of the information.

2. (//FGI C) Mark the portions that contain the foreign government information to indicate the country of origin and the classification level. Substitute the words "Foreign Government Information" or "FGI" where the identity of the specific government must be concealed. The identity of the concealed foreign source in this example must be maintained with the record copy and adequately protected.

3. (//FGI S) This paragraph contains information marked "SECRET" by the government of a foreign country. The "Derived From" citation should cite the title and date of the document provided. Declassification date, event, or exemption category is carried forward, if known.

Derived From: Foreign Government Source Document, (fill-in date)

Declassify On: X5, FGI

//FGI SECRET//X5

CLASSIFICATION MARKINGS FOR TRAINING PURPOSES ONLY

Figure D-3. Sample of Marking Foreign Government Intelligence Information

## Appendix E Security Procedures for Documents Created for and on Automated Information Systems and Internet Web–based Display

#### Section I

Documents Produced by AIS Equipment and Electronic Mail

#### E–1. Introduction

*a*. As the level of technological advances increases more and more, it is beginning to be commonplace for units to create and communicate via automated information systems. Reports, training, inventories, and other day-to-day operations are created, transmitted and displayed with computers. Many units and agencies now create and maintain Internet websites, not only on the open unclassified systems, but also on all levels of classified Local Area Networks (LANs) and Wide Area Networks (WANs). Until now, there have been no definitive instructions regarding the handling of these types of materials and systems. Particularly, this is true when it comes to classification marking. The creation of these "digital documents" is probably the only time that the rules governing who can be the originating authority do not apply. Hopefully by explaining the procedures for handling these materials it will eliminate the possibility of accidental security violations.

b. The World Wide Web (WWW) provides the Department of the Army with a powerful tool to convey information quickly and efficiently on a broad range of topics relating to its activities, objectives, policies, programs, and personnel. The global reach of the web makes this information easily accessible to the men and women of the Armed Forces, their families, the American public, and the international audience. Recently, however, it has become apparent that some information on publicly accessible websites provides too much detail on DOD capabilities, infrastructure, personnel, and operational procedures, past, present, and future. While it may be true that the majority of this information is wholly, and truly, unclassified, when combined with information from other sources and/or sites, they can become sensitive and even classified. They also may increase the vulnerability of DOD systems and potentially be used to threaten or harass DOD personnel and/or their families. All DA commanders who establish publicly accessible websites are responsible for ensuring that the information published on their sites do not compromise national security or place DA personnel at risk. The commander's responsibility extends beyond general public affairs considerations regarding the release of information into the realm of operational security and force protection. Commanders must apply comprehensive risk management procedures to ensure that the considerable mission benefits gained by using the web are carefully balanced against the potential security and privacy risks created by having aggregated DOD information more readily accessible to a worldwide audience than ever before.

#### E-2. Printed Documents Produced by AIS Equipment

There are no special provisions for documents produced by Automated Information Systems (AIS) which function as word processing systems. Documents produced on an AIS will be marked like any other documents. For other AIS–generated documents, special exceptions may apply where the application of the marking requirements are not feasible. These exceptions are:

a. Classification markings on interior pages of fan-folded printouts. These markings will be applied and can be done so on the AIS equipment, however they may not meet the standard requirement of being conspicuous.

*b*. Special warning notices, identification of classification sources, and declassification (and downgrading, where applicable) instructions will either be marked on the cover of the document (or on the first page if there is no cover) or will be placed on a separate notice attached to the front of the document.

c. Pages or other portions of AIS printouts removed for separate use or maintenance will be marked in the standard manner as individual documents.

#### E-3. Classified electronic mail

With advent of electronic mail (E-mail or e-mail), communication between units and personnel has increased exponentially. However, with this increase in e-mail, there is also a decrease in security awareness when it comes to proper classification markings on these messages. Department of the Army personnel take for granted that since they are communicating over a classified LAN they do not have to use proper markings. This is incorrect. Proper page and portion markings will be used on all correspondence just as if it were a typed or hand-written paper. If the entire message, on a classified LAN, is UNCLASSIFIED, it can be marked on its face, top and bottom: "UNCLASSIFIED,", and a statement added: "All portions of this message are UNCLASSIFIED." No further markings are required. All other classified messages will be marked according to chapter 4 of this regulation.

## E-4. E-mail sensitivity

The availability and use of commercial e-mail programs within the Department of the Army has added another potential for security violations. These programs offer what is called "sensitivity" categories for messages. The most commonly available categories are: (1) NORMAL; (2) PERSONAL; (3) PRIVATE; and (4) CONFIDENTIAL. The use of this last category could open the user to potential security violations since classified information will not be sent over an unclassified system. The term "CONFIDENTIAL" is reserved strictly for classification purposes only. Therefore, this sensitivity category will not be used on any e-mail, sent or received on or by a government owned and/or operated machine and/or system.

## Section II Internet Websites

## E-5. General Overview

One of the most popular, and convenient, forms of spreading information about a unit or agency is to post it on the Internet. The cost of "surfing" has diminished to the point where an overwhelming majority of people have Internet access, not only at work, but also at home. It is as common as the family television set. It is estimated that by the next millenium, the majority of unit data exchange will be via Internet websites. It is for this reason that measures need to be instituted now before it is too late. For purposes of this appendix, the term "page" is used to represent every webpage or file related to, and linked from, any files that contain Department of Defense information on a unit or activity's Internet site, regardless if it is on an unclassified or a classified LAN. This includes embedded items, such as graphics, multimedia, etc.

## E-6. Unclassified websites

Much as classified and unclassified documents are easily marked and recognized, so must be unclassified webpages. For each unclassified Internet homepage/website, a banner stating that the website contains only unclassified, non-sensitive, and non-Privacy Act information is required as the first page visitors will come to. A banner similar to the one at figure E-1 will be used and no further markings are required.

WARNING!!!
UNCLASSIFIED, NON-SENSITIVE, NON-PRIVACY ACT USE ONLY
This is a Department of Defense (DoD) interest computer system. This system is
monitored to ensure proper operation to verify the functioning of applicable security
features and for other like purposes. Anyone using this system or any other DoD
computer system expressly consents to such monitoring and is advised that if such
monitoring reveals possible evidence of criminal activity, system personnel may provide
the evidence of such monitoring to law enforcement officials. Unauthorized attempts to
upload or change information; to defeat or circumvent security features, or to utilize this
system for other than its intended purposes are prohibited.

## Figure E-1. Unclassified Warning Banner

## E-7. Classified websites

Homepages/websites, on a classified network, are the same thing as pages in a classified document, especially after they have been printed out. It is vitally important that proper marking and handling procedures are implemented to reduce the risk of a security violation. For each classified homepage/website, a banner similar to the one at figure E-2 is required on the first page visitors come to.

# Intelink Security Banner All Intelink telecommunications and automated information systems are for the communication, transmission, processing, and storage of U.S. Government information only. These systems and equipment are subject to monitoring to ensure proper functioning, to protect against improper or unauthorized use or access, and to verify the presence or performance of applicable security features or procedures, and for like purposes. Such monitoring may result in the acquisition, recording, and analysis of all data being communicated, transmitted, processed, or stored in this system by a user. If monitoring reveals possible evidence of criminal activity, such evidence may be provided to law enforcement personnel.

ANYONE USING THIS SYSTEM EXPRESSLY CONSENTS TO SUCH MONITORING

Figure E-2. Intelink Security Banner

## E-8. Marking Classified Websites

Homepages/websites, on a classified network, will be handled in much the same manner, when it comes to marking and handling, as you would for a printed classified document.

a. Every page will bear the following markings:

(1) The overall (highest) classification of the information.

(2) The command, office of origin, date, and if not evident by the name of the command, the fact that the page was generated by the Army.

(3) Identification of the specific classified information in the page and its level of classification (page and portion markings).

(4) Identification of the source(s) of the classified information ("Classified By" or "Derived From" line).

(5) Declassification instructions ("Declassify On" line), and downgrading instructions, if any downgrading applies.

(6) Warning notices, if any, and other markings, if any, that apply to the page.

b. Each section, part, paragraph, and similar portion of each classified webpage will be marked to show the highest level of classification of information it contains, or that it is wholly UNCLASSIFIED. "Portion marking" is the term used to meet this requirement. The term "paragraph marking" is generally used interchangeably with "portion marking". Whether referred to as portion or paragraph marking, the term includes the marking of all portions of a webpage, not just paragraphs. When deciding whether a subportion (such as a subparagraph) will be marked separately as a "similar subportion", the deciding factor is whether or not the marking is necessary to eliminate doubt about the classification of its contents. Unless the original classification authority or originator of the source document indicates otherwise, each classified portion of a webpage will be presumed to carry the declassification instructions (date, event, or exemption category) of the overall document.

(1) Each portion of text will be marked with the appropriate abbreviation ("TS" for TOP SECRET, "S" for SECRET, "C" for CONFIDENTIAL, "U" for UNCLASSIFIED, "SBU" for Sensitive But Unclassified), placed in parentheses immediately before the beginning of the portion. If the portion is numbered or lettered, the abbreviation will be placed in parentheses between the letter or number and the start of the text. Some agencies permit portion marking at the end of the portion, rather than at the beginning. The Army does not. When extracts from non–Army source documents are made and incorporated into Army webpages, the portion marking will be placed at the beginning of the portion.

(2) Portions containing Restricted Data and Formerly Restricted Data will have abbreviated markings ("RD" or "FRD") included with the classification marking, for example, "(S–RD) or (S–FRD)". Critical Nuclear Weapons Design Information (CNWDI) will be marked with an "N" in separate parentheses following the portion marking, for example, "(S–RD)(N)".

(3) The abbreviation "FOUO" will be used in place of "U" when a portion is UNCLASSIFIED but contains For Official Use Only (FOUO) information. DODD 5200.1–R, appendix C, contains the definition and policy application of FOUO markings.

(4) Portions of Army webpages containing foreign government or NATO information will include identification of the foreign classification in the marking in parentheses. For example, "(UK–S)" for information classified "SECRET" by the United Kingdom; and "(NATO–C)" for NATO information classified as "CONFIDENTIAL".

(5) The subject and title of classified webpages will be marked to show the classification of the information in the subject or title. The same abbreviations ("TS", "S", "C", "U", or "FOUO") will be used but the abbreviations will be placed in parentheses at the end of the subject or title.

(6) Charts, graphs, photographs, illustrations, figures, tables, drawings, and similar items will be marked with the unabbreviated classification such as "UNCLASSIFIED", based on the level of classified information revealed. The marking will be placed within the chart, graph, etc., or next to it, such as on the frame holding the document. Captions and titles of charts, graphs, etc., will be marked as required for text portions (such as paragraphs) and will be placed at the beginning of the caption or title.

(7) See appendix D, of this regulation, for an explanation of portion marking certain intelligence control markings (for instance, ORCON and PROPIN). Portion marking of those intelligence control markings will follow the same policy as stated in this section of the regulation.

#### E-9. Publicly Accessible Websites

Figure E–3, of this appendix, contains a memorandum, from the Director of Information Systems for Command Control, Communications, and Computers (DISC4), that provides guidance for the establishment and operation of publicly accessible, non-restricted, U.S. Army World Wide Web websites (Army websites). Publicly accessible, non-restricted Army websites will only provide information that has been properly cleared for release. The organization's leadership is responsible for the release of all information on the organization's website.

Guidance for Management of Publicly Accessible U.S. Army Websites 30 November 1998

SUBJECT: Guidance for Management of Publicly Accessible U.S. Army Websites

1. Purpose

a. This memorandum provides guidance for the establishment and operation of publicly accessible, non-restricted, U.S. Army World Wide Web (WWW) websites (Army websites). Publicly accessible, non-restricted Army websites will only provide information that has been properly cleared for release.

b. The World Wide Web is an efficient and effective means for the U.S. Army to share information. Army websites should focus on providing value-added information services and products to the organization's users, customers, the Army, and the public through the sharing of accurate, timely, and relevant information. To ensure that the Army fully leverages the capabilities of the WWW in a manner that is efficient, focused on saving resources, and moving toward a digital environment, the following guidance is provided.

2. Proponent and Exception Authority

a. The proponent for this memorandum is the Director of Information Systems for Command Control, Communications, and Computers (DISC4).

b. The DISC4 has the authority to approve exceptions to this memorandum that are consistent with controlling law and regulation. The DISC4 may delegate the authority to approve exceptions to this policy, in writing, to a division chief under his supervision within the proponent agency that holds the grade of Colonel or GM/GS-15.

3. References

a. This policy supersedes Director of Information Systems for Command, Control, Communications, and Computers, 301404Z October 1996, Guidance for the Management of Army Websites.

b. 5 USC Chapter 35, "Paperwork Reduction Act", as amended.

c. Public Law 100-235, Computer Security Act of 1986.

d. For guidance on use of government-owned computing equipment and resources

(e.g., non-duty related email use and web browsing in the workplace), see DoD 5500.7-

R, Joint Ethics Regulation (JER) (30 August 1993) and Change 2 (25 March 1996).

e. For all DoD newspapers, including electronic publications, DoD Instruction 5120.4,

DoD Newspapers and Civilian Enterprise Publications (May 29, 1996) applies.

f. For image manipulation standards, DoD Directive 5040.5, Alteration of Official DoD Imagery (August 29, 1995) applies.

g. AR 25-1, The Army Information Resource Management Program (25 March 1997).

h. AR 25-55, Army Freedom of Information Act Program (1 November 1997).

i. For access-controlled websites, AR 380-19, Information System Security (27 August 1998) applies.

- j. AR 340-21, Army Privacy Act Program (5 July 1985).
- k. AR 360-5, Public Information (31 May 1989).

I. AR 380-5, Department of the Army Information Security Program (25 February 1988). m. AR 530-1, Operational Security (3 March 1995).

n. HTML 3.2 Reference Specification, World Wide Web Consortium (W3C) Recommendation (11 January 1997).

o. HTML 4.0 Specification, World Wide Web Consortium (W3C) Recommendation (24 April 1998).

- 4. Definitions and Explanation of Abbreviations
- a. WWW World Wide Web
- b. HTML Hypertext Markup Language
- c. W3C World Wide Web Consortium
- d. CGI Common Gateway Interface
- e. GILS Government Information Locator Service
- f. FOUO For Official Use Only
- g. FOIA Freedom of Information Act
- h. MACOM Major Command
- i. GO/SES General Officer/Senior Executive
- j. DoD Department of Defense

k. Webpage - an individual HTML-compliant electronic file accessible through a TCP/IP network

I. TCP/IP network - a data communication network that uses Transport Control

Protocol/Internet Protocol (TCP/IP); the public internet and the DoD Non-Classified IP

Router Network (NIPRNET) are examples of TCP/IP networks

m. Website - a collection of HTML-compliant electronic files designed to provide

information, services, or goods to users through a TCP/IP network

n. Homepage - the single, top-level, webpage designed to be the first file accessed by a

user visiting a website; also known as an "index" or "default" page

5. Responsibilities

a. The leader of each organization that operates an official U.S. Army website (leadership), regardless of location or echelon (e.g., unit, office, installation, major command), will--

(1) exercise ultimate control over the content of the organization's website,

(2) define the purpose of the website in terms of how it supports the mission of the organization,

(3) define the core functions, products, and information that will be made available through the organization's website,

(4) ensure compliance with all applicable policies, including AR 530-1, Operational Security (3 March 1995), and

(5) periodically reevaluate each website under their control to ensure performance of the responsibilities in paragraphs 5.a.(1) through 5.a.(4).

b. The organization's leadership may delegate the execution of this responsibility to one or more website managers and other appropriate officials. Where appropriate, the organization's leadership may delegate these responsibilities to a team of subject matter experts, the exact composition of which is left to the discretion of the leadership. This team may be composed of subject matter experts from one or more of the following communities: Public Affairs, Communications/Computers, Intelligence, Legal, and others.

6. Policy

a. Specifications and Standards.

(1) Official U.S. Army websites should be implemented in such a way as to support the widest range of potential users and computing platforms. Use of non-standard or proprietary website elements will not benefit the widest range of potential users.
(2) For hypertext markup language (HTML) documents, official U.S. Army websites must use any of the HTML specifications listed by the World Wide Web Consortium (W3C). As an alternative, official U.S. Army websites may employ any HTML specification that is a W3C Proposed Recommendation, or any non-HTML specification that is a W3C Recommendation. W3C Technical Reports (including Recommendations and Proposed Recommendations) are found online at "http://www.w3.org/TR/".
(3) Official U.S. Army websites may employ non-standard (e.g., browser-specific) HTML tags and browser extensions (plug-in). However, official U.S. Army websites may not require or encourage users to use any particular browser product or "plug-in" technologies. Official U.S. Army websites may not be designed to support any particular browser product.

b. Requirements for Organizations Operating Websites.

(1) Every Army organization that maintains a website must register it with the U.S. Army Homepage through the online registration form found on the U.S. Army Homepage at "http://www.army.mil/register3.htm".

(2) Every Army organization that maintains a website must notify the U.S. Army Homepage Webmaster, via e-mail at webmaster@hqda.army.mil, when the Universal Resource Locator (URL) or any of the point of contact information required as part of the registration process changes.

(3) Every Army organization that maintains a website must register it with the

Government Information Locator Service (GILS), see "http://www.dtic.mil/index/."

(4) Every Army organization that maintains a website must display a Privacy and Security Notice.

(5) U.S. Army organizations operating an official website will provide the following information or hyperlinks to the following information on their homepage:

(a) Organization missions and functions.

(b) Organizational structure, listing or hyperlinking to parent and subordinate command or organization websites. Organizational charts containing individuals' names and other personal information should not be made available to the public unless privacy and security concerns have been addressed; posting such information for members of deployable units and others in sensitive positions could make them potential targets of hostile organizations or individuals.

(c) Electronic mail address, phone number, or mail address of the point of contact responsible for the website content.

(d) A hyperlink to the U.S. Army Homepage using "http://www.army.mil".

Figure E-3. Guidance for Management of Publicly Accessible U.S. Army Websites-Continued

c. Requirements for Website Managers

A website manager is the organizations' leader, or an individual or group that has been

delegated the following responsibilities by the organization's leadership. Website

Managers (webmasters) will--

(1) Ensure that information published on their website is accurate, timely, represents the official Army position, and is properly cleared for public dissemination;

(2) Ensure appropriate security and access controls are in place, commensurate with

the perceived threats, and to ensure that the following types of information is not made

available to unauthorized individuals or organizations:

(a) Classified

(c) Information that cannot be disclosed under the Privacy Act

(d) For Official Use Only (FOUO)

(e) Freedom of Information Act (FOIA)-exempt information (including, but not limited to

draft policies and regulations, and pre-decisional information)

(f) Copyrighted information for which releases from the copyright owner have not been obtained

(3) Provide the highest practicable level of assurance that information made available to

or received from the public does not contain malicious software code (e.g., viruses,

trojan horses), or if it does, to sufficiently notify the user before the download of such

information begins;

(4) Respond to email, direct queries to the appropriate source of information, or

otherwise fulfill or redirect requests for information;

(5) Ensure that the organization's website provides point of contact information for the webmaster.

d. Requirements for Webpages.

All U.S. Army webpages will display the date when that page was last updated,

reviewed, or cleared for public release.

e. Release of Information.

(1) The organization's leadership will institute a review process to ensure that information provided on their website(s) is current, timely, and cleared for public release. The organization's leadership is responsible for the release of all information on the organization's website.

(2) The following types of information will not be made available to the public through the WWW:

(a) Classified

(b) Unclassified but sensitive

(c) Information that cannot be disclosed under the Privacy Act

(d) For Official Use Only (FOUO)

(e) Freedom of Information Act (FOIA)-exempt information (e.g., draft policies and regulations, or pre-decisional information)

(f) Copyrighted information for which releases from the copyright owner have not been obtained

(3) Commanders of Army major commands (MACOM Commander), or equivalent, may authorize a waiver to the restrictions at paragraphs 6c(2) and 6e(2) for draft doctrinal and draft technical information only. The ability to waive this prohibition may not be delegated below the GO/SES-level in the MACOM Headquarters. To authorize a waiver in such cases, a MACOM Commander (or the delegated MACOM Headquarters authority) must:

(a) Sign a memorandum waiving the prohibition against releasing draft doctrinal or draft technical information to the public.

(b) Addresses the intelligence and national security, public affairs, legal, and contractual issues pertinent to the public release of the draft doctrinal or draft technical information to the public.

f. Commercial Advertising and Sponsorship.

(1) Commercial advertising on official U.S. Army websites is prohibited. Corporate or product logos and trademarks (other than text or hyperlinked text) are considered commercial advertisements, and may not be served from official U.S. Army websites.

(2) No money, services, products, or in-kind payment (e.g., website hosting, site management, site design) will be accepted in exchange for a link to a non-Army web resources placed on an official U.S. Army website.

(3) No product endorsement will be served from an official U.S. Army website. Official

U.S. Army websites will not provide preferential treatment to non-U.S. Government

entities.

g. External Linked Content.

The ability to hyperlink to resources external to the Army is a fundamental feature of the World Wide Web, and can add value and functionality to Army websites.

(1) Hyperlinks to web resources other than official U.S. Army (non-Army) web resources are permitted if the organization's leadership certifies them to be in support the organization's mission.

(2) Official U.S. Army websites may use only text or hyperlinked text to direct users to non-Army software download sites.

(3) Army websites that provide links to non-Army web resources must display a

disclaimer in accordance with DoD policy.

h. Collection of Information.

Army websites that collect standardized information from 10 or more members of the

public must comply with:

(1) DoD Memorandum, Establishing and Maintaining a Publicly Accessible Department

of Defense Web Information Service, 18 July 1997, found at

"http://www.defenselink.mil/policy97.html."

(2) The Paperwork Reduction Act of 1995 (as amended).

i. Personal Use.

(1) Personal use of government resources generally is improper.

(2) Hyperlinks on Army websites to homepages, websites, or other web resources of a personal and non-mission related nature are prohibited.

(3) Army Internet users are subject to DoD 5500.7-R, change 2, Joint Ethics Regulation (JER), 25 Mar 1996.

j. Restricted Access.

(1) In addition to not posting certain information to Army websites as noted above in

paragraph 6.e., webmasters shall ensure that Army websites do not provide direct

hyperlinks (or other methods to bypass access-controls, such as hyperlinking to

webpages below password protection webpages) to the following types of information:

- (a) Classified
- (b) Unclassified but sensitive
- (c) Information that cannot be disclosed under the Privacy Act
- (d) For Official Use Only (FOUO)

(e) Freedom of Information Act (FOIA)-exempt information (including, but not limited to

draft policies and regulations, and pre-decisional information)

(f) Copyrighted information for which releases from the copyright owner have not been obtained

(2) Publicly accessible Army websites may provide hyperlinks to access-controlled websites only through intervening access-control mechanisms or procedures that are sufficient to address the perceived level of threat and sensitivity of the information.

(3) Army websites must not use inflammatory or threatening language when describing access-controls and procedures, and must avoid the perception that the Army is hiding or withholding information that otherwise would be available to the public.

7. Point of Contact

Point of contact for this policy is the Army Homepage Webmaster. Their e-mail address is "webmaster@hqda.army.mil."

## Appendix F Management Control

## Section I Management Control Evaluation Checklist

## F-1. Purpose

The purpose of this checklist is to assist Command Security Managers and Management Control Administrators (MCAs) in evaluating the key management controls outlined below. It is not intended to cover all controls. It is to be answered in a YES/NO/NA format. A negative response (NO) is to be explained at the end of the question. The locations in bold are provided as reference points within AR 380–5.

## F-2. Instructions

Answers must be based on the actual testing of key management controls (e.g., document analysis, direct observation, sampling, simulation, etc.). Answers that indicate deficiencies must be explained and corrective action indicated in supporting documentation. These key management controls must be formally evaluated at least once every five years. Certification that this evaluation has been conducted must be accomplished on DA Form 11-2-R (Management Control Evaluation Certification Statement). All Department of the Army units, with access to classified information, will develop and implement an inspection program for annual inspections. This checklist can serve as the base for the annual inspection and can include other questions as determined by the agency or command performing the inspection. Inspection programs are used to evaluate the effectiveness of these key management controls as well as other requirements of this regulation as they apply to the command being inspected.

## F-3. Test questions

## Chapter 1 General Provisions and Program Management

Paragraph 1-5

Responsibilities of Deputy Chief of Staff for Intelligence (DCSINT), Headquarters Department of the Army.

#### Does the DCSINT, DAMI-CH-

*a.* Promulgate (or cause to promulgate) policy, procedures, and programs necessary for the implementation of EO 12958 and resulting national and DOD directives?

b. Monitor, evaluate, and report on the administration of the Army Information Security Program?

c. Ensure the Major Army Commands (MACOM), and other agencies, establish and maintain an ongoing self-inspection program, to include periodic reviews and assessments of their classified products?

d. Respond to information security matters pertaining to classified information that originated in an Army activity that no longer exists and for which there is no successor in function?

e. Delegate SECRET and CONFIDENTIAL Original Classification Authority (OCA) to other Army officials?

f. Commit needed resources for effective policy development and oversight of the programs established by this regulation?

Paragraph 1–6

Responsibilities of the Commander.

Does the Commander-

a. Establish written local information security policies and procedures?

*b*. Initiate and supervise measures or instructions necessary to ensure continual control of classified information and materials?

c. Assure that persons requiring access to classified information are properly cleared?

d. Continually assess the individual trustworthiness of personnel who possess a security clearance?

e. Designate a Command Security Manager by written appointment? Is the Security Manager of sufficient rank or grade to effectively discharge assigned duties and responsibilities?

f. Make sure the Command Security Manager is afforded security training consistent to the duties assigned?

g. Make sure adequate funding and personnel are available to allow security management personnel to manage and administer applicable information security program requirements?

*h*. Review and inspect annually the effectiveness of the Information Security Program in subordinate commands? *i*. Make sure prompt and appropriate responses are given, or forward for higher echelon decision, any problems, suggestions, requests, appeals, challenges, or complaints arising out of the implementation of this regulation?

Paragraph 1-7

Responsibilities of the Command Security Manager.

Does the Command Security Manager-

*a*. Advise and represent the Commander on matters related to the classification, downgrading, declassification, and safeguarding of national security information?

b. Establish and implement an effective security education program, as required by chapter 9 of this regulation? c. Establish procedures for assuring that all persons handling classified material are properly cleared? Is the clearance status of each individual recorded and accessible for verification?

d. Advise and assist officials on classification problems and the development of classification guidance?

e. Ensure that classification guides for classified plans, programs, and projects are properly prepared and maintained?

*f.* Conduct a periodic review of classifications assigned within the activity to ensure that classification decisions are proper?

g. Review all classified documents, in coordination with the agency or command records management officer, to ensure consistency with operational and statutory requirements?

h. Continually reduce, by declassification, destruction, or retirement, unneeded classified material?

*i.* Submit, in a timely manner, Standard Form 311 (Information Security Program Data Report) to DAMI-CH, annually, as required?

*j*. Supervise or conduct security inspections and spot checks and notify the Commander regarding compliance with this regulation and other applicable security directives?

*k*. Assist and advise the Commander on matters pertaining to the enforcement of regulations governing the dissemination, reproduction, transmission, safeguarding, and destruction of classified material?

*l*. Make recommendations on requests for visits by foreign nationals, and provide security and disclosure guidance if visit is approved?

*m*. Make sure of the completion of inquiries and reporting of security violations occur including compromises or other threats to the safeguarding of classified information?

*n*. Recommend to the decision official whether or not administrative sanction is warranted, and/or indicate corrective action that should be taken concerning security violations?

o. Make sure proposed public releases on classified programs are reviewed to preclude the release of classified information, or other sensitive unclassified information covered under the Freedom of Information Act?

p. Establish and maintain visit control procedures for visitors are authorized access to classified information?

q. Issue contingency plans for the emergency destruction of classified information and, where necessary, for the safeguarding of classified information used in or near hostile or potentially hostile areas?

r. Act as the single point of contact to coordinate and resolve classification or declassification problems?

s. Report data as required by this regulation?

#### Paragraph 1-8

Responsibilities of the Supervisor.

Does the Supervisor-

*a*. Make sure subordinate personnel who require access to classified information are properly cleared and are given access only to that information for which they have a need-to-know?

b. Make sure subordinate personnel are trained in, understand, and follow the requirements of this regulation and local command policy and procedures concerning the information security program?

c. Continually assess the eligibility for access to classified information of subordinate personnel and report to the Command Security Manager any information that may have a bearing on that eligibility?

*d*. Supervise personnel in the execution of procedures necessary to allow the continuous safeguarding and control of classified information?

*e*. Include the management of classified information as a critical element/item/objective in personnel performance evaluations, where deemed appropriate, in accordance with Army personnel policy and paragraph 1–5c of this regulation? (A supervisor should include the protection of classified information as a performance evaluation factor or objective for other personnel as the supervisor deems appropriate.)

*f.* Lead by example? Does he/she follow Command and Army policy and procedures to properly protect classified information and to appropriately classify and declassify information as stated in this regulation?

#### **Chapter 2 Classification**

#### Paragraph 2–1

Are personnel designated, in writing, by either the Secretary of the Army (SECARMY) or HQDA, as Original Classification Authorities (OCA)?

#### Paragraph 2–3:

Are requests for original classification authority submitted through command channels to DAMI-CH?

#### Paragraph 2–4:

Do officials who have been delegated as an Original Classification Authority receive training, as required by chapter 9 of this regulation, before exercising this authority?

#### Paragraph 2–5:

Do derivative classifiers make sure that the classification is properly applied based on the original source material marking and local security classification guides?

#### Paragraph 2–6:

Do personnel accomplishing derivative classification:

a. Observe and respect the classification determinations made by Original Classification Authorities?

b. Apply markings or other means of identification to the derivatively classified material, as required by this regulation, at the level and for the duration specified by the classification guide or source document?

c. Use only authorized sources such as classification guides, other forms of official classification guidance, and markings on source material from which the information is extracted, to determine the material's classification?

d. Use caution when paraphrasing or restating information extracted from a classified source to determine whether the classification could have been changed in the process?

e. Take appropriate and reasonable steps to resolve doubt or conflicts in classification?

*f*. Make a list of sources used when material is derivatively classified based on "Multiple Sources" (more than one security classification guide, classified source document, or any combination)? Is a copy of this list included in or attached to the file and/or record copy of the material?

g. Contact the classifier of the source document for resolution in cases in which the derivative classifier believes the classification applied to the information is not accurate?

#### Paragraph 2–7:

In making a decision to originally classify an item of information, do Original Classification Authorities:

a. Determine that the information has not already been classified?

b. Determine that the information is eligible for classification pursuant to paragraph 2-8 of this regulation?

c. Determine that classification of the information is a realistic course of action and that it can only be protected from unauthorized disclosure when classified?

*d*. Decide that unauthorized disclosure of the information could reasonably be expected to cause damage to the National Security that this disclosure is identifiable and can be described?

*e*. Select the appropriate level or category of classification to be applied to the information, based on a judgement as to the degree of damage unauthorized disclosure could cause?

*f*. Determine and include the appropriate declassification, and when applicable, downgrading instruction to be applied to the information?

g. Make sure that the classification decision is properly communicated so that the information will receive appropriate protection?

#### Paragraph 2–8:

U.S. classification can only be applied to information that is owned by, produced by or for, or is under the control of the United States Government. Does the Original Classification Authority determine that the unauthorized disclosure of the information reasonably could be expected to result in damage to the national security, and that the information falls within one or more of the categories specified in section 1.5 of Executive Order 12958?

#### Paragraph 2–9:

Does the Original Classification Authority determine that, if classification is applied or reapplied, there is a reasonable possibility that the information will be provided protection from unauthorized disclosure?

#### Paragraph 2-10:

Once a decision is made to classify, information will be classified at one of three levels. For each level, is the Original Classification Authority able to identify or describe the damage that unauthorized disclosure reasonably could be expected to cause to the national security?

#### Paragraph 2–11:

a. Is information declassified as soon as it no longer meets the standards for classification?

*b*. At the time of original classification, does the Original Classification Authority attempt to establish a specific date or event for declassification based upon the duration of the national security sensitivity of the information?

c. If unable to determine a date or event that is ten years or less, does the Original Classification Authority assign an exemption designation to the information if the information qualifies for exemption from automatic declassification in ten years?

#### Paragraph 2–16:

Is a security classification guide issued for each system, plan, program, or project in which classified information is involved?

#### Paragraph 2–17:

Do security classification guides, at a minimum, include the information outlined in paragraph 2-17a through f?

#### Paragraph 2–18:

*a*. Are security classification guides personally approved in writing by the Original Classification Authority who is authorized to classify information at the highest level designated by the guide, and who has program support or supervisory responsibility for the information or for the organization's Information Security Program?

b. Are security classification guides distributed to those commands, contractors, or other activities expected to be derivatively classifying information covered by the guide?

#### Paragraph 2-19:

a. Are security classification guides revised whenever necessary to promote effective derivative classification? b. Are security classification guides reviewed by the originator for currency and accuracy at least once every five years, or if concerning a defense acquisition program, prior to each acquisition program milestone, whichever occurs first?

#### Paragraph 2–22a:

Does the Commander establish procedures through which authorized holders of classified information, within their Commands, can challenge a classification decision, and make sure that Command personnel are made aware of the established procedures?

#### Paragraph 2–22b:

Does each Original Classification Authority establish:

a. A system for processing, tracking, and recording formal challenges to classification?

*b*. Provide an acknowledgment or written response to the challenge within 60 calendar days following the receipt of the challenge?

c. Advise the challenger of the right to appeal the decision, if the challenge is denied and the Original Classification Authority determines that the information is properly classified?

#### Paragraph 2–22c:

Is information, that is the subject of a classification challenge, continue to be classified and appropriately safeguarded until a decision is made to declassify it?

#### Chapter 3 Declassification, Regrading, and Destruction

#### Paragraph 3–1:

a. Is information declassified when it no longer meets the standards and criteria for classification?

b. Do MACOMs establish programs to make sure that records are reviewed and either declassified or exempted prior to the date for automatic declassification?

#### Paragraph 3–3:

Is declassification of RD and FRD information only with the express specific approval of the Original Classification Authority for the information?

#### Paragraph 3–4:

Is information classified by other U.S. Executive Branch agencies or by foreign governments or international organizations (including foreign contractors) referred to the originating agency (or its successors in function) prior to declassification? (In the case of a foreign government, refer to its legitimate successor for a declassification decision.)

#### Paragraph 3–5:

*a*. Does the automatic declassification requirement exist for all records of permanent historical value as they become 25 years old?

b. Are Army files, determined to be of permanent historical value under Title 44, USC? (Unless that information has been reviewed and exempted, automatic declassification will occur whether or not the records have been reviewed.)

#### Paragraph 3–6:

In accordance with EO 12958, the Army has identified and proposed specific designated file series, with description and identification of information in those file series, to be exempt from the 25-year automatic declassification.

*a*. Are files containing information described in the list of exempt file series, located and marked to reflect the exemption from automatic declassification, the applicable exemption category, and the date or event for future declassification?

*b*. Is classified information, not contained in the exempted file series, exempted from declassification if they fall within one of the nine exemption categories that are listed in paragraph 3–6e?

c. Are requests for exemption for other than listed exemptions reported to DAMI-CH?

Paragraph 3–7:

Are documents that are exempted from automatic declassification after 25 years, marked with the designation "25X", followed by the number of the exemption category (see categories listed in paragraph 3–6e), or by a brief reference to the pertinent exemption?

Paragraph 3–10:

Is information downgraded to a lower level of classification when the information no longer requires protection at the originally assigned level and can be properly protected at a lower level?

Paragraph 3–12:

Is classified information upgraded to a higher level of classification only by officials who have been delegated the appropriate level of Original Classification Authority? Do they also notify holders of the change in classification?

## Paragraph 3–13:

Is classified material destroyed completely to preclude recognition or reconstruction of the classified information contained in or on the material? Is destruction accomplished in accordance with the guidelines outlined in this regulation?

## **Chapter 4 Marking**

## Paragraph 4–1:

Is classified material identified clearly by marking, designation, electronic labeling, or if physical marking of the medium is not possible, by some other means of notification?

## Paragraph 4–3:

Does classified material bear the following markings:

a. The overall (highest) classification of the information?

*b*. The Command, office of origin, date, and if not evident by the name of the Command, the fact that the document was generated by the Department of the Army?

c. Identification of the specific classified information in the document and its level of classification (page and portion markings)?

*d.* Identification and date of the source(s) of classification ("Classified by" or "Derived from" line), and, for originally classified information, the concise reason(s) for classification?

*e*. Declassification instructions ("Declassify on" line), and downgrading instructions, if any downgrading applies? *f*. Warning notices and other markings, if any, that apply to the document?

Paragraph 4–12:

a. Are required warning notices used, where applicable?

b. Is the required warning notice on all government owned or operated automated information systems?

## Paragraph 4–15:

Are all unclassified Army records, including For Official Use Only (FOUO), regardless of media, and all classified Army records through SECRET, marked according to the MARKS system, to include disposition instructions?

#### Paragraph 4–16:

If a classified document has components likely to be removed and used or maintained separately, is each component marked as a separate document?

## Paragraph 4–19:

Are translations of U.S. classified information into a foreign language marked with the appropriate U.S. classification markings and the foreign language equivalent?

#### Paragraph 4–21:

Are documents which contain no classified information, but are marked with classification markings for training purposes, marked to clearly show that they are actually UNCLASSIFIED?

Paragraph 4–22:

Are files, folders, and similar groups of documents containing classified information clearly marked as to the highest classification of information contained therein?

Paragraph 4–23:

Are documents produced on an AIS marked like other documents except where designated by paragraph 4-23?

Paragraph 4–24:

When classified information is contained in equipment, hardware, AIS media, or on film, tape, or other audiovisual media, or in another form not commonly thought of as a document, is the marking provisions of this regulation met in a way that is compatible with the type of material?

Paragraph 4–33:

Do System Managers make sure that AIS, including word processing systems, provide for classification designation of data stored in internal memory or maintained on fixed storage media?

Paragraph 4–45:

Are Joint Chiefs of Staff (JCS) papers, including extractions from such papers, safeguarded in accordance with this regulation?

Paragraph 4–56:

Are classified documents originated by NATO, if not already marked with the appropriate classification in English, so marked?

#### **Chapter 5 Controlled Unclassified Information**

Paragraph 5–2:

"For Official Use Only (FOUO)" is a designation that is applied to unclassified information, which is exempt from mandatory release to the public under the Freedom of Information Act (FOIA). The FOIA specifies nine categories of information which can be withheld from release if requested by a member of the public.

*a*. Does information to be exempt from mandatory release, fit into one of the qualifying categories and is there a legitimate government purpose served by withholding it?

b. Is information which has been determined to qualify for FOUO status indicated, by markings, when included in documents and similar material?

Paragraph 5–5:

a. During working hours, are reasonable steps taken to minimize risk of access by unauthorized personnel?

b. After working hours, is FOUO information stored in approved methods outlined in paragraph 5-5?

c. Are FOUO documents and material transmitted by approved means?

Paragraph 5–10:

Is SBU information protected as that required for FOUO information?

Paragraph 5–12:

Are UNCLASSIFIED documents containing DEA sensitive information marked as required by paragraph 5-12?

Paragraph 5–14:

*a*. Are UNCLASSIFIED documents containing DEA sensitive information transmitted outside CONUS by a means approved for transmission of SECRET material?

b. Is DEA sensitive material destroyed by a means approved for the destruction of CONFIDENTIAL material?

#### Paragraph 5–16:

Are UNCLASSIFIED documents and material containing DOD UCNI marked as required by paragraph 5-16?

#### Paragraph 5–22:

Is Information on DOD AIS systems, which is determined to be "sensitive," within the meaning of the Computer Security Act of 1987, provided protection as required by paragraph 5–22?

#### Paragraph 5–24:

Are distribution statements placed on technical documents no matter if they are classified or unclassified?

#### Chapter 6 Access, Control, Safeguarding & Visits

#### Paragraph 6–2:

*a.* Prior to granting access to classified information, do all unit personnel receive a briefing outlining their responsibility to protect classified information and have they signed the Classified Information Nondisclosure Agreement (NDA), Standard Form SF 312 or the SF 189?

b. Are SF 312 and SF 189 NDAs retained for 50 years from the date of signature?

#### Paragraph 6–3:

Prior to execution of the NDA, does the command have proof of clearance?

#### Paragraph 6–4:

a. If a person refuses to sign the form, is the individual advised of the applicable portions of the NDA?

b. If after the five days the individual still refuses to sign the NDA, is classified access, if it had been previously granted, formally suspended and the individual not permitted any access to classified information?

c. Is the Department of the Army's Central Clearance Facility notified concerning clearance revocation or denial action, and the matter reported as required by AR 380-67?

#### Paragraph 6–5:

*a*. Are all DA personnel who are retiring, resigning, being discharged, or will no longer have access to classified information, out-processed through the Command's Security Manager's office or other designated command office, and are they formally debriefed?

b. Do all personnel sign a debriefing statement during out-processing? (The debriefing statement will be either DA Form 2962 or the NDA Security Debriefing Acknowledgement section of the SF 312.)

c. For all DA military personnel retiring, resigning, or separating from military service, is the termination portion of the NDA executed and maintained on file by the Command Security Manager, or other designated command official, for two years?

d. Does the original NDA, for civilian employees, who retire or resign from government service, remain in the employee's OPF and will be retired as part of the OPF?

#### Paragraph 6–7:

Is access to RD (less CNWDI) and FRD by DA personnel at Army facilities, under the same conditions as for all other classified information, based on the appropriate security clearance and need-to-know for the information?

#### Paragraph 6–8:

Is access to classified information or material by Congress, its committees, subcommittees, members, and staff

representatives, in accordance with DOD Directive 5400.4?

Paragraph 6–9:

Do Commands maintain a system of control measures that ensures that access to classified information is limited only to authorized persons?

Paragraph 6–10:

*a.* Is classified material removed from storage kept under constant surveillance and control by authorized personnel? *b.* Are classified document cover sheets, Standard Forms 703, 704, and 705 placed on classified documents or files not in security storage?

Paragraph 6–11:

Do Commands that access, process, or store classified information establish a system of security checks at the close of each working day to ensure that all classified material is properly secured? Is Standard Form 701 used to record these checks?

Paragraph 6–12:

Have Commands developed plans for the protection, removal, and destruction of classified material in case of fire, flood, earthquake, other natural disasters, civil disturbance, terrorist activities, or enemy action, to minimize the risk of its compromise?

Paragraph 6–13:

Is classified information only discussed, in telephone conversations, over secure communication equipment, such as a STU–III, and circuits approved for transmission of information at the level of classification being discussed?

Paragraph 6–15:

Are storage containers and information processing equipment, which had been used to store or process classified information, inspected by cleared personnel before removal from protected areas or before unauthorized persons are allowed unescorted access to them?

Paragraph 6–16:

Have Commands established procedures to control access to classified information by visitors?

Paragraph 6–18:

When a DA command wishes to be authorized to serve as the security sponsor for an association-related classified meeting, have they made the request, through command channels, to DAMI-CH, and is it received at least 120 days in advance of the meeting?

Paragraph 6–19:

Does the Command have security procedures that prescribe the appropriate safeguards to prevent unauthorized access to non-COMSEC-approved equipment, that are used to process classified information, and replace and destroy equipment parts, pursuant to the level of the classified material contained therein, when the information cannot be removed from them?

Paragraph 6–20:

*a*. Has the Command developed procedures to protect incoming mail, bulk shipments, and items delivered by messenger, until a determination is made whether classified information is contained in the mail?

b. Have screening points been established to limit access to classified information?

Paragraph 6–21:

*a.* Has the Command established procedures, tailored to the individual situation, in accordance with the principles of risk management, for the control and accountability of the TOP SECRET material they hold?

*b.* Have TOP SECRET Control Officers (TSCO) been designated within offices which handle or maintain TOP SECRET material?

*c*. Are TOP SECRET materials accounted for by a continuous chain of receipts, and are the receipts maintained for 5 years?

*d.* Is TOP SECRET material inventoried at least once annually? Does the inventory reconcile the TOP SECRET accountability register and records with 100 percent of the TOP SECRET material held? Is the inventory conducted by the TSCO, or alternate, and one properly cleared disinterested individual?

*e*. Before leaving the command, does the TSCO or alternate conduct a joint inventory with the new TSCO or alternate of all TOP SECRET material for which they have custodial responsibility?

#### Paragraph 6–22:

Has the Command established procedures to control all SECRET and CONFIDENTIAL information and material originated, received, distributed, or routed to subelements within the command, and all information disposed of by the command by transfer of custody or destruction?

#### Paragraph 6-24:

Are Working papers containing classified information handled IAW paragraph 6-24?

#### Paragraph 6–25:

*a*. Has the Command established and enforced procedures for the reproduction of classified material which limit reproduction to that which is mission essential and do they make sure that appropriate countermeasures are taken to negate or minimize any risk?

b. Are all copies of classified documents reproduced for any purpose, including those incorporated in working papers, subject to the same safeguards and controls prescribed for the document from which the reproduction is made?

c. Is stated prohibition against reproduction of information at any classification level prominently displayed?

d. Is specific equipment designated for the reproduction of classified information and prominently marked as such?

#### Paragraph 6–26:

Have Command established procedures which ensure that appropriate approval is granted before classified material is reproduced?

#### Paragraph 6-27:

*a*. Are classified documents and other material retained only if they are required for effective and efficient operation of the Command or if their retention is required by law or regulation?

b. Are documents which are no longer required for operational purposes disposed of in accordance with the provisions of the Federal Records Act (44 USC Chapters 21 and 33) as implemented by AR 25-400-2?

c. Do Commanders ensure that the management of the retention of classified material is included in oversight and evaluation of program effectiveness?

#### Paragraph 6–28:

Are classified documents and materials destroyed by burning or, when meeting the standards contained in Chapter 3, of this regulation, by melting, chemical decomposition, pulping, pulverizing, cross-cut shredding, or mutilation, sufficient to preclude recognition, or reconstruction of the classified information?

#### Paragraph 6-29:

a. Are records of destruction for TOP SECRET documents and material created at the time of destruction?

*b*. Are records of destruction for NATO and foreign government documents at the SECRET level created at the time of destruction?

c. Are records of destruction maintained for 5 years from the date of destruction?

#### Paragraph 6–35:

Do the heads of DA MACOMs, units, activities, and agencies establish and maintain a self-inspection program based on program needs and the degree of involvement with classified information?

#### Chapter 7 Storage and Physical Security Standards

Paragraph 7-4:

a. Is classified information, that is not under the personal control and observation of an authorized person, guarded or stored in a locked security container, vault, room, or area, pursuant to the level of classification and this regulation?

*b*. Do Commands establish administrative procedures for the control and accountability of keys and locks whenever key-operated, high-security padlocks are utilized?

#### Paragraph 7–8:

*a*. Is the requirement that there will be no external mark revealing the level of classified information, authorized to be stored in a given container or vault, followed?

b. Is the requirement that there will be no external mark revealing priorities for emergency evacuation and destruction marked or posted on the exterior of storage containers, vaults, or secure rooms, followed?

c. For identification and/or inventory purposes only, does each vault or container bear, externally, an assigned number or symbol?

d. Are Combinations changed:

(1) When placed in use?

(2) Whenever an individual knowing the combination no longer requires access?

(3) When the combination has been subject to possible compromise?

(4) At least once annually?

(5) When taken out of service, are built-in combination locks reset to the standard combination 50-25-50, and combination padlocks reset to the standard combination 10-20-30?

(6) Annually, per U.S. Central Registry, when NATO information is stored in the security container, vault, or secure room?

*e*. Is the combination of a container, vault or secure room used for the storage of classified information treated as information having a classification equal to the highest category of the classified information stored inside?

f. Is a record maintained for each vault or secure room door, or container used for storage of classified information, using Standard Form 700, "Security Container Information," or the equivalent?

g. Is access to the combination of a vault or container used for the storage of classified information granted only to those individuals who are authorized access to the classified information that is to be stored inside?

*h*. Are entrances to secure rooms or areas either under visual control at all times during duty hours, to preclude entry by unauthorized personnel, or is the entry equipped with electric, mechanical, or electro–mechanical access control devices to limit access during duty hours?

#### Paragraph 7–9:

Have there been unapproved modifications or repairs to security containers and vault doors? (Considered a violation of the container's or door's integrity and the GSA label will be removed.) If so, has the GSA label been removed?

#### Paragraph 7–10:

Have MACOMs established procedures concerning repair and maintenance of classified material security containers, vaults, and secure rooms, to include a schedule for periodic maintenance?

#### Paragraph 7–11:

Is security equipment inspected before turn-in or transfer to ensure that classified material is not left in the container?

#### Paragraph 7–21:

Do new purchases of combination locks, for GSA-approved security containers, vault doors and secure rooms, conform to Federal Specification FF-L-2740A?

## Chapter 8 Transmission & Transportation

Paragraph 8–1:

Have Commands established local procedures to meet the minimum requirements to minimize risk of compromise while permitting use of the most effective transmission or transportation means?

Paragraph 8–2:

Is TOP SECRET information transmitted only as outlined in paragraph 8-2?

Paragraph 8–3:

Is SECRET information transmitted only as outlined in paragraph 8-3?

Paragraph 8–4:

Is CONFIDENTIAL information transmitted only as outlined in paragraph 8-4?

Paragraph 8–5:

Is NATO Restricted Information transmitted only as outlined in paragraph 8-5?

Paragraph 8-6:

Is classified information or material approved for release to a foreign government in accordance with AR 380–10? Will it be transferred only between authorized representatives of each government in compliance with the provisions of Chapter 8, of this regulation?

#### Paragraph 8–8:

Where applicable, have Commands established procedures for shipment of bulk classified material as freight, to include provisions for shipment in closed vehicles when required, appropriate notice to the consignee concerning the shipment, procedures at transshipment activities, and action to be taken in case of non-delivery or unexpected delay in delivery?

#### Paragraph 8–9:

When classified information is transmitted, is it enclosed in two opaque, sealed envelopes, wrappings, or containers, durable enough to properly protect the material from accidental exposure and to ease in detecting tampering, except where exempted by paragraph 8–9?

#### Paragraph 8–10:

*a*. Is the outer envelope or container for classified material addressed to an official government activity or to a DOD contractor with a facility clearance and appropriate storage capability?

*b*. Does the inner envelope or container show the address of the receiving activity, the address of the sender, the highest classification of the contents, including, where appropriate, any special markings, and any other special instructions?

c. Is the requirement that the outer envelope or single container not bear a classification marking or any other unusual marks that might invite special attention to the fact that the contents are classified strictly followed?

Paragraph 8–12:

Is handcarrying of classified material limited to situations of absolute necessity and carried out to make sure it does not pose an unacceptable risk to the information, IAW the provisions of paragraph 8–12, of this regulation?

#### Paragraph 8-13:

*a.* Do responsible officials provide a written statement to all individuals escorting or carrying classified material authorizing such transmission?

b. Do travelers who are authorized to carry classified material on international flights, or by surface conveyance if crossing international borders, have courier orders?

#### Paragraph 8–14:

*a*. When classified material is handcarried for delivery to a foreign government representative, or when classified information is discussed with or otherwise disclosed to foreign national personnel, are the requirements of AR 380–10 strictly followed?

b. The DOD requires that a request for travel outside the United States contain a written statement by the traveler that classified information will or will not, as applicable, be disclosed during the trip. If the foreign disclosure of classified information is involved, is there an additional written statement that disclosure authorization has been obtained in accordance with DOD Directive 5230.11?

#### Paragraph 8–15:

*a*. Is the individual designated as courier in possession of a DOD or contractor-issued identification card that includes a photograph, descriptive data, and signature of the individual? (If the identification card does not contain date of birth, height, weight, and signature, these items must be included in the written authorization.)

b. Does the courier have the original of the authorization letter since a reproduced copy is not acceptable? (The traveler will have sufficient authenticated copies to provide a copy to each airline involved.)

#### **Chapter 9 Security Education**

#### Paragraph 9–1:

Has the Commander established a Security Education Program?

#### Paragraph 9–3:

Have all personnel who could be expected to play a role in the Information Security Program been given an initial orientation?

#### Paragraph 9-4:

a. As a minimum, does the Information Security Program include the following points:

(1) The nature of U.S. and foreign government classified information, its importance to the national security, and the degree of damage associated with each level of classification?

(2) How to recognize U.S. and foreign government classified information that personnel may encounter, including markings, etc.?

(3) The individual's responsibility for protection of classified information, and the consequences of failing to do so?

- (4) Procedures and criteria for authorizing access to classified information?
- (5) Procedures for safeguarding and control of classified information in the individual's work environment?

(6) Proper reaction to discovery of information believed to be classified in the public media?

(7) The security management and support structure within the command, to include sources of help with security problems and questions and proper procedures for challenging classifications believed to be improper?

(8) Penalties associated with careless handling or compromise of classified information?

b. Before being granted access to classified information, have all employees signed a Standard Form 312?

#### Paragraph 9–6:

Are personnel who are not cleared for access to classified information included in the security education program if they will be working in situations where inadvertent access to classified information might occur or they will have access to unclassified information which might be of value to intelligence collectors?

#### Paragraph 9–7:

As a minimum, are DA employees who have access to, create, process, or handle classified information provided refresher training in their responsibilities at least once a year?
### Paragraph 9–9:

Are DA personnel, who are in positions which require performance of specified roles in the Information Security Program, provided security education sufficient to permit quality performance of those duties? Is the training provided before, concurrent with, or not later than six months following assumption of those positions?

### Paragraph 9–10:

Are officials who have been granted original classification authority educated in their responsibilities before they exercise the delegated authority?

### Paragraph 9–11:

Are all DA personnel, whose responsibilities include derivative classification, trained in requirements and procedures appropriate to the information and material they will be classifying, to include the proper use of classification guides and source documents?

### Paragraph 9–12:

Are Command Security Managers, security staff members, and others with significant responsibility for management of the Information Security Program, trained and educated to fulfill their roles?

### Paragraph 9–14:

Does Command include in their security education programs, either in the general program or as part of special briefings to select personnel affected, provisions regarding special education and training for personnel who:

a. Use automated information systems to store, process, or transmit classified information?

*b*. Will be traveling to foreign countries where special concerns about possible exploitation exist or will be attending professional meetings or conferences where foreign attendance is likely?

- c. Will be escorting, handcarrying, or serving as a courier for classified material?
- d. Are authorized access to classified information requiring special control or safeguarding measures?
- e. Are involved with international programs?

f. Regardless of clearance and/or access level held, do all DA personnel receive SAEDA training, at a minimum of every two years, pursuant to Interim Change IO1, AR 381–12?

### Paragraph 9–16:

a. Do DA Commanders ensure that security education programs are appropriately evaluated during self-inspections and during oversight activities of subordinate commands or organizational units?

b. Do Commands maintain a record of the programs offered and of the personnel that participated? Are these records maintained for two years and available for review during oversight inspections and assistance visits?

# Chapter 10 Unauthorized Disclosure & Other Security Incidents

### Paragraph 10–2:

Are personnel aware of their responsibilities in the event of an actual or possible compromise or loss of classified information or material?

### Paragraph 10–3:

a. When an incident of possible loss or compromise of classified information is reported, does the Command immediately initiate a preliminary inquiry into the incident?

*b*. Does the person appointed to conduct the preliminary inquiry have the appropriate security clearance, the ability and available resources to conduct an effective inquiry, and is not likely to have been involved, directly or indirectly, in the incident?

c. In cases of apparent loss of classified material, has the person conducting the preliminary inquiry ensured that a thorough search for the material has been conducted, and has documented the steps taken to locate the material?

d. Does the preliminary inquiry focus on answering all the basic interrogatives (who, what, where, why, how, when)?

e. As a result of the preliminary inquiry, has one of the following alternatives been chosen?

(1) Compromise of classified information did not occur.

(2) Compromise of classified information may have occurred.

(3) Compromise of classified information did occur, but there is no reasonable possibility of damage to the national security.

(4) Compromise of classified information did occur and damage to the national security may result.

*f*. If at any time, during the preliminary inquiry, it appears that deliberate compromise of classified information may have occurred, has the situation been immediately reported to the chain of command and supporting counterintelligence unit?

g. Have apparent violations of other criminal law been reported to the supporting criminal investigative activity?

#### Paragraph 10-5:

When notified of possible or actual compromise, has the holder of that information or material ensured that the original classification authority, responsible for each item of information, is notified of the incident?

### Paragraph 10-6:

In cases where a person has had unauthorized access to classified information, has the person been debriefed to enhance the probability that he/she will properly protect it?

### Paragraph 10-7:

*a*. Have Department of the Army Commands, and especially MACOMs, established necessary reporting and oversight mechanisms, to make sure that inquiries are conducted when required, that they are done in a timely and effective manner, and that appropriate management action is taken to correct identified problem areas?

*b*. Have Commands, and especially MACOMs, established a system of controls and procedures to make sure that reports of security inquiries and damage assessments are conducted, when required, and that their results are available as needed?

Paragraph 10–9:

When an individual who has had access to classified information is absent without authorization, commits or attempts to commit suicide, or is temporarily or permanently incapacitated, has the command inquired into the situation to see if there are indications of activities, behavior, or associations, that could indicate classified information might be at risk?

### F-4. Comments

This appendix is designed to assist you in evaluating your management controls. Please submit any comments and/or suggestions to the following address: DEPARTMENT OF THE ARMY DAMI-CHS, 2511 Jefferson Davis Highway, Suite #9300, Arlington, VA 22202–3910

# Section II Recordkeeping Requirements

### F-5. Updated requirements

The matrix below (table F-1) shows file titles and dispositions for records created and maintained under the purview of this regulation. The matrix will assist users to determine how long to keep the records in the current file areas (CFA) and when to transfer them to the Record Holding Area (RHA) or destroy them. The chart at table F-2 gives an expanded description of which files fall under what MARKS number.

### Table F–1 File Titles and Dispositions for Records

	•	1							1		
						Orgar	nizational	Level			
MARKS Number	File Title	NARA Authority	Privacy Act Systems Notice	A	в	с	D	E	F	G	
380	General security corre- spondence files (no longer needed for current operations)	NN-167 NN-165-192	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5a	Security briefings and debriefings (transfer or separation of person)	NC1-AU-78-116	A0380-67-DAMI	KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5b	Security inspections and surveys (next compara- ble inspection or survey)	NN-173-72		KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5d	Classified material ac- cess files (transfer, reas- signment, or separation of the person, or when obsolete)	NN-166-204	OPM/GOVT-1	KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5g	Classified Information Nondisclosure Agree- ment (NDA) maintained separately from the indi- vidual's official personnel folder	GRS 18–25	A0380-67-DAMI	T75	T75	T75	T75	T75	T75	T75	
380–5g	Classified Information Nondisclosure Agree- ment (NDA) maintained separately from the indi- vidual's official personnel folder	GRS 18–25	A0380-67-DAMI	T75	T75	T75	T75	T75	T75	T75	
380–5j	TOP SECRET document records (related docu- ment is downgraded, transferred, or destroyed)	GRS 18–5b	A0001DAMI	KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5k	Security classification files – Offices in charge of issuance (after final declassification action)	NC1-AU-78-117	N/A	TE10	TE10	TE10	TE10	TE10	TE10	TE10	
380–5k	Security classification files – Offices in charge of issuance (after final declassification action)	NC1-AU-78-117	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5n	Office non-registered classified document de- struction certificates	NC1-AU-79-27	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5r	Security information ex- changes	NC1-AU-78-117	N/A	T20	T20	T20	T20	T20	T20	T20	
380–5s	Security compromise cases (completion of final corrective or disciplinary action)	NC1-AU-330-76-1	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6	
380–5s	Security compromise cases: records of viola- tions of a sufficiently seri- ous nature to be classi- fied as felonies (comple- tion of final corrective or disciplinary action)	NC1-AU-330-76-1	N/A	TEP	TEP	TEP	TEP	TEP	TEP	TEP	
380–5u	Security equipment files (termination of the exception)	NN-166-204	N/A	TE10	TE10	TE10	TE10	TE10	TE10	TE10	

# Table F–1 File Titles and Dispositions for Records—Continued

				Organizational Level						
MARKS Number	File Title	NARA Authority	Privacy Act Systems Notice	A	в	с	D	E	F	G
380–5w	Security regrading cases – Offices in charge of regrading	NC1-AU-78-117	N/A	T15	T15	T15	T15	T15	T15	T15
380–5w	Security regrading cases – Other offices	NC1-AU-78-117	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6
380–5x	Security information ac- cess cases – Offices having Army-wide re- sponsibility	NC1-AU-78-117	A0380-67-DAMI	T20	T20	T20	T20	T20	T20	T20
380–5x	Security information ac- cess cases – Other of- fices	NC1-AU-78-117	A0380-67-DAMI	KE6	KE6	KE6	KE6	KE6	KE6	KE6
380–5y	Security information re- leases – Offices having Army–wide responsibility	NCI-AU-78-117	N/A	T20	T20	T20	T20	T20	T20	T20
380–5y	Security information re- leases – Other offices	NCI-AU-78-117	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6
380–5z	Non-cryptographic regis- tered document accounts (superseded by a new report if all information referred to is accounted for either by a report of destruction or inclusion in the new report)	II-NNA-1002	A0001DAMI	KE6	KE6	KE6	KE6	KE6	KE6	KE6
380–5aa	TOP SECRET material accountability (after all items on each page have been destroyed, down- graded, dispatched, or when entries are trans- ferred to a new page)	GRS 18–5a	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6
380–5bb	Industrial information se- curity	NC1-AU-83-28	N/A	TP	TP	TP	TP	TP	TP	TP
380–5dd	Activity Entry/Exit Inspec- tion Program	GRS 18–8	N/A	K6	K6	K6	K6	K6	K6	K6
380–150d	Atomic information ex- changes	NC1-AU-76-51	N/A	K6	K6	K6	K6	K6	K6	K6
380–381j	SAP security manage- ment files: Interim secu- rity classification guides – sponsoring program/ac- tivity (after supersession or final declassification action)	N1-AU-92-1	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6
380–381j	SAP security manage- ment files: Final security classification guides – Offices with Army-wide responsibility	N1-AU-92-1	N/A							
380–381j	SAP security manage- ment files: Final security classification guides – Other offices	N1-AU-92-1	N/A	KE6	KE6	KE6	KE6	KE6	KE6	KE6

Code: Organizational Level: A = Company and below B = Battalion C = Brigade

#### Table F-1

#### File Titles and Dispositions for Records—Continued

				Organizational Level						
MARKS		NARA	Privacy Act		1	•	_	_	_	
Number	File Title	Authority	Systems Notice	A	В	С	D	E	F	G

D = Division, Installation

E = Corps, Major Subordinate Command

F = Individual Army, Major Command

G = HQDA, Secretariat

Disposition Instruction:

K6 = Keep until no longer needed for conducting business, but not longer than 6 years.

KE6 = After a specific event occurs, keep until no longer needed for conducting business, but not longer than 6 years.

T (followed by a numeral) = Transfer to records holding area when no longer required for conducting business. The time period specified is when the Records Holding Area (RHA) will destroy the record, based on the creation date.

TE (or TE followed by a numeral) = Transfer to the RHA in accordance with event instructions. The time period shown is when the RHA will destroy the record, based on the event date.

TP or TEP = Transfer to Records Holding Area when no longer required for conducting business. The P indicates that these area permanent records. They will be transferred to the National Archives within 30 years in accordance with the governing records retention schedule.

NOTE: Records requiring transfer to the RHA will be prepared and transferred by deployed units monthly and at the end of the deployment. All other units will prepare and transfer annually by fiscal or calendar year. See your Installation Records Manager for instructions on transferring records.

### Table F-2

File Numbers and Descriptions for Records

FN: 380-5a

Title: Security briefings and debriefings

Authority: NC1-AU-78-116

Privacy Act: A0380-67DAMI

Description: Information on classified material, to include Sensitive Compartmented Information (SCI), and security briefing and debriefing of personnel authorized access to classified material. Included are briefing statements, debriefing statements and certificates, and related information.

Disposition: Destroy 2 years after transfer or separation of person.

FN: 380–5b

Title: Security inspections and surveys Authority: NN–173–72

Privacy Act: Not applicable

Description: Information on inspections and surveys that are conducted by security officers. This includes SCI security inspections and surveys, and routine after-duty-hours security inspections conducted under the supervision of a security officer to decide the adequacy of measures taken to protect security information against the hazards of fire, explosion, sabotage, and unauthorized access. Included are reports, recommendations, and related information.

Disposition: Destroy after next comparable inspection or survey.

FN: 380–5d

Title: Classified material access files Authority: NN-166-204 Privacy Act: OPM/GOVT-1

Description: Information showing authorization to have access to classified files. This includes forms containing person's name and signature, classification of files concerned, information desired, signature of an official authorizing access, and similar data. Disposition: Destroy on transfer, reassignment, or separation of the person, or when obsolete.

FN: 380-5g

Title: Classified Information Nondisclosure Agreement (NDA)

Authority: GRS 18, Item 25

Privacy Act: A0380-67DAMI

Description: Copies of nondisclosure agreements; such as SF 312 or SF 189 and DD Form 1847–1 (Sensitive Compartmented Information Nondisclosure Statement/SCI NDA or similar forms; signed by military or civilian personnel including employees of contractors, licensees, or grantees with access to information that is classified under standards put forth by Executive Orders governing security classification. These forms should be maintained separately from personnel security clearance files. Agreements for civilian employees working for elements of the intelligence community must be maintained separately from the official personnel folder. For all other persons, these forms may be filed in the individual's official military personnel folder (for uniformed military personnel) or on the right side of the official personnel folder (for civilian employees).

Disposition: a. If maintained separately from the individual's official personnel folder. Destroy when 70 years old. b. If maintained in the individual's official personnel folder. Apply the disposition for the official personnel folder.

FN: 380–5j Title: TOP SECRET document records Authority: GRS 18, Item 5b Privacy Act: A0001DAMI

### Table F-2

File Numbers and Descriptions for Records—Continued Description: Information used to record the names of persons having had access to TOP SECRET information and copies of extracts distributed. Disposition: Destroy when related document is downgraded, transferred, or destroyed. FN: 380-5k Title: Security classification files Authority: NC1-AU-78-117 Privacy Act: Not applicable Description: Information relating to the security classification or grading system involving the classification or downgrading of information. Included are correspondence or memorandums and reports on security classification. It excludes other files described in this record series. Disposition: a. Offices in charge of issuance: Destroy 10 years after final declassification action. b. Other offices and TOE units: Destroy after 3 years. FN: 380-5n Title: Office nonregistered classified document destruction certificates Authority: NC1-AU-79-27 Privacy Act: Not applicable Description: Forms and other types of information that show the destruction of classified information. Disposition: Destroy after 2 years, or earlier when approved by HQDA (DAMI-CIS) WASH DC 20310. FN: 380-5r Title: Security information exchanges Authority: NC1-AU-78-117 Privacy Act: Not applicable Description: Information on the exchange of security classified information with other Government agencies, industry, and foreign governments. Included are correspondence on the exchange of information, exchange agreements, and related information. Disposition: Destroy after 20 years. FN: 380-5s Title: Security compromise cases Authority: NC1-330-76-1 Privacy Act: Not applicable Description: Information on investigations of alleged security violations. Included are SCI security violations such as missing information, unauthorized disclosure of information, unattended open security containers, and information not properly safeguarded. Disposition: Destroy 2 years after completion of final corrective or disciplinary action. Records of violations of a sufficiently serious nature to be classified as felonies are permanent. FN: 380-5u Title: Security equipment files Authority: NN-166-204 Privacy Act: Not applicable Description: Information gathered for the determination of uses and types of security equipment for protecting classified information and materials. They include safes, alarm systems, and other security equipment. Disposition: Destroy after 10 years. Destroy exceptions to storage standards 10 years after termination of the exception. FN: 380-5w Title: Security regrading cases Authority: NC1-AU-78-117 Privacy Act: Not applicable Description: Information on the review of specific classified information or equipment for the purpose of regrading the information or equipment. Disposition: a. Offices in charge of regrading: Destroy after 15 years. b. Other offices: Destroy after 3 years. FN: 380-5x Title: Security information access cases Authority: NC1-AU-78-117 Privacy Act: A0380-67DAMI Description: Information on the review of specific requests for access to classified files or equipment for purposes of research and study. Disposition: a. Offices having Army-wide responsibility: Destroy after 20 years. b. Other offices: Destroy after 2 years. FN: 380-5v Title: Security information releases Authority: NCI-AU-78-117 Privacy Act: Not applicable Description: Information on the review of classified or potentially classified documentary materials for dissemination of information to sources outside the Army, such as the review of manuscripts, photography, lectures, radio, and television scripts, and other materials. Disposition: a. Offices having Army-wide responsibility: Destroy after 20 years. b. Other offices: Destroy after 2 years.

FN: 380–5z Title: Noncryptographic registered document accounts Authority: II–NNA–1002 Privacy Act: A0001DAMI File Numbers and Descriptions for Records—Continued Description: Information showing the accountability of non-Army noncryptographic registered information. Included are semiannual inventory or report of transfer, transfer reports, and similar information. Disposition: Destroy when superseded by a new report if all information referred to is accounted for either by a report of destruction or inclusion in the new report. FN: 380-5aa Title: TOP SECRET material accountability Authority: GRS 18, Item 5a Privacy Act: A0001DAMI Description: Information showing the identity, receipt, dispatch, downgrading, source, movement from one office to another, destruction, and current custodian of all TOP SECRET material for which the TOP SECRET control office is responsible. Disposition: Destroy 5 years after all items on each page have been destroyed, downgraded, dispatched, or when entries are transferred to a new page. FN: 380-5bb Title: Industrial information security Authority: NC1-AU-83-28 Privacy Act: Not applicable Description: Information on the protection of classified information in the possession of industry, including information on the issuance of clearance certificates, and related information. Disposition: Permanent. FN: 380-5dd Title: Activity Entry and Exit Inspection Program Authority: GRS 18, Item 8 Privacy Act: Not applicable Description: Documents collected at MACOMs, SUBMACOMs, and Staff agencies in the conduct of an inspection program to deter and detect unauthorized introduction or removal of classified material from DOD owned or leased installations and facilities. Included area. The date(s) and number of entry and exit inspections conducted by the activity and subordinate elements during the previous quarter. b. The number of instances during the quarter when persons handcarried classified information without apparent authorization. c. Problems encountered in the conduct of the entry and exit inspection program. Note: Use FN 380-5s to file information on investigations on alleged security violations. Disposition: Cut off at end of calendar year. Destroy 2 years after cut off.

FN: 380–150d Title: Atomic information exchanges Authority: NC1–AU–76–51 Privacy Act: Not applicable Description: Information relating to requests from various foreign governments or allies for atomic information, to include requests for regulations, manuals, reports, and other related information. Disposition: Destroy after 1 year.

# Appendix G Security Classification Guide Preparation

# G-1. Preparation guidance

a. This appendix discusses preparation of security classification guides. Due to the wide variety of systems, plans, and projects for which guides must be published, this appendix provides very general guidance only. The effort must be tailored to fit the specific nature of the guide subject and the classification guidance which must be provided.

*b.* This appendix supplements DOD Directive 5200.1–H. DOD Directive 5200.1–H has not been revised, as of yet, to include new classification criteria and marking requirements of Executive Order 12958. But it still provides useful guidance on preparing classification guides.

c. The question of classifying guides themselves requires careful consideration.

(1) Guides should be published in unclassified form if possible; however, classified guides may be published in classified form if necessary. To avoid classifying a guide, it is sometimes possible to include all necessary classified information in a classified supplement. This is also a good method of dealing with Special Access Programs (SAPs) information. See appendix I, of this regulation and AR 380–381 for more information.

(2) Preparers of guides must be careful that descriptions of classified information in the guide do not inadvertently disclose classified information.

(3) Like any other classified document, classified guides must be portion marked and marked with the identity of the classifier and declassification instructions.

d. Section II provides instructions on changing and reissuing guides.

e. Section III contains instructions for preparing and submitting DD Form 2024. These instructions supplement those found on the reverse of the form.

*f*. Figure G–1 is a reprint of DODD 5200.1–H. Note: Except for sections 1, 2, 6, and 7, the sample guide contains only a sampling of topics that normally would be included. Comments are enclosed in parentheses. References are to DODD5200.1–R, tailored as best as possible to this regulation.

### G-2. Procedures for changing security classification guides

*a*. Whenever a security classification involving a guide subject changes, that change must be reflected by revision or reissuance of the guide. It may also be necessary to clarify or correct information in the guide or provide additional information or instructions.

b. The terms "revision" and "reissuance" have specific meanings which must be understood because they are used differently on DD Form 2024.

(1) The term "revision" includes the following three types of actions:

(a) A change, which actually modifies some provision(s) of a guide (e.g., a classification, declassification or review date, the description of an item of information, a note).

(b) An errata sheet, which corrects an error (typographical or otherwise) in a guide.

(c) An addendum, which adds new material to a guide. A single revision may serve more than one of these purposes (e.g., add new material and correct an error). A revision may be titled "change," "errata sheet," or "addendum," but in the interests of clarity the term "revision" is recommended. Each revision should bear an identifying number. Revisions may be pen–and–ink changes or page changes whichever is more efficient.

(2) Revisions will show the date of approval.

c. The term "reissuance" applies only to republication of an entire guide to incorporate modifications. The date of the reissuade guide will be the date of approval of the reissuance. The original issue of the guide or the latest prior reissuance (if any) will be shown under "Supersessions."

d. Revisions and reissued guides must be approved, distributed, and reported as required in section 5 of DODD 5200.1-R.

### G–3. Instructions for preparing DD Form 2024

*a*. Submission of DD Form 2024, required by paragraph 2–18d of this regulation, allows the listing of a security classification guide in DODD 5200.1–I.

*b.* DD Form 2024 must be prepared at least in four copies. One copy is to be retained with the record copy of the SCG. In the case of new guides, revisions, or reissuances, the DD Form 2024 should accompany the copies of the SCG submitted to the Administrator, Defense Technical Information Center (DTIC).

c. The correct edition date for DD Form 2024 is July 1986. Previous editions are obsolete and should not be used.

d. The reverse side of DD Form 2024 provides instructions for completing the form. The following guidance supplements and clarifies these instructions.

(1) Paragraph B of the instructions defines the meanings of the six blocks in Item 1 of the form.

(a) The "new guide" block should be checked when reporting any of the following:

(1) The issue of a guide covering a subject on which no previous guide has been issued.

(2) Reissue of a guide which is not currently indexed in DODD 5200.1-I.

(b) Do not check the "new guide" block just because the title of a guide has been changed. The "revision" block should be checked when the form is prepared to reflect a partial change (errata sheet, addendum, page change) to an SCG. The "reissuance" block should be checked when the entire guide is republished to include changes. The "correction" block is used when DD Form 2024 is submitted to correct information entered on a previous DD Form 2024, not to correct information in the SCG itself.

(2) Item 2 of the form should show only numbered publications which contain or transmit SCGs (for example, ARs, TBs, LOIs, major command or local regulations). If the guide is separately published, letters of transmittal, etc., should not be listed; "NONE" should be entered in Item 2.

(3) Never complete Item 3 by entering "Security Classification Guide for..." The item or project name should be entered as it appears on the guide; the phrases "Security Classification Guide" or "Classification Guidance for" are unnecessary.

(4) Item 4 must contain a date as shown in paragraph D on the back of the form. For a new guide, enter the date of its approval. When reporting a reissuance, enter the date of approval of the reissued guide. For other submissions, enter the date of the latest issue of the guide.

(5) In Item 7, enter a date two years from the date of issue, reissue, or the last review, whichever is latest.

(6) When completing Item 8, enter "NONE" if no revisions to the SCG have been published. Record revisions

according to this example: A guide was issued on 5 May 76. A revision was issued on 12 Jul 77; the DD Form 2024 for the revision showed "01770712" in Item 8. A second revision was issued on 20 Nov 78; Item 8 of that DD Form 2024 read "02781120."

(7) Item 9, "Subject Matter Index Terms," is an extremely important portion of the form, since the accuracy and validity of these terms determine the usefulness of DODD 5200.1–I. Carefully study the list of index terms found in DODD 5200.1–I and select terms that best apply to the SCG subject. Within the Department of the Army, only terms listed in the current edition of the index will be used. Other terms may not be used without the permission of the Directorate of Counterintelligence and Security Countermeasures, ODSCINT, HQDA. If none of the current terms are appropriate, submit recommended additions to Headquarters Department of the Army, DCSINT, ATTN: DAMI–CH, 2511 Jefferson Davis Hwy, Suite 9300, Arlington, VA 22202–3910. Ensure that the index terms do not disclose classified information about the SCG subject.

(8) The index sequence number entered in Item 11 is essential to the accuracy of DODD 5200.1–I. If DD Form 2024 reflects a revision, reissuance, biennial review, cancellation, or correction, the index sequence number following the SCG title in the current edition of DODD 5200.1–I must be entered in Item 11. Two special circumstances deserve mention:

(a) If a guide is to be reissued with a new title (due to a change in equipment nomenclature or project title), the index sequence number of the old guide must be entered in Item 11. The computer program used to compile the index will key on that number, delete the old title, and replace it with the new title. Do not submit a DD Form 2024 to list it under its new title. This might cause the SCG to be deleted from the index entirely.

(b) In a very few cases, a revision or correction to a recently issued SCG which has not yet been listed in DODD 5200.1-I will be issued. In such cases, leave item 11 blank.

(9) Item 14, "Remarks," may be used to advise the recipients of DD Form 2024 of any additional information considered appropriate. Information placed in Item 14 will not appear in DODD 5200.1–I. Item 14 may be left blank.

(10) DD Form 2024 states that completing Item 16a, "Action Officer," is optional. Within the Department of the Army, an action officer will be indicated in Item 16a each time DD Form 2024 is prepared. The person listed should be the current action officer for the SCG.

*e*. Sometimes an SCG will be replaced by two or more SCGs dealing with individual projects or items of equipment. For example, an SCG dealing with aircraft survivability equipment was replaced with separate SCGs for each item of equipment. In other cases, two or more SCGs may be combined. For example, SCGs covering two pieces of equipment may be replaced by one SCG for a system in which both items are included. When this occurs, the proponent of each superseded SCG must submit a DD Form 2024 to cancel the old SCG. A statement in Item 14 of the DD Form(s) 2024 for the new SCG(s) will not cancel the superseded SCG(s).

*f*. When the proponent of a SCG changes due to organizational changes or progress in system development, the following instructions apply:

(1) If responsibility is transferred from one U.S. Army element to another, a "correction" or "cancellation" DD Form 2024 will not be submitted. The change of proponent should not be reported until the SCG is revised or reissued by the new proponent.

(2) If responsibility is assumed by a U.S. Army element from an agency outside the U.S. Army, the change should be reflected when the gaining activity revises the SCG. The "new guide" block in Item 1 of the DD Form 2024 will be checked; no index sequence number will be entered. Canceling the old SCG is the responsibility of the former proponent.

(3) If responsibility is assumed by a non-Army agency from a U.S. Army element, the losing element should make sure they know when the new proponent republishes the SCG. The losing U.S. Army element is responsible for submitting a "cancellation" DD Form 2024 when, but not before, the gaining agency republishes the SCG.

DOD Directive 5200.1-H

DEPARTMENT OF DEFENSE

HANDBOOK FOR WRITING

SECURITY CLASSIFICATION GUIDANCE

1998

THE ASSISTANT SECRETARY OF DEFENSE FOR

COMMAND, CONTROL, COMMUNICATIONS, AND

INTELLIGENCE

FOREWORD

This Handbook is issued under the authority of DoD Directive 5200.1, "DoD Information Security Program, " December 13, 1996. Its purpose is to assist in the development of the security classification guidance required under paragraph 2-500 of DoD Regulation 5200.1-R, for each system, plan, program, or project in which classified information is involved.

DoD 5200.1-H, "Department of Defense Handbook for Writing Security Classification Guidance," March 18, 1986, is hereby canceled.

This Handbook is effective immediately.

Users of this Handbook are encouraged to direct comments to the Director, Security; Office of the Deputy Assistant Secretary of Defense (Security and Information Operations), 6000 Defense, The Pentagon, Washington, DC 20301-6000.

//signed//

Christopher K. Mellon

Deputy Assistant Secretary of Defense

(Security and Information Operations)

Figure G-1. DOD Directive 5200.1–H

Distribution of this Handbook is authorized to U.S. Government Agencies and their contractors. Administrative or Operational Use, August 1992 Other requests for this document shall be referred to the Security Programs Directorate, Office of the Deputy Assistant Secretary of Defense (Security and Information Operations), Washington, DC 20301-6000

# REFERENCES

(a) Executive Order 12958, "Classified National Security Information," April 20, 1995

(b) Information Security Oversight Office Directive No. I, "Classified National Security Information," October 13,1995

(c) DoD Regulation 5200.1-R, "Information Security Program," January 14, 1997,

authorized by DoD Directive 5200.1, December 16, 1996

(d) DoD Regulation 5400.7-R, "DOD Freedom of Information Act Program," May 1997,

authorized by DoD Directive 5400.7, "DoD Information Act Program," September 1997

(e) Deputy Secretary of Defense Memorandum, "Web Site Administration," December

1998

Figure G-1. DOD Directive 5200.1–H—Continued

### C.1. CHAPTER 1 INTRODUCTION

# C.1.1.

Good security classification practice in an organization as large and widespread as the Department of Defense, calls for the timely issuance of comprehensive guidance regarding security classification of information concerning any system, plan, program, or project; the unauthorized disclosure of which reasonably could be expected to cause damage to the national security. Precise classification guidance is prerequisite to effective and efficient information security and can do much to assure that security resources are expended to protect only that which truly warrants protection in the interests of national security. Executive Order 12958 (reference (a)) and its implementing Information Security Oversight Office Directive No. 1 (reference (b)), provide general requirements and standards concerning the issuance of security classification guides.

# C.1.2.

Information is classified to assist in ensuring that it is provided an appropriate level of protection. Therefore, it is essential that a classification guide be concerned with identifying the specific items of information and the level of protection required, as well as the time period for which protection must be continued.

# C.1.3.

A classification guide should be issued as early as practical in the life cycle of the classified system, plan, program or project. Any uncertainty in application of the policies and procedures contained in DOD Regulation 5200.1–R, "Information Security Program," (reference (c)), which implements the provisions of reference (a) and (b) within DOD, will result in a less than satisfactory security classification guide. Accordingly, the requirements of DOD 5200.1–R regarding classification, declassification, downgrading, marking, and security classification guides should be reviewed and understood before proceeding with the task of writing a security classification guide.

# C.1.4.

DOD information that does not, individually or in compilation, qualify for classification, must be reviewed in accordance with DOD Regulation 5400.7 (reference (d)), prior to its release outside DOD. In addition, such information must also be reviewed for compliance with the provisions of Deputy Secretary of Defense Memorandum, dated December 7, 1998 (reference (e)), prior to its placement on any publicly accessible DOD web site.

# C.2. Chapter 2

CLASSIFICATION AND DECLASSIFICATION

# C.2.1. GENERAL

Since the primary purpose of this Handbook is to provide assistance to those who are responsible for the writing of a security classification guide, some discussion of classification and declassification principles is warranted.

# C.2.2. CLASSIFICATION

2.2.1 Basically, information is classified in one of two ways, either originally or derivatively. Original classification occurs when information is developed which intrinsically meets the criteria for classification under Executive Order 12958 (reference (a)) and such classification cannot reasonably be derived from a previous classification still in force involving in substance, the same or closely related information. A security classification guide is, in effect, the written record of an original classification decision or series of decisions regarding a system, plan, program, or project. Derivative classification occurs when the information under consideration fits the description of information already known to be classified.

2.2.2 Classification may be applied only to information that is owned by, produced by or for, or is under the control of the United States Government. Information may be considered for classification only if it concerns one of the categories specified in Section 1.5a of Executive Order 12958 (reference (a)):

2.2.2.1 Military plans, weapon systems, or operations;

2.2.2.2 Foreign government information;

2.2.2.3 Intelligence activities (including special activities), intelligence sources or methods, or cryptology;

2.2.2.4 Foreign relations or foreign activities of the United States, including confidential sources;

2.2.2.5 Scientific, technological, or economic matters relating to the national security;

2.2.2.6 United States Government programs for safeguarding nuclear materials or facilities; or

2.2.2.7 Vulnerabilities or capabilities of systems, installations, projects or plans relating to the national security.

2.2.3 An original classification authority is confronted with the need to decide whether certain information should be classified. To make this determination there are a number of steps to follow. These steps may be laid out as a series of questions.

2.2.3.1 Is the information owned by, produced by or for, or under the control of the United States Government? 2.2.3.2 Does the information fall within one or more of the several categories of information in paragraph C.2.2.1 through C.2.2.7 above? If the answer to this question is "no," the information cannot be classified. If the answer is "yes," then the next question applies.

2.2.3.3 Can the unauthorized disclosure of the information reasonably be expected to cause damage to the national security? If the answer is "no," the information cannot be classified. If the answer is "yes," then the third question applies.

2.2.3.4 What is the degree of damage to the national security that is expected in the event of an unauthorized disclosure

of the information? If the answer to this question is "damage" you have arrived at a decision to classify the information Confidential. If the answer is "serious damage," you have arrived at a decision to classify the information Secret. If the answer is "exceptionally grave damage," you have arrived at a decision to classic the information Top Secret.

# C.2.3. Declassification

The declassification decision determines duration of protection, and is as important as the original classification determination. At the time an item of information is classified, original classifiers shall:

2.3.1 Assign a date within ten years from the date of classification upon which the information can be automatically declassified;

2.3.2 Determine a specific event, reasonably expected to occur within years, that can beset as the signal for automatic declassification; or

2.3.3 Designate the information as being automatically declassified on a date ten years from the date of its original classification.

2.3.3 An original classifier may extend classification beyond ten years only if:

2.3.3.1 The unauthorized disclosure of the information could reasonably be expected to cause damage to the national security for a period in excess of 10 years, and

2.3.3.2 Release of the information could reasonably be expected to:

2.3.3.2.1 Reveal an intelligence source, method, or activity, or a cryptologic system or activity;

2.3.3.2.2 Reveal information that would assist in the development or use of weapons of mass destruction;

2.3.3.2.3 Reveal information that would impair the development or use of technology within a United States weapon system;

2.3.3.2.4 Reveal United States military plans, or national security emergency preparedness plans;

2.3.3.2.5 Reveal foreign government information;

2.3.3.2.6 Damage relations between the United States and a foreign government, reveal a confidential source, or seriously undermine diplomatic activities that are reasonably expected to be ongoing for a period greater than 10 years; 2.3.3.2.7 Impair the ability of responsible United States Government officials to protect the President, the Vice President, and other individuals for who protection services are authorized;

2.3.3.2.8 Violate a statute, treaty, or international agreement.

### C.2.4. DOWNGRADING

Executive Order 12958 (reference (a)) does permit an original classifier to provide for downgrading of classification to a lower level at predetermined points in time, or upon the occurrence of specified events. You are encouraged to specify in your guide, downgrading to a lower level of classification when the lower level will provide adequate protection.

# C.3. Chapter 3

A PLAN OF ACTION FOR WRITING CLASSIFICATION GUIDES

# C.3.1. STEP 1. CONSIDER RELATED CURRENT GUIDANCE

# C.3.1.1.

Before the actual writing of a security classification guide begins, it is necessary to find out what, if any, classification guidance already issued is applicable to items of information concerning the system, plan, program or project for which the classification guide is being constructed. Any existing guidance may affect your effort, and should be considered carefully. Uniformity and consistency in the exercise of classification authority, especially in the form of a security classification guide, are essential. Be alert to conflicts between the guide you will be developing and any already approved guide.

# C.3.1.2.

In some fields of interest, guides have been issued that apply to a broad spectrum of activities. Such guides often are issued as DOD Instructions through the DOD Directives System. DOD 5200.1–1 (reference (e)) provides a listing of most guides issued within the Department of Defense. Many of the listed guides are available from the Defense Technical Information Center. Always check reference (e), but be aware that some classification guides are too sensitive to be identified in that document. In addition, there may be other classification guides issued along functional lines by activities outside the Department of Defense that could have a bearing on your effort. Seek the advice of those who have knowledge of classification in the subject area under consideration or in closely related fields. If your activity has an information security specialist, that individual may be a particularly valuable source of advice and assistance.

# C.3.2. STEP 2. DETERMINE STATE OF THE ART STATUS

Reasonable classification determinations cannot be made in the scientific and technical field without analysis of what has been accomplished, and what is being attempted and by whom. Make use of scientific and information services; consult technical and intelligence specialists; obtain whatever assistance is available from any proper source. Learn about the state–of–the–art, the state of development and attainment in the field of work, and what is known and openly published about it, including:

- 3.2.1 The known or published status, foreign and domestic.
- 3.2.2 The known but unpublished (probably classified) status in the United States.
- 3.2.3 The known but unpublished status in friendly and unfriendly countries.
- 3.2.4 The extent of foreign knowledge of the unpublished status in the United States.

# C.3.3. STEP 3. IDENTIFY ADVANTAGE FACTORS

The subject matter of your guide must be looked at as a totality. Decide what it does or seeks to accomplish that will result in a net national advantage. Cover all the values, direct and indirect, accruing or expected to accrue to the United States. In the final analysis, the decision to classify will be related to one or more of the following factors, producing directly or indirectly, the actual or expected net national advantage.

3.3.1 Fact of interest by the U.S. Government in the particular effort as a whole or in specific parts that are being considered or emphasized.

3.3.2 Fact of possession by the United States.

- 3.3.3 Capabilities of the resulting product in terms of quality, quantity, and location.
- 3.3.4 Performance, including operational performance, as it relates to capabilities.
- 3.3.5 Vulnerabilities, weaknesses, countermeasures, and counter-countermeasures.
- 3.3.6 Uniqueness, exclusive knowledge by the United States.
- 3.3.7 Lead time, which is related to state-of-the-art.
- 3.3.8 Surprise, which is related to possession and capability to use.

3.3.9 Specifications, which may be indicative of goals, aims, or achievements.

3.3.10 Manufacturing technology. 3.3.11 Associations with other data or activities.

# C.3.4. STEP 4. MAKE INITIAL CLASSIFICATION DETERMINATION

Making the analyses outlined in sections C.3.2 and C.3.3 above, will lead to conclusions on the ways the effort will result in net national advantage, and hence, what it is that requires classification to protect that advantage. Although at this stage of the guide's preparation you are concerned primarily with information relating to the overall effort, consideration must be given to some of the more particular information or data such as that covering performance capabilities, and possible vulnerabilities and weaknesses. Appendix A has been designed to help in that consideration.

# C.3.5. STEP 5. IDENTIFY SPECIFIC ITEMS OF INFORMATION THAT REQUIRE CLASSIFICATION

3.5.1 The real heart of a classification guide is the identification and enunciation of the specific details of information warranting security protection. Regardless of the size or complexity of the subject matter of the guide, or the level at which the classification guide is issued, there are certain identifiable features of the information that create or contribute to actual or expected national security advantage. There also may be certain items of information that need to be protected to prevent or make it more difficult for hostile forces to develop or apply timely and effective countermeasures. The problem is to identify and state those special features or critical items of information and to decide how and why they are related to the net national advantage. Several substeps to this problem of identification of classifiable details are laid out in appendices B and C. The important thing is that the statements of classification guide. (See chap 4 for a complete discussion on classifying hardware items.)

3.5.2 It is equally important that you specify precisely and clearly the level of classification to be applied to each item of information identified in the guide. Broad guidance such as "U–S" meaning Unclassified to Secret does not provide sufficient instruction to users of the guide, unless you also delineate the exact circumstances under which each level of classification should be applied. The exact circumstances may be supplied in amplifying comments, for example, "Unclassified ("U") when X is not revealed"; "Confidential when X is revealed;" and "Secret when X and Y are revealed." Failure to provide such guidance will result in users of the guide making their own interpretations which may, or may not, be consistent with your intent.

# C.3.6. STEP 6. DETERMINE HOW LONG CLASSIFICATION MUST CONTINUE

3.6.1 Equally important to determinations to classify, are the decision on how long the classification should remain in effect. The following are factors that may influence this decision:

3.6.1.1 At the conceptual stage of a new effort there may be good reason to classify more information about the effort than will be necessary in later phases. Typically, information loses its sensitivity and importance in terms of creating or contributing to the national advantage over time.

3.6.1.2 At certain stages in production, or deployment, it may not be practical or possible to protect certain items of information from disclosure. It is also possible that design improvements may have eliminated exploitable vulnerabilities.

3.6.1.3 Official public releases have a direct affect on the duration of classification.

3.6.2 With these factors in mind, and considering the provisions of Chapter 2.3., proceed with the determination of the appropriate declassification instructions for each item of classified information.

3.6.3 Always look at the possibility of providing for automatic downgrading of the classification that is assigned. Future downgrading is an option that is always open when information is originally classified at "S" or "TS" levels. Consider it carefully in every instance, and provide for downgrading at fixed future points in time when the damage that is expected to result from an unauthorized disclosure will be reduced to a level prescribed for lower classification.

# C.3.7. STEP 7. WRITING THE GUIDE

3.7.1 Having determined exactly what warrants security classification, it is then necessary to set down in clear, precise language, statements describing which items of information require classification. It is also advisable to include items that are unclassified as this assures users of the guide that this information is, in fact, unclassified and was not inadvertently omitted. While there is no mandatory DOD-wide format for security classification guides, the one illustrated in appendix D will be adequate in many applications; consider it first. (Also see appendix E for some format variations.) Place significant words of the guide's title first, for example, "FA-5B Aircraft Security Classification Guide."

3.7.2 There are a number of administrative requirements for security classification guides. Bear in mind that the security classification guide you are writing must:

3.7.2.1 State precisely the specific information elements to be protected.

3.7.2.2 Identify the classification levels "TS," "S," or "C" and any additional control marking such as Restricted Data (RD), Formerly Restricted Data (FRD) or NO FOREIGN DISSEMINATION (NOFORN), that may apply to each element of information, or when it will serve a useful purpose, specify that the information is unclassified.

3.7.2.3 Identify the reason for classification.

3.7.2.4 Specify the duration of classification for each element of information (except RD and FRD). RD and FRD is subject to the provisions of the Atomic Energy Act, therefore, no declassification determination should be entered for this information.

3.7.2.5 State any downgrading action that is to occur, and when such action is to take place.

3.7.2.6 Identify the original classification authority who personally approved the guide in writing, and who has program or supervisory responsibility over the information addressed in the guide as well as the Office of Primary Responsibility that can be contacted for clarification or additional information.

3.7.2.7 Include amplifying comments whenever appropriate to explain the exact application of classification.

# C.4. Chapter 4

CLASSIFYING HARDWARE ITEMS

# C.4.1. GENERAL

A piece of hardware may convey information which is every bit as sensitive as words printed upon a piece of paper.

# C.4.2. BASIC CONSIDERATIONS

Hardware items may be classified if they reveal information or information can be obtained from them. The following are some basic considerations:

4.2.1 An item of hardware does not necessarily need to be classified simply because it is part of a classified product or effort.

4.2.2 Unclassified off-the-shelf items, unless modified in some particular way to make them perform differently, can never be classified even though they constitute a critical element, become an integral part of a classified end product, or produce a properly classified effect. However, the association of otherwise unclassified hardware with a particular effort or product may reveal something classified about that effort or product. Common integrated circuits that control frequencies are notable examples. In such cases it is the association with the effort or product that reveals the classified information, not the circuits themselves. Decisions regarding what aspect of the system to classify may be difficult, but are necessary to delineate for users of the guide, what information requires protection.

4.2.3 Frequently, classified information pertaining to a hardware item can be restricted to the paper work associated with the item.

4.2.4 Unusual, unique, or peculiar uses or modifications of ordinarily available unclassified materials or hardware may create a classifiable item of information In another instance, the mere fact of use of a particular material in a particular effort might reveal a classifiable research or development interest. In such cases, it is especially important to accurately identify the classified information in order to determine whether it is the hardware or material that reveals classified information, or that it is the association of use of the hardware with a particular effort that reveals such information.

4.2.5 At some stage in a production effort, production and engineering plans are drawn. Usually a family-tree type diagram is prepared to assist in determining what components, parts, and materials will be required. This diagram supplies a good basis to determine where and when classified information will be involved in the production effort. 4.2.6 Another usual step in production engineering is the development of drawings for all the individual elements that go into the final product. These drawings show design data, functions, and specifications, all of which are closely tied with items of information that may be classified. From these drawings it is possible to determine exactly which elements of the final product will reveal classified information. It is also possible to determine associations that may reveal classified information. This is a prime opportunity to identify and isolate classification requirements.

# C.4.3. USER CONSIDERATIONS

Know who will be using your classification guide.

4.3.1 Usually management and staff supervisory personnel need to have a fairly broad knowledge of classification requirements. Farther down the line however, foremen and workers usually need to know only which hardware items are classified, the appropriate levels of classification and which items are unclassified. Therefore, as soon as possible in the production planning process, make a listing of all classified hardware items according to part number or other identifier, and when necessary for understanding, a listing of unclassified items. Such a listing will be valuable to procurement and logistics (shipping, handling, and storage) personnel. The listing should preferably be unclassified, but should be reviewed carefully to ensure that classified information is not revealed by the listing itself, particularly through association.

4.3.2 When planning a production line, careful attention is needed to delay as long as possible the insertion of classified hardware items.

4.3.3 Test equipment rarely embodies classified information. When such equipment is used to test tolerances, specifications, performance, and other details that are classified, the equipment would still be unclassified unless it was calibrated or set in such a way as to reveal the classified information pertaining to the item being tested. This is one example of a situation where it may be possible to limit the classified information to the paper work involved and to the test operator's personal knowledge, precluding the necessity for classifying the test equipment itself.

# C.5. CHAPTER 5

CLASSIFYING MILITARY OPERATIONS INFORMATION

# C.5.1. GENERAL

The security classification of military operations information is subject to many of the considerations described in Chapter C.3. and Appendix C of this Handbook. While there are no hard and fast rules for classification of military operations information, and while each Military Service and command may require a unique approach to operations security (OPSEC), there are basic concepts that can be applied.

# C.5.2. MILITARY OPERATIONS INFORMATION

Military operations is defined for the purpose of this Handbook as information pertaining to a strategic or tactical military action, including training, movement of troops and equipment, supplies, and other information vital to the success of any battle or campaign.

# C.5.3. MILITARY OPERATIONS CLASSIFICATION CONSIDERATIONS

5.3.1 Successful battle operations depend largely upon our ability to assess correctly the capability and intention of enemy forces at each stage of the battle while concealing our own capabilities and intentions, and to communicate an effective battle doctrine throughout our forces. Classifiable information would include:

5.3.1.1 The number, type, location, and strengths of opposing units.

5.3.1.2 The capabilities and vulnerabilities of weapons in enemy hands, and how he normally applies the weapons. 5.3.1.3 The morale and physical condition of the enemy force.

5.3.2 In considering classification guidance for military operations, there may be good reason to classify more information about the operations in the beginning than will be necessary later. Certain elements of information such as troop movements may no longer require protection after a certain date or event. When this point is reached, downgrading or even declassification should be considered.

5.3.3 The following are examples of information relating to military operations that may warrant classification:

Table C–1 Classification guidance			
TOPIC	CLASS	DECLASS	REMARKS
Overall operational plans	"S"	Date, event, date w/in 10 yrs	
System operational deployment or employment	"C"	After deployment or Employ- ment	
Initial Operational Capability (IOC) Date	"C"	After IOC Date	
Planned location of operational units	"S"	After arrival on site	
Equipage dates, readiness dates, Operational employment dates	"S"	After these events	
Total manpower or personnel Requirements for total operational Force	"C"	After operation	
Coordinates of selected operational sites	"S"	"C" after site activation; "U" on termination of site.	
Specific operational performance Data which relates to the effec- tiveness of the control of forces and data on specific vulnerabilities and weaknesses.	"S"	Date/event, date w/in ten years	
Existing OPSEC and COMSEC Procedures, projections, and techniques.	"S"	Date/event, date w/in ten years	
Target characteristics	"S"	Date/event, date w/in ten years	

# C.6. CHAPTER 6

CLASSIFYING INTELLIGENCE INFORMATION

# C.6.1. CLASSIFICATION CONSIDERATIONS

Producers of intelligence must be wary of applying so much security that they are unable to provide a useful product to their consumers. Consequently, an intelligence product should be classified only when its disclosure could reasonably be expected to cause some degree of damage to national security. The following are some basic considerations, but are not necessarily all–inclusive:

6.1.1 In general, resource information should not be classified unless it reveals some aspect of the intelligence mission, and its revelation would jeopardize the effectiveness of a particular function. An example of classifiable resource information is the intelligence contingency fund.

6.1.2 Intelligence concerning foreign weapons systems may be classified based on what is generally known about a particular system or its components. Normally, the less that is publicly known about a particular system or component, the higher its level of classification.

6.1.3 Intelligence identifying a sensitive source or method is classified, as well as the evaluation of the particular source or method.

6.1.4 Intelligence which does not identify or reveal a sensitive source or method is usually not classified unless the information contains other classified information such as intelligence activities including intelligence plans, policies, or operations.

6.1.5 Intelligence that reveals the identity of a conventional source or method normally does not require classification. However, if the information is communicated to the Department of Defense by a foreign government, whether under a formal government–to–government agreement or, simply with the understanding that the information is provided in confidence, the information must be protected at the level and for the length of time the United States and the transmitting government agree to. If the information is obtained from a foreign government without any agreement or restrictions, the classification, if any, should be based solely on the content of the information provided.

6.1.6 Intelligence that reveals the identification of all known and possible enemy capabilities to collect and exploit information from a given or similar operation is classified. This threat would include enemy intelligence collection and analysis capabilities, efforts, and successes. An integral part of this data is an assessment of enemy human intelligence, signals intelligence, and reconnaissance satellite capabilities.

6.1.7 Security classification assigned to intelligence received from non-Defense sources must be respected by Defense users.

6.1.8 An intelligence estimate is normally classified since it contains sensitive sources, methods, or raw or evaluated intelligence.

6.1.9 An intelligence requirement is classified when it reveals what is not known, what is necessary to know, and why.

Moreover, the requirement may recommend a sensitive source or method, other military intelligence required, or contain technical and operational characteristics of classified weapons systems.

6.1.10 The classification of relationships with foreign intelligence organizations is related to the following considerations:

6.1.10.1 Normally, the fact of broad, general intelligence cooperation with foreign countries or groups of countries with which the United States maintains formal military alliances or agreements (e.g. NATO) is not classified.

6.1.10.2 The fact of intelligence cooperation between the United States and a specific governmental component in an allied country, or general description of the nature of intelligence cooperation between the United States and any allied country may be classified. The fact of intelligence cooperation between the United States and specifically named countries or their governmental components with which the United States is NOT allied is always classified.

6.1.10.3 Details of or specifics concerning any intelligence exchange agreements are classified. In some instances, the mere fact of such an agreement may be classified.

6.1.10.4 The identities of foreign governmental or military personnel who provide intelligence under such agreements or liaison relationships may be classified.

6.1.11 Information that reveals counterintelligence activities, identities of undercover personnel or units or clandestine human agents, methods of operations and analytical techniques for the interpretation of intelligence data is classified. 6.1.12 Cryptologic information (including cryptologic sources and methods) is classified.

6.1.13 Information concerning electronics intelligence, telemetry intelligence, and electronic warfare is usually classified.

6.1.14 The intelligence community normally considers the following categories of information to be classified:

6.1.14.1 Cryptologic, cryptographic, signals intelligence, or imagery intelligence.

6.1.14.2 Counterintelligence.

6.1.14.3 Special access programs.

6.1.14.4 Information which identifies clandestine organizations, agents, sources, or methods.

6.1.14.5 Information on personnel under official or nonofficial cover, or revelation of a cover arrangement.

6.1.14.6 Covertly obtained intelligence reports and the derivative information which would divulge intelligence sources or methods.

6.1.14.7 Methods or procedures used to acquire, produce, or support intelligence activities.

6.1.14.8 Intelligence organizational structure, size, installations, security, objectives, and budget.

6.1.14.9 Information that would divulge intelligence interests, value, or extent of knowledge on a subject.

6.1.14.10 Training provided to or by an intelligence organization which would indicate its capability or identify personnel.

6.1.14.11 Personnel recruiting, hiring, training, assignment, and evaluation policies.

6.1.14.12 Information that could lead to foreign political, economic, or military action against the United States or its allies.

6.1.14.13 Events leading to international tension that would affect U.S. foreign policy.

6.1.14.14 Diplomatic or economic activities affecting national security or international security negotiations.

6.1.14.15 Information affecting U.S. plans to meet diplomatic contingencies affecting national security.

6.1.14.16 Nonattributable activities conducted abroad in support of U.S. foreign policy.

6.1.14.17 U.S. surreptitious collection in a foreign nation that would affect relation with the country.

6.1.14.18 Covert relationships with international organizations or foreign governments.

6.1.14.19 Information related to political or economic instabilities in a foreign country threatening American lives and installation there.

6.1.14.20 Information divulging U.S. intelligence and assessment capabilities.

6.1.14.21 United States and allies' defense plans and capabilities that enable a foreign entity to develop countermeasures.

6.1.14.22 Information disclosing U.S. systems and weapons capabilities or deployment.

6.1.14.23 Information on research, development, and engineering that enables the United States to maintain an advantage of value to national security.

6.1.14.24 Information on technical systems for collection and production of intelligence.

6.1.14.25 U.S. nuclear programs and facilities.

6.1.14.26 Foreign nuclear programs, facilities, and intentions.

6.1.14.27 Contractual relationships that reveal the specific interest and expertise of an intelligence organization.

6.1.14.28 Information that could result in action placing an individual in jeopardy.

6.1.14.29 Information on secret writing when it relates to specific chemicals, reagents, developing, and microdots.

6.1.14.30 U.S. Military space programs.

# C.6.2. INTELLIGENCE DECLASSIFICATION CONSIDERATIONS

Normally intelligence will remain classified for a longer duration than other types of classified information, but still only as long as is necessary to protect a certain source or method. The outline in Chapter 3 of this Handbook on determining how long classification must continue is applicable to all information, including intelligence.

# C.6.3. CLASSIFICATION GUIDE ILLUSTRATIONS

The treatment of Classifying Details (Appendix B) and Recommended Format for a Security Classification Guide (Appendix D) are applicable to the development of an intelligence security classification guide. In addition, the following is provided as an example of security classification guidance that might be applied to a Human Intelligence (HUMINT) effort:

Table C–2 HUMINT classification guide			
TOPIC	CLASS	DECLASS	REMARKS
Biographic information taken exclusively from open source, where no intelligence connection is shown.	"U"		
Positive identification of an individual as po- tential source to a U.S. intelligence agency.	"S–TS"	Date/event w/in ten years, or 10 years from origination.	"TS" if identified as an actual source.
Identity of a target installation or target per- sonality when not linked to a specific collec- tion operation.	"S"	Date/event w/in ten years, or 10 years from origination.	"TS" when linked to an actual source or specific collection oper- ation.
Interest in specific events for collection exploi- tation, including specific areas of technology.	"S"	Date/event w/in ten years, or 10 years from origination.	
Names of collection agency case officers in conjunction with a specific collection operation.	"C"	Date/event w/in ten years, or 10 years from origination.	
Information on collection agency HUMINT pol- icy plans, resources, Methods, or accomplish- ments.	"S"	Date/event w/in ten years, or 10 years from origination.	

# C.7. CHAPTER 7

CLASSIFYING FOREIGN RELATIONS INFORMATION

# C.7.1. GENERAL

The Department of State (DoS) is the agency primarily responsible for the development and execution of the foreign policy of the United States, and thus is also the primary agency responsible for the security classification of foreign relations information. Most Defense classification determinations in the area of foreign relations will be derivative in nature. However, there will be instances where Defense projects and programs involve foreign relations information for which security classification guidance must be developed.

# C.7.2. FOREIGN RELATIONS CLASSIFICATION CONSIDERATIONS

The following are some of the types of information or material involving foreign relations that warrant classification consideration:

7.2.1 All material that reveals or recommends U.S. Government positions or options in a negotiation with a foreign government or group of governments, or that comments on the merits of foreign government positions in such negotiations.

7.2.2 All material that comments on the quality, character, or attitude of a serving foreign government official, whether elected or appointed, and regardless of whether the comment is favorable or critical. Illustrations of the types of information covered in this category are records revealing:

7.2.2.1 A foreign official speaking in a highly critical manner of his own government's policy.

7.2.2.2 A foreign official suggesting how pressure might effectively be brought to bear on another part of his own government.

7.2.2.3 A foreign official acting in unusually close concert with U.S. officials where public knowledge of this might be harmful to that foreign official.

7.2.2.4 A foreign official whose professional advancement would be beneficial to U.S. interest, especially if any implication has been made of U.S. efforts to further his advancement, or if public knowledge of this might place the person or his career in jeopardy.

7.2.3 All unpublished, adverse comments by U.S. officials on the competence, character, attitudes, or activities of a serving foreign government official.

7.2.4 All material which constitutes or reveals unpublished correspondence between heads of state or heads of government.

7.2.5 Statements of U.S. intent to defend, or not to defend, identifiable areas, or along identifiable lines, in any foreign country or region.

7.2.6 Statements of U.S. intent to attack militarily in stated contingencies, identifiable areas in any foreign country or region.

7.2.7 Statements of U.S. policies or initiatives within collective security organizations, e.g., NATO.

7.2.8 Agreements with foreign countries for the use of, or access to, military or naval facilities.

7.2.9 Contingency plans insofar as they involve other countries, the use of foreign bases, territory, or airspace; or the use of chemical, biological, or nuclear weapons.

7.2.10 Defense surveys of foreign territories for purposes of basing or using in contingencies.

7.2.11 Statements relating to any use of foreign bases not authorized under bilateral agreements.

# C.7.3. CLASSIFICATION GUIDE ILLUSTRATIONS

7.3.1 The treatment of Classifying Details (Appendix B) and Recommended Format for a Security Classification Guide (Appendix D) are applicable to the development of a foreign relations security classification guide. The following is provided as an example of the impact that foreign government information might have on the development of classification guidance.

7.3.1.1 A DOD Component is involved in negotiating some arrangement with country "X." In the process of the negotiations, the foreign counterpart states that his country does not want discussion on the subject to become public knowledge. At the same time, the foreign official makes it clear that his country has announced publicly its intention to seek U.S. views on the subject of the discussions.

7.3.1.2 The nature of business being discussed is such that the United States would not require protecting the discussions from public disclosure. Moreover, the subject matter is one that would not ordinarily be classified. The DoD Component, however, does classify the notes and transcripts pertaining to the discussion because of the expressed wishes of the foreign government. The information fits the description of foreign government information. Thus, a classification guide on the subject might contain the following topics:

Table C–3 Classification of notes and transcripts						
ТОРІС	CLASS	DECLASS	REMARKS			
Apple orchard negotiations with country "X."	"U"		Mere fact of negotiations only, and elaboration may be classified, see next topic.			
Transcripts of apple orchard negotiations and substantive notes pertaining to them.	"C"	Requires consultation with for- eign government				

7.3.1.3 The foregoing scenario illustrates a brief classification guide involving the foreign relations of the United States as well as foreign government information. The guide could not have been written until after the opening of the negotiations at which point the foreign official made known the two critical elements of information. In anticipation that the negotiations will involve a large number of personnel from several U.S. agencies and will last several years, a classification guide such as this one, brief as it is, can serve a very useful purpose.

7.3.1.4 To illustrate a scenario with military implications, let's presume that two countries in Europe have secretly granted the United States permission to fly over their territory, but only at high (50,000 feet) altitudes. One of the countries ("Y") indicated that serious damage would occur to our relations if the information became public while the other ("Z") indicated that it did not want the information to be in the public domain. Classification guide topics might read as follows:

Table C–4 Classification topics							
ТОРІС	CLASS	DECLASS	REMARKS				
<ul> <li>(U) Fact of U.S. overflights – Europe</li> <li>a. (S) Country "Y"</li> <li>b. (C) Country "Z"</li> <li>(U) Other European</li> </ul>	"S" "C" "U"	Requires written approval of foreign government involved.	(S) Must be at least 50,000 feet alti- tude; lower flights not permitted in "Y" and "Z"				
Notes:							

In this example, the guide itself would have to be classified "S" as it reveals the information that country "Y" has determined would result in serious damage.

# Appendix A CLASSIFICATION FACTORS

The following questions, answers, and potential actions will assist in systematically determining whether certain broad aspects of an effort warrant security classification:

# **CLASSIFICATION FACTORS**

The following questions, answers, and potential actions will assist in systematically



determining whether certain broad aspects of an effort warrant security classification:



Figure A-1. CLASSIFICATION FACTORS—Continued



Figure A-1. CLASSIFICATION FACTORS—Continued

# Appendix B CLASSIFYING DETAILS

Having considered the factors involved in making classification determinations concerning the overall effort, it is now necessary to take the second step and consider the classification of certain specific details of the effort. Providing answers to the following questions will assist in systematically reviewing the details of the effort to determine security classification. The questions are not presented in any order of priority. A listing of specific items of information to consider is contained in Appendix C.

# **B.1. PERFORMANCE OR CAPABILITY**

*b.1.1.* What will this do (actual or planned) that is more, better, faster, or cheaper (in terms of all kinds of resources) than anything like it?

*b.1.2.* How does this degree or kind of performance contribute to or create a national security advantage? How much of an advantage?

b.1.3. How long can this data be protected? The advantage

b.1.4. How would knowledge of these performance details help an enemy, or damage the success of the effort?

*b.1.5.* Would statement of a particular degree of attained performance or capability be of value to hostile intelligence in assessing U.S. capabilities? In spurring a foreign nation to similar effort, or in developing or planning countermeasures?

### **B.2.. UNIQUENESS**

*b.2.1.* What information pertaining to this effort is know or believed to be the exclusive knowledge of the United States?

b.2.2. Is it known or reasonable to believe that other nations have achieved a comparable degree of success or attainment?

*b.2.3.* What information, if disclosed, would result in or assist other nations in developing a similar item or arriving at a similar level of achievement?

b.2.4. In what way or ways does the uniqueness of this item contribute to a national security advantage?

b.2.5. In what way or ways has the end product of this effort or any of its parts been modified, developed, or applied so at to be unique to this kind of effort? How unique is this?

*b.2.6.* Is the method of adaptation or application of the end product or any of its parts the source of the uniqueness and a national security advantage? In what way or ways? Is it in itself a unique adaptation of application in this kind of effort?

### **B.3. TECHNOLOGICAL LEAD TIME**

*b.3.1.* How long did it take to reach this level of performance or achievement?

b.3.2. How much time and effort have been expended? Was this a special concerted effort, or only a gradual developmental type of activity?

b.3.3. If all or some of the details involved in reaching this stage of development or achievement were known, how much sooner could this goal have been reached? Which details would contribute materially to a shortening of the time for reaching this goal? Can these details be protected? For how long?

b.3.4. Have other nations reached this level of development or achievement?

b.3.5. Do other nations know how far we have advanced in this kind of effort?

*b.3.6.* Would knowledge of this degree of development or achievement spur a foreign nation to accelerate its efforts to diminish our lead in this field? What details of knowledge would be likely to cause such acceleration?

b.3.7. How important, in terms of anticipated results, is the lead time we think we have gained?

b.3.8. What national security advantage actually results from this lead time?

b.3.9. How long is it practical to believe that this lead time will represent an actual advantage?

b.3.10. How long is it practical to expect to be able to protect this lead time?

### **B.4. SURPRISE**

b.4.1. Do other nations know we have reached this level of development or achievement?

*b.4.2.* Will operational use of the end item of this effort give us an immediate advantage that would be less or lost if it were known that we have achieved this particular goal?

b.4.3. What is the nature of the advantage resulting from surprise use of this end item?

*b.4.4.* When will this element of surprise be lost?

### **B.5. VULNERABILITIES AND WEAKNESSES**

*b.5.1.* What are the weak spots in this effort that make it vulnerable to failure? What is the rate or effect of this failure?

b.5.2. How will the failure of the effort in whole or in part affect the national security advantage expected upon completion of this effort, or use of the resulting end item?

b.5.3. What elements of this effort are subject to countermeasures?

b.5.4. How would knowledge of these vulnerable elements assist in planning or carrying out countermeasures?

b.5.5. Can information concerning these weak or vulnerable elements be protected from unauthorized disclosure – or are they inherent in the system?

b.5.6. Can these weaknesses or vulnerabilities be exploited to reduce or defeat the success o this effort? How could this be done?

b.5.7. What measures are planned or have been taken to offset these weaknesses or vulnerabilities?

b.5.8. Are the counter-countermeasures obvious, special, unique, unknown to outsiders or other nations?

*b.5.9.* How would knowledge of these counter–countermeasures assist in carrying out or planning new countering efforts?

*b.5.10.* Would knowledge of specific performance capabilities assist in developing or applying specific countermeasures? How? What would be the effect on the expected national security advantage?

### **B.6. SPECIFICATIONS**

b.6.1. What would details of specification reveal:

(6.1.1) A special or unusual interest that contributes to the resulting or expected national security advantage?

(6.1.2) Special or unique compositions that contribute to the resulting or expected national security advantage?

(6.1.3) Special or unique levels of performance that are indicative of a classifiable level of achievement or goal?

(6.1.4) Special, or unique use of certain materials that reveals or suggests the source of a national security advantage?

(6.1.5) Special or unique size, weight, or shape that contributes to the resulting or expected national security advantage?

*b.6.2.* Are any specification details in themselves contributory to the resulting or expected national security advantage? How?

*b.6.3.* Can details of specifications be protected? For how long?

### **B.7. CRITICAL ELEMENTS**

b.7.1. What are the things that really make this effort work?

*b.7.2.* Which of these critical elements contribute to the resulting of expected national security advantage? How? To what extent?

b.7.3. Are these critical elements the source of weakness or vulnerability to countermeasures?

*b.7.4.* What details of information pertaining to these critical elements disclose or reveal the national security advantage, weakness or vulnerability?

*b.7.5.* Can details of information pertaining to these critical elements be protected by classification? For how long?

### **B.8. MANUFACTURING TECHNOLOGY**

*b.8.1.* What manufacturing methods, techniques, or modes of operation were developed to meet the requirements of this effort?

*b.8.2.* Which of these manufacturing innovations are unique to this effort or this product? Are they generally known or suspected?

b.8.3. Are these manufacturing innovations essential to successful production of the product?

b.8.4. What kind of lead time results from these innovations?

### **B.9. ASSOCIATIONS**

b.9.1. Are there any associations between this effort and others that raise classification questions?

*b.9.2.* Are there associations between information in this effort, and already publicly available information (unclassified), that raise classification problems?

*b.9.3.* Is it necessary or possible to classify items of information in this effort because of their association with other unclassified or classified information would diminish or result in the loss of a national security advantage?

### **B.10. PROTECTABILITY**

b.10.1. Can the information effectively be protected from unauthorized disclosure by classification? For how long? b.10.2. If not, what alternative means can be used to ensure protection?

# Appendix C ITEMS OF INFORMATION

Table C-1 shows items of information that may disclose present or future strategic or tactical capabilities and vulnerabilities and which should be considered when preparing classification guidance:

Performance And Capabilities Accuracy	Payload	
Alert time	Penetration	
Altitude	Range (range scales)	
Maximum	Rate of fire	
Optimum	Reaction time	
Ballistics	Reliability/failure rate data	
Initial	Resolution	
Terminal	Response time	
Control	Sensitivity	
Countermeasures (proven, unproven)	Sequence of events	
Counter-countermeasures	Signature Characteristics	
Decoys	Acceptance	
Electronic	Analysis	
Penetration aids	Distinguishment	
Shield materials	Identification	
Depth/height (also of burst)	Speed/velocity	
Maximum	Acceleration/deceleration	
Optimum	Cruise	
Duration (flight)	Intercept	
Effectiveness	Landing	
Frequencies (bands, specific, command,	Maximum	
operating, infrared, microwave, radio,	Minimum	
COMSEC)	Optimum	
Heating	Stability	
Impulse	Target data	
Intercept	Details	
Lethality/critical effects	Identification	
Lift	Illumination	
Limitations	Impact predicted	
Maneuverability	Preliminary	
Military strength	Priority	
Actual	Range determination	
Planned, predicted, anticipated	Thresholds	
Miss distance	Thrust	
Noise Figure	Toxicity	
Operational readiness time cycle	·	

(Detailed, Basic, Subsidiary)

Balance Burn rate Capacity (system) Center of gravity Codes Composition Configuration/contour Consumption Energy requirements Specific Total Filler Loading/loads Mass factor (propellant) Moment of inertia On-station time Output data Payload Power requirements Purity Size, weight, shape Stability (static, dynamic) Strength of members, frames Stresses

TOPIC	CLASS DECLASS	REMARKS
Fineness	Thickness	
Gain configuration	Tolerance	
Hardness, degree	Туре	
nput data		
VULNERABILITY		
Countermeasures/counter	Signature characteristics	
Countermeasures	Acoustic	
Dynamic pressure (supersonic)	Electrical	
EMP (radiation)	Infrared	
Ground or air shock	Magnetic	
Jamming	Pressure	
	Radar	
	Static overpressure	
PROCUREMENT AND PRODUCTION		
Completion date or dates	Progress/schedules (mile-	
Numbers	stones)	
Dispersion (numbers per unit	Stock density	
of force)	Supply plans and status	
On-hand stockpile	Tactical deployment	
Planned or programmed (totals scheduled)		
Rate of delivery or production		
Requirements		
Spares		
OPERATIONS		
Countdown time	Plans	
Deployment data	Command and control	
Environment	Results	
Location	Analysis, Conclusions,	
Numbers available	reports	
Objectives	Sequences of events	
Mission of program	Staging techniques	
Specific or general	Statement.concept	
Test, broad or detailed	Tactical	
	Build–up, units per force	
	activation dates,	
	personnel	

# Appendix D Recommended Format For A Security Classification Guide

This Appendix illustrates a format for a security classification guide. (A cover page is recommended showing essentially the following:)

# Section 1 NAME OF THE PROGRAM, PROJECT, SYSTEM OR STUDY

(If necessary, use an acronym, short title or project number in order to keep title unclassified)

# SECURITY CLASSIFICATION GUIDE

(Date of the guide)

ISSUED BY: (Name and address of issuing office.)

**APPROVED BY:** (Original Classification Authority) (Statement of supersession of any previous guides.)

(Distribution Limitation Statement for the Defense Technical Information Center per DOD Directive 5230.24 (reference (e))

PROGRAM, PROJECT, SYSTEM (ETC.) SECURITY CLASSIFICATION GUIDE (Date of the guide):

# Section I GENERAL INSTRUCTIONS

# 1. Purpose.

To provide instructions and guidance on the classification of information involved in (insert name of the program, project, etc., using an unclassified identification of the effort).

# 2. Authority.

This guide is issued under authority of (state any applicable departmental or agency regulations authorizing or controlling the issuance of guides, such as DOD 5200.1–R, "Information Security Program"). Classification of information involved in (identification of the effort) is governed by, and is in accordance with, (cite any applicable classification guidance or guides under which this guide is issued). This guide constitutes authority, and may be cited as the basis for classification, regrading, or declassification of information and material involved in (identification of the effort). Changes in classification required by application of this guide shall be made immediately. Information identified as classified in this guide is classified by (complete title or position of classifying authority).

# 3. Office of Primary Responsibility (OPR):

This guide is issued by, and all inquiries concerning content and interpretation, as well as any recommendations for changes, should be addressed to:

(Name, code, mailing address of issuing office)

(An administrative or security office in the issuing activity may be used. Inclusion of the action officer's name and phone number/fax and e-mail is desirable.)

# 4. Classification Challenges.

If at any time, any of the security classification guidance contained herein is challenged, the items of information involved shall continue to be protected at the level prescribed by this guide until such time as a final decision is made on the challenge by appropriate authority. Classification challenges should be addressed to the OPR.

# 5. Reproduction, Extraction and Dissemination.

Authorized recipients of this guide may reproduce, extract, and disseminate the contents of this guide, as necessary, for application by specified groups involved in (identification of the effort), including industrial activities. Copies of separate guides issued to operating activities in application of this guide shall be sent to the OPR.

NOTE: If it is necessary to classify the guide, you may have to modify this paragraph to express any required limitations.

# 6. Public Release.

The fact that this guide shows certain details of information to be unclassified does not allow automatic public release of this information. Proposed public disclosures of unclassified information regarding (identification of effort) shall be processed through appropriate channels for approval. NOTE: It may be desirable to indicate the office to which requests for public disclosure are to be channeled.

# 7. Foreign Disclosure.

Any disclosure to foreign officials of information classified by this guide shall be in accordance with the procedures set forth in (identify applicable issuances implementing DOD foreign disclosure policy). If a country with which the Department of Defense has entered into a reciprocal procurement memorandum of understanding or offset arrangement, expresses an interest in this effort, a foreign disclosure review should be conducted prior to issuance of a solicitation. (If it is known that foreign participation cannot be permitted because of the sensitivity of the effort, this fact should be stated.)

### 8. Definitions.

(Include in this paragraph the definitions of any items for which there may be various meanings to ensure common understanding of the details of information that are covered by the guide.)

### Section II OVERALL EFFORT

## 9. Identification.

(Include in this paragraph any necessary statements explaining the classifications, if any, to be assigned to various statements identifying the effort. These statements should be consistent with other program documentation.)

### 10. Goal, Mission, Purpose.

(Include in this paragraph any necessary statements identifying information concerning the purpose of the effort that can be released as unclassified and that which must be classified. Take care to ensure that unclassified statements do not reveal classified information.)

### 11. End Item.

(Include in this paragraph statements of the classification to be assigned to the end products of the effort, whether paperwork or hardware. In this connection it is important to distinguish between classification required to protect the fact of the existence of a completed end item, and classification required because of what the end item contains or reveals. In some instances classified information pertaining to performance, manufacture, or composition of incorporated parts or materials is not ascertainable from mere use of or access to the end item. In others, the classifiable information is that which concerns total performance, capabilities, vulnerabilities, or weaknesses of the end item itself, rather than any of the parts or materials.)

# Section III

# PERFORMANCE AND CAPABILITIES

(This section includes characteristics of performance and capability of an end item, or an end item's components, parts, or materials, the performance or capabilities of which require classification In this section, also provide, in sequentially numbered items, statements that express details of performance and capabilities planned and actual. Include both those elements that warrant classification and those that are unclassified. These statements normally would not set forth the numeric values that indicate degree of performance or capability, planned or attained, but merely should identify the specific elements of performance or capability that are covered. When it is necessary to state certain limiting figures above or below which classification is required, the statement itself may warrant classification. For clarity, continuity, or ease of reference it may be desirable to include performance classification data in the sections dealing with the end item or the components or parts to which the performance data apply. Use a "Remarks" column for explanations, limitations, special conditions, associations, etc.)

Table D–1 Performance and capabilities topic	s		
TOPIC	CLASS	DECLASS	REMARKS
1. Range a. Actual	"S"	15 June 1999	
2. Accuracy/range rate a. Predicted	"C"	30 Jan 2000	
3. Altitude: Operational Maximum	"C" "C"	30 Jan 2000 30 Jan 2000	The statement "in excess of 50,000 feet" is "U"
4. Receiver sensitivity, selectivity, and frequency coverage.	"S"	15 Apr 2005	If standard commercial receivers are used, their characteristics are "U" but their application to this effort shall be "S."
<ul><li>5. Resolution Thermal</li><li>a. Maximum</li><li>b. Operational optimum</li></ul>	"S" "S"	15 Apr 2001 15 Apr 2001	Planned or actual attained thermal resolutions above 0.25 degrees C. are "U."

Table D–1 Performance and capabilities topics—Continued						
TOPIC	CLASS	DECLASS	REMARKS			
c. Operational attainment	"S"	15 Apr 2001				
6. Speed						
a. Maximum	"S"	15 Jan 2001	Downgrade to "C" upon IOC.			
b. Rate of climb	"S"	15 Jan 2001	Reference to "supersonic speed" is "U."			
c. Intercept	"S"	15 Jan 2001	· ·			

# Section IV SPECIFICATIONS

This section includes items of information describing standards for qualities of materials and parts; methods or modes or construction, manufacture or assembly; and specific dimensions in size, form, shape, and weight, that require classification. Because they are contributory to the national security advantage resulting from (identification of this effort), or which frequently require classification but are unclassified in (identification of this effort). Classification of specifications pertaining to performance capability are covered in section 3. (Actual figures do not need to be given, merely statements identifying clearly the specific items of information involved. If figures are necessary to establish classification levels, it may be necessary to classify the statements themselves. When necessary for clarity, continuity or ease of reference, specification data may be included in sections on the end product or components or parts to which the data apply. Use a "Remarks" column for explanations, limitations, special conditions, associations, etc.)

Table D–2 Specification topics			
TOPIC	CLASS	DECLASS	REMARKS
1. Burn rate	"C"	17 Sep 2001	
2. Power requirement	"S"	17 Sep 2001	Only when associated with advanced model ##, otherwise "U."

# Section V CRITICAL ELEMENTS

This section is used only if there are specific elements that are critical to the successful operation of the end item of this effort, and are unique enough to warrant classification of some data concerning them. Provide in sequentially numbered paragraph each significant item of information peculiar to these critical elements and the classification applicable. Also include in this section the classification to be assigned to information pertaining to components, parts, and materials that are peculiar and critical to the successful operation of the end item in this effort when such items of information are the reason for or contribute to the national security advantage resulting from this effort. Performance data pertaining to such critical elements can be included in this section instead of section 3.

# Section VI VULNERABILITIES AND WEAKNESSES

This section is used to specify classification to be assigned to details of information that disclose inherent weaknesses

that could be exploited to defeat or minimize the effectiveness of the end product of this effort. Classification assigned to details of information on countermeasures and counter-countermeasures should be included in this section.

# Section VII ADMINISTRATIVE DATA

This section is used only if particular elements of administrative data, such as program information, procurement schedules, production quantities, schedules, programs, or status of the effort, and data on shipments, deployment, or transportation and manuals (field, training, etc.) warrant classification.

Table D–3 Administrative data topics					
ТОРІС	CLASS	DECLASS	REMARKS		
<ol> <li>Planned delivery rate.</li> <li>Actual routing of delivery of end items.</li> </ol>	"C" "C"	13 Mar 2001 See remarks	See item 3, below. Classify upon selection of route, and declassify upon completion of last delivery to site		
3. Shipping dates and times.	"C"	See remarks	Classify upon decision to ship, and declassify upon arrival at site.		

### Section VIII HARDWARE

The degree of specificity to be included in this section will depend largely upon:

a. The level from which issued. When issued from a headquarters level, probably the only classification to be applied to hardware would be to the end item itself.

*b*. The channels or hands through which the guidance will travel to the ultimate user. The closer the issuer is to the user, the more detailed the guidance may become. Intermediate levels may be required to expand or elaborate on the guidance, and cover more details concerning materials, parts, components, subassemblies, and assemblies, and the classification, if any, to be assigned. Any such expansion or elaboration should be fully coordinated the he headquarters issuing the basic guide.

c. The ease of determining when classified information could be revealed by a particular hardware item. Obscure connections and associations that could reveal classified information may require the issuer of the guide to state classification for certain hardware items. In such cases it probably would be advisable to explain why classification is necessary.

*d*. Whether there are factors that require consideration and action at a headquarters level. National or DOD policy, intelligence data, broad operational requirements, extraneous factors, or other matters not ordinarily available below headquarters, or which require high level consideration may result in decisions to classify certain hardware items.

Table D–4 Hardware classification			
INFORMATION			
REVEALING	CLASS	DECLASS	REMARKS
1. End item hardware:			
a. AN/APR-999	"C"	20 Aug 2000	External views of the assembled AN/AR-999 are "U"
<ol> <li>Analyzer unit</li> <li>Threat display</li> <li>Preamplifier</li> </ol>	"C" "U" "U"	20 Aug 2000	
b. AN/APR-OOOO	"C"		

# Appendix E FORMAT VARIATIONS

This appendix illustrates column headers and arrangements that are different from those used in appendix D. These headers and arrangements may be employed in the construction of your classification guide, and modified to suit your style and need in a given effort. For example, a column for downgrading action would not be necessary if the guide did not provide for it, or if only one or two items of information are to be downgraded. In the later case, the downgrading instruction could be placed in a "Remarks" or "Comments" column.

Table E–1 Format variation topics			
Example 1			
TOPIC 1.4.1 System capacity	CLASSIFICATION "S"	DECLASSIFY 30 Jun 2004	COMMENTS Downgrade to "C" upon TOC
1.4.1 Signature characteristics	"C"	19 Jun 2001	
Example 2			
DESCRIPTION 1.4.1 System capacity 1.4.2	CLASSIFIED "S" "C"	UNTIL 30 Jun 2004 19 Jun 2001	REMARKS Downgrade to "C" upon IOC.
Example 3			
INFORMATION			
REVEALING 1.4.1 System capacity 4.1 Signature characteristics	CLASSIFICATION/DECLASSIFICATION "S" DCL on 30 Jun 2004 "C" DCL on 19 Jun 2001		REMARKS Downgrade to "C" upon IOC

### **COVER BRIEF**

TO: SENIOR CIVILIAN OFFICIAL, OASD(C31)

THROUGH: PDASD(C31)

DASD(S&10) DASD(S&10)

**FROM:** DIRECTOR, SECURITY PROGRAMS Prepared by WHBell/695–2686

SUBJECT: Revision of DOD 5200.1-H, "DOD Handbook for Writing Security Classification Guidance – ACTION MEMORANDUM

PURPOSE: To obtain SCO signature on the SD 106 at Tab A.

# **DISCUSSION:**

In implementation of Executive Order 12958, "Classified National Security Information," and Information Security Oversight Office Directive No. 1, DOD Regulation 5200.1–R requires that a security classification guide be issued for each system, plan, program, or project in which classified information is involved. The Handbook provides guidance and illustrations to assist DOD Component personnel in developing such guides. The SD 106 at Tab A has been prepared to obtain DOD Component coordination on the proposed revision of the Handbook.

### COORDINATION: OASD(C31)(Policy)\_\_\_\_

### **RECOMMENDATION:**

That the SCO sign the SD 106 at Tab A.

# Appendix H Instructions Governing Use of Code Words, Nicknames, and Exercise Terms

### Section I Definitions

### H-1. Using Component

The DOD component to which a code word is allocated for use, and which assigns to the word a classified meaning, or which originates nicknames and exercise terms using the procedure established by the Joint Chiefs of Staff.

### H-2. Code Word

Word selected from those listed in Joint Army–Navy–Air Force Publication (JANAP) 299 and later volumes, and assigned a classified meaning by appropriate authority to ensure proper security concerning intentions, and to safeguard information pertaining to actual military plans or operations classified as Confidential or higher. A code word will not be assigned to test, drill or exercise activities. A code word is placed in one of three categories:

a. Available. Allocated to the using component. Available code words individually will be unclassified until placed in the active category.

b. Active. Assigned a classified meaning and current.

c. Canceled. Formerly active, but discontinued due to compromise, suspected compromise, cessation, or completion of the operation to which the code word pertained. Canceled code words individually will be unclassified and remain so until returned to the active category.

#### H-3. Nickname

A combination of two separate unclassified words which is assigned an unclassified meaning and is employed only for unclassified administrative, morale, or public information purposes.

### H-4. Exercise term

A combination of two words, normally unclassified, used exclusively to designate a test, drill, or exercise. An exercise term is employed to preclude the possibility of confusing exercise directions with actual operations directives.

### Section II Policy and Procedure

#### H–5. Code Words

The Joint Chiefs of Staff are responsible for allocating words or blocks of code words from JANAP 299 to DOD components. DOD components may request allocation of such code words as required and may reallocate available code words within their organizations, in accordance with individual policies and procedure, subject to applicable rules set forth herein.

a. A permanent record of all code words will be maintained by the Joint Chiefs of Staff.

b. The using component will account for available code words and maintain a record of each active code word. Upon being canceled, the using component will maintain the record for 2 years; thence the record of each code word may be disposed of in accordance with current practices, and the code word returned to the available inventory.

c. The Deputy Chief of Staff for Operations and Plans (DCSOPS), HQDA will control and allot blocks of code words from JANAP 299 to MACOMs and U.S. Army commands on request. Commands are authorized to make assignments from these code word blocks, subject to rules in this regulation. The DCSOPS will allocate code words to HQDA agencies, as needed. Requirements will be submitted in writing to Headquarters DA, DAMO–ODS, Washington, D.C. 20310–0440.

#### H-6. Nicknames

*a.* Nicknames may be assigned to actual events, projects, movement of forces, or other non-exercise activities involving elements of information of any classification category, but the nickname, the description or meaning it represents, and the relationship of the nickname and its meaning must be unclassified. A nickname is not designed to achieve a security objective.

b. Nicknames, improperly selected, can be counterproductive. A nickname must be chosen with sufficient care to ensure that it does not—

(1) Express a degree of hostility inconsistent with traditional American ideals or current foreign policy;

(2) Convey connotations offensive to good taste or derogatory to a particular group, sect, or creed; or,

(3) Convey connotations offensive to our allies or other Free World nations.

c. The following will not be used as nicknames:

(1) Any two-word combination voice call sign found in JANAP 119 or ACP 110. (However, single words in JANAP 119 or ACP 110 may be used as part of a nickname if the first word of the nickname does not appear in JANAP 299 and later volumes.)

(2) Combination of words including word "project," "exercise," or "operation."

(3) Words that may be used correctly either as a single word or as two words, such as "moonlight."

(4) Exotic words, trite expressions, or well-known commercial trademarks.

d. The Joint Chiefs of Staff will-

(1) Establish a procedure by which nicknames may be authorized for use by DOD Components.

(2) Prescribe a method for the using components to report nicknames used.

e. The heads of DOD components will-

(1) Establish controls within their components for the assignment of nicknames authorized under paragraph H–6a.

(2) Under the procedures established, advise the Joint Chiefs of Staff of nicknames as they are assigned.

(3) All requests for and changes in nicknames, including assignments, meanings, changes to meanings, cancellations, deletions, and possible compromises will be submitted in writing to Headquarters DA, DAMO-ODS, Washington, D.C. 20310-0440.

### H-7. Exercise terms

*a*. Exercise terms may be assigned only to tests, drills, or exercises for the purpose of emphasizing that the event is a test, drill, or exercise and not an actual operation. The exercise term, the description or meaning it represents, and the relationship of the exercise term and its meaning can be classified or unclassified. A classified exercise term is designed to simulate actual use of DOD code words and must be employed using identical security procedures throughout the planning, preparation, and execution of the test, drill, or exercise to ensure realism.

b. Selection of exercise terms will follow the same guidance as contained in paragraph 6a.

c. The Joint Chiefs of Staff will-

(1) Establish a procedure by which exercise terms may be authorized for use by DOD components.

(2) Prescribe a method for using components to report exercise terms used.

d. The heads of DOD components will-

(1) Establish controls within their component for the assignment of exercise terms authorized under paragraph 7a.

(2) Under the procedures established, advise the Joint Chiefs of Staff of exercise terms as they are assigned.

(3) All requests for and changes in exercise terms, including assignments, meanings, changes to meanings, cancellations, deletions, and possible compromises, will be submitted in writing to Headquarters DA, DAMO–ODS, Washington, D.C. 20310–0440 in accordance with AR 525–1 and JCS PUB 6, Vol. II.

### H-8. Assignment of classified meanings to code words

*a*. The DOD component responsible for the development of a plan or the execution of an operation will be responsible for determining whether to assign a code word.

b. Code words will be activated for the following purposes only:

(1) To designate a classified military plan or operation;

(2) To designate classified geographic locations in conjunction with plans or operations referred to in subparagraph 3b(1), above; or,

(3) To conceal intentions in discussions and messages or other documents pertaining to plans, operations, or geographic locations referred to in subparagraphs (1) and (2), above.

c. The using component will assign to a code word a specific meaning classified TOP SECRET, SECRET, or CONFIDENTIAL, commensurate with military security requirements. Code words will not be used to cover unclassified meanings. The assigned meaning need not in all cases be classified as high as the classification assigned to the plan or operation as a whole.

d. Code words will be selected by each using component in such manner that the word used does not suggest the nature of its meaning.

*e*. A code word will not be used repeatedly for similar purposes; that is, if the initial phase of an operation is designated "Meaning," succeeding phases should not be designated "Meaning II" and "Meaning III," but should have different code words.

f. Each DOD component will establish policies and procedures for the control and assignment of classified meanings to code words, subject to applicable rules set forth herein.

# H-9. Notice of assignment, dissemination, and cancellation of code words and meanings

*a.* The using component will promptly notify the Joint Chiefs of Staff when a code word is made active, indicating the word, and its classification. Similar notice will be made when any changes occur, such as the substitution of a new word for one previously placed in use. MACOMs, Army Staff Agencies, and Field Operating Agencies will notify Headquarters DA, DAMO–ODS, Washington, D.C. 20310–0440, of all codeword transactions as specified above.

b. The using component is responsible for further dissemination of active code words and meanings to all concerned activities, to include classification of each.

(1) Dissemination of the code word and its meaning to other DOD agencies will be made by ODCSOPS at the request of the assigning authority.

(2) The assigning authority is responsible for disseminating code words and their meanings to activities within its jurisdiction.

(3) When a MACOM or HQDA agency receives classified meanings and related code words from an agency outside DA, the receiving activity will provide this information to activities under its jurisdiction when needed for security reasons.

(4) A MACOM that receives a code word and its classified meaning from an agency outside of the U.S. Army, for which there is no required action, will retain that information in the office responsible for maintaining records of code words. No internal distribution of the meaning will be made without approval from the original using agency.

(5) If MACOMs or HQDA agencies receive documents or messages that contain code words but do not have the associated meaning, that information may be requested, in writing, from the DCSOPS if officially needed. Requests for the classified meaning will contain justification for the need.

(6) When a non-DOD agency furnishes a word that has a special meaning for use within DOD, recipients will be informed that it originated outside the DOD and is not subject to the DOD's code word policy. Words of this type will be safeguarded if required by the classification assigned by the originator.

c. The using component is responsible for notifying the Joint Chiefs of Staff of canceled code words. This cancellation report is considered final action, and no further reporting or accounting of the status of the canceled code word will be required.

### H-10. Classification and downgrading instructions

*a.* During the development of a plan, or the planning of an operation by the headquarters of the using component, the code word and its meaning will have the same classification. When dissemination of the plan to other DOD components or to subordinate echelons of the using component is required, the using component may downgrade the code words assigned below the classification assigned to their meanings in order to facilitate additional planning implementation, and execution by such other components or echelons, but code words will, at a minimum, be classified CONFIDENTIAL.

*b*. A code word which is replaced by another code word due to a compromise or suspected compromise, or for any other reason, will be canceled, and classified CONFIDENTIAL for a period of 2 years, after which the code word will become unclassified.

c. When a plan or operation is discontinued or completed, and is not replaced by a similar plan or operation but the meaning cannot be declassified, the code word assigned thereto will be canceled and classified CONFIDENTIAL for a period of 2 years, or until the meaning is declassified, whichever is sooner, after which the code word will become unclassified.

d. In every case, whenever a code word is referred to in documents, the security classification of the code word will be placed in parentheses immediately following the code word, for example, "Label (C)."

e. When the meaning of a code word no longer requires a classification, the using component will declassify the meaning and the code word and return the code word to the available inventory.

### H–11. Security practices

*a.* The meaning of a code word may be used in a message or other document, together with the code word, only when it is essential to do so. Active code words may be used in correspondence or other documents forwarded to addressees who may or may not have knowledge of the meaning. If the context of a document contains detailed instructions or similar information which indicates the purpose or nature of the related meaning, the active code word will not be used.

b. In handling correspondence pertaining to active code words, care will be used to avoid bringing the code words and their meanings together. They should be handled in separate card files, catalogs, indexes, or lists, enveloped separately, and dispatched at different times so they do not travel through mail or courier channels together.
c. Code words will not be used for addresses, return addresses, shipping designators, file indicators, call signs, identification signals, or for other similar purposes.

### H-12. Disposition

All code words formerly categorized as "inactive" or "obsolete" will be placed in the current canceled category and classified Confidential. Unless otherwise restricted, all code words formerly categorized as "canceled" or "available" will be individually declassified. All records associated with such code words may be disposed of in accordance with current practices, provided such records have been retained at least 2 years after the code words were placed in the former categories of "inactive," "obsolete," or "canceled."

Appendix I Special Access Programs (SAPs)

Section I General

## I-1. Policy

It is the policy of the Department of the Army to use the security classification categories, and the applicable sections of EO 12958 and its implementing ISOO Directives, to limit access to classified information on a "need-to-know" basis for only those personnel who have been determined to meet requisite personnel security requirements. Further, it is DA policy to rigorously apply the need-to-know principle in the normal course of controlling collateral classified information, so that Special Access Programs (SAPs) controls will be used only when exceptional security measures are required, based on threat and/or vulnerability (e.g. sensitivity or value of the information) associated with the SAPs. The following sections are excerpts from AR 380–381, and should be reviewed prior to initiation of any SAPs.

### I-2. Restrictions on using Special Access Programs

a. Only approved Prospective Special Access Programs (PSAPs) and SAPs may use the extraordinary security measures outlined in this regulation and AR 380-381.

*b.* Proponents of acquisition, intelligence, or operations and support activities, who identify a particularly sensitive piece of information that they believe merits SAPs protection, should report this information, through their chain of command, for a security policy review. If a determination is made that the information warrants SAPs controls, the MACOM/Program Executive Office (PEO) will report this to the Chief, Technology Management Office (TMO), who coordinates a security review at HQDA. Some examples of potential SAPs are-

(1) A specific technology with potential for weaponization that gives the United States a significant technical lead or tactical advantage over potential adversaries.

(2) Sensitive technology that is especially vulnerable to foreign intelligence exploitation without special protection.

(3) An emerging technology, proposed operation, or intelligence activity risking the compromise of other SAPs.

(4) Exposure of sensitive activities that could jeopardize the lives of U.S. citizens.

(5) A capability that is so unique or sensitive that it requires protection beyond normal procedures.

(6) An extremely sensitive activity requiring special protection from disclosure to prevent significant damage to national security or the reputation or interests of the United States.

(7) Methods used to acquire foreign technology or equipment.

(8) Sensitive support to DOD and non-DOD agencies.

*c*. In compliance with DOD policy, HQDA and its subordinate units and activities will not establish, disestablish, implement, fund, categorize, create carve–out status, or change the mission or scope of a SAP, without written approval of the Deputy Secretary of Defense (DEPSECDEF).

*d*. The Department of Defense National Industrial Security Program Supplement Overprint, dated February 14, 1998, contains further and more specific guidance, especially regarding the enhanced security requirements, procedures, and options involved with the National Industrial Security Program Operating Manual (NISPOM) for:

(1) Critical restricted data (RD) classified at the Secret and Top Secret levels.

(2) Special Access Programs and SAP-type compartmented efforts established and approved by the Executive Branch.

(3) Sensitive Compartmented Information (SCI) or other DCI SAPs-type compartmented programs under the Director of Central Intelligence which protect intelligence sources and methods.

(4) Acquisition, intelligence and operations and support SAPs.

### Section II Responsibilities

### I-3. The Secretary of the Army

The Secretary of the Army (SA) has overall responsibility for SAPs within the Department of the Army. The SA willa. Make recommendations to the DEPSECDEF concerning the establishment, disestablishment, categorization, carve-out status, and changes of mission and scope of Army SAPs.

- b. Ensure adequate oversight of Army SAPs.
- c. Delegate, at SA discretion, management of Army SAPs.

### I-4. The Under Secretary of the Army

The Under Secretary of the Army will-

a. Approve SAPs reprogramming actions.

b. Serve as co-chairman of the SAPs Program Performance and Budget Execution Review System (PPBERS).

## I-5. The Assistant Secretary of the Army (Acquisition, Logistics and Technology)

The Assistant Secretary of the Army (Acquisition, Logistics and Technology (ASA(ALT)) will-

*a*. Serve as the Army Acquisition Executive for all Army programs, including SAPs, and as the principal assistant to the SA for matters relating to Acquisition SAPs (AQ–SAPs).

b. Ensure a single subordinate commander of a MACOM or Program Executive Officer is responsible for each AQ-SAPs throughout its life cycle.

c. Conduct periodic reviews of secure environment contracting conducted in support of SAPs.

d. Ensure SAPs protection and procedures for procurement and fielding of systems, components, and modifications are developed and acquired under SAPs provisions.

e. Coordinate with the Office of the Deputy Chief of Staff for Intelligence (ODCSINT) on issues concerning technology transfer.

f. Coordinate within Army and other DOD components to eliminate duplication of effort and ensure consistent security classification for similar technologies.

g. Coordinate technical review of PSAPs.

h. Evaluate proposed acquisition strategies and plans for Army SAPs.

*i*. Coordinate with Office of the Deputy Chief of Staff for Logistics (ODCSLOG) to integrate logistics support and property accountability considerations into AQ-SAPs efforts and products.

### I-6. The Assistant Secretary of the Army (Manpower and Reserve Affairs)

The Assistant Secretary of the Army (Manpower and Reserve Affairs) (ASA(M&RA)) will-

a. Review and assist in developing policy regarding personnel and personnel security support to SAPs.

*b.* Provide guidance concerning the documentation process to ensure that Tables of Distribution of Allowances (TDA) accurately reflect Army requirements consistent with approved SAPs missions and the Army Authorization Document.

c. Evaluate and approve requests for special pays, as appropriate, in support of SAPs missions.

d. In coordination with the ASA(FM&C), assist in establishing guidance to ensure proper control and accountability of financial data pertaining to Army personnel assigned to SAPs.

#### I–7. The Assistant Secretary of the Army (Financial Management and Comptroller)

The Assistant Secretary of the Army (Financial Management and Comptroller) (ASA(FM&C)) will-

- a. Provide financial and budget policy and guidance for SAPs.
- b. Provide liaison with Congress for SAPs budgets.

c. Coordinate with Defense Finance and Accounting (DFAS) to ensure DFAS provides a secure finance and accounting network to process sensitive financial transactions.

d. Provide financial quality assurance oversight through the Special Review Office (SRO).

e. Coordinate the Army's Budget Estimate Submission for SAPs with OSD.

#### I-8. The Director of Information Systems for Command, Control, Communications, and Computers

The Director of Information Systems for Command, Control, Communications, and Computers (DISC4) will-

- a. Coordinate information systems support for SAPs.
- b. Assist TMO in developing information systems policy for SAPs.
- c. Validate and approve Information System Support Plans (ISSPs).
- d. Through USACECOM Technology Applications Office (TAO):

(1) Provide information management support in preparing Information Systems Requirements Packages (ISRPs) and Information Management Support Plans (IMSPs).

(2) Provide technical advice and support in preparing ISRPs and IMSPs.

## I-9. The General Counsel

The General Counsel (GC) will-

a. Review Army SAPs and prospective Army SAPs for legality and propriety before submission to OSD.

b. Advise the SA on legal and policy issues.

c. Conduct policy reviews.

## I-10. The Inspector General

The Inspector General (TIG) will-

a. Evaluate managerial procedures and practices pertaining to operations, personnel, materiel, funding, secure environment, contracting, and security of SAPs.

b. Identify issues, situations, or circumstances that affect SAPs mission performance.

c. Provide a secure system for program personnel to report fraud, waste, and abuse without fear of reprisal or unnecessary disclosure of information.

d. Conduct non-criminal investigations as directed by the Vice Chief of Staff, Army.

e. Inspect Army SAPs and Army involvement in non-Army SAPs.

f. Develop and coordinate an annual inspection plan with TMO, other inspection/audit agencies, MACOMs, and PEOs.

## I-11. The Auditor General

The Auditor General (TAG) will-

a. Maintain auditors with appropriate clearance and access to perform audits of SAPs.

b. Coordinate with TMO when performing audits of SAPs.

## I-12. Chief of Public Affairs

The Chief of Public Affairs (PA) will-

- a. Staff media queries on SAPs and provide releasable information.
- b. Provide public affairs guidance on SAPs matters.

#### I-13. The Chief of Staff, Army

The Chief of Staff, Army (CSA) will develop, coordinate, review, and conduct oversight of all Army SAPs.

#### I-14. The Vice Chief of Staff, Army

The Vice Chief of Staff, Army (VCSA) will-

*a*. Review SAPs through the Special Access Programs Oversight Committee (SAPOC) and serve as chairman of the SAPOC.

b. Serve as the chairman of the Executive Fix-It Committee.

c. Serve as the co-chairman of the SAPs PPBERS.

d. Provide guidance and direction to Chief, TMO.

## I-15. The Deputy Chief of Staff for Personnel

The Deputy Chief of Staff for Personnel (DCSPER) will-

a. Provide policy on SAPs personnel matters.

b. Coordinate with ODCSOPS to establish procedures ensuring MACOM SAPs properly use allocated personnel spaces to resource the SAPs.

c. Ensure that the U.S. Army Total Personnel Command (PERSCOM) coordinates designated DA approved personnel assignment actions for SAPs.

### I-16. The Deputy Chief of Staff for Intelligence

The Deputy Chief of Staff for Intelligence (DCSINT) will-

a. Oversee Army Intelligence SAPs (IN-SAPs) and serve as IN-SAP Army Staff (ARSTAF) proponent.

b. Establish security, counterintelligence, and intelligence policy for SAPs.

c. Coordinate necessary counterintelligence support for the execution of Army SAPs.

d. Provide OPSEC and threat assessments for responsible MACOMs and program managers to SAPs and present these to the Working SAPOC for inclusion in the SAPOC revalidation briefing.

e. Advise the SAPOC on whether a program or activity warrants SAPs protection.

f. Review SAPs security plans and guides for accuracy and completeness.

g. Provide input as requested to Office of the Deputy Under Secretary of Defense for Policy (Policy Support) (ODUSD(P)(PS)) concerning Army SAPs security classification guides.

h. Coordinate intelligence property issues for Army SAPs with DCSLOG.

i. Coordinate policy for polygraph support to Army SAPs.

*j*. Review and approve disclosure of official Army information (classified and unclassified) for release to foreign governments and international agencies. Coordinate with TMO and Director for Special Programs, Office of the Under Secretary of Defense for Policy (Policy Support) (OTUSD(P)(PS)) for release of information and technology identified by SAPs proponents for release to foreign governments and international agencies.

## I-17. The Deputy Chief of Staff for Operations and Plans

The Deputy Chief of Staff for Operations and Plans (DSCOPS) will-

- a. Oversee Operations and Support SAPs (OS-SAPs) and serve as the ARSTAF proponent.
- b. Provide policy guidance and standards for Operations Security (OPSEC) measures appropriate for Army SAPs.
- c. Develop Army policy and guidance for materiel requirements for SAPs.
- d. Establish and validate Army acquisition priorities for SAPs.

*e*. Coordinate and approve manpower requirements, allocate man-power resources, and prepare Tables Of Distribution And Allowances (TDA) documents for SAPs.

f. Conduct manpower and workload validations of SAPs to support HQDA and PEO/PMs.

g. Task U.S. Army Force Management Support Agency (USAFMSA) to provide necessary support and analysis of SAPs manpower requirements.

## I-18. The Deputy Chief of Staff for Logistics

The Deputy Chief of Staff for Logistics (DCSLOG) will-

- a. Integrate logistics support for all Army materiel development or acquisition for SAPs.
- b. Provide policy guidance on property accountability and logistics support for SAPs.

### I-19. Chief of Engineers

The Chief of Engineers (COE) will provide secure architectural-engineering, construction, real estate, and contracting support to SAPs as required.

### I-20. The Judge Advocate General

The Judge Advocate General (TJAG) will provide legal and policy advice on SAPs matters to CSA and the ARSTAF.

## I-21. Chief of Legislative Liaison

The Chief of Legislative Liaison (CLL) will-

- a. Coordinate congressional briefings on Army SAPs.
- b. Provide required reports to selected congressional committees on Army SAPs.
- c. Assist TMO in updating clearance information for individuals in Congress accessed to Army SAPs.
- d. Assist TMO in verifying access of individuals in Congress.

## I-22. Director, Program Analysis and Evaluation Directorate

The Director, Program Analysis and Evaluation Directorate (PAED), will-

a. Ensure that SAPs compete with other Army programs for resources in the Program Objective Memorandum (POM) development process.

b. Coordinate with the ARSTAF and TMO to develop SAPs program funding profiles and provide copies of approved profiles to TMO.

- c. Provide program analyses for reprogramming actions.
- d. Coordinate SAPs POM during the program review process with OSD.

## I-23. Chief, Technology Management Office

The Chief, Technology Management Office (TMO), will-

a. Serve as the Army primary point of contact for the management and oversight of Army and Army-supported SAPs.

b. Establish policy for the management of SAPs.

- c. Coordinate establishment, maintenance, and disestablishment of SAPs.
- d. Act as the approval authority for establishment and disestablishment of Army P-SAPs.

e. Approve the creation or closure of SAPs subcompartments when there is no change to the categorization, carve-out status, mission, or scope of the parent SAPs. In cases where creation or closure of a subcompartment will

change the mission or scope of the parent SAPs, Chief, TMO will submit the action through SA to the Deputy Secretary of Defense for approval.

- f. Serve as the Executive Secretary for the SAPOC, SAPs PPBERS, and Fix-It committees.
- g. Provide quarterly update reviews to the senior Army leadership.
- h. Assist CLL in coordinating congressional SAPs access briefings and congressional notifications.
- *i*. Monitor budget and financing associated with SAPs.

*j.* Review SAPs for compliance with authorizations, legal constraints, funding, and continued enhanced security measures.

- k. Serve as the POC for Army sensitive support to DOD and non-DOD agencies.
- l. Maintain a registry of Army involvement in SAPs and sensitive activities.
- m. Maintain the Army baseline billet roster.
- n. Coordinate indoctrination of ARSTAF principals to Army SAPs.
- o. Direct the Sensitive Records and Information Agency (SRIA).

### I-24. Commanding General, U.S. Army Training and Doctrine Command

The Commanding General, U.S. Army Training and Doctrine Command (CG, TRADOC), will-

a. Institute procedures to ensure early identification and protection of combat developments, concepts, and systems with SAPs potential.

b. Identify support requirements for SAPs-developed products deployed to the field.

## I-25. Commanding General, U.S. Army Materiel Command

Commanding General, U.S. Army Materiel Command (CG, AMC), will-

a. Institute procedures to ensure early identification and protection of potential research and development (R&D) breakthroughs that may warrant SAPs protection.

- b. Conduct appropriate technology feasibility reviews of AMC SAPs.
- c. Provide support and oversight for AMC SAPs.

## I-26. Commanding General, Forces Command

The Commanding General, Forces Command (CG, FORSCOM), will institute procedures to ensure early identification and protection of activities, operational concepts, and combat developments requiring SAPs status.

#### I-27. Commanding General, U.S. Army Space and Missile Defense Command

The Commanding General, U.S. Army Space and Missile Defense Command (USASMDC), will institute procedures to ensure early identification and protection of potential R&D breakthroughs within the USASMDC that may warrant SAPs protection and coordinate potential release of SAPs information through DA, DCSINT to SAAL–SO and TMO prior to initiating or engaging in preliminary discussions with a foreign government or international organization. DA DCSINT will coordinate, as required, with Director for Special Programs, OTUSD(P)(PS), prior to any release.

## I-28. Commanding General, U.S. Army Intelligence and Security Command

The Commanding General, U.S. Army Intelligence and Security Command (CG, USAINSCOM), will-

a. Institute procedures to ensure early identification and protection of sensitive intelligence activities that may warrant SAPs protection.

b. Provide dedicated counterintelligence, security, and OPSEC support to commanders, program managers, or heads of DA activities having proponency for Army SAPs or Army supported SAPs. This support includes Counterintelligence (CI) assessments of Army SAPs, P–SAPs and contractor facilities involved in Army SAPs contracts.

*c*. Provide DCSINT with counterintelligence assessments of the threat posed to SAPs by Foreign Intelligence Services (FIS) and technology assessments of foreign research and development efforts related to SAPs technologies. Coordinate with DCSINT to provide this information to organizations and installations supporting SAPs.

*d*. Provide to DCSINT an annual counterintelligence evaluation of the OPSEC and security posture of Army SAPs and Army-supported SAPs.

e. Manage and execute the Army polygraph program in support of SAPs.

f. Provide Technical Surveillance Countermeasures (TSCM), TEMPEST, ADP security and Counter-SIGINT support to SAPs.

g. Conduct security reviews of SAPs disestablishment actions, security plans, and CI support plans.

## I-29. Commanding General, U.S. Army Criminal Investigation Command

The Commanding General, U.S. Army Criminal Investigation Command (CG, USACIC), will-

a. Maintain criminal investigators with appropriate clearances and access to conduct investigations of criminal activity in or directed against SAPs.

b. Maintain effective liaison, through individuals well acquainted with special access procedures, with TMO to ensure quick response to investigative requirements.

c. Conduct criminal investigations in all instances of suspected criminal activity in or directed against Army SAPs in accordance with applicable federal statutes, DODD 5205.7, DODI 5505.2 and AR 195-2.

d. Conduct periodic economic crime threat assessments.

e. Coordinate with TMO to conduct crime prevention surveys on SAPs.

#### I-30. Department of the Army Staff

The Department of the Army (ARSTAF) sections having SAPs proponency or support requirements for SAPs willa. Designate a central point of contact for SAPs.

- b. Provide appropriate staff oversight for the planning, programming, budgeting, and execution of SAPs.
- c. Act as SAPs managers when appointed to do so.

#### I-31. Major Army Commands and Program Executive Officers

The Major Army Commands (MACOMs) and Program Executive Officers (PEOs) who supervise managers of SAPs will-

a. Assist program managers in managing their programs.

- b. Establish internal inspection programs for SAPs.
- c. Conduct periodic property reviews to validate new requirements and document materiel assets in support of SAPs.

d. Coordinate with ARSTAF proponents and DCSINT for SAPs intelligence, counterintelligence, and threat assessments.

e. Ensure that all SAPs are incorporated into the Internal Review and Audit Compliance (IRAC) program as described in chapter 4, AR 380-381.

*f.* Coordinate potential release of SAP information through DA DCSINT to SAAL–SO and TMO prior to initiating or engaging in preliminary discussions with a foreign government or international organization. DA DCSINT will coordinate with Director for Special Programs, OTUSD(P)(PS), prior to any release.

g. Coordinate with USAFMSA for manpower support for TDA documentation.

#### I-32. Program/Project/Product Managers of SAPs

In addition to the duties and responsibilities normally incumbent on managers and those delineated by law and regulation, Program/Project/Product Managers (PMs) of SAPs will-

a. Maintain essential SAPs information including establishment, documentation, security plans, access rosters, and security inspection records.

- b. Plan, prepare, and implement security and OPSEC programs designed to protect critical program information.
- c. Coordinate with TMO for information systems advice and support.
- d. Establish and maintain a viable records management program.
- e. Coordinate with SRIA for records management assistance.
- f. Coordinate with the Defense Security Service (DSS) for industrial facility reviews.

g. Identify, establish, maintain, and forward to the SAPs Archive, SRIA, appropriate historical record files pertaining to the program and its operations.

*h*. Coordinate potential release of SAP information through DA DCSINT to SAAL–SO and TMO prior to initiating or engaging in preliminary discussions with a foreign government or international organization. DA DCSINT will coordinate with Director for Special Programs, OTUSD(P)(PS), prior to any release.

## I-33. Sensitive Records and Information Agency (SRIA)

The Sensitive Records and Information Agency (SRIA) will-

*a.* Operate the Army Records Center for Sensitive Records and Information. Accept, review, process, archive, and destroy Army sensitive records in accordance with DODD 5205.7, AR 380–381, and AR 25–400–2. Respond to requests for information. Conduct records review and disposition.

b. Conduct Army-wide document searches for sensitive information. Compile and prepare document indexes and responsive documents for forwarding to requesting agencies. Coordinate declassification reviews.

c. Maintain the Billet Structure Management System (BSMS) for the Army. Develop, maintain and distribute the BSMS system software. Identify BSMS hardware and software system requirements for Army SAPs. Maintain a consolidated BSMS database for Army SAPs. Ensure consistency between program office databases and SRIA master database through regular periodic updates and data transfers.

#### I-34. Defense Security Service (DSS)

The Defense Security Service (DSS) will-

a. Maintain industrial security inspectors with appropriate clearances and access to perform facility clearance inspections and industrial security reviews of contractor facilities supporting Army SAPs.

b. Coordinate with TMO in the conduct of industrial security reviews.

## Section III SAP Categories and Types

### I-35. SAP Categories

DOD recognizes four generic categories of SAPs: Acquisition (AQ–SAPs), Intelligence (IN–SAPs) and Operations and Support (OS–SAPs), and SCI Programs (SCI–SAPs). Within the Army, ASA(ALT) is the proponent for AQ–SAPs, DCSINT for IN–SAPs, and DCSOPS for OS–SAPs.

a. AQ-SAPs protect sensitive Research, Development, Test, and Evaluation (RDT&E), modification or procurement efforts.

*b.* IN–SAPs protect the planning and execution of especially sensitive intelligence or counterintelligence units or operations, including the collection, analysis, and exploitation of intelligence. IN–SAPs also protect especially sensitive programs to procure and exploit foreign materiel.

c. OS-SAPs protect the planning, execution, and support to especially sensitive military operations. This type of SAPs may protect organizations, property, operational concepts, plans, or activities.

d. SCI-SAPs protect sensitive compartmented information or other Director Central Intelligence programs which protect intelligence sources and methods.

#### I–36. SAP Categories of Protection

*a.* Army SAPs. SAPs categories reflect the sensitivity of protected material and the degree of protection needed beyond collateral security management. Categories I through III delineate programs requiring varying degrees of protection with CAT I being the most sensitive and CAT III being the least sensitive. The Army SAPOC determines the category designation for each SAPs. SAPs categories may change as a program matures based on changing needs for protection or the sensitivity of the information being protected. The categories are-

(1) Category I (CAT I): Army SAPs that are extremely sensitive and include only the most critical technical developments and the most sensitive intelligence and operational activities. Compromise would cause exceptionally grave damage to national security. Only Army CAT I SAPs are eligible for consideration for waived SAP status.

(2) Category II (CAT II): Army SAPs that include critical acquisition, intelligence and operational activities that do not meet the criteria for CAT I. Compromise would cause serious damage to national security.

(3) Category III (CAT III): Army SAPs usually having significant non-Army participation, relatively short duration or are not suitable for a billet structure.

(4) Unless waived by the SAPOC or the DEPSECDEF, if necessary, CAT I, II and III SAPs follow security procedures listed in AR 380-381.

b. Non-Army SAPs.

(1) Category N (CAT-N) is the category designation for SAPs or sensitive activities where the Army is the executive agent but not the sponsor. Security measures for CAT-N SAPs are in accordance with AR 380-381 as amended by Memoranda of Agreement (MOAs) between Army and the program sponsor. CAT-N programs have the same management oversight as Army SAPs unless otherwise directed by the VCSA.

(2) Army elements may also participate in SAPs that are neither sponsored nor executed (primarily) by Army. Approval for, and management of, Army participation in SAPs executed by other organizations are governed by AR 380–381. Security measures for these SAPs are in accordance with the executing organization's security procedures.

#### I-37. SAPs Types

*a.* There are two types of SAPs, acknowledged and unacknowledged. An acknowledged SAPs is a program which may be openly recognized or known, however, specifics are classified within that SAPs. The existence of an unacknowledged SAPs, or an unacknowledged program, will not be made known to any person not authorized for this information.

b. Within DOD, three levels of SAPs protection apply. The three levels are waived SAPs, unacknowledged SAPs, and acknowledged SAPs. These SAPs levels are further explained in DODD 0–5207.7 and DODI 0–5205.11.

#### Section IV Security of DA SAPs

#### I-38. Programs Ineligible for SAP Security

a. The extraordinary security measures approved for use with SAPs may not be used by non-SAPs. No collateral program, including programs with approved limited dissemination controls, may use program access non-disclosure agreements (read-on statements), classified code words for program identification, "Carve-out" contracting, "Special

Access Required" markings or cover sheets, a program billet structure, or personnel security investigative or adjudicative requirements more stringent than those required for a comparable level of classified information. These measures are reserved for SAPs.

b. SAPs derive enhanced security primarily from the restricted access features of these programs. Generally, with respect to other security features (for example, physical security, technical security, and so forth), SAPs are held to the same standards as collateral programs at the same classification level.

### I-39. Physical Security

a. Security level. As a general rule, SAPs base the level of physical security on the classification level of the information processed or stored by the SAPs.

(1) Category I SAPs and SAPs processing or storing Sensitive Compartmented Information (SCI) adhere to standards established by DCID 1/21 and the applicable requirements of DOD 5105.21–M–1, DOD TS–5105.21–M–2, and DOD TS–5105.21–M–3.

(2) Category II and III SAPs that do not process or store SCI will follow AR 380-5.

b. Risk assessment.

(1) Program managers of CAT II and III SAPs conduct risk assessments during the PSAPs process to determine whether they must implement physical security measures above the standards prescribed in AR 380-5.

(2) Program offices coordinate this risk assessment with supporting INSCOM CI elements and include the results in their SAPs security plan.

(3) The risk assessment incorporates-

(a) Intelligence Threat Report. This is a multi-disciplined counterintelligence report that addresses the general and specific collection threat to the program.

(b) Counterintelligence Assessment. This is an in-depth analysis of the counterintelligence factors affecting the program's overall security and CI/OPSEC posture. INSCOM normally supports SAPs by developing Intelligence Threat Reports and the Counterintelligence Assessments for the programs at their request.

(c) Program Security Assessment. Program Managers integrate information from the Intelligence Threat Report and the Counterintelligence Assessment to determine whether the program requires further technical protection. If so, the PM requests a TSCM Survey (see AR 380–381, para 5–5).

c. Two Person Integrity (TPI). TPI is required for Category I SAPs only.

*d.* Entry/Exit Searches. The Entry/Exit Inspection Program (EEIP) is, effective by this regulation, no longer a Department of the Army–wide requirement, however, it remains an effective tool that can be used in command security programs to deter and detect the unauthorized removal of classified information. When effectively implemented the Entry/Exit Inspection Program provides visibility and emphasis to the command security program. Its use is a command option.

#### I-40. Document/Information Security

*a*. Marking authors of classified information mark documents in accordance with chapter 4, AR 380–5. Additionally, authors of SAPs material will–

(1) Mark SAPs nicknames at the top and bottom of the outside front cover, title page, all interior pages and on the outside back cover of all documents (for example, SECRET/BROKEN BRIDGE/SPECIAL ACCESS REQUIRED).

(2) Mark the subcompartment nickname in accordance with subparagraph (1) above (for example, SECRET/ BROKENBRIDGE/STURDY TWIG/SPECIAL ACCESS REQUIRED). The SAPs nickname and the wording "INCLUSIVE" may be used when a report contains information on all subcompartments (for example, SECRET/ BROKENBRIDGE/INCLUSIVE/SPECIAL ACCESS REQUIRED).

(3) Use the caveat "SPECIAL ACCESS REQUIRED" on the top and bottom of all pages that contain SAPs information (for example, SECRET/BROKEN BRIDGE/SPECIAL ACCESS REQUIRED).

(4) Mark the beginning of each paragraph with the initial of the classification and the initials of the unclassified nickname of the SAPs and subcompartment if applicable (for example, S/BB). Programs must include the parent SAPs initials before the subcompartment for portions containing subcompartment information (for example, S/BB/ST). Partial paragraphs which begin a page will be marked with appropriate paragraph markings.

*b*. Further information on marking SAPs, as well as transmission, dissemination, storage, destruction, accountability, receipting, and reproduction, can be found in AR 380–381.

# Glossary

## Section I Abbreviations

AASA Administrative Assistant to the Secretary of the Army

ADP Automated Data Processing

# AMC

U.S. Army Materiel Command

AR Army Regulation

ARNG Army National Guard

ARSTAF Army Staff

ASA(ALT) Assistant Secretary of the Army (Acquisition, Logistics, and Technology)

ASA(FM&C) Assistant Secretary of the Army (Financial Management and Comptroller)

ASA(M&RA) Assistant Secretary of the Army (Manpower and Reserve Affairs)

ASD(C3I) Assistant Secretary of Defense for Command, Control, Communications, and Intelligence

**CCF** Central Clearance Facility

**CCTV** Closed Circuit Television

**CFIUS** Counterintelligence Foreign Influence in the U.S.

CG Commanding General

**CI** Counterintelligence

CIA Central Intelligence Agency

CINC Commander-in-Chief

CLL Chief of Legislative Liaison

CNWDI

Critical Nuclear Weapons Design Information

**COE** Chief of Engineers

**COMINT** Communications Intelligence

COMSEC Communications Security

**CONUS** Continental/Contiguous U.S.

CPO Civilian Personnel Office

CSA Chief of Staff, Army

**DA** Department of the Army

DCI Director of Central Intelligence

**DCSLOG** Deputy Chief of Staff for Logistics

**DCSOPS** Deputy Chief of Staff for Operations

**DCSPER** Deputy Chief of Staff for Personnel

**DEFCON** Defense Readiness Condition

**DEPSECDEF** Deputy Secretary of Defense

**DES** Data Encryption Standard

**DIS** Defense Investigative Service

DNG Downgrade to (used in electronic messages)

**DOD** Department of Defense

**DODD** Department of Defense Directive

**DODI** Department of Defense Instruction

**DOE** Department of Energy DOS Disk Operating System

DTS Defense Transportation System

EO Executive Order

ELINT Electronic Intelligence

**ELSEC** Electronic Security

FAA Federal Aviation Administration

**FAX** Facsimile

FMS Foreign Military Sales

FOA Field Operating Agency

FOIA Freedom of Information Act

**FORSCOM** U.S. Army Forces Command

FOUO For Official Use Only

FRD Formerly Restricted Data

GAO General Accounting Office

GC General Counsel

GPO Government Printing Office

GSA General Services Administration

JANAP Joint Army-Navy-Air Force Publication

JCS Joint Chiefs of Staff

JOPS Joint Operation Planning System JSCP Joint Strategic Capabilities Plan

JSPDSA Joint Strategic Planning Document Supporting Analysis

MACOM Major Army Command

MARKS The Modern Army Recordkeeping System

MOA Memoranda of Agreement

MOU Memoranda of Understanding

MSC Major Subordinate Command

NATO North Atlantic Treaty Organization

NCA National Command Authority

NDP National Disclosure Policy

NIE National Intelligence Estimate

NSA National Security Agency

O&M Operation and Maintenance

**OCONUS** Outside Continental U.S.

**ODCSOPS** Office of the Deputy Chief of Staff for Operations

OIC Officer in Charge

OMB Office of Management and Budget

**OPF** Official Personnel File

**OPSEC** Operations Security

PA Public Affairs PCU Premise Control Unit

**POE** Port of Embarkation

**RD** Restricted Data

**RDTE** Research, Development, Test, and Evaluation

SAEDA Subversion and Espionage Directed Against the U.S. Army

SCI Sensitive Compartmented Information

SF Standard Form

SIGINT Signals Intelligence

SIGSEC Signals Security

SNIEs Special National Intelligence Estimates

**TDA** Tables of Distribution and Allowances

**TDY** Temporary Duty

TIG The Inspector General

TJAG The Judge Advocate General

TOE Tables of Organization and Equipment

TRADOC U.S. Army Training and Doctrine Command

TS TOP SECRET

**TSCO** TOP SECRET Control Officer

**TSEC** Telecommunications Security

USC United States Code UCMJ Uniform Code of Military Justice

USAR

U.S. Army Reserve

VCSA Vice Chief of Staff, Army

WNRC Washington National Records Center

Section II Terms

Access

The ability and opportunity to obtain knowledge of classified information.

## Acknowledged Special Access Programs (A-SAPs)

A SAP whose existence is known, to include association with another classified program, and which is publicly acknowledged.

### Agency

An organization specified as such in EO 12958, as amended by EO 12972. The Army is an agency (an Army command is not an agency, but rather is part of an agency, the Army). Within the Department of Defense (DOD), this term includes the DOD, and the Department of the Army, the Department of the Navy, and the Department of the Air Force.

### **Applicable Associated Markings**

Markings, other than those which designate classification level, that are required to be placed on classified documents. These include the "classified by" line, downgrading and declassification instructions, special warning notices, Special Access Program caveats, etc.

## Automated Information System (AIS)

An assembly of computer hardware, software, or firmware configured to collect, create, communicate, compute, disseminate, process, store, or control data or information.

#### Automatic Declassification

The declassification of information based solely upon: (1) the occurrence of a specific date or event as determined by the original classification authority; or (2) the expiration of a maximum time frame for duration of classification established under EO 12958 (unless an exception or exemption to the automatic declassification has been authorized).

#### **Carve-Out**

A classified contract for which the Defense Security Service has been relieved of inspection responsibility in whole or in part.

#### Classification

The act or process by which information is determined to be classified information.

### **Classification Guidance**

Any instruction or source that prescribes the classification of specific information.

### **Classification Guide**

A documentary form of classification guidance issued by an original classification authority that identifies the elements of information regarding a specific subject that must be classified and establishes the level and duration of classification for each such element.

### Classified National Security Information (or "classified information")

Information and material that has been determined, pursuant to EO 12958 or any predecessor order, to require protection against unauthorized disclosure and is marked to indicate its classified status when in documentary or readable form.

## Classifier

An individual who makes a classification determination and applies a security classification to information or material. A classifier may be an original classification authority or a person who derivatively assigns a security classification based on a properly classified source or a classification guide.

### Code Word

A single word assigned a classified meaning by appropriate authority to ensure proper security concerning intentions and to safeguard information pertaining to actual, real-world military plans or operations classified as CONFIDEN-TIAL or higher.

### **Collateral Information**

Information identified as National Security Information under the provisions of EO 12958 but which is not subject to enhanced security protection required for SAP Information.

### Command

Headquarters, Department of the Army (HQDA) to include the Office of Secretary of the Army and the Army Staff, Major Commands (MACOMs), Major Subordinate commands and other organizations formed within the Army to support HQDA or a MACOM.

### **Communications Security (COMSEC)**

The protection resulting from all measures designed to deny unauthorized persons information of value which might be derived from the possession and study of telecommunications and to ensure the authenticity of such communications. COMSEC includes cryptosecurity, emission security, transmission security, and physical security of COMSEC material and information.

### Compilation

Items of information that are individually unclassified can be classified if the combined information reveals an additional association or relationship that matches criteria for classification pursuant to section 2–8 of this regulation.

#### Compromise

An unauthorized disclosure of classified information.

## CONFIDENTIAL

Classification that will be applied to information in which the unauthorized disclosure could reasonably be expected to cause damage to the national security.

## **Confidential Source**

Any individual or organization that has provided, or that may reasonably be expected to provide, information to the United States on matters pertaining to the national security with the expectation that the information or relationship, or both, are to be held in confidence.

## Continental (or "Contiguous") United States (CONUS)

The 48 contiguous states in the United States territory, including adjacent territorial waters, located within the North American continent, between Canada and Mexico.

## Controlled Cryptographic Item (CCI)

A secure telecommunications or information handling equipment ancillary device, or associated cryptographic component, which is unclassified but controlled. (Equipment and components so designated bear the designator "Controlled Cryptographic Item" or CCI.)

#### **Controlled Unclassified Information (CUI)**

Other types of information that require application of controls and protective measures, for a variety of reasons, not to include those that qualify for formal classification.

### Counterintelligence (CI)

Those activities which are concerned with identifying and counteracting the threat to security (of the U.S. Army and U. S. Government to include, but not limited to, its technology or industrial base) posed by foreign intelligence services or organizations, or by individuals engaged in espionage, sabotage, sedition, subversion, or terrorism.

## Critical Nuclear Weapon Design Information (CNWDI)

That TOP SECRET Restricted Data or SECRET Restricted Data revealing the theory of operation or design of the

components of a thermonuclear or implosion-type fission bomb, warhead, demolition munition or test device. Specifically excluded is information concerning arming, fusing, and firing systems; limited life components; and total contained quantities of fissionable, fusionable, and high explosive materials by type. Among these excluded items are the components which DOD personnel set, maintain, operate, test, or replace.

### Cryptoanalysis

The analysis of encrypted messages; the steps or processes involved in converting encrypted messages into plain text without initial knowledge of the system of key employed in the encryption.

### Cryptography

The branch of cryptology which treats the principles, means, and methods of designing and using cryptosystems.

### Cryptology

The branch of knowledge which treats the principles of cryptography and cryptoanalytics; and the activities involved in producing signals intelligence (SIGINT) and maintaining communications security (COMSEC).

### **DA** Personnel

Includes any active or reserve military personnel or National Guard, assigned or attached to a Department of the Army installation or activity and persons employed by, assigned to, or acting for an activity within the Department of the Army, including contractors, licensees, certificate holders, and grantees, and persons otherwise acting at the direction of such an activity.

#### Damage to the National Security

Harm to the national defense or foreign relations of the United States from the unauthorized disclosure of information.

#### Declassification

The authorized change in the status of information from classified information to unclassified information.

#### **Declassification Authority**

An official position, in writing, not assigned to a person by name, applied to:

a. The official who authorized the original classification, if that official is still serving in the same position;

- b. The originator's current successor in function;
- c. A supervisory official of either; or

d. Officials delegated declassification authority in writing by the agency head (Secretary of the Army) or the senior agency official (DAMI-CH).

#### **Declassification Guide**

Written instructions issued by a declassification authority that describes the elements of information regarding a specific subject that may be declassified and the elements that must remain classified.

#### **Derivative Classification**

The process of determining whether information has already been originally classified and, if it has, ensuring that it continues to be identified as classified by marking or similar means when included in newly created material.

#### Document

Any physical medium in or on which information is recorded or stored, to include written or printed matter, audiovisual materials, and electromagnetic storage medium or media.

## DOD

Abbreviation for the Department of Defense.

#### **DOD** Component

The Office of the Secretary of Defense (OSD), the Military Departments, the Organization of the Joint Chiefs of Staff (OJCS), the Unified Combatant Commands, and the Defense Agencies. The Army is a DOD Component.

#### Downgrading

A determination that information classified at a specific level will be classified at a lower level.

## Entry/Exit Inspection Program (EEIP)

An inspection program to deter and detect unauthorized introduction or removal of classified material from DOD

owned or leased installations and facilities. This program did not replace existing programs for facility and installation security and law enforcement inspection requirements. This requirement has since been cancelled.

#### Event

An occurrence or happening that is reasonably certain to occur and which can be set as the signal for automatic declassification of information.

#### **Exercise Term**

A combination of two words, normally unclassified, used exclusively to designate a test, drill, or exercise. An exercise term is employed to preclude the possibility of confusing exercise directions with actual operations directives.

#### **File Series**

Documentary material, regardless of its physical form or characteristics, that is arranged in accordance with a filing system or maintained as a unit because it pertains to the same function or activity.

#### **Foreign Government Information**

Information, either in soft-copy or hard-copy form, that is:

a. Government or governments, an international organization of governments, or any element thereof, with the expectation that the information, the source of the information, or both, are to be held in confidence;

b. Information produced by the United States pursuant to or as a result of a joint arrangement with a foreign government or governments, or an international organization of governments, or any element thereof, requiring that the information, the arrangement, or both, are to be held in confidence; or

c. Information received and treated as "Foreign Government Information" under the terms of a predecessor order to EO 12958.

### Formerly Restricted Data (FRD)

Information removed from the Restricted Data category upon a joint determination by the Department of Energy (or antecedent agencies) and the Department of Defense that such information relates primarily to the military utilization of atomic weapons and that such information can be safeguarded adequately as classified defense information. For purposes of foreign dissemination, this information is treated in the same manner as Restricted Data.

#### Hard Copy

Printed format output of a document.

### Homepage (or "home page")

The single, top-level, webpage designed to be the first file accessed by a user visiting a website; also known as an "index" or "default" page.

#### Information

Any knowledge that can be communicated or documentary material, regardless of its physical form or characteristics, that is owned by, produced by or for, or is under the control of the United States Government. "Control" means the authority of the agency that originates information, or its successor in function, (proponent), to regulate access to the information.

#### Information Security (INFOSEC)

The system of policies, procedures, and requirements established under the authority of EO 12958 to protect information that, if subjected to unauthorized disclosure, could reasonably be expected to cause damage to the national security.

#### Infraction

Any knowing, willful, or negligent action contrary to the requirements of EO 12958 or its implementing directives that does not comprise a violation as defined below.

#### Integrity

The state that exists when information is unchanged from its source and has not been accidentally or intentionally modified, altered, or destroyed.

### **Intelligence** Activity

An activity that an agency within the Intelligence Community is authorized to conduct under EO 12333.

### Lost Cluster

A group of disk sectors. The operating system assigns a unique number to each cluster and then keeps track of files according to which clusters they use. Occasionally, the operating system marks a cluster as being used even though it is not assigned to any file. This is called a lost cluster.

#### Mandatory Declassification Review

Review for declassification of classified information in response to a request for declassification that meets the requirements under section 3.6 of EO 12958.

### Material

Any product or substance on or in which information is embodied or associated.

### **Multiple Sources**

Two or more source documents, classification guides, or a combination of both.

#### **National Security**

The national defense or foreign relations of the United States.

#### Need-to-Know (or "need-to-know")

A determination made by an authorized holder of classified information that a prospective recipient requires access to specific classified information in order to perform or assist in a lawful and authorized governmental function.

#### Network

A system of two or more computers that can exchange data or information.

#### Nickname

A combination of two separate unclassified words that is assigned an unclassified meaning and is employed only for unclassified administrative, morale, or public information purposes.

### **Operations Security (OPSEC)**

The process of denying adversaries information about friendly capabilities and intentions by identifying, controlling, and protecting indicators associated with the planning and conducting of military operations and other activities.

#### **Original Classification**

An initial determination that information requires, in the interest of national security, protection against unauthorized disclosure.

### **Original Classification Authority (OCA)**

An individual's position, which has been authorized in writing, either by the President, Secretary of the Army, or the HQDA, DCSINT, to originally classify information up to and including a certain classification level.

#### **Permanent Historical Value**

Those records that have been identified in an agency records schedule as being permanently valuable. For Army records, see AR 25–400–2, The Modern Army Recordkeeping System (MARKS).

#### **Personal Identifier**

Any grouping of letters or numbers, used in an organization code, that the command uses to identify a particular position.

#### Prospective Special Access Programs (P-SAPs)

An Army program or activity for which enhanced security measures have been proposed and approved to facilitate security protection prior to establishing the effort as an Army SAPs.

#### **Protective Security Service**

A transportation protective Service provided by a cleared commercial carrier qualified by the Military Traffic Management Command (MTMC) to transport shipments up to and including the SECRET classification level. The carrier must provide continuous attendance and surveillance of the shipment by qualified carrier representatives and maintain a signature and tally record. In the case of air movement, however, observation of the shipment is not required during the period it is stored in the carrier's aircraft in connection with flight, provided the shipment is loaded into a compartment that is not accessible to an unauthorized person aboard. Conversely, if the shipment is loaded into a compartment of the aircraft that is accessible to an unauthorized person aboard, the shipment must remain under the constant surveillance of a cleared escort or qualified carrier representative.

## Regrade

To raise or lower the classification assigned to an item of information.

## Restricted Data (RD)

All data concerning:

- a. Design, manufacture or utilization of atomic weapons;
- b. The production of special nuclear material; or

c. The use of special nuclear material in the production of energy, but will not include data declassified or removed from the Restricted Data category under section 142 of the Atomic Energy Act of 1954, as amended.

### Safeguarding

Measures and controls that are prescribed to protect classified information. Safeguarding and protection of classified information are synonymous terms.

### SECRET

Level of classification that will be applied to information in which the unauthorized disclosure could reasonably be expected to cause serious damage to the national security.

### Security Clearance

A determination that a person is eligible under the standards of AR 380–67 (implementation for DOD Directive 5200. 2–R) for access to classified information.

### Security In–Depth

A determination by an authorized command official (Commander or Security Manager or other official where so designated) that a facility's security program consists of layered and complementary security controls sufficient to deter and detect unauthorized entry and movement within the facility. Examples include, but are not limited to use of perimeter fences, employee and visitor access controls, use of an Intrusion Detection System, random guard patrols throughout the facility especially during non-working hours, closed circuit video monitoring or other safeguards that mitigate the vulnerability of unalarmed storage areas and security storage cabinets during non-working hours.

#### Self–Inspection

The internal review and evaluation of the individual command's activities and the command as a whole (to include subordinate elements) with respect to the implementation of the Information Security Program and this regulation which implements the requirements established under EO 12958 and its implementing directives.

#### Senior Agency Official (or "Senior Official")

An official appointed, in writing, by the head of an agency, under the provisions of section 5.6.c of EO 12958, to be responsible for direction and administration of the Information Security Program. Within the Department of the Army, the Secretary of the Army has appointed the HQDA, Deputy Chief of Staff for Intelligence (DCSINT) as the Senior Agency Official.

### Senior Official of the Intelligence Community (SOIC)

The head of an agency, office, bureau, or other intelligence element as identified in section 3 of the National Security Act of 1947, as amended, 50 USC 401a(4), and section 3.4(f) (1 through 6) of Executive Order 12333.

## Sensitive But Unclassified (SBU)

Information originated within the Department of State which warrants a degree of protection and administrative control and meets the criteria for exemption from mandatory public disclosure under the Freedom of Information Act.

## Sensitive Compartmented Information (SCI)

Classified information concerning or derived from intelligence sources, methods, or analytical processes, which is required to be handled within formal access control systems established by the Director of Central Intelligence.

### Sensitive Compartmented Information Facility (SCIF)

A physical space which has been designated as an area containing safeguards and security interests which dictate the need for the imposition of physical protection measures, as a minimum entailing control of access to and from the

designated area, in order to protect DA interests. Functionally specialized security areas where measures applicable to each type of security area are tailored to the protection needs of the security interests contained therein.

#### Slack Space

Data that is located in the space, or the empty space itself, between the "End of File" marker and the end of the cluster. The DOS and Windows file systems use fixed-size clusters. DOS and older Windows systems use a 16-bit file allocation table (FAT), which results in very large cluster sizes for large partitions. Even if the actual data being stored requires less storage than the cluster size, an entire cluster is reserved for the file. Unless a file is EXACTLY one or more clusters in length, which is unlikely, there will always be space between the file's "End of File" marker and the end of the cluster associated with that file. Any data that is located in that space, or the empty space itself, is referred to as file slack.

### Soft Copy

A document that is in digital format, either on an AIS or storage media.

#### Source Document

An existing document that contains classified information that is incorporated, paraphrased, restated, or generated in new form into a new document.

#### Special Access Programs (SAPs)

Any DOD program or activity (as authorized in E.0. 12958), employing enhanced security measures (e.g., safeguarding, access requirements, etc.) exceeding those normally required for information at the same level. The term "activity" is not in this section meant to be construed as organizational activity (such as a command). Policy and procedures for enhanced protection measures for that command that do not exceed beyond a command are at command discretion and are not to be confused with SAPs. See AR 380–381 for Army policy on SAPs.

#### **Special Activity**

An activity, or functions in support of such activity, conducted in support of national foreign policy objectives abroad that is planned and executed so that the role of the United States is neither apparent nor acknowledged publicly; but that is not intended to influence US political processes, public opinion, policies, or media, and does not include diplomatic activities or the collection and production of intelligence or related support functions.

#### Systematic Declassification Review

The review process for declassification of classified information contained in records that have been determined by the Archivist of the United States ("Archivist") to have permanent historical value in accordance with chapter 33 of title 44, United States Code.

#### Telecommunications

The preparation, transmission, or communication of information by electronic means.

#### **TCP/IP** Network

A data communication network that uses Transport Control Protocol/Internet Protocol (TCP/IP); the public internet and the DOD Non-Classified IP Router Network (NIPRNET) are examples of TCP/IP networks.

#### TOP SECRET

Level of classification that will be applied to information in which the unauthorized disclosure could reasonably be expected to cause exceptionally grave damage to the national security.

#### **Two–Person Integrity (TPI)**

Condition by which an individual was allowed to work with TOP SECRET material as long as another cleared individual was present. In the past, because of the extreme importance to the national security of Top Secret information and information controlled within approved Special Access Programs, employees were not permitted to work alone in areas where such information is in use or stored and accessible by those employees. This requirement has since been cancelled.

#### Unacknowledged Special Access Programs (U-SAPs)

A SAPs, the existence of which is not acknowledged, affirmed, or made known to any person not authorized for access.

#### **Unallocated Space**

The space on a storage medium that the disk operating system (DOS) regards as available for use when needed. As far

as DOS is concerned, the space is empty. In actuality the space may contain deleted files, which are not gone until they are overwritten, or partial fragments of old files to include old file slack.

### Unauthorized Disclosure

A communication or physical transfer of classified information to an unauthorized recipient.

### Upgrade

To raise the classification of an item of information from one level to a higher one.

### Using Component

The DOD Component to which a code word is allocated for use, and which assigns to the word a classified meaning, or which originates nicknames and exercise terms using the procedure established by the Joint Chiefs of Staff.

### Violation

Condition by which:

a. Any knowing, willful, or negligent action that could reasonably be expected to result in an unauthorized disclosure of classified information;

b. Any knowing, willful, or negligent action to classify or continue the classification of information contrary to the requirements of EO 12958 or its implementing directives; or

c. Any knowing, willful, or negligent action to create or continue Special Access Programs contrary to the requirements of EO 12958.

### Waived Special Access Programs (W-SAPs)

A SAPs for which the Secretary of Defense has waived applicable reporting requirements of section 1 19. Title 10, USC, is identified as a "Waived SAPs" and therefore has more restrictive reporting and access controls.

### Webpage (or "web page")

An individual HTML-compliant electronic file accessible through a TCP/IP network.

### Website (or "web site")

A collection of HTML-compliant electronic files designed to provide information, services, or goods to users through a TCP/IP network.

## Section III

### Special Abbreviations and Terms

This publication uses the following abbreviations, brevity codes, or acronyms not contained in AR 310–50. These include use for electronic publishing media and computer terminology.

#### AECS

Automated Entry Control System

#### AIMA

Army Intelligence Materiel Activity

## AIRMP

Army Information Resources Management Program

## AIS

Automated Information Systems

AQ-SAP Acquisition SAP

AT/FP Antiterrorism/Force Protection

**B&P** Bid and Proposal BMS Balanced Magnetic Switch

BSMS Billet Structure Management System

C CONFIDENTIAL

C3I Command, Control, Communications & Intelligence

CAPCO Controlled Access Program Coordination Office

CCO Handle Via COMINT Channels Only

CGI Common Gateway Interface

CLAS Classified by (used in electronic messages)

**COMPUSEC** Computer Security

CS Chief of Staff

CSA Cognizant Security Agency

CSM Command Security Manager

CSS Constant Surveillance Service

CUI Controlled Unclassified Information

DAMI-CH Chief of the Counterintelligence, Human Intelligence Division

DCAA Defense Contract Audit Agency

DCID Director of Central Intelligence Directive

DCMC Defense Contract Management Command

DCS Defense Courier Service

DCSINT Deputy Chief of Staff for Intelligence

## DEA

Drug Enforcement Administration

## DECL

Declassify on (used in electronic messages)

## DERV

Derived from (used in electronic messages)

## DISC4

Director of Information Systems for Command, Control, Communications, and Computers

## DODPSP

Department of Defense Personnel Security Program

DSI Defense Security Institute

DSS Defense Security Service

**DSSTO** Defense Security Service Training Office

DTIC Defense Technical Information Center

EEIP Entry/Exit Inspection Program

**E-mail or e-mail** Electronic Mail

ENDEX End of Exercise

**FAT** File Allocation Table

FGI Foreign Government Information

FIS Foreign Intelligence Services

FISINT Foreign Instrumentation Signals Intelligence

GILS Government Information Locator Service

GO/SES General Officer/Senior Executive

HQDA Headquarters, Department of the Army

HTML Hypertext Markup Language HVCCO Handle Via COMINT Channels Only

IACSE Interagency Advisory Committee on Security Equipment

**ID** Identification

IDE Intrusion Detection Equipment

IDS Intrusion Detection System

IMA Intelligence Materiel Activity

IMD Intelligence Materiel Division

INFOSEC Information Security/Information Systems Security

IN-SAPs Army Intelligence SAPs

IR&D Independent Research and Development

ISCAP Interagency Security Classification Appeals Panel

ISDN Integrated Service Digital Network

ISOO Information Security Oversight Office

ISPAC Information Security Policy Advisory Council

ISS Information Systems Security

IRR Individual Ready Reserve

JER Joint Ethics Regulation

JSPS Joint Strategic Planning System

LAN Local Area Network

LOA Letter of Offer and Acceptance LOU Limited Official Use (obsolete)

MCA Management Control Administrators

MSM Magnetic Storage Media

NACSI National Communications Security Instruction

NC Not releasable to Contractors/Consultants

NDA Nondisclosure Agreement

NFESC Naval Facilities Engineering Service Center

NIPRNET Non-Classified IP Router Network

NISP National Industrial Security Program

NISPOM National Industrial Security Program Operating Manual

NISPOMSUP National Industrial Security Program, Operating Manual Supplement

NIST National Institute of Standards

**NOCONTRACT** Not Releasable to Contractors/Consultants

**NOFORN** Not Releasable to Foreign Nationals

NRC Nuclear Regulatory Commission

NSTS National Secure Telephone System

NTISSC National Telecommunications and Information Systems Security Committee

OADR Originating Agency's Determination Required (obsolete)

OASD(C3I) Director, Counterintelligence and Security Programs

OC

Dissemination and extraction of information controlled by originator

OCA Original Classification Authority

**ODUSD(P)** Office of the Deputy Under Secretary of Defense (Policy)

OJCS Office of the Joint Chief Secretariat

**ORCON** Dissemination and extraction of information controlled by originator

**OS–SAP** Operations and Support SAP

OUSD Office of the Under Secretary of Defense

PAED Director, Program Analysis and Evaluation Directorate

PEO Program Executive Office

PIN Personal Identification Number

**PR** Proprietary Information

**PROPIN** Proprietary Information Involved

**PSS** Protective Security Service

**REL TO** Authorized for release to

S SECRET

S&G Sargent and Greenleaf

SA Secretary of the Army

SAPOC SAP Oversight Committee

SAPs Special Access Programs

SBU Sensitive But Unclassified

SCG Security Classification Guide SCI-SAP SCI Programs SAP

SEALS Security Equipment and Locking Systems

**SECARMY** Secretary of the Army

SNM Special Nuclear Material

SOIC Senior Official of the Intelligence Community

SPB Security Policy Board

SPM Special Program Manager

SRG Senior Review Group

SRIA Sensitive Records and Information Agency

SSO Special Security Office/Officer

STU Secure Telephone Unit

**TA** Traffic Analysis

TAG The Auditor General

TCP/IP Transport Control Protocol/Internet Protocol

TDS Technical Data Sheet

**TPI** Two–Person Integrity

**TSCM** Technical Surveillance Countermeasures

U UNCLASSIFIED

UCNI Unclassified Controlled Nuclear Information

UN United Nations URL Universal Resource Locator

USACIC U.S. Army Criminal Investigation Command

USAINSCOM U.S. Army Intelligence and Security Command

USASMDC U.S. Army Space and Missile Defense Command

USD(P) Under Secretary of Defense for Policy

USD(SP) Under Secretary of Defense (Special Programs)

W3C World Wide Web Consortium

WAN Wide Area Network

Webmaster Website Manager

WHS Washington Headquarters Service

WNINTEL Warning Notice – Intelligence Sources or Methods Involved (obsolete)

WWW World Wide Web

# Index

This index is organized alphabetically by topic and subtopic. Topics and Subtopics are identified by paragraph number.

380 - 5Abbreviations, 1-3 Applicability, 1-10 Legal authority, 1–12 Principles, 1-11 Purpose, 1-1 Recordkeeping requirements, 1-13 References, 1-2 Requirements for recordkeeping, 1-13 Terms, 1-3 What is also addressed, 1-1 What it does not implement, 1-1 What it establishes, 1-1 What it implements, 1-1 Abbreviations 380-5, 1-3 Absences Inquiries, 10-9 Unauthorized, 10-9 Access Classified information by persons outside the Executive Branch, 6-8 CNWDI, 6-7 Congress, 6-8 Combinations, 7-8 Computer Security Act of 1987, 5-21 Critical Nuclear Weapons Design Information, 6-7 DEA sensitive information, 5-13 Debriefing, 6-5 DOD UCNI, 5-17 DOD Unclassified Controlled Nuclear Information, 5-17 Drug Enforcement Administration sensitive information, 5-13 Emergency request to Department of Energy information, 6-17 Emergency request to DOE information, 6-17 Former Presidential appointees, 6-8 Formerly Restricted Data, 6-7 FRD, 6-7 FOUO information, 5-4 GAO, 6-8 General Accounting Office, 6-8 Government Printing Office, 6-8 GPO, 6-8 Historical researchers, 6-8 Information – by persons outside the Executive Branch, 6-8 Joint Chiefs of Staff papers, 4-46 Judicial proceedings, 6-8 Open storage area control, 7-19 Other situations, 6-8 RD, 6-7 Request to Department of Energy information, 6-17 Request to DOE information, 6-17 Restricted Data, 6-7 SBU, 5-9 Sensitive But Unclassified, 5-9

Sensitive information (Computer Security Act of 1987), 5-21 Termination, 6-5 UCNI, 5-17 Unauthorized access debriefing, 10-6 Unclassified Controlled Nuclear Information, 5-17 When authorized, 1-11 Access request To Department of Energy information, 6-17 To DOE information, 6-17 Accompanying documentation Destruction, 6-29 Disposition, 6-27 Accountability Keys, 7-4 Locks, 7-4 TOP SECRET information, 6-21 Accountability and administrative procedures CONFIDENTIAL information, 6-22 Foreign government material information, 6-23 NATO information, 6-23 SECRET information, 6-22 TOP SECRET information, 6-21 Working papers, 6-24 Acquisition systems Classification, 2–14 Addressing Preparation of material, 8-10 Administration Information Security Oversight Office, 1-15 ISOO, 1-15 Advice Destruction - technical, 3-17 Technical regarding destruction, 3-17 AECS, 7–19 Agency of origin Marking, 4–5 Air Transportation Security Field Offices, 8–15 AIS Exceptions to marking, E-2 General, E-1 Marking exceptions, E-2 Misconceptions, 4–32 Printed documents, E-2 Security procedures, E-1 Warning notices, 4-12 AIS storage media Marking, 4–32 AIS storage media - fixed and internal Marking, 4–33 Alternatives Preliminary inquiry judgement, 10-3 Annual requirement SF 311, 1-7 Applicability 380-5, 1-10 Application

MARKS, 4-15 Modern Army Recordkeeping System, 4-15 Sanctions, 1-21 Appointment Derivative classification, 2-1 Inquiry investigator, 10-3 Appropriate techniques and methods Destruction, 3-16 Approval Classification guides, 2-18 Security classification guides, 2-18 **Approval authority** Conferences, 6-18 Meetings, 6-18 Reproduction, 6-26 Approved methods Destruction, 3-15 Approving authority For waiver, 1–19 **Army Declassification Program** General, 3-1 **Army Service Schools** Release of Joint Chiefs of Staff papers, 4-51 Assessment Damage assessment reevaluation, 10-5 Assignment Classified meanings to code words, H-8 Notice of code word, H-9 Assistance Regarding physical security storage, 7-9 Assistant Secretary of the Army (Acquisition, Logistics, and Technology (ASA(ALT)) Special Access Programs (SAPs) responsibilities, I-5 Assistant Secretary of the Army (Financial Management and Comptroller) (ASA(FM&C)) Special Access Programs (SAPs) responsibilities, I-7 Assistant Secretary of the Army (Manpower and Reserve Affairs) (ASA(M&RA)) Special Access Programs (SAPs) responsibilities, I-6 Association-related classified meetings Conferences, 6–18 Meetings, 6-18 Atomic Energy Act of 1954, 2-15 Atomic energy information, 1-16 Auditor General (AG) Special Access Programs (SAPs) responsibilities, I-11 Authority 380-5, 1-12 For waiver approval, 1-19 Release, 6–8 Reproduction approval, 6-26 Authority and purpose Downgrading, 3-11 Authority delegation OCA, 2–3 Original Classification Authority, 2-3 Authority for approval Conferences, 6-18 Meetings, 6-18

Automated Entry Control System (AECS), 7-19 Automated Information Systems (AIS) Exceptions to marking, E-2 General, E-1 Marking exceptions, E-2 Misconceptions, 4-32 Printed documents, E-2 Security procedures, E-1 Warning notices, 4-12 Automatic declassification Review, 3-5 Automatic declassification exemption List of qualifiers, 2-11 Qualifier list, 2-11 Automatic Declassification System Exemption list, 3–6 Exemption request for other than listed, 3-6 Exemption requirements, 3-6 Foreign documents, 3-4, 3-6 Foreign government information, 3-4, 3-6 General, 3-5 Marking of exempted documents, 3-7 Requirements, 3-5 Requirements for exemption, 3-6 **Biometrics devices**, 7–19 **Blueprints** Marking, 4–26 **Briefing** CNWDI, 9-13 Critical Nuclear Weapons Design Information, 9-13 Foreign Travel. 9-8 On refusal to sign NDA, 9-5 On refusal to sign Nondisclosure Agreement, 9-5 On refusal to sign SF 312, 9-5 Other special security education briefings, 9-14 Bulk quantities of material Remarking, 4-38 **Bulky** material Consignor/consignee responsibilities for shipment, 8-16 Storage, 7-4 Burning Destruction. 3–15 Cancellation Classification guides, 2-19 Notice of code word, H-9 Security classification guides, 2-19 Care during working hours Control measures, 6-10 Cover Sheets, 6-10 SF 702, 6-10 Signs on security containers, 6-10 Categories Classification, 2–8 Special Access Programs (SAPs), I-35 **Categories of protection** 

Special Access Programs (SAPs), I-36 **CD-ROM** Destruction, 3-18 **Certified Mail** Transmission of CONFIDENTIAL, 8-4 Challenge Appeal of denial, 2-22 Classification, 2-22 Foreign government information, 2-22 Procedures for formal classification challenges, 2-22 Requirements for Original Classification Authority (OCA), 2-22 To classification - timeline, 2-22 Challenges to classification Procedures, 2-22 Change Classification guide, G-2 Combinations-when to change, 7-8 Markings, 4-35 Security classification guide, G-2 Charts Marking, 4-6, 4-26 Checklist Management Control Evaluation Checklist, F-1 Chief of Engineers (COE) Special Access Programs (SAPs) responsibilities, I-19 Chief of Legislative Liaison (CLL) Special Access Programs (SAPs) responsibilities, I-21 Chief of Public Affairs (PA) Special Access Programs (SAPs) responsibilities, I-12 Chief of Staff, Army (CSA) Special Access Programs (SAPs) responsibilities, I-13 Chief of the Counterintelligence, Human Intelligence Division Responsibilities, 1-5 Chief, Technology Management Office Responsibilities, 1-5 Special Access Programs (SAPs) responsibilities, I-23 Choppers Destruction, 3–15 Civilian personnel NDA, 6-3 Nondisclosure agreement, 6-3 Classification Acquisition systems, 2-14 Categories, 2-8 Challenge denial appeal, 2-22 Challenge timeline, 2-22 Challenges, 2-22 Criteria, 2-8 Decisions to classify, 2-7 Derivative, 2-1 Derivative policy, 2-5 Downgrading during original classification, 3-11 Duration of, 2-11 Extensions – duration of classification, 4–39 Instructions for code words, H-10 Levels of, 1-11, 2-10

Limitations and prohibitions, 2-15 Marking - overall, 4-4 Original, 2-1 Original process, 2-7 Overall marking, 4-4 Policy for derivative, 2-5 Principles, 2-1 Procedures for formal challenges, 2-22 Reason for and classification guides, 2-17 Reason for and security classification guides, 2-17 Reason for combination of original and derivative, 4-9 Reasons - original classification, 4-9 Reclassification, 2-9 Security education for derivative classifiers, 9-11 Security education for original classifiers, 9-10 Sources overview, 4-7 Upgrading, 3-12 When it is done, 1-11 **Classification challenge** Appeal of denial, 2-22 Denial appeal, 2-22 Foreign government information, 2-22 Procedures, 2-22 Timeline, 2-22 **Classification determination** Non-government information, 2-21 Classification Guide(s) Approval, 2-18 Cancellation, 2-19 Changing, G–2 Classification reason, 2-17 Content requirements, 2-17 Distribution, 2–18 General, G-1 Instructions for preparing DD Form 2024, G-3 OCA, 2–17 Original Classification Authority, 2-17 Point of contact, 2-17 Reason for classification, 2-17 Report of review, revision, and/or cancellation, 2-19 Requirements for content, 2-17 Review, 2-19 Revision, G-2 Sample, G-1 Warning notice, 2-17 Classified Discussions, 7-6 Loss of classified material, 10-1 E-mail, E-3 Electronic mail, E-3 Websites - marking, E-7 **Classified** access Debriefing, 6-5 Termination, 6-5 **Classified By line** Combination of original and derivative, 4-8

General, 4-8 Procedures, 4-8 **Classified** information Access by Congress, 6-8 Access by former Presidential appointees, 6-8 Access by GAO, 6-8 Access by General Accounting Office, 6-8 Access by Government Printing Office, 6-8 Access by GPO, 6-8 Access by historical researchers, 6-8 Access by persons outside the Executive Branch, 6-8 Access for judicial proceedings, 6-8 Access in other situations, 6-8 DOD personnel responsibilities, 6-1 Into combat/hostile areas, 1–18 Personnel responsibilities, 6-1 Safeguarding in foreign countries, 7-7 Storage policy, 7-1 Procedures for transmission to foreign government, 8-7 Transmission to foreign government, 8-6 When maintained, 1–11 **Classified material** Destruction, 6-27 Destruction standards, 3-13 Disposition, 6–27 Escort – general, 8–12 Handcarrying – general, 8–12 Level of protection required, 1-11 Receipt, 6-20 Remarking and using old classified material, 4-40 Reproduction policy, 6–25 Review prior to destruction, 6-27 Review prior to disposition, 6-27 CLASSIFIED SCI label - SF 712, 4-34 **Classified shipment** From direct commercial sales to foreign government, 8-7 Outside the U.S. to foreign government, 8-7 Classify OCA communication of decision, 2-12 Original Classification Authority communication of decision, 2-12 Clearance(s) Department of Energy, 6-17 DOE, 6-17 NRC, 6-17 Nuclear Regulatory Commission, 6-17 **Cleared** personnel Security education, 9-4 Clearing Media, 3–18 **CNWDI** Access, 6-7 Briefing, 9-13 Marking portion(s), 4–6 Portion marking, 4-6 **Code Word** Assignment of classified meanings, H-8

Classification instructions, H-10 Definition, H-2 Disposition, H–12 Downgrading instructions, H-10 Notice of assignment, H-9 Notice of cancellation, H-9 Notice of dissemination, H-9 Policy, H-5 Procedures, H-5 Security practices, H-11 Combat/hostile areas Classified information and, 1-18 **Combination(s)** Access, 7-8 Marking record of, 7-8 Record of, 7-8 Security containers and security, 7-8 SF 700, 7-8 Storage equipment, 7-8 When to change, 7-8 **Combination** locks Replacement, 7-4 Combination original and derivative information Reasons for classification, 4-9 Command Responsibilities regarding classified meetings and conferences, 6-18 **Command security inspection** Commander, 1-24 Responsibilities, 1-24 **Command Security Manager** Communication, 6–6 Cooperation, 6-6 Position requirements, 1-7 Requirements for position, 1-7 Responsibilities, 1-7 Commander Command security inspection, 1-24 Communication, 6–6 Cooperation, 6-6 Delegation by, 1-6 Responsibilities, 1–6 Commanding General, Forces Command (CG, FORSCOM) Special Access Programs (SAPs) responsibilities, I-26 Commanding General, U.S. Army Criminal Investigation Command (CG, USACIC) Special Access Programs (SAPs) responsibilities, I-29 Commanding General, U.S. Army Intelligence and Security Command (CG, USAINSCOM) Special Access Programs (SAPs) responsibilities, I-28 Commanding General, U.S. Army Materiel Command (CG, AMC) Special Access Programs (SAPs) responsibilities, I-25 Commanding General, U.S. Army Space and Missile Defense Command (USASMDC) Special Access Programs (SAPs) responsibilities, I-27 Commanding General, U.S. Army Training and Doctrine Command (CG, TRADOC) Special Access Programs (SAPs) responsibilities, I-24 Commercial passenger aircraft Escorting aboard, 8-15 Handcarrying aboard, 8-15
Communication Command Security Manager, 6-6 Commander, 6–6 OCA decision to classify, 2-12 Original Classification Authority decision to classify, 2-12 **Communications directories** Marking, 4-25 Communications Security (COMSEC), 1-17 **Compensatory measures** Waivers, 6-31 Compilation Marking, 4-6 Marking guidance, 4-6 Marking upgrade, 4-6 Of UNCLASSIFIED items, 2-13 Requirements, 2-13 Upgrade marking, 4-6 **Component** parts Marking documents, 4-16 Compromise Deliberate compromise reporting requirements, 10-3 General, 10-1 Comptroller of the Army, HQDA Responsibilities, 1–5 **Computer Security Act of 1987** Access, 5-21 Description, 5–19 Guidance, 5–23 Marking, 5-20 Protection, 5-22 COMSEC General, 1-17 Transmission, 8-1 Transportation, 8-1 **COMSEC** material Incidents involving, 10-1 Warning notice, 4-12 Concepts Destruction, 3-14 Conferences Approval authority, 6–18 Association-related classified meetings, 6-18 Authority for approval, 6–18 General, 6-18 In-house classified meetings, 6-18 Obtaining authorization, 6-18 CONFIDENTIAL Definition of, 2-10 **CONFIDENTIAL** information Accountability and administrative procedures, 6-22 Storage requirements, 7-4 Transmission, 8-4 CONFIDENTIAL label - SF 708, 4-34 Confidential source or relationship Marking, 4-2 Congress

Access to classified information, 6-8 **Consignor/Consignee** Bulky material-responsibilities for shipment, 8-16 **Constant Surveillance Service (CSS)** Transmission of CONFIDENTIAL, 8-4 Construction Standards for open storage areas, 7-13 Standards for secure rooms, 7-13 Standards for vaults, 7-13 Container Inspection, 7-10 Maintenance, 7-10 Storage records, 7-8 Preparation of material, 8–9 **Content requirements** Classification guides, 2-17 Security classification guides, 2-17 Contractor Debriefing cleared contractor, 10-6 Debriefing uncleared contractor, 10-6 **Contractor** personnel NDA, 6-2 Nondisclosure agreement, 6-2 Control Keys, 7-4 Locks, 7-4 Open storage area access, 7-19 **Control marking** Authorized for intelligence information, D-3 Obsolete, 4-13 **Control measures** Care during working hours, 6-10 Cover sheets, 6-10 Emergency planning, 6–12 End-of-day security checks, 6-11 Responsibilities, 6-9 SF 702, 6-10 Signs on security containers, 6-10 Speakerphone guidance, 6-14 Telephone conversations, 6-13 **Controlled Unclassified Information (CUI)** General, 5-1 Cooperation Command Security Manager, 6-6 Commander, 6-6 **Corrective actions** Commander's responsibilities, 1-20 General, 1-20 **Cover sheet** Care during working hours, 6-10 Control measures, 6-10 **Criminal violations** Reporting requirements, 10-4 Criteria Classification, 2-8 Critical Nuclear Weapons Design Information (CNWDI)

Access, 6-7 Briefing, 9-13 Marking portion(s), 4-6 Portion marking, 4-6 **Cryptologic information** Declassification, C-2 General, C-1 Guidelines, C-1 Identification words, C-2 INFOSEC, C-1 Review, C-2 SIGINT, C-1 CSS Transmission of CONFIDENTIAL, 8-4 CUI General, 5–1 **DA** Consultants NDA, 6-3 Nondisclosure agreement, 6-3 **DA** Personnel Definition of, 1-1 Damage Security container repair, 7-9 Damage assessment Reevaluation, 10-5 Steps taken, 10-5 DAMI-CH Responsibilities, 1-5 Data density concept Destruction, 3-14 Data loss Lost cluster, 4-32 Slack space, 4-32 Unallocated space, 4-32 Date Decision to extend declassification date, 2-11 **Date determination** Declassification, 2–11 **Date extension** Declassification, 2-11 Date not specified Declassification, 2-11 DCS Transmission, 8-2 DCSINT Responsibilities, 1-5 DCSPER, 3–2 Responsibilities, 1–5 **DEA** sensitive information Access, 5-13 Description, 5-11 Marking, 5-12 Protection, 5-14 Debriefing Access, 6-5 Classified access, 6-5

Cleared contractor, 10-6 Legal counsel, 10-6 Non-U.S. government personnel, 10-6 U.S. government personnel, 10-6 Unauthorized access, 10-6 Uncleared contractor, 10-6 Decision to classify OCA communication, 2-12 Original Classification Authority communication, 2-12 Decisions To classify, 2-7 Decisions to extend Declassification date, 2-11 Declassification As a result of disclosure, 2-9 Cryptologic information, C-2 Date determination, 2-11 Date extension, 2-11 Date not specified, 2-11 Decisions to extend date, 2-11 Designation of exemption, 2-11 Determination of date, 2-11 Earlier than scheduled, 4-36 Exemption designation, 2-11 Exemption list, 2–11 Extension of date, 2-11 Foreign government information, 3-4 Formerly Restricted Data, 3-3 FRD, 3-3 In accordance with markings, 4–35 Mandatory review, 3-8 Media, 3-18 OADR - instructions, 4-10 Other than Army information, 3-4 RD, 3-3 Remarking earlier than scheduled, 4–36 Restricted Data, 3-3 Review for mandatory declassification, 3-8 Review for systematic declassification, 3-9 SIGINT, C-2 Systematic review, 3-9 Time or event phased, 2-11 **Declassification date** Time or event phased, 2-14 **Declassification exemption** List of qualifiers, 2-11 Oualifier list, 2–11 **Declassification instructions** Marking, 4–10 **Declassification procedures** Marking, 4-35 **Declassification process review** Special Program Manager, 3-2 SPM, 3-2 **Declassify On line** Past ten years, 4-10

**Defense Courier Service (DCS)** Transmission, 8-2 **Defense Security Service (DSS)** Special Access Programs (SAPs) responsibilities, I-34 Defense Security Service Academy, 9–10, 9–12 Definition Code word, H-2 CONFIDENTIAL, 2-10 Nickname, H–3 SECRET, 2-10 TOP SECRET, 2-10 Using Component, H-1 Degaussing Media, 3-18 Delegation By Commanders, 1–6 By MACOM Commanders for waiver, 1-19 OCA, 1-5 SECRET and CONFIDENTIAL OCA, 1-5 TOP SECRET OCA, 1-5 **Delegation of authority** OCA, 2-3 Original Classification Authority, 2-3 **Deliberate compromise** Reporting requirements, 10–3 Department of Defense (DOD) Components Responsibilities, 1-4 **Department of Energy (DOE)** Clearances, 6-17 Preparation of material for mail through DOE channels, 8-11 Department of Energy (DOE) facilities Visits, 6–17 Department of Energy (DOE) information Emergency access request, 6-17 Department of Energy (DOE) personnel Visits, 6-17 Department of the Army (ARSTAF) Special Access Programs (SAPs) responsibilities, I-30 Deputy Chief of Staff for Intelligence Responsibilities, 1-5, F-1 Special Access Programs (SAPs) responsibilities, I-16 Deputy Chief of Staff for Logistics (DCSLOG) Special Access Programs (SAPs) responsibilities, I-18 Deputy Chief of Staff for Operations and Plans (DSCOPS) Special Access Programs (SAPs) responsibilities, I-17 Deputy Chief of Staff for Personnel (DCSPER) General, 3-2 Responsibilities, 1-5 Special Access Programs (SAPs) responsibilities, I-15 Derivative and original combination Reasons for classification, 4-9 **Derivative classification** Appointment, 2-1, 2-5 Definition, 2–1 Guidance, 2-1, 2-5 Policy, 2-5

Requirements, 2-1, 2-5 **Derivative classifiers** Responsibilities, 2–5 Security education, 9-1 **Derivatively classified documents** Declassify On line - marking, 4-10 Marking Declassify On line, 4-10 **Derived From line** Multiple sources, 4-8 Procedures, 4-8 Description Computer Security Act of 1987, 5-19 DEA sensitive information, 5-11 DOD UCNI, 5-15 DOD Unclassified Controlled Nuclear Information, 5-15 Drug Enforcement Administration sensitive information, 5-11 For Official Use Only, 5-2 FOUO, 5-2 Limited Official Use information, 5-7 LOU, 5-7 SBU, 5-7 Sensitive But Unclassified, 5-7 Sensitive information (Computer Security Act of 1987), 5-19 UCNI, 5-15 Unclassified Controlled Nuclear Information, 5-15 **Designated equipment** Reproduction, 6–26 **Designation** – restricted Marking, 4-56 **Designation of exemption** Declassification, 2-11 **Designations** Storage equipment, 7-8 Designee SOIC, 1-5 Destroying Media, 3-18 Destruction Accompanying documentation, 6-29 Advice, 3-17 Appropriate techniques and methods, 3-16 Approved methods, 3–5 Burning, 3-15 CD-ROM, 3-18 Choppers, 3-15 Classified material, 6-27 Clearing media, 3-18 Concepts, 3-14 Data density concept, 3-14 Declassifying media, 3-18 Destroying media, 3-18 Devices, 3-16 Documentation accompanying, 6-29 Equipment, 3-16 FOUO information, 5-5 General, 3-13

Hammer mills, 3-15 Hoggers and hybrids, 3-15 Magnetic Storage Media (MSM), 3-16 MARKS, 6-27 Methods, 3–15, 6–28 Microfiche - duplicate, 3-16 Microfiche - original, 3-16 Microfilm - duplicate, 3-16 Microfilm - original, 3-16 Modern Army Recordkeeping System, 6-27 Non water-soluble or similar material, 3-16 Non-paper-based products, 3-16 Photographic material - silver-based - original, 3-16 Plastic acetate or similar material, 3-16 Plastic coated or similar material, 3-16 Policy, 6-27 Pulping, 3-15 Pulverizing, 3-15 Purging media, 3–18 Records, 6-29 Request for technical advice, 3-17 Review of classified material prior to, 6-27 SCI, 3-18 Secure volume concept, 3-14 Sensitive Compartmented Information, 3–18 Shredding, 3-15 Silver-based photographic material - original, 3-16 Solid objects, 3-16 Standards, 6-28 Standards for classified material, 3-13 Technical advice request, 3-17 Typewriter ribbons and cassettes (mylar, nylon, and cotton-based ribbon), 3-16 Waxed paper or similar material, 3-16 **Destruction standards** Classified material, 3-13 Determination Non-government information classification, 2-20 Permanent historical value – options, 4–11 **Determination of date** Declassification, 2-11 Deviations Open storage areas-minimum standards to construction, 7-20 Devices Destruction, 3-16 Director of Central Intelligence Directive (DCID) 1/7 Security controls, D-1 Director, Information Security Oversight Office (ISSO) Responsibilities, 1-15 **Director**, ISSO Responsibilities, 1-15 Director, Program Analysis and Evaluation Directorate (PAED) Special Access Programs (SAPs) responsibilities, I-22 **Directories** Marking, 4–25 DISC4 Special Access Programs (SAPs) responsibilities, I-8

Disclosure Declassification as a result of, 2-9 **Discovery of incident** Reaction, 10-2 Discussions Classified, 7-6 Disposition Accompanying documentation, 6-27 Classified material, 6-27 Code words, H-12 Documentation accompanying, 6-27 MARKS, 6-27 Methods, 6-27 Modern Army Recordkeeping System, 6-27 Policy, 6-27 Records, 6-27 Review of classified material prior to, 6-27 Standards, 6-27 Dissemination Notice of code word, H-9 Security controls on dissemination of intelligence information, D-1 Distribution Classification Guides, 2-18 Joint Chiefs of Staff papers, 4-48 Joint Operation Planning System (JOPS) documents, 4-50 Joint Strategic Planning System (JSPS) documents, 4-49 Security classification guides, 2-18 **Distribution statements** Examples, 4-12 Technical documents, 4-12, 5-24 Documents - technical, 5-24 **Document(s)** FOUO information in transmittal documents, 5-3 Internal documents exemption, 5-2 Marking those exempted from the Automatic Declassification System, 3-7 Marking those subject to final storage, 3-7 Marking special types, 4-16 NATO – marking, 4–56 Printed on AIS, E-2 Printed on Automated Information Systems, E-2 Produced by AIS equipment - marking, 4-23 Statement for technical document distribution, 4-12 **Document Security** Special Access Programs (SAPs), I-40 Documentation Escort, 8-13 Handcarrying, 8–13 Waivers, 6-30 **Documents** – foreign government information In DA documents - marking, 4-59 Marking, 4-58 **Documents - NATO information** In DA documents - marking, 4-59 **Documents** – restricted Foreign government information in DA documents - marking, 4-59 NATO information in DA documents - marking, 4-59

**Documents** – technical Distribution statements, 5-24 General, 5-24 Statements – distribution, 5–24 Documents marked for training purposes Marking, 4-21 Documents with component parts Marking, 4-16 **DOD** Component(s) Responsibilities, 1-4 **DOD UCNI** Access, 5-17 Description, 5-15 Marking, 5-16 Protection, 5-18 **DOD Unclassified Controlled Nuclear Information (UCNI)** Access, 5-17 Description, 5-15 Marking, 5-16 Protection, 5-18 DOE Clearances, 6-17 Preparation of material for mail through DOE channels, 8-11 **DOE** facilities Visits, 6–17 **DOE** information Emergency access request, 6-17 **DOE** personnel Visits, 6-17 Downgrading At later date, 3-11 Authority and purpose, 3-11 During original classification, 3-11 Earlier than scheduled, 4-36 In accordance with markings, 4-35 Instructions for code words, H-10 Marking, 4-14 Purpose and authority, 3–11 Remarking earlier than scheduled, 4-36 Downgrading procedures Marking, 4-35 **Drawings** Marking, 4–6 Drug Enforcement Administration (DEA) sensitive information Access, 5-13 Description, 5-11 Marking, 5-12 Protection, 5-14 Duration Of classification, 2-11 Waivers, 6-32 **Duration of classification** Extensions, 4-39 E-mail Classified, E-3 Sensitivity, E-4

## Education

Foreign travel briefings, 9-8 Other special security education briefings, 9-14 Oversight of security education program, 9-16 Security education continuing requirements, 9-7 Security education for cleared personnel, 9-4 Security education for derivative classifiers, 9-11 Security education for original classifiers, 9-10 Security education for uncleared personnel, 9-6 Security education methodology, 9-2 Security education policy, 9-1 Security education program - points to include, 9-4 Security education program requirements, 9-1 Security education special requirements, 9-9 EEIP Inspections, 6–36 Electronic mail (E-mail or e-mail) Classified, E-3 Sensitivity, E-4 Electronically transmitted messages Marking, 4–20 Emergency Access request to Department of Energy (DOE) information, 6-17 **Emergency access request** To Department of Energy information, 6-17 To DOE information, 6-17 **Emergency** planning Control measures, 6-12 **Emergency situations** Exceptional situations, 1-18 End-of-day security checks Control measures, 6-11 Entrance Security, 7–8 Entry Exit Inspection Program (EEIP) Inspections, 6–36 **Envelopes** Preparation of material, 8-9 Equipment Combinations, 7-8 Destruction, 3-16 Designated for reproduction, 6-25 Designations, 7-8 IDS, 7-14 Intrusion Detection System, 7-14 Procurement of new storage equipment, 7-5 Security procedures, 6–19 Storage standards, 7-3 Transfer, 7-11 Turn-in, 7-11 Equivalent foreign security classification, 4-55 Escort Documentation, 8-13 General, 8–12 Escorting Aboard commercial passenger aircraft, 8-15

**Espionage** laws Extracts, 10-11 Establishes 380-5, 1-1 Establishment Information Security Oversight Office, 1-15 ISOO, 1-15 **Evaluation** Management Control Evaluation Checklist, F-1 Examples Distribution statements, 4-12 Marking classification, 4–3 Warning Notices, 4-12 **Exception**(s) Marking, 4-2 Marking on AIS, E-2 Marking on Automated Information System, E-2 Policy, 1-19 **Exceptional situations** Emergency situations, 1–18 Marking, 4–6 Military exercises, 1-18 Military operations, 1–18 Exceptions to policy, 1-19 **Executive Orders** 12958, 1-1 12972, 1-1 13142, 3-5 Exemption(s) FOIA categories, 5-2 Freedom of Information Act categories, 5-2 Internal documents, 5-2 Law enforcement records, 5-2 List for automatic declassification, 3-6 Mandatory review, 3-8 Marking, 4-10 Marking for multiple, 4-10 Multiple marking, 4–10 Request for other than listed-automatic declassification, 3-6 Review for mandatory declassification, 3-6 Ten Year Rule, 4-10 **Exemption** – multiple Marking, 4-10 **Exemption designation** Declassification, 2-11 Exemption from automatic declassification List of qualifiers, 2-11 Qualifier list, 2-11 **Exemption list** Automatic Declassification System, 3-6 Declassification, 2-11 **Exemption** request For other than listed-automatic declassification, 3-6 **Exemption** requirements Automatic Declassification System, 3-6 **Exercise Term** 

Definition, H-4 Heads of DOD Components, H-7 Joint Chiefs of Staff, H-7 Policy, H-7 Procedures, H–7 **Express Mail** Transmission of SECRET, 8-3 Extend Decision to extend declassification date, 2-11 Extension of date Declassification, 2-11 **Extensions** Duration of classification, 4-39 **External marks** Storage containers, 7-8 Extract Espionage laws, 10-11 Federal statutes, 10-11 Federal Aviation Administration, 8–15 **Federal statutes** Extracts, 10-11 **Figures** Marking, 4-6 File(s) Marking, 4–22 Financial information, 5–2 First Class Mail Transmission of CONFIDENTIAL, 8-4 Flights Handcarrying on international flights, 8-14 FMS. 8–7 FOIA Exemption from public release categories, 5-2 Folders Marking, 4–22 For Official Use Only (FOUO) Description, 5-2 For Training Purposes Only Marking, 4-21 **Foreign countries** Safeguarding of U.S. classified information, 7-7 **Foreign documents** Automatic Declassification System, 3-6 Foreign government Classified shipment from direct commercial sales to, 8-7 Classified shipment from outside the U.S. to, 8-7 Procedures for transmission of classified information to, 8-7 Transmission of classified information to, 8-7 Foreign government designations Equivalent U.S. classification designations, 4-55 Foreign government documents Marking, 4–57 Foreign government information Automatic Declassification System, 3-6 Challenge to classification, 2-22 Classification challenge, 2-22

Declassification, 3-4 In DA documents - marking, 4-59 Incidents involving, 10-1 Marking portion(s), 4-6 Policy, 4-54 Portion(s) marking, 4-6 Provided in confidence - marking, 4-58 Request for waiver, 1-19 Warning notices, 4–12 Foreign government material information Accountability and administrative procedures, 6-23 Foreign intelligence agencies Incidents involving, 10-1 Foreign Military Sales, 8–7 Foreign travel briefing Content requirements, 9-8 Former Presidential appointees Access to classified information, 6-8 Formerly Restricted Data (FRD) Access, 6-7 Declassification, 3–3 General, 1-16 Request for waiver, 1-19 Marking, 4-2 Marking portion(s), 4–6 Portion marking, 4-6 Warning notices, 4–12 Forwarding for filing NDA, 6-5 Nondisclosure agreement, 6-5 FOUO Description, 5–2 Marking, 4-6 **FOUO** information Access, 5-4 Destruction, 5-5 Guidance, 5–6 Identification, 5-3 Marking, 5-3 Procedures, 5-3 Protection, 5-5 Transmission, 5–5 Transmittal documents, 5-3 FRD Access, 6-7 Declassification, 3-3 General, 1-16 Marking, 4-2 Marking portion(s), 4–6 Portion marking, 4-6 Warning notices, 4-12 Freedom of Information Act (FOIA) Exemption from public release categories, 5-2 Freight Shipment, 8-8 Function

Information Security Oversight Office, 1-15 ISOO, 1-15 GAO Access to classified information, 6-8 General AIS, E-1 Army Declassification Program, 3-1 Automated Information System, E-1 Automatic Declassification System, 3-5 Classification guide, G-1 Cryptologic information, C-1 Compromise, 10-1 Conferences, 6–18 Controlled Unclassified Information, 5-1 CUI, 5-1 Documents - technical, 5-24 Destruction, 3-13 Escort, 8-12 Handcarrying, 8-12 Loss of classified material, 10-1 Maintenance, 7-10 Mandatory review, 3-8 Meetings, 6-18 NDA, 6-2 Nondisclosure agreement, 6-2 Operating instructions, 7-10 Review for systematic declassification, 3-9 Security classification guide, G-1 Security incidents, 10-1 Systematic review, 3-9 Technical documents, 5-24 Termination briefing, 9–15 Unauthorized disclosure, 10-1 Waivers, 6-30 Website, E-5 General Accounting Office (GAO) Access to classified information, 6-7 **General Counsel (GC)** Special Access Programs (SAPs) responsibilities, I-9 **Government Printing Office (GPO)** Access to classified information, 6-7 GPO Access to classified information, 6-7 Granting of OCA, 2–3 Original Classification Authority, 2-3 Graphs Marking, 4-6 **Groups of documents** Marking, 4–22 Guidance Compilation marking, 4-6 Computer Security Act of 1987, 5-23 Derivative classification, 2-1 FOUO information, 5-6 Marking compilations, 4-6

Sensitive information (Computer Security Act of 1987), 5-23 Speakerphone, 6-14 Website, E-5 Guidelines Cryptologic information, C-1 Inquiry, 10–3 Hammer mills Destruction, 3-15 Handcarrying Aboard commercial passenger aircraft, 8-15 Documentation, 8-13 General, 8–12 NATO information, 8-14 Security Requirements for temporary duty (TDY) travel outside the U.S., 8-14 Travelers on international flights, 8-14 Heads of DOD Components Exercise terms, H-7 Nicknames, H-6 Responsibilities, H-6, H-7 Historical researchers Access to classified information, 6-7 Historical value, 2–11 Hoggers and hybrids Destruction, 3-15 IACSE, 7-9 ID badges, 7-18 Identification FOUO information, 5-3 **Identification words** Cryptologic information, C-2 IDS Equipment selection, 7-15 Installation, 7-18 Maintenance, 7-18 Monitoring, 7–18 Selection of equipment, 7-15 Standards, 7-14 System requirements, 7-17 Transmission security, 7-16 Illustrations Marking, 4-6 Implement 380-5, 1-1 In-house classified meetings Conferences, 6-18 Meetings, 6-18 Incapacitation Inquiries, 10-9 Incidents Involving COMSEC material, 10-1 Involving foreign government information, 10-1 Involving foreign intelligence agencies, 10-1 Involving SAPs, 10-1 Involving Special Access Programs, 10-1 Involving the public media, 10-2

Reaction to discovery of incident, 10-2 Reporting, 1-20 Individual Responsibilities, 1–9 Information Access by Congress, 6-8 Access by former Presidential appointees, 6-8 Access to FOUO, 5-4 Access by GAO, 6-7 Access by General Accounting Office, 6-7 Access by Government Printing Office, 6-7 Access by GPO, 6-7 Access by historical researchers, 6-8 Access by persons outside the Executive Branch, 6–7 Access for judicial proceedings, 6-7 Access in other situations, 6–7 Classified storage policy, 7-1 CONFIDENTIAL information accountability and administrative procedures, 6-22 CONFIDENTIAL storage requirements, 7-4 CONFIDENTIAL transmission, 8-4 Control markings authorized for intelligence information, D-3 Cryptologic guidelines, C-1 Declassification of cryptologic information, C-2 Destruction of FOUO, 5-5 Emergency request for access to Department of Energy information, 6-17 Emergency request for access to DOE information, 6-17 Foreign government material information accountability and administrative procedures, 6–23 FOUO guidance, 5-5 FOUO procedures, 5-3 Handcarrying NATO information, 8-14 Incidents involving foreign government information, 10-1 Identification words and cryptologic information, C-2 INFOSEC and cryptologic information, C-1 Intelligence information responsibilities, D-1 Marking FOUO information, 5-3 Marking intelligence information, D-1 NATO CONFIDENTIAL transmission, 8-4 NATO information accountability and administrative procedures, 6–23 Procedures for transmission to foreign government, 8-7 Protection of FOUO, 5-5 Request for access to Department of Energy information, 6-17 Request for access to DOE information, 6-17 Review of cryptologic information, C-2 Safeguarding classified in foreign countries, 7-7 SECRET information accountability and administrative procedures, 6-22 Security controls on dissemination of intelligence information, D-1 Security procedures for AIS, E-1 Security procedures for Automated Information System, E-1 SECRET storage requirements, 7-4 SECRET transmission, 8-3 SIGINT and cryptologic information, C-1 TOP SECRET information accountability and administrative procedures, 6-21 TOP SECRET information inventory, 6-21 TOP SECRET information receipt accountability, 6-21 TOP SECRET storage requirements, 7-4 TOP SECRET transmission, 8-2

TPI inspections, 6-36 Transmission of FOUO, 5-5 Transmission to foreign government, 8–6 Two Person Integrity inspections, 6-36 Warning notices on intelligence information, D-1 Working papers accountability and administrative procedures, 6-24 Information - classified DOD personnel responsibilities, 6-1 Personnel responsibilities, 6-1 Information – foreign government information Provided in confidence - marking, 4-58 Information processing equipment Removal, 6-15 Security procedures, 6-19 **Information Security** Special Access Programs (SAPs), I-40 Information Security Oversight Office (ISSO) Director's responsibilities, 1-15 Responsibilities, 1–15 **INFOSEC** Cryptologic information, C-1 Initial orientation Requirements, 9-3 Inquiry Additional investigation, 10-8 Alternatives for preliminary judgement, 10-3 Applicable regulations, 10–1 Appointment of an investigator, 10-3 Incapacitation, 10-9 Management, 10-7 Negligence, 10–10 Oversight, 10-7 Preliminary report sample, 10-4 Public media leak, 10-3 Purpose, 10-1 Questions to focus on, 10-3 Reporting results, 10-4 Responsibilities, 1–7 Suicides, 10-9 Unauthorized absences, 10-9 **Inspection**(s) Container, 7-10 Entry Exit Inspection Program (EEIP), 6-36 Responsibilities, 1-7 Security, 1-24 Security container, 7-10 Self-inspections, 6-35 TPI, 6-36 Two Person Integrity for TOP SECRET information, 6-36 **Inspector General (IG)** Special Access Programs (SAPs) responsibilities, I-10 Installation IDS, 7-18 Intrusion Detection System, 7-18 Instructions Code word classification instructions, H-10

Code word downgrading instructions, H-10 Declassification - OADR, 4-10 Marking for downgrading, 4–14 Preparing DD Form 2024, G-3 Intelligence Incidents involving foreign intelligence agencies, 10-1 **Intelligence control markings** Marking, 4-6 **Intelligence** information Authorized control markings, D-3 Marking, D-3 Responsibilities, D-2 Security controls on dissemination, D-1 Warning notices, 4-12, D-1 Interagency Advisory Committee on Security Equipment(IACSE), 7-9 **Internal documents** Exemption, 5-2 **International flights** Handcarrying, 8-14 Internet web-based display Security procedures, E-1 Intrusion Detection System (IDS) Equipment selection, 7–15 Installation, 7-18 Maintenance, 7-18 Monitoring, 7-18 Selection of equipment, 7–15 Standards, 7-14 System requirements, 7-17 Transmission security, 7-16 Inventory TOP SECRET information, 6-21 Investigation Additional inquiries, 10-8 AR 15-6 investigation, 10-8 Investigator Appointment for the inquiry, 10-3 ISSO Director's responsibilities, 1-15 Responsibilities, 1-15 Joint Chiefs of Staff Exercise terms, H-7 Nicknames, H-6 Responsibilities, H-6, H-7 Joint Chiefs of Staff paper(s) Access, 4-46 Army responsibilities, 4-44 Distribution, 4–48 References, 4–43 Release to Army schools, 4-51 Release to organizations outside of DA, 4-52 Reproduction, 4-53 Requirements, 4-45 Safeguarding, 4–45 Joint Operation Planning System (JOPS) document(s) Distribution, 4-50

Release, 4-50 Joint Strategic Planning System (JSPS) document(s) Distribution, 4–49 Release, 4-49 Judge Advocate General (TJAG) Special Access Programs (SAPs) responsibilities, I-20 Judicial proceedings Access to classified information, 6-7 Key card system Standards, 7-19 Key cards, 7-19 Keys Accountability, 7-4 Control, 7-4 Law enforcement records Exemption, 5–2 Leak inquiry Public media, 10-3 Legal authority 380-5, 1-12 Legal counsel Debriefing, 10-6 Level Classification, 2-10 Classified material - protection required, 1-11 Of classification, 1-11 Limitations and prohibitions To classification, 2-15 Limited Official Use Information (LOU) Description, 5-7 Lock(s) Accountability, 7-4 Combination lock replacement, 7-4 Control, 7-4 Replacement priorities, 7-21 Lock-bar cabinets, 7-4, 7-5 Lock-outs, 7-9 Loss of classified material General, 10-1 Lost cluster Data loss, 4-32 LOU Description, 5-7 MACOM Responsibilities regarding classified meetings and conferences, 6-18 Magnetic storage media (MSM) Destruction, 3-16 Mail Canadian Registered Mail transmission of SECRET, 8-3 Certified Mail transmission of CONFIDENTIAL, 8-4 Classified electronic mail, E-3 Classified e-mail, E-3 Constant Surveillance Service transmission-CONFIDENTIAL, 8-4 CSS transmission of CONFIDENTIAL, 8-4 Express Mail transmission of SECRET, 8-3

First Class Mail transmission of CONFIDENTIAL, 8-4 Next-day mail transmission of SECRET, 8-3 Overnight mail transmission of SECRET, 8-3 Preparation of material for mail through DOE channels, 8-11 Protective Security Service transmission of SECRET, 8-3 PSS transmission of SECRET, 8-3 Registered Mail transmission of CONFIDENTIAL, 8-4 Registered Mail transmission of SECRET, 8-3 Transmission of CONFIDENTIAL by ship, 8-4 U.S. Government contract vehicle transmission of SECRET, 8-3 U.S. Government vehicle transmission of SECRET, 8-3 Maintain Classified information, 1-11 Maintenance Container, 7-10 General, 7-10 IDS, 7-18 Intrusion Detection System, 7-18 Security container, 7-10 Major Army Commands (MACOMs) Special Access Programs (SAPs) responsibilities, I-31 Management Control evaluation checklist, F-1 Inquiries, 10-7 Security program management personnel, 9–12 Management Control Evaluation Checklist, F-1 Management of classified information Performance rating and, 1-5 Mandatory review Declassification, 3-8 Exemptions, 3-8 General, 3-8 Maps Marking, 4-26 Marking Agency of origin, 4–5 AIS equipment produced printed documents, 4-23 AIS storage media, 4-32 AIS storage media - fixed and internal, 4-33 Blueprints, 4-26 Bulk quantities of material - remarking, 4-38 Changes, 4–35 Charts, 4–6, 4–26 Classification - overall, 4-4 Classified websites, E-8 CNWDI portion(s), 4-6 Combination record, 7-8 Communication directories, 4-25 Compilations, 4-6 Compilations guidance, 4-6 Compilations upgrade, 4-6 Computer Security Act of 1987, 5-20 Confidential source or relationship, 4-2 Critical Nuclear Weapons Design Information portion(s), 4-6 DEA sensitive information, 5-12 Declassification instructions, 4-10

Declassification procedures, 4-35 Declassify On line - derivatively classified documents, 4-10 Declassify On line procedures, 4-10 Derivatively classified documents - Declassify On line, 4-10 Designation - restricted, 4-56 Directories, 4-25 Documents - foreign government, 4-57 Documents - foreign government information-DA documents, 4-59 Documents - NATO, 4-56 Documents - NATO information in DA documents, 4-59 Documents - restricted foreign government information in DA documents, 4-59 Documents - restricted NATO information in DA documents, 4-59 Documents exempt from Automatic Declassification System, 3-7 Documents subject to final storage, 3-7 Documents marked for training purposes, 4-21 Documents with component parts, 4-16 DOD UCNI, 5-16 DOD Unclassified Controlled Nuclear Information, 5-16 Downgrading instructions, 4-14 Downgrading procedures, 4-35 Drawings, 4-6 Drug Enforcement Administration sensitive information, 5-12 Electronically transmitted messages, 4-20 Exceptional situation, 4-6 Exceptions, 4-2 Exceptions for AIS, E-2 Exceptions for Automated Information System, E-2 Exemption to Ten Year Rule, 4-10 Exemption(s), 4-10 Exemption(s) multiple, 4-10 Figures, 4-6 Files, 4-22 Fixed and internal AIS storage media, 4-33 Folders, 4-22 For Training Purposes Only, 4-21 Foreign government documents, 4-57 Foreign government information in DA documents, 4-59 Foreign government information portion(s), 4–6 Foreign government information provided in confidence, 4-58 Formerly Restricted Data, 4-2 Formerly Restricted Data portion(s), 4-6 FOUO, 4-6 FOUO information, 5-3 FRD, 4-2 FRD portion(s), 4-6 Graphs, 4-6 Groups of documents, 4-22 Illustrations, 4-6 Information - foreign government information provided in confidence, 4-58 Intelligence control markings, 4-6 Intelligence information, D-1 Intelligence information authorized control markings, D-3 Maps, 4–26 Material - special types, 4-24 Microfiche, 4-31 Microfilm, 4-31

Motion picture films, 4-29 NATO documents, 4-56 NATO information in DA documents, 4-59 NATO information portion(s), 4-6 Negatives, 4–27 NOFORN, 4-6 Obsolete, 4-13 Office of origin, 4-5 ORCON, 4-6 Overall classification, 4-4 Page, 4-6 Photographs, 4-6, 4-27 Policy, 4-1 Portion, 4–6 Portion(s) foreign government information, 4-6 Portion(s) Formerly Restricted Data, 4-6 Portion(s) FRD, 4-6 Portion(s) NATO information, 4-6 Portion(s) RD, 4-6 Portion(s) Restricted Data, 4-6 Portion(s) that are unclassified, 4-6Printed documents produced by AIS equipment, 4-23 PROPIN, 4-6 Public media, 4-2 Purpose, 4-1 RD, 4-2 RD portion(s), 4-6 Recordings - sound, 4-30 Relationship - confidential, 4-2 Remarking for upgrading, 4–37 Removable AIS storage media, 4-32 Requirements, 4-3 Restricted Data, 4-2 Restricted Data portion(s), 4-6 Restricted designation, 4-56 Restricted foreign government information-DA documents, 4-59 Restricted NATO information in DA documents, 4-59 SAPs, 4-6 SBU, 5-8 Schematics, 4-26 Sensitive But Unclassified, 5-8 Sensitive information (Computer Security Act of 1987), 5-20 SF labels, 4-34 Slides, 4-28 Sound recordings, 4-30 Source - confidential, 4-2 Sources created prior to 1976, 4-11 Special Access Programs, 4-6 Special types of documents, 4-16 Special types of material, 4-24 Standard Form (SF) labels, 4-34 Subject, 4-6 Tables, 4-6 Telephone directories, 4-25 Title, 4-6 Translations, 4-19

Transmittal documents, 4-17 Transparencies, 4-28 UCNI, 5-16 Unclassified Controlled Nuclear Information, 5-16 Unclassified websites, E-6 Unclassified portion(s), 4-6 Unprocessed film, 4-27 Upgrade of compilations, 4-6 US ONLY, 4-6 Videotapes, 4-29 Warning notices, 4-12 MARKS Application, 4-15 Destruction, 6-27 Disposition, 6-27 Principles, 4–15 Purpose, 4-15 Material Address preparation, 8-10 Bulky storage, 7-4 Classified material reproduction policy, 6-25 Consignor/consignee responsibilities for shipment of bulky material, 8-16 Incidents involving COMSEC material, 10-1 Loss of classified material, 10-1 NATO Restricted transmission, 8-5 Preparation for transmission, 8-9 Preparation of containers, 8-9 Preparation of envelopes, 8-9 Material - classified Destruction, 6-28 Disposition, 6-27 Review prior to destruction, 6–28 Review prior to disposition, 6-27 Media Clearing, 3-18 Declassification, 3-18 Degaussing, 3-18 Destroying, 3–18 Incidents involving the public media, 10-2 Public media leak inquiry, 10-3 Purging, 3-18 Meetings Approval authority, 6–18 Association-related classified meetings, 6-18 Authority for approval, 6-18 General, 6-18 In-house classified meetings, 6-18 Obtaining authorization, 6-18 **Meetings and Conferences** Command and MACOM responsibilities, 6-18 Messages Marking electronically transmitted messages, 4-20 Methodology Security education, 9-2 Methods Destruction, 3-15, 6-28

Disposition, 6-27 Microfiche Marking, 4–31 Microfiche – duplicate Destruction, 3–16 Microfiche – original Destruction, 3-16 Microfilm Marking, 4-31 Microfilm – duplicate Destruction, 3-16 Microfilm - original Destruction, 3-16 Military exercises Exceptional situations, 1-18 **Military** operations Exceptional situations, 1-18 Military personnel NDA, 6-3 Nondisclosure agreement, 6-3 **Misconceptions** AIS, 4-32 Automated Information System, 4-32 Modern Army Recordkeeping System (MARKS) Application, 4–15 Destruction, 6-28 Disposition, 6-27 Further guidance, 4-15 Principles, 4–15 Purpose, 4–15 Monitoring IDS, 7-18 Intrusion Detection System, 7-18 Motion picture films Marking, 4-29 **Multiple** sources Derived From line, 4-8 National Industrial Security Program Operating Manual (NISPOM) Special Access Programs (SAPs), I-2 NATO classified information, 4-12 NATO CONFIDENTIAL information Transmission, 8–4 **NATO** documents Marking, 4–56 **NATO** information Accountability and administrative procedures, 6-23 Handcarrying, 8-14 In DA documents - marking, 4-59 Marking portion(s), 4-6 Portion(s) marking, 4-6 Request for waiver, 1-19 Warning notices, 4-12 NATO Restricted material Transmission, 8-5 NDA Civilian personnel, 6-3

Contractor personnel, 6-2 General, 6-2 Military personnel, 6-3 Non-U.S. Government personnel, 6-3 Refusal to execute, 6-4 Refusal to sign, 6-5 Refusal to sign - briefing, 9-5 Retention, 6-2 Signing and filing, 6-3 Who must sign, 6-2Negatives Marking, 4-27 Negligence Inquiries, 10–10 New storage equipment Procurement, 7–5 Next-day mail Transmission of SECRET, 8-3 Nickname Definition, H-3 Heads of DOD Components responsibilities, H-6 Joint Chiefs of Staff responsibilities, H-6 Policy, H-6 Procedures, H-6 Words not to be used, H-6 NISPOM, 8-3 NOCONTRACT, 4–13 **NOFORN** Marking, 4-6 Non-government information Determination of classification, 2-21 Policy, 2-20 Non-paper-based products Destruction, 3-16 Non-U.S. Government personnel NDA, 6-3 Nondisclosure agreement, 6-3 Non-water-soluble or similar material Destruction, 3-16 Nondisclosure agreement (NDA) Civilian personnel, 6-3 Contractor personnel, 6-2 General, 6-2 Military personnel, 6-3 Non-U.S. Government personnel, 6-3 Refusal to execute, 6-4 Refusal to sign, 6–5 Refusal to sign - briefing, 9-5 Retention, 6–2 Signing and filing, 6-3 Who must sign, 6-2 Notice Classification guides, 2-17 Code word assignment, H-9 Code word cancellation, H-9 Code word dissemination, H-9

Security classification guides, 2-17 Warning – Foreign government information, 4–12 Warning – Formerly Restricted Data, 4–12 Warning - FRD, 4-12 Warning – Intelligence information, 4–12 Warning - Marking, 4-12 Warning - Other, 4-12 Warning - RD, 4-12 Warning - Restricted Data, 4-12 Warning – SAPs, 4–12 Warning - Special Access Programs, 4-12 Warning notices on intelligence information, D-1 NRC Clearances, 6-17 Nuclear Regulatory Commission (NRC) Clearances, 6-17 OADR Declassification instructions, 4-10 General, 2-11 Obsolete Control markings, 4-13 Restrictions, 4-13 **Obtaining authorization** Conferences, 6-18 Meetings, 6-18 **OCA** Authority delegation, 2-3Challenge requirements, 2-22 Classification guides, 2-17 Communication of decision to classify, 2-12 Delegation, 1-5 Delegation of authority, 2-3 Granting of, 2-3 Request submission, 2-3 Required training, 2-4 Sanctions, 1-21 Security classification guides, 2-17 Submission of request, 2–3 Training required, 2-4 Office of origin Marking, 4-5 **Old documents** Remarking, 4-40 **Open** storage Access controls, 7-19 **Open storage areas** Construction standards, 7-13 Minimum standards-deviations to construction standards, 7-20 **Operating instructions** General, 7-10 **Options** Permanent historical value determination, 4-11 ORCON Marking, 4-6 Organizations outside DA Release of Joint Chiefs of Staff papers, 4-52

Orientation Initial orientation requirements, 9-3 Original Classification process, 2-7 Original and derivative combination Classified By line, 4-6 Reasons for classification, 4-9 **Original classification** Definition, 2–2 Downgrading during, 3-11 General, 2-1, 2-2 Policy, 2-2 Process, 2-7 Reasons, 4-9 **Original Classification Authority (OCA)** Authority delegation, 2–3 Challenge requirements, 2-22 Classification guides, 2-17 Communication of decision to classify, 2-12 Delegation, 1-5 Delegation of authority, 2-3 Granting of, 2-3 Request submission, 2-3 Required training, 2-4 Sanctions, 1-21 Security classification guides, 2-17 Submission of request, 2-3 Training required, 2-4 **Original classifiers** Security education, 9-10 Originating Agency's Determination Required (OADR), 2-11 Other Warning notices, 4–12 **Other Situations** Access to classified information, 6-7 Other than Army information Declassification, 3-4 Outprocessing Procedures, 6-5 **Overnight mail** Transmission of SECRET, 8-3 Oversight Of security education program, 9-16 Inquiries, 10-7 Overview Sources of classification, 4-7 Page Marking, 4-6 **Papers** Safeguarding Joint Chiefs of Staff papers, 4-42 Performance rating and Management of classified information, 1-5 Permanent historical value, 2-11 Determination options, 4-11 Personal Identification Number (PIN), 7-19 Personal performance appraisal

Responsibilities, 1-8 Personnel Debriefing non-U.S. Government personnel, 10-6 Debriefing U.S. Government personnel, 10-6 Definition of, 1–1 Security education for cleared personnel, 9-4 Security education for uncleared personnel, 9-6 Security program management, 9-12 **Photographs** Marking, 4-6, 4-27 Photographic material - silver-based - original Destruction, 3-16 Physical security Assistance regarding storage, 7-9 Policy, 7-2 Standards, 7–12 Special Access Programs (SAPs), I-39 Physical security standards SCI, 7-19 Sensitive Compartmented Information, 7-19 Plastic acetate or similar material Destruction, 3-16 Plastic coated or similar material Destruction, 3-16 Point of contact Classification guides, 2-17 Security classification guides, 2-17 Policy Classified information storage, 7-1 Code words, H-5 Derivative classification, 2-5 Destruction, 6–28 Disposition, 6-27 Exceptions, 1-19 Exercise terms, H-7 Foreign government information, 4-54 Marking, 4-1 Nicknames, H-6 Non-government information, 2-20 Original classification, 2-2 Physical security, 7-1 Public media, 4-2 Reproduction of classified material, 6-25 SAPs, I-1 Security classification guides, 2-16 Security education, 9-1 Special Access Programs, I-1 Termination briefing, 9-15 Portion Marking, 4-6 Possibility Protection, 2-9 **Postal Service** Canadian Registered Mail transmission of SECRET, 8-3 Certified Mail transmission of CONFIDENTIAL, 8-4 Constant Surveillance Service transmission-CONFIDENTIAL, 8-4

CSS transmission of CONFIDENTIAL, 8-4 Express Mail transmission of SECRET, 8-3 First Class Mail transmission of CONFIDENTIAL, 8-4 Next-day Mail transmission of SECRET, 8-3 Overnight Mail transmission of SECRET, 8-3 Registered Mail transmission of CONFIDENTIAL, 8-4 Registered Mail transmission of SECRET, 8-3 Transmission of CONFIDENTIAL by ship, 8-4 **Practices** Code word security, H-11 Preliminary inquiry report Sample, 10-4 **Preliminary judgement** Alternatives for inquiry, 10-3 Preparation Material for transmission, 8–9 Preparation of material Addressing, 8-10 Containers, 8-9 Envelopes, 8-9 Mail channels with the Department of Energy (DOE), 8-11 Principles 380-5, 1-11 Classification, 2-1 MARKS, 4-15 Modern Army Recordkeeping System, 4-15 **Printed documents** Produced by AIS equipment - marking, 4-23 Prior waivers, 6-34 **Priorities** Lock replacement, 7-21 Procedures AIS security, E-1 Automated Information System security, E-1 Code words, H-5 Challenges to classification, 2-22 Classification challenges, 2-22 Classification sources, 4-8 Classified By line, 4-7 CONFIDENTIAL information accountability/administration, 6-22 Declassification marking, 4-35 Declassify On line, 4-8 Declassify On line - past ten years, 4-10 Derived From line, 4-7 Downgrading marking, 4-35 Exercise terms, H-7 Foreign government material information accountability and administration, 6-23 Formal challenges to classification, 2-22 Formal classification challenges, 2–22 FOUO information, 5-3 Information Security Oversight Office, 1-15 Internet web-based display security, E-1 ISOO, 1-15 NATO information accountability and administration, 6-23 Nicknames, H-6 Outprocessing, 6-5

SECRET information accountability and administration, 6-22 TOP SECRET information accountability and administration, 6-21 Transferring, 6-5 Transmission-classified information to foreign government, 8-6 Working papers accountability and administration, 6-24 Process Original classification, 2-7 Procurement New storage equipment, 7-5 Program Oversight of security education program, 9-16 Security Education Program - points to include, 9-4 Security Education Program requirements, 9-1 Security program management personnel, 9-12 **Program Executive Officers (PEOs)** Special Access Programs (SAPs) responsibilities, I-31 Program/Project/Product Managers (PMs) Special Access Programs (SAPs) responsibilities, I-32 Programs ineligible for SAPs security Special Access Programs (SAPs), I-38 PROPIN Marking, 4-6 Protection Computer Security Act of 1987, 5-22 DEA sensitive information, 5-14 DOD UCNI, 5-18 DOD Unclassified Controlled Nuclear Information, 5-18 Drug Enforcement Administration sensitive information, 5-14 FOUO information, 5-15 Possibility of, 2-9 SBU, 5-10 Sensitive But Unclassified, 5-10 Sensitive information (Computer Security Act of 1987), 5-22 UCNI, 5-18 Unclassified Controlled Nuclear Information, 5-18 **Protective Security Service (PSS)** Transmission of SECRET, 8-3 PSS Transmission of SECRET, 8-3 **Public media** Marking, 4-2 Incidents involving, 10-2 Leak inquiry, 10-3 Policy, 4-2 **Public releases** Responsibilities, 1-8 Pulp Destruction, 3–15 Pulverizing Destruction, 3-15 Purging Media, 3-18 Purpose 380-5, 1-1 Inquiry, 10-1 Marking, 4-1

MARKS, 4–15 Modern Army Recordkeeping System, 4–15 **Purpose and authority** Downgrading, 3–11

## Questions

To focus on during inquiry, 10-3

## RD

Access, 6-7 Declassification, 3-3 General, 1-16 Marking, 4–2 Marking portion(s), 4-6 Portion marking, 4-6 Request for waiver, 1-19 Warning notices, 4-12 Reaction Discovery of incident, 10-2 Reason Combination of original and derivative classification, 4-9 For classifying classification guides, 2-17 For classifying security classification guides, 2-17 Original classification, 4–9 Receipt Classified material, 6-20 TOP SECRET information accountability, 6-21 Reclassification, 2-9 **Recordings** – sound Marking, 4-30 Recordkeeping Requirements, 1-13, F-5 Records Combination and security, 7-8 Combination and SF 700, 7-8 Destruction, 6-29 Disposition, 6-27 Law enforcement records exemption, 5-2 Marking record of combination, 7-8 Storage containers, 7-8 Reevaluation Damage assessment, 10-5 Steps taken for damage assessment, 10-5 References 380-5, 1-2 Joint Chiefs of Staff papers, 4-43 List of, 1-2 Refusal to execute NDA, 6-4 Nondisclosure agreement (NDA), 6-4 Refusal to sign NDA, 6-5, 9-5 Nondisclosure agreement, 6-5, 9-5 SF 312, 9-5 **Registered Mail** Transmission of CONFIDENTIAL, 8-4 Transmission of SECRET, 8-3

Registered Mail - Canadian Transmission of SECRET, 8-3 Regrading, 3–10 Regulation Applicable to inquiries, 10-1 **Relationship** – confidential Marking, 4–2 Release Joint Chiefs of Staff papers to Army schools, 4-51 Joint Chiefs of Staff papers-organizations outside of DA, 4-52 Joint Operation Planning System (JOPS) documents, 4-50 Joint Strategic Planning System (JSPS) documents, 4-49 Release Authority, 5-2, 6-8 Remarking Bulk quantities of material, 4-38 Declassification earlier than scheduled, 4–36 Downgrading earlier than scheduled, 4-36 Old documents, 4-40 Upgrading, 4-37 Remarking and using old classified material, 4-40 **Removable AIS storage media** Marking, 4–32 Removal Information processing equipment, 6-15 Storage containers, 6–15 Repair Security container damage, 7-9 Replacement Combination locks, 7-4 Priorities for locks, 7-21 Report of review, revision, and/or cancellation Classification guides, 2-19 Security classification guides, 2-19 Reporting Criminal violations reporting requirements, 10-4 Deliberate compromise reporting requirements, 10-4 Incidents, 1-22 Inquiry results, 10-4 Requirements, 1-23 **Reports** General, 1–23 Sample preliminary inquiry report, 10-4 Reproduction Approval authority, 6-26 Designated equipment, 6-25 Joint Chiefs of Staff papers, 4-53 Policy for classified material, 6-25 **Reproduction Notices**, 4–12 Request Access to Department of Energy information, 6-17 Access to DOE information, 6-17 Emergency access to Department of Energy information, 6-17 Emergency access to DOE information, 6-17 For waiver, 1-19 Request for technical advice Destruction, 3-17

**Request submission** OCA, 2-3 Original Classification Authority, 2-3 **Required training** OCA, 2-4 Original Classification Authority, 2-4 Requirements Automatic declassification system, 3-5 Challenges and Original Classification Authority (OCA), 2-22 Classification guides for content, 2-17 Compilation, 2-13 CONFIDENTIAL information storage, 7-4 Criminal violation reporting, 10-3 Deliberate compromise reporting, 10-3 Derivative classification, 2-1, 2-5 Exemption from automatic declassification, 3-6 Foreign Travel briefing, 9-8 Handcarrying on temporary duty (TDY) travel outside U.S., 8-14 IDS, 7-17 Initial orientation, 9-3 Intrusion Detection System, 7-17 Joint Chiefs of Staff papers, 4-45 Marking, 4–3 Recordkeeping, 1-11, F-5 Reporting, 1-23 Search, 10-3 SECRET information storage, 7-4 Security classification guides for content, 2-17 Security education - continuing requirements, 9-7 Security Education Program, 9-1 Security plan, 6-18 SF 311, 1-7 Special for security education, 9-9 TOP SECRET information storage, 7-4 Transportation plan, 8-7 Visit request, 6–16 **Residential areas** Storage, 7-6 Responsibilities Army for Joint Chiefs of Staff papers, 4-44 Chief, Counterintelligence, Human Intelligence Division, 1-5 Chief, Technology Management Office, HQDA, 1-5 Command regarding classified meetings and conferences, 6-18 Command security inspection, 1-24 Command Security Manager, 1-7 Commander, 1-6 Commanders and corrective actions, 1–19 Comptroller of the Army, HQDA, 1-5 Consignor/consignee for shipment of bulky material, 8-16 Control measures, 6-9 DAMI-CH, 1-5 DCSINT, 1-5 DCSPER, 1-5 Department of Defense components, 1-4 Deputy Chief of Staff for Intelligence, 1-5 Deputy Chief of Staff for Personnel, 1-5

Derivative classifiers, 2-6 Director, Information Security Oversight Office, 1-15 Director, ISSO, 1–15 DOD components, 1-4 DOD personnel, 6-1 Heads of DOD Components, H-6, H-7 Individual, 1-1, 1-9 Information Security Oversight Office, 1-15 Inquiry, 1–7 Inspections, 1-7 Intelligence information, D-1 ISSO, 1–15 Joint Chiefs of Staff, H-6, H-7 MACOM Special Program Manager, 3-2 MACOM SPM, 3-2 MACOM regarding classified meetings and conferences, 6-18 Personal performance appraisal, 1-8 Personnel, 6-1 Public releases, 1-7 SAPs, I-3 Security Manager, 1-7 Special Access Programs, I-3 Special Program Manager, 3-2 SPM, 3-2 Supervisor, 1–8 TMO, 1-5 Visits, 6-16 Restricted Foreign government information in DA documents - marking, 4-59 NATO information in DA documents - marking, 4-59 **Restricted Data (RD)** Access, 6-7 Declassification, 3-3 General, 1-16 Marking, 4-2 Marking portion(s), 4–6 Portion marking, 4–6 Request for waiver, 1-19 Warning notices, 4-12 **Restricted designation** Marking, 4-56 Restrictions Obsolete, 4-13 SAPs, I-2 Special Access Programs, I-2 **Results** Inquiry reporting, 10–4 Retention NDA, 6-2 Nondisclosure agreement, 6-2 Review Automatic declassification, 3-5 Classification guides, 2-19 Classified material prior to destruction, 6-28 Classified material prior to disposition, 6-27 Cryptologic information, C-2

Exemptions from mandatory declassification, 3-8 For systematic declassification, 3-9 General for mandatory declassification, 3-8 Mandatory declassification - general, 3-8 Mandatory for declassification, 3-8 Security classification guides, 2-19 SIGINT, C-2 Systematic declassification - general, 3-9 **Review for declassification process** Special Program Manager, 3-2 SPM, 3-2 Revision Classification guide, G-2 Security classification guide, G-2 **Risk management factors** Before submitting a request for waiver, 1-19 Safeguarding Classified information in foreign countries, 7-7 Joint Chiefs of Staff papers, 4-42 Sample Classification guide, G-1 Distribution statements, 4–12 Marking classification, 4-3 Preliminary inquiry report, 10-4 Security classification guide, G-1 Warning Notices, 4–12 Sanctions Application, 1-21 General, 1-21 OCA, 1-21 Original Classification Authority, 1-21 Types, 1–21 SAPs Categories, I-35 Categories of protection, I-36 Document security, I-40 General, 1-1, 1-17 Incidents involving, 10-1 Information Security, I-40 Marking, 4-6 National Industrial Security Program Operating Manual (NISPOM), I-2 Physical security, I-39 Policy, I-1 Programs ineligible for SAP security, I-38 Responsibilities, I-3 Restrictions, I-2 Security of DA SAPs, I-39 Types, I-37 Warning notice, 4–12 SBU Access, 5-9 Definition, 5-7 Description, 5–7 Marking, 5–8 Protection, 5-10 **Schematics** 

Marking, 4-26 SCI, 1-17 Destruction, 3–18 Physical security standards, 7-19 Scientific research, 2–15 SEALS, 7-9 Search Requirements, 10–3 SECRET Definition of, 2-10 SECRET and CONFIDENTIAL OCA Delegation, 1-5 **SECRET** information Accountability and administrative procedures, 6-22 Storage requirements, 7-4 Transmission, 8–3 SECRET label - SF 707, 4-34 Secretary of the Army Special Access Programs (SAPs) responsibilities, I-3 Secure room Construction standards, 7-13 Secure volume concept Destruction, 3-14 Security Code word security, H-11 Combination and SF 700, 7-8 Combination records, 7-8 Combinations to security containers, 7-8 Controls on dissemination of intelligence information, D-1 Entrances, 7–8 Heads of DOD components, H-6, H-7 Inspections, 1-24 Joint Chiefs of Staff, H-6, H-7 Physical security standards, 7-12 Physical security standards for SCI, 7-19 Physical security standards for Sensitive Compartmented Information, 7-19 Procedures for AIS, E-1 Procedures for Automated Information System, E-1 Procedures for Internet web-based display, E-1 Repair of damaged container, 7-9 Requirements for handcarrying while on temporary duty (TDY) travel outside the United States, 8–14 Specialized equipment storage, 7-4 Transmission for IDS, 7-16 Transmission for Intrusion Detection System, 7-16 Security classification guide(s) Approval, 2-18 Cancellation, 2-19 Changing, G-2 Classification reason, 2-17 Content requirements, 2-17 Distribution, 2-18 General, G-1 Instructions for preparing DD Form 2024, G-3 Notice, 2-17 OCA, 2-17 Original Classification Authority, 2-17
Point of contact, 2-17 Policy, 2-16 Reason for classification, 2-17 Report of review, revision, and/or cancellation, 2-19 Requirements for content, 2–17 Review, 2-19 Revision, G-2 Sample, G-1 Warning notice, 2-17 Security container Combinations and security, 7-8 Inspection, 7–10 Maintenance, 7-10 Repair of damage, 7–9 Signs, 6-10 Security controls Director of Central Intelligence Directive 1/7, D-1 Security education Cleared personnel, 9-4 Continuing requirements, 9-7 Derivative classifiers, 9-11 Foreign Travel briefing, 9-8 General, 9-1 Methodology, 9-2 Original classifiers, 9-10 Other special briefings, 9-14 Program oversight, 9-16 Special requirements, 9-9 Uncleared personnel, 9-6 Security Education Program Points to include, 9-4 Requirements, 9-1 Security equipment Transfer, 7-11 Turn-in, 7-11 Security Equipment and Locking Systems (SEALS), 7-9 Security incidents General, 10-1 Security Manager Position requirements, 1-6 Requirements for position, 1–6 Responsibilities, 1-7 Security of DA SAPs Special Access Programs (SAPs), I-39 Security plan Requirements, 6-18 Security procedures Equipment, 6-19 Information processing equipment, 6-19 Security program management Personnel, 9-12 Selection of equipment IDS, 7-15 Intrusion Detection System, 7-15 Self-inspection(s) Inspections, 6-35

Sensitive But Unclassified (SBU) Access, 5-9 Definition, 5–7 Description, 5-7 Marking, 5–8 Protection, 5-10 Sensitive Compartmented Information (SCI) Destruction, 3-18 General, 1-17 Physical security standards, 7-19 Sensitive information (Computer Security Act of 1987) Access, 5-21 Description, 5-19 Guidance, 5-23 Marking, 5-20 Protection, 5–22 Sensitive Records and Information Agency (SRIA) Special Access Programs (SAPs) responsibilities, I-33 Sensitivity E-mail, E-4 Electronic mail, E-4 SF 311 Annual requirement, 1-7 Requirement, 1-7 SF 312 Refusal to sign - briefing, 9-5 SF 700 – Combination record, 7–8 SF 702 Care during work hours, 6-10 Control measures, 6-10 SF 706 - TOP SECRET label, 4-34 SF 707 - SECRET label, 4-34 SF 708 - CONFIDENTIAL label, 4-34 SF 710 - UNCLASSIFIED label, 4-34 SF 712 - CLASSIFIED SCI label, 4-34 SF labels Marking, 4-34 Shredding Destruction, 3-15 Ship Transmission of CONFIDENTIAL, 8-4 Shipment Freight, 8-8 Shipment of freight, 8-8 SIGINT Cryptologic information, C-1 Declassification, C-2 Review, C-2 Signing and filing NDA, 6-2 Nondisclosure agreement, 6-2 Silver-based photographic material - original Destruction, 3-16 Slack space Data loss, 4-32 Slides

Marking, 4–28 SNM, 5–15 SOIC Designee, 1–5 Solid objects Destruction, 3-16 Sound recordings Marking, 4-30 Source – confidential Marking, 4–2 Sources created prior to 1976 Marking, 4-11 Sources of classification Overview, 4-7 Procedures, 4-8 Speakerphone Guidance, 6-14 Speakerphone guidance Control measures, 6-14 Special Access Programs (SAPs) Categories, I-35 Categories of protection, I-36 Document security, I-40 General, 1-1, 1-17 Incidents involving, 10-1 Information Security, I-40 Marking, 4-6 National Industrial Security Program Operating Manual (NISPOM), I-2 Physical security, I-39 Policy, I-1 Programs ineligible for SAP security, I-38 Responsibilities, I-3 Restrictions, I-2 Security of DA SAPs, I-39 Types, I-37 Warning notice, 4–12 Special Access Programs (SAPs) - responsibilities Assistant Secretary of the Army (Acquisition, Logistics, and Technology (ASA(ALT)), I-5 Assistant Secretary of the Army (Financial Management and Comptroller) (ASA(FM&C)), I-7 Assistant Secretary of the Army (Manpower and Reserve Affairs) (ASA(M&RA)), I-6 Auditor General (AG), I-11 Chief of Engineers (COE), I-19 Chief of Legislative Liaison (CLL), I-21 Chief of Public Affairs (PA), I-12 Chief of Staff, Army (CSA), I-13 Chief, Technology Management Office (TMO), I-23 Commanding General, Forces Command (CG, FORSCOM), I-26 Commanding General, U.S. Army Criminal Investigation Command (CG, USACIC), I-29 Commanding General, U.S. Army Intelligence and Security Command (CG, USAINSCOM), I-28 Commanding General, U.S. Army Materiel Command (CG, AMC), I-25 Commanding General, U.S. Army Space and Missile Defense Command (USASMDC), I-27 Commanding General, U.S. Army Training and Doctrine Command (CG, TRADOC), I-24 Defense Security Service (DSS), I-34 Department of the Army (ARSTAF), I-30 Deputy Chief of Staff for Intelligence (DCSINT), I-16 Deputy Chief of Staff for Logistics (DCSLOG), I-18

Deputy Chief of Staff for Operations and Plans (DSCOPS), I-17 Deputy Chief of Staff for Personnel (DCSPER), I-15 Director, Program Analysis and Evaluation Directorate (PAED), I-22 DISC4, I-8 General Counsel (GC), I-9 Inspector General (IG), I-10 Judge Advocate General (TJAG), I-20 Major Army Commands (MACOMs), I-31 Program Executive Officers (PEOs), I-31 Program/Project/Product Managers (PMs), I-32 Secretary of the Army, I-3 Sensitive Records and Information Agency (SRIA), I-33 Under Secretary of the Army, I-4 Vice Chief of Staff, Army (VCSA), I-14 Special briefings Foreign Travel Briefing, 9-8 Other special security education briefings, 9-14 Special Nuclear Material (SNM), 5-15 Special Program Manager (SPM) Declassification process review, 3-2 General, 3–2 MACOM responsibilities, 3-2 Responsibilities, 3-2 Review for declassification process, 3-2 **Special requirements** Foreign Travel briefing, 9-8 Security education, 9–9 Special types of documents Marking, 4-16 Special types of material Marking, 4–24 Specialized security equipment Storage, 7-4 SPM Declassification process review, 3-2 General, 3–2 MACOM responsibilities, 3-2 Responsibilities, 3–2 Review for declassification process, 3-2 Standard Form (SF) labels Marking, 4-34 Standards Classified material destruction, 3-13 Destruction, 6-28 Destruction of classified material destruction, 3-13 Disposition, 6-27 Construction of open storage areas, 7–13 Construction of secure rooms, 7-13 Construction of vaults, 7-13 For storage equipment, 7-3 IDS, 7-14 Intrusion Detection System, 7-14 Key card system, 7-19 Open storage area deviations to construction, 7-20 Physical security, 7-12 Physical security for SCI, 7-19

Physical security-Sensitive Compartmented Information, 7-19 Statement Technical document distribution, 4-12 Statement - distribution Documents – technical, 5–24 Technical documents, 5-24 Storage Bulky material, 7-4 CONFIDENTIAL information requirements, 7-4 Equipment combinations, 7-8 Equipment designations, 7-8 Open storage access controls, 7-19 Policy for classified information, 7-1 Residential areas, 7-6 SECRET information requirements, 7-4 Specialized security equipment, 7-4 Standards for storage equipment, 7-3 Technical assistance regarding physical security, 7-9 TOP SECRET information requirements, 7-4 Storage containers External marks, 7-8 Records, 7-8 Removal, 6-15 Subject Marking, 4–6 Submission of request OCA, 2–3 Original Classification Authority, 2-3 Suicides Inquiries, 10-9 Supervisor Responsibilities, 1-8 System Requirements for IDS, 7-17 Requirements for Intrusion Detection System, 7-17 Systematic review Declassification, 3-9 General, 3-9 Table(s) Marking, 4-6 Technical advice request Destruction. 3–17 **Technical documents** Distribution statements, 4-12, 5-24 General, 5-24 Statements - distribution, 5-24 **Telephone** conversations Control measures, 6-13 **Telephone directories** Marking, 4-25 Temporary (TDY) duty Requirements for handcarrying outside the United States, 8-14 Ten-year rule, 2-11 Exemption, 4-10 Termination Access, 6-5

Classified access, 6-5 **Termination briefing** General, 9-15 Policy, 9-15 Terms 380-5, 1-3 Time or event phased Declassification, 2-11 Declassification date, 2-17 Timeline Classification challenge, 2-22 Title Marking, 4-6 тмо Responsibilities, 1-5 TOP SECRET Definition of, 2-10 TOP SECRET Control Officers (TSCO), 6-21 Alternate, 6-21 **TOP SECRET** information Accountability and administrative procedures, 6-21 Inventory, 6-21 Receipt accountability, 6-21 Storage requirements, 7-4 Transmission, 8–2 Two Person Integrity (TPI) inspections, 6-36 TOP SECRET label - SF 706, 4-34 TOP SECRET OCA Delegation, 1-5 TPI Inspections, 6-36 Trade secrets, 5–2 Training required Command Security Manager, 2-12 Derivative classifier, 2-11 OCA, 2-4 Original Classification Authority, 2-4 Original classifier, 9-10 Security Managers, 2-12 Security Program Management Personnel, 2-12 Transfer Equipment, 7–11 Security equipment, 7-11 Transferring Procedures, 6-5 Translations Marking, 4–19 Transmission By ship – CONFIDENTIAL, 8–4 Canadian Registered Mail - SECRET, 8-3 Certified Mail - CONFIDENTIAL, 8-4 Classified information to foreign government, 8-6 COMSEC, 8-1 CONFIDENTIAL information, 8-4 Constant Surveillance Service - CONFIDENTIAL, 8-4 CSS – CONFIDENTIAL, 8–4

DCS, 8-2 Defense Courier Service, 8-2 First Class Mail - CONFIDENTIAL, 8-4 FOUO information, 5-5 NATO CONFIDENTIAL information, 8-4 NATO Restricted Material, 8-5 Next-day mail - SECRET, 8-3 Overnight mail - SECRET, 8-3 Preparation of Material, 8-9 Procedures-classified information to foreign government, 8-6 Protective Security Service - SECRET, 8-3 PSS - SECRET, 8-3 Registered Mail - CONFIDENTIAL, 8-4 Registered Mail - SECRET, 8-3 SECRET - Express Mail, 8-3 SECRET information, 8-3 Security for IDS, 7-16 Security for Intrusion Detection System, 7-16 TOP SECRET information, 8-2 U.S. Government contract vehicle - SECRET, 8-3 U.S. Government vehicle - SECRET, 8-3 Transmittal documents FOUO information, 5-3 Marking, 4-17 Transparencies Marking, 4–28 Transportation COMSEC, 8-1 **Transportation plan** Requirements, 8-7 Travel Foreign Travel Briefing, 9-8 Turn-in Equipment, 7-11 Security equipment, 7-11 **Two Person Integrity (TPI)** Inspections, 6-36 Types Sanctions, 1-21 Special Access Programs (SAPs), I-37 Typewriter ribbons and cassettes (mylar, nylon, and cotton-based ribbon) Destruction, 3–16 U.S. classification designations Equivalent foreign government designations, 4-55 U.S. Government contract vehicle Transmission of SECRET, 8-3 U.S. Government vehicle Transmission of SECRET, 8-3 UCNI Access, 5-17 Description, 5-15 Marking, 5-16 Protection. 5-18 **Unallocated** space Data loss, 4-32 Unauthorized absences

Inquiries, 10-9 Unauthorized access Debriefing, 10-6 Unauthorized disclosure General, 10-1 UNCLASSIFIED Marking portion(s), 4-6 Portion(s) marking, 4-6 Websites - marking, E-6 Unclassified Controlled Nuclear Information (UCNI) Access, 5-17 Description, 5-15 Marking, 5-16 Protection, 5-18 Unclassified items Compilation of, 2–13 UNCLASSIFIED label - SF 710, 4-34 Uncleared personnel Security education, 9-6 Under Secretary of the Army Special Access Programs (SAPs) responsibilities, I-4 Unique situation Waivers, 6-31 **Unprocessed Film** Marking, 4-27 Upgrading Classification, 3–12 General, 3-12 Remarking, 4-37 Upgrading remarking Marking, 4-37 US ONLY Marking, 4-6 Using Component Definition, H-1 Vault Construction standards, 7-13 Vice Chief of Staff, Army (VCSA) Special Access Programs (SAPs) responsibilities, I-14 Videotapes Marking, 4-29 Violation(s) Criminal violations reporting requirements, 10-3 Visit request Requirements, 6-16 Visits By Department of Energy personnel, 6-17 By DOE personnel, 6-17 Responsibilities, 6-16 To Department of Energy facilities, 6-17 To DOE facilities, 6-17 Waiver(s) Approving authority, 1-19 Authority to approve, 1-19 Compensatory measures, 6-31

Delegation by MACOM Commanders, 1-19 Documentation, 6-31 Duration, 6-32 Foreign government information, 1-19 Formerly Restricted Data (FRD), 1-19 General, 1-19, 6-30 Granted prior to this regulation, 1-19 NATO information, 1-19 Prior waivers, 6-34 Request for, 1-19 Restricted Data (RD), 1-19 Risk management factors, 1-19 Unique situation, 6-31 Warning notice AIS, 4-12 Automated Information System, 4–12 Classification guides, 2-17 COMSEC material, 4-12 Examples, 4–12 Foreign government information, 4-12 Formerly Restricted Data, 4-12 FRD, 4-12 Intelligence information, 4-12, D-1 Marking, 4-12 NATO information, 4-12 Other, 4-12 RD, 4–12 Restricted Data, 4-12 Samples, 4-12 SAPs, 4–12 Security classification guides, 2-17 Special Access Programs, 4-12 Waxed paper or similar material Destruction, 3-16 Web-based display Security procedures on the Internet, E-1 Website Classified – marking, E-8 General, E-5 Guidance, E-5 Unclassified - marking, E-6 Who must sign NDA, 6-2 Nondisclosure agreement, 6-2 WNINTEL, 4–13 Working papers Accountability and administrative procedures, 6-24

Army Regulation 105–19 AFR 82-2 NAVMATINST 10550.14 MCO 10550.8

**Communications-Electronics** 

# Joint Electronics Type Designation System

Headquarters Departments of the Army, the Air Force, the Navy, and United States Marine Corps Washington, DC 11 February 1974

UNCLASSIFIED

# SUMMARY of CHANGE

AR 105-19/AFR 82-2/NAVMATINST 10550.14/MCO 10550.8 Joint Electronics Type Designation System

Headquarters Departments of the Army, the Air Force, the Navy, and United States Marine Corps Washington, DC 11 February 1974

#### Army Regulation 105–19 AFR 82–2 NAVMATINST 10550.14 MCO 10550.8

Effective 11 February 1974

#### **Communications-Electronics**

#### Joint Electronics Type Designation System

By Order of the Secretaries of the Army, the Navy, and the Air Force:

Official:

VERNE L. BOWERS Major General, United States Army The Adjutant General

JACK R. BENSON, Colonel, USAF Director of Administration

#### History. Not available.

**Summary.** This regulation prescribes policies and procedures for the administration of the Joint Electronics Type Designation System(JETDS), as it is established in MIL-STD-196. It designates the Joint Equipment and Standardization Panel of the US Military Communications-Electronics Board (MCEB) to provide guidance on policy matters concerning the JETDS. It establishes and assigns responsibilities of the Department of Defense Control Point and of Military Departments requesting designations.

**Applicability.** This regulation applies to Military Departments in the identification for design control of electronic (excluding communications security) materiel and certain associated equipment.

#### Proponent and exception authority.

The Army agency of primary interest in this

Official:

CREIGHTON W. ABRAMS General, United States Army Chief of Staff

K. R. WHEELER Vice Admiral, Supply Corps, USN Vice Chief of Naval Material

joint publication is Office, Assistant Chief of Staff for Communications-Electronics.

Army management control process. Not applicable.

Supplementation. Not applicable.

**Suggested Improvements.** Army users are invited to send comments and suggested improvements on DA Form 2028 (Recommended Changes to Publications) direct to HQDA (DACE-CSC-S) WASH DC 20310.

Distribution. Army:

To be distributed in accordance with DA Form 12-9A requirements for AR, Communications-Electronics: Active Army: C (Qty Rqr Block No. 131) ARNG: None USAR: D (Qty Rqr Block No. 132) *Air Force:* B. *Navy:* A4A (50 cys) FKA1A (AIR 5202) 25 cys; FKA1B (ELEX

HARRY C. OLSON Major General, U.S. Marine Corps Deputy Chief of Staff for Installations and Logistics

GEORGE S. BROWN. General, USAF Chief of Staff

04652) 25 cys: FKA1D (ORD 05211A) 10 cys; FKL8 (SEC 6124) 10 cvs: FKR3A (NAVAIRENGCEN Code X) 10 cys: Stocked: Commanding Officer, Navy Publications & Forms Center 5801, Tabor Avenue, Philadelphia, Pa. 19120 Marine Corps: MARCORPS CODE: 6025001 (10) 6025002, 003 (5) 6901 (4) 6600 (3) 1025001, 002, 008/2005/2030/3700001, 002, 003, 004 009/6902/6905/ 7221/7230001, 004, 005, 011/7801006 (2) Copy to: 8145001

**Contents** (Listed by paragraph and page number) Terms Explained. • 1, *page 1* Objectives. • 2, *page 1* Policies. • 3, *page 1* Responsibilities. • 4, *page 1* Administration. • 5, *page 1* 

**UNCLASSIFIED** 

RESERVED

#### 1. Terms Explained.

a. Nomenclature. The combination of an item name and type designation.

*b. Item name.* A name published in Federal Cataloging Handbook H-1, or that name developed by the requester in accordance with MIL-STD-100, in that portion applicable to drawing titles.

*c. Type Designation.* A specific combination of letters and numerals, structured in accordance with MIL-STD-196, that provides a standard means of uniquely identifying electronic materiel by design configuration.

*d. Department of Defense Control Point (DODCP).* The official focal point within the Department of Defense that is responsible for administration of the JETDS and is authorized to assign type designations.

*e. Department Control Point (DCP).* The official focal point within a Military Department that is authorized to obtain a type designation from the DODCP.

#### 2. Objectives.

The policy and procedures established in this regulation constitute a management system for the JETDS. The objectives of this system are to insure that the JETDS—

a. Is properly administered and controlled.

*b*. Serves as a viable tool for identification of electronic materiel and for use in configuration management of this material.

c. Provides identification of electronic materiel type similarities and significant differences.

*d*. Is sufficiently flexible and broad in scope to be applicable to present and future electronic materiel.

#### 3. Policies.

*a.* The administration of the JETDS will be consistant with the objectives specified in MIL-STD-196, Defense Standardization Manual 4120.3-M and this regulation.

b. The assignment of type designations will-

(1) Provide visibility of electronic materiel in the service inventories.

(2) Be responsive to user requirements.

(3) Provide singular, common identification.

(4) Further the standardization of electronic materiel identification.

c. The assignment of type designations will be consistent with-

(1) DOD Instruction 5010.21.

(2) MIL-STD-196.

(3) MIL-STD-280.

*d*. Type designations will be used as assigned. Changes may be requested if they are consistent with established policy.

e. Application of the JETDS is amplified as follows:

(1) Type designations may be assigned to classified materiel in the development stage to provide an unclassified means of identification.

(2) Other departmental designations may be replaced by JETDS type designations.

(3) US Communications Security (COMSEC) materials that are under the National Security Agency (NSA) Telecommunications Security(TSEC) Nomenclature Systems will not be redesignated under JETDS.

(4) Revisions of the JETDS will not require redesignation action. *f.* Matters of controversy which cannot be resolved by the DODCP and the DCPs will be referred to the MCEB through normal military department administrative channels.

#### 4. Responsibilities.

*a.* The US Military Communications-Electronics Board (MCEB-)will provide guidance on the JETDS and aid in the resolution of problems and controversies. This function will be accomplished by its Joint Equipment Standardization Panel, which will—

(1) Exercise continuing monitorship of the JETDS.

(2) Provide guidance on policy matters concerning the JETDS.

(3) Resolve those matters of controversy which cannot be resolved by the DODCP and the DCPS.

*b.* The Department of the Army will designate an appointment activity to be the DODCP. This DODCP will—

(1) Administer the JETDS in coordination with the DCPs.

(2) Respond to guidance from the MCEB.

(3) Continuously refine the JETDS in coordination with the DCPs.

(4) Collect, maintain and distribute MIL-STD-196 technical data for each assigned type designation.

(5) Communicate directly with the DCPs on matters pertaining to the JETDS.

*c*. Each military Department/agency will designate a DCP. More than one DCP may be established where functional alignment and responsibilities necessitate. The DCP(s) will—

(1) Act as the Military Department/agency focal point for processing type designation requests to the DODCP.

(2) Coordinate with the DODCP on the administration of the JETDS.

(3) Coordinate with the DODCP on refinements to the JETDS.(4) Refer matters of controversy which cannot be resolved with the DODCP to the Joint Equipment and Standardization Panel of the MCEB through Military Department administrative channels.

#### 5. Administration.

a. Technical data. The DODCP collects, maintains, and distributes technical data for each assigned type designation.

(1) Regrading of existing classified technical data is accomplished in accordance with the automatic downgrading directives indicated in b(2) below. Earlier regrading action is encouraged and may be accomplished by DCP submission of a DD Form 61, Request for Nomenclature, to the DODCP.

(2) Dissemination of technical data will be limited to those DOD activities which have justified a need and to those specifically designated by the DOD.

(3) Access to classified technical data will be authorized by the DCP which requested the nomenclature.

*b. Request for nomenclature.* Requests for nomenclature are submitted by the DCP to the DODCP on DD Form 61, Request for Nomenclature.

(1) Requests for nomenclature will be completed in accordance with MIL-STD-196.

(2) Security classification.

(a) Requests for nomenclature shall include both the security classification of the item (hardware) described and that of the information (data) provided thereon. Unclassified requests pertaining to classified equipment will be stamped "UNCLASSIFIED" on the top and bottom.

(b) All classified requests for nomenclature will bear the appropriate marking required by Air Force Regulation 205-1, Army Regulation 380-5, OPNAV Instruction 5500.40B, or other applicable security directive.

(c) Type designation reservations. Prior to submission of a classified request for nomenclature to the DODCP, the DCP will ensure that each data element on the request is marked with the appropriate security classification symbol immediately preceding and to the left of that data element involved. The symbols (S), (C), and (U) shall be used respectively for SECRET, CONFIDENTIAL, and UNCLASSIFIED.

*c*. Reservation of type designations is limited to Military Department high priority requirements and emergencies. Reservations will not be made without sufficient information to permit determination of full nomenclature. The procedure for type designation reservation is as follows:

(1) The DCP submits the request to the DODCP. The request must be accurate and clear in its recommendations to preclude delays and erroneous assignments. Security classification considerations will be based on the provisions of b(2) above.

(a) The request may be by letter, message, or telephone, depending upon its urgency. Telephone request will be confirmed by message or letter as soon as possible.

(b) The request must cite the requesting DCP, request number, item name, requested type designator, and whether or not development and/or production designation is needed.

(c) When available, manufacturer's drawing number, part number, or model number will be included.

(2) The DOCP will confirm all reservations using the same medium as the request. Telephone confirmations will be followed by a message or letter as soon as possible.

(3) The DCP will submit a DD Form 61 or notification of requirement cancellation within 60 days after obtaining the reservation.



# Department of Defense DIRECTIVE

NUMBER 3222.3

August 20, 1990

ASD(C3I)

SUBJECT: Department of Defense Electromagnetic Compatibility Program

# References: (a) DoD Directive 3222.3, "Department of Defense Electromagnetic Compatibility Program," July 5, 1967 (hereby canceled)

- (b) DoD Directive 5160.57, "Electromagnetic Compatibility Analysis Center (ECAC)," September 23, 1966 (hereby canceled)
- (c) DoD Directive 4120.3, "Defense Standardization and Specification Program," February 10, 1979
- (d) <u>DoD Directive 4650.1</u>, "Management and Use of the Radio Frequency Spectrum," July 24, 1987
- (e) DoD Directive C-3222.5, "Electromagnetic Compatibility (EMC) Management Program for SIGINT Sites (U)," April 22, 1987
- (f) "Manual of Regulations and Procedures for Federal Radio Frequency Management," authorized by U.S. Department of Commerce, May 1989 Edition (with revisions for September 1989 and January 1990)<sup>1</sup>

# 1. <u>REISSUANCE AND PURPOSE</u>

This Directive reissues reference (a) to:

1.1. Update DoD policy on an integrated EMCP to ensure EMC of all military electronic and telecommunications equipments, subsystems, and systems during their conceptual, design, acquisition, and operational phases.

1.2. Assign specific and joint responsibilities to DoD Components for leadership in the following EMCP areas:

1.2.1. Database and analysis capability.

<sup>&</sup>lt;sup>1</sup> Available through the U.S. Department of Commerce, National Telecommunications and Information Administration, Room 1605, 14th and Pennsylvania Avenue, NW, Washington, DC 20230.

1.2.2. Standards and specifications.

1.2.3. Education for EMC.

1.2.4. Design.

1.2.5. Doctrine, tactics, techniques, and procedures.

1.2.6. Operational problems.

1.2.7. Test and validation.

1.3. Provide the following EMCP objectives:

1.3.1. Achievement of EMC of all electronic and electrical equipments, subsystems, and systems that are produced and operated by DoD Components. Operational compatibility with the equipment, subsystem, or systems when used in their intended environment is part of, and the main focus of, that objective.

1.3.2. Attainment of built-in design compatibility instead of after-the-fact remedial measures.

1.3.3. Fostering of common DoD-wide philosophies, approaches, and techniques in the design, production, test, and operation of military telecommunications equipments.

1.4. Incorporate DoD Directive 5160.57 (reference (b)) into this Directive to specify the functions, responsibilities, operational relationships, and fiscal arrangements for the ECAC.

# 2. <u>APPLICABILITY</u>

This Directive applies to the Office of the Secretary of Defense (OSD); the Military Departments; the Chairman, Joint Chiefs of Staff and Joint Staff; the Unified and Specified Commands; the Inspector General, Department of Defense (IG, DoD); the Defense Agencies; and the DoD Field Activities (hereafter referred to collectively as the "DoD Components"). The term "Military Services," as used herein, refers to the Army, the Navy, the Air Force, and the Marine Corps.

### 3. <u>DEFINITIONS</u>

3.1. <u>Electromagnetic (EM) Compatibility (EMC)</u>. The ability of telecommunications equipments, subsystems, and systems to operate in their intended operational environments without suffering or causing unintentional unacceptable degradation because of EM radiation or response.

3.2. <u>Telecommunication</u>. Any transmission, emission, or reception of signs, signals, writings, images, sounds, or, information of any nature by wire, radio, visual, or other EM systems.

### 4. <u>POLICY</u>

4.1. Policy guidance for the EMCP and the ECAC shall be provided jointly by the Assistant Secretary of Defense (Command, Control, Communications, and Intelligence) (ASD(C3I)) and the Chairman, Joint Chiefs of Staff (CJCS), or designees.

4.2. EMCP applies to the conceptual, design, acquisition, and operational phases for all military electronic and telecommunications equipments, subsystems, and systems. The EMCP includes the following areas:

4.2.1. <u>Database and Analysis Capability</u>. Acquisition of effective databases and mathematical and statistical tools for EM analysis of any telecommunications component, circuit, equipment, subsystem, system, environment, concept, or doctrine and the ability to apply those tools to predict, prevent, and correct incompatibilities. That shall require:

4.2.1.1. A DoD-wide data collection and verification plan to ensure complete and current databases adequate to describe any probable telecommunications environment in sufficient technical and operational detail.

4.2.1.2. Common data processing and analytical techniques to provide rapid and timely summaries of data and analyses of equipments within known or expected environments, site selection and evaluation, analyses of concepts and doctrine for the use of telecommunications equipment in support of military operations, and solution of existing EM operational problems.

4.2.2. <u>Standards and Specifications</u>

4.2.2.1. Adequate and useful military standards and specifications for design, development, procurement, production, test, and measurement related to EMC shall be developed. Retrofits, modifications, and upgrades of fielded telecommunications equipments, systems, and subsystems shall be subject to that process, consistent with projected modification costs and expected operational life of the equipment or system, and shall be done in accordance with DoD Directive 4120.3 (reference (c)).

4.2.2.2. Adherence by all DoD Components to all DoD EMC standards and specifications shall be mandatory for the applicable operational telecommunications equipments, subsystems, and systems, unless duly waived.

4.2.2.3. Where required standards and specifications for EMC either do not exist or need correction, they shall be developed or updated promptly.

4.2.2.4. Authority for waiver control over any of the EMCP standards and specifications shall best at a level determined by the DoD Component concerned. That authority shall be delegated with careful discretion to prevent evasion of the EMC standards and specifications. Notification of all waiver actions must be provided concurrently to the CJCS, or designee of EMCP issues, for review and maintenance of records.

4.2.3. <u>Measurement Techniques and Instrumentation</u>. DoD Components shall maintain the capability to determine EM interference (EMI) effects and verify EMC through measurement. Basic electronic and electrical engineering tools and automated measurement techniques, rather than special purpose instrumentation, shall be used when available.

4.2.4. <u>Education for EMC</u>. Awareness of the effects of EMC deficiencies on the part of all DoD personnel concerned with the design, development, production, test, operational use, and maintenance of military electronic and telecommunications equipment shall be attained through:

4.2.4.1. Training of designers and engineers in the design and production methods and techniques for achieving EMC.

4.2.4.2. Training of operating and maintenance personnel in field techniques to optimize EMC.

4.2.4.3. Emphasis on EMC considerations as a portion of basic electronic and electrical engineering techniques.

4.2.4.4. Inclusion of EMCP within the training curricula for acquisition managers.

4.2.5. <u>Design</u>. Emphasis shall be placed on designing systems, from their conception, to achieve desired EMC performance levels. System design requires the selection of those techniques, circuits and components during the research, development, test, and evaluation (RDT&E) phases necessary to achieve compatible system operation.

4.2.6. <u>Doctrine, Tactics, Techniques, and Procedures</u>. EMC and EMI factors in the field employment of telecommunications equipments, subsystems, and systems shall be considered in the development of doctrine, tactics, techniques, and procedures. To minimize the impact of EMI factors shall require:

4.2.6.1. Analysis for EMC of all current and proposed doctrine, tactics, techniques, and procedures in the earliest possible timeframe to ensure that they shall not be invalidated by degradation of sensors or communications equipment from mutual or external interference.

4.2.6.2. Consideration of EMC factors in war-gaming to ensure awareness of the total EM environment in the evolution of new doctrine, tactics, techniques, and procedures.

4.2.7. <u>Operational Problems</u>. Development of a capability for detecting, reporting, solving, and correcting current time-frame operational EMC problems shall require:

4.2.7.1. Procedures for detecting and channels for reporting EM incompatibilities and EMI, which degrade combat effectiveness in the field.

4.2.7.2. Application of existing measurement and analysis techniques to identify the sources of the problems and determine corrective action.

4.2.7.3. Procedures for rapid implementation of required corrective action.

4.2.8. Test and Validation. Field engineering test facilities and testing in the

intended operational environments are required to verify predicted performance and to establish confidence in EMC design, based on standards and specifications, and in EMC analysis methodology, thus providing:

4.2.8.1. Problem parameter measurements.

4.2.8.2. Evaluation of EMC analysis and predictions in appropriate (real or emulated) environments.

### 5. <u>RESPONSIBILITIES</u>

5.1. The <u>Assistant Secretary of Defense (Command, Control, Communications,</u> <u>and Intelligence)</u> (ASD(C3I)) and the <u>Chairman, Joint Chiefs of Staff</u> (CJCS), or designees, shall be responsible, jointly, for:

5.1.1. Oversight of the EMCP.

5.1.2. Providing policy guidance and direction to the ECAC.

5.1.3. Providing specific direction, as necessary, to ensure a well-coordinated and current EMCP.

5.1.4. Reviewing ECAC's EMCP plan for providing EMC analysis capabilities and use of the EMC databases, annually, and, if required, designating DoD Components to carry out specific requirements of that plan.

5.1.5. Ensuring that adequate database and analysis support is provided by the ECAC to the DoD Components having those responsibilities. The functions of frequency management have a strong influence on operational compatibility, and require database and analysis support. DoD Directive 4650.1 (reference (d)) assigns responsibilities in this area.

5.2. The <u>Chairman, Joint Chiefs of Staff (CJCS)</u>, or designee, shall be responsible for:

5.2.1. Collection of information for EMC databases. Those collection efforts should especially consider EMC program needs for spectrum signatures, characteristics, locations, and operating plans for newly fielded or modified systems.

5.2.2. Submission of doctrine, tactics, techniques, and procedures for joint operations to the ECAC for analysis of EMC impact.

5.2.3. Developing and implementing procedures and channels for detecting and reporting current joint operational EMC and EMI problems.

5.3. The <u>Director, Electromagnetic Compatibility Analysis Center</u> (ECAC), shall be responsible for:

5.3.1. The DoD joint EMC program that shall include:

5.3.1.1. Development and maintenance of a coordinated plan for providing EMC analysis capabilities and use of the EMC databases.

5.3.1.2. Development, maintenance, and distribution of EMC databases and EMC analysis models.

5.3.1.3. Providing technical and/or operational EMC analysis support to the ASD(C3I), the CJCS, and the Military Communications Electronics Board (MCEB).

5.3.2. The sponsored EMC program that shall include:

5.3.2.1. Providing EMC analysis on a reimbursable basis to DoD Components developing or operating telecommunications equipment.

5.3.2.2. Providing EMC analysis on a reimbursable basis to other Departments of the U.S. Government and to others, as authorized by the ASD(C3I) (e.g., foreign governments and industry under contract to the U.S. Government).

NOTE: In furtherance of that sponsored program, the ECAC is authorized to communicate directly with all of the DoD Components and other non-DoD activities.

5.4. The Secretary of the Air Force shall:

5.4.1. Be designated as the administrative agent for ECAC.

5.4.2. Program, budget, and finance the joint EMC program at the ECAC, described in paragraph 5.3.1., above. The joint EMC program budget and changes to the joint EMC program budget (e.g., reprogrammings) shall be coordinated with the ASD(C3I) and the CJCS, or designees.

5.4.3. Appoint the Director, ECAC, who shall be a Colonel, subject to approval by the ASD(C3I) and the CJCS, or designees.

5.5. The Secretaries of the Military Departments shall:

5.5.1. Provide a suitably qualified field grade officer to serve as a Deputy Director, ECAC. The responsibilities of the Deputy Directors at ECAC are to:

5.5.1.1. Assist the Director, ECAC, in providing quality EMCP support to all users.

5.5.1.2. Monitor the status of Service-related projects.

5.5.1.3. Assist the spectrum management offices of each respective Service.

5.5.2. Provide such personnel to ECAC, as designated for that Service by the Joint Staff Table of Distribution (JTD).

5.6. The <u>Director, National Security Agency</u> (NSA), shall be responsible for the signals intelligence (SIGINT) portions of the EMCP, as assigned by DoD Directive C-3222.5 (reference (e)).

5.7. The <u>Heads of DoD Components</u> shall be jointly responsible for leadership in each of the following EMCP areas:

5.7.1. Database and Analysis Capability

5.7.1.1. Ensuring that ECAC databases are complete and current for equipments, subsystems, and systems that are acquired, developed, or operated by their respective DoD Components.

5.7.1.2. Using ECAC capabilities maximally, rather than developing duplicate databases or EMC analysis tools. The need for separate databases should decrease as communications between data processing systems improve.

5.7.1.3. Developing new databases and analytical techniques when required for intra-DoD Component problems that, with minimum modification, may be exchanged with and used by the DoD Components.

5.7.2. Standards and Specifications

5.7.2.1. Developing and maintaining a complete range of component, circuit, equipment, subsystem, and system EMC standards. Related standards for prediction, measurement, and validation of EMC shall be included.

5.7.2.2. Ensuring that all specifications for telecommunications equipment cite appropriate EMC standards.

5.7.3. Education for EMC

5.7.3.1. Ensuring that properly balanced emphasis on EMC is included in all formal courses in design, maintenance, and operation of telecommunications components, circuits, equipment subsystems, and systems conducted within their organization.

5.7.3.2. Maintaining current handbooks describing the most effective techniques for meeting the standards for EMC. Adoption of other DoD Component handbooks, which are adequate, is encouraged.

5.7.3.3. Ensuring adequate participation by appropriate members of their staff in symposia, conferences, and other professional activities of the industry organizations and technical societies concerned with EMC and electronic engineering.

5.7.4. <u>Design</u>. Emphasizing EMC in RDT&E of telecommunications equipments, subsystems, and systems and ensuring exchange of information regarding results of those efforts.

5.7.5. <u>Doctrine, Tactics, Techniques, and Procedures</u>. Providing proper EMC-impact consideration in the formulation of their intra-DoD Component doctrine, tactics, techniques, and procedures.

5.7.6. <u>Operational Problems</u>. Developing and implementing procedures and channels for detecting, reporting, solving, and correcting their intra-DoD Component operational EMC problems. They shall provide feedback from that to the standards, design, concepts, and doctrine, and educational and analytical elements of the EMCP.

# 5.7.7. Test and Validation

5.7.7.1. Development of individual procedures and, as appropriate, in the development of an inter-DoD Component coordinated plan for test and validation in support of the EMCP.

5.7.7.2. Examining and supporting test validation of the EMC performance of their various telecommunications equipments in joint operational environments, as specified by the CJCS, or designee. This involves a dedicated effort in understanding and/or verifying the equipments' performance due to EMC and EMI factors.

#### 6. <u>RELATIONSHIPS</u>

6.1. Other Government Agencies and the Civilian Community EMC problems are common to all users of the EM spectrum. A successful program must consider and serve all who use telecommunications equipment. Within the constraints of national security, and fund or facility availability, the capabilities attained under this EMCP may be made available to non-DoD Components; e.g., the National Telecommunications and Information Administration (NTIA), the Federal Communications Commission (FCC), and other Government Agencies; the International Telecommunications Union (ITU); and the civilian community. The "Manual of Regulations and Procedures for Federal Radio Frequency Management" (reference (f)) gives the interrelationships among the Department of Defense, NTIA, FCC and ITU.

6.2. Electronic Countermeasures (ECM), Electronic Counter-Countermeasures (ECCM), Electronic Combat (EC), EM effects, EM Pulse (EMP), and Radiation Hazards (RADHAZ) Programs. Those programs are specific aspects of the use of or defense against effects of EM radiations. Their existence, as separate programs, is predicated on either military requirements or on overriding urgency due to danger to personnel. As the EMCP progresses, it should augment, be used by, and, in some instances, be integrated with those programs. Advances in EMC that are based upon basic advances in EM technology should be shared among applicable programs. As a minimum, other DoD programs shall be so conducted that equipments and systems developed for their special purposes shall meet all applicable EMC standards of conventional telecommunications equipments, subsystems, and systems. The EMC community should be alert also for techniques developed in other programs that have general application (i.e., ECCM techniques that may also be effective against unintentional interference).

#### 7. EFFECTIVE DATE

This Directive is effective immediately.

Timeled fatured

Donald J. Atwood Deputy Secretary of Defense



# Department of Defense DIRECTIVE

NUMBER 5200.1

December 13, 1996

ASD(C3I)

SUBJECT: DoD Information Security Program

# References: (a) DoD Directive 5200.1, subject as above, June 7, 1982 (hereby canceled)

- (b) Executive Order 12958, "Classified National Security Information," April 20, 1995, as amended
- (c) Information Security Oversight Office Directive, "Classified National Security Information," October 13, 1995
- (d) DoD Instruction 5230.21, "Protection of Classified National Security Council and Intelligence Information," March 15, 1982 (hereby canceled)
- (e) through (i), see enclosure l

### 1. REISSUANCE AND PURPOSE

This Directive:

1.1. Reissues reference (a) to update policy and responsibili-ties for the DoD Information Security Program under references (b) and (c).

1.2. Replaces references (d) through (f).

1.3. Continues to authorize the publication of DoD 5200.1-R (reference (g)), in accordance with DoD 5025.1-M (reference (h)).

### 2. <u>APPLICABILITY</u>

This Directive applies to the Office of the Secretary of Defense, the Military Departments, the Chairman of the Joint Chiefs of Staff, the Combatant Commands, the Inspector General of the Department of Defense, the Defense Agencies, and the DoD Field Activities (hereafter referred to collectively as "the DoD Components").

#### 3. <u>DEFINITIONS</u>

3.1. <u>Compromise</u>. A communication or physical transfer of classified information to an unauthorized recipient.

3.2. <u>Information</u>. Any knowledge that may be communicated or documentary material, regardless of its physical form or characteristics, that is owned by, produced by or for, or is under the control of, the Department of Defense.

3.3. <u>National Security</u>. The national defense or foreign relations of the United States.

#### 4. POLICY

It is DoD policy that:

4.1. National security information shall be classified, declassified and safeguarded, in accordance with national-level policy issuances. Misclassification shall be avoided.

4.2. Declassification of information shall receive equal attention with classification to ensure that information remains classified only as long as required by national security considerations.

4.3. The volume of classified national security information shall be reduced to the minimum necessary to meet operational requirements.

4.4. An active security education and training program shall be established and maintained to ensure that DoD military and civilian personnel who require access to classified national security information in the conduct of official business are familiar with their responsibilities for protecting such information from unauthorized disclosure.

#### 5. <u>RESPONSIBILITIES</u>

5.1. The <u>Assistant Secretary of Defense for Command, Control, Communications,</u> and <u>Intelligence</u> shall: 5.1.1. Serve as the Senior Agency Official for the Department of Defense under subsection 5.6.(c) of E.O. 12958, as amended (reference (b)).

5.1.2. Direct, administer, and oversee the DoD Information Security Program to ensure that the program is efficient, recognizes assigned authorities and responsibilities, and that appropriate management safeguards are in place to prevent fraud, waste, and abuse.

5.1.3. Approve, when appropriate, requests for exceptions to DoD Information Security Program policies and procedures.

5.1.4. Approve and publish DoD Instructions and Publications, as necessary, to guide, direct, or help DoD Information Security Program activities, consistent with DoD 5025.1-M (reference (h)).

5.1.5. Encourage liaison between the DoD Components and industry; professional associations; academia; Federal, State, and local government organizations; and international organizations to acquire information that may be of use in improving the DoD Information Security Program.

5.1.6. Assist the Under Secretary of Defense for Acquisition and Technology, as required, in implementing the DoD Acquisition Systems Protection Program, both by establishing security policy and providing technical security support to that program.

5.2. The Under Secretary of Defense for Policy shall:

5.2.1. Direct, administer and oversee that portion of the DoD Information Security Program pertaining to Special Access Programs, foreign government (including North Atlantic Treaty Organization) classified information, the National Disclosure Policy and security arrangements for international programs.

5.2.2. Approve, when appropriate, requests for exception to policy involving any programs listed in paragraph 5.2.1., above.

5.3. The Assistant Secretary of Defense for Public Affairs shall:

5.3.1. Direct and administer a DoD Mandatory Declassification Review Program under subsection 3.6. of E.O. 12958 (reference (b)).

5.3.2. Establish policies and procedures for processing mandatory declassification review requests, including appeals consistent with subsection 3.6.(d) of reference (b) and Section 2001.13 of the Information Security Oversight Office Directive (reference (c)), which make maximum use of DoD Component resources and systems established to implement DoD Directive 5400.7 (reference (i)).

5.4. The <u>Under Secretary of Defense for Acquisition and Technology</u> shall serve as the office of primary responsibility and provide day-to-day direction and management of the DoD Acquisition Systems Protection Program.

5.5. The <u>Secretaries of the Military Departments</u>, as Agency Heads under reference (b), and the <u>Heads of the Other DoD Components</u>, shall:

5.5.1. Designate a senior agency official for their respective Departments who shall be responsible for the direction and administration of the Department's information security program, to include active oversight, classification, declassification and security education and training programs to ensure effective implementation of reference (b) and DoD 5200.1-R (reference (g)).

5.5.2. Ensure that funding and resources are adequate to carry out such oversight, classification, declassification and security education and training programs.

5.5.3. Consider and take action on complaints and suggestions from persons in or outside the Government regarding the Department's Information Security Program.

5.6. The <u>Director, National Security Agency</u>, shall, as the designee of the Secretary of Defense, when necessary, impose special requirements on the classification, declassification, marking, reproduction, distribution, accounting, and protection of and access to classified cryptologic information.

# 6. <u>EFFECTIVE DATE</u>

This Directive is effective immediately.

John P. White Deputy Secretary of Defense

Enclosures - 1 1. References

### E1. ENCLOSURE 1

#### **REFERENCES**, continued

- (e) DoD Instruction O-5230.22, "Security Controls on the Dissemination of Intelligence Information," August 17, 1988 (hereby canceled)
- (f) DoD Directive 5200.12, "Conduct of Classified Meetings," July 27, 1992 (hereby canceled)
- (g) DoD 5200.1-R, "Department of Defense Information Security Program Regulation," January 17, 1997 authorized by this Directive
- (h) DoD 5025.1-M, "DoD Directives System Procedures," August 1994, authorized by DoD Directive 5025.1, June 24, 1994
- (i) DoD Directive 5400.7, "DoD Freedom of Information Act Program," May 13, 1988

#### ANNEX J

### **Guidance for Determination of Necessary Bandwidth**

#### **1.1 INTRODUCTION**

This Annex contains guidance relating to the necessary bandwidth parameter. Necessary bandwidth forms part of the emission designator used for frequency management purposes and is used as a parameter in spectrum standards, frequency assignments, etc., throughout this Manual.

#### 2.1 GENERAL

Except for radars, the necessary bandwidth may be determined by one of the following methods with the order of preference shown:

1. Use of the appropriate formula from Table A in this Annex.<sup>1</sup>

2. Computation in accordance with the Recommendations ITU-R SM.328-8 (1994) and SM.853 (1994).

3. Measurements of specialized modulations not covered by 1. or 2. above.

4. Use of the best available technical information from other sources.

The value so determined shall be used when the full designation of an emission is required for example, as indicated in Chapter 9.

See Section 5.1.5 for the desired relationship of occupied bandwidth to necessary bandwidth.

#### **3.1 RADAR SYSTEMS**

For radars the necessary bandwidth shall be determined at a point that is 20 dB below the peak envelope value of the spectrum by one of the following with the order of preference shown:

1. Computation in accordance with the following equations which assume trapezoidal pulse modulation, with equal rise and fall times. a. for non-FM pulsed radars<sup>2</sup> (including spread spectrum or coded pulse radars):

$$B(-20 dB) = \frac{1.79}{\sqrt{t_r t}}$$
 or  $\frac{6.36}{t}$  whichever is less

b. for FM-pulse radars (intentional FM)<sup>2</sup>

$$B(-20dB) = \frac{1.79}{\sqrt{t_r t}} + 2B_c$$

c. for FM-pulse radars<sup>2,3</sup> (intentional FM with frequency hopping)

$$B(-20 dB) = \frac{1.79}{\sqrt{t_r t}} + 2B_c + B_s$$

d. For frequency hopping radars using non-FM pulses<sup>2,3</sup> (including spread spectrum or coded pulses)

$$B(-20 dB) = \frac{1.79}{\sqrt{t_r t}} + B_s$$

e. for CW radars<sup>4</sup>

- 2. If  $t_f$  is less than  $t_r$ , then  $t_f$  is to be used in place of  $t_r$  when performing the necessary bandwidth calculations.
- 3. These formulas yield the total composite B(-20dB) bandwidth of a frequency hopping radar as if all channels included within  $B_s$  were operating simultaneously. Individual channels have a B(-20 dB) radar necessary bandwidth given by the equations in Annex J, Paragraphs 3.1.1.a and b.
- 4. The emission bandwidth of a CW transmitter typically will not be zero due to noise and frequency tolerance considerations. However, designating zero as the necessary bandwidth is a valid method for identifying such equipment.

<sup>1.</sup> Individual formulas may be based on theoretical models for the modulation technique.

B(necessary) = 0

f. for FM/CW radars

 $B(necessary) = 2B_d$ 

Where:

B = necessary bandwidth in MHz.

 $B_c$  = bandwidth of the frequency deviation (the total frequency shift during the pulse duration) in MHz.

 $B_d$  = bandwidth of the frequency deviation (peak difference between instantaneous frequency of the modulated wave and the carrier frequency for FM/CW radar systems) in MHz.

 $B_s = maximum range in MHz$  over which the carrier frequency will be shifted for a frequency hopping radar. t = emitted pulse duration in : sec at 50% amplitude (voltage) points. For coded pulses, the pulse duration is the interval between 50% amplitude points of one chip (sub-pulse). The 100% amplitude is the nominal flat top level of the pulse.

 $t_r$  = emitted pulse rise time in : sec from the 10% to the 90% amplitude points on the leading edge. For coded pulse, it is the rise time of a sub-pulse; if the subpulse rise time is not discernible, assume it is 40% of the time to switch from one phase or sub-pulse to the next.

 $t_f$  = emitted pulse fall time in  $\mu$  sec from the 90% to the 10% amplitude points on the trailing edge.

2. Computation in accordance with ITU-R Appendix 1 and ITU-R M1138. If ITU Regulations are used, the value of K should be 4.5, for the formula  $B_n=2K/t$ .

3. Results of actual measurement.

4. Use of the best available technical information from other sources.

#### 4.1 ANALOG FM

The basis of the formulas in Table A for the necessary bandwidth of analog FM and FDM/FM systems is Carson's Rule. This bandwidth is given by  $B_1=2$  (D+M)=2(a+1)M, where D is the peak frequency deviation, "a" is the peak modulation index and M is the maximum modulating frequency. This rule represents an additive combination of the bandwidth expressions for extreme high

 $(B_1 \sim 2D=2aM)$  and low  $(B_1 \sim 2M)$  modulation index conditions. One of these two expressions prevails over the other for  $\alpha \gg 1$  or  $\alpha \ll 1$ , so that their linear superposition always yields the bandwidth measure for extreme index conditions.

An accepted relationship between analog FM bandwidth and a measure of performance such as allowable distortion as a function of the modulation index is not available. There is no distortion measure or criterion that is generally accepted for evaluation purposes, because of difficulties arising from the variety of modulating signal characteristics and models that occur in practice.

The normalized FM bandwidth ( $B_1/M$ ) for single tone sinusoidal modulation is shown in Figure 1 for various power percentages included. Each stepped line corresponds to a fixed power percentage (p). The solid stepped line represents p=99% power included. The normalized bandwidth based on Carson's Rule is given by  $(B_1/M)=2(a+1)$ , shown in Figure 1 by the solid straight line. Carson's Rule essentially follows the p=99% line for indices in the 0.9 < a < 4.3 range. It also includes more power at lower indices, but falls progressively below the 99% power curve at higher indices outside this range.

The case of a random modulating signal with a uniform baseband spectrum has also been analyzed using included power as the band-limiting distortion criterion. A peak to rms load ratio of 11 dB has been assumed to simulate representative conditions of FDM/FM telephony. The resultant normalized bandwidth can be estimated by  $(B_1/M)=2Z(a,q)$  where Z is a function of "a" and the fractional power rejected q=1-(p/100) as follows (Refs b and c):

$$Z(q,a) = a[\sqrt{1 - \log_{(q^{3/7},3^3)}} - 0.05] + 0.75$$

This expression is an effective approximation to a complicated integral formulation for moderate index values (1 < a < 5). The normalized bandwidth (B<sub>1</sub>/M) is shown in Figure 2 for various (q) values, along with the bandwidth formula corresponding to Carson's Rule. The latter can be noted to represent a power rejection in the  $10^{-10} < q < 10^{-8}$  range, which is negligible.

The modulation cases shown in Figures 1 and 2 are extreme energy distribution conditions, in that one has all the baseband energy concentrated on a single frequency while the other has it spread uniformly over the baseband. The implication of Figures 1 and 2 is that Carson's Rule represents an effective bound to calculating analog FM bandwidth from a power included standpoint for modulation indices below five. The results also indicate that Carson's Rule includes considerably more power when the baseband modulation has a spread rather than concentrated spectral characteristic. Carson's Rule represents a q=0.01 power rejection sinusoidal modulation. for simple and  $10^{-10} < q < 10^{-8}$  power rejection for a random modulation with a uniform baseband spectrum.

The necessary bandwidth of analog FM systems with modulation indices greater than 5.0 should be based on the methods of subparagraphs 2, 3 and 4 of the above GENERAL section.

See References a, b, and c.

#### 5.1 SYMBOLS

As appropriate, the following table shall be used for calculation of necessary bandwidth. The following symbols are used in this table:  $B_n =$  Necessary bandwidth

B = Digital symbol rate for telegraphy (i.e. baud)

N = Maximum possible number of black plus white elements to be transmitted per second, in facsimile

M = Maximum modulation frequency

C = Sub-carrier frequency

D = Peak deviation, i.e., half the difference between the maximum and minimum values of the instantaneous frequency.

t = Pulse duration in seconds at half-amplitude

K = An overall numerical factor which varies according to the emission and which depends upon the allowable signal distortion

 $N_c$  = Number of baseband channels in radio systems employing multichannel multiplexing

 $f_p$  = Continuity pilot sub-carrier frequency (continuous signal utilized to verify performance of frequency-division multiplex systems).

P = Continuity pilot sub-carrier frequency in hertz<sup>5</sup>

S = Number of equivalent non-redundant signaling states.

R = Digital information rate

<sup>5.</sup> The frequency (P) of a continuity pilot sub-carrier in frequency modulated radio relay systems may exceed M.



**MODULATION INDEX (a)** 

FIGURE 1.

FM Bandwidth Occupancy and Power Preservation with Sinusoidal Modulation.

(Note: Carson's Rule is the Straight Line)

(Legend: p is the Power Percentage Preserved)


## **MODULATION INDEX (a)**

FIGURE 2.

## FM Bandwidth Occupancy and Power Preservation with Band-Limited White Modulation.

(Note: Carson's Rule is the Dotted Line)

(Legend: q is the Power Fraction Rejected)

## TABLE A

## Necessary Bandwidth Calculations

## I. UNMODULATED SIGNAL

Necessary Bandwidth				
Description of EmissionFormulaSample CalculationDesignation of Emission				
Continuous wave emission			NON	

## **II. AMPLITUDE MODULATION**

## 1. Signal with Quantized or Digital Information

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Continuous wave telegraphy, Morse code	$B_n = BK$ , K=5 for fading circuits, K=3 for non-fad- ing circuits	25 words per minute; B=20, K=5, $B_n$ =100 Hz	100H00A1AAN
Telegraphy by on-off keying of a tone modulated carrier, Morse code	$B_n = BK + 2M, K = 5$ for fading circuits, K = 3 for non-fading circuits	25 words per minute; $B=20$ M=1000, K=5; $B_n=2100$ Hz=2.1 kHz	2K10A2AAN
Selective calling signal using sequential single frequency code, single-sideband full car- rier	B <sub>n</sub> =M	Maximum code frequency is 2110 Hz, $M=2110$ ; $B_n=2110$ ; Hz=2.11 kHz	2K11H2BFN
Direct printing telegraphy using a frequency shifted modulating sub-carrier, with error-correc- tion, single-sideband, sup- pressed carrier (single channel)	$B_n = 2M + 2DK;$ $M = \frac{B}{2}$	B=50, D=35 Hz (70 Hz shift) K=1.2; B <sub>n</sub> =134 Hz	134H00J2BCN
Telegraphy, multi-channel with voice frequency, error correc- tion, some channels are time- division multiplexed, single- sideband, reduced carrier	$B_n = (highest central frequency) + M + DK;$ $M = \frac{B}{2}$	15 channels; highest central is: 2805 Hz, B=100, D=42.5 Hz (85 Hz shift), K=0.7; $B_n=2885 Hz=2.885 kHz$	2K89R7BCW

## 2. Telephony (Commercial Quality)

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Telephony, double-sideband (single channel)	B <sub>n</sub> =2M	M=3000 $B_n=6000$ Hz=6 kHz	6K00A3EJN

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Telephony, single-sideband, full carrier (single channel)	B <sub>n</sub> =M	M=3000; B <sub>n</sub> =3000 Hz=3 kHz	3K00H3EJN
Telephony, single-sideband, suppressed carrier (single chan- nel)	B <sub>n</sub> =M-lowest modulation frequency;	M=3000, lowest modula- tion is 300 Hz; $B_n$ =2700 Hz=2.7 kHz	2K70J3EJN
Telephony with separate fre- quency modulated signal to con- trol the level of de-modulated speech signal, single-sideband, reduced carrier (Lincompex) (single channel)	B <sub>n</sub> =M	Maximum control fre- quency is 2990 Hz, M=2990; $B_n = 2990 Hz=2.99 kHz$	2K99R3ELN
Telephony with privacy, single- sideband, suppressed carrier (two or more channels)	B <sub>n</sub> =N <sub>c</sub> M lowest modulation fre- quency in the lowest chan- nels	$N_c=2$ , M=3000, lowest modulation fre- quency is 250 Hz; $B_n=5750=5.75$ kHz	5K75J8EKF
Telephony independent side- band (two or more channels)	B <sub>n</sub> =sum of M for each side-band	Two channels, M=3000; $B_n=6000 \text{ Hz}=6 \text{ kHz}$	6K00B8EJN

# 3. Sound Broadcasting

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Sound broadcasting double-side- band	$B_n=2M$ , M may vary between 4000 and 10000 depending on the quality desired	Speech and music, M=4000; B <sub>n</sub> =8000 Hz=8 kHz	8K00A3EGN
Sound broadcasting, single-side- band reduced carrier (single channel)	$B_n = M$ , M may vary between 4000 and 10000 depending on the quality desired	Speech and music, M=4000; B <sub>n</sub> =4000 Hz=4 kHz	4K00R3EGN
Sound broadcasting, single-side- band, suppressed carrier	B <sub>n</sub> =M-lowest modulation frequency	Speech and music, M=4500, lowest modula- tion frequency=50 Hz; $B_n$ =4450 Hz=4.45 kHz	4K45J3EGN

## 4. Television

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Television, vision and sound	Refer to Recommendations ITU-R BT.470-3 (1994) and BO. 650 (1994) for the bandwidths of the com- monly used television sys- tems	Number of lines = 525; Number of lines per sec- ond = 15,750; Video band- width: 4.2 MHz; Total visual band-width: 5.75 MHz; FM aural bandwidth including guardbands: 250 kHz Total bandwidth: 6 MHz	5M75C3F; 250K00F3EGN; 6M25C3F

## 5. Facsimile

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Analog facsimile by sub-carrier frequency modulation of a sin- gle-sideband emission with reduced carrier, monochrome	$B_n = C + \frac{N}{2} + DK$ K=1.1 (typically)	N=1100, corresponding to an index of cooperation of 352 and a cycler rotation speed of 60 rpm. Index of cooperation is the product of the drum diameter and number of lines per unit of length; C=1900, D=400 Hz; B <sub>n</sub> =2890 Hz=2.89 kHz	2K89R3CMN
Analog facsimile; frequency modulation of an audio fre- quency sub-carrier which modu- lates the main carrier, single- sideband suppressed carrier	$B_n = 2M + 2DK;$ $M = \frac{N}{2}$ $K = 1.1 \text{ (typically)}$	N=1100, D=400 Hz; B <sub>n</sub> =1980 Hz=1.98 kHz	1K98J3C

# 6. Composite Emissions

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Double-sideband television relay	B <sub>n</sub> =2C+2M+2D	Video limited to 5 MHz, audio on 6.5 MHz fre- quency modulation sub-car- rier, sub-carrier deviation= 50  kHz $C=6.5X10^{6}$ $D=5X10^{3} \text{ Hz}$ M=15000 $B_n=13.13X10^{6}$ Hz=13.13  MHz	13M10A8W
Double-sideband radio- relay system, frequency division multiplex	B <sub>n</sub> =2M	10 voice channels occupy- ing baseband between 1 kHz and 164 kHz; M=164000 $B_n=328000$ Hz=328 kHz	328K00A8E
Double-sideband emission of VOR with voice (VOR=VHF omni-direc- tional radio range)	B <sub>n</sub> =2C <sub>max</sub> +2M+2DK; K=1 (typically)	The main carrier is modu- lated by: - a 30 Hz sub-carrier - a carrier resulting from a 9960 Hz tone frequency modulated by a 30 Hz tone - a telephone channel - a 1020 Hz keyed tone for continual Morse identifica- tion $C_{max}=9960, M=30,$ D=480 Hz $B_n=20940 Hz=20.94 kHz$	20K90A9WWF
Independent sidebands; sev- eral telegraph channels with error-correction together with several telephone channels with privacy; fre- quency division multiplex	B <sub>n</sub> =sum of M for each sideband	Normally composite sys- tems are operated in accor- dance with standardized channel arrangements (e.g. ITU-R Rec. 348-4 (1994)). 3 telephone channels and 15 telegraphy channels require the bandwidth 12000 Hz=12  kHz	12K00B9WWF

## **J-10**

## **III. ANGLE MODULATION**

## 1. Signal with Quantized or Digital Information

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Telegraphy without error-cor- rection (single channel)	$B_n = 2M + 2DK$ $M = \frac{B}{2}$ $K = 1.2 \text{ (typically)}$	B=100, D=85 Hz (170 Hz shift) B <sub>n</sub> =304 Hz	304H00F1BBN
Telegraphy, narrow-band direct-printing with error-cor- rection (single channel)	$B_n = 2M + 2DK$ $M = \frac{B}{2}$ $K = 1.2 \text{ (typically)}$	B=100, D=85 Hz (170 Hz shift) $B_n=304 Hz$	304H00F1BCN
Selective calling signal	$B_n = 2M + 2DK$ $M = \frac{B}{2}$ $K = 1.2 \text{ (typically)}$	B=100, D=85 Hz (170 Hz shift) $B_n=304$ Hz	304H00F1BCN
Four-frequency duplex telegra- phy	$B_n = 2M + 2DK, B = Modu-$ lation rate in bauds of the faster channel. If the chan- nels are synchronized $M = \frac{B}{2}$ (otherwise M=2B), K=1.1 (typically)	Spacing between adjacent frequencies = 400 Hz; Synchronized channels B=100, M=50, D=600 Hz $B_n=1420$ Hz=1.42 kHz	1K42F7BDX

# 2. Telephony (Commercial Quality)

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Commercial telephony	$B_n=2M+2DK, K=1$ (typically, but under cer- tain conditions a higher value may be necessary)	For an average case of com- mercial telephony: $D=5000$ Hz, $M=3000$ $B_n=16000$ Hz=16 kHz	16K00F3EJN

# 3. Sound Broadcasting

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Sound Broadcasting	$B_n=2M+2DK, K=1$ (typically)	Monaural D=75000 Hz, M=15000 $B_n=180000 Hz=180 kHz$	180K00F3EGN

# 4. Facsimile

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Facsimile by direct frequency modulation of the carrier; black and white	$B_n = 2M + 2DK;$ $M = \frac{B}{2}$ $K = 1.1 \text{ (typically)}$	N=1100 elements/ sec, D=400 Hz; $B_n$ =1980 ; Hz=1.98 kHz	1K98F1C
Analogue facsimile	$B_n=2M+2DK; M=N$ OVER 2 K=1.1 (typically)	N=1100 elements/ sec, D=400 Hz; $B_n$ =1980; Hz=1.98 kHz	1K98F3C

# 5. Composite Emissions

Necessary Bandwidth <sup>a</sup>			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Radio-relay systems, frequency division multiplex	B <sub>n</sub> =2f <sub>p</sub> +2DK K=1 (typically)	60 all voice telephone chan- nels occupying baseband between 60 kHz and 300 kHz; rms per channel devia- tion: 200 kHz; continuity pilot at 331 kHz produces 100 kHz rms deviation of main carrier For X=-5.6: $D=(200X10^3)(3.76)$ $(1.19)=8.95X10^5$ Hz; $f_p=0.331X10^6$ Hz; $B_n=2.45X10^6$ Hz=2.45 MHz	2M45F8EJF

Necessary Bandwidth <sup>a</sup>			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Radio-relay system; frequency division multiplex	B <sub>n</sub> =2M+2DK K=1 (typically)	960 all voice telephone channels occupying base- band between 60 kHz and 4028 kHz; rms per channel deviation: 200 kHz; conti- nuity pilot at 4715 kHz pro- duces 140 kHz rms deviation of main carrier For X=-19.6: D=(200X10 <sup>3</sup> ) (3.76) (3.24)=2.43X10 <sup>6</sup> Hz; M=4.028X10 <sup>6</sup> ; f <sub>p</sub> =4.715X10 <sup>6</sup> ; (2M+2DK)>2f <sub>p</sub> ; B <sub>n</sub> =12.9X10 <sup>6</sup> Hz=12.9 MHz	12M9F8EJF
Radio-relay system, frequency division multiplex	B <sub>n</sub> =2f <sub>p</sub>		17M00F8EJF
Radio-relay system; frequency division multiplex	B <sub>n</sub> =2M+2DK K=1 (typically)	960 data channels that oper- ate at a uniform power level of -15 dBM occupying baseband between 60 kHz and 4028 kHz; rms per channel deviation: 200 kHz; continuity pilot at 4715 kHz produces 140 kHz rms devi- ation of main carrier $D=200X10^3$ (3.76) (5.5)= $4.13X10^6$ Hz; $M=4.028X10^6$ ; $f_p=4.715X10^6$ , $(2M+2DK) > 2f_p$ ; $B_n=16.32X10^6$ Hz=16.32 MHz	16M32F8DJF

5/2003

Necessary Bandwidth <sup>a</sup>			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Stereophonic sound broadcast- ing with multiplexed subsidiary telephone sub-carrier	$B_n = 2M + 2DK$ K=1 (typically)	Pilot tone system; M=75000, D=75000 Hz $B_n=300000 Hz=300 kHz$	300K00F8EHF
TV microwave relay system	B <sub>n</sub> =2M+2DK K=1 (typically)	Aural program on 7.5 MHz, aural sub-carrier deviation "150 kHz; conti- nuity pilot at 8.5 MHz pro- duces 140 kHz rms deviation of main carrier; $D=3.7X10^{6}$ Hz (visual) plus $0.3X10^{6}$ Hz (aural) Computation of $B_n$ : $M=(2.0+0.2)X10^{6}$ ; $P=8.5X10^{6}$ Hz; $D=(3.7+0.3)X10^{6}$ Hz; (2M+2DK) > 2P $B_n=23.3X10^{6}$ Hz=23.3 MHz	23M3F3WJF
TV microwave relay system	B <sub>n</sub> =2P	Aural program on 6.9 MHz sub-carrier; aural sub-car- rier deviation "150 kHz; continuity pilot at 8.5 MHz produces 50 kHz rms devia- tion of main carrier $D=2X10^6$ (visual) plus $0.2X10^6$ (aural); Computa- tion of B <sub>n</sub> : $M=(2.0+0.2)X10^6$ Hz; $D=6.16X10^6$ Hz; $K=1$ ; $P=8.5X10^6$ ; (2M+2DK) < 2P $B_n=17X10^6$ Hz=17 MHz	17M00F3WJF
Binary Frequency Shift Keying <sup>b</sup>	If $\left(0.03 < \frac{2D}{R} < 1.0\right)$ ; Then $B_n = 3.86D + 0.27R$ ; If $\left(1.0 < \frac{2D}{R} < 20\right)$ ; Then $B_n = 2.4D + 1.0R$	Digital modulation used to send 1 megabit per second by frequency shift keying with 2 signaling sates and 0.75 MHz peak deviation of the carrier. $R=1X10^6$ bits per second; $D=0.75X10^6$ Hz; $B_n=2.8$ MHz	2M80F1DBC

Necessary Bandwidth <sup>a</sup>			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Multilevel Frequency Shift Key- ing	$B_n = (R/\log_2 S) + 2DK$	Digital modulation to send 10 megabits per second by use of frequency shift key- ing with four signaling states and 2 MHz peak devi- ation of the main carrier $R=10X10^6$ bits per sec- ond; $D=2$ MHz; $K=1$ ; $S=4$ ; $B_n=9$ MHz	9M00F7DDT
Phase Shift Keying	B <sub>n</sub> =2RK/log <sub>2</sub> S	Digital modulation used to send 10 megabits per sec- ond by use of phase shift keying with 4 signaling states $R=10X10^6$ bits per second; $K=1$ ; $S=4$ ; $B_n=10$ MHz <sup>6</sup>	10M00G7DDT
Minimum Shift Keying	2-ary: $B_n = R(1.18)$ 4-ary $B_n = R(2.34)$	Digital modulation used to send 2 megabits per second using 2-ary minimum shift keying $B_n=2.36X10^6$ Hz=2.36 MHz	2M36G1DBN

a. See Table B for FM/FDM multiplying factors when "D" is not known.

b. See References g, h, and i for further details.

## IV. AMPLITUDE-MODULATED AND ANGLE-MODULATED

## **1. Quadrature Amplitude Modulation**

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Quadrature Amplitude Modulation (QAM)	B <sub>n</sub> =2R/log <sub>2</sub> S	64 QAM is used to send 135 Mbps has the same nec- essary bandwidth as 64-PSK used to send 135 Mbps; $R=135X10^6$ bps; $S=64$ ; $B_n=45$ MHz	45M00WXD

#### V. PULSE MODULATION

#### 1. Radar

See paragraph 3.1 of this Annex for specific instruction on calculating necessary bandwidth for RADARS.

#### 2. Composite Emissions

Necessary Bandwidth			
Description of Emission	Formula	Sample Calculation	Designation of Emission
Radio-relay system	$B_n = \frac{2K}{t}$ $K = 1.6$	Pulse position modulated by 36 voice channel baseband: pulse width at half ampli- tude=0.4 <i>us</i> ; $B_n=8X10^6$ Hz=8 MHz (Bandwidth independent of the number of voice chan- nels)	8M00M7EJT
Composite transmission digital modulation using DSB-AM (Microwave radio relay system)	B <sub>n</sub> =2RK/log <sub>2</sub> S	Digital modulation used to send 5 megabits per second by use of amplitude modu- lation of the main carrier with 4 signaling states $R=5X10^6$ bits per second; K=1; $S=4B_n=5 MHz$	5M00K7DD

#### TABLE B

### MULTIPLYING FACTORS FOR USE IN COMPUTING D, PEAK FREQUENCY DEVI-ATION, IN FM FREQUENCY DIVISION MULTIPLEX (FM/FDM) MULTI-CHANNEL EMISSIONS

For FM/FDM systems the necessary bandwidth is (for systems having no continuity pilot sub-carrier or having a continuity pilot sub-carrier whose frequency is not the highest modulating the main carrier):  $B_n = 2M + 2DK$ .

The value of D, or peak frequency deviation, in these formulas for  $B_n$  is calculated by multiplying the rms value of per-channel deviation by the appropriate "multiplying factor" shown below. In the case where a continuity pilot of frequency  $f_p$  exists above the maximum modulation frequency, M, the general formula becomes:

$$B_{\rm n} = 2f_{\rm p} + 2DK$$

In the case where the modulation index of the main carrier produced by the pilot is less than 0.25, and the rms frequency deviation of the main carrier produced by the pilot is less than or equal to 70 percent of the rms value of per-channel deviation, or in a radio system for television, the rms deviation of the main carrier due to the pilot does not exceed 3.55 percent of the peak deviation of the main carrier, the general formula becomes either:  $B_n=2_p$  or  $B_n=2M+2DK$  whichever is greater.

The selection of the values used to determine the multiplying factor are highly dependent upon the information transfer requirements placed upon the FM/FDM systems. Available technical information indicates that (depending on the number of channels) a value of "X" of -2, -5.6 or -19.6 should be appropriate for modern commercial telephone circuits where most of the channels are actual speech. In smaller or older FM/FDM systems and those where most of the circuits are used for data

transmission, "X" values of +2.6, -1.0 or -15 should be appropriate since typical commercial multichannel data circuits operate at power levels from -13 to -15 dBmO

Number telephone channels N <sub>c</sub>	Multiplying factors	Limits of X(P <sub>avg</sub> (dBm0)
3 <n<sub>c&lt;12</n<sub>	4.47 x antilog x $\frac{x}{20}$	Not applicable
	X = a value in dB specified by the equipment manufacturer or station lic- ensee, subject to NTIA approval	
$12 < N_c < 60$	3.76 antilog $\frac{(X+2\log_{10}N_c)}{20}$	X: -2 to +2.6
$60 < N_c < 240$	3.76 antilog $\frac{(X+4\log_{10}N_c)}{20}$	X: -5.6 to -1.0
N <sub>c</sub> >240	3.76 antilog $(X+10 \log_{10} N_c)$ 20	X: -19.6 to -15.0

Where  $N_c$  is the number of circuits in the multiplexed message load; 4.47 corresponds to a peak load factor of 13.0 dB, and 3.76 corresponds to a peak load factor of 11.5 dB.



FIGURE 3. Percent Power Inside vs. "K" for BPSK and QPSK

#### REFERENCES

a. Taub and Schilling (1971), "Principles of Communication Systems", McGraw-Hill Book Company, Chapter 4.

b. Filippi, C. (1983), "FM Spectral Modeling and FDM/FM Simulation Programs", NTIA Report 83-134, available from National Technical Information Service, Springfield, VA 22161.

c. Medhurst, R. G. and Plotkin, S. C. (1968), "Comments on FM Bandwidth as a Function of Distortion and Modulation Index", IEEE transaction on Communication Technology, p. 500 (June).

d. Bellamy, J. (1982), "Digital Telephony, John Wiley & Sons, Chapter 6.

e. Cohen, D. (1983), "Necessary Bandwidth and Spectral Properties of Digital Modulation", NTIA-TR 84-168, available from National Technical Information Service, Springfield, VA 22161.

f. Prabhu, V. K. (1981), "MSK and Offset QPSK Modulation with Bandlimiting Filters", IEEE Transactions on Aerospace and Electronic Systems, Volume AES-17, No. 1 (January).

g. K. L. McAdoo, "Speech Volumes on Bell System Message Circuits-1960 Survey", Bell System Technical Journal, Vol. 42, No. 5, September 1963. h. W. C. Ahern, F. P. Duffy, J. A. Maher, "Speech Signal Power in the Switched Message Network", Bell System Technical Journal, Vol. 57, No. 7, Part 2, September 1978, pp 2695-2726.

i. Federal Communications Commission (FCC), Report and Order, "In the Matter of

Amendment of Parts 2 and 21, of the Commissions' Rules Concerning Calculations of Necessary Bandwidth for Frequency Modulation Microwave Radio Relay Systems", General Docket No. 81-743, Rule Making 3625, adopted March 3, 1983.

#### **Authorized Bandwidth**

For purposes of this Manual, the term "authorized bandwidth" is defined as the necessary bandwidth (bandwidth required for the transmission and reception of intelligence) and does not include allowance for transmitter drift or doppler shift. See, in addition, Chapter 6 for the definitions of special terms including authorized bandwidth and mean power.

#### **Resolution Bandwidth**

Resolution bandwidth is the 3 dB bandwidth of the measurement system used, e.g., in power spectral density measurements. The appropriate resolution bandwidth of the measurement system varies depending on the modulation type and frequency band but should not be greater than the necessary bandwidth of the transmitter being measured.

#### Power (RR)

Power is designated as:

peak envelope power (PX or pX) mean power (PY or pY)

carrier power (PZ or pZ)

p denotes power expressed in watts

P denotes power in dB relative to a reference level

#### Logarithm

In this chapter,  $Log = Log_{10}$ 

#### 5.2 FREQUENCY TOLERANCES AND **UNWANTED EMISSIONS**

#### **5.2.1** Table of Frequency Tolerances

Frequency tolerance standards applicable to Federal stations are specified in Table 5.2.1. The table specifies standards for station types arranged within frequency bands.

Transmitter frequency tolerance is the maximum permissible departure from the assigned frequency by the center frequency of the frequency band occupied by an emission.

Receiver frequency tolerance is the maximum permissible departure of the center frequency of the IF passband from the desired center frequency of the IF passband.

In Table 5.2.1 the units for frequency tolerance are expressed in  $(\pm)$  parts per million (ppm) unless otherwise stated. For the purpose of this Manual, the " $\pm$ " symbol will always be implied. For example,  $\pm 10$  ppm will appear as 10 ppm.

The power shown for the various categories of stations is the peak envelope power for single sideband transmitters and the mean power for all other transmitters, unless otherwise indicated. (RR)

#### Table 5.2.1 Table of Frequency Tolerances

Frequency Band 9 kHz to 535 kHz	Frequency Tolerance
I. Fixed Stations	
A. 9 - 50 kHz	100
B. 50 - 535 kHz	50
II. Mobile Stations	
A. Aeronautical Stations	
1. Aeronautical	50
2. Aircraft	50
3. Survival craft	500
B. Land Mobile Stations	
1. Base (TIS) (530 kHz)	100 Hz
2. Land Mobile	20
3. Direct Printing telegraph and data.	10 Hz
C. Maritime Mobile Stations	
1. Coast	100
<ul> <li>2. Ship</li> <li>a. Direct printing telegraph and data.</li> <li>b. Other that above</li> </ul>	10 Hz 200
3. Ship Emergency Transmitters	500 (a)
4. Survival Craft	500

Г

Frequency Band 9 kHz to 535 kHz	Frequency Tolerance
III. Radiodetermination Stations	100
Frequency Band 535 kHz to 1605 kHz	Frequency Tolerance
I. Broadcasting Stations	10 Hz (b)
Frequency Band 1605 kHz to 4000 kHz	Frequency Tolerance
I. Fixed Stations	
A. All, except SSB	10
B. SSB radiotelephone	20 Hz
II. Mobile (Aeronautical, Land, Maritime	e) Stations
A. Aeronautical Mobile Stations	
1. Aeronautical a. pY < 200W b. pY > 200W c. SSB radiotelephone	20 10 10 Hz (c)
<ul><li>2. Aircraft</li><li>a. All except SSB</li><li>b. SSB radiotelephone</li></ul>	20 20 Hz (d)
B. Land Mobile Stations	
<ol> <li>Base         <ol> <li>pY &lt; 200W, except SSB</li> <li>pY &gt; 200W, except SSB</li> <li>SSB radiotelephone</li> </ol> </li> </ol>	20 (e) 10 20 Hz
<ul><li>2. Land Mobile</li><li>a. All except SSB</li><li>b. SSB radiotelephone</li></ul>	50 20 Hz
C. Maritime Mobile Stations	
1. Coast a. pY $\leq$ 200W, except c and d below b. pY $\geq$ 200W, except c and d	100 50
below c. SSB Radiotelephone d. Direct printing telegraph and data	20 Hz 10 Hz
<ul> <li>2. Ship <ul> <li>a. All except below</li> <li>b. SSB radiotelephone</li> <li>c. Direct printing telephony and data</li> </ul> </li> </ul>	40 (f) 40 Hz 40 Hz

Frequency Band 1605 kHz to 4000 kHz	Frequency Tolerance
<ol> <li>Survival Craft         <ol> <li>EPIRB</li> </ol> </li> </ol>	100
D. Radiodetermination Stations	
<ol> <li>Radionavigation         <ul> <li>a. pY ≤ 200W</li> <li>b. pY &gt; 200W</li> </ul> </li> </ol>	20 10
2. Radiolocation	10
E. Broadcasting Stations	10 Hz

Frequency Band 4 to 29.7 MHz	Frequency Tolerance
I. Fixed Stations	
A. pY $\leq$ 500W, except C and D below	20
B. pY > 500W, except C and D below	10
C. SSB/ISB Radiotelephone	20 Hz
D. Class F1B emissions	10 Hz
II. Mobile (Aeronautical, Land, Maritim	e) Stations
A. Aeronautical mobile stations	
<ol> <li>Aeronautical         <ol> <li>pY &lt; 500W, except SSB</li> <li>pY &gt; 500W, except SSB</li> <li>SSB Radiotelephone</li> </ol> </li> </ol>	30 10 10 Hz (c)
<ul><li>2. Aircraft</li><li>a. All except SSB</li><li>b. SSB Radiotelephone</li></ul>	30 20 Hz
B. Land mobile stations	
<ol> <li>Base         <ol> <li>pY &lt; 500W, except SSB</li> <li>pY &gt; 500W, except SSB</li> <li>SSB Radiotelephone</li> </ol> </li> </ol>	20 10 20 Hz
<ul><li>2. Land Mobile</li><li>a. All except SSB</li><li>b. SSB Radiotelephone</li></ul>	30 20 Hz
C. Maritime mobile stations	
<ol> <li>Coast         <ol> <li>SSB radiotelegraph</li> <li>Direct printing telegraph and data</li> <li>Other than above</li> </ol> </li> </ol>	20 Hz 10 Hz 20 Hz (g)

Frequency Band 4 to 29.7 MHz	Frequency Tolerance
2. Ship	
a. Class A1A emission.	10
b. Other than A1A emissions	
(1) SSB Radiotelephone	50 Hz
(2) Direct printing, telegraphy and	10 Hz
data	
(3) Other than above	50 Hz (h)
3. Survival craft	50
III. Broadcasting stations	2
IV. Space and earth stations	20

Frequency Band 29.7 to 108 MHz	Frequency Tolerance
I. Fixed stations	
A. 29.7-50 MHz, single-channel ana- log/digital FM/PM	5 (i)
B. Other than above	
1. pY_10W	20
2. $pY > 10W$	5
II. Mobile (Aeronautical, Land, Maritime	e) Stations
A. 29.7- 50 MHz, analog and digital FM/PM	
1. Land and mobile	5 (i)
2. Portables	20 (i)
B. Other than above	
1 pY <u>&lt;</u> 10W	20 (j)
2. $pY > 10W$	5
III. Aeronautical Radionavigation sta- tions (Marker beacons on 75 MHz)	50
IV. Broadcasting stations	
A. TV sound and vision	500 Hz (k)(l)
B. Other than TV	
1. pY <u>&lt;</u> 10 W	3000 Hz
2. $pY > 10 W$	2000 Hz
V. Space and earth stations	20

Frequency Band 108 to 470 MHz	Frequency Tolerance
I. Fixed stations.	
A. 108 - 406.1 MHz, all except below.	5
<ul> <li>B. 138 - 150.8 and 162 - 174 MHz, narrowband analog /digital FM/</li> <li>PM except C below</li> </ul>	1.5
<ul> <li>C. 162 - 174 MHz, low power and splinter channels</li> <li>1. pY &lt; 10W</li> <li>2. pY &gt; 10W</li> </ul>	5 2
D. 406-470 MHz	
<ol> <li>406.1-420 MHz         <ol> <li>Multi-Channel</li> <li>Analog/Digital FM/PM</li></ol></li></ol>	2.5 (m)(n) 2.5 (i) 1.0 (i)
<ul> <li>2. Other than above</li> <li>a. pY ≤ 10 W</li> <li>b. pY &gt; 10 W</li> </ul>	5 2.5
II. Mobile (Aeronautical, Land, Maritim	e) Stations
A. Aeronautical mobile stations	
1. Aeronautical	
a. Analog/digital FM/PM	
<ul><li>(1) 162-174 MHz</li><li>(a) Wideband</li><li>(b) Narrowband</li></ul>	5(i) 1.5(o)
<ul><li>(2) 406.1-420 MHz</li><li>(a) Wideband</li><li>(b) Narrowband</li></ul>	2.5 (i) 1.0 (o)
B. Other than above	20
2. Aircraft	
a. 156-174 and 406.1-420 MHz	5
<ul> <li>(1) 162-174 MHz analog/digital FM/PM</li> <li>(a) Wideband</li> <li>(b) Narrowband</li> <li>(2) 406.1-420 MHz analog/digital FM/PM</li> </ul>	5(i) 2.5 (o)
<ul><li>(a) Wideband</li><li>(b) Narrowband</li></ul>	5(i) 2.5 (o)
b. Other than above	20

Frequency Band 108 to 470 MHz	Frequency Tolerance
B. Land mobile stations	
1. Base	
a. 108 - 406.1 MHz, all except below	5
<ul> <li>b. 138-150.8 and 162 - 174 MHz analog/digital FM/PM</li> <li>(1) Wideband</li> <li>(2) Narrowband</li> </ul>	5(i) 2.5 (o)
c. 162 - 174 MHz, splinter channel (1) pY <u>&lt;</u> 10 W (2) pY > 10 W	5 2
d. 220 - 222 MHz, single-channel, narrowband	0.1
e. 406.1 - 470 MHz	
(1) 406.1 - 420 MHz analog/digi- tal	
(a) Wideband (b) Narrowband	2.5 (i) 1.0 (o)
<ul> <li>(2) Other than above</li> <li>(a) pY ≤ 10 W</li> <li>(b) pY &gt; 10 W</li> </ul>	5 2.5
2. Land Mobile	
a. 138 - 150.8 and 162 - 174 MHz, all except below	5 (j)
<ul> <li>b. 138 - 150.8 and 162 - 174 MHz, analog/digital FM/PM</li> <li>(1) Wideband</li> <li>(2) Narrowband</li> </ul>	5 (i) 2.5 (o)
<ul> <li>c. 162 - 174 MHz (splinter channels)</li> <li>(1) pY ≤ 10 W</li> <li>(2) pY &gt; 10 W</li> </ul>	5 2
d. 220 - 222 MHz (single channel, narrowband)	1.5 (p)
<ul> <li>e. 406.1- 420 MHz</li> <li>(1) Analog/digital FM/PM</li> <li>(2) Other than above</li> </ul>	5 (i) 2
f. Other than above	15
C. Maritime mobile stations	
1. Coast	

Frequency Band 108 to 470 MHz	Frequency Tolerance
a. 150.8 - 162.0125 MHz	
(1) FM (a) $pY < 3 W$ (b) $3 W \le pY \le 50 W$	100 (q) 50 (q)
<ul> <li>(2) Other than above</li> <li>(a) pY &lt; 3 W</li> <li>(b) 3 W ≤ pY &lt; 100 W</li> <li>(c) pY ≥ 100 W</li> </ul>	10 5 2.5
b. Outside of 150.8 - 162.0125 MHz	
<ul> <li>(1)162 - 174 MHz, analog/digital FM/PM</li> <li>(a) Wideband</li> <li>(b) Narrowband</li> </ul>	5 (i) 1.5 (o)
<ul> <li>(2) 406.1 - 420 MHz, analog/digital FM/PM</li> <li>(a) Wideband</li> <li>(b) Narrowband</li> </ul>	2.5 (i) 1.0 (o)
(2) Other than above	10
2. Ship	
a. 150.8 - 162.0125 MHz (FM, pY < 25 W)	100 (q)(r)
b. 156 - 162 MHz	10
<ul> <li>c. 162 - 174 MHz, analog/digital FM/PM</li> <li>(1) Wideband</li> <li>(2) Narrowband</li> </ul>	5 (i) 2.5 (o)
<ul> <li>d. 406.1 - 420 MHz, analog/digital FM/PM</li> <li>(1) Wideband</li> <li>(2) Narrowband</li> <li>(3) Other than above</li> </ul>	5 (i) 2 (o) 5
e. 450 - 470 MHz	5
f. Outside above bands	20 (r)
3. Survival craft	
a. 156 - 174 MHz	10 (r)
b. Other than above	20 (s)
III. Radiodetermination Stations	
A. Radionavigation stations	
1. Radar	50
2. Other than radar	20

Frequency Band 108 to 470 MHz	Frequency Tolerance
B. Radiolocation stations	
1. Radar	50 (t)
2. Other than radar	50
IV. Broadcasting Stations	
A. TV sound and vision	500 Hz (k)(o)
B. Other than TV	2000 Hz
V. Space and Earth Stations	20

Frequency Band 470 to 960 MHz	Frequency Tolerance
I. Fixed Stations	
A. Point-to-Multipoint (932 - 932.5, 941 - 941.5MHz)	1.5 (n)
B. Point-to-Point (932.5 - 935, 941.5 - 944 MHz)	2.5 (n)
C. Other than above	5
II. Mobile (Aeronautical, Land, Maritime) Stations	
A. Land (Aeronautical, Base, Coast)	5
B. Mobile (Aircraft, Land Mobile, Ship)	
1. pY <u>&lt;</u> 3 W	20
2. $pY > 3 W$	5
III. Radiolocation Stations	400
IV. Broadcasting Stations	
A. TV Broadcasting	500 Hz (k)(i)
B. TV Broadcasting Translators	200
V. Space and Earth Stations	20

Frequency Band 960 to 1215 MHz	Frequency Tolerance
I. Aeronautical Radionavigation Stations	
A. Aeronautical and Ship Stations	10
B. Aircraft	50

Frequency Band 960 to 1215 MHz	Frequency Tolerance
II. IFF/ATCRBS of similar type station	
A. Interrogators 1030 MHz	200 kHz
B. Transponders 1090 MHz	3 MHz

Frequency Band 1215 to 10500 MHz	Frequency Tolerance
I. Fixed Stations	
A. pY <u>&lt;</u> 100 W	
1. 1215 to 4000 MHz	30 (n)
2. 4 to 10500 MHz	50 (n)
B. pY > 100 W	10 (n)
II. Mobile (Aeronautical, Land, Maritime) Stations	
A. 1215 to 2450 MHz	20
B. 2450 to 4000 MHz	30
C. 4000 to 10500 MHz	50
III. Radiodetermination Stations	
A. 1215 to 2450 MHz	500
B. 2450 to 4000 MHz	800
C. 4000 to 10500 MHz	1250
IV. Space and Earth Stations	20

Frequency Band 10.5 to 30 GHz	Frequency Tolerance
I. Fixed Stations	
A. 21.2 - 23.6 GHz	300
B. 21.8 - 22 GHz and 23 -23.2 GHz	500 (u)
C. Other than above	50 (n)
II. Mobile (Aeronautical, Land, Maritime) Stations	100
III. Radiodetermination Stations	2500
IV. Space and Earth Stations	50

Frequency Band	Frequency
Greater than 30 GHz	Tolerance
I. Fixed Stations	75

Frequency Band Greater than 30 GHz	Frequency Tolerance
II. Mobile (Aeronautical, Land, Maritime) Stations	150
III. Radiodetermination Stations	5000
IV. Space and Earth Stations	75

#### **Notes For Frequency Tolerances**

(a) If the emergency transmitter is used as the reserve transmitter for the main transmitter, the tolerance for ship station transmitters applies.

(b) In the area covered by the North American Regional Broadcasting Agreement (NARBA), the tolerance of 20 Hz may continue to be applied.

(c) 20 Hz is applicable to other than Aeronautical Mobile (R) frequencies.

(d) The tolerance for aeronautical stations in the Aeronautical Mobile (R) service is 10 Hz.

(e) Travelers Information Stations (TIS) on 1630 kHz have a tolerance of 100 Hz.

(f) For A1A emissions the tolerance is 50 ppm.

(g) For A1A emissions the tolerance is 10 ppm.

(h) For ship station transmitters in the band 26.175-27.5 MHz, on board small craft, with a carrier power not exceeding 5 W operating in or near coastal waters and utilizing A3E or F3E and G3E emissions, the frequency tolerance is 40 ppm.

(i) This tolerance is based on emissions with an analog input and a necessary bandwidth of 16 kHz. Stations with digital inputs may require a different necessary bandwidth, but still must meet all other standards. It does not apply to military equipment used for tactical and/or training operations, FM wireless microphone systems whose pY < 0.1 watts, equipment on splinter channels, and fixed stations with multichannel emissions. Also, in the band 162-174 MHz, it does not apply to equipment operating on channels designated for low power systems as set forth in Sections 4.3.8 and 5.3.6, or NOAA Weather Radio Transmitters. The measurement method for the receiver frequency tolerance is contained in paragraph 2.1.2.E.1 of Annex M.

(j) 50 ppm applies to wildlife telemetry with mean power output less than 0.5 W.

(k) In the case of television stations of:

(1) 50 W (vision peak envelope power) or less in the band 29.7-100 MHz;

(2) 100 W (vision peak envelope power) or less in the band 100-965 MHz and which receive their input from other television stations or which serve small isolated communities. It may not, for operational reasons, be possible to maintain this tolerance. For such stations, this tolerance is 1000 Hz.

(l) For transmitters for system M(NTSC) the tolerance is 1000 Hz. However, for low power transmitters using this system note (m) applies.

(m) The receiver frequency tolerance shall be maintained within 10 ppm.

(n) See Annex M, paragraph 2.1.2.C.1.(a), for the measurement method of (1) multichannel equipment in the 406.1-420 MHz band, (2) point-to-point and point-to-multipoint equipment in the bands 932-935/941-944 MHz, or (3) point-to-point and transportable type equipment operating between 1710 MHz and 15.35 GHz (except for systems designed to use scatter techniques).

(o) This tolerance is for stations with emissions having a necessary bandwidth of 11 kHz or less. It does not apply to military equipment used for tactical and/or training operations, FM wireless microphone systems whose mean output power does not exceed 0.1 watt, equipment operating on channels designated for low power systems as set forth in Sections 4.3.8 and 5.3.6, and NOAA Weather Radio Transmitter.

(p) This standard is for narrowband operations with a necessary bandwidth of 4 kHz or less.

(q) The frequency tolerance standard is for maritime mobile stations using FM emissions in the band 150.8-162.0125 MHz with a necessary bandwidth of less than or equal to 16 kHz. See Annex M, paragraph 2.1.2.B, for the measurement method.

(r) Outside band 156-174 MHz, for transmitters used by on-board communications stations, a tolerance of 5 ppm shall apply.

(s) For transmitters used by on-board communications stations, a tolerance of 5 ppm applies.

(t) A frequency tolerance of 10 ppm applies to wind profiler radars operating on the frequency 449 MHz. (u) Applies to frequency pairs 21.825 GHz, 23.025 GHz; 21.875, 23.075 GHz; 21.925, 23.125GHz; and, 21.975 GHz, 23.175 GHz only.

# 5.2.2 Location of Standards for Levels of Unwanted Emissions

## 5.2.2.1 Location of Specific Standards

The location of levels of unwanted emission standards are provided in Table 5.2.2.1 below. The table specifies the section number for each standard by station type.

#### Table 5.2.2.1

Station Type: FIXED STATIONS	Location of Standards
Single Sideband and Independent Sideband Equipment (2-29.7 MHz)	§5.3.1
Multichannel (406.1-420 MHz) Point-to- point and point-to-multipoint (932-935/ 941-944 MHz) Point-to-point and transportable, except for systems using scatter techniques (1.71- 15.35 GHz)	§5.3.3
Analog or Digital FM/PM Wideband Operations (29.7-50, 162-174, and 406.1- 420 MHz)	§5.3.5.1
Analog or Digital FM/PM Narrowband Operations (138-150.8,162-174, and 406.1-420 MHz)	§5.3.5.2
Low Power Channels and Splinter Chan- nels (162-174 MHz and 406.1-420 MHz)	§5.3.6
Telemetry, Terrestrial (1435-1535, 2200- 2290 and 2310-2390 MHz)	§5.3.7
Analog Transmissions and Low Power Transmit (21.2-23.6 GHz)	§5.3.9
Other than above	§5.2.2.2

Station Type:	Location of
LAND and MOBILE STATIONS	Standards
Single Sideband and Independent Sideband Equipment (2-29.7 MHz)	§5.3.1

Station Type: LAND and MOBILE STATIONS	Location of Standards
Maritime Mobile Stations using FM (150.8-162.0125 MHz)	§5.3.2
Land Mobile, Single Channel Narrow- band Operations (220-222 MHz)	§5.3.4
Analog or Digital FM/PM Wideband Operations (29.7-50, 162-174, and 406.1-420 MHz)	§5.3.5.1
Analog or Digital FM/PM Narrowband Operations (138-150.8, 162-174 MHz and 406.1-420. MHz)	§5.3.5.2
Low Power Channels and Splinter Chan- nels (162-174 MHz and 406.1-420 MHz)	§5.3.6
Telemetry, Terrestrial (1435-1535, 2200-2290 and 2310-2390 MHz)	§5.3.7
Other than above	§5.2.2.2

Station Type: RADIODETERMINATION STATIONS	Location of Standards
Primary radars including spacebased radars on a case-by-case bases (100 MHz to 40 GHz)	Part 5.5
Other than above	§5.2.2.2

Station Type:	Location of
BROADCASTING STATIONS	Standards
All bands	§5.2.2.2

Station Type: EARTH and SPACE STATIONS (excluding spacebased radars)	Location of Standards
Below 470 MHz	§5.2.2.2
470 MHz and above	Part 5.6

**XR--**Experimental Research Station: An experimental station used in basic studies concerning scientific investigation looking toward the improvement of the art of radiocommunications.

**XT--**Experimental Testing Station: An experimental station used for the evaluation or testing of electronics equipment or systems, including site selection and transmission path surveys, which have been developed for operational use.

# 6.1.4 Table of Services, Station Classes, and Stations

Table A is used to determine the proper Station Class (STC) symbol (see Section 9.8.2, paragraph 15a through 15c) to be used versus the Service in which the transmitting station will operate. Frequency bands are allocated to Service(s) based upon the U.S. Government Table of Frequency Allocations (see Section 4.1.3).

TABLE A           Table of Services, Station Classes, and Stations		
Service	Station Class	Station
1. Amateur	None	Amateur
2. Broadcasting	BC BT	Broadcasting (sound) Broadcasting (television)
3. Broadcasting-Satellite	EB EV	Space (sound) Space (television)
4. Earth Exploration-Satellite Meteorological-Satellite	EW TW EM TM	Space Earth Space Earth
5. Fixed Aeronautical Fixed	FX FXD FXE FXH AX	Fixed Telecommand Fixed Telemetering Fixed Hydrologic and Meteorological Fixed Aeronautical Fixed
6. Fixed-Satellite	EC TC VA TB TI TY	Space Earth Land Earth Earth Coast Earth Base Earth
7. Inter-Satellite	ES	Space
8. Meteorological Aids	SA SAR SM SMB SMD SMRG	Meteorological Aids Mobile Station Radiosonde Meteorological Aids Base Station Radar Beacon Precipitation Gage Meteorological Radar Radiosonde Ground

TABLE A           Table of Services, Station Classes, and Stations		
Service	Station Class	Station
9. Mobile	FL	Land
	FLD	Telecommand Land
	FLE	Telemetering Land
	FLEA	Aeronautical Telemetering Land
	FLEB	Flight Telemetering Land
	FLEC	Surface Telemetering Land
	FLH	Hydrologic and Meteorological Land
	FLU	Aeronautical Utility Land
	MO	Mobile
	MOB	Radio Beacon Mobile
	MOD	Telecommand Mobile
	MOE	Telemetering Mobile
	MOEA	Aeronautical Telemetering Mobile
	MOEB	Flight Telemetering Mobile
	MOEC	Surface Telemetering Mobile
	MOH	Hydrologic and Meteorological Mobile
	МОР	Portable Mobile
	MOU	Aeronautical Utility Mobile
Aeronautical Mobile	FA	Aeronautical
	FAB	Aeronautical Broadcast
	FAC	Airdrome Control
	FAD	Telecommand Aeronautical
	FAT	Flight Test
	MA	Aircraft
	MAD	Telecommand Aircraft
	МАР	Portable Aircraft
Aeronautical Mobile (OR)	FG	Aeronautical
Aeronautical Mobile (R)	FD	Aeronautical
Land Mobile	FB	Base
	FBD	Telecommand Base
	ML	Land Mobile
	MLD	Telecommand Land Mobile
	MLP	Portable Land Mobile
Maritime Mobile	FC	Coast
· ···· ··· ····	FCB	Marine Broadcast
	FCD	Telecommand Coast
	MS	Ship/Telecommand Ship
	MSD	Portable Ship
	MSP	Oceanographic Data
	OD	Oceanographic Data Interrogating
	OE	
10. Mobile-Satellite	UA	Mobile Earth
	TE	Satellite EPIRB
	EI	Space
	VA	Land Earth
Aeronautical Mobile-Satellite	EI	Space
	TB	Earth
	TJ	Aircraft Earth
	-	

TABLE A           Table of Services, Station Classes, and Stations		
Service	Station Class	Station
Land Mobile-Satellite	EU TU TY	Space Land Mobile Earth Base Earth
Maritime Mobile-Satellite	EG TG TI	Space Ship Earth Coast Earth
11. Radio Astronomy	RA	Radio Astronomy
12. Radiodetermination	None RG	Radiodetermination Radio Direction-Finding
Radiolocation	LR MR MRP	Land Mobile Portable
Radionavigation	NR RNL RN	Mobile LORAN Land
Aeronautical Radionavigation	AL ALA ALB ALC ALG ALL ALO ALR ALS ALTM ALTO AM AMA	Land Marker Beacon Radio Beacon Radar Beacon (Racon) Glide Path (Slope) Localizer /Omnidirectional Range Radio Range Surveillance Radar Land Test (Maintenance) Land Test (Operational) Mobile Altimeter
Maritime Radionavigation	NL NLC NLM	Land Radar Beacon (Racon) Marine Radio Beacon
13. Radiodetermination-Satellite	EF TF TL	Space Earth Mobile Earth
Radionavigation-Satellite	EN TN UM	Space Fixed Earth Mobile Earth
Aeronautical Radionavigation-Satellite	EO TO TZ	Space Mobile Earth Earth
Maritime Radionavigation-Satellite	EQ TQ TX	Space Mobile Earth Earth
14. Space Operation	ET TT	Space Earth

TABLE A           Table of Services, Station Classes, and Stations		
Service	Station Class	Station
15. Space Research	EH TH	Space Earth
16. Standard Frequency and Time Signal	SS	Standard Frequency and Time Signal
17. Standard Frequency and Time Signal-Sat- ellite	EE	Space
18. No Specific Service	DGP ED EK ER SN SP TD TK TR XC XD XC XD XE XM XR XR	Differential-Global-Positioning-System Space Telecommand Space Space Tracking Space Space Telemetering Space Sounder Network Sounder Prediction Space Telecommand Earth Space Telecommand Earth Space Telemetering Earth Experimental Contract Developmental Experimental Developmental Experimental Developmental Experimental Export Experimental Research /Experimental Testing

### 6.2 FREQUENCY NOMENCLATURE

The terms "allocation," "allotment," and "assignment" are used to describe the distribution of frequencies or bands of frequencies. Allocations are made to radio services, for example, the fixed service, the aeronautical mobile service, and the space research service. Allotments are made to areas or countries and, within the U.S. Government, to specific uses. Assignments are instruments of authorization of discrete frequencies to specific radio stations.

In the application of the provisions of this Manual, letters or numbers shall not be used to designate specific bands of frequencies, e.g., S-band, Xband, Ku-band. Such designations create confusion, because the band limits vary from one designator system or user group to another. This can be avoided by using the international system of citing in Hertz the actual numerical limits of specific frequency bands. Further, broader portions of the spectrum can be described by the following internationally adopted terms:

Frequency Subdivision	Frequency Range
VLF (very low)	3 kHz to 30 kHz
LF (low)	30 kHz to 300 kHz
MF (medium)	300 kHz to 3000 kHz
HF (high)	3 MHz to 30 MHz
VHF (very high)	30 MHz to 300 MHz
UHF (ultra high)	300 MHz to 3000 MHz
SHF (super high)	3 GHz to 30 GHz
EHF (extremely high)	30 GHz to 300 GHz 300 GHz to 3000 GHz

```
Hz=hertz=cycles per second
k=kilo (10^3)
M=Mega (10^6)
G=giga (10^9)
T=tera (10^{12})
```

4. *Estimated Termination Date* (where applicable).

5. *Estimated Initial Cost of the System:* This item is for information to show the general size and complexity of the system. It is not intended to be a determining factor in system reviews.

6. *Target Date:* Submit dates on which spectrum-related decisions must be made relative to system planning, development, procurement, and employment.

7. System Relationship and Essentiality: Submit for all stages a statement of the relationship between the proposed system and the function or operation it is intended to support. Include a brief statement of the essentiality to the supported function or operation.

8. *Replacement Information:* Identify the existing system(s) and associated frequency assignments to be replaced by the proposed system, where applicable.

9. Out-of-Band Operations: Submit a justification for any telecommunication system or subsystem that does not operate in accordance with the applicable Tables of Frequency Allocations as required by Section 10.0.1, and details of how operations on a noninterference, unprotected basis are feasible. This information is required for systems or subsystems that must operate out-of-band in the United States, its possessions, or in space.

10. War Emergency Function: A statement as to whether the proposed system, if it becomes operational, will support a war emergency function.

### 10.7.2 Attachment 1.7.2

Attachment 1 to the cover letter shall provide line diagram(s) showing the links, direction of transmission, and frequency band(s).

## 10.7.3 Attachment 2 -- For Space Systems

#### 1. Stage 1 and 2 Requirements:

a. Satellite orbital characteristics (longitude for geostationary satellites, and apogee, perigee, and inclination for non-geostationary satellites).

b. Satellite transmitter maximum spectral power density for each transponder.

c. Earth station locations (city, state, and coordinates) within the US&P, and frequencies or frequency bands used at each.

2. Stage 3 Requirements -- For Each Earth Station Transmitter and Receiver Site:

a. Frequencies or frequency bands and satellites accessed.

b. Coordinates.

c. Emission designator for each frequency or frequency band.

d. Maximum spectral power density and output power for each frequency or frequency band.

e. Lowest equivalent satellite link noise temperature and associated value of transmission gain for each frequency or frequency band (geostationary satellites with simple frequency-changing transponders only).

f. Antenna gain and beamwidth.

g. Minimum elevation angle of antenna main beam.

h. Range of azimuth angles.

i. Lowest total receiver noise temperature (when e. is not appropriate).

3. Stage 3 Requirements -- For Each Space Station Transmitter and Receiver:

a. Frequency or frequency bands and cooperating earth stations.

b. Satellite orbital information.

c. Emission designator for each frequency or frequency band.

d. Peak power and spectral power density for each frequency or frequency band for transmitters.

e. Receiver noise temperature.

f. Transmitter antenna pattern (only if PFD limits are exceeded).

4. *Stage 4 Requirements:* In addition to satisfying all Stage 3 requirements, the following data items are required for each earth station:

a. Horizon elevation angle diagram.

b. Antenna altitude above ground.

5. The format for providing these data is left to the discretion of each agency. However, for unclassified space systems which have not been waived from the requirements of international registration as described in Part 3.3, similar information must be prepared in specific formats and submitted to the SSS in accordance with instructions in Part 3.3. The data required by the SSS to satisfy the specifications in Appendix 4 of the ITU Radio Regulations for:

a. Advance Publication shall be submitted at the same time as the Stage 2 system review request,

and may be used in lieu of the Attachment 2 data for Stage 1 and 2 system review requests; and,

b. Coordination and Notification data shall be submitted at the same time as Stage 3 system review requests, and may be used in lieu of the Attachment 2 data for Stage 3 and 4 system review requests. If any of the frequency bands proposed for the satellite network or system are subject to coordination under the ITU Radio Regulations, the simplified advance publication information for those bands, as set forth in Sub-Section IB of Article 9 of the Radio Regulations, cannot be used in lieu of Attachment 2 data. In those cases where the simplified advance publication information is required by the ITU, the simplified information as well as the Attachment 2 data must be submitted.

# 10.7.4 Attachment 2 -- For Terrestrial Systems (all stages)

The following information may be included in the line diagram submitted as Attachment 1 above, if desired:

1. Station class(es),

2. Number of units (for mobile systems),

3. Station locations and/or areas of operation, as appropriate (geographical coordinates required for Stages 2, 3, and 4),

4. Frequency requirements (i.e., band(s) or discrete frequencies required, bandwidth and emission designators, and netting information, where appropriate),

5. Proposed date of activation.

#### 10.7.5 Attachment 3 -- Related Analysis Data

For all stages, submit reports of any previous EMC studies, predictions, analyses, and prototype EMC testing that are relevant to the assessment of the system under review, or references thereto if previously provided to the IRAC/SPS, including references to previous system reviews of the same system or its predecessors.

#### 10.7.6 Attachment 4 -- Equipment Characteristics

1. Submit completed forms NTIA-33, NTIA-34, and NTIA-35 for each equipment (transmitter, receiver, antenna) intended for use in the

system under review. All applicable data items shall be completed for Stages 3 and 4 (estimated values or ranges of values may be submitted for Stage 3 in the absence of other available data). For Stages 1 and 2, provide actual equipment data, or in the absence of such data, estimated data and ranges of values shall be stated on the forms sufficient to support a realistic preliminary assessment of frequency availability and EMC characteristics.

2. NTIA forms need not be submitted for equipment whose required characteristics have been previously provided to the SPS or which are contained in the Equipment Characteristics File. In such cases, indicate in Attachment 4 the Government nomenclature or manufacturer's model number of such equipment.

3. Instructions for the completion of the NTIA forms are contained in Sections 10.7.7 through 10.7.9 below.

#### 10.7.7 Instructions for Completing the Transmitter Characteristics Form (NTIA-33)

1. Security Classification: This form will be classified in accordance with appropriate agency security directions. Downgrading instructions will be indicated. The items or relationship of items which make this completed form classified will be stated in the remarks section (e.g., "The association of the frequency range and the equipment nomenclature are classified \_\_\_\_\_"). Alternatively the classification may be indicated by a (U), (C), or (S) alongside each item as appropriate.

2. *Item 1:* Enter the Government alphanumeric equipment designation. Use the official designation as it appears or will appear on the nameplate of the transmitter (e.g., T128). If a Government nomenclature has not been designated, enter the manufacturer's model number (e.g., MTT 502). If neither a Government nomenclature nor a manufacturer's model number has been designated, enter a short descriptive title (e.g., ATS-6 Telemetry Transmitter).

3. *Item 1a:* Enter the manufacturer's name. In those cases where a Government nomenclature has not been entered in Item 1, this item is mandatory.

4. *Item 2:* Enter the system nomenclature. The system nomenclature is defined as that nomenclature which has been assigned to a combination of equipment (e.g., AN/GRC-27).

## **EMISSION DESIGNATOR**

The EMISSION DESIGNATOR consists of the *necessary bandwidth* and *emission classification symbols*.

- a. The amount of radio spectrum which may be occupied by an authorized frequency assignment does not extend beyond:
  - (1) the ASSIGNED FREQUENCY BAND which is determined by:
    - a. subtracting from the specific operating frequency ½ the *necessary bandwidth* plus the absolute value of the frequency tolerance and,
    - b. adding to the *specific operating frequency* ½ the *necessary bandwidth* plus the absolute value of the frequency tolerance; or
  - (2) the lower and upper limits shown for FREQUENCY BAND ASSIGNMENTS.
- b. The *authorized use* is determined by the designated:
  - (1) type of modulation:
  - (2) nature of the signal(s) modulating the main carrier; and
  - (3) type of information to be transmitted.
- c. An EMISSION DESIGNATOR must be entered for each type of transmission such as CW or tone modulated telegraphy.
- d. The EMISSION DESIGNATOR consists of the following:
  - (1) Necessary bandwidth
    - a. See Necessary Bandwidth for methods to calculate it.
    - b. Doppler Shift shall not be included in the frequency tolerance or necessary bandwidth.

## (2) Emission classification symbols:

- a. First symbol Indicates the type of modulation of the main carrier.
- b. Second symbol Indicates the nature of signal(s) modulating the main carrier.
- c. Third symbol Indicates the type of information to be transmitted.
- d. Fourth symbol Indicates the details of the signal (Optional recommend use when applicable).
- e. Fifth symbol Indicates the nature of multiplexing (Optional recommend use when applicable).
- e. The EMISSION DESIGNATOR shall be entered according to the following rules:
  - (1) Necessary Bandwidth. To express the necessary bandwidth portion of the emission designator, enter a maximum of five numerals and one letter. The letter occupies the position of the decimal point and represents the unit of bandwidth as follows:
    - H for Hertz K for kilohertz M for Megahertz G for Gigahertz

Fractional bandwidths may be expressed to a maximum of two decimal places following the letter. The first character of the *necessary bandwidth* shall always be greater than zero unless the *necessary bandwidth* is less than 1 Hertz in which case the first character shall be the letter H. Express the necessary bandwidths according to the following.

a. *Necessary bandwidths* between .01 and 999.99 Hz shall be expressed in Hertz using the letter H in lieu of the decimal; e.g., 15H is 15 Hz of bandwidth; 15H01 is 15.01 Hz of bandwidth.

- b. *Necessary bandwidths* between 1.00 and 999.99 kHz shall be expressed in kilohertz using the letter K in lieu of the decimal; e.g., 2K is 2 kHz bandwidth; 2K85 is 2.85 kHz of bandwidth.
- c. *Necessary bandwidths* between 1.00 and 999.99 MHz shall be expressed in Megahertz using the letter M in lieu of the decimal; e.g., 6M is 6 MHz of bandwidth; 6M25 is 6.25 MHz of bandwidth.
- *d. Necessary bandwidths* between 1.00 and 999.99 GHz shall be expressed in Gigahertz using the letter G in lieu of the decimal; e.g., 10G is 10 GHz of bandwidth; 10G05 is 10.05 GHz of bandwidth.
- (2) Emission Classification Symbols. Enter the basic emission designator of three symbols, and if desired, two optional additional symbols, as derived from the following;

Symbol	Type of Emission
UNMODULATED	
N	Emission of an unpopulated carrier
AMPLITUDE-MODULATED	
<i>Emission in which the main carrier is amplitude-modulated (including cases where sub-carriers are angle-modulated):</i>	
A	Double-sideband
В	Independent sidebands
С	Vestigial sideband
Н	Single-sideband, full carrier
J	Single-sideband, suppressed carrier
R	Single-sideband, reduced or variable level carrier

a. First symbol – designates the type of modulation of the main carrier:

Symbol	Type of Emission
	ANGLE-MODULATED
Emission in which the	carrier is angle-modulated <u>:</u>
F	Frequency modulation
G	Phase modulation
AMPLITUDE-MODULATED AND ANGLE-MODULATED	
D	Emission in which the main carrier is amplitude- modulated and angle-modulated either simultaneously or in a preestablished sequence.
	PULSE
Emission of pulses: (Emissions, where the main carrier is directly modulated by a signal which has been coded into quantized form (e.g., pulse code modulation), shall be designated either an emission in which the main carrier is amplitude-modulated, or an emission in which the main carrier is angle-modulated).	
Р	Sequence of unmodulated pulses.
A sequence of pulses:	
К	Modulated in amplitude
L	Modulated in width or duration
М	Modulated in position or phase
Q	Carrier is angle-modulated during the period of the pulse
V	A combination of the foregoing or produced by other means
COMBINATION	
W	Cases not covered above, in which an emission consists of the main carrier modulated, either simultaneously or in a combination of two or more of the following modes; amplitude, angle, pulse.
X	Cases not otherwise covered.

 b. Second symbol – designates the nature of signal(s) modulating the main carrier:

Symbol	Type of Emission
0	No modulating signal.
1	A single channel* containing quantized or digital signals without the use of a modulating subcarrier. (This excludes time-division multiplex.)
2	A single channel* containing a quantized or a digital signal with the use of modulating subcarrier.
3	A single channel* containing an analogue signal.
7	Two or more channel* containing quantized or digital signals.
8	Two or more channels* containing analogue signals.
9	A composite system with one or more channels* containing quantized or digital signals, together with one or more channels containing analogue signals.
Х	Cases not otherwise covered.
*In this context, the word "Channel(s)" refers to the radio frequency (RF)	

channel.

c. Third symbol – designates the type of information to be transmitted. (In this context the word "information" does not include information of a constant, unvarying nature such as provided by standard frequency emissions, continuous wave and pulse radars, etc.)

Symbol	Type of Emission
N	No information transmitted.
A	Telegraphy – for aural reception.
В	Telegraphy _ for automatic reception.
С	Facsimile.
D	Data transmission, telemetry, telecommand; (the symbol D indicated that data, telemetry, or telecommand information is being transmitted individually or, that any combination of the three are being transmitted simultaneously. If any combination is being transmitted simultaneously, one of the multichannel symbols, 7, 8 or 9, must be used for the second symbol.
E	Telephony (including sound broadcasting).
F	Television (video).
W	Combination of the above. (Use only for multi-channel systems having the capability of transmitting all information simultaneously.)
X	Cases not otherwise covered.

# d. Fourth symbol – designates the details of signal(s):

<b>A</b>	
Symbol	Type of Emission
A	Two-condition code with elements of differing numbers and/or durations.
В	Two-condition code with elements of the same number and duration without error-correction.
С	Two-condition code with elements of the same number and duration with error-correction.
D	Four-condition code in which each condition represents a signal element (or one or more bits).
E	Multi-condition code in which each condition represents a signal element (or one or more bits)

F	Multi-condition code in which each condition or combination of conditions represents a character.
G	Sound of broadcasting quality (monophonic).
Н	Sound of broadcasting quality (stereophonic or quadraphonic).
J	Sound of commercial quality (excluding categories defined for symbols K and L below).
К	Sound of commercial quality with the use of frequency inversion or band-splitting.
L	Sound of commercial quality with separate frequency-modulated signals to control the level of demodulated signal.
М	Monochrome.
N	Color.
W	Combination of the above.
Х	Cases not otherwise covered.

e. Fifth symbol – designates the nature of multiplexing:

Symbol	Type of Emission
N	None.
C	Code-division multiplex.
	(This includes bandwidth expansion techniques.)
F	Frequency-division multiplex.
Т	Time-division multiplex.
W	Combination of frequency-division multiplex and time-division multiplex.
Х	Other types of multiplexing.

# Examples:

Designator	Type of Emission
NON	Continuous wave
1K24F1B	1.24 kHz necessary bandwidth for frequency modulated single channel telegraphy.
16KF3EJN	16 kHz necessary bandwidth for commercial telephony.